This is a reproduction of a library book that was digitized by Google as part of an ongoing effort to preserve the information in books and make it universally accessible.



https://books.google.com



CODE OF FEDERAL REGULATIONS



TITLE 47
Parts 0 to 19
Revised as of January 1, 1970

CONTAINING A CODIFICATION OF DOCUMENTS OF GENERAL APPLICABILITY AND
FUTURE EFFECT AS OF JANUARY 1, 1970

With Ancillaries

Published by the Office of the Federal Register, National Archives and Records Service
General Services Administration, as a Special Edition of the Federal Register
Pursuant to Section 11 of the Federal Register Act as Amended



As of January 1, 1970

Title 47, Parts 0 to 19

Revised as of January 1, 1969

Replaced by This Volume

U.S. GOVERNMENT PRINTING OFFICE WASHINGTON: 1970

For sale by the Superintendent of Documents, U.S. Government Printing Office Washington, D.C. 20402 - Price \$1.75 (paper cover)

Table of Contents

								Page
Explanation	•	•	•	•		•	•	
√Title 47:								
Chapter I—Federal Communications Commission	•	•	•	•	•	•	•	3
Tru Jin a Atala								
Finding Aids: Table of CFR Titles and Chapters	•				•		•	393
								~
Alphabetical List of CFR Subtitles and Chapters	•	•	•	•	•	•	•	403
Redesignation Table		•	•	•	•	•	•	409
List of Sections Affected						,		421
Tist of pections wheelen	•	•	•	•		•		741

Explanation

This book contains the rules and regulations which constitute Parts 0-19 of Title 47, revised as of January 1, 1970. It replaces the volume entitled "Title 47—Telecommunication, Parts 0 to 19, Revised as of January 1, 1969," and becomes an integral part of the Code of Federal Regulations.

The text in this revision is derived from the latest text of the rules and regulations, general and permanent in nature, duly promulgated in the Federal Register on or before December 31, 1969. Source materials from which the text is derived are cited with the text and should be consulted to determine the effective date of any given provision. All dates appearing in the source citations are dates of publication in the Federal Register and should not be construed as effective dates.

Current regulatory material appearing in the daily issues of the Federal Register follows the numbering system used herein and serves as a daily supplement hereto. This book contains the following finding aids: a table of CFR titles and chapters; an alphabetical list of CFR subtitles and chapters; redesignation table indicating former part numbers of regulations codified in this title; and a list of sections affected which furnishes Federal Register citations to all changes in this book beginning January 1, 1964, including changes which are no longer in effect.

A compilation of the list of sections affected for the entire Code of Federal Regulations from January 1, 1949, through December 31, 1963, is published in a separate volume entitled "List of Sections Affected, 1949–1963."

A General Explanation of the Code of Federal Regulations, a list of current CFR volumes, and a list of superseded CFR volumes appear in the volume containing Title 1 and in the General Index volume. A list of titles, subtitles, chapters, subchapters, and parts of the entire Code of Federal Regulations is also published in the General Index volume.

This volume is published pursuant to Part 30 of the regulations of the Administrative Committee of the Federal Register (34 F.R. 19106; 1 CFR Part 30), under the authority contained in section 11 of the Federal Register Act, as amended (44 U.S.C. 1510). The contents of the Federal Register and the Code of Federal Regulations are by law prima facie evidence of the text of the original documents and are required to be judicially noticed (44 U.S.C. 1507, 1510). The preferred citation of the Code is "CFR". Thus the citation "47 CFR 0.1" refers to section 0.1 of this Title.

DAVID C. EBERHART

January 2, 1970.

Digitized by Google

Cite this Code CFR thus: 47 CFR 0.1

Title 47—Telecommunication

(This book contains Parts 0 to 19)

CHAPTER I—FEDERAL COMMUNICATIONS COMMISSION

(Parts 0 to 19)

SUBCHAPTER A-GENERAL

Part	
0	Commission organization.
1	Practice and procedure.
2	Frequency allocations and radio treaty matters; general rules and regulations
5	Experimental radio services (other than broadcast).
13	Commercial radio operators.
15	Radio frequency devices.
17	Construction, marking and lighting of antenna structures.
18	Industrial, scientific, and medical equipment.
19	Employee responsibilities and conduct.

Cross References:

Excise taxes on radio and television sets: Internal Revenue, 26 CFR Part 48. Excise taxes on communications services and facilities: Internal Revenue, 26 CFR Part 49.

Tax on gain from sale or exchange of radio broadcasting stations to effectuate policies of FCC: Internal Revenue, 26 CFR 1.1071.

Federal financial assistance for noncommercial education television broadcast facilities:
Department of Health, Education, and Welfare, General Administration, Public Welfare: 45 CFR, Part 60.

SUPPLEMENTAL PUBLICATIONS:

Annual Reports of the Federal Communications Commission to Congress.

Federal Communications Commission Reports of Orders and Decisions.

Communications Act of 1934 (with amendments and index thereto), revised to Sept. 13, 1960; Packet No. 1, revised pages, Sept. 1960 to Dec. 1961; Packet No. 2, revised pages, Dec. 1961 to Oct. 1962.

Statistics of Communications Common Carriers, 1957, 1959, 1960, and 1962.

MISCELLANEOUS PUBLICATIONS: Study Guide and Reference Material for Commercial Radio Operator Examinations, May 1955 edition. Figure M-3, Estimated AM Ground Conductivity of the United States (set of two maps).



Title 47—Chapter I

SUBCHAPTER A-GENERAL

	PART 0—COMMISSION	_	COMMON CARRIER BUREAU
	ORGANIZATION	Sec.	
		0.91	Functions of the Bureau.
	Subpart A—Organization	0.9 2 0.98	Units in the Bureau. Field Offices.
	GENERAL	0.50	
Sec.			FIELD ENGINEERING BUREAU
0.1	The Commission.	0.111	Functions of the Bureau.
0.3	The Chairman.	0.112	Units in the Bureau.
0.4	Standing committees of Commis-	0.113	
0.5	sioners.	0.114	
0.5	General description of Commission	0.115	Field Offices Division.
	organization and operations.	0.116	Monitoring Systems Division.
	OFFICE OF EXECUTIVE DIRECTOR	0.121	Location of field offices and monitor-
0.11	Functions of the Office.		ing stations.
0.12	Units in the Office.	SAFET	Y AND SPECIAL RADIO SERVICES BUREAU
	OFFICE OF CHIEF ENGINEER	0.131	
		0.131	Functions of the Bureau. Units in the Bureau.
0.31	Functions of the Office.	0.202	
0.32	Units in the Office.		OFFICE OF HEARING EXAMINERS
0.33	Immediate Office of the Chief Engi-	0.151	Functions of the Office.
0.34	neer. Research Division.	0.152	
0.85	Technical Division.		Examiner.
0.36	Laboratory Division.		REVIEW BOARD
0.37	Frequency Allocation and Treaty Di-		
	vision.	0.161	Functions of the Board.
	OFFICE OF GENERAL COUNSEL		OFFICE OF OPINIONS AND REVIEW
			Functions of the Office.
0.41	Functions of the Office.		
0.42 0.43	Units in the Office.	Ι	DEFENSE AND EMERGENCY PREPAREDNESS
0.43	Immediate Office of the General Counsel.		Functions
0.44	Litigation Division.	0.181	The Defense Commissioner.
0.45	Legislation Division.	0.182	Executive Director.
0.46	Administrative Law and Treaties Di-	0.183	Office of Emergency Communications.
	vision.	0.184	Regional Liaison Officers of the Office
0.47	Enforcement and Defense Division.		of Executive Director, Emergency
	OFFICE OF THE SECRETARY		Communications Division.
A E 1		0.185	Responsibilities of the bureaus and
0.51	Functions of the Office.		staff offices.
0.52 0.53	Units in the Office. Dockets Division.	0.186	Emergency Relocation Board.
0.54	Minute and Rules Division.		Submert B. Dolometians of Authority
0.55	Library Division.		Subpart B—Delegations of Authority
0.56	Technical Assistance Division.		GENERAL
	OFFICE OF INFORMATION	0.201	General provisions.
		0.203	Authority of person, panel, or board
0.61	Functions of the Office.		to which functions are delegated.
	BROADCAST BURRAU	0.204	The exercise of delegated authority.
0.71	Functions of the Bureau.		COMMISSIONERS
0.72	Units in the Bureau.		
0.73	Office of the Bureau Chief.	0.211	Chairman.
0.74	Broadcast Facilities Division.	0.212	
0.75	Renewal and Transfer Division.	0.214	
0.76	Hearing Division.	0.215	Telephone Committee.
0.77	Rules and Standards Division.	0.216	•
0.78	License Division.		Committee.
0.79	Research and Education Division.	0.218	Authority of, and delegated to, an in-
0.80	Office of Network Study.		dividual Commissioner or Commis-
0.81	Complaints and Compliance Division.		sioners.
	CATV TASK FORCE		EXECUTIVE DIRECTOR
0.85	Functions of the Task Force.	0.231	
J. J	a amounding of min 1888 Lougs.	U.201	Authority delegated.



Sec.	CHIEF ENGINEER	Сн	ief, Safety and Special Radio Services
0.241	Authority delegated to the Chief En-	Sec.	BUREAU
	gineer.	0.331	Authority delegated.
0.243	Authority delegated to the Chief En-	0.332	Additional authority delegated.
	gineer upon securing concurrence of the General Counsel.	0.333	Authority delegated jointly to the Chiefs of Common Carrier and
0.247			Safety and Special Radio Services
	GENERAL COUNSEL	0.337	Bureaus. Record of actions taken .
0.251	Authority delegated.	0.001	HEARING EXAMINERS
	SECRETARY	0.341	
0.261	Authority delegated.	0.347	Authority of hearing examiner. Record of actions taken.
	CHIEF, BROADCAST BUREAU		CHIEF HEARING EXAMINER
0.281	Authority delegated.	0.351	Authority delegated.
0.287	Record of actions taken.		REVIEW BOARD
	CHIEF, CATV TASK FORCE	0.361	General authority.
0.289	Authority delegated.	0.365	Authority delegated to the Review
	CHIEF, COMMON CARRIER BUREAU		Board on a regular basis.
0.291	Authority concerning radio matters.	Cı	hief, Office of Opinions and Review
0.292	Additional authority concerning radio	0.371	Authority delegated.
0.293	matters.	Dı	EFENSE AND EMERGENCY PREPAREDNESS
0.293	Authority concerning position of officer.		DELEGATIONS
0.294	Authority concerning section 214 of the Act.	0.381 0.383	Defense Commissioner.
0.295	Authority concerning section 220 of	0.385	Emergency Relocation Board. FCC representatives assigned to Re-
	the Act.		gional Preparedness Committees of
0.296	Authority concerning section 221(a) of the Act.		the Office of Emergency Planning
0.297	Authority concerning tariff regula-	0.386	during a wartime emergency period. Record of actions taken.
0.298	tions. Authority concerning Communica-	0.387	Other defense and emergency pre-
	tions Satellite matters.		paredness delegations; cross refer-
0.302	Authority concerning records and papers.		ence.
0.303	Authority concerning extension of		Subpart C—General Information
0.304	time and waivers.		GENERAL
0.504	Authority delegated jointly to Chiefs of Common Carrier and Safety and	0.401	Location of Commission offices.
	Special Radio Services Bureaus.	0.403 0.405	
0.307	Record of actions taken.	0.406	
	FIELD ENGINEERING BUREAU		PRINTED PUBLICATIONS
0.311		0.411	General reference materials.
	to the Deputy Chief of the Field Engineering Bureau.	0.412	Nongovernment publications.
0.314	Authority delegated to the Engineers	0.413	The Commission's printed publica-
0.315	in Charge. Authority delegated to Engineers in	0.414	tions. The Communications Act and other
0.010	Charge who serve as FCC representa-		statutory materials.
	tives to Office of Civil Defense	0.415	The rules and regulations (looseleaf
	(OCD) Regional Civil Defense Co- ordinating Boards and Regional	0.416	service). The Federal Communications Com-
	Preparedness Committees of the	0.210	misson Reports.
0.010	Office of Emergency Planning (OEP).	0.417	The Annual Reports.
0.316	Authority delegated to Marine Super- visors at marine offices, to engineers	0.420	Other Commission publications.
	engaged in ship inspection duties	Fo	rms and Documents Available Upon
	at radio district offices, and to radio engineers at suboffices.	0.491	REQUEST
0.317	engineers at subomces. Record of actions taken.	0.421 0.422	Application forms. Current action documents and public
0.318	Authority delegated to operator ex-		notices.
	aminer.	0.423	Information bulletins.

200

LISTS CONTAINING INFORMATION COMPILED BY

0.431	The	FCC	Service	Frequency	Lists.
0.432	The	NAR	BA List.		

0.433 The Radio Equipment Lists.

0.434 Lists of authorized broadcast stations and pending broadcast applications.

PUBLIC INFORMATION AND INSPECTION OF RECORDS

0.441 General.

0.443 General information office.

0.445 Publication, availability and use of opinions, orders, policy statements, interpretations, administrative manuals, and staff instructions.

0.451 Inspection of records.

0.453 Public reference rooms.

0.455 Other locations at which records may be inspected.

0.456 General correspondence files.

0.457 Records not routinely available for public information.

0.459 Requests that materials or information submitted to the Commission be withheld from public inspection.

0.461 Requests for inspection of materials not routinely available for public inspection.

0.463 Demand by competent authority for the production of documents or testimony concerning information contained therein.

0.465 Request for copies of materials which are available, or made available, for public inspection.

0.466 Fees.

0.467 Certified copies.

PLACES FOR MAKING SUBMITTALS OR REQUESTS FOR FILING APPLICATIONS, AND FOR TAKING EXAMINATIONS

0.471 Miscellaneous submittals or requests.

0.473 Reports of violations.

0.475 Applications for employment.

0.481 Place of filing applications for radio authorizations.

0.483 Applications for amateur station and operator license and/or commercial operator license.

0.485 Amateur and commercial operator examination points.

0.487 Applications for interim ship station licenses.

0.489 Applications for ship radio inspection and periodical survey.

0.491 Application for exemptions from compulsory ship radio requirements.

0.493 Non-radio common carrier applications.

AUTHORITY: The provisions of this Part 0 issued under secs. 4, 303, 48 Stat. 1066, 1082, as amended; 47 U.S.C. 154, 303, unless otherwise noted. Implement; 5 U.S.C. 552.

Subpart A-Organization

AUTHORITY: \$\$ 0.1 to 0.186 also issued under sec. 5, 48 Stat. 1068, as amended; 47 U.S.C. 155.

SOURCE: The provisions of this Subpart A appear at 28 F.R. 12392, Nov. 22, 1963, unless otherwise noted.

GENERAL

§ 0.1 The Commission.

The Federal Communications Commission is composed of 7 members, who are appointed by the President subject to confirmation by the Senate. Normally, one Commissioner is appointed or reappointed each year, for a term of 7 years. [32 F.R. 10569, July 19, 1967]

§ 0.3 The Chairman.

- (a) One of the members of the Commission is designated by the President to serve as Chairman, or chief executive officer, of the Commission. As Chairman, he has the following duties and responsibilities:
- (1) To preside at all meetings and sessions of the Commission.
- (2) To represent the Commission in all matters relating to legislation and legislative reports; however, any other Commissioner may present his own or minority views or supplemental reports.

(3) To represent the Commission in all matters requiring conferences or communications with other governmental officers, departments or agencies.

(4) To coordinate and organize the work of the Commission in such a manner as to promote prompt and efficient disposition of all matters within the jurisdiction of the Commission.

(b) The Commission will, in the case of a vacancy in the Office of the Chairman of the Commission, or in the absence or inability of the Chairman to serve, temporarily designate one of its members to act as Chairman until the cause or circumstance requiring such designation has been eliminated or corrected.

[32 F.R. 10569, July 19, 1967]

§ 0.4 Standing committees of Commissioners.

There are 3 standing committees of Commissioners; the Telegraph Committee, the Telephone Committee, and the Subscription Television Committee, each composed of 3 Commissioners. These committees are delegated authority to act or study and report on certain telegraph, telephone and subscription television matters from time to time. [32 F.R. 10569, July 19, 1967]

§ 0.5 General description of Commission organization and operations.

- (a) Principal staff units. The Commission is assisted in the performance of its responsibilities by its staff, which is divided into the following principal units:
 - (1) Office of Executive Director.
 - (2) Office of Chief Engineer.
 - (3) Office of General Counsel.
 - (4) Broadcast Bureau.
 - (5) Common Carrier Bureau.
- (6) Safety and Special Radio Services Bureau.
 - (7) Field Engineering Bureau.
 - (8) Office of Hearing Examiners.
 - (9) Review Board.
 - (10) Office of Opinions and Review.
 - (11) Office of the Secretary.
 - (12) Office of Information.
 - (13) CATV Task Force.
- (b) Staff responsibilities and functions. The organization and functions of these major staff units are described in detail in §§ 0.11-0.171. The defense and emergency preparedness functions of the Commission are set forth separately, beginning at § 0.181. For a complete description of staff functions, reference should be made to these provisions. (See also the U.S. Government Organization Manual, which contains a chart showing Commission's organization, names of the members and principal staff officers of the Commission, and other information concerning the Commission.) So that the public may more readily inform itself concerning the operations of the Commission as a whole, concerning the staff officials who exercise responsibility over matters in which they are interested and concerning the relationship between the several staff units in such matters, however, a brief overall description of staff functions and responsibilities is set forth in this paragraph.
- (1) The Executive Director. The Executive Director is directly responsible to the Commission, works under the supervision of the Chairman, and assists him in carrying out the Commission's organizational and administrative responsibilities. His principal role is to see that other staff units work together and promptly dispose of the matters for which they are responsible. He is directly responsible for internal administrative matters such as personnel and budget

planning, and supervises implementation of the Public Information Act of 1966.

- (2) The Chief Engineer and the General Counsel. Though primary responsibility in most established areas of regulation is lodged in other staff units, the Chief Engineer and the General Counsel are responsible for advising the Commission concerning any engineering or legal matter involved in the making and implementation of policy or in the decision of cases. For example, while policies solely to broadcasting are relating primarily the responsibility of the Broadcast Bureau, and the preparation of Commission opinions in hearing cases is primarily the responsibility of the Office of Opinions and Review, the Chief Engineer and the General Counsel may be called upon for advice and assistance in either area. The Chief Engineer and the General Counsel, in addition, exercise primary responsibility in areas of regulation which transcend the responsibilities of a single bureau. Thus, for example, the General Counsel is primarily responsible for the Rules of Practice and Procedure, Part 1 of this chapter, and the Chief Engineer is primarily responsible for frequency allocation and for other areas of regulation under Parts 2, 5, and 15. The General Counsel also represents the Commission in litigation in the courts and coordinates the preparation of the Commission's legislative program. Both the Chief Engineer and the General Counsel exercise responsibility in matters pertaining to international communications.
- (3) The operating bureaus. The principal work load operations of the Commission are conducted by the four operating bureaus.
- (i) Three of these bureaus: The Broadcast Bureau, Common Carrier Bureau, and Safety and Special Radio Services Bureau—exercise primary responsibility in the three principal areas of regulation into which the Commission has divided its responsibilities. Broadcast Bureau, as its name indicates, is responsible for the regulation of broadcast stations (see Part 73 of this chapter) and related facilities (see Part 74). The Common Carrier Bureau is responsible for the regulation of communications common carriers whether carriage involves the use of wire or radio facilities (see Parts 21-66). The Safety and Special Radio Services Bureau is responsible for the regulation of all other

radio stations with minor exceptions (e.g., experimental stations licensed under Part 5). These include amateur stations and numerous other categories of stations engaged in communication for safety, commercial or personal purposes (see Parts 81-99). Within its area of responsibility, each of these bureaus is responsible for developing and implementing a regulatory program; for processing applications for radio licenses or other filings: for the consideration of complaints and the conduct of investigations: for participation in Commission hearing proceedings as appropriate; and for the performance of such other functions as may be related to its area of responsibility.

(ii) The fourth operating bureau: The Field Engineering Bureau-maintains field offices and monitoring stations throughout the United States. It is responsible for detecting violations of regulations pertaining to the use of radio and, in this connection, monitors radio transmissions, periodically inspects stations, and investigates complaints of radio frequency interference. It issues violation notices to the station in question, thereby affording it an opportunity to take corrective measures. If formal enforcement action is appropriate, the proceedings are conducted by the staff unit which exercises primary responsibility over the station in questionusually one of the other operating bureaus. The Field Engineering Bureau, in addition, exercises responsibility over commercial radio operator matters (see Part 13 of this chapter), antenna structures (see Part 17), and the use of radio for purposes other than communication (see Part 18). It also conducts amateur operator examinations.

(4) Staff units which exercise responsibility for the decision of hearing cases. The Office of Hearing Examiners, the Review Board, and the Office of Opinions and Review exercise responsibility for the decision of hearing cases. The hearing examiners preside over hearing cases and issue initial decisions. In most cases, initial decisions are subject to review by the Revew Board, which is a permanent body composed of three or more senior Commission employees. Initial decisions may also be reviewed by one or more Commissioners designated by the Commission. In such cases, the Board or

designated Commissioner(s) issues a final decision, which is subject to possible review by the Commission. In other cases, the initial decision is reviewed directly by the Commission en banc. The Office of Opinions and Review assists and advises the Commission, and any Commissioner(s) designated to review an initial decision, in the decision of cases which come before them.

(5) The Secretary. With minor exceptions, the Secretary signs all correspondence and documents adopted by the Commission. He is custodian of the Commission's seal and records. He maintains records of Commission actions and the dockets of hearing proceedings, and is responsible for their accuracy, authenticity, and completeness. Except as otherwise provided in this chapter, he is the proper addressee and recipient of papers filed with the Commission.

(6) Office of Information. The Office of Information is responsible for informing the public concerning actions which have been taken by the Commission and is the contact point for the press, the industry and the public in the matter of general information relating to the Commission and its activities.

(7) The CATV Task Force. The CATV Task Force is responsible for the development and implementation of a regulatory program for community antenna television systems and community antenna relay stations (see Subparts J and K of Part 74 of this chapter). The licensing of related microwave radio facilities is coordinated with the Task Force by the Common Carrier Bureau and the Safety and Special Radio Services Bureau.

(c) Delegations of authority to the staff. Pursuant to section 5(d) of the Communications Act, the Commission has delegated authority to its staff to act on matters which are minor or routine or settled in nature and those in which immediate action may be necessary. See Subpart B of this Part. Actions taken under delegated authority are subject to review by the Commission, on its own motion or on an application for review filed by a person aggrieved by the action. Except for the possibility of review, actions taken under delegated authority have the same force and effect as actions taken by the Commission. The delegation of authority to a staff officer, however, does not mean that he will exercise that

authority in all matters subject to the delegation. In non-hearing matters, the staff is at liberty to refer any matter at any stage to the Commission for action. upon concluding that it involves matters warranting the Commission's consideration, and the Commission may instruct the staff to do so. In like manner, in hearing cases, pursuant to § 0.361(b) and (c), the Commission may direct that matters pending before the Review Board be certified to the Commission for decision, and the Board may itself certify such matters to the Commission, with a request that they be acted upon by the Commission.

(d) Commission action. Matters requiring Commission action, or warranting its consideration, are dealt with by the Commission at regular weekly meetings, or at special meetings called to consider a particular matter. Meetings are normally held at the principal offices of the Commission in the District of Columbia, but may be held elsewhere in the United States. In appropriate circumstances, Commission action may be taken between meetings "by circulation", which involves the submission of a document to the Commissioners for his each of approval.

[32 F.R. 10569, July 19, 1967, as amended at 32 F.R. 13125, Sept. 15, 1967; 33 F.R. 8227, June 1, 1968]

OFFICE OF EXECUTIVE DIRECTOR

§ 0.11 Functions of the Office.

The Executive Director is designated by the Commission and is directly responsible to the Commission under the supervision of the Chairman. He has the following duties and responsibilities:

- (a) To provide sustained administrative leadership and coordination of staff activities in carrying out the policies of the Commission, through overall supervision and coordination, but not control, of such staff activities. In this capacity, he coordinates the activities of policy making staff officers to assure that adequate information and recommendations in important policy areas are expeditiously considered by the staff and brought promptly to the attention of the Commission.
- (b) To review with the Commission and with heads of the several bureaus and offices, the program and procedures of the Commission and to make recommendations thereon as may be necessary

to administer the Communications Act most effectively in the public interest.

(c) To assist the Chairman in carrying out the administrative and executive responsibilities delegated to the Chairman as the administrative head of the agency and, in connection therewith, to plan, direct, coordinate, and manage the administrative affairs of the Commission with respect to the functions of personnel, budget, planning, office services, mail and files, messenger and chauffeur services, and international telecommunications settlements.

(d) Under the general direction of the Defense Commissioner, and with the advice and assistance of the heads of the several bureaus and offices, the Executive Director coordinates the defense activities of the Commission, and has the following duties and responsibilities:

- (1) To act as Alternate Defense Coordinator in representation with other agencies with respect to planning for the continuity of the essential functions of the Commission under national emergency conditions, and serves as the alternate representative of the Commission to the Interagency Emergency Planning Committee of the Office of Emergency Planning.
- (2) To serve as the alternate representative of the Commission to the Interagency Civil Defense Committee of the Office of Civil Defense, Department of the Army.
- (3) To keep the Defense Commissioner informed as to significant developments in this area.
- (e) With the concurrence of the General Counsel, to interpret the rules and regulations pertaining to application filing fees.

[28 F.R. 12392, Nov. 22, 1963, as amended at 29 F.R. 3202, Mar. 10, 1964; 29 F.R. 14664, Oct. 28, 1964]

§ 0.12 Units in the Office.

- (a) Immediate Office of the Executive Director.
 - (b) Administrative Services Division.
 - (c) Budget and Fiscal Division.
 - (d) Data Processing Division.
- (e) Emergency Communications Division.
- (f) Management Information Division.
 - (g) Personnel Division.
- (h) Property Management Division. [34 F.R. 14330, Sept. 12, 1969]

OFFICE OF CHIEF ENGINEER

§ 0.31 Functions of the Office.

The Office of the Chief Engineer has the following duties and responsibilities:

- (a) To plan and direct broad programs to develop information on the progress of communication techniques and equipment, radio wave propagation, and new uses for communications, and to advise the Commission and bureaus in such matters.
- (b) To represent the Commission on various national and international organizations devoted to the progress of communications and the development of information and standards relative thereto.
- (c) To conduct scientific and technical studies in advanced phases of terrestrial and space communications and to conduct special projects to obtain theoretical and experimental data on new or improved techniques.
- (d) To develop and administer procedures to acquire, store, and retrieve scientific and technical information useful in the engineering work of the Commission.
- (e) To advise and represent the Commission on the allocation of radio frequencies, including international agreements pertaining to frequency allocations and usage.
- (f) In cooperation with the General Counsel, to render advice to the Commission, participate in and coordinate the staff work with respect to general frequency allocation proceedings and other proceedings not within the jurisdiction of any single bureau, and to render advice with respect to rule making matters and proceedings affecting more than one bureau.
- (g) To collaborate with the bureaus in the formulation of the technical requirements of the rules and regulations, and to advise the Commission on such matters.
- (h) To administer Part 5 of this chapter, including licensing, record keeping, and rule making.
- (i) To perform all engineering and management functions of the Commission with respect to formulating rules and regulations, technical standards, and general policies for Parts 15 and 18 of this chapter, and for type approval, type acceptance, and certification of radio equipment for compliance with the Commission's rules.

- (j) To maintain liaison with other agencies of government and with technical experts representing foreign governments, and to deal with members of the public and of the industries concerned.
- (k) To calibrate and standardize technical equipments and installations used by the Commission.
- (1) To exercise such authority as may be assigned or referred by the Commission pursuant to section 5(d) of the Communications Act of 1934, as amended.

§ 0.32 Units in the Office.

The Office of Chief Engineer comprises the following units:

- (a) Immediate Office of the Chief Engineer.
 - (b) Research Division.
 - (c) Technical Division.
 - (d) Laboratory Division.
- (e) Frequency Allocation and Treaty Division.

§ 0.33 Immediate Office of the Chief Engineer.

The immediate Office of the Chief Engineer advises the Commission and bureaus on technical matters and directs and coordinates the functions of the Office.

§ 0.34 Research Division.

The Research Division analyzes and disseminates to the Commission and the several operating bureaus, technical and scientific data relating to advanced phases of terrestrial and space communications; conducts scientific studies in wave propagation, satellite and space communications; studies technical aspects of potential new uses of radio: conducts special projects to obtain theoretical and experimental data on new or improved techniques; provides engineering assistance to the Commission and the several bureaus in the development of computer programs and automatic data processing techniques; develops procedures to acquire, store, and retrieve scientific and technical information; and maintains familiarity with relevant research activities by means of visits, conferences, and correspondence.

§ 0.35 Technical Division.

The Technical Division develops practical applications for results of research, including research conducted by the Re-



search Division: develops technical rules and standards for the various radio services in consultation with the bureaus concerned: conducts technical studies of equipment design and performance; acts on applications for type acceptance of equipment and issues lists of type accepted and type approved equipment; in consultation with the Chief, Broadcast Bureau, acts on applications for advance approval of subscription television technical systems and issues lists of approved subscription television technical systems; administers Part 5 of this chapter regulating the Experimental Radio Services; performs engineering and management functions with respect to formulating rules, technical standards, and general policy for Parts 15 and 18 of this chapter; performs rule making for that portion of Part 2 of this chapter relative to equipment; provides Commission representation at national and international conferences; and reviews patents as they are issued by the Patent Office to determine which patents may dominate radio services regulated by the Commission and notifies the General Counsel as to such findings.

[34 F.R. 14374, Sept. 13, 1969]

§ 0.36 Laboratory Division.

The Laboratory Division studies new phenomena, proposed new systems, and equipment looking toward greater use of radio, the reduction of interference, and the establishment of appropriate rules and regulations; participates in various intergovernmental, national, and international organizations looking toward the standardization of equipment and measuring units and methods as well as the more efficient use of the radio spectrum or the reduction of interference; designs and assembles apparatus for special tests and studies, and performs special tests and studies concerning propagation, equipment or systems, and evaluates the results of such tests or studies with regard to the Commission's problems, often looking toward new or modified rules; makes type approval tests on equipment including those equipments under Parts 18, 73, 74, 83, and 95 of this chapter requiring type approval, and makes recommendations regarding type approval; provides information and comments on test procedures and test results to assist the Technical

Research Division in its evaluation of material supporting certifications and applications for type acceptance: conducts special tests of equipments for the Technical Research Division in connection with the certification and type acceptance program; studies equipment problems of data procurement and enforcement and develops, designs, and constructs equipment for use in connection with the Commission's Field Engineering Bureau activities as well as other Commission activities; standardizes and calibrates equipment and installation for the Field Engineering Bureau: and makes tests of radio devices for other government departments.

§ 0.37 Frequency Allocation and Treaty Division.

The Frequency Allocation and Treaty Division makes continuing studies of new technical developments affecting frequency requirements and of utilization of the frequencies between the several radio services to establish their allocation requirements; proposes adjustments in the Table of Frequency Allocation when necessary; provides Commission representation on, and coordination with, the Interdepartment Radio Advisory Committee and, as may be required, on other national and international telecommunication bodies; coordinates frequency allocation policy matters involving government users of radio with the Office of Emergency Planning; maintains the Commission's master frequency record of assignments made; notifies United States frequency assignments to the International Frequency Registration Board of the International Telecommunication Union: performs staff functions relating to international communications conferences and agreements having to do with frequency allocation and assignment: and communicates as necessary with administrations in foreign countries, through appropriate channels, concerning matters which relate to assignment of radio frequencies and to control of radio interference.

OFFICE OF GENERAL COUNSEL

§ 0.41 Functions of the Office.

The Office of the General Counsel has the following duties and responsibilities:

(a) To advise and represent the Commission in matters of litigation.

- (b) To advise and make recommendations to the Commmission with respect to proposed legislation and to coordinate the preparation of Commission views thereon for submission to Congress.
- (c) To participate in international conferences and in the implementation of international agreements.
- (d) To interpret the statutes, international agreements, and international regulations affecting the Commission.
- (e) To prepare and make recommendations and interpretations concerning procedural rules of general applicability and to review all rules for consistency with other rules, uniformity, and legal sufficiency.
- (f) To conduct research in legal matters as directed by the Commission.
- (g) In cooperation with the Chief Engineer, to participate in, render advice to the Commission, and coordinate the staff work with respect to general frequency allocation proceedings and other proceedings not within the jurisdiction of any single bureau, and to render advice with respect to rule making matters and proceedings affecting more than one bureau.
- (h) To perform all legal functions with respect to experimental operations under Part 5 of this chapter; the operation of restricted radiation devices under Parts 15 and 18 of this chapter; and type approval and type acceptance of radio equipment.
- (i) To exercise such authority as may be assigned or referred to it by the Commission pursuant to section 5(d) of the Communications Act of 1934, as amended.
- (j) To cooperate with the Common Carrier Bureau and the Office of Chief Engineer on all matters pertaining to space satellite communications.
- (k) To study the licensing practices of patentees and assignees in communications services regulated by the Commission, particularly in situations referred to it by the Office of Chief Engineer; and, in collaboration with the Office of Chief Engineer, to develop appropriate recommendations for Commission action.
- (1) To interpret statutes and executive orders affecting the Commission's national defense responsibilities, and to perform such functions involving implementation of such statutes and executive orders as may be assigned to it by the Commission or the Defense Commissioner.

- (m) To perform all legal functions with respect to leases, contracts, tort claims and such other internal legal problems as may arise.
- (n) To prepare the bound volumes of the FCC Reports.
- (o) To review the national security aspects of applications for commercial operator licenses or permits forwarded to it by the Field Engineering Bureau, and of applications for amateur operator and station licenses submitted to it by the Safety and Special Radio Services Bureau.

§ 0.42 Units in the Office.

The Office of General Counsel is divided into the following units:

- (a) Immediate Office of the General Counsel.
 - (b) Litigation Division.
 - (c) Legislation Division.
- (d) Administrative Law and Treaties Division.
- (e) Enforcement and Defense Division.

§ 0.43 Immediate Office of the General Counsel.

The Immediate Office of the General Counsel directs and coordinates the functions of the Office.

§ 0.44 Litigation Division.

The Litigation Division advises and represents the Commission in all matters of litigation to which the Commission is a party, advises the Commission as to legal questions involved in proposed actions and policies in the light of past and pending litigation, and conducts research in legal matters as directed by the General Counsel.

§ 0.45 Legislation Division.

The Legislation Division advises and makes recommendations to the Commission with respect to proposed legislation and coordinates the preparation of Commission views thereon for submission to Congress, interprets statutes affecting the Commission, and conducts research in legal matters as directed by the General Counsel.

§ 0.46 Administrative Law and Treaties Division.

The Administrative Law and Treaties Division has the following duties and responsibilities:

(a) To conduct research in legal matters and prepare legal opinions as directed by the General Counsel.

- (b) To participate in international conferences and in the implementation of international agreements, and to interpret international agreements and international regulations affecting the Commission.
- (c) To prepare and/or coordinate recomendations concerning procedural rules of general applicability, and to maintain liaison with Government agencies and private organizations concerned with improving the Commission's procedures.
- (d) To coordinate rule making matters and proceedings affecting more than one Bureau.
- (e) In cooperation with the Office of Chief Engineer, to participate in general frequency allocation proceedings affecting more than one Bureau and in rule making proceedings involving Parts 5, 15, or 18 of this chapter.
- (f) To cooperate with the Common Carrier Bureau and the Office of Chief Engineer on matters pertaining to satellite communications.
- (g) To study and develop recommendations with regard to the licensing practices of patentees and assignees in communications services regulated by the Commission.
- (h) To perform any additional functions assigned to it by the General Counsel.

§ 0.47 Enforcement and Defense Divi-

The Enforcement and Defense Division has the following duties and responsibilities:

- (a) To advise the Commission and to take action, usually upon recommendations submitted by the operating bureaus, as to civil and criminal proceedings to enforce the Communications Act, the rules and regulations of the Commission, and Commission orders in the courts of the United States.
- (b) To interpret statutes and executive orders affecting the Commission's national defense responsibilities, and to perform the functions of the Office of General Counsel involving implementation of such statutes and executive orders.
- (c) To perform all legal functions involved in administering and enforcing the rules and regulations of the Commission pertaining to the Experimental Radio Service (Part 5 of this chapter), Radio Frequency Devices (Part 15 of

this chapter), the Industrial, Scientific, and Medical Service (Part 18 of this chapter), and type approval and type acceptance of radio equipment, including educational functions directed to obtaining voluntary compliance with the provisions of Parts 15 and 18 of this chapter.

(d) To perform all legal functions with respect to leases, contracts, tort claims and such other internal legal

problems as may arise.

(e) To prepare the bound volumes of the FCC Reports.

- (f) To review the national security aspects of applications for commercial operator licenses or permits submitted for review by the Field Engineering Bureau, and of applications for amateur operator and station licenses submitted for review by the Safety and Special Radio Services Bureau.
- (g) To perform any additional functions assigned to it by the General Counsel.

OFFICE OF THE SECRETARY

§ 0.51 Functions of the Office.

The Office of the Secretary has the following duties and responsibilities:

- (a) To maintain minutes and records of official Commission actions and, with minor exceptions, to sign all correspondence and documents adopted by the Commission.
- (b) Upon obtaining the approval of the chief of the bureau or head of the staff office primarily responsible for the particular part or section of the rules involved, to make nonsubstantive, editorial revisions of the Commission's rules and regulations.
- (c) To maintain dockets of all Commission hearing proceedings, except that dockets are not to be established in revocation proceedings concerning licenses in the Safety and Special Radio Services unless and until respondents file timely notices of appearance and requests for hearing.
- (d) To maintain library and reference facilities.
- (e) To direct and supervise the development and execution of a technical assistance program in cooperation with the Foreign Operations Administration and other Government agencies.

[28 F.R. 12392, Nov. 22, 1963, as amended at 33 F.R. 5302, Apr. 3, 1968; 33 F.R. 8227, June 1, 1968]

§ 0.52 Units in the Office.

The Office of the Secretary is divided into the following units:

- (a) Dockets Division.
- (b) Minute and Rules Division.
- (c) Library Division.
- (d) Technical Assistance Division.

§ 0.53 Dockets Division.

The Dockets Division maintains the official dockets of all Commission hearing cases, except that dockets are not to be established in revocation proceedings concerning licenses in the Safety and Special Radio Services unless and until respondents file timely notices of appearance and requests for hearing.

[33 F.R. 5302, Apr. 3, 1968]

§ 0.54 Minute and Rules Division.

The Minute and Rules Division maintains official minutes of all Commission actions and notations of Comsubmits to mission meetings; the Federal Register Office of all official Commission documents which require publication: maintains the Commission's official copy of the rules and regulations with history and background; and acts as provided for in § 0.51(b).

§ 0.55 Library Division.

The Library Division catalogues and maintains books, periodicals, and other reference material; provides reading and reference room facilities and service; and provides library research and reference service to Commissioners and staff members.

§ 0.56 Technical Assistance Division.

The Technical Assistance Division develops, in cooperation with the Agency for International Development, technical assistance programs with respect to communications in accordance with Commission policies, and maintains liaison with the Agency for International Development and other government agencies concerning technical assistance.

OFFICE OF INFORMATION

§ 0.61 Functions of the Office.

The Office of Information is responsible for releasing public announcements of the Commission; is the central depository of this material for reference and call; prepares certain informational publications and material, including

annual reports; provides an internal information service for the Commissioners and staff; and is the contact point for the press, industry and public in the matter of general information relating to the Commission and its activities.

[32 F.R. 13125, Sept. 15, 1967]

BROADCAST BUREAU

§ 0.71 Functions of the Bureau.

The Broadcast Bureau assists, advises, and makes recommendations to the Commission with respect to the development of a regulatory program for the broadcast services and is responsible for the performance of any work, function, or activities to carry out that program in accordance with applicable statutes, international agreements, rules and regulations, and policies of the Commission.

The Bureau performs the following functions:

- (a) Receives, examines, files, indexes, records and processes applications in the broadcast services, makes recommendations to the Commission thereon and issues authorizations in accordance with Commission instructions and directions.
- (b) Processes petitions in broadcast matters (protests and orders of designation) not specified in § 0.171.
- (c) By notification in accordance with the procedures set forth in the North American Regional Broadcasting Agreement, establishes priorities for United States stations under this and associated agreements. Receives and examines notifications of assignments in other North American countries, and resolves conflicts by negotiation.
- (d) Participates in hearings involving applications, rule making, and other matters which pertain to the radio broadcast services, including proceedings pursuant to sections 312 and 316 of the Communications Act of 1934, as amended.
- (e) Makes recommendations to the Commission concerning the promulgation of rules and standards in the broadcast services.
- (f) Participates in international conferences with respect to broadcast services, and in implementation of international broadcast agreements.
- (g) Studies frequency requirements in the broadcast services and makes recommendations with respect to the allocation of frequencies and the drafting of frequency assignment plans in such services

- (h) Confers with government and industry groups interested in the problems of broadcast services.
- (i) Studies and establishes technical requirements for equipment in the broadcast services in accordance with standards established by the Commission.
- (j) Exercises such authority as may be assigned or referred by the Commission pursuant to section 5(d) of the Communications Act of 1934, as amended.

§ 0.72 Units in the Bureau.

The Broadcast Bureau is divided into the following units:

- (a) Office of the Bureau Chief.
- (b) Broadcast Facilities Division.
- (c) Renewal and Transfer Division.
- (d) Hearing Division.
- (e) Rules and Standards Division.
- (f) License Division.
- (g) Research and Education Division.
- (h) Office of Network Study.
- (i) Complaints and Compliance Division.

§ 0.73 Office of the Bureau Chief.

The Office of the Bureau Chief is composed of the immediate offices of the Chief and the Assistant Chiefs of the Broadcast Bureau, and of the Administrative Assistant. It assists the Chief of the Broadcast Bureau in planning, directing, coordinating and executing the functions of the Bureau.

§ 0.74 Broadcast Facilities Division.

The Broadcast Facilities Division is responsible for all functions indicated in the statement contained in § 0.71, insofar as such functions pertain to standard (AM), FM, television, international, experimental, and auxiliary broadcast services, excluding functions stated in §§ 0.75, 0.76, 0.77, and 0.81.

§ 0.75 Renewal and Transfer Division.

The Renewal and Transfer Division is responsible for all functions indicated in the statement contained in § 0.71, insofar as such functions pertain to applications for renewal or assignment of a license or construction permit, or for transfer of control of a corporate licensee or permittee.

§ 0.76 Hearing Division.

The Hearing Division is responsible for all functions indicated in the statement contained in § 0.71, insofar as such functions pertain to applications which have been designated for hearing or to proceedings pursuant to sections 312 and 316 of the Communications Act of 1934, as amended.

§ 0.77 Rules and Standards Division.

The Rules and Standards Division is responsible for all functions indicated in the statement contained in § 0.71, insofar as such functions relate to the development or revision of rules and standards, to international conferences, and to special projects in the broadcast services.

§ 0.78 License Division.

The License Division exercises responsibility for the receipt, initial examination, filing, indexing, recording and routing of all applications in the broadcast services; the clerical functions with respect to the issuance of broadcast authorizations; and for the maintenance of a public reference room in cooperation with the Dockets Division of the Office of the Secretary where public information with respect to all broadcast matters and all docketed proceedings may be examined.

§ 0.79 Research and Education Division.

The Research and Education Division compiles data and prepares reports to the Commission on the condition and status of the broadcast industry; studies the social and economic factors affecting communications; and advises the Bureau and the Commission with respect to the development and promotion of the educational broadcasting and the commercial broadcasting services.

§ 0.80 Office of Network Study.

The Office of Network Study conducts studies and compiles data relating to radio and television network operations necessary for the Commission to develop and maintain an adequate regulatory program.

§ 0.81 Complaints and Compliance Division.

The Complaints and Compliance Division is responsible for all functions indicated in the statement contained in § 0.71, insofar as such functions pertain to complaints concerning the conduct of radio and television broadcast operations and compliance with statutes, international agreements, and policies and rules of the Commission, except insofar as functions are specifically delegated to

other bureaus and staff offices of the Commission.

CATV TASK FORCE

§ 0.85 Functions of the Task Force.

The CATV Task Force assists, advises and makes recommendations to the Commission with respect to the development of a regulatory program for community antenna television systems and related private and common carrier microwave radio facilities. The Task Force performs the work and activities involved in the implementation of the Commission's regulatory program as it pertains to the regulation of community antenna television systems (see Subpart K, Part 74 of this chapter); and performs the work and activities involved in the licensing and regulation of Community Antenna Relay stations, after coordination with the Broadcast Bureau (see Subpart J, Part 74). The licensing and regulation of related common carrier microwave facilities are coordinated with the Task Force by the Common Carrier Bureau (see Subpart I, Part 21). The licensing and regulation of related microwave facilities in the Business Radio Service are coordinated with the Task Force by the Safety and Special Radio Services Bureau (see Subpart L, Part 91). The Task Force performs the following functions:

- (a) Processes proposals, applications, and requests regarding community antenna television systems, makes recommendations thereon to the Commission, and takes appropriate actions in accordance with Commission instructions and directions.
- (b) Makes recommendations to the Commission concerning the promulgation of rules and regulations affecting community antenna television systems.
- (c) Participates in hearings involving applications, rulemaking, and other matters which may have a significant impact on the Commission's regulatory program for community antenna television systems.
- (d) Confers with Government and industry groups interested in problems relating to community antenna television systems.
- (e) Exercises such authority as may be assigned or referred by the Commission pursuant to section 5(d) of the Communications Act of 1934, as amended.

[31 F.R. 16714, Dec. 30, 1966]

COMMON CARRIER BUREAU

§ 0.91 Functions of the Bureau.

The Common Carrier Bureau develops, recommends, and administers policies and programs with respect to the regulation of rates, services, accounting, and facilities of communication common carriers involving the use of wire, cable, radio, and space satellites. The Bureau performs the following functions:

- (a) Advises and makes recommendations to the Commission and represents the Commission in matters pertaining to common carrier regulation and licensing.
- (b) Participates in all phases of international conferences involving common carrier matters.
- (c) Collaborates with representatives of State regulatory commissions and with the National Association of Railroad and Utilities Commissioners in the conduct of cooperative studies of regulatory matters of common concern.
- (d) Participates, as a party, in adjudicatory hearings in which significant common carrier issues are involved.
- (e) Advises and assists members of the public and the industries regulated on communication matters.
- (f) Makes recommendations to the Commission on policy and technical determinations regarding the use of space satellites for other than common carrier communication purposes.
- (g) Exercises such authority as may be assigned or referred to it by the Commission pursuant to section 5(d) of the Communications Act of 1934, as amended.

[31 F.R. 16315, Dec. 21, 1966]

§ 0.92 Units in the Bureau.

- (a) Office of the Bureau Chief.
- (b) International and Satellite Communication Division.
 - (c) Domestic Rates Division.
 - (d) Domestic Radio Division.
- (e) Domestic Services and Facilities Division.
 - (f) Field Operations Division.
- (g) Economic Studies Division. [31 F.R. 16315, Dec. 21, 1966]

§ 0.93 Field offices.

Common Carrier Bureau field offices are located at 90 Church Street, New York, N.Y. 10007; and Room 1016, 208 North Broadway, St. Louis, Mo. 63102. [32 F.R. 20859, Dec. 28, 1967]

FIELD ENGINEERING BUREAU

§ 0.111 Functions of the Bureau.

The Field Engineering Bureau is responsible for all Commission engineering activities performed in the field relating to radio stations including station inspections, surveys, monitoring, direction finding, signal measurement and investigations: for those enforcement activities performed in the field dealing with the suppression of interference and the inspection of devices possessing electromagnetic radiation characteristics; and such other field inspections or investigations as might be required by the Commission, or the bureaus and staff offices. The Bureau also performs the following functions:

(a) Develops rules and regulations which will provide such classes of commercial operators as may be required in the various radio services regulated by the Commission, and administers and enforces such rules and regulations.

(b) Prepares and conducts commercial radio operator examinations and issues commercial operator licenses.

(c) Conducts amateur examinations, and upgrades amateur licenses in situations where the upgrading can be accomplished by endorsements.

(d) Processes data with respect to proposed new or modified antenna structures covered in Part 17 of this chapter to determine whether such proposed construction will create hazards to air navigation.

(e) Represents the Commission and participates in interagency committees and in international conferences with respect to matters for which the Bureau is responsible.

(f) Enforces and administers Parts 15 and 18 of this chapter relative to equipment, interference, and related problems in the industrial, scientific and medical services arising from restricted radiation devices.

(g) Subject to the policy guidance of the Defense Commissioner, and in consultation with the Executive Director, and the Chief, Office of Emergency Communications.

(1) Prepares detailed plans for and coordinates the development of the FCC owned, leased, and operated communications systems designated as assets of the National Communications System:

(2) Selects and prepares detailed plans for the physical facilities designated as FCC Emergency Relocation facilities, including plans for the utilization of such facilities on a day-to-day basis;

(3) Prepares detailed operational plans, procedures, and arrangements for certain classified activities with other Government Agencies, as related to the FCC owned, leased and operated communications systems;

(4) Prepares the detailed emergency operational staffing plans and procedures for performing certain classified functions of the Field Engineering Bureau under national emergency conditions, including selection, assignment, emergency notification, transportation, and specialized training of personnel and Special Task Forces (FEB) personnel; and

(5) Investigates violations of pertinent law and regulations in an emergency involving occupancy of the radio spectrum, and develops procedures designated to initiate, recommend, or otherwise bring about appropriate enforcement action required in the interest of national security.

(h) Subject to the policy guidance of the Defense Commissioner, provides the principal representation for the Commission to the Regional Preparedness Committees of the Office of Emergency Planning for the purpose of carrying out detailed planning for the continuity of essential Government functions in a national emergency.

(i) In cooperation with the Office of Emergency Communications, and subject to the policy guidance of the Defense Commissioner, provides alternate representation for the Commission to the Regional Civil Defense Coordinating Boards.

(j) Exercises such authority as may be assigned or referred by the Commission pursuant to section 5(d) of the Communications Act of 1934, amended.

[29 F.R. 14664, Oct. 28, 1964, as amended at 29 F.R. 15758, Nov. 24, 1964]

§ 0.112 Units in the Bureau.

The Bureau consists of the following units:

(a) Office of the Bureau Chief.

(b) Engineering and Facilities Division.

(c) Field Offices Division and its associated field organization consisting of district offices, sub-offices, marine offices, and mobile enforcement units.

(d) Monitoring Systems Division and its associated field organization consisting of monitoring stations.

§ 0.113 Office of Bureau Chief.

The Office of the Bureau Chief plans. directs, and coordinates the functions of the Bureau. It includes a Deputy Chief: an Assistant Chief; an administrative branch, which performs the administrative functions of the Bureau and is responsible for planning, coordinating and administering the Bureau's employee technical supervisory training programs; a Special Assistant to the Chief of the Bureau responsible for directing the detailed planning and development of the FCC owned, leased and operated communications systems designated as assets of the National Communications System; and an Attorney Advisor, who advises the Chief of the Bureau and the divisions of the Bureau on legal questions relative to such matters as applications for commercial operator licenses. interference problems, investigations, and other activities of the Bureau.

[29 F.R. 15758, Nov. 24, 1964]

§ 0.114 Engineering and Facilities Division.

The Engineering and Facilities Division is responsible for all functions indicated in § 0.111 insofar as technical engineering standards are concerned; for the development of measuring procedures and techniques for the guidance of field personnel in performing engineering measurements; for the determination of technical equipment and facilities requirements of all Bureau field installations in the conduct of their assigned duties and for the provision of such equipment and facilities; the administration, interpretation and revision of Part 17 of this chapter governing construction, marking, and lighting of antenna structures, including the processing of data concerning proposed new or modified antenna construction to insure no hazard to air navigation results from the proposed construction; for liaison with the Federal Aviation Administration in respect to matters concerning antenna tower construction and antenna hazards; and for liaison with the Office of the Chief Engineer and with other Government agencies with respect to technical engineering matters of Bureau

[28 F.R. 12392, Nov. 22, 1963, as amended at 32 F.R. 20859, Dec. 28, 1967]

§ 0.115 Field Offices Division.

The Field Offices Division is responsible for:

- (a) The examination functions of the Bureau, including responsibility for Part 13 of this chapter concerning the licensing of commercial radio operators, and recommending action on matters of noncompliance with rules, acts, or treaties by these operators.
- (b) The inspection functions of the Bureau, including responsibility for the development and making of recommendations with respect to the priority and frequency of station inspections.
- (c) Planning enforcement and investigative programs for the Bureau; directing and supervising investigations by the field offices; and maintaining liaison with other bureaus and offices of the Commission with respect to the conduct of investigations in their behalf.
- (d) The administration of Parts 15 and 18 of this chapter relative to equipment, interference and related problems involving the devices and equipment regulated by these parts, including review and action upon certifications submitted to the Commission pursuant to Part 18 of this chapter.
- (e) Supervising and managing the activities of the Engineers in Charge of the district offices, the Radio Engineers in Charge of suboffices, the Marine Supervisors and the Engineers in Charge of mobile TV enforcement units; and performing those administrative activities which will insure the most effective performance of the field operations of the Bureau.

§ 0.116 Monitoring Systems Division.

The Monitoring Systems Division is responsible for standards, techniques and field operations indicated in § 0.111 for the radio monitoring function, including coordinating Commission monitoring activities with centralizing offices for international monitoring in other countries, with other Federal Government agencies, and with industry self-help groups. The monitoring stations, whose operations are directed and the results evaluated by this Division, perform surveillance of the radio spectrum, detect and locate illegal radio stations and sources of interference, enforce radio and regulations, gather facts through monitoring and engineering measurements to resolve interference problems and to assist the Commission in rulemaking, participate through direction finding in search and rescue



operations involving distressed ships and aircraft, and perform monitoring and direction finding work under contractual arrangements with other Federal governmental agencies.

- § 0.121 Location of field offices and monitoring stations.
- (a) District Offices and their suboffices are located at the following addresses:

Radio	Address of the Engineer in Charge	Territory within district			
district		States	Counties		
1	1600 Customhouse, Boston, Mass. 02109.	Connecticut	All counties.		
		Massachusetts New Hampshire Rhode Island Vermont	Do. Do. Do. Do.		
2	748 Federal Bldg., 641 Washington St., New York, N.Y.	New Jersey	Bergen, Essex, Hudson, Hunterdon, Mercer,		
		New York	Middlesex, Monmouth, Morris, Passaic, Somerset, Sussex, Union, and Warren. Albany, Bronx, Columbia, Delaware, Dutchess, Greene, Kings, Nassau, New York, Orange, Putnam, Queens, Rensselaer, Richmond, Rockland, Schenectady, Suffolk, Sullivan, Ulster, and Westchester.		
3	1005 New U.S. Customhouse, Philadelphia, Pa. 19106.	Delaware New Jersey	New Castle. Atlantic, Burlington, Camden, Cape May, Cumberland, Gloucester, Ocean, and Salem.		
		Pennsylvania	Adams, Berks, Bucks, Carbon, Chester, Cumberland, Dauphin, Delaware, Lancaster, Lebanon, Lehigh, Monroe, Montgomery, Northampton, Perry, Philadelphia, Schuylkill, and York. Kent and Sussex.		
4	Room 819, Federal Building,	Delaware	Kent and Sussex.		
	Baltimore, Md. 21201.	Maryland Virginia	All except District 24. Clarke, Fairfax all except District 24, Fauquier, Frederick, Loudoun, Page, Prince William,		
		West Virginia	Rappahannock, Shenandoah, and Warren. Barbour, Berkeley, Grant, Hampshire, Hardy, Harrison, Jefferson, Lewis, Marion, Mineral, Monongalia, Morgan, Pendleton, Preston, Randolph, Taylor, Tucker, Upshur.		
5	Room 400, Federal Bldg., Nor-	North Carolina	All except District 6.		
6	folk, Va. 23510. 1602 Gas Light Tower, 235 Peachtree Street NE,	Virginia	All except Districts 4 and 24. All except District 8. All counties.		
	Atlanta, Ga. 30303. Suboffice: Post Office Box 8004, Room 238, Post Office Bldg., Savannah, Ga. 31402.	North Carolina	Ashe, Avery, Buncombe, Burke, Caldwell, Cherokee, Clay, Cleveland, Graham, Hay- wood, Henderson, Jackson, McDowell, Macon, Madison, Mitchell, Polk, Rutherford, Swain, Transylvania, Watauga, Yancey.		
		South Carolina	All counties.		
7	Room 919, 51 Southwest First	Tennessee	Do. All except District 8.		
-	Ave., Miami, Fla. 33130.	1	•		
8	829 Federal Office Bldg., 600 South St., New Orleans, La.	Alabama	Baldwin and Mobile. All counties.		
	70130.	Florida	Escambia.		
	Suboffice: 439 U.S. Courthouse and Customhouse, Mobile.	Louisiana	All counties.		
. 9	and Customhouse, Mobile, Ala. 36602.	Texas	City of Texarkana only.		
. 8	New Federal Office Bldg., 515 Rusk Ave., Room 5636, Houston, Tex. 77002. Suboffice: 239 Federal Bldg.,	1 (405	Angelina, Aransas, Atascosa, Austin, Bandera, Bastrop, Bee, Bexar, Blanco, Brazoria, Brazos,		
	Suboffice: 239 Federal Bldg.		eron, Chambers, Colorado, Comal, De Witt		
	300 Willow St., Beaumont, Tex. 77701.		Dimmit, Duval, Edwards, Fayette, Fort		
	161. 77701.		Gonzales, Grimes, Guadalupe, Hardin, Harris		
			Bastrop, Bee, Bexar, Blanco, Brazoria, Brazos, Brooks, Burleson, Caldwell, Calhoun, Cam- eron, Chambers, Colorado, Comal, De Witt, Dimmit, Duval, Edwards, Fayette, Fort Bend, Frio, Galveston, Gillespie, Goliad, Gonzales, Grimes, Guadalupe, Hardin, Harris Hays, Hidalgo, Jackson, Jasper, Jefferson, Jim Hogg, Jim Wells, Karnes, Kendall, Ken- edy, Kerr, Kinney, Kleberg, La Salle, Lavaca, Lee, Liberty, Live Oak, Madison, Matagorda Maverick, McMullen, Medina, Montgomery Nacogdoches, Newton, Nueces, Orange, Polk		
			Lee, Liberty, Live Oar, Madison, Matagorda Maverick, McMullen, Medina, Montgomery Nacogdoches, Newton, Nueces, Orange, Polk Real, Refugio, Sabine, San Augustine, Sar Jacinto, San Patricio, Starr, Travis, Trinity Tyler, Uvalde, Val Verde, Victoria, Walker Waller, Washington, Webb, Wharton, Willacy Williamson, Wilson, Zapata, and Zavala.		
10	1214 Wood St. Doom 707	Oklahoma	Waller, Washington, Webb, Wharton, Willacy Williamson, Wilson, Zapata, and Zavala.		
10	1314 Wood St., Room 707, Dallas, Tex. 75202.	Oklahoma Texas	An counties.		

		Tomiton within district				
Radio district	Address of the Engineer in Charge		Territory within district			
		States	Counties			
11	Room 1758, U.S. Courthouse, 312 North Spring St., Los Angeles, Calif. 90012. Suboffice: Fox Theatre Bldg.,	ArizonaCalifornia	All counties. Imperial, Inyo, Kern, Los Angeles, Orange, Riverside, San Bernardino, San Diego, San			
	1245 Seventh Ave., San	Nevada	Luis Obispo, Santa Barbara, and Ventura. Clark.			
12	323-A Customhouse, 555 Battery St., San Francisco, Calif. 94111.	California Nevada	All except District 11. All except Clark.			
13	314 Multnomah Building, 319 Southwest Pine St., Port- land, Oreg. 97204.	Idaho Oregon Washington	All except District 14. All counties. Clark, Cowlitz, Klickitat, Skamania, and Wahkiakum.			
14	8012 Federal Office Bldg., First Ave. and Marion, Seattle, Wash. 98104.	Idaho Montana	Benewah, Bonner, Boundary, Clearwater, Idaho Kootenai, Latah, Lewis, Nez Perce, Shoshone.			
15	504 New Customheuse, Denver, Colo. 80202.	Washington Celorade Utah Wyoming	All except District 13. All counties. Do.			
		Nebraska New Mexico South Dakota	Banner, Box Butte, Cheyenne, Dawes, Deuel, Garden, Kimball, Morrill, Scotts Bluff, Sheridan, Sioux. All counties. Butte, Custer, Fall River, Lawrence, Meade, Pennington, Shannon, Washabaugh.			
16	691 Federal Building and U.S. Courthouse, Fourth and Robert Sts., St. Paul, Minn. 55101.	Minnesota Michigan	Pennington, Shannon, Washaosugh. All counties. Alger, Baraga, Chippewa, Delta, Dickinson, Gogebic, Houghton, Iron, Keweenaw, Luce, Mackinsc, Marquette, Menominee, Onto- nagon, and Schoolcraft.			
		South Dakota North Dakota Wisconsin	nagon, and Schoolcraft. All counties except District 15. All counties. All counties except District 18.			
17	1703 Federal Bldg., 601 East 12th St., Kansas City, Mo. 64106.	Iowa Kansas Missouri Nebraska	All except District 18. All counties. Do.			
18	1872 New U.S. Courthouse and Federal Office Bidg., 219 South Dearborn St., Chi- cago, Ill. 60604.	Illinois Indiana Iowa	All counties. Do. Allamakee, Buchanan, Cedar, Clayton, Clinton, Delaware, Des Moines, Dubuque, Fayette, Henry, Jackson, Johnson, Jones, Lee, Linn, Louisa, Muscatine, Scott, Washington, and			
		Wisconsin	Brown, Calumet, Columbia, Crawford, Dane, Dodge, Door, Fond du Lac, Grant, Green, Iowa, Jefferson, Kenceha, Kewaunee, Lafayette, Manitowoc, Marinette, Milwaukee, Oconto, Outagamie, Ozaukee, Racine, Richland, Rock, Sauk, Sheboygan, Walworth, Washington, Waukesha, and Winnebago. All counties except District 19. Bath Ball Boone Bourbon, Boyd, Bracken.			
19	1029 New Federal Bldg., Detroit, Mich. 48226.	Kentucky	Breathitt, Campbell, Carter, Clark, Clay, Elliott, Estill, Fayette, Fleming, Floyd, Franklin, Gallatin, Garrad, Grant, Greenup, Harlan, Harrison, Jackson, Jessamine, Johnson, Kenton, Knott, Knox, Laurel, Lawrence, Lee, Lesile, Letcher, Lewis, Lincoln, Madison, Magoffin, Martin, Mason, McCreary, Menifee, Montgomery, Morgan, Nicholas, Owen, Owsley, Pendleton, Perry, Pike, Powell, Pulaski, Robertson, Rockcastle, Rowan, Scott, Wayne, Whitley, Wolf, and Woodford.			
		West Virginia	All counties except District 16. All counties except District 4.			
20 21	328 Federal Bldg., Buffalo, N.Y. 14203. 502 Federal Bldg., Post Office Box 1021, Honolulu, Hawaii	New York Pennsylvania Hawaii and outlying Pacific possessions.	All except District 2. All except District 3.			
22	96808. Post Office Box 2987, 322-323 Federal Bldg., San Juan,	Puerto Rico				
23	P.R. 00903. Post Office Box 644, Room 53, U.S. Post Office and Court-	Alaska				
64	house Bldg., Anchorage, Alaska 99501.	District of Columbia				
24	Room 216, 1919 M St. NW., Washington, D.C. 20554.	District of Columbia and 10 miles beyond the boundary of the District of Colum- bia in each direction.				

- (b) The Marine Offices are located at the following addresses:
- Federal Communications Commission, 738 Federal Office Building, 500 Zack Street, Tampa, Fla. 33602.
- Federal Communications Commission, Post Office Box 3009, Terminal Island, 300 South Ferry Street, Room 2525, San Pedro, Calif. 90731.
- (c) The Class A monitoring stations are located at the following addresses:
- Federal Communications Commission, Allegan Monitoring Station, Post Office Box 89, Allegan, Mich. 49010.
- Federal Communications Commission, Anchorage Monitoring Station, Post Office Box 6303 Annex, Anchorage, Alaska 99502.

Federal Communications Commission, Canandaigqua Monitoring Station, Post Office Box 374, Canandaigqua, N.Y. 14424.

Federal Communications Commission, Fort

Lauderdale Monitoring Station, Post Office Box 22836, Fort Lauderdale, Fla. 33315. Federal Communications Commission, Grand Island Monitoring Station, Post Office Box 788, Grand Island, Nebr. 68801.

Federal Communications Commission, Kingsville Monitoring Station, Post Office Box 632, Kingsville, Tex. 78363.

Federal Communications Commission, Laurel Monitoring Station, Post Office Box 40, Laurel, Md. 20810.

Federal Communications Commission, Livermore Monitoring Station, Post Office Box 311, Livermore, Calif. 96551.

311, Livermore, Calif. 96551.

Federal Communications Commission, Powder Springs Monitoring Station, Post Office Box 85, Powder Springs, Ga. 30073.

Federal Communications Commission, Santa Ana Monitoring Station, Post Office Box 5126, Santa Ana, Calif. 92704.

Federal Communications Commission, Waipahu Monitoring Station, Post Office Box 1035, Waipahu, Hawaii 96797.

- (d) The Class B monitoring stations are located at the following addresses:
- Federal Communications Commission, Ambrose Monitoring Station, Post Office Box 1126, Denison, Tex. 75020.
- Federal Communications Commission, Chillicothe Monitoring Station, Post Office Box 251, Chillicothe, Ohio 45601.
- Federal Communications Commission, Douglas Monitoring Station, Post Office Box 6, Douglas, Ariz. 85607.
- Federal Communications Commission, Spokane Monitoring Station, Post Office Box 191, Spokane, Wash. 99210.
- (e) The Class C monitoring stations are located at the following addresses:
- Federal Communications Commission, Marietta Monitoring Station, Post Office Box 339, Bellingham, Wash. 98225.
- Federal Communications Commission, Wahiawa Monitoring Station, Post Office Box 529, Wahiawa, Hawali 96786.

- Federal Communications Commission, Winter Harbor Monitoring Station, Post Office Box 64, Prospect Harbor, Maine 04669.
- Federal Communications Commission, Imperial Beach Monitoring Station, Post Office Box 1087, Imperial Beach, Calif. 92032.
- Federal Communications Commission, Post Office Box 181, Sabana Seca, P.R. 00749.
- (f) An examination office for amateur operator license applicants is located at 334 York Street, Gettysburg, Pa.

[28 F.R. 12392, Nov. 22, 1963, as amended at 39 F.R. 8837, July 14, 1965; 32 F.R. 20859, Dec. 28, 1967; 34 F.R. 5656, Mar. 26, 1969; 34 F.R. 6480, Apr. 15, 1969; 34 F.R. 14690, Sept. 23, 1969; 34 F.R. 18123, Nov. 11, 1969]

SAFETY AND SPECIAL RADIO SERVICES BUREAU

§ 0.131 Functions of the Bureau.

The Safety and Special Radio Services Bureau develops, recommends, and administers policies and programs for the development and regulation of the Safety and Special Radio Services. These services include nationwide and international uses of radio by persons, businesses, state and local governments, and other organizations licensed to operate their own communication systems for their own use as an adjunct of their primary business or other activity. This program includes, among others, (1) the compulsory use of radio for safety at sea purposes, and (2) the regulation, jointly with the Common Carrier Bureau, of certain classes of radio stations which render communication service for hire. The Bureau performs the following functions:

- (a) Advises and makes recommendations to the Commission and acts for the Commission in matters pertaining to the regulation and development of the Safety and Special Radio Services. These matters include: Rule making, waivers of rules, action on applications for authorizations, adjudicative hearings, enforcement activities, legislation, and defense matters.
- (b) Participates in treaty activities and all phases of international conferences concerning the Safety and Special Radio Services.
- (c) Conducts studies of frequency requirements in the Safety and Special Radio Services; recommends allocations of frequencies and drafts frequency assignment plans for these services.

(d) Studies technical requirements for equipment for the Safety and Special Radio Services in accordance with standards established by the Chief Engineer.

(e) Collaborates and coordinates with Federal and State Government agencies in matters involving the Safety and

Special Radio Services.

(f) Advises and assists members of industry and user groups interested in the Safety and Special Radio Services. [31 F.R. 6832, May 7, 1966]

§ 0.132 Units in the Bureau.

The detailed operations of the Bureau are performed within five major units, as follows:

- (a) Legal, Advisory, and Enforcement Division:
- (b) Industrial and Public Safety Rules Division:
- (c) Industrial and Public Safety Facilities Division;
- (d) Aviation and Marine Division;
- (e) Amateur and Citizens Division.[31 F.R. 6832, May 7, 1966]

OFFICE OF HEARING EXAMINERS

§ 0.151 Functions of the Office.

The Office of Hearing Examiners consists of a Chief Hearing Examiner, and Assistant Chief Hearing Examiner, and as many other Hearing Examiners qualified and appointed pursuant to the requirements of section 11 of the Administrative Procedure Act as the Commission may find necessary. It is responsible for hearing and conducting all adjudicatory cases designated for any evidentiary adjudicatory hearing other than those designated to be neard by the Commission en banc or by one or more members of the Commission, and for such other hearings as the Commission may assign.

§ 0.152 Responsibilities of the Chief Hearing Examiner.

The Chief Hearing Examiner and, in his absence, an Acting Chief Hearing Examiner appointed by the Chief Hearing Examiner, have the following administrative responsibilities:

(a) To coordinate and supervise administratively the activities of the Office of Hearing Examiners.

(b) To prepare and maintain hearing calendars, showing the time and place of the commencement of hearings.

(c) To receive, correlate and approve recommendations of the Hearing Ex-

aminers of the Commission on matters relating to changes in rules and regulations governing hearing procedures, and recommend to the Commission desirable changes in said rules and regulations to promote the simplified and expeditious conduct of proceedings.

(d) To require and prepare reports, statistical data and other information requested or required by the U.S. Civil Service Commission or other offices or agencies of the U.S. Government concerned with the proper operation of the

Office of Hearing Examiners.

(e) To serve, upon instruction of the Commission or the Chairman, as liaison for the Commission and the Office of Hearing Examiners in the making of appropriate arrangements for the securing of advice or information from representatives of other agencies, bar associations, and other interested persons in connection with the formulation and improvement of administrative procedures and practices applicable to the Commission's proceedings.

(f) To exercise such authority as may be assigned to him by the Commission pursuant to section 5(d) of the Commu-

nications Act. as amended.

REVIEW BOARD

§ 0.161 Functions of the Board.

The Review Board is a permanent body with continuing functions, composed of three or more Commission employees designated by the Commission. The Board reviews initial decisions and other hearing matters referred to it by the Commission, takes original action on certain interlocutory matters which arise during the course of hearing proceedings, and performs such additional duties not inconsistent with these functions as may be assigned to it by the Commission.

OFFICE OF OPINIONS AND REVIEW

§ 0.171 Functions of the Office.

The Office of Opinions and Review is composed of legal, engineering, accounting, and other personnel and is headed by a Chief. It assists and makes recommendations to the Commission, and to individual commissioners designated to review initial decisions, in the disposition of matters arising in cases of adjudication (as defined in the Administrative Procedure Act and such other cases as by Commission policy are handled in the same manner) which have been designated for hearing.



DEFENSE AND EMERGENCY PREPAREDNESS FUNCTIONS

§ 0.181 The Defense Commissioner.

A Defense Commissioner and two Alternate Defense Commissioners are designated by the Commission. The Defense Commissioner directs the defense activities of the Commission and has the following duties and responsibilities:

(a) To keep the Commission informed as to significant developments in the field of emergency preparedness, defense mobilization, and any defense activities that involve formulation or revision of Commission policy in any area of responsibility of the Commission.

(b) To represent the Commission in national defense matters requiring conferences or communications with other governmental officers, departments, or

agencies.

- (c) To act as the Defense Coordinator in representations with other agencies with respect to planning for the continuity of the essential functions of the Commission under national emergency conditions, and to serve as the principal representative of the Commission to the Interagency Emergency Planning Committee of the Office of Emergency Planning.
- (d) To serve as the principal representative of the Commission to the Interagency Civil Defense Committee of the Office of Civil Defense, Department of the Army.
- (e) To serve as the principal point of contact for the Commission on all matters pertaining to the National Communications System.
- (f) To take such measures as will assure continuity of the Commission's functions under any foreseeable circumstances with a minimum of interruption.
- (g) In the event of enemy attack, or the imminent threat thereof, or other disaster resulting in the inability of the Commission to function at its offices in Washington, D.C., to assume all of the duties and responsibilities of the Commission and the Chairman, until relieved or augmented by other Commissioners or members of the staff, as set forth in §§ 0.186 and 0.383.
- (h) To perform such other duties and assume such other responsibilities related to the Commission's defense activities as may be necessary for the continuity of functions and the protection of

personnel and property of the Commission

[29 F.R. 14664, Oct. 28, 1964]

§ 0.182 Executive Director.

Under the general direction of the Defense Commissioner, and with the advice and assistance of the heads of the several bureaus and offices, the Executive Director coordinates the defense activities of the Commission, and has the following duties and responsibilities.

- (a) To act as Alternate Defense Coordinator in representations with other agencies with respect to planning for the continuity of the essential functions of the Commission under national emergency conditions, and to serve as the alternate representative of the Commission to the Interagency Emergency Planning Committee of the Office of Emergency Planning.
- gency Planning.

 (b) To serve as the alternate representative of the Commission to the Interagency Civil Defense Committee of the Office of Civil Defense, Department of the Army.
- (c) To keep the Defense Commissioner informed as to significant developments in this area.

[29 F.R. 14664, Oct. 10, 1964]

§ 0.183 Office of Emergency Communications.

- (a) The Chief of the Office of Emergency Communications plans, directs and coordinates the functions of the Office. The Office of Emergency Communications under the supervision and direction of the Executive Director and in collaboration with, and with the concurrence of the responsible Bureau Heads and Staff Officers, develops, prepares, and recommends to the Defense Commissioner national emergency plans and develops preparedness programs covering (See Executive Orders 10312, 10346, as amended, 11007, 11092, and 11179):
- (1) Provision of service by common carriers, broadcasting facilities, and the safety and special radio services under national emergency conditions;
- (2) Assignment of radio frequencies to Commission licensees under national emergency conditions:
- (3) Preparation of data with respect to facilities operated by the non-government communications industry for use by the National Resources Evaluation Center;
- (4) Control of radio stations in an emergency, except those belonging to, or

operated by, any department or agency of the United States Government;

- (5) Investigations of violations of pertinent law and regulations in an emergency, and development of procedures designated to initiate, recommend, or otherwise bring about the appropriate enforcement actions required in the interest of national security:
- (6) Provision of financial, credit, or other assistance to common carriers and Commission licensees who might need such assistance in various conditions of mobilization, particularly those resulting from attack;
- (7) Stimulation of the development by common carriers and by licensees, of standby plans for the conservation and salvage of supplies and equipment and the rehabilitation, restoration, or replacement of essential communication facilities after an attack;
- (8) Preparation, as claimant agency for the non-government communications industry, to claim materials, manpower, equipment, supplies and services needed in support of the common carriers, and Commission licensees from the appropriate resource agencies, and work with such agencies in developing programs to insure availability of such resources in an emergency;
- (9) Provision of advice and guidance to achieve such industry protection as is necessary to maintain the integrity of the facilities and services provided by common carriers and radio station licensees, and promote a national program to stimulate disaster preparedness and damage control, including, but not limited to, organizing and training facility employees, personnel shelter, evacuation and relocation plans, records protection, continuity of management, security, emergency repair and recovery of facilities, deconcentration and dispersal of facilities, and mutual aid associations for emergency;
- (10) Development and maintenance of a capability to assess the effects of attack on communication facilities and services subject to Commission regulation, which are essential in a national emergency, and provide data to the Department of Defense; and
- (11) Fostering or conducting research in areas directly concerned with the Commission's emergency preparedness responsibilities;
- (b) The Office of Emergency Communications, in collaboration with, and with the concurrence of, Bureau Heads and

- Staff Officers, prepares plans for the continuity of Government functions of the Commission in the event of a national emergency (except for certain designated functions of the Field Engineering Bureau, as set forth in § 0.111) as prescribed by Executive Order 10346, as amended, including plans for emergency mobilization of the Commission's personnel; positioning, maintenance, and protection of supplies, material and essential records; and selection, training. transportation and emergency assignment, notification and funding arrangements for the Commission's Executive Reserve Unit of the National Defense Executive Reserve Program under the provisions of Executive Order 11179 (except for certain designated functions of the Field Engineering Bureau as set forth in § 0.111).
- (c) The Office of Emergency Communications provides the Executive Secretariat for the National Industry Advisory Committee, its sub-committees at the Regional, State and Local level, and special working groups as may be formed for specific purposes by the Defense Commissioner.
- (d) The Office of Emergency Communications is composed of the Office of the Chief, including the NIAC Secretariat and the FCC Mobilization Planning Officer, the Emergency Communication Systems Division, and the Emergency Communication Resources Plans Division, with functions as set forth in paragraphs (e), (f), and (g) of this section.
- (e) Emergency Communication Systems Division: The Emergency Communication Systems Division develops and recommends plans and procedures for the following:
- (1) Broadcasting service. Construction, activation, deactivation of broadcasting facilities and services, the continuation or suspension of broadcasting services and facilities, and issuance of appropriate authorizations for such facilities, services, and personnel in an emergency.
- (2) Safety and special radio services. Authorization, operation, and use of specified safety and special radio services, facilities, and personnel in the national interest in an emergency.
- (3) Electromagnetic radiation. Control of radio stations in an emergency, except for those belonging to, or operated by, any department or agency of the United States Government.

- (4) Facilities protection. Provide advice and guidance to achieve such industry protection as is necessary to maintain the integrity of the facilities and station licensees and promote a national program to stimulate disaster preparedness and damage control. The guidance shall include, but not be limited to, organizing and training facility employees, personnel shelter, evacuation and relocation plans, records protection, continuity of management, security, emergency repair and recovery of facilities, deconcentration and dispersal of facilities, and mutual aid associations for emergency.
- (5) Research. Foster or conduct research in areas directly concerned with the Commission's emergency preparedness responsibilities.
- (f) Emergency Communication Resources Plans Division: (1) The Emergency Communication Resources Plans Division develops and recommends plans and procedures for the following:
- (i) Common carrier service. (a) Extension, discontinuance, or reduction of common carrier facilities or services, and issuance of appropriate authorizations for such facilities, services, and personnel in an emergency; and control of all rates, charges, practices, classifications, and regulations for service to Government and non-Government users during an emergency, in consonance with national economic stabilization policies.
- (b) Development and administration of priority systems for public correspondence and for the use and resumption of leased inter-city private line service in an emergency.
- (c) Use of common carrier facilities and services to overseas points to meet vital needs in an emergency.
- (ii) Safety and special radio services. Authorization, operation and use of specified safety and special radio services, facilities, and personnel in the national interest in an emergency.
- (iii) Radio frequency assignment. Assignment of radio frequencies to, and their use by, Commission licensees in an emergency.
- (iv) Resource data. Provision of assistance and data with respect to facilities operated by Commission licensees for national resource evaluation purposes of the Office of Emergency Planning.
- (v) Financial aid. Provision of financial, credit, or other assistance to common carriers and Commission licensees who might need such assistance in vari-

ous conditions of mobilization, particularly those resulting from attack.

(vi) Conservation, salvage, and rehabilitation. Stimulation of the development, by common carriers and by licensees, of standby plans for the conservation and salvage of supplies and equipment and the rehabilitation, restoration, or replacement of essential communication facilities after an attack.

(vii) Claimancy. Preparation, as claimant agency for the non-Government communications industry, to claim materials, manpower, equipment, supplies, and services needed in support of the common carriers and Commission licenses from the appropriate resource agencies, and insure availability of such resources in an emergency.

(viii) Damage assessment. Maintain a capability to assess the effects of attack on communication facilities and services subject to Commission regulation which are essential in a national emergency, and provide data to the Department of Defense.

(2) The Emergency Communication Resources Plans Division coordinates and directs the activities of the Commission's Damage Assessment Unit, and coordinates the activities of the Commission's Resource Data Officer assigned to the National Resources Evaluation Center.

(3) The Emergency Communication Resources Plans Division conducts the review of plans developed by the State Task Groups and the Commission's representatives to the Regional Preparedness Committees of the Office of Emergency Planning.

- (g) FCC Mobilization Planning Officer: (1) The FCC Mobilization Planning Officer, pursuant to the provisions of Executive Order 10346, as amended, and in consultation with the Bureau Heads and Staff Officers of the Commission, has responsibility for the preparation of plans for recommendation to the Defense Commissioner for maintenance of the Commission's essential functions at the seat of Government, and elsewhere, during a national emergency (except for certain designated functions of the Field Engineering Bureau as set forth in § 0.111) including the following specific functions:
- (i) Prepares detailed plans for the emergency mobilization, assignment and utilization of present Commission personnel and augmentation staff, including emergency notification, transporta-

tion, and continuity of funding arrangements.

(ii) Makes detailed advance arrangements for selection, positioning, maintenance, and protection of supplies and materials required by emergency relocation personnel for the execution of the plans for continuity of the Commission's essential functions in the event of a national emergency.

(iii) Prepares recommendations for the selection, positioning, maintenance, and protection of essential records for the performance of essential functions of the Commission in the event of a na-

tional emergency.

(iv) Prepares plans for the selection, training, emergency assignment, emergency notification, transportation, and emergency funding arrangements for personnel in the Commission's Executive Reserve Unit of the National Defense Executive Reserve Program under the provisions of Executive Order 11179.

(v) Selects and prepares detailed plans for training, emergency assignment, emergency notification, and transportation of the Emergency Task Forces assigned to the Special Facilities Division of the Office of Emergency Planning and to the Commission's Principal Emergency Relocation Facility.

(vi) In cooperation with the Chief, Emergency Communication Systems Division, and the Chief, Emergency Communication Resources Plans Division, makes detailed plans for appropriate investigation, enforcement and carrying out by Emergency Task Forces of all approved emergency preparedness plans (except for certain functions of the Field Engineering Bureau) under the provisions of Executive Orders 11092 and 10312.

(vii) Prepares plans for and conducts tests and exercises (except for certain activities of the Field Engineering Bureau as set forth in § 0.111) to determine the state of operational readiness of the Commission and its emergency facilities to execute its essential emergency functions and emergency mobilization plans in the event of a national emergency; and performs such other duties related to the Commission's continuity of Government activities and responsibilites as directed.

(viii) Prepares data and information with respect to continuity of the essential functions of the Commission in a national emergency (except for the Field Engineering Bureau as set forth in § 0.-

- 111), for use by the Defense Coordinator and Alternate Defense Coordinator at Interagency Emergency Planning Committee meetings, as directed, and prepares initial drafts of documents to implement any required actions as a result of such meetings.
- (h) Executive Secretariat of the National Industry Advisory Committee: The Executive Secretary of the National Industry Advisory Committee provides secretariat service to the National Industry Advisory Committee, and its subcommittees at the Regional, State and Local level and special working groups as may be formed for specific purposes by the Defense Commissioner, in accordance with the provisions of Executive Order 11007, as follows:
- (1) Preparation of appointments to Committees or Working Groups as directed by the Defense Commissioner and maintenance of membership rosters, records and files of all committees and requests for appropriate security clearance for selected members.
- (2) Preparation, certification, and appropriate distribution of agendas and associated material for meetings of all committees.
- (3) Arrangements for preparation of appropriate minutes of meetings, and distribution thereof.
- (4) Preparation of interagency coordination of National Industry Advisory Committee recommendations as directed.
- (5) Coordinates preparation of the required agenda items for consideration by the Commission to effect implementation of National Industry Advisory Committee recommendations.

[29 F.R. 15758, Nov. 24, 1964]

- § 0.184 Regional Liaison Officers of the Office of Executive Director, Emergency Communications Division.
- (a) Field Liaison officers of the Office of Emergency Communications (OEC) are designated as FCC representatives to Regional Civil Defense Coordinating Boards of the Office of Civil Defense, Department of the Army (DOA). (Engineers in Charge of Field Engineering Bureau (FEB) district offices at Boston, Massachusetts; Baltimore, Maryland; Atlanta, Georgia; Detroit, Michigan; Dallas, Texas; Denver, Colorado; San Francisco, California; and Seattle, Washington are designated as alternates.)
- (b) Field Liaison officers of the OEC are designated to serve as alternate FCC



representatives to the OEP Regional Preparedness Committees. (The Engineers in Charge listed in paragraph (a) of this section are principal representatives.)

(c) The field offices of the Office of emergency Communications are located at the following addresses:

OCD/ OEP Region	Address of the FCC Field Liaison Officer	States
1	OCD Region One, Oak Hill Road, Harvard, Mass., 01451.	Connecticut, Maine, Massachusetts, New Hampshire, New Jersey, New York, Rhode Island,
2	OCD Region Two, Olney, Md., 20632.	Vermont. Delaware, District of Columbia, Ken- tucky, Maryland, Ohio, Pennsylvania, Virginia, West Virginia.
3	OCD Region Three, Thomasville, Ga., 31792.	Alabama, Florida, Georgia, Mississippi, North Carolina, South Carolina, Tennessee, Virgin Islands, Puerto Rico.
4	OCD Region Four, Federal Center, Battle Creek, Mich., 49016.	Illinois, Indiana, Michigan, Minne- sota, Wisconsin.
5	OCD Region Five, Denton Federal Center, Denton, Tex., 76201.	Arkansas, Louisiana, New Mexico, Oklahoma, Texas.
6	OCD Region Six, Denver Federal Center, Bldg. 50, Denver, Colo., 80225.	Colorado, Iowa, Kansas, Missouri, Nebraska, North Dakota, South Dakota, Wyoming.
7	OCD Region Seven, Post Office Box 1300, Santa Rosa, Calif., 95403.	American Samoa, Arizona, California, Guam, Hawaii, Nevada, Utah.
8	OCD Region Eight, Everett, Wash., 98201.	Alaska, Idaho, Oregon, Montana, Washington.

FCC Field Supervisor, Eastern United States (OCD Regions 1, 2, 3, 4) located at OCD Region 3, Thomas-

ville, Ga.
FCC Field Supervisor, Western United States (OCD Regions 5, 6, 7, 8) located at OCD Region 7, Santa Rosa, Calif.

(d) The FCC Field Liaison Officers listed in paragraph (c) of this section are designated as the FCC representatives to attend and supervise all Regional and State Industry Advisory Committee meetings within their respective Regions and States pursuant to the requirements of Executive Order 11007.

[29 F.R. 14665, Oct. 28, 1964, as amended at 32 F.R. 20860, Dec. 28, 1967]

§ 0.185 Responsibilities of the bureaus and staff offices.

The heads of each of the several bureaus and staff offices, in rendering advice and assistance to the Executive Director in the performance of his duties with respect to defense activities will have the following duties and responsibilities:

(a) To keep the Executive Director informed of the instigation, progress, and completion of programs, plans, or activities with respect to defense in which they are engaged or have been requested to engage.

(b) To render assistance and advice to the Executive Director on matters which relate to the functions of their respective

bureaus or staff offices.

(c) To render such assistance and advice to other agencies as may be consistent with the functions of their respective bureaus or staff offices and the Commission's policy with respect thereto.

(d) To perform such other duties related to the Commission's defense activities as may be assigned to them by the

Commission.

[29 F.R. 14665, Oct. 28, 1964]

§ 0.186 Emergency Relocation Board.

- (a) An Emergency Relocation Board, to be convened by the Commission's relocation headquarters, performs the functions of the Commission in the event of the inability of the Commission to function at its offices in Washington, D.C., resulting from disaster or the threat of imminent disaster from enemy attack or from natural causes, either:
- (1) At the time of or under circumstances specified in a directive from the President.
- (2) In the absence thereof, upon receipt of a warning signal indicating that an attack on the capital is likely.
- (3) In the absence of either a directive or warning signal, immediately following an actual attack.
- (b) The Board shall comprise such Commissioners as may be present and able to act or, if no Commissioner is present and able to act, the occupant of the following positions, in the order listed, who is present and able to act:
- (1) The Chief, Field Engineering Bureau.
 - (2) The General Counsel.
 - (3) The Chief Engineer.
- (4) The Chief, Safety and Special Radio Services Bureau.
 - (5) The Chief, Broadcast Bureau.
- (6) The Chief, Common Carrier Bureau.
 - (7) The Executive Director.

- (8) The Deputy Chief, Field Engineering Bureau.
 - (9) The Deputy General Counsel.
 - (10) The Associate Chief Engineer.
- (11) The Assistant Chief, Safety and Special Radio Services Bureau.
- (12) An Assistant Chief, Broadcast Bureau.
- (13) The Associate Chief, Common Carrier Bureau.
- (14) The Chief of Division, ranking in the same order as indicated in subparagraphs (1) to (7) of this paragraph. [28 F.R. 12392, Nov. 22, 1963, as amended at 29 F.R. 15923, Nov. 28, 1964]

Subpart B—Delegations of Authority

AUTHORITY: §§ 0.201 to 0.387 also issued under sec. 5, 48 Stat. 1068, as amended; 47 U.S.C. 155.

SOURCE: The provisions of this Subpart B appear at 28 F.R. 12402, Nov. 22, 1963, unless otherwise noted.

GENERAL

§ 0.201 General provisions.

- (a) There are three basic categories of delegations made by the Commission pursuant to section 5(d) of the Communications Act of 1934, as amended:
- (1) Delegations to act in non-hearing matters and proceedings. The great bulk of delegations in this category are made to bureau chiefs and other members of the Commission's staff. This category also includes delegations to individual commissioners and to boards or committees of commissioners, such as the Telegraph and Telephone Committees.
- (2) Delegations to rule on interlocutory matters in hearing proceedings. Delegations in this category are made to the Review Board and to the Chief Hearing Examiner.

NOTE: Interlocutory matters which are delegated neither to the Review Board nor to the Chief Hearing Examiner are ruled on by the presiding officer by virtue of the authority vested in him to control the course and conduct of the hearing. This authority stems from section 7 of the Administrative Procedure Act and section 409 of the Communications Act rather than from delegations of authority made pursuant to section 5(d) of the Communications Act. (See §§ 0.218 and 0.341.)

(3) Delegations to review an initial decision. Delegations in this category

- are made to individual commissioners, to panels of commissioners, or to the Review Board.
- (b) Delegations are arranged in this subpart under headings denoting the person, panel, or board to whom authority has been delegated, rather than by the categories listed in paragraph (a) of this section.
- (c) Procedures pertaining to the filing and disposition of interlocutory pleadings in hearing proceedings are set forth in §§ 1.291-1.298 of this chapter. Procedures pertaining to appeals from rulings of the presiding officer are set forth in § 1.301. Procedures pertaining to reconsideration of the presiding officer's rulings are set forth in § 1.303. Procedures pertaining to reconsideration and review of actions taken pursuant to delegated authority are set forth in §§ 1.101, 1.102, 1.104, 1.106, 1.113, 1.115, and 1.117. Procedures pertaining to exceptions to initial decisions are set forth in §§ 1.276-1.279.
- (d) The Commission, by vote of a majority of the members then holding office, may delegate its functions either by rule or by order, and may at any time amend, modify, or rescind any such rule or order.
- (1) Functions of a continuing or recurring nature are delegated by rule. The rule is published in the FEDERAL REGISTER and is included in this subpart.
- (2) Functions pertaining to a particular matter or proceeding are delegated by order. The order is published in the Federal Register and associated with the record of that matter or proceeding, but neither the order nor any reference to the delegation made thereby is included in this subpart.
- § 0.203 Authority of person, panel, or board to which functions are delegated.
- (a) The person, panel, or board to which functions are delegated shall, with respect to such functions, have all the jurisdiction, powers, and authority conferred by law upon the Commission, and shall be subject to the same duties and obligations.
- (b) Except as provided in § 1,102 of this chapter, any action taken pursuant to delegated authority shall have the same force and effect and shall be made, evidenced, and enforced in the same manner as actions of the Commission.

§ 0.204 The exercise of delegated authority.

(a) Authority to issue orders and to enter into correspondence. Any official (or group of officials) to whom authority is delegated in this subpart is authorized to issue orders (including rulings, decisions, or other action documents) pursuant to such authority and to enter into general correspondence concerning any matter for which he is responsible under this subpart or subpart A of this part.

(b) Authority of subordinate officials. Authority delegated to any official to issue orders or to enter into correspondence under paragraph (a) of this section may be exercised by that official or by appropriate subordinate officials acting

for him.

(c) Signature.

(1) A final decision of the Review Board is signed by the Board member responsible for its preparation.

(2) Other orders made by a committee, board or panel identify the body and are signed by the Secretary.

(3) Upon signing an order, the Secretary affixes the Commission's seal.

(4) General correspondence by a committee or board is signed by the committee or board chairman.

(5) All other orders and letters are signed by the official who has given final

approval of their contents.

- (6) With the exception of certain license forms also signed by the issuing Engineer in Charge, license forms bear only the signature of the Secretary and the seal of the Commission.
- (d) Form of orders. Orders may be issued in any appropriate form (e.g., as captioned orders, letters, telegrams) and may, if appropriate, be issued orally. Orders issued orally shall, if practicable, be confirmed promptly in writing.
- (e) Minutes entries. Except as otherwise provided in this subpart, actions taken as provided in paragraph (d) of this section shall be recorded in writing and filed in the official minutes of the Commission.

[33 F.R. 8227, June 1, 1968]

COMMISSIONERS

§ 0.211 Chairman.

The responsibility for the general administration of internal affairs of the Commission is delegated to the Chairman of the Commission. The Chairman will keep the Commission advised concerning

his actions taken under this delegation of authority. This authority includes:

(a) Actions of routine character as to which the Chairman may take final action.

- (b) Actions of non-routine character which do not involve policy determinations. The Chairman may take final action on these matters but shall specifically advise the Commission on these actions.
- (c) Actions of an important character or those which involve policy determinations. In these matters the Chairman will develop proposals for presentation to the Commission.
- (d) To act within the purview of the Federal Tort Claims Act, as amended, 28 U.S.C. § 2672, upon tort claims directed against the Commission where the amount of damages does not exceed \$5,000.

[28 F.R. 12402, Nov. 22, 1963, as amended at 31 F.R. 11756, Sept. 8, 1966; 32 F.R. 20860, Dec. 28, 1967]

§ 0.212 Board of Commissioners.

(a) Whenever the Chairman or Acting Chairman of the Commission determines that a quorum of the Commission is not present or able to act, he may convene a Board of Commissioners. The Board shall be composed of all Commissioners present and able to act.

(b) The Board of Commissioners is authorized to act upon all matters normally acted upon by the Commission

en banc, except the following:

(1) The final determination on the merits of any adjudicatory or investigatory hearing proceeding or of any rule making proceeding, except upon a finding by the Board that the public interest would be disserved by waiting the convening of a quorum of the Commission.

(2) Petitions for reconsideration of

Commission actions.

(3) Applications for review of actions taken pursuant to delegated authority.

(c) The Board of Commissioners is authorized to act upon all matters normally acted upon by an individual Commissioner (when he or his alternates are not present or able to act) or by a committee of Commissioners (in the absence of a quorum of the committee).

(d) Actions taken by the Board of Commissioners shall be recorded in the same manner as actions taken by the

Commission en banc.

(e) This section has no application in circumstances in which the Commission

is unable to function at its offices in Washington, D.C. See §§ 0.181-0.186 and 0.381-0.387.

[30 F.R. 9814, July 27, 1965]

§ 0.214 Telegraph Committee.

A Telegraph Committee, composed of three Commissioners, designated as such by the Commission or a majority thereof, will act, except as otherwise ordered by the Commission, upon all applications or requests submitted under sections 214 or 319 of the Communications Act of 1934, as amended, by carriers engaged principally in record communication, for certificates or authorization for the construction, acquisition, operation, or extension of telegraph wire, cable, or radio facilities, for temporary or emergency telegraph service, for supplementing existing telegraph facilities, or for discontinuance, reduction, or impairment of telegraph service, except those covered by §§ 0.291 and 0.294. [28 F.R. 12402, Nov. 22, 1963, as amended at 33 F.R. 8228, June 1, 1968]

§ 0.215 Telephone Committee.

A Telephone Committee, composed of three Commissioners, designated such by the Commission, or a majority thereof, will act, except as otherwise ordered by the Commission, upon all applications or requests submitted under sections 214 or 319 of the Communications Act of 1934, as amended, by carriers engaged principally in telephone communication, for certificates or authorizations for the construction, acquisition, operation, or extension of telephone or telegraph wire, cable, or radio facilities, for temporary or emergency telephone service, for supplementing existing telephone or telegraph facilities, or for discontinuance, reduction, or impairment of telephone or telegraph service, except those covered by §§ 0.291 and 0.294.

[28 F.R. 12402, Nov. 22, 1963, as amended at 33 F.R. 8228, June 1, 1968]

§ 0.216 Delegation to Subscription Television Committee.

(a) A Subscription Television Committee, composed of three Commissioners, designated as such by the Commission, or a majority thereof, will act, except as otherwise ordered by the Commission, upon all requests and other matters pertaining to trial subscription television operations authorized in ac-

cordance with the provisions of the Third Report in Docket No. 11279, with the exception of applications for authority to conduct trial subscription television operations on stations not previously engaged in such operations.

(b) All actions which are taken by the Subscription Television Committee which dispose of requests or application shall be recorded in writing and filed in the official minutes of the Commission. [28 F.R. 13746, Dec. 17, 1963]

§ 0.218 Authority of, and delegated to, an individual Commissioner or Commissioners.

- (a) One or more members of the Commission may be designated to preside in a hearing proceeding. The Commissioner or Commissioners designated to preside at such a hearing shall fix the time and place of the hearing and shall act upon all motions, petitions or other matters which may arise while the proceeding is in hearing status.
- (b) One or more members of the Commission may be designated to review an initial decision issued in any hearing case.
- (c) Except for actions taken during the course of a hearing and upon the record thereof, actions taken by a Commissioner or Commissioners pursuant to the provisions of this section shall be recorded in writing and filed in the official minutes of the Commission.

EXECUTIVE DIRECTOR

§ 0.231 Authority delegated.

- (a) The Executive Director, or his designee, is delegated authority to exempt Standard, FM, and Television broadcast licensees from the requirement of installing and maintaining the necessary equipment to receive Emergency Action Notifications and Terminations and arranging for either an associated listening watch, or automatic alarm, or both. (See § 73.922 of this chapter.)
- (b) The Executive Director, or his designee, upon securing concurrence of the General Counsel, is delegated authority to execute in the name of the Commission all agreements pertaining to the loan of United States Government property to radio station licensees for national defense purposes.
- (c) The Executive Director, or his designee, upon securing concurrence of the General Counsel, is delegated authority to act upon requests for waiver



of the filing fee requirements for modification applications occasioned by natural disasters.

[29 F.R. 14666, Oct. 28, 1964, as amended at 30 F.R. 14106, Nov. 9, 1965]

CHIEF ENGINEER

§ 0.241 Authority delegated to the Chief Engineer.

The Chief Engineer is delegated authority to act upon the following matters which are not in hearing status:

- (a) With respect to type approval of radio frequency equipment and certification of radio frequency equipment as acceptable for licensing, to exercise all functions of the Commission pertaining to the:
- (1) Examination of all applications for type approval of radio frequency equipment as provided for in this chapter; performance of tests, analysis of data, and determination of compliance with the provisions of this chapter; issuance of type approval certificates for radio frequency equipment which complies with the provisions of this chapter; and issuance of lists of type approved equipment.
- (2) Examination of all applications for certification of radio frequency equipment as acceptable for licensing as provided for in this chapter; notification to the applicant that an examination of the certified technical information and data submitted in accordance with the provisions of this chapter indicates that the equipment does or does not appear to be acceptable for licensing in the particular radio service or services involved; and issuance of lists of radio frequency equipment certified as acceptable for licensing.
- (3) Examination of all applications for certification (approval) of subscription television technical systems as acceptable for use under a subscription television authorization as provided for in this chapter; notification of the ap-plicant that an examination of the certified technical information and data submitted in accordance with the provisions of this chapter indicates that the system does or does not appear to be acceptable for authorization as a subscription television system; and issuance of a list of subscription television systems certified as acceptable for authorization. The delegation granted in this subparagraph shall be exercised in consultation with the Chief, Broadcast Bureau.

- (b) With respect to stations operating in the Experimental Radio Service that are not engaged in the development of an established service and administered by a single Bureau, to act upon the following matters:
- (1) Applications to assign frequencies, power, emission, and types of equipment and to approve the apparatus to be employed by these radio stations so as to provide the maximum results from the experimentation which can reasonably be expected of the licensee.

(2) Applications from existing licensees which involve a change in frequency, power, or type of emission.

- (3) Applications which involve only a change in location, type, model, design or number of transmitters.
- (4) Requests by licensees or permittees for cancellation of their station licenses, construction permits or other authorizations.
- (5) Applications for license or modification of license to cover construction permit.
- (6) Applications for renewal of licenses.
- (7) Applications for extension of the expiration date of construction permits.
- (8) For addition, modification, or coordination of programs of research or experimentation so as to provide the maximum results from the experimentation which can be reasonably expected of the licensee or licensees.
- (9) Requests for operation with a modified antenna system where formal application is not required.
- (10) Requests for extension of time within which to comply with technical requirements specified in authorizations, orders, rules, or releases of Commission.
- (11) Representation of compliance with technical requirements specified in authorizations, orders, rules, or releases of the Commission (except formal applications).
- (12) Requests to operate at a temporary location with a temporary antenna system.
- (13) Request for special conditions of operation necessary to comply with technical requirements specified in authorizations, orders, rules, or releases of the Commission.
- (14) Request for special temporary authority in emergency cases, at times outside of the regular office hours of the Commission and requiring immediate action during those hours.

- (15) Request for exemption from station identification requirements.
- (16) Request for exemption from transmitter control requirements.
- (17) Request for exemption from limitation on authorized points of communication.
- (c) To assign new or modified call signs to stations in all of the radio services except the Citizens and Amateur Radio Services. See §§ 0.332(a) and 1.550 of this chapter.

[28 F.R. 12402, Nov. 22, 1963, as amended at 28 F.R. 14502, Dec. 31, 1963; 29 F.R. 15577, Nov. 20, 1964; 34 F.R. 14375, Sept. 13, 1969]

§ 0.243 Authority delegated to the Chief Engineer upon securing concurrence of the General Counsel.

- (a) The Chief Engineer, upon securing concurrence of the General Counsel, is delegated authority with respect to stations operating in the experimental radio services, other than experimental and developmental stations operating in established services under the jurisdiction of a single bureau, to act upon the following matters:
- (1) Applications for construction permits for new stations.
- (2) All requests for withdrawal of papers in accordance with § 1.8 of this chapter.
- (3) Applications for consent to assignment and transfer of control of station authorizations.
- (4) The extension of time previously ordered by the Commission within which transfers of control or assignment of licenses be effectuated.
- (b) The Chief Engineer, upon securing concurrence of the General Counsel, is authorized to determine under § 1.80 of this chapter whether forfeiture liability has been incurred in connection with the operation of any station governed by Part 5 of this chapter, and to issue notices of apparent liability as provided in § 1.80.
- (c) The Chief Engineer, upon securing concurrence of the General Counsel, is delegated authority to act upon requests for waiver of the all-channel television receiver rules contained in Part 15 of this chapter where the receiver involved in such requests is an integral part of a hospital communications system. In such cases, a grant may be made when there is no danger of ad-

- versely affecting the audience potential of present or future UHF television stations, and benefits are to be derived from the grant of the requests.
- (d) The Chief Engineer, upon securing concurrence of the General Counsel, is authorized to dismiss, as repetitious, any petition for reconsideration of a Commission order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order.
- (e) The Chief Engineer, upon securing concurrence of the General Counsel, is authorized to dismiss or deny petitions for rule making which are repetitive or moot or which, for other reasons, plainly do not warrant consideration by the Commission.

[28 F.R. 12402, Nov. 22, 1963, as amended at 29 F.R. 10585, July 20, 1964; 31 F.R. 3074, Feb. 24, 1966; 31 F.R. 4456, Mar. 16, 1966]

§ 0.247 Record of actions taken.

The application and authorization files and other appropriate files of the Office of the Chief Engineer are designated as the official minute entries of actions taken pursuant to §§ 0.241 and 0.243.

[33 F.R. 8228, June 1, 1968]

GENERAL COUNSEL

§ 0.251 Authority delegated.

- (a) [Reserved]
- (b) Insofar as authority is not delegated to any other Bureau or Office, and with respect only to matters which are not in hearing status, the General Counsel is delegated authority to act upon requests for extension of time within which briefs, comments or pleadings may be filed.
- (c) The General Counsel is delegated authority to dismiss, as repetitious, any petition for reconsideration of a Commission order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order.
- (d) The General Counsel is delegated authority to dismiss or deny petitions for rule making which are repetitive or moot or which, for other reasons, plainly do not warrant consideration by the Commission
- [28 F.R. 12402, Nov. 22, 1963, as amended at 31 F.R. 3074, Feb. 24, 1966; 31 F.R. 4456, Mar. 16, 1966; 31 F.R. 11756, Sept. 8, 1966; 31 F.R. 13474, Oct. 19, 1966]

SECRETARY

§ 0.261 Authority delegated.

- (a) The Secretary is delegated authority to make nonsubstantive, editorial revisions of the Commission's rules and regulations upon approval of the bureau or staff office primarily responsible for the particular part or section involved.
- (b) The Secretary is delegated authority, upon securing the approval of the Field Engineering Bureau, to delete or modify, from time to time, as need may appear, the location of radio operator examination points as set forth in § 0.485 and in the appendix to Part 97 of this chapter.

[28 F.R. 12402, Nov. 22, 1963, as amended at 32 F.R. 20860, Dec. 28, 1967; 33 F.R. 8228, June 1, 1968]

CHIEF, BROADCAST BUREAU

§ 0.281 Authority delegated.

The Chief of the Broadcast Bureau is delegated authority to act upon applications, requests, and other matters which are not in hearing status relating to broadcast services as follows:

- (a) Formal applications for broadcast authorizations:
- (1) For construction permits for new or changed standard, FM, noncommercial educational FM, television, television translator, and UHF television booster facilities, or for modification thereof; for licenses or modification thereof; for renewal of licenses for such facilities for the normal license term, which applications comply fully with the requirements of the Communications Act and the provisions of this chapter, accord with Commission policy and standards, are not mutually exclusive with any other application, and concerning which no petition to deny pursuant to § 1.580 of this chapter or other substantial objection has been filed.
- (2) For assignment of broadcast licenses or permits or for transfer of control of broadcast licensees or permittees, which applications comply fully with the requirements of the Communications Act and the provisions of this chapter, accord with Commission policy and standards, and concerning which no petition to deny pursuant to § 1.580 of this chapter or other substantial objection has been filed.
- (3) For new or modified experimental, developmental, and auxiliary broadcast authorizations covered by Part 74 of this

- chapter, or for renewal of authority for such facilities.
- (4) For reinstatement of expired construction permits.
- (5) For new or modified Subsidiary Communications Authorizations, or for renewal of such Authorizations.
- (b) Designate for hearing, upon appropriate issues, mutually exclusive applications for new or modified standard, FM, noncommercial FM, and television facilities.
 - (c) Requests for temporary authority:
- (1) For temporary waiver of technical operating requirements relating to overall system performance or elements thereof and rules requiring that specified items of equipment be employed.
- (2) For operation with temporary antenna system.
- (3) For operation with auxiliary transmitter as main transmitter.
- (4) For operation with new or modified equipment pending repair of existing equipment or pending receipt and action upon a formal application.
- (5) For operation with reduced power or to make other changes in operation of authorized equipment for technical reasons.
- (6) For special operation necessary to facilitate equipment, program and service tests or to comply with technical requirements specified in authorizations, orders, rules or releases of the Commission.
- (7) For operation with licensed, new, or modified equipment at a temporary site with a temporary antenna system when in case of an emergency it becomes impossible to continue operating at the regularly authorized site.
- (8) For special operation of stations in the experimental, developmental and auxiliary broadcast services covered by Part 74 of this chapter.
- (9) For temporary authority in emergency cases, at times outside of the regular office hours of the Commission, which require immediate action during such hours.
- (10) For authority for television broadcast stations to operate with visual-to-aural power other than that specified in this chapter.
- (d) Miscellaneous applications and requests:
- (1) For temporary operation, for a lesser period of time than specified by §§ 73.71, 73.261, and 73.651 of this chapter, or to remain silent for temporary periods.

- (2) For extension of time within which a transfer of control or assignment of license may be effectuated.
- (3) For authority for FM broadcast stations to transmit multiplex facsimile in accordance with § 73.266 of this chapter.
- (4) For authority to rebroadcast when authorization is required under Parts 73 and 74 of this chapter.
- (5) For any permit required by the provisions of section 325(b) of the Communications Act.
- (6) For cancellation of licenses, construction permits or other authorizations.
- (7) For withdrawal of papers in accordance with § 1.8 of this chapter.
- (8) For extension of time within which to file briefs, comments and pleadings in rule making proceedings.
- (9) For conducting equipment and program tests.
- (10) For operation during daytime for specified periods with the nighttime facilities in order to check measurements and operation.
- (11) For extensions of time within which to comply with technical requirements specified in authorizations, orders, and rules or releases of the Commission.
 - (12) For television site survey.(13) For standard broadcast special
- field test authorizations.
- (14) For authority to relocate the main studio outside the corporate limits of the community to which the station is assigned, when no change in station location or identification is involved.
- (15) For waiver of the transmitter inspection requirements imposed by §§ 73.-93(e), 73.265(e), and 73.565(e) of this chapter.
- (16) To waive the provisions of the note to §§ 1.571 and 73.37 of this chapter to the extent necessary to accept for filing an application by an existing standard broadcast facility for change of site or antenna efficiency, which would result in new or increased cochannel or first adjacent channel overlap, if it is found that good cause for the change exists, and such overlap is not in excess of 2 miles along the line of maximum penetration.
- (17) For waiver of the provisions of § 73.652(a) of this chapter to permit multiple-city identification, where the additional community or communities with which identification is sought, are provided with the requisite field intensity

- specified in § 73.685(a) of this chapter by the station seeking such authority.
- (e) Applications or requests concerning experimental or developmental broadcast stations involving:
- (1) Assignment from time to time of the frequency or frequencies, power, emission, and type of equipment to be employed by any experimental or developmental broadcast station, so as to provide the maximum results from the experimentation with the minimum interference.
- (2) Addition, modification, or coordination of programs of research or experimentation of any experimental or developmental broadcast station, so as to provide the maximum results from the experimentation which can be reasonably expected of the licensee or licensees.
- (f) To withdraw authorizations for equipment and service or program tests where the terms of the construction permit have not been met.
- (g) Requests for modification of tower painting and lighting specifications.
- (h) To issue such National Defense Emergency Authorizations as may be required to permit stations licensed by the Commission to participate in approved National Defense Plans during a national emergency, and to issue such further authorizations as may be appropriate under Executive Order 11092.
- (i) To dismiss applications without prejudice (1) as provided in § 1.568 (a) and (b) of this chapter, or (2) where an application is not timely filed under the Commission's rules in order to receive comparative consideration with an application or applications with which it is mutually exclusive.
- (j) To give written consent to applicants who request authority to make minor changes in effecting transfers of control, or assignment of licenses, previously authorized by the Commission.
- (k) To advise licensees to cease operation in the event renewal applications are not filed with the Commission prior to the expiration date of the particular license
- (1) To deter action on those renewal license applications received subsequent to the fifteenth day of the month prior to the expiration date of the particular license.
- (m) To grant, for good cause shown, requests for temporary authority to continue operation for a period not to exceed 90 days, where an application for renewal

of license has been filed subsequent to the expiration of the particular license.

- (n) To dismiss or return applications or petitions which are not acceptable under .Commission rules.
- (0) To extend the time to file oppositions to petitions relating to broadcast applications not designated for hearing.
- (p) To administer, interpret, and apply orders or rules of practice and procedure promulgated by the Commission relating to financial and statistical data of stations in the broadcast service and broadcast networks and chains, including applications for extension of time in which to file financial and statistical statements and reports.
- (q) To declare a construction permit for a broadcast facility automatically forfeited if the station authorized by the construction permit is not ready for operation within the time specified therein or within such further time as the Commission may have allowed for completion, and if no application for extension of construction permit has been granted by the Commission or timely flied by the permittee, and to place a notation of the forfeiture in the records of the Commission as of the expiration date of the construction permit.
- (r) After coordination with the U.S. Information Agency, to authorize the use of frequencies and frequency hours by international broadcast stations; and to grant, upon adequate showing of need, waivers for operation in excess of the frequency-hour limitations imposed by § 73.702(1) and the note to § 1.574(c) of this chapter.
- (s) To determine whether an application for modification constitutes a major change in facilities, and whether an amendment to an application constitutes a major amendment; and, if so, to so designate such change or amendment.
- (t) To direct standard broadcast stations to refrain from pre-sunrise operation with their daytime facilities pursuant to § 73.87 of this chapter.
- (u) To dismiss petitions and other pleadings which have clearly been rendered moot.
- (v) To extend the time to file responses to official correspondence.
- (w) With the concurrence of the General Counsel, to issue rulings and interpretations with respect to, and to act upon complaints arising under, section 315 of the Communications Act and § 73.120, 73.290, 73.590, or 73.657 of this chapter.

- (x) To issue Notices of Apparent Liability in amounts not in excess of \$250, under section 503(b) of the Communications Act and § 1.621 of this chapter.
- (y) To issue lists of applications for standard broadcast facilities establishing a "cut-off" date pursuant to the provisions of § 1.571(c) of this chapter.
- (z) To dismiss, subject to request within 30 days for reinstatement and hearing, applications for extensions of time in which to construct a station where it appears that the failure to complete was due to causes under the control of the grantee.
- (aa) To dismiss, as repetitious, any petition for reconsideration of a Commission order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order.
- (bb) To dismiss or deny petitions for rule making which are repetitive or moot or which, for other reasons, plainly do not warrant consideration by the Commission.
- (cc) To act on requests for temporary exemption from the provisions of § 73.242 (a) of the Commission's rules and regulations, and if good cause is shown to grant temporary exemption for a period of no more than 3 months.
- (dd) To accept and grant reinstatement applications beyond the 30-day period in § 1.534(b) of this chapter.
- (ee) To grant temporary operation by remote control pending receipt and consideration of a formal application.
- (ff) To waive the provisions of the note to §§ 1.571 and 73.37 of this chapter to the extent necessary to accept and grant an application by an existing standard broadcast facility for change of site or antenna efficiency, which would result in new or increased co-channel or first adjacent channel overlap, if it is found that good cause for the change exists, and such overlap is not in excess of 2 miles along the line of maximum penetration.
- (gg) To grant, grant in part, or dismiss, as appropriate, informal applications for Presunrise Service Authority (PSA) in accordance with § 73.99 of this chapter, and to suspend, modify, or withdraw such authority under the circumstances outlined therein.
- (hh) To grant temporary authority for subchannel operation.
- (ii) In conjunction with the Office of Chief Engineer, to rule on objections

based on claimed phonetic similarity arising under § 1.550 of this chapter.

(jj) In conjunction with the Office of Chief Engineer, to waive the provisions of § 1.550(d)(1) of this chapter if an examination of the call signs of the broadcast stations notified by applicant clearly indicates that no significant likelihood of public confusion could arise and that no purpose would be served by the waiting period prescribed.

(kk) For waiver of the provisions of §§ 73.117 and 73.287 of this chapter to permit multiple-city identification, where the additional community or communities with which identification is sought are provided with the minimum principal city signal intensity specified in §§ 73.188(b) and 73.315(a) of this chapter for AM and FM broadcast stations, respectively; and to dismiss requests for multiple-city identification which do not meet principal city coverage requirements.

(11) To act on requests for waiver of § 73.651(c) of this chapter, where operation under such requests will not exceed ten hours per week, to permit operation by a noncommercial educational television broadcast station of its aural transmitter to broadcast music accompanied by slides, films or other visual transmissions.

[28 F.R. 12402, Nov. 22, 1963, as amended at 29 F.R. 145, Jan. 7, 1964; 29 F.R. 344, Jan. 15, 1964; 29 F.R. 11160, Aug. 1, 1964; 30 F.R. 8678, July 9, 1965; 30 F.R. 14598, 14599, Nov. 24, 1965; 31 F.R. 3074, Feb. 24, 1966; 31 F.R. 4456, 4457, Mar. 16, 1966; 31 F.R. 6627, May 4, 1966; 31 F.R. 13474, Oct. 19, 1966; 31 F.R. 16567, Dec. 28, 1966; 32 F.R. 10363, July 14, 1967; 32 F.R. 15472, Nov. 1, 1967; 33 F.R. 2445, Feb. 1, 1968; 34 F.R. 6524, Apr. 16, 1969; 34 F.R. 11144, July 2, 1969]

§ 0.287 Record of actions taken.

The history card, the station file, and other appropriate files are designated to be the official record of the action taken by the Chief of the Broadcast Bureau. [33 F.R. 8228, June 1, 1968]

CHIEF, CATV TASK FORCE

§ 0.289 Authority delegated.

(a) The Chief, CATV Task Force, in coordination with the Broadcast Bureau, is delegated authority to act upon the following applications for Community Antenna Relay station authorizations, if such authorizations comply fully with the requirements of the Communications Act, the provisions of this chapter (see,

in particular, Subpart J of Part 74 of this chapter), and Commission policy and standards; if no mutually exclusive application has been filed; and if no petition to deny or other substantial objection to the application has been filed:

(1) Applications for construction permits for new or changed stations.

(2) Applications for licenses to cover construction permits.

(3) Applications for modification, assignment, transfer of control, or renewal.

- (b) All minutes of actions taken by the Chief of the CATV Task Force pursuant to the authority delegated in this section shall be maintained for public inspection by the CATV Task Force.
- (c) The Chief, CATV Task Force, is delegated authority to act upon the following matters:
- (1) Requests for extensions of time in which to comply with Commission orders directing carriage and/or program exclusivity by a CATV system for a reasonable period of time upon a showing of good cause.
- (2) For withdrawal of papers in accordance with § 1.8 of this chapter.
- (3) To dismiss or return applications or petitions which are not acceptable under Commission rules.
- (4) For extension of time within which to file briefs and comments with respect to rule making.
- (5) To extend the time to file pleadings related to waiver requests or applications not designated for hearing.
- (6) To dismiss petitions and other pleadings which have clearly been rendered moot.
- (7) To initiate official correspondence where appropriate, and to extend the time to file responses to official correspondence.
- (8) To dismiss, as repetitious, any petition for reconsideration of a Commission order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order.
- (9) To dismiss or deny petitions for rule making which are repetitive or moot or which, for other reasons, plainly do not warrant consideration by the Commission.
- (10) to dismiss premature requests for waivers of the Commission's rules.
- (11) To grant requests for waiver of §§ 74.1037 and 74.1067 of this chapter.
- (12) To act on unopposed proposals to import distant educational television sig-

nals into the 100 largest television markets, as defined in § 74.1107(a) of this chapter.

[31 F.R. 16715, Dec. 30, 1966, as amended at 32 F.R. 3831, Mar 8, 1967; 32 F.R. 6283, Apr. 21, 1967; 32 F.R. 7835, May 30, 1967; 33 F.R. 8228, June 1, 1968; 34 F.R. 15416, Oct. 3, 1969]

CHIEF, COMMON CARRIER BUREAU

§ 0.291 Authority concerning radio matters.

The Chief of the Common Carrier Bureau is delegated authority to act upon the following applications, requests, and other matters which are not in hearing status, involving the use of radio, insofar as they apply to common carrier services (except marine and aeronautical), where the estimated construction cost is less than \$2,000,000:

(a) From existing licensees for instruments of authorization, for the fixed public or fixed public press radio services.

- (1) New points of communication, not already authorized to a station of the licensee at some other location or not already authorized by an outstanding construction permit or special temporary authorization; unless the application for a new point of communication is for the transmission or control of transmission or reception of addressed program or facsimile material.
- (2) Changes in transmitter location other than local in character.
- (3) The establishment of a new type of service.
- (b) For the domestic public radio services, and for the fixed public services, in the possessions of the United States and in the State of Hawaii.
- (c) For developmental stations which render or propose to render a common carrier service.
- (d) For special temporary authorizations, including those filed at times outside of the regular office hours of the Commission in emergency cases.
- (e) For operation with new or modifled equipment pending repair of existing equipment, or pending receipt of and action upon formal applications; operation with licensed, new or modified equipment at a temporary location with a temporary antenna system in case of an emergency when, due to causes beyond the control of the licensee, it becomes impossible to continue operating at the licensed location; and, special operations necessary to facilitate equipment and service tests, or to comply

with technical requirements specified in authorizations, orders, rules or releases of the Commission.

- (f) For a change in location, type or model, design or number of transmitters; representations of compliance with technical requirements specified in authorizations, orders, rules or releases of the Commission (except formal applications).
- (g) For equipment and service tests, or extensions thereof.
- (h) To determine whether an application for modification constitutes a major change in facilities, and whether an amendment to an application constitutes a major amendment; and, if so, to so designate such change or amendment.
- (i) To dismiss, as repetitious, any petition for reconsideration of a Commission order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order.
- (j) To dismiss or deny petitions for rule making which are repetitive or moot or which, for other reasons, plainly do not warrant consideration by the Commission.

(k) To

waive the requirement of § 21.23(a) of this chapter that amendments to applications be served by the applicant upon persons who have filed petitions to deny the application or to designate it for hearing, upon a showing by the applicant that the requirement is unreasonably burdensome, and to prescribe such alternative procedures as may be appropriate under the circumstances to protect petitioners' interests and to avoid undue delay in the proceed-Requests for waiver shall be served on petitioners. Oppositions to the petition may be filed within 5 days after the petition is filed and shall be served on the applicant. Replies to oppositions

will not be entertained. [28 F.R. 12402, Nov. 22, 1963, as amended at 31 F.R. 3074, Feb. 24, 1966; 31 F.R. 4456, Mar. 16, 1966; 32 F.R. 5273, Mar. 29, 1967; 33 F.R. 15291, Oct. 15, 1968]

§ 0.292 Additional authority concerning radio matters.

The Chief of the Common Carrier Bureau is delegated authority to act on the following matters:

(a) To designate for hearing all mutually exclusive applications for radio facilities filed pursuant to Parts 21, 23, and 25 of this chapter.

(b) To determine under § 1.80 of this chapter whether forfeiture liability has been incurred in connection with the operation of any station governed by Parts 21, 23, and 25 of this chapter, and to issue notices of apparent liability as provided in § 1.80 of this chapter.

[33 F.R. 15291, Oct. 15, 1968]

§ 0.293 Authority concerning position of officer.

The Chief of the Common Carrier Bureau is delegated authority to act upon applications under section 212 of the Communications Act for authority to hold the position of officer or director of more than one carrier subject to the Act, and to act upon applications for a finding that a carrier owns more than fifty percent of the stock of another or other carriers, or that a person owns fifty percent or more of the stock of two or more carriers.

§ 0.294 Authority concerning section 214 of the Act.

The Chief of the Common Carrier Bureau is delegated authority to act upon the following applications or requests under section 214 of the Communications Act:

- (a) For a certificate authorizing the construction, acquisition, operation, or extension of lines, or for an authorization for temporary or emergency service or the supplementing of existing facilities involving an estimated construction or purchase cost of less than \$2 million, or an annual rental of less than \$250,000.
- (b) For modification of a certificate or authorization under this section of the Act where such amendment or modification involves an estimated construction or purchase cost of less than \$2 million or an annual rental of less than \$250,000.
- (c) For authority to discontinue, reduce, or impair telephone service where the applications are filed pursuant to the provisions of §§ 63.63, 63.65, and 63.66 of this chapter.
- (d) For an authorization for temporary or emergency closure of telegraph offices, for any closure of a telegraph office at a military establishment, for closure of railroad-operated agency offices, for closure of company-operated main offices where substitute service is to be provided by telephone, teleprinter or facsimile-operated agency office in the same community and for any reduction in the hours of telegraph service in a community or part of a community in

those cases where applicable Commission policy has been established.

(e) For informal requests for authority to discontinue, reduce or impair telegraph service filed pursuant to the provisions of §§ 63.63, 63.64, and 63.66 through 63.69 of this chapter.

(f) For reconsideration and dismissal of applications to discontinue, reduce or impair service where authority has been granted but will not be used by applicant because of conditions arising subsequent to the filing of the application.

(g) For special temporary authorizations in the event of cable failures, satellite outages, or similar occurrences, including those requested at times outside of regular office hours of the Commission

in emergency cases, without regard to the cost of obtaining such emergency restored facilities.

[28 F.R. 12402. Nov. 22, 1963, as amended at

§ 0.295 Authority concerning section 220 of the Act.

33 F.R. 15291, Oct. 15, 1968]

The Chief of the Common Carrier Bureau is delegated authority to interpret the regulations and to act upon the administration of such regulations promulgated by the Commission pursuant to section 220 of the Communications Act, relating to accounts, records and memoranda to be kept by carriers subject to the jurisdiction of the Commission.

§ 0.296 Authority concerning section 221(a) of the Act.

The Chief of the Common Carrier Bureau is delegated authority:

- (a) To determine upon consideration of all relevant factors whether hearings shall be held on applications filed under section 221(a) of the Communications Act where no request therefor has been made by a telephone company, an association of telephone companies, a State Commission or local government authority;
- (b) To fix the time and place for hearings he determines shall be held under paragraph (a) of this section or where a request therefor has been made by a telephone company, an association of telephone companies, a State Commission or local government authority; and to give reasonable notice in writing to the Governor of each of the States in which the physical property affected, or any part thereof, is situated, to the State

Commission having jurisdiction over telephone companies, and to such other persons as he may deem advisable; and

(c) To act in all other cases upon applications filed under section 221(a) where the proposed expenditure for consolidation, acquisition or control is less than \$500,000.

§ 0.297 Authority concerning tariff regulations.

The Chief of the Common Carrier Bureau is delegated authority to act upon all matters arising in connection with the administration of tariff regulations promulgated by the Commission pursuant to section 203 of the Communications Act, and, in connection with the administration of that section insofar as it relates to the modification of requirements thereof or made pursuant thereto, as authorized in particular instances by subsection (b) thereof, and to the rejection of tariffs as authorized by subsection (d) thereof.

§ 0.298 Authority concerning Communications Satellite matters.

- (a) The Chief of the Common Carrier Bureau is delegated authority to act upon applications from communications common carriers for authorization to purchase stock in the Communications Satellite Corporation, pursuant to the provisions of section 304(b) of the Communications Satellite Act and of Subpart H of Part 25 of this chapter.
- (b) The Chief of the Common Carrier Bureau is delegated authority to act upon notification statements required to be filed under § 25.166 of this chapter by the Communications Satellite Corporation, other communications common carriers, and their respective prime contractors with respect to the procurement of apparatus, equipment, and services required for the establishment and operation of the communications satellite system and satellite terminal stations. This delegation is limited to taking the following actions:
- (1) Determination that no action should be taken to prevent the proposed contracts from being awarded at any time subsequent to 10 days after the filing of the statements with the Commission;
- (2) Notice that such statements are defective:
- (3) Issuance of a public notice announcing a 30-day investigation period

to determine whether the rules have been complied with:

- (4) Determination that no action should be taken to prevent the proposed contract from being awarded at any time subsequent to 30 days after the date of the public notice.
- (c) The Chief of the Common Carrier Bureau is delegated authority to act upon applications, requests, and other matters which are not in hearing status relating to communication satellite facilities and services as follows:
- (1) For modifications of communication satellite earth station or space station construction permits; for operating licenses or modification or renewal of operating licenses for such facilities: for authorizations for developmental purposes covered by Part 25, Subpart E, of this chapter; and for modifications of approval to participate in construction of satellites intended for use in the global communication satellite system, when the applications involved comply with the requirements of the Communications Act of 1934, as amended, and the Communications Satellite Act of 1962, are in accord with Commission policy and standards, are not mutually exclusive with any other applications, and concerning which no petition to deny has been filed.
- (2) For authorizations to establish channels of communications with new points, to obtain units of satellite utilization, and to engage in promotional demonstrations or experimental operations.
- (3) Temporary or emergency authorizations with respect to any of the foregoing matters.

[29 F.R. 4913, Apr. 8, 1964, as amended at 33 F.R. 15291, Oct. 15, 1968]

§ 0.302 Authority concerning records and papers.

The Chief of the Common Carrier Bureau is delegated authority to act upon the following matters insofar as they apply to records or papers involving common carriers:

- (a) To dismiss petitions and other pleadings relating to matters not in hearing status which have clearly been rendered moot.
- (b) Requests for withdrawal of papers in accordance with § 1.8 of this chapter. [28 F.R. 12402, Nov. 22, 1963, as amended at 30 F.R. 12125, Sept. 23, 1965]

§ 0.303 Authority concerning extension of time and waivers.

The Chief of the Common Carrier Bureau is delegated authority to act upon the following requests:

(a) For the extension of time in which to file annual, monthly and special reports required by the Commission pursuant to section 219 of the Communi-

cations_Act.

(b) For the extension of the time prescribed in § 43.51 of this chapter for the filing of documents specified therein.

(c) For the extension of time within which briefs, comments and pleadings may be filed in common carrier rule

making proceedings.

(d) For extension of time previously ordered by the Commission within which the transfer of control or assignment of license be effectuated insofar as these apply to fixed public, fixed public press, domestic public or common carrier experimental radio services.

(e) For extension of time within which to comply with technical requirements specified in authorizations, orders, and rules or releases of the Commission insofar as these apply to fixed public, fixed public press, domestic public or common carrier experimental radio serv-

ices.

- (f) For waiver of or exception to any rule, regulation or requirement relating to the services under his jurisdiction and to act upon petitions or requests relating to the assignment of frequencies to common carrier operations but requiring action under \$ 2.102 of this chapter, when he finds that the operation for which permission is sought (1) is of a nonrecurring nature and does not warrant a rule making proceeding with a view to establishing it on regular basis, (2) will not exceed 90 days, and (3) will cause no harmful interference to any service operating in accordance with the Table of Frequency Allocations. delegation does not apply to requests for renewals of any authority to operate granted hereunder.
- (g) For extension of time within which to file pleadings concerning applications filed in the fixed public, fixed public press, domestic public, or common carrier experimental radio services which are not in hearing status.

[28 F.R. 12402, Nov. 22, 1963, as amended at 30 F.R. 12125, Sept. 23, 1965; 31 F.R. 13474, Oct. 19, 1966]

§ 0.304 Authority delegated jointly to Chiefs of Common Carrier and Safety and Special Radio Services Bureaus.

Authority is delegated jointly to the Chief of the Common Carrier Bureau and the Chief of the Safety and Special Radio Services Bureau to act upon applications involving common carrier matters in the maritime mobile service, aeronautical mobile service, and in the fixed service in Alaska. (For record of actions taken under this section, see § 0.337.)

[29 F.R. 14405, Oct. 20, 1964]

§ 0.307 Record of actions taken.

The application and authorization files in the License Branch and the appropriate central files of the Common Carrier Bureau are designated as the Commission's official records of actions by the Chief of the Bureau pursuant to authority delegated to him. In the case of joint authority exercised by the Chief, Common Carrier Bureau, and the Chief, Safety and Special Radio Services Bureau, § 0.337 applies.

[33 F.R. 8228, June 1, 1968]

FIELD ENGINEERING BUREAU

- § 0.311 Authority delegated to the Chief and to the Deputy Chief of the Field Engineering Bureau.
- (a) The Chief of the Field Engineering Bureau is delegated authority to act upon the following matters which are not in hearing status:
- (1) Except as otherwise provided in § 1.61 of this chapter, with respect to the construction, marking, and lighting of antenna towers and supporting structures, to exercise the functions of the Commission as set forth in Part 17 of this chapter: Provided, however, That in cases in which the Federal Aviation Agency recommends denial of any application, the Chief of the Field Engineering Bureau advises the bureau concerned in order that it may submit the application to the Commission for appropriate action.
- (2) Waiver of rules, regulations and orders of the Commission relating to the proper time for filing of renewals of commercial and amateur radio operator licenses.
- (3) Temporary operation by radio station licensees, except licensees of standard and FM broadcast stations, with a

licensed operator of lower grade than normally required or for waiver of other technical requirements for operators.

- (4) With respect to the operation of industrial, scientific, and medical equipment subject to Part 18 of this chapter, to issue, in accordance with section 312 (c) of the Act, (i) orders to show cause why a cease and desist order pursuant to section 312(b) should not be issued; and (ii) cease and desist orders after waiver of hearing and certification of the proceeding to the Commission pursuant to § 1.92 of this chapter.
- (5) To suspend commercial operator licenses and to designate the matter for hearing, as provided for in § 1.85 of this chapter.
- (6) To act on requests for a waiver of the English language provisions of §§ 13.22 and 13.23 of this chapter in the case of Spanish-speaking applicants in Puerto Rico and vicinity, and to issue licenses bearing appropriate restrictions to those applicants found qualified.
- (7) To dismiss without prejudice applications for new and renewal commercial radio operator licenses in cases where prior to designation of such applications for hearing an applicant has failed to answer official correspondence or a request for additional information from the Commission.
- (8) To act on requests for a provisional radio operator certificate.
- (9) To act on requests by holders of a pilot certificate issued by the Civil Aeronautics Administration, the Federal Aviation Agency or the Federal Aviation Administration for a waiver of the U.S. citizenship requirement under section 303(1) of the Communications Act of 1934, as amended.
- (10) To act on requests for waiver of the written examination requirements of §§ 13.21 and 13.22 of this chapter and to authorize oral examination in lieu thereof.
- (11) To reject certifications of industrial heating equipment as provided in § 18.118(a) of this chapter and to accept certifications of industrial heating equipment as provided in § 18.118(d) of this chapter.
- (12) To determine under § 1.80 of this chapter whether forfeiture liability has been incurred by any licensed commercial radio operator (except where the operator is also subject to liability as licensee of the station in question); to issue notices of apparent liability in such

cases; and to act in such cases on requests that the forfeiture be cancelled or reduced.

Note: Where the operator is also subject to liability as licensee of the station, these matters are acted on by the Bureau or Office which exercises authority over the class of station involved. See §§ 0.243(b), 0.292(b), and 0.332 (d) and (e).

- (13) To act on requests for a waiver of the geographical restriction in § 13.4 (c) of this chapter on the validity of operator licenses issued to alien aircraft pilots.
- (14) To dismiss, as repetitious, any petition for reconsideration of a Commission order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order.
- (15) To dismiss or deny petitions for rule making which are repetitive or moot or which, for other reasons, plainly do not warrant consideration by the Commission.
- (b) The Chief and the Deputy Chief of the Field Engineering Bureau are authorized to declare that a state of general communications emergency exists and to act on behalf of the Commission pursuant to the provision of § 97.107 of this chapter with respect to the operation of amateur stations during a state of general communications emergency.

[28 F.R. 12402, Nov. 22, 1963, as amended at 29 F.R. 12773, Sept. 10, 1964; 29 F.R. 15923, Nov. 28, 1964; 30 F.R. 7521, June 9, 1965; 31 F.R. 3074, Feb. 24, 1966; 31 F.R. 4457, Mar. 16, 1966; 32 F.R. 20860, Dec. 28, 1967; 33 F.R. 3637, Mar. 1, 1968]

§ 0.314 Authority delegated to the Engineers in Charge.

The Engineers in Charge at each headquarters office of the 24 districts of the Field Engineering Bureau are delegated authority to act upon the following applications, requests, or other matters which are not in hearing status:

- (a) For new, modified replacement, duplicate or renewal commercial radio operator license and provisional radio operator certificate.
- (b) For temporary permission to operate standard and FM broadcast stations with licensed operators of lesser grade than normally required by the Commission's rules or for waiver of other technical requirements of operators at such stations.
- (c) In cases of informal requests from broadcast stations to extend temporary

authority for operation without any of the following: modulation monitor, frequency monitor, plate ammeter, plate voltmeter, base current meter, common point meter, and transmission line meter from FM and television stations.

(d) Relating to the time within which an applicant for amateur or commercial radio operator license may take an examination after having failed a previous examination (§§ 13.27 and 97.33 of this

chapter).

(e) For periodic survey as required by section 385 of the Communications Act of 1934, as amended, and issuance of Communications Act radiotelephony certificates in accordance with § 83.512 of this chapter.

(f) Applications, in any acceptable form, filed at Commission field offices located in Alaska, for special temporary operator license authorization, in lieu of regular commercial radio operator license, when it is shown that there is a need for such authorization for use in connection with the protection of life or property during an emergency period.

(g) Requests for interim ship station and for interim radar ship station licenses as provided by Part 83 of this chapter governing stations on shipboard

in the maritime services.

- (h) For special operation necessary to facilitate equipment, program and service tests or to comply with technical requirements specified in authorizations, orders, rules or releases of the Commission.
- (i) Operation during daytime for specified periods with the nighttime facilities in order to check measurements and operation.
- (j) For ship radio inspection and certification of the ship radio license, pursuant to the requirements of section 362
 (b) of the Communications Act of 1934, as amended.
- (k) For a Safety Radiotelegraphy Certificate or a Safety Radiotelephony Certificate in accordance with the terms of Regulations 11 and 12, Chapter I, of the Safety Convention.
- (1) For inspection or periodical survey as required by Article 11 of the Great Lakes Agreement and certification prescribed by Articles 12 and 13 thereof.
- (m) Issuance of notices and orders to operators of industrial heating equipment, as provided in §§ 18.120 and 18.121 of this chapter.

(n) Requests for permission to resume operation of industrial heating equipment on a temporary basis, as provided in §§ 18.120 and 18.121 of this chapter, and requests for extensions of time within which to file final reports, as provided by § 18.122(b) of this chapter.

(o) Informal applications filed in accordance with the provisions of § 83.47 of this chapter for temporary waivers of annual inspection of vessels as required by section 362(b) of the Communications Act of 1934, as amended.

[28 F.R. 12402, Nov. 22, 1963, as amended at 32 F.R. 20860, Dec. 28, 1967]

- 30.315 Authority delegated to Engineers in Charge who serve as FCC representatives to Office of Civil Defense (OCD) Regional Civil Defense Coordinating Boards and Regional Preparedness Committees of the Office of Emergency Planning (OEP).
- (a) Engineers in Charge of Field Engineering Bureau district offices at Boston, Massachusetts; Baltimore, Maryland; Atlanta, Georgia; Detroit, Michigan; Dallas, Texas; Denver, Colorado; San Francisco, California; and Seattle, Washington, are designated as the principal FCC representatives to Regional Preparedness Committees. (FCC Field Liaison Officers of the Office of Emergency Communications are designated as alternates. The authority delegated to FCC representatives to Regional Preparedness Committees is set forth in § 0.385.)
- (b) Engineers in Charge of Field Engineering Bureau district offices at Boston, Mass.; Baltimore, Md.; Atlanta, Ga.; Detroit, Mich.; Dallas, Tex.; Denver, Colo.; San Francisco, Calif.; and Seattle, Wash., are designated as the alternate FCC representatives to the Regional Civil Defense Coordinating Boards. (FCC Regional Liaison Officers of the Emergency Communications Division are designated as the principal representatives to the Regional Civil Defense Coordinating Boards.)
- (c) The Engineers in Charge listed in paragraph (a) of this section are designated as FCC representatives to State Emergency Telecommunications Task Groups established by the OEP in its Organization and Planning Guide of September 1962 (OEP Circular 3300.1).

[29 F.R. 14665, Oct. 28, 1964, as amended at 32 F.R. 20860, Dec. 28, 1967]

- § 0.316 Authority delegated to Marine Supervisors at marine offices, to engineers engaged in ship inspection duties at radio district offices, and to radio engineers at suboffices.
- (a) The Radio Engineer at each suboffice of a district headquarters office of
 the Field Engineering Bureau is delegated authority to act upon ali matters
 contained in § 0.314, except paragraph
 (b).
- (b) The Marine Supervisor at the marine office of the Field Engineering Bureau at Tampa, Fla., is delegated authority to act upon matters set forth in § 0.314 (a), (d), (e), (g), (j), (k), and (o). The Marine Supervisor at the marine office of the Field Engineering Bureau at San Pedro, Calif., is delegated authority to act upon matters set forth in § 0.314 (e), (g), (j), (k), and (o), and to act upon applications for restricted radiotelephone operator permits and requests for provisional radio operator certificates related thereto.
- (c) Engineers engaged in ship inspection duties at radio district offices of the Field Engineering Bureau are delegated authority to act upon matters set forth in § 0.314 (e), (j), and (l).

[28 F.R. 12402, Nov. 22, 1963, as amended at 29 F.R. 15288, Nov. 14, 1964]

§ 0.317 Record of actions taken.

All actions taken under § 0.314 or § 0.316 shall be recorded in the field office where such action has been taken, and except for actions taken under § 0.314 (c), (d), (h), and (i), a report thereof shall be sent to the Washington, D.C., office of the Field Engineering Bureau quarterly.

§ 0.318 Authority delegated to operator examiner.

The operator examiner at the examination office in Gettysburg, Pa., is authorized to act on requests for waiver of the waiting time requirement applying to applicants for amateur radio operator licenses who have failed a previous examination (§ 97.33 of this chapter).

[30 F.R. 8837, July 14, 1965]

CHIEF, SAFETY AND SPECIAL RADIO SERVICES BUREAU

§ 0.331 Authority delegated.

The Chief of the Safety and Special Radio Services Bureau is delegated authority to act upon the following appli-

- cations, requests, and other matters which are not in hearing status:
- (a) Consistent with and pursuant to the rules governing the Disaster Communications Service and the Maritime. Aviation, Public Safety, Industrial, Amateur. Land Transportation, and Citizens Radio Services, and the Fixed Service in Alaska, on all applications for construction permits, modification of construction permits, extension of construction permits, station licenses, modification of station licenses, transfer of control or assignment of assignment of construction permits or station licenses, renewal of station construction licenses, amateur operator licenses, renewal or modification of amateur operator licenses and special temporary authorizations. With respect to applications involving common carrier matters. this authority shall be exercised jointly with the Chief, Common Carrier Bureau pursuant to §§ 0.304 and 0.333.
- (b) On the following matters insofar as they involve the Disaster Communications Service or the Maritime, Aviation, Public Safety, Industrial, Amateur, Land Transportation, or Citizens Radio Services:
- (1) Requests for extensions of time for equipment or service tests or within which to comply with technical requirements specified in authorizations, orders and rules or releases of the Commission.
 - (2) [Reserved]
- (3) Requests for withdrawal of papers in accordance with § 1.8 of this chapter.
- (4) Requests for extension of time within which briefs, comments and pleadings may be filed in rule making proceedings.
- (5) To make the finding of emergency involving danger to life or property or due to damage to equipment, as provided by section 308(a) of the Communications Act of 1934, as amended.
- (6) Cancellation of station licenses, construction permits or other authorizations upon the request of the licensee or permittee or upon abandonment of the station.
- (7) Petitions or requests seeking waiver of or exception to any rule, regulation or requirement, and to act upon petitions or requests relating to the assignment of frequencies to the various safety and special radio services but requiring action under \$2.102 of this chapter, when he finds that the operation for which permission is sought (i) is

of a nonrecurring nature and does not warrant rule making proceedings with a view to establishing it on a regular basis. (ii) will not exceed 180 days, and (iii) will cause no harmful interference to any service operating in accordance with the Table of Frequency Allocations. delegation does not apply to requests for renewals of any authority to operate granted hereunder: Provided, however. That none of the foregoing limitations shall apply to petitions or requests for waiver of or exception to any requirement set forth in §§ 97.13, 97.27, 97.51, and 97.95 of this chapter, or to the requirements with regard to types of emission set forth in § 97.111.

(8) To issue, in accordance with section 312(c) of the Communications Act of 1934, as amended, (i) orders to show cause why an order of revocation pursuant to section 312(a) of the Act. or a cease and desist order pursuant to section 312(b) of the Act, should not be issued: and (ii) after waiver of hearing and termination of the hearing proceeding in revocation cases and cease and desist cases as prescribed by §1.92 of this chapter, to issue orders of revocation and orders to cease and desist, and also orders that a revocation or cease and desist order shall not issue.

(9) To grant the authorizations provided for in § 2.102(c) of this chapter.

(10) To cancel amateur operator licenses as provided in § 97.35 of this chapter.

(11) To act upon all requests (to the extent that they relate to qualifications various amateur operator licenses) for waiver of the requirement of Part 97 of this chapter where it is alleged that, by reason of a protracted or permanent physical disability, the applicant is unable to meet the requirements of such rules.

(12) To issue, in accordance with section 316 of the Act, orders to modify ship radiotelephone stations by deleting therefrom any frequency when the use thereof has resulted in harmonic emissions found to be capable of causing interference to other radio services, and if a hearing thereon is requested, to designate such matters for hearing.

(13) To grant, in the absence of unusual circumstances, a six month waiver of the type acceptance requirement of § 83.139 of this chapter in cases substantially the same as those in which the Commission en banc has taken similar action.

(14) To deny requests for waiver of type acceptance requirement of § 83.139 of this chapter and dismiss associated applications for ship station licenses in cases substantially the same as those in which the Commission en banc has taken similar action.

(15) To act on requests for waiver of the type acceptance requirement of § 83.139 of this chapter where the applicant is operating a ship radiotelephone transmitter on frequencies between 4 and 27.5 Mc/s and where authority is needed to operate non-type accepted ship radiotelephone equipment pending conversion to single sideband.

(16) To grant waivers of the type acceptance requirements of § 87.77 of this chapter for a period not to extend beyond January 1, 1970, in cases substantially the same as those in which the Commission en banc has taken similar action.

(17) To grant waivers of the type acceptance requirements of § 89.117 of this chapter for a period not to extend beyond January 1, 1967, in cases substantially the same as those in which the Commission en banc has taken similar action.

(18) To act on requests for waiver of application procedures to allow a licensee to submit a request for the identical modification or assignment of a number of outstanding authorizations without filing a separate application for each station. Action taken under this delegation does not include authority to waive or reduce applicable fee requirements which shall be determined as if separate applications were filed for each station.

(19) To act on requests for waiver of the restrictions of §1.916 of this chapter subsequent to revocation of a radio station license.

(20) With the concurrence of the Chief Engineer, to act on applications for the use of frequencies in the 216-220 Mc/s band for telemetering purposes and to waive appropriate rules as may be necessary to grant such applications.

(c) Consistent with the provisions of section 309(f) of the Communications Act of 1934, as amended, to grant or deny requests for temporary authorizations and to issue orders stating the reasons therefor, but only prior to the time a petition to deny the application involved has been filed or the application has been designated for hearing.

(47 U.S.C. 154, 303) [28 F.R. 12402, Nov. 22, 1963, as amended at 29 F.R. 3202, Mar. 10, 1964; 29 F.R. 18815, Oct. 7, 1964; 29 F.R. 14405, Oct. 20, 1964; 29 F.R. 18373, Dec. 24, 1964; 30 F.R. 8678, July 9, 1965; 31 F.R. 10125, July 27, 1966; 31 F.R. 13474, Oct. 19, 1966; 31 F.R. 16316, Dec. 21, 1966; 32 F.R. 3831, Mar. 8, 1967; 33 F.R. 7153, May 15, 1968; 33 F.R. 10849, July 31, 1968]

§ 0.332 Additional authority delegated.

The Chief of the Safety and Special Radio Services Bureau is delegated authority to act upon the following applications, requests and other matters:

- (a) Requests for assignment of call signs to new stations in the Citizens and Amateur Radio Services and for changes in the call signs of existing stations in those services.
- (b) Applications or requests for exemption, pursuant to the provisions of sections 352(b) and 383 of the Communications Act; Regulation 4, Chapter I of the Safety Convention; Regulation 5, Chapter IV of the Safety Convention; Regulation 12(b), Chapter V of the Safety Convention; or Article 6 of the Great Lakes Radio Agreement:
- (1) For emergency and renewal exemption of vessels;
- (2) For initial exemption of vessels subject to Title III, Part III of the Act;
- (3) For initial exemption of vessels of less than 100 gross tons subject to Title III, Part II of the Act or the Safety Convention;
- (4) For exemption from Title III, Part II, of the Act of vessels operated in the Gulf of Mexico which participate in oil well drilling operations when the circumstances are substantially the same as those in precedent cases decided by the Commission en banc; and
 - (5) [Reserved]
- (c) Applications or requests for approval of schedules of transmissions by coast stations of information for the general benefit of mariners to a plurality of mobile stations or to designated fixed locations.
- (d) Make determination and notification of incurrence of forfeitures under the provisions of sections 364, 386, 507, and 510 of the Communications Act, with reference to stations governed by the provisions of Parts 81-99 of this chapter.
- (e) Applications made pursuant to section 504(b) of the Communications Act of 1934, as amended, for mitigation

or remission of forfeitures imposed under sections 364, 386, 507, and 510 of the Act.

(f) Issue orders suspending the licenses of amateur operators in accordance with section 303(m)(1) of the Act and if a hearing thereon is requested, to designate such matters for hearing.

(g) Requests pursuant to the provisions of §§ 81.104(b) (1) and 81.189(c) (1) of this chapter for waiver or exception to the requirements of those sections concerning 2182 kc/s coast station facilities and watch.

(h) To dismiss applications without prejudice in cases where, prior to designation of such application for hearing, an applicant has failed to answer official correspondence or a request for additional information from the Commission.

(i) To grant or deny applications for permits and to modify, suspend, or cancel such permits, pursuant to Subpart G, Part 97 of this chapter.

(j) Requests for extension of time within which to file pleadings concerning applications filed in the Safety and Special Radio Services which are not in hearing status.

(k) To dismiss petitions and other pleadings relating to matters not in hearing status which have clearly been rendered moot.

(1) To dismiss, as repetitious, any petition for reconsideration of a Commission order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order.

(m) To dismiss or deny petitions for rule making which are repetitive or moot or which, for other reasons, plainly do not warrant consideration by the Commission.

(n) To maintain the official record of hearing proceedings for the revocation of licenses in the Safety and Special Radio Services until such matters are docketed.

[28 F.R. 12402, Nov. 22, 1963, as amended at 30 F.R. 2706, Mar. 3, 1965; 30 F.R. 6779, May 19, 1965; 30 F.R. 12125, Sept. 23, 1965; 31 F.R. 3074, Feb. 24, 1966; 31 F.R. 4467, Mar. 16, 1966; 33 F.R. 5302, Apr. 3, 1968]

§ 0.333 Authority delegated jointly to Chiefs of Common Carrier and Safety and Special Radio Services Bureau.

Authority is delegated jointly to the Chief of the Common Carrier Bureau and the Chief of the Safety and Special Radio Services Bureau to act upon applications involving common carrier matters in the maritime mobile service, aeronautical mobile service, and in the fixed service in Alaska.

[29 F.R. 14406, Oct. 20, 1964]

§ 0.337 Record of actions taken.

The history card pertaining to a certain station is designated to be the official record of the action taken by the Chief, Safety and Special Radio Services Bureau, in pursuance of the authority delegated to him in §§ 0.331 and 0.332 or jointly to him and the Chief, Common Carrier Bureau, in § 0.333. In cases where no history card is prepared, the application and authorization file pertaining to the station in question is designated to be the official record of the action taken by the Chief of the Bureau, or by him jointly with the Chief of the Common Carrier Bureau.

[33 F.R. 8228, June 1, 1968]

HEARING EXAMINERS

§ 0.341 Authority of hearing examiner.

- (a) After a hearing examiner has been designated to preside at a hearing and until he has issued an initial decision or certified the record to the Commission for decision, or the proceeding has been transferred to another hearing examiner, all motions, petitions and other pleadings shall be acted upon by such hearing examiner, except the following:
- (1) Those which are to be acted upon by the Commission. See § 1.291(a) (1) of this chapter.
- (2) Those which are to be acted upon by the Review Board under § 0.365 (b) and (d).
- (3) Those which are to be acted upon by the Chief Hearing Examiner under § 0.351.
- (b) Any question which would be acted upon by the hearing examiner if it were raised by the parties to the proceeding may be raised and acted upon by the hearing examiner on his own motion.
- (c) Any question which would be acted upon by the Chief Hearing Examiner, the Review Board or the Commission, if it were raised by the parties, may be certified by the hearing examiner, on his own motion, to the Chief Hearing Examiner, the Review Board or the Commission, as the case may be.

[29 F.R. 6442, May 16, 1964]

§ 0.347 Record of actions taken.

Except for actions taken during the course of the hearing and upon the record thereof, actions taken by a hear-

ing examiner pursuant to \$0.341 shall be recorded each week in writing and filed in the official minutes of the Commission.

CHIEF HEARING EXAMINER

§ 0.351 Authority delegated.

The Chief Hearing Examiner shall act on the following matters in proceedings conducted by hearing examiners:

(a) Initial specifications of the time and place of hearings where not otherwise specified by the Commission and excepting actions under authority delegated by § 0.296.

(b) Designation of the hearing ex-

aminer to preside at hearings.

- (c) Orders directing the parties or their attorneys to appear at a specified time and place before the hearing examiner for an initial prehearing conference in accordance with § 1.251(a) of this chapter. (The hearing examiner named to preside at the hearing may order an initial prehearing conference although the Chief Hearing Examiner may not have seen fit to do so and may order supplementary prehearing conferences in accordance with § 1.251(b) of this chapter.)
- (d) Petitions requesting a change in the place of hearing where the hearing is scheduled to begin in the District of Columbia or where the hearing is scheduled to begin at a field location and all appropriate proceedings at that location have not been completed. (See § 1.253 of this chapter.)

(e) In the absence of the hearing examiner who has been designated to preside in a proceeding, to discharge the hearing examiner's functions.

(f) All pleadings filed, or matters which arise, after a proceeding has been designated for hearing, but before an examiner has been designated, which would otherwise be acted upon by the examiner, including all pleadings filed, or matters which arise, in cease and desist and/or revocation proceedings prior to the designation of a presiding officer.

(g) All pleadings (such as motions for extension of time) which are related to matters to be acted upon by the Chief Hearing Examiner.

[29 F.R. 6443, May 16, 1964]

REVIEW BOARD

§ 0.361 General authority.

(a) The Review Board is a permanent body with continuing functions. The main function of the Board is to review matters referred to it by the Commission in hearing proceedings. The Board also takes original action on certain interlocutory matters which arise during the course of hearing proceedings. hearing matters referred to the Board on a regular basis are listed in § 0.365. Other hearing matters may be referred to the Board for review on a case by case basis, either at the time of designation for hearing or upon consideration of exceptions. The Commission may, from time to time, assign the Board additional inconsistent duties not with functions.

- (b) Any matter referred to the Board on a regular basis or otherwise may, on its own motion or upon its consideration of the motion of any party, be certified by the Board to the Commission, with a request that the matter be acted upon by the Commission, if in the Board's judgment the matters at issue are of such a nature as to warrant Commission review of any decision which the Board might otherwise have made. If a majority of the members of the Commission then holding office vote to grant the Board's request, the matter shall be acted upon by the Commission.
- (c) Whenever the Commission determines that a matter pending before the Board involves a novel or important issue of law or policy, it may, on its own motion, by the vote of a majority of the members then holding office, direct that any matter before the Board be certified to the Commission for decision. However, no petition requesting the Commission to take such action will be entertained.
- (d) The Review Board shall decide each matter before it by majority vote in accordance with the Communications Act of 1934, as amended, rules and regulations, case precedent, and established policies of the Commission. In reviewing initial decisions referred to it, the Review Board is authorized to perform all of the review functions which would otherwise have been performed by the Commission under §§ 1.273–1.282 of this chapter.
- (e) The Review Board is composed of three or more Commission employees. Members of the Board are designated by the Commission, serve indefinitely on a full-time basis, and are responsible only to the Commission. Neither the Commission nor any of its members will

discuss the merits of any matter pending before the Board with the Board or any of its members.

- (f) A minimum of three members will participate in each case referred to the Board. A majority of the members who participate in a case shall constitute a quorum. Any member assigned to a case who is not present at oral argument may, after reading the transcript of oral argument, participate in the Board's decision. However, so far as practicable, all of the members of the Board assigned to a case shall hear oral argument.
- (g) Except in interlocutory matters, each opinion of the Review Board will be signed by one of its members, who shall be responsible for its preparation.

 [28 F.R. 12402, Nov. 22, 1963, as amended at 29 F.R. 6443, May 16, 1964; 33 F.R. 8228,

§ 0.365 Authority delegated to the Review Board on a regular basis.

June 1, 1968]

- (a) Review of initial decisions. Unless the Commission specifies to the contrary at the time of designation for hearing or otherwise, the Review Board shall review initial decisions of hearing examiners in all adjudicative proceedings (including mixed adjudicative and rule making proceedings), except for proceedings involving the renewal or revocation of a station license in the Broadcast Radio Services or the Common Carrier Radio Services.
- (b) Original action on interlocutory matters. In adjudicative proceedings conducted by hearing examiners (including mixed adjudicative and rule making proceedings), the Review Board shall take original action on the following interlocutory matters and upon any question with respect to such matters which is certified to it by the presiding examiner (see § 1.291 of this chapter):
- (1) Petitions to amend, modify, enlarge, or delete issues upon which the hearing was ordered.
- (2) Joint requests for approval of agreements filed pursuant to § 1.525 of this chapter and, if further hearing is not required on issues other than those arising out of the agreement, to terminate the proceeding and make appropriate disposition of all applications. (In considering such requests, the Review Board may in its discretion, hold informal conferences with counsel for parties to the proceeding.)
- (c) Action on interlocutory appeals from rulings of hearing examiners. The

Review Board shall act on interlocutory appeals from rulings of hearing examiners in adjudicative proceedings (including mixed adjudicative and rule making proceedings). See § 1.301 of this chapter.

(d) Action on pleadings filed in cases or matters which are before the Board. The Review Board shall act on all pleadings filed in cases or matters which are before the Board.

[29 F.R. 6443, May 16, 1964]

CHIEF, OFFICE OF OPINIONS AND REVIEW § 0.371 Authority delegated.

The Chief, Opinions and Review, is delegated authority to act upon the following matters in hearing proceedings which are pending before the Commission en banc:

- (a) Motions or petitions for extension of time.
 - (b) Pleadings which are moot.
- (c) To dismiss, as repetitious, any petition for reconsideration of a Commission order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order.
- (d) To issue orders, in accordance with Commission instructions, specifying or changing the day or hour of oral argument.
- (e) Requests for permission to file pleadings in excess of the length prescribed by the provisions of this chapter. See §§ 1.48 and 1.204 of this chapter.

[33 F.R. 8228, June 1, 1968, as amended at 33 F.R. 19181, Dec. 24, 1968]

DEFENSE AND EMERGENCY PREPAREDNESS DELEGATIONS

§ 0.381 Defense Commissioner.

The authority delegated to the Commission under Executive Order 11092 is redelegated to the Defense Commissioner.

[33 F.R. 8228, June 1, 1968]

§ 0.383 Emergency Relocation Board.

- (a) During any period in which the Commission is unable to function because of the circumstances set forth in § 0.186 (a), all work, business or functions of the Federal Communications Commission arising under the Communications Act of 1934, as amended, is assigned and referred to the Emergency Relocation Board.
- (b) The Board, acting by a majority thereof, shall have the power and au-

thority to hear and determine, order, certify, report or otherwise act as to any of the said work, business or functions so assigned or referred to it, and in respect thereof shall have all the jurisdiction and powers conferred by law upon the Commission, and be subject to the same duties and obligations.

(c) Any order, decision or report made or other action taken by the said Board in respect of any matters so assigned or referred shall have the same effect and force, and may be made, evidenced, and enforced in the same manner, as if made or taken by the Commission.

[28 F.R. 12402, Nov. 22, 1963, as amended at 33 F.R. 8228, June 1, 1968]

§ 0.385 FCC representatives assigned to Regional Preparedness Committees of the Office of Emergency Planning during a wartime emergency period.

Engineers in Charge of Field Engineering Bureau district offices at Boston, Massachusetts; Baltimore, Maryland; Atlanta, Georgia; Detroit, Michigan; Dallas, Texas; Denver, Colorado; San Francisco, California; and Seattle. Washington, are designated as FCC representatives to Regional Preparedness Committees. FCC Field Liaison Officers of the FCC Office of Emergency Communications are designated as alternates. These Committees will be activated in wartime emergency periods or during Civil Defense exercises. These FCC representatives will give advice and assistance to Regional Directors of the Office of Emergency Planning and to other Federal agencies within the respective Office of Emergency Planning region on matters relating to the functions and responsibilities of the FCC. They will also coordinate FCC operations within the region in the event of a regional "cut-off" situation. For the purpose of delegating authority to FCC representatives to act for the Commission, a re-gional "cut-off" situation shall be considered as existing when either no communication to higher authority within the FCC can be effected, or such communication as exists could not be expected to provide an answer from higher authority within the FCC in time to authorize urgent actions that the FCC representative determines cannot be delayed until such answer is received. The delegations of authority set forth in paragraphs (a), (b), and (c) of this section are made to these Engineers in Charge

or to their alternates, to enable them to carry out their responsibilities.

- (a) When communication is available to higher authority within the FCC. When instructions can be obtained from higher authority within the FCC, the authority of FCC representatives as-signed to Regional Preparedness Committees shall comprise that delegated to the Engineers in Charge in nonemergency periods under § 0.314. Action under this delegation must be consistent with any defense rules that supersede rules for normal conditions, wartime plans and emergency orders that may be adopted by the Commission, emergency legislation, Executive Orders, and any pertinent actions of other Government agencies taken pursuant to authority delegated to them under section 606 of the Communications Act of 1934, as amended.
- (b) During a temporary "cut-off" period when communication to higher authority within the FCC is disrupted. When instructions cannot be obtained from higher authority within the FCC and when immediate action must be taken, delegation of authority to FCC representatives assigned to Regional Preparedness Committees of the Office of Emergency Planning shall comprise the following in addition to the delegations included in paragraph (a) of this section:
- (1) Matters delegated to the Chief and Deputy Chief of the Field Engineering Bureau under § 0.311(b).

(2) Matters delegated to the Chief, Broadcast Bureau by § 0.281.

- (3) Matters delegated to the Chief, Safety and Special Radio Services Bureau by §§ 0.331-0.333, except for the authority to act on requests for waiver of rules governing the availability of frequencies below 25 megacycles in the Aviation and Maritime Services.
- (4) Matters delegated to the Chief, Common Carrier Bureau by §§ 0.291, 0.294, 0.303 (d)-(f), 0.304, 0.305, and 0.307.

(5) Matters delegated to the Chief Engineer by §§ 0.241 and 0.243.

- (6) Authority to act on requests for temporary authorizations for new or modified radio station operations, subject to the provisions of the Communications Act of 1934, as amended.
- (7) Authority to act on requests for temporary authorization for all classes of radio operators, subject to the provisions of the Communications Act of 1934, as amended.

- (8) Authority to waive temporarily any provisions of this chapter applicable to radio stations and radio operators, subject to the provisions of the Communications Act of 1934, as amended: Provided, however, That such authority shall not include waiver of rules governing the availability of frequencies below 25 megacycles in the Aviation and Maritime Services, nor any of the Emergency Broadcast System Rules.
- (c) General. (1) All authorizations granted pursuant to authority contained in paragraph (b) of this section shall be reported to higher Commission authority at the earliest opportunity. Such authorizations shall be subject to review and cancellation or revision, without hearing, by the FCC representatives assigned to Regional Preparedness Committees or by higher Commission authority when the need for such action arises.
- (i) Written authorizations granted pursuant to authority contained in paragraph (b) of this section shall show plainly on their face that they are temporary authorizations, not to exceed 30 days from the date of issue, and subject to review and cancellation or revision without hearing.
- (ii) Where immediate oral authorizations are necessary, the applicant shall be orally informed of the limitations enumerated above in this subparagraph and the oral authorization shall be followed as soon as possible by a written authorization bearing the same date of issue as the date of oral authorization. If the "cut-off" period exceeds 30 days in duration, such authorizations may be renewed for additional periods of 30 days each.
- (2) Actions taken under any delegation of authority must take into full account, and be in conformance with, any defense rules that supersede rules for normal conditions; wartime emergency plans and orders of the Commission; emergency legislation; Executive Orders; and any pertinent actions of other Government agencies taken pursuant to authority delegated to them under section 606 of the Communications Act of 1934, as amended.
- (3) No actions shall be taken under any delegation of authority until full consideration is given to the effect of such actions on the continuance of vital radio communications, both Government and non-Government, on a world-wide or nationwide basis and in adja-

cent regions. If the FCC representative assigned to the Regional Preparedness Committee of the Office of Emergency Planning determines that interference to radio operations outside of a "cut-off" region may result by authorizing new or modified radio facilities (for example, a change of frequency or power of a radio station), he shall withhold such authorization unless directed to grant the temporary authorization by higher authority.

(4) Coordination shall be effected, where possible, with the FCC representatives assigned to Regional Preparedness Committees of the Office of Emergency Planning in adjacent regions before acting to authorize temporary requests for

radio operations.

(5) New obligations incurred pursuant to delegation of authority specified in this section for personal service, procurement, contract agreements, and similar items shall not exceed the amount authorized by the Commission's Budget Officer in periodic emergency allotment authorizations.

[29 F.R. 14665, Oct. 28, 1964, as amended at 32 F.R. 20860, Dec. 28, 1967]

§ 0.386 Record of actions taken.

Actions taken under § 0.385(b) by FCC representatives assigned to Regional Preparedness Committees of the Office of Emergency Planning shall be reported in writing to the Chief, Field Engineering Bureau at the earliest opportunity.

[29 F.R. 15123, Nov. 28, 1964]

- § 0.387 Other defense and emergency preparedness delegations; cross reference.
- (a) For authority of the Chief of the Broadcast Bureau to issue National Defense Emergency Authorizations, see §§ 0.281(h) and 1.502.
- (b) For authority of the Chief and Deputy Chief of the Field Engineering Bureau to declare a general communications emergency, see § 0.311(b).

[28 F.R. 12411, Nov. 22, 1963, as amended at 29 F.R. 15923, Nov. 28, 1964]

Subpart C—General Information GENERAL

- § 0.401 Location of Commission offices.
- (a) The main offices of the Commission are located at 1919 M Street NW., Washington, D.C.
- (1) Documents submitted by mail to those offices should be addressed to:
- Federal Communications Commission, Washington, D.C. 20554.

- (2) Hand-carried applications accompanied by a filing fee should be delivered to:
- Mail and Files Division, Office of Executive Director, 1919 M Street NW., Washington, D.C.
- (3) Other hand-carried documents should be delivered to:
- Office of the Secretary, 1919 M Street NW., Washington, D.C.
- (b) The Commission's laboratory is located north of Laurel, Md. The mailing address is:
- Federal Communications Commission Laboratory, Post Office Box 40, Laurel, Md. 20810.
- (c) The locations of the field offices of the Field Engineering Bureau are listed in § 0.121.
- (d) The locations of the field offices of the Common Carrier Bureau are listed in § 0.94.
- (e) The Commission also maintains an office at Gettysburg, Pennsylvania. The mailing address for this office is:

Federal Communications Commission, 334 York Street, Gettysburg, Pennsylvania, 17325.

- (f) The location of the field offices of the Office of Emergency Communications are listed in $\S 0.184(c)$.
- [28 F.R. 12411, Nov. 22, 1963, as amended at 29 F.R. 12370, Aug. 28, 1964; 29 F.R. 14666, Oct. 28, 1964; 32 F.R. 12795, Sept. 7, 1967; 32 F.R. 20860, Dec. 28, 1967]

§ 0.403 Office hours.

The main offices of the Commission are open from 8 a.m. to 4:30 p.m., Monday through Friday, excluding legal holidays. The Immediate Office of the Secretary and the Mail and Files Branch, Office of the Executive Director, will remain open until 5 p.m. to receive documents filed with the Commission. The Office of Information will also remain open until 5 p.m.

[34 F.R. 9282, June 12, 1969]

§ 0.405 Statutory provisions.

The following statutory provisions, among others, will be of interest to persons having business with the Commission:

(a) The Federal Communications Commission was created by the Communications Act of 1934, 48 Stat. 1064, June 19, 1934, as amended, 47 U.S.C. 151-609.

- (b) The Commission exercises authority under the Submarine Cable Landing Act, 42 Stat. 8, May 27, 1921, 47 U.S.C. 34–39. See section 5 of Executive Order 10530, 19 F.R. 2709, May 10, 1954, as amended, 3 CFR 1965 ed., p. 463.
- (c) The Commission exercises authority under the Communications Satellite Act of 1962, 76 Stat. 419, August 31, 1962, 47 U.S.C. 701–744.
- (d) The Commission operates under the Administrative Procedure Act, 60 Stat. 237, June 11, 1946, as amended, originally codified as 5 U.S.C. 1001–1011. Pursuant to P.L. 89–554, September 6, 1966, 80 Stat. 378, the provisions of the Administrative Procedure Act now appear as follows in the Code:

§ 0.406 The rules and regulations.

Persons having business with the Commission should familiarize themselves with those portions of its rules and regulations pertinent to such business. All of the rules have been published and are readily available. See §§ 0.411(b), 0.412, and 0.415. For the benefit of those who are not familiar with the rules, there is set forth in this section a brief description of their format and contents.

(a) Format. The rules are set forth in the Code of Federal Regulations as Chapter I of Title 47. Chapter I is divided into parts numbered from 0-99. Each part, in turn, is divided into numbered sections. To allow for the addition of new parts and sections in logical sequence, without extensive renumbering. parts and sections are not always numbered consecutively. Thus, for example, Part 2 is followed by Part 5, and § 1.8 is followed by § 1.10; in this case, Parts 3 and 4 and § 1.9 have been reserved for future use. In numbering sections, the number before the period is the part number; and the number after the period locates the section within that part. Thus, for example, § 1.1 is the first section of Part 1 and § 5.1 is the first section in Part 5. Except in the case of accounting regulations (Parts 31-35), the period should not be read as a decimal point; thus, § 1.511 is not located between § 1.51 and § 1.52 but at a much later point in the rules. In citing the Code of Federal Regulations, the citation, 47 CFR 5.1, for example, is to § 5.1 (in Part 5) of Chapter I of Title 47 of the Code, and permits the exact location of that rule. No citation to other rule units (e.g., subpart or chapter) is needed.

- (b) Contents. Parts 0-19 of the rules have been reserved for provisions of a general nature. Parts 20-69 have been reserved for provisions pertaining to common carriers. Parts 70-79 have been reserved for provisions pertaining to broadcasting. Parts 80-99 have been reserved for provisions pertaining to the Safety and Special Radio Services. In the rules pertaining to common carriers, Parts 21, 23, and 25 pertain to the use of radio; Parts 31-66 pertain primarily to telephone and telegraph companies. In the rules pertaining to broadcasting, Part 74, Experimental, Auxiliary and Special Broadcast Services, also contains provisions for regulation of community antenna television (CATV) systems and community antenna relay (CAR) stations. Persons having business with the Commission will find it useful to consult one or more of the following parts containing provisions of a general nature in addition to the rules of the radio or wire communication service in which they are interested:
- (1) Part 0, Commission organization. Part 0 describes the structure and functions of the Commission, lists delegations of authority to the staff, and sets forth information designed to assist those desiring to obtain information from, or to do business with, the Commission. This Part is designed, among other things, to meet certain of the requirements of the Administrative Procedure Act, as amended.
- (2) Part 1, practice and procedure. Subpart A of Part 1 contains the general rules of practice and procedure. Except as expressly provided to the contrary, these rules are applicable in all Commission proceedings and should be of interest to all persons having business with the Commission. The subpart also contains certain other miscellaneous provisions. Subpart B contains the procedures applicable in formal hearing proceedings (see § 1.201). Subpart C contains the procedures followed in making or revising the rules and regulations. Subpart D contains rules applicable to applications for licenses in the Broadcast Radio Services, including the forms to be used,

the filing requirements, the procedures for processing and acting on such applications, and certain other matters. Subpart E contains general rules and procedures applicable to common carriers. Additional procedures applicable to certain common carriers by radio are set forth in Part 21. Subpart F contains rules applicable to applications for licenses in the Safety and Special Radio Services, including the forms to be used, the filing requirements, the procedures for processing and acting on such applications. and certain other matters. Subpart G contains rules pertaining to application filing fees. Subpart H, concerning ex parte presentations, sets forth standards governing communication with Commission personnel in hearing proceedings and contested application proceedings. Subparts G and H will be of interest to all applicants, and Subpart H will, in addition, be of interest to all persons involved in hearing proceedings.

(3) Part 2, frequency allocations and radio treaty matters; general rules and regulations. Part 2 will be of interest to all persons interested in the use of radio. It contains definitions of technical terms used in the rules and regulations; provisions governing the allocation of radio frequencies among the numerous uses made of radio (e.g., broadcasting, land mobile) and radio services (e.g., television, public safety), including the Table of Frequency Allocations (§ 2.106); technical provisions dealing with emissions; provisions dealing with call signs and emergency communications; provisions governing type acceptance and type approval of radio equipment; and a list of treaties and other international agreements pertaining to the use of radio.

(4) Part 5, experimental radio services (other than broadcast). Part 5 deals with the temporary use of radio frequencies for research in the radio art, for communication involving other research projects, and for the development of equipment, data, or techniques.

(5) Part 13, commercial radio operators. Part 13 describes the procedures to be followed in applying for a commercial operator license, including the forms to be used and the examinations given, and sets forth rules governing licensed operators. It will be of interest to applicants for such licenses, licensed operators, and the licensees of radio stations which may be operated only by persons holding a commercial radio operator license.

- (6) Part 15, radio frequency devices. Part 15 contains regulations designed to prevent harmful interference to radio communication from radio receivers and other devices which radiate radio frequency energy, and provides for the certification of radio receivers. It also provides for the certification of low power transmitters and for the operation of certificated transmitters without a license.
- (7) Part. 17, construction, marking, and lighting of antenna structures. Part 17 contains criteria for determining whether applications for radio towers require notification of proposed construction to the Federal Aviation Administration, and specifications for obstruction marking and lighting of antenna structures.
- (8) Part 18, industrial, scientific and medical equipment. Part 18 contains regulations designed to prevent harmful interference to radio communication from ultrasonic equipment, industrial heating equipment, medical diathermy equipment, radio frequency stabilized are welders, and other equipment which uses radio energy for purposes other than communication.

(9) Part 19, employee responsibilities and conduct. Part 19 prescribes standards of conduct for the members and staff of the Commission.

[32 F.R. 10571, July 19, 1967, as amended at 32 F.R. 12180, Aug. 24, 1967]

PRINTED PUBLICATIONS

Source: The provisions of §§ 0.411 to 0.420 appear at 32 F.R. 10571, July 19, 1967, unless otherwise noted.

§ 0.411 General reference materials.

The following reference materials are available in many libraries and may be purchased from the Superintendent of Documents, U.S. Government Printing Office, Washington, D.C. 20402:

- (a) Statutory materials. Laws pertaining to communications are contained in Title 47 of the United States Code. Laws enacted since the printing of the last supplement to the Code are printed individually as slip laws, and these are compiled chronologically in the United States Statutes at Large. The Acts of Congress from 1910–62 pertaining to radio have been compiled in a single volume, Radio Laws of the United States (1962 ed.). See §§ 0.405 and 0.414.
- (b) Regulatory materials—(1) The Code of Federal Regulations. The rules and regulations of the Commission are

contained in Chapter I of Title 47 of the Code of Federal Regulations, Chapter I is divided into the following 4 subchapters, which may be purchased separately: Subchapter A—General; Subchapter B— Common Carrier Services; Subchapter C—Broadcast Radio Services: and Subchapter D-Safety and Special Radio Services. Most persons will find that they need Subchapter A, containing the general rules, and one of the other volumes, depending upon their area of interest. These four volumes are revised annually to reflect changes in the rules. See §§ 0.406, 0.412, and 0.415. The Code of Federal Regulations is fully indexed and contains numerous finding aids. See 1 CFR Appendix C.

(2) The Federal Register. As rules are adopted, amended, or repealed, the changes are published in the Federal Register, which is published daily except on days following legal holidays. Notices of proposed rule making, other rule making documents, statements of general policy, interpretations of general applicability, and other Commission documents having general applicability and legal effect are also published in the Federal Register. The Federal Register is fully indexed and contains numerous finding aids. See 1 CFR Appendix C.

§ 0.412 Nongovernment publications.

(a) Pike and Fischer Radio Regulation. This multi-volume service contains the text of statutes, treaties and regulations pertaining to radio; Commission and court decisions; and other materials, including a digest. Information concerning this service may be obtained from Pike and Fischer, Inc., 1726 M Street NW., Washington, D.C. 20036.

(b) Rules Service Company service. This service contains Parts 0, 1, 17, 73, 74, and 87 of the rules and regulations and other materials. Information concerning this service may be obtained from the Rules Service Company, 1001 15th Street NW., Washington, D.C. 20005.

Note: Other published research materials pertaining to communications will be listed in this section upon request of the publisher.

§ 0.413 The Commission's printed publications.

The Commission's printed publications are described in §§ 0.414-0.420. These publications may be purchased from the Superintendent of Documents, U.S. Government Printing Office, Washington, D.C. 20402. The Commission does not

furnish copies of these publications but will furnish a price list (Administration Bulletin No. 1) upon request. Requests for copies of that list should be directed to the Office Services Division, Office of Executive Director, Federal Communications Commission, Washington, D.C. 20554.

§ 0.414 The Communications Act and other statutory materials.

This publication, with packets of revised pages, contains the Communications Act of 1934, with amendments through 1964; the Administrative Procedure Act, with amendments through 1964; the Judicial Review Act; the Communications Satellite Act of 1962; and selected sections of the Criminal Code pertaining to communications. It also contains indexes to the Communications Act and the Administrative Procedure Act. Persons who do not have ready access to the United States Code, or who refer frequently to these materials, may find this volume to be useful.

§ 0.415 The rules and regulations (loose-leaf service).

In this service, the rules and regulations are divided into 10 volumes, each containing several related parts. Each volume may be purchased separately from the Superintendent of Documents. The purchase price for a volume includes a subscription to replacement pages reflecting changes in the rules contained therein until such time as the volume is revised. Each volume is revised periodically, depending primarily on the frequency with which the rules it contains have been amended. When a volume is revised, the revised volume and replacement pages therefor will be furnished to those who renew their subscriptions.

§ 0.416 The Federal Communications Commission Reports.

All documents currently adopted by the Commission having precedential or historical significance are published in the FCC Reports. The FCC Reports are published weekly in pamphlet form. The pamphlets are available on a subscription basis, and are subsequently compiled and published in bound volumes. Earlier volumes contain Commission decisions and reports but are less comprehensive than those currently being published. Supplements (to those earlier volumes) containing additional documents having precedential or historical

significance will be issued from time to time. Current bound volumes contain indexes, tables of cases and other finding aids.

§ 0.417 The Annual Reports.

At the end of each fiscal year, the Commission publishes an Annual Report containing general information concerning the Commission and the history of regulation, a summary of developments during the year, and selected industry statistics.

§ **0.420** Other Commission publications.

The following additional Commission publications may be purchased from the Superintendent of Documents:

(a) Statistics of Commnnications Common Carriers (for the year 1965).

(b) Study Guide and Reference Material for Commercial Radio Operator Examinations (May 1965).

Figure M-3, Estimated AM Ground Conductivity of the United States (set of two maps).

(d) Television Network Program Procurement Report, 2d Interim Report, Part 2, by the Office of Network Study.

FORMS AND DOCUMENTS AVAILABLE UPON REQUEST

§ 0.421 Application forms.

All forms for use in submitting applications for radio authorizations, together with instructions and information as to filing such forms, may be obtained at the Washington offices of the Commission or at any of the engineering field offices listed in § 0.121. For information concerning the forms to be used and filing requirements, see Subparts D, E, F, and G of Part 1 of this chapter and the appropriate substantive rules.

[32 F.R. 10572, July 19, 1967]

§ 0.422 Current action documents and public notices.

A limited number of copies of the text of documents adopted by the Commission, public notices of Commission actions, and other public releases is made available at the Office of Information when they are issued. Back issues of public releases are available for inspection in this office.

[32 F.R. 10572, July 19, 1967, as amended at 32 F.R. 20860, Dec. 28, 1967]

§ 0.423 Information bulletins.

The bulletins listed in this section have been prepared with the specific purpose of providing information to the public concerning communications and the Federal Communications Commission. Copies of these bulletins are available upon request. Many of the bulletins contain references to additional materials and the manner in which they may be obtained, including some which are available from the Commission without charge upon request.

(a) Copies of the following information bulletins issued by the Commission are available in the Office of Information and will be furnished upon request.

(1) An A-B-C of the FCC (No. 3-G).

(2) Radio Stations and Other Lists (No. 4-G)

(3) Publications and Services (No. 6-G).

(4) A Short History of Electrical Communication (No. 7-G)

(5) Radio Station Call Signs (No. 13-G).

(6) Regulation of Wire and Radio Communication (No. 14-G).

(7) Frequency Allocation (No. 15-G).

(8) Memo to All Young People Interested in Radio (No. 17-G).

(9) Letter to a Schoolboy (No. 18-G).

(10) Policing the Airwaves and Other FCC Field Services (No. 19-G).

(11) Broadcast Application and Hearing Procedures (No. 1-B).

(12) Broadcast Primer (No. 2-B).

(13) Educational Television (No. 16-B).

(14) Subscription TV and the FCC (No. 20-B).

(15) Educational Radio (No. 21-B).

(16) Common Carrier Primer (No. 12-C).

(17) Safety and Special Radio Services Primer (No. 11-S).

(b) Copies of the following information bulletins issued by the Office of Chief Engineer are available in that office and will be furnished upon request:

(1) **Type** Approved Miscellaneous

Equipment (OCE Bull. No. 5).

(2) Type Approved Medical Diathermy Equipment (OCE Bull. No. 7).

(3) Industrial Radio Frequency Heaters Require Periodic Inspection (OCE Bulletin No. 8).

(4) Attachments to Type Approved Equipment Illegal (OCE Bulletin No. 10).

(5) Does My Transmitter Need a License (OCE Bulletin No. 11).

(6) Operation in the Broadcast Band Without a License (OCE Bulletin No. 12).

- (7) Type Approved Wireless Microphones and Telemetering Transmitters (OCE Bulletin No. 13).
- (8) Editorial Revision of FCC Rules, Parts 15 and 18 (OCE Bulletin No. 14).
- (9) Type Acceptance Program (OCE Bulletin No. 15).
- (c) Copies of the following information bulletins issued by the Safety and Special Radio Services Bureau are available from the Office Services Division, Office of Executive Director, upon request:
- (1) Citizens Radio Service (SS Bulletin No. 1001).
- (2) Use of Citizens Radio by Telephone Answering Services and Similar Organizations (SS Bull. No. 1001d).
- (3) Citizens Radio Service—Selecting Class C and Class D Station Equipment (SS Bulletin No. 1001g).
- (4) Aircraft Radio Station (SS Bulletin No. 1002).
- (5) Aeronautical Advisory Stations (SS Bulletin No. 1002a).
- (6) Aeronautical Public Service Stations (SS Bulletin No. 1002c).
- (7) Amateur Radio Service (SS Bulletin No. 1003).
- (8) Amateur Radio Operation Away from the Licensed Location (SS Bulletin No. 1003b).
- (9) International Amateur Radiocommunication (SS Bulletin No. 1103c).
- (10) Assignment of Amateur Radio Station Call Signs (SS Bulletin No. 1003d).
- (11) Renewal of Amateur Radio Licenses (SS Bulletin No. 1003e).
- (12) Reciprocal Amateur Operation (SS Bulletin No. 1003f).
- (13) Land Transportation Radio Services (SS Bulletin No. 1004).
- (14) Industrial Radio Services (SS Bulletin No. 1005).
- (15) Use of the Same Transmitting Equipment by More than One Station Licensee in the Public Safety, Industrial and Land Transportation Radio Services (SS Bulletin No. 1006a).
- (16) Ship Radiotelephone and Radar (SS Bulletin No. 1007).
- (17) Public Safety Radio Services (SS Bulletin No. 1009).
- (18) Study Questions for Amateur Novice Class Examination (SS Bulletin No. 1035).
- (19) Mutual Recognition of Certain Mobile and Amateur Radio Licenses Issued by the United States or Canada (SS Bulletin No. 1065).

- (20) Notice to Licensees and Operators of Land Mobile Radio Stations (SS Bulletin No. 1097).
- [32 F.R. 10572, July 19, 1967, as amended at 32 F.R. 12180, Aug. 24, 1967; 32 F.R. 13125, Sept. 15, 1967]

LISTS CONTAINING INFORMATION COMPILED BY THE COMMISSION

SOURCE: The provisions of §§ 0.431 to 0.434 appear at 32 F.R. 10573, July 19, 1967, unless otherwise noted.

§ 0.431 The FCC Service Frequency Lists.

Lists of frequency assignments to radio stations authorized by the Commission are recapitulated periodically by means of a machine record system. All stations licensed by the Commission are included. except the following: Aircraft, Amateur, Citizens (except Class A), Civil Air Patrol, and Disaster. The resulting documents, the FCC Service Frequency Lists, consist of several volumes arranged by nature of service, in frequency order, including station locations, call signs and other technical particulars of each assignment. These documents are available for public inspection at each of the Commission's Field Engineering Bureau field offices (see § 0.121) and, in Washington, D.C., at the Commission's Broadcast and Docket Reference Room and in the offices of the Chief Engineer. They may be purchased from Cooper-Trent, Inc., 1130 19th Street NW., Washington, D.C. 20006.

§ 0.432 The NARBA List.

Pursuant to the North American Regional Broadcast Agreement and the United States/Mexican Agreement, appropriate countries are notified of standard broadcast station assignments as they are made. The information thus supplied by notice includes frequency, station location, call letters, power and other technical particulars. Every months, a recapitulative list containing this information for all existing standard broadcast stations, arranged in frequency order, is prepared by the Commission. This is the so-called NARBA List. These lists are available for public examination at each of the Commission's Field Engineering Bureau field offices (see § 0.121) and, in Washington, D.C., at the Commission's Broadcast and Docket Reference Room, They may be purchased from Cooper-Trent, Inc., 1130 19th Street NW., Washington, D.C. 20006.



§ 0.433 The Radio Equipment Lists.

Lists of type approved and type accepted equipment (the Radio Equipment Lists) are prepared periodically by the Commission. These documents are available for public inspection at each of the Commission's Field Engineering Bureau field offices (see § 0.121) and in Washington, D.C., in the offices of the Chief Engineer. They may be purchased from Cooper-Trent, Inc., 1130 19th Street NW., Washington, D.C. 20006.

§ 0.434 Lists of authorized broadcast stations and pending broadcast applications.

Periodically the Commission prepares lists containing information about authorized broadcast stations and pending applications for such stations. These lists, which are prepared by an addressing machine, contain frequency, station locations, and other particulars. They are available for public examination at the Commission's Broadcast and Docket Reference Room, Washington, D.C., and may be purchased from Cooper-Trent, Inc., 1130 19th Street NW., Washington, D.C. 20006.

(a) For standard broadcast stations the lists are arranged as follows:

(1) Authorized stations arranged in frequency order, alphabetically by State and city, and by call letters.

(2) Pending applications for new stations and major changes in existing facilities arranged in frequency order and alphabetically by State and city.

(b) For FM broadcast stations the lists

are arranged as follows:

- (1) Authorized stations arranged by call letters and alphabetically by State and city.
- (2) Pending applications for new stations and major changes in existing facilities arranged alphabetically by State and city.
- (c) For television broadcast stations only one list is prepared. This list contains authorized stations and pending applications for new stations and major changes in existing facilities, and is arranged alphabetically by State and city.

(d) For television broadcast translator stations only one list is prepared. This list contains authorized stations and pending applications for new stations and major changes in existing facilities and is arranged alphabetically by State and city.

PUBLIC INFORMATION AND INSPECTION OF RECORDS

SOURCE: The provisions of §§ 0.441 to 0.467 appear at 32 F.R. 10573, July 19, 1967, unless otherwise noted.

§ 0.441 General.

Any person desiring to obtain information may do so by writing or coming in person to any of the Commission's offices. A broader range of information and more comprehensive information facilities are available at the Commission's main office in Washington, D.C., however, and inquiries of a general nature should ordinarily be submitted to that office.

§ 0.443 General information office.

The Office of Information is located at 1919 M Street, N.W., Washington, D.C. Here the public may obtain copies of public notices of Commission actions, formal documents adopted by the Commission and other public releases as they are issued. Back issues of public releases are available for inspection in this Office. Copies of fact sheets which answer recuring questions about the Commission's functions may be obtained from this Office

[32 F.R. 10573, July 19, 1967, as amended at 32 F.R. 13125, Sept. 15, 1967]

§ 0.445 Publication, availability and use of opinions, orders, policy statements, interpretations, administrative manuals, and staff instructions.

(a) All opinions and orders of the Commission (including concurring and dissenting opinions) are mailed to the parties and, as part of the record, are available for inspection in accordance with §§ 0.453 and 0.455.

(b) All final decisions and other documents currently adopted by the Commission having precedential or historical significance are published in the FCC Reports. Older materials of this nature are either published in the FCC Reports or are in the process of being published. In the event that such other materials are not yet published in the FCC Reports, reference should be made to Pike and Fischer Radio Regulations. See §§ 0.412(a) and 0.416.

(c) All rule making documents are published in the Federal Register. See

§ 0.411(b)(2).

(d) Formal policy statements and interpretations designed to have general applicability and legal effect are published in the FEDERAL REGISTER and the

FCC Reports. See §§ 0.411(b) (2) and 0.416. Commission decisions and other Commission documents not entitled formal policy statements or interpretations may contain substantive interpretations and statements regarding policy, and these are published as part of the document in the FCC Reports. Gereral statements regarding policy and interpretations furnished to individuals, in correspondence or otherwise, are not ordinarily published. A series of individual interpretations may be collected and published in the Federal Register and the

FCC Reports. (e) If the documents described in paragraphs (a)-(d) of this section are published in the FEDERAL REGISTER, the FCC Reports, or Pike and Fischer Radio Regulations, they may be relied upon, used or cited as precedent by the Commission or private parties in any matter. If they are not so published, they may not be relied upon, used or cited as precedent, except against persons who have actual notice of the document in question or by such persons against the Commission. No person is expected to comply with any requirement or policy of the Commission unless he has actual notice of that requirement or policy or unless a document stating it has been published as provided in this parargaph. Nothing in this paragraph, however, shall be construed as precluding a reference to the rationale set forth in a recent document which is pending publication if the requirement or policy to which the rationale relates is contained in a published document or if actual notice of that requirement or policy has been given.

(f) The FEDERAL REGISTER, the FCC Reports, and Pike and Fischer Radio Regulations are indexed. If the documents described in paragraphs (a)-(d) of this section are not published, they are neither indexed nor relied upon, except as provided in paragraph (e) of this section.

(g) There are two Commission staff manuals, the FCC Administrative Manual and the FEB Manual. They have not been published. The FCC Administrative Manual (excepting Part IX, concerning Civil Defense, which contains materials classified under E.O. 10501) is available for inspection in the Office of the Executive Director. Portions of the FEB Manual which pertain to administrative matters are available for inspection in the Field Engineering Bureau. Portions of the FEB Manual which pertain to enforcement matters are not available for

inspection. (See Attorney General's Memorandum on the Public Information Section of the Administrative Procedure Act, June 1967, at pages 16–17.) The manuals are not indexed but are organized by subject, with tables of contents, and the materials contained therein can be located without difficulty.

(h) Subparts A and B of this part describe the functions of the staff and list the matters on which authority has been delegated to the staff. Except as provided in paragraph (g) of this section, all general instructions to the staff and limitations upon its authority are set forth in those subparts. As part of the Commission's rules and regulations, the provisions of these subparts are indexed in the Federal Regulations. Instructions to the staff in particular matters or cases are privileged and are not published or made available for public inspection.

(i) To the extent required to prevent a clearly unwarranted invasion of personal privacy, the Commission may delete identifying details when it makes available or publishes any document described in this section. The justification for any such deletion will be fully explained in a preamble to the document.

§ 0.451 Inspection of records.

(a) Records which are routinely available for public inspection. Sections 0.453 and 0.455 list examples of the Commission's records which are routinely available for public inspection, and the places at which those records may be inspected. Subject to the following limitations, and to the provisions of § 0.466, those records will be made available for inspection to any person upon request:

(1) The person desiring to inspect those records must appear at the location specified during the office hours of the Commission and must inspect the records at that location.

(2) The request must be reasonable in scope, and the records in question must be so identified as to permit their location by staff personnel without undue disruption of their regular duties. The information needed to locate the records will vary, depending upon the records requested. Advice concerning the kind of information needed to locate particular records will be furnished in advance upon request. Members of the public will not be given access to the area in which the records are kept and will not be permitted to search the files.

(3) Current records may be in use by the staff when the request is made. Older records may have been forwarded to another location for storage. In these and similar circumstances, there may be a delay in furnishing the records requested. To avoid inconvenience in such circumstances, arrangements may be made in advance, by telephone or correspondence, to make the records available for inspection on a specified date.

(b) Records not routinely available for public inspection. (1) Section 0.457 lists records which are not routinely available for public inspection. In some cases, the Commission is prohibited from permitting the inspection of records. In other cases, the records are the property of another agency, and the Commission has no authority to permit their inspection. In still other cases, the Commission is authorized, for reasons of policy, to withhold records from public inspection, but is not required to do so. The statutory basis for withholding records from public inspection and the underlying policy considerations are outlined briefly in § 0.457, with the intent of assisting those who may wish to file requests for inspection under § 0.461. See subparagraph (3) of this paragraph.

(2) Section 0.459 governs requests from members of the public that materials they submit to the Commission not be made available for public inspection.

- (3) Except where the disclosure of records by the Commission is prohibited or where the records are the property of another agency, the Commission will consider requests that records withheld from public inspection be made available for inspection, and reach a judgment whether such requests should be granted in the public interest, taking into account the policies of Public Law 89-487, 5 U.S.C. 552. If the records are the property of another agency, the request will be referred to that agency. Procedures governing requests for inspection are set forth in §§ 0.461 and 0.466. Procedures governing demands by competent authority for inspection of records are set forth in § 0.463.
- (4) Except as provided in §§ 0.461 and 0.463, no officer or employee of the Commission shall permit the inspection of records which are not routinely available for inspection under § 0.453 or § 0.455, or disclose information contained therein.
- (c) Records not listed in § 0.453 or § 0.455. To be as helpful as possible to the public, numerous examples of records

which are routinely available for inspection are listed in §§ 0.453 and 0.455. Though the examples cover the bulk of the Commission's records and should cover most requests for inspection, the listing is inevitably not complete. If such example listing proves helpful, it may be supplemented from time to time.

(d) Copies. Section 0.465 applies to requests for copies of Commission records which are routinely available for public inspection under §§ 0.453 and 0.455 and those which are made available for inspection under § 0.461. Section 0.467 applies to certified copies of Commission records.

§ 0.453 Public reference rooms.

The Commission maintains the following public reference rooms at its offices in Washington, D.C.:

(a) The Broadcast and Dockets Reference Room. The following documents, files and records are available for inspection at this location:

- (1) Files containing the record of all docketed cases. A file is maintained for each docketed hearing case and for each docketed rule making proceeding. Cards summarizing the history of such cases are available for inspection in the Dockets Division.
- (2) Broadcast applications and related files.
- (3) Files containing petitions for rule making and related papers.
- (4) Rulings under the fairness doctrine and section 315 of the Communications Act, and related materials.
- (b) The Amateur License Reference Room. Information concerning amateur radio operators is available for inspection at this location.
- (c) The Library. Various legal and technical publications, and legislative history compilations, related to communications are available for inspection in the Library.

[32 F.R. 10573, July 19, 1967, as amended at 33 F.R. 5302, Apr. 3, 1968]

§ 0.455 Other locations at which records may be inspected.

Except as provided in §§ 0.453, 0.457, and 0.459, records are routinely available for inspection in the offices of the Bureau or Office which exercises responsibility over the matters to which those records pertain (see § 0.5), or will be made available for inspection at those offices upon request. Upon inquiry to the appropriate Bureau or Office, persons de-

siring to inspect such records will be directed to the specific location at which the particular records may be inspected. A list of Bureaus and Offices and examples of the records available at each is set out below.

(a) Office of Chief Engineer. (1) Experimental application and license files.
(2) The Master Frequency Records

(Standard Form 128).

- (b) Broadcast Bureau. (1) Applications for broadcast authorizations and related files are available for public inspection in the Broadcast and Dockets Reference Room. See § 0.453(a) (2). Certain broadcast applications, reports and records are also available for inspection in the community in which the main studio of the station in question is located or proposed to be located. See § 1.526 of this chapter.
- (2) Ownership reports filed by licensees of broadcast stations pursuant to § 1.615 of this chapter.
- (3) Contracts relating to network service filed on or after the 1st day of May 1969, under § 1.613 of this chapter.
- (c) Common Carrier Bureau. (1) Annual reports filed by carriers and certain affiliates under § 43.21 of this chapter.

(2) Monthly reports filed by carriers under § 43.31 of this chapter.

- (3) Reports on pensions and benefits filed by carriers under § 43.42 of this chapter.
- (4) Reports of proposed changes in depreciation rates filed by carriers under § 43.43 of this chapter.
- (5) Reports regarding division of international telegraph communication charges filed under § 43.53 of this chapter.
- (6) Reports regarding services performed by telegraph carriers filed under § 43.54 of this chapter.
- (7) Reports of public coast station operators filed under § 43.71 of this chapter.
- (8) Valuation reports filed under section 213 of the Communications Act, including exhibits filed in connection therewith, unless otherwise ordered by the Commission, with reasons therefor, pursuant to section 213(f) of the Communications Act. See § 0.457(c) (2).
- (9) A list of other reports filed by common carriers.
- (10) Contracts and other arrangements filed under § 43.51 and reports of negotiations regarding foreign communication matters filed under § 43.52 of this chapter, except for those kept confiden-

tial by the Commission pursuant to section 412 of the Communications Act. See § 0.457(c) (3).

(11) Tariff schedules for all charges for interstate and foreign wire or radio communications filed pursuant to section 203 of the Communications Act, all documents filed in connection therewith, and all communications related thereto.

(12) All applications for common carrier authorizations, both radio and non-

radio, and files relating thereto.

(13) All formal and informal complaints against common carriers filed under §§ 1.711-1.735 of this chapter, all documents filed in connection therewith, and all communications related thereto.

(14) Files relating to submarine cable landing licenses, except for maps showing the exact location of submarine cables, which are withheld from inspection under section 4(j) of the Communi-

cations Act. See § 0.457(c) (1) (i).

(d) Safety and Special Radio Services Bureau. (1) All applications for authorizations in the Safety and Special Radio Services and files relating thereto. These materials are available at the offices of the Divisions of the Bureau which process the applications in question. See § 1.951 of this chapter. Information concerning amateur radio operators is available for inspection at the Amateur License Reference Room (see § 0.453(b)).

(e) Field Engineering Bureau. (1) Commercial radio operator application files. See, however, § 0.457(f) (3).

(2) Files pertaining to the certification of plants or equipment under Part 18 of this chapter.

(f) Office of the Secretary. (1) All minutes of Commission actions, containing a record of all final votes, except for minutes of actions on classified matters and internal management matters as provided in § 0.457(b) (1) and (c) (1) (ii). These minutes are available for inspection in the Minute and Rules Division.

(2) Files containing information concerning the history of the Commission's rules. These files are available for inspection in the Minute and Rules

Division.

(g) Office of Information. See § 0.443.

- (h) The Commission's offices in Gettysburg, Pennsylvania. (1) Amateur and Citizens Radio Service application files.
- (2) Commercial radio operator application files. See, however, § 0.457(f) (3).

(i) CATV Task Force. CATV petitions, requests and related files. [32 F.R. 10573, July 19, 1967, as amended at 32 F.R. 13125, Sept. 15, 1967; 34 F.R. 5949, Mar. 29, 1969]

§ 0.456 General correspondence files.

Due to the general nature of the Commission's correspondence files, the contents of those files will be made available for inspection under procedures set forth in § 0.461. Before correspondence is made available for inspection under the provisions of that section, it will be reviewed by the Commission's staff to determine whether it should be withheld from inspection under § 0.457. In view of the burden which could be imposed by requests lacking in specificity, persons desiring to inspect materials contained therein will be required to identify those materials with particularity. Requests of a general nature will not be granted. Details of a personal nature, including the name and address of the correspondent, may be deleted from correspondwhich is made available for inspection, if their deletion is warranted under § 0.457(f) or (g).

§ 0.457 Records not routinely available for public inspection.

The records listed in this section are not routinely available for public inspection. The records are listed in this section by category, according to the statu-tory basis for withholding those records from inspection; and under each category, if appropriate, the underlying policy considerations affecting the withholding and disclosure of records in that category are briefly outlined. Except where the records are not the property of the Commission or where the disclosure of those records is prohibited by law. the Commission will entertain requests from members of the public under § 0.461 for permission to inspect particular records withheld from inspection under the provisions of this section, and will weigh the policy considerations favoring non-disclosure against the reasons cited for permitting inspection in the light of the facts of the particular case. In making such requests, it is important to appreciate that there may be more than one basis for withholding particular records from inspection. The listing of records by category is not intended to imply the contrary but is solely for the information and assistance of persons making such requests.

(a) Materials that are specifically required by executive order to be kept secret in the interest of the national defense or foreign policy, 5 U.S.C. 552(b) (1). (1) E.O. 10450, "Security Requirements for Government Employees," 18 F.R. 2489, April 27, 1953, 3 CFR, 1949–1953 Comp., p. 936. Pursuant to the provisions of E.O. 10450, reports and other material and information developed in security investigations are the property of the investigative agency. If they are retained by the Commission, it is required that they be maintained in confidence and that no access be given to them without the consent of the investigative agency. Such materials and information will not be made available for public inspection. See also paragraphs (f) and (g) of this section.

(2) E.O. 10501, "Safeguarding Official Information in the Interests of the Defense of the United States," 18 F.R. 7049, November 10, 1953, as amended, 3 CFR, 1965 ed., p. 450. E.O. 10501, as amended, provides for the classification of official information which requires protection in the interests of national defense, and prohibits the disclosure of classified information except as provided therein. Classified materials and information will not be made available for public inspection. See also, E.O. 10033, February 8, 1949, 14 F.R. 561, 3 CFR, 1949–1953 Comp., p. 226, and 47 U.S.C. 154(j).

(b) Materials that are related solely to the internal personnel rules and practices of the Commission, 5 U.S.C. 552(b)(2). (1) Materials related solely to internal management matters, including minutes of Commission actions on such matters. Such materials may be made available for inspection under § 0.461, however, unless their disclosure would interfere with or prejudice the performance of the internal management functions to which they relate, or unless their disclosure would constitute a clearly unwarranted invasion of personal privacy (see paragraph (f) of this section).

(2) Materials relating to the negotiation of contracts.

(c) Materials that are specifically exempted from disclosure by statute, 5 U.S.C. 552(b)(3). The Commission is authorized under the following statutory provisions to withhold materials from public inspection:

(1) Section 4(j) of the Communications Act, 47 U.S.C. 154(j), provides, in part, that, "The Commission is author-

ized to withhold publication of records or proceedings containing secret information affecting the national defense." Pursuant to that provision, it has been determined that the following materials should be withheld from public inspection (see also paragraph (a) of this section):

- (i) Maps showing the exact location of submarine cables.
- (ii) Minutes of Commission actions on classified matters.
- (iii) Maps of nation-wide point-topoint microwave networks.
- (2) Under section 213(f) of the Communications Act, 47 U.S.C. 213(f), the Commission is authorized to order, with the reasons therefor, that records and data pertaining to the valuation of the property of common carriers and furnished to the Commission by the carriers pursuant to the provisions of that section, shall not be available for public inspection. If such an order has been issued, the data and records will be withheld from public inspection, except under the provisions of § 0.461. Normally, however, such data and information is available for inspection. See § 0.455(c) (8).
- (3) Under section 412 of the Communications Act, 47 U.S.C. 412, the Commission may withhold from public inspection certain contracts, agreements and arrangements between common carriers relating to foreign wire or radio communication. Reports of negotiations regarding such foreign communication matters, filed by carriers under § 43.52 of this chapter, may also be withheld from public inspection under section 412. Any person may file a petition requesting that such materials be withheld from public inspection. To support such action, the petition must show that the contract, agreement or arrangement relates to foreign wire or radio communications; that its publication would place American communication companies at a disadvantage in meeting the competition of foreign communication companies; and that the public interest would be served by keeping its terms confidential. If the Commission orders that such materials be kept confidential, they will be made available for inspection only under the provisions of § 0.461.
- (4) Section 605 of the Communications Act, 47 U.S.C. 605, provides, in part, that, "no person not being authorized by the sender shall intercept any communi-

cation [by wire or radio] and divulge or publish the existence, contents, substance, purport, effect, or meaning of such intercepted communications to any person." In executing its responsibilities, the Commission regularly monitors radio transmissions (see § 0.116). Except as required for the enforcement of the communications laws, treaties and the provisions of this chapter, or as authorized in section 605, the Commission is prohibited from divulging information obtained in the course of these monitoring activities; and such information, and materials relating thereto, will not be made available for public inspection.

(5) Section 1905 of the Criminal Code, 18 U.S.C. 1905, prohibits the unauthorized disclosure of certain confidential information. See paragraph (d) of this section.

(d) Trade secrets and commercial or financial information obtained from any person and privileged or confidential, 5 U.S.C. 552(b) (4) and 18 U.S.C. 1905. Section 552(b)(4) is specifically applicable to trade secrets and commercial or financial information but is not limited to such matters. Under this provision, the Commission is authorized to withhold from public inspection materials which would be privileged as a matter of law if retained by the person who submitted them and materials which would not customarily be released to the public by that person, whether or not such materials protected from disclosure by a privilege. See, Attorney General's Memorandum on the Public Information Section of the Administrative Procedure Act, June 1967, at pages 32-34.

(1) Materials submitted to the Commission which contain trade secrets, or which contain commercial, financial or technical data which would customarily be guarded from competitors by the person submitting it, will not ordinarily be made available for inspection. A persuasive showing as to the reasons for inspection of such materials, and as to the Commission's authority to make disclosure in view of 18 U.S.C. 1905, will be required in requests for inspection submitted under § 0.461. This category includes the following materials:

(i) Financial reports submitted by licensees of broadcast stations pursuant to § 1.611 of this chapter or by radio and television networks.

(ii) Technical data submitted in connection with type acceptance, type ap-

proval or certification of equipment, except as set out in the Radio Equipment Lists. See § 0.433.

(2) Prior to July 4, 1967, the rules and regulations provided that certain materials submitted to the Commission would not be made available for public inspection or provided assurance, in varying degrees, that requests for non-disclosure of certain materials would be honored. See, e.g., 47 CFR (1966 ed.) 2.557, 5.204, 5.255, **15.70**, 21.406, 81.506, 83.436, 87.153, 89.215, 91.208, 91.605 and 93.208. Materials submitted under these provisions are not routinely available for public inspection. If a request for inspection is submitted under § 0.461, the Commission will then determine whether it is appropriate to withhold the materials in question from public inspection. To the extent that such materials were accepted on a confidential basis under the then existing rules, no disclosure of such materials will be made, absent a most compelling showing. See § 0.461 for provision for comments on a request for disclosure. The rules referred to above are superseded by the provisions of this subpart.

(3) Requests that materials submitted to the Commission on or after July 4, 1967, be withheld from public inspection are governed by § 0.459.

Note: Section 1905 of the Criminal Code, 18 U.S.C. 1905, prohibits the unauthorized disclosure of certain confidential information by employees of the United States. That section reads as follows:

Section 1905. Disclosure of confidential in-

formation generally.

Whoever, being an officer or employee of the United States or of any department or agency thereof, publishes, divulges, discloses, or makes known in any manner or to any extent not authorized by law any information coming to him in the course of his employment or official duties or by reason of any examination or investigation made by, or return, report or record made to or filed with, such department or agency or officer or employee thereof, which information concerns or relates to the trade secrets, processes. operations, style of work, or apparatus, or to the identity, confidential statistical data, amount or source of any income, profits, losses, or expenditures of any person, firm, partnership, corporation, or association; or permits any income return or copy thereof or any book containing any abstract or particulars thereof to be seen or examined by any person except as provided by law; shall be fined not more than \$1,000, or imprisoned not more than one year, or both; and shall be removed from office or employment (June 25, 1948, ch. 645, 62 Stat. 791).

(e) Interagency andintra-agency memorandums or letters, 5 U.S.C. 552(b) (5). Interagency and intra-agency memorandums or letters and the work papers of members of the Commission or its staff will not be made available for public inspection, except in accordance with the procedures set forth in § 0.461. Only if it is shown in a request under § 0.461 that such a communication would be routinely available to a private party through the discovery process in litigation with the Commission will the communication be made available for public inspection. Normally such papers are privileged and not available to private parties through the discovery process, since their disclosure would tend to restrain the commitment of ideas to writing, would tend to inhibit communication among Government personnel, would, in some cases, involve premature disclosure of their contents.

(f) Personnel, medical and other files whose disclosure would constitute a clearly unwarranted invasion of personal privacy, 5 U.S.C. 552(b)(6). (1) Under Executive Order 10561, 19 F.R. 5963, September 13, 1954, 3 CFR 1954-1958 Comp., page 205, the Commission maintains an Official Personnel Folder for each of its employees. Such folders are under the jurisdiction and control, and are a part of the records, of the U.S. Civil Service Commission. Except as provided in the rules of the Civil Service Commission (5 CFR 294.701-294.703), such folders will not be made available for public inspection by the Commission. In addition, other records of the Commission containing private, personal or financial information concerning particular employees will be withheld from public inspection.

(2) [Reserved]

(3) Information submitted to the Commission by applicants for commercial radio operator licenses concerning the character and mental or physical health of the applicant is available for inspection only under procedures set forth in § 0.461. Except in this respect, or where other aspects of a similar private nature warrant nondisclosure, commercial radio operator application files are available for inspection.

- (g) Investigatory files compiled for law enforcement purposes except to the extent available by law to a private party. 5 U.S.C. 552(b) (7). Papers relating to the institution or conduct of an investigation are placed in an investigatory file. Except as provided in this paragraph, such files are not available for public inspection.
- (1) Complaints. Except with respect to complaints filed under Section 208 of the Communications Act (see § 0.455(c)). when a complaint is received by the Commission, an initial determination is made as to whether the matters related in the complaint warrant (or may warrant) an investigation. If it is determined that such matters warrant (or may warrant) an investigation, the complaint is placed in an investigatory file. A complaint which is placed in an investigatory file may be made available for inspection upon request if it appears that its disclosure will not prejudice the conduct of the investigation (e.g., if information concerning the complaint has otherwise been disclosed). Except as provided in subdivisions (i) and (ii) of this subparagraph, the complaint will be made available for inspection, upon request, when it has been determined that no investigation should be conducted or when the investigation has been completed.
- (i) If the complainant has requested that his identity not be disclosed, or if there is reason to believe that disclosure of complainant's identity could embarrass him or subject him to harassment, the complaint will not be made available for inspection without his consent. If no investigation is conducted or if the investigation has been completed, however, such complaints will be made available for inspection upon request if, and to the extent that, it is possible to do so without disclosing the complainant's identity.
- (ii) Complaints or portions of complaints containing scurrilous or defamatory statements will not be made available for inspection, except upon a compelling showing that the public interest so requires.
- (2) Work papers and memoranda. Personal work papers, memoranda, or reports prepared by Commission personnel relating to the institution, conduct, or outcome of an investigation are placed in

- an investigatory file and are retained in that file. They are not available for public inspection. Requests for inspection of such materials will be considered under § 0.461. Materials of this nature received from other Government agencies are placed and retained in an investigatory file; requests for inspection of such materials will be referred to the agency from which they were received.
- (3) Statements and documents. Statements and documents obtained by Commission investigators in the course of an investigation are handled in the same manner as records referred to in subparagraph (2) of this paragraph.
- [32 F.R. 10573, July 19, 1967, as amended at 32 F.R. 12180, Aug. 24, 1967; 32 F.R. 11213, Aug. 2, 1967]
- § 0.459 Requests that materials or information submitted to the Commission be withheld from public inspection.
- (a) Any person submitting information or materials to the Commission may submit therewith a request that such information not be made available for public inspection. A copy of the request shall be attached to and shall cover all of the materials to which it applies and all copies of those materials. If feasible, the materials to which the request applies shall be physically separated from the materials to which the request does not apply; if this is not feasible, the portion of the materials to which the request applies shall be identified.
- (b) Each such request shall contain a statement of the reasons for withholding the materials from public inspection (see § 0.457) and of the facts upon which those reasons are based. If the request is that the materials be withheld from public inspection for a limited period of time, that period shall be specified.
- (c) If the materials are submitted voluntarily (i.e., absent any request or direction by the Commission), the person submitting them may request the Commission to return the materials without consideration rather than make them available for public inspection. If the request for confidentiality should be denied, the materials will ordinarily be returned (e.g., an application will be returned if it cannot be treated on a confidential basis); only in the unusual instance where the public interest so requires will the material be made public,

and then only after an appropriate period for comment by the person affected and judicial review. If the submission of the materials is requested or required by the Commission, and the request for confidentiality is denied, the materials will be made available for public inspection. (As to maintenance of confidentiality of complainants, see § 0.457(g).)

(d) If no request for confidentiality is filed, the Commission assumes no obligation to consider the need for non-disclosure but, in the unusual instance, may determine, on its own motion, that the materials should be withheld from pub-

lic inspection under § 0.457.

(e) Upon a determination by the appropriate Bureau or Office Chief that a request for confidentiality is well-founded, the materials will not routinely be made available for public inspection. Such person will, however, prepare a brief memorandum generally identifying the materials and specifically stating the reasons for nondisclosure. This memorandum will be placed in the public file in lieu of the materials withheld from inspection.

(f) If the Bureau or Office Chief is unable to make a determination on a request for confidentiality, the request will be referred to the Executive Director for

action.

- (g) If a request for confidentiality is denied, the materials will not be returned or made available for public inspection until the person who filed the request has been notified of the action and has been afforded an opportunity to file an application for review by the Commission and to petition for judicial review of the Commission's action.
- (h) Any person desiring to inspect materials withheld from public inspection under the provisions of this section may submit a request for inspection under § 0.461.
- § 0.461 Requests for inspection of materials not routinely available for public inspection.
- (a) Any person desiring to inspect documents not open to routine inspection under § 0.456, § 0.457, or § 0.459 may file a request for inspection. An original and one copy shall be submitted. Each such request shall identify, with particularity, the materials to be inspected and shall set forth the reasons for permitting inspection and the facts in support thereof.

- (b) In the case of materials not open to routine inspection under § 0.457(d) or § 0.459, or if, in the judgment of the Executive Director, the person who submitted the materials should be afforded an opportunity to file a response, the duplicate copy of the request for inspection will be mailed to the person who submitted the materials, and he will be afforded a reasonable period (normally 15 days) in which to comment on the request and to indicate any objections thereto. If a response is submitted, the respondent shall serve a copy on the person who filed the request for inspection. Except as specifically authorized or directed by the Executive Director, additional pleadings may not be filed.
- (c) Requests for inspection will be acted upon by the Executive Director as

follows:

- (1) If the Commission is prohibited from disclosing the materials in question, the request for inspection will be denied with a statement setting forth the specific grounds for denial.
- (2) If the materials are the property of another agency, the request will be referred to that agency and the person who submitted that request will be so advised, with the reasons therefor.
- (3) If it is determined that the Commission does not have authority to withhold the materials from public inspection, the request will be granted.
- (4) If it is determined that the Commission does have authority to withhold the materials from public inspection, the considerations favoring disclosure and nondisclosure will be weighed in the light of the facts presented, and the request will be granted, either conditionally or unconditionally, or denied.
- (d) (1) If the request is granted conditionally or denied and disclosure has not been opposed under paragraph (b) of this section, the person who filed the request may file an application for review by the Commission under § 1.115 of this chapter. Responsive pleadings provided for in § 1.115 of this chapter may not be filed.
- (2) If disclosure has been opposed under paragraph (b) of this section, the person who submitted the materials or the person who filed the request for inspection may file an application for review by the Commission within 15 days



after the order acting on the request for inspection is released. Responsive pleadings may be filed in accordance with § 1.115(d) of this chapter. No order granting the request for inspection shall be implemented until the opportunities for review by the Commission and judicial review have been afforded, unless the Commission finds, for reasons set forth in the order, that the public interest requires earlier inspection. In the latter event, the order will not be implemented until the opportunity to obtain a judicial stay of the Commission's Order has been afforded.

(e) (1) If the request is related to a hearing proceeding and is filed by a party to that proceeding, it shall be served on all other parties to the proceeding.

(2) The presiding officer will act on the request under the procedures and criteria set out in paragraphs (b) and (c) of this section and may also call for and consider pleadings filed by parties to the proceeding.

- (3) Any party to the hearing and any person who has participated in proceedings on the request may appeal the presiding officer's ruling to the Commission. Appeals shall be filed within 5 days after the order is released. Oppositions to the appeal shall be filed within 5 days after the time for filing appeals has expired. Additional pleadings may be filed only if specifically authorized or requested by the Commission.
- (4) Orders granting requests for inspection are stayed pending appeal and judicial review, in accordance with subparagraph (2) of paragraph (d) of this section.

[32 F.R. 10573, July 19, 1967, as amended at 33 F.R. 463, Jan. 12, 1968]

§ 0.463 Demand by competent authority for the production of documents or testimony concerning information contained therein.

(a) In the event that a demand (subpoena, order or other demand) is made by a court or other competent authority outside the Commission, upon any officer or employee of the Commission for the production of records or files or for testimony concerning information contained therein, he shall promptly advise the Executive Director of such demand, the nature of the papers or information sought, and all other relevant facts and

circumstances. The Commission will thereupon issue such instructions as it may deem advisable.

(b) Unless specifically authorized to produce such records or files or to testify with respect thereto, any officer or employee of the Commission who is served with a demand for the production of records or files or his testimony concerning the same, shall appear in response to the demand and respectfully decline to produce such records or files or to testify concerning them, basing his refusal upon this rule.

§ 0.465 Request for copies of materials which are available, or made available, for public inspection.

- (a) The Commission annually awards a contract to one or more commercial firms to make copies of Commission records and offer them for sale to the public. The contract is awarded on the basis of the lowest cost to the public. Currently, the contractor is Cooper-Trent, Inc., 1130 19th Street NW., Washington, D.C., 20006. Except as provided in paragraphs (b) and (c) of this section and in § 0.467, requests for copies of the records listed in §§ 0.453 and 0.455, and those made available for inspection under § 0.461, should be directed to the contractor. The contractor maintains master files of the following materials specifically for reproduction and sale to the public:
- (1) Horizontal directional antenna patterns filed with the Commission since February 1948.
- (2) FCC Form 402 microwave applications filed with the Commission since July 1961.
- (3) FCC Form 401 applications for common carrier microwave relay stations serving community antenna television (CATV) systems.
- (4) Grade A and Grade B contour maps on television broadcast stations filed with the Commission since July 1966.
- (b) The Commission annually awards a contract to a commercial firm to transscribe Commission proceedings in which a verbatim record is kept and to offer copies of the transcript for sale to the public. The contract is awarded on the basis of the lowest cost to the public and to the Commission. Except as authorized by the Commission, the firm is required

to retain the capacity to furnish copies of the transcript for a period of 5 years. and may retain that capacity for a longer period, even though another firm is currently transcribing Commission proceedings. Requests for copies of the transcript of current proceedings should be directed to the current contractor. C.S.A. Reporting Corp., 300 Seventh Street S.W., Washington, D.C., 20004, Requests for transcripts of older proceedings will be forwarded by the Commission to the firm which made the transcript in question; and the names of contracting firms for past years will be furnished upon request. If a transcript cannot be obtained from the reporting firm, it can be obtained from the general duplicating contractor, as provided in paragraph (a) of this section.

- (c) (1) This section has no application to printed publications, which may be purchased from the Superintendent of Documents or private firms (see §§ 0.411-420). Nor does it apply to application forms or information bulletins, which are prepared for the use and information of the public and are available upon request (see §§ 0.421 and 0.423).
- (2) Contractual arrangements which have been entered into with commercial firms, as described in this section, do not in any way limit the right of the public to inspect Commission records or to extract therefrom whatever information may be desired. Any person may, in addition, make photocopies of Commission records with his own equipment at locations where those records may be inspected. Coin-operated photocopy machines are available for use by the public in the Broadcast and Dockets Reference Room and the Common Carrier Bureau for the duplication of records available for inspection at those locations.
- (3) The Commission has reserved the right to make copies of its records for its own use or for the use of other agencies of the U.S. Government. When it serves the regulatory or financial interests of the U.S. Government, the Commission will make and furnish copies of its records free of charge. In other circumstances, however, if it should be necessary for the Commis-

sion to make and furnish copies of its records for the use of others, the fee for this service shall be the same as that charged by the general duplicating contractor for copies of those records under contractual arrangements described in paragraph (a) of this section.

(4) Requests for copies by representatives of foreign governments or persons residing in foreign countries shall be submitted to the Commission and will be reviewed by the Commission under criteria established by the Department of Commerce for controlling the export of technical data.

\$ 0.466 Fees.

Note: Provisions relating to private charges for furnishing copies of documents are set forth in §§ 0.465(c)(3) and 0.467. Provisions relating to fees for the location and production of records for inspection are under consideration and will be dealt with at a subsequent time.

§ 0.467 Certified copies.

Copies of documents which are available, or made available, for inspection under §§ 0.451-0.465 will be prepared and certified by the Secretary, under seal, on written request specifying the exact documents, the number of copies desired, and the date on which they will be required. The request shall allow a reasonable time for the preparation and certification of copies. The fee for preparing copies shall be the same as that charged by the general duplicating contractor for the same work under contractual arrangements described in § 0.465(a). The fee for certification shall be \$1 for each document.

PLACES FOR MAKING SUBMITTALS OR RE-QUESTS, FOR FILING APPLICATIONS, AND FOR TAKING EXAMINATIONS

§ 0.471 Miscellaneous submittals or requests.

Persons desiring to make submittals or requests of a general nature should communicate with the Office of the Secretary of the Commission.

[32 F.R. 10578, July 19, 1967]

§ 0.473 Reports of violations.

Reports of violations of the Communications Act or of the Commission's rules



and regulations may be submitted to the Commission in Washington or to any field office.

[32 F.R. 10578, July 19, 1967]

§ 0.475 Applications for employment.

Persons who wish to apply for employment should communicate with the Chief, Personnel Division.

[32 F.R. 10578, July 19, 1967]

§ 0.481 Place of filing applications for radio authorizations.

Class of station	Method of filing	Number of copies
(a) Alaskan fixed public and Alaskan public coastal.	VIA Engineer In Charge Radio District No. 14 Seattle, Wash., 98104.	2.
(b) Amateur	See §§ 0.483 and 0.485.	As specified in form.
(c) Interim ship station license.	See § 0.487	Do.
(d) Citizens	To Federal Com- munications Commission, Gettysburg, Pa., 17325.	Do.
(e) Ship (FCC Forms 502 & 405-B).	Do	Do.
(f) Aircraft (FCC Forms 404 & 405-B).	D0	Do.
(g) All others	Directly to the main Washington, D.C., office of the Commission. See § 0.401.	Do.

[31 F.R. 6112, Apr. 21, 1966, as amended at 32 F.R. 10578, July 19, 1967. Redesignated at 32 F.R. 10578, July 19, 1967]

§ 0.483 Applications for amateur station and operator license and/or commercial operator license.

(a) Application for a new amateur operator license, or for a combination of new amateur operator and station license, which will require examination supervised by Commission personnel, shall be filed in the appropriate engineering field office listed in § 0.121. All other applications for amateur radio licenses shall be submitted to the Federal Communications Commission, Gettysburg, Pennsylvania, 17325. Only one copy of the application is required.

(b) Application for commercial operator license of a class for which exami-

nation is required, for a verification card (FCC Form 758-F) or for a verification of operator license (FCC Form 759) shall be filed with an engineering field office listed in § 0.121 (a) or (b) at which the applicant desires his application to be considered and acted upon, except that application for replacement or duplicate license of such class shall be filed with the office of issue.

(c) Application for commercial operator license for which examination is not required (Restricted Radiotelephone Operator Permit) shall be submitted to the Federal Communications Commission. Gettysburg, Pa., 17325, with the following exceptions:

(1) When the applicant is located in Alaska, Hawaii, Puerto Rico, or the Virgin Islands of the United States, the application may be submitted by mail or in person to the nearest engineering field office.

(2) When the applicant is at any other location and the application is accompanied by a written showing by the applicant of immediate need for a permit for safety purposes and presented in person by the applicant or his agent, the application may be submitted to the nearest engineering field office.

(3) When application is from an alien aircraft pilot (see § 13.4(c) of this chapter), it shall be submitted to the Federal Communications Commission, Washington, D.C. 20554.

[28 F.R. 12412, Nov. 22, 1963, as amended at 29 F.R. 12371, Aug. 28, 1964; 32 F.R. 2626, Feb. 8, 1967. Redesignated at 32 F.R. 10578, July 19, 1967]

§ 0.485 Amateur and commercial operator examination points.

(a) Examinations for amateur and commercial radio operator licenses are conducted at each of the Field Engineering Bureau district offices listed in § 0.121 on the days designated by the Engineer in Charge of the district office. Examination schedules may be obtained from the Engineer in Charge.

(b) Examinations for amateur radio operator licenses are given frequently, by appointment, at the Commission's offices in Gettysburg, Pa. Examinations for all classes of radio operator licenses are

given frequently, by appointment, at the Commission's offices in the following cities:

Mobile, Ala. Anchorage, Alaska. San Diego, Calif.

Tampa, Fla. Savannah, Ga. Beaumont, Tex.

(c) Examinations are also given at greater intervals in the cities listed below, which are visited for that purpose by Commission examiners from the field offices for such locations. For current schedules, exact time, place, and other details, inquiry should be addressed to the office conducting examinations at the chosen point.

QUARTERLY POINTS

Albany, N.Y. Birmingham, Ala. Phoenix, Ariz. Little Rock, Ark. Fresno, Calif. Fort Wayne, Ind. Indianapolis, Ind. Davenport, Iowa. Des Moines, Iowa. Louisville, Ky. Grand Rapids, Mich. St. Louis, Mo. Omaha, Nebr. Syracuse, N.Y. Winston-Salem, N.C. Cincinnati, Ohio

Cleveland, Ohio.
Columbus, Ohio.
Ooklahoma City, Okla.
Tulsa, Okla.
Pittsburgh, Pa.
Williamsport, Pa.
Sioux Falls, S. Dak.
Knoxville, Tenn.
Memphis, Tenn.
Nashville, Tenn.
Corpus Christi, Tex.
San Antonio, Tex.
Sait Lake City, Utah.
Charleston, W. Va.
Milwaukee, Wis.

SEMIANNUAL

Fairbanks, Alaska. Tucson, Ariz. Hartford, Conn. Jacksonville, Fla. Bolse, Idaho. Wichita, Kans. Portland, Maine. Jackson, Miss. Las Vegas, Nev. Albuquerque, N. Mex. Wilmington, N.C. El Paso, Tex. Lubbock, Tex. Salem, Va. Spokane, Wash.

ANNUAL

Bakersfield, Calif. Hilo, Hawaii. Lihue, Hawaii. Wailuku, Hawaii. Bangor, Maine. Marquette, Mich. Billings, Mont. Great Falls, Mont. Missoula, Mont. Jamestown, N. Dak. Klamath Falls, Oreg. Rapid City, S. Dak.

(d) Arrangements have also been made, with the cooperation of other Federal agencies, for Extra Class and General Class examinations in outlying areas, as follows:

Alaska: Stations of the Alaska Communications System.

Guam: District Communications Officer, United States naval station.

Hawaii: At not exceeding one point on any island, by the Engineer in Charge (Honolulu).

[28 F.R. 12413, Nov. 22, 1963, as amended at 29 F.R. 10397, July 25, 1964; 29 F.R. 15082, Nov. 7, 1964; 29 F.R. 15289, Nov. 14, 1964; 30 F.R. 4612, Apr. 9, 1965; 30 F.R. 7755, June 16, 1965; 30 F.R. 8837, July 14, 1965; 30 F.R. 14932, Dec. 2, 1965; 31 F.R. 16567, Dec. 28, 1966; 34 F.R. 752, Jan. 17, 1969; 34 F.R. 18459, Nov. 20, 1969. Redesignated at 32 F.R. 10578, July 19, 1967]

§ 0.487 Applications for interim ship station licenses.

Formal applications for ship station licenses for use of radiotelephone or radar transmitting apparatus or both and applications for modification of such licenses shall, when accompanied by requests for interim ship station licenses, be filed in accordance with § 83.35 of this chapter and presented in person by applicants or their agents at the nearest field office of the Commission as shown in § 0.121 (a) and (b) or at the Commission's main office in Washington, D.C.: Provided, That, as an alternative procedure, an applicant, in Alaska, for such a ship station license may submit an application by mail to the Commission's Field Engineering Office at Anchorage, Alaska, when accompanied by a written request for an interim ship station license. Applications for renewal of ship station licenses are not subject to the provisions of this section.

[28 F.R. 12413, Nov. 22, 1963. Redesignated at 32 F.R. 10578, July 19, 1967]

§ 0.489 Applications for ship radio inspection and periodical survey.

Applications for ship radio inspection or for periodical survey shall be forwarded to the radio district office nearest the desired port of inspection or place of survey.

[28 F.R. 12413, Nov. 22, 1963. Redesignated at 32 F.R. 10578, July 19, 1967]

§ 0.491 Applications for exemption from compulsory ship radio requirements.

Applications for exemption filed under the provisions of section 352(b) or 383 of the Communications Act; Regulation 4, Chapter I of the Safety Convention; Regulation 5 or 6, Chapter IV of the Safety Convention; Regulation 12(b), Chapter V of the Safety Convention; or

Sec. Article 6 of the Great Lakes Agreement. 1.51 Number of copies of pleadings, briefs, shall be filed at the Commission's offices and other documents. in Washington, D.C. 1 52 Subscription and verification. [28 F.R. 12413, Nov. 22, 1963. Redesignated at GENERAL APPLICATION PROCEDURES 32 F.R. 10578, July 19, 1967] 1.61 Procedures for handling applications Non-radio common carrier ap-§ 0.493 special requiring aeronautical plications. study. 1.62 Operation pending action on renewal All such applications shall be filed at application. the Commission's offices in Washington, 1.65 Substantial and significant changes in information furnished by appli-[28 F.R. 12413, Nov. 22, 1963. Redesignated at cants to the Commission. 32 F.R. 10578, July 19, 1967] 1.68 Action on application for license to cover construction permit. PART 1—PRACTICE AND PROCEDURE 1.70 Procedures for handling applications involving the use of certain lands Subpart A-General Rules of Practice and and reservations under the jurisdiction of the U.S. Government. Procedure 1.77 Detailed application procedures; GENERAL. cross references. Sec. 1.1 MISCELLANEOUS PROCEEDINGS Proceedings before the Commission. 1.2 Declaratory rulings. 1.80 Forfeiture proceedings (excluding 1.3 Suspension, amendment, or waiver of those pertaining to broadcast lirules. censees and permittees or ships and 1.4 Computation of time. ship masters). 1.5 Mailing address furnished by licen-1.81 Other forfeiture procedures; cross references. 16 Availability of station logs and rec-1.83 Application for radio operator liords for Commission inspection. cense 1.8 Withdrawal of papers. 1.84 Procedure with respect to commercial 1.10 Transcript of testimony; copies of radio operator license applications. documents submitted. 1.85 Suspension of operator licenses. 1.12 Notice to attorneys of Commission 1.87 Modification of license or construcdocuments. tion permit on motion of the Com-1.14 Citation of Commission documents. mission. 1.89 Notice of violations. PARTIES AND PRACTITIONERS. AND WITNESSES 1.91 Revocation and/or cease and desist 1.21 Parties. proceedings; hearings. 1.22 Authority for representation. 1.92 Revocation and/or cease and desist 1.23 Persons who may be admitted to proceedings; after waiver of hearpractice. ing. 1.24 Censure, suspension, or disbarment RECONSIDERATION AND REVIEW OF ACTIONS of attorneys. TAKEN PURSUANT TO DELEGATED AUTHORITY; 1.25 Former Commissioners and employ-RECONSIDERATION OF ACTIONS TAKEN BY THE ees. 1.26 COMMISSION Appearances. 1.27 Witnesses; right to counsel. 1.101 General provisions. 1.102 Effective dates of actions taken pur-PLEADINGS, BRIEFS, AND OTHER PAPERS suant to delegated authority. 1.41 Informal requests for Commission 1.104 Preserving the right of review; deaction. ferred consideration of applica-1.42 Applications, reports, complaints; tion for review. cross-reference. 1.106 Petition for reconsideration of final action taken by the Commission 1.43 Requests for stay; cross-reference. en banc or by a designated author-1.44 Separate pleadings for different reity pursuant to a delegation. quests. 1.108 Reconsideration on Commission's 1.45 Pleadings; filing periods. own motion. 1.46 Motions for extension of time. 1.110 Partial grants; rejection and desig-1.47 Service of documents and proof of nation for hearing. service. 1.111 Petitions for reconsideration of an 1,48 Length of pleadings. order designating a case for hear-1.49 Specifications as to pleadings and ing. documents. 1.113 Action modified or set aside by per-

son, panel, or board.

1.50

Specifications as to briefs.

Part 1	art 1 Title 47—Chapter I			
Sec.		Sec.		
1.115	Application for review of action taken	1.294	Oppositions and replies.	
1.117	pursuant to delegated authority. Review on motion of the Commis-	1.296 1.297	Service. Oral argument.	
	sion.	1.298	Rulings; time for action.	
1.120	Protests of grants without hearing.	APPEA	L AND RECONSIDERATION OF PRESIDING	
	Subpart B—Hearing Proceedings		OFFICER'S RULING	
	GENERAL	1.301	Appeal from the presiding officer's	
1.201 1.202	Scope.	1 000	adverse ruling; effective date.	
1.202	Official reporter; transcript. The record.	1.303	Reconsideration of presiding officer's adverse ruling.	
1.204	Pleadings; definition.	T	<u> </u>	
1.205	Continuances and extensions.		ISCOVERY AND PRESERVATION OF EVIDENCE	
1.207	Interlocutory matters, reconsidera-	1.311 1.313	General. Protective orders.	
1.209	tion and review; cross references. Identification of responsible officer in	1.315	Depositions upon oral examination—	
	caption to pleading.		notice and preliminary procedure.	
1.211	Service.	1.316	Depositions upon written interroga-	
	PARTICIPANTS AND ISSUES		tories—notice and preliminary pro-	
1.221	Notice of hearing; appearances.	1.318	cedure. The taking of depositions.	
1.223	Petitions to intervene.	1.319	Objections to the taking of deposi-	
1.225	Participation by non-parties; con- sideration of communications.		tions.	
1.227	Consolidations.	1.321	Use of depositions at the hearing.	
1.229	Motions to enlarge, change, or delete	1.323 1.325	Interrogatories to parties. Discovery and production of docu-	
	issues.	1.020	ments and things for inspection,	
	PRESIDING OFFICER		copying, or photographing.	
1.241	Designation of presiding officer.	1.831	Who may sign and issue.	
1.243	Authority of presiding officer.	1.333 1.334	Requests for issuance of subpens. Motions to quash.	
1.245	Disqualification of presiding officer.	1.335	Rulings.	
	PREHEARING PROCEDURES	1.886	Service of subpenas.	
1.246	Admission of facts and genuineness	1.337	Return of service.	
1.248	of documents. Prehearing conferences; hearing con-	1.338 1.339	Subpena forms. Witness fees.	
,	ferences.	1.340	Attendance of witness; disobedience.	
1.249	Prehearing statement.		EVIDENCE	
HE	ARING AND INTERMEDIATE DECISION	1.351	Rules of evidence.	
1.250	Discovery and preservation of evi-	1.352	Cumulative evidence.	
7.000	dence; cross-reference.	1.353	Further evidence during hearing.	
1.253 1.254	Time and place of hearing. Nature of the hearing; burden of	1.354	Documents containing matter not material.	
-a.mvz	proof.	1.355	Documents in foreign language.	
1.255	Order of procedure.	1.356	Copies of exhibits.	
1.258	Closing of the hearing.	1.357	Mechanical reproductions as evi- dence.	
1.260 1.261	Certification of transcript. Corrections to transcript.	1.358	Tariffs as evidence.	
1.263	Proposed findings and conclusions.	1.359	Proof of official record; authentica-	
1.264	Contents of findings of fact and con-		tion of copy.	
1 007	clusions.	1.360 1.361	Proof of lack of record.	
1.267	Initial and recommended decisions.	1.362	Other proof of official record. Production of statements.	
	REVIEW PROCEEDINGS			
1.271	Delegation of review function.		Subpart C—Rule Making Proceedings	
1,273	Waiver of initial or recommended decision.	;	Petitions and Related Pleadings	
1.274	Certification of the record to the	1.401	Petitions for rule making.	
	Commission for initial or final de-	1.403	Notice and availability.	
1.276	cision.	1.405 1.407	Responses to petitions; replies. Action on petitions.	
1.276	Appeal and review of initial decision.		Date Marie December	

1.282 Final decision of the Commission. Interlocutory Actions in Hearing Proceedings

Appeal and review of initial decision. Exceptions; oral arguments.

Limitation of matters to be reviewed.

1.291 General provisions.

1.292 Number of copies.

1.276 1.277

1.279

1.411 Commencement of rule making proceedings. 1.412 Notice of proposed rule making. 1.413 Content of notice. 1.415 Comments and replies.

RULE MAKING PROCEEDINGS

1.417 Statutory requirement for hearing.

Sec.	Horm of comments and replica: num-	Sec. 1.547	Application for permission to use
1.419	Form of comments and replies; num- ber of copies.	1.021	Application for permission to use lesser grade operators.
1.421 1.423	Further notice of rule making. Oral argument and other proceed-	1.548	Application to operate by remote control.
2.220	ings.	1.549	Requests for extensions of authority
1.425 1.427	Commission action. Effective date of rules.		to operate without certain indicat- ing instruments.
	ubpart D—Broadcast Applications and	1.550	Requests for new or modified call
31	Proceedings		sign assignments.
	GENERAL	Aı	PPLICATION PROCESSING PROCEDURES
1.501	Scope. Emergency Broadcast System Au-	1.561	Staff consideration of applications
1.502	thorizations.		which receive action by the Commission.
	GENERAL FILING REQUIREMENTS	1.562	Staff consideration of applications
1.511	Applications required.		which do not require action by the Commission.
1.512	Where to file; number of copies.	1.564	Acceptance of applications.
1.518 1.514	Who may sign applications. Content of applications.	1.566	Defective applications.
1.516	Specification of facilities.	1.568 1.569	Dismissal of applications. Applications for frequencies adjacent
1.518	Inconsistent or conflicting applica-	000	to Class I–A channels.
1.519	tions. Repetitious applications.	1.570	Standard broadcast applications in-
1.520	Multiple applications.		volving other North American countries.
1.522 1.525	Amendment of applications. Agreements between parties for	1.571	Processing of standard broadcast ap-
1.020	amendment or dismissal of, or	1.572	plications.
	failure to prosecute broadcast	1.012	Processing of television broadcast applications.
1.526	applications. Records to be maintained locally for	1.573	Processing of FM and noncommer-
	public inspection by applicants,		cial educational FM broadcast applications.
	permittees, and licensees.	1.574	Processing of international broad-
APPLIC	ATION FORMS AND PARTICULAR FILING	1	cast applications.
4 -01	REQUIREMENTS	1.578	Amendments to applications for re- newal, assignment and/or trans-
1.531 1.533	Formal and informal applications. Application forms for authority to		fer of control.
2.000	construct a new station or make	1.580	Local notice of the filing of broadcast
1 504	changes in an existing station.		applications, and timely filing of petitions to deny them.
1.534	Application for extension of con- struction permit or for construc-	1.587	Procedure for filing informal objec-
	tion permit to replace expired con-		tions.
1.536	struction permit. Application for license to cover con-		Action on Applications
1.000	struction permit.	1.591	Grants without hearing of authori-
1.537	Application for license to use former		zations other than licenses pur- suant to construction permits.
	main transmitter or antenna as an auxiliary.	1.592	Conditional grant.
1.538	Application for modification of li-	1.593	Designation for hearing.
4 500	cense.	1.594	Local notice of designation for hearing.
1.539 1.540	Application for renewal of license. Application for voluntary assign-	1.597	Procedures on transfer and assign-
	ment or transfer of control.		ment applications.
1.541	Application for involuntary assign-	1.598	Period of construction.
	ment of license or transfer of con- trol.	1.599 1.601	Forfeiture of construction permit. License, simultaneous modification
1.542	Application for temporary authoriza-		and renewal.
1.543	tion. Application for renewal or modifica-	1.603	Special waiver procedure relative to
1.040	tion of special service authoriza-	1.605	applications. Retention of applications in hearing
	tion.	1.000	status after designation for hear-
1.544	Application for standard broadcast station experimental operation.		ing
1.545	Application concerning programs to	FORMS	AND INFORMATION TO BE FILED WITH
	be transmitted to foreign radio stations.		THE COMMISSION
1.546	Application to determine operating	1.611	Financial report.
	power by direct measurement of	1.613	Filing of contracts. Ownership reports.
	antenna power.	1.615	Ownersmb repores.

Part 1 Title 47—Chapter I			
FORFE	TURES RELATING TO BROADCAST LICENS- EES AND PERMITTEES	SPECII	FIC TYPES OF APPLICATIONS UNDER TITLE II OF COMMUNICATIONS ACT
Sec.		Sec.	0 00
1.621	Forfeitures relating to broadcast li-	1.761	Cross-reference.
	censees and permittees.	1.762	Interlocking directorates.
Subpar	E—Complaints, Applications, Tariffs, and	1.763	Construction, extension, acquisition,
Josephan	Reports Involving Common Carriers	1.764	or operation of lines.
	_ -	1.704	Discontinuance, reduction, or impairment of service.
	GENERAL	1.765	Consolidation or acquisition of tele-
1.701	Show cause orders.		phone companies.
1.703	Appearances.	1.766	Consolidation of domestic telegraph
	COMPLAINTS		carriers.
1.711	Formal or informal complaints.	1.767	Cable landing licenses.
1.713	Satisfaction of complaints; damages.		Tariffs
	INFORMAL COMPLAINTS	1.771	Filing.
		1.772	Application for special tariff permis-
1.716	Form. Procedure.	1.773	sion. Petitions for suspension of tariff
1.717 1.718	Unsatisfied informal complaints; for-	1.110	schedules.
1.110	mal complaints relating back to	_	
	the filing dates of informal com-	CONT	RACTS, REPORTS, AND REQUESTS REQUIRED
	plaints.		To Be Filed by Carriers
	FORMAL COMPLAINTS	1.781	Requests for extension of filing time.
1 701			CONTRACTS
1.721 1.722	Form. Statement of issues and facts.	1.783	Filing.
1.723	Damages; allegations with certainty.	FINA	ANCIAL AND ACCOUNTING REPORTS AND
1.724	Specific tariff schedule references.		REQUESTS
1.725	Joinder of complainants and causes	1.785	Annual financial reports.
	of action.	1.786	Monthly financial reports.
1.726	Discrimination, preference, or prejudice.	1.787	Reports of proposed changes in de- preciation rates.
1.727	Supplemental complaints.	1.788	Reports regarding pensions and
1.728	Cross complaints.	1.789	benefits.
1.729	Copies; service.	1.769	Reports regarding division of inter- national telegraph communica-
1.730	Answers to complaints, supplemental complaints, amended complaints,	4	tion charges.
	and cross complaints.	1.790	Reports relating to traffic by inter- national carriers.
1.731	Motions to dismiss complaints or to	1.791	Reports and requests to be filed
	make them more definite and cer- tain.		under Part 31 of this chapter.
1.732	Replies to answers or amended	1.792	Reports and requests to be filed
1.102	answers: motions to make answers	4 500	under Part 33 of this chapter.
	more definite and certain.	1.793	Reports and requests to be filed
1.733	Oppositions to motions to dismiss	1.794	under Part 34 of this chapter. Reports and requests to be filed
	complaints or to make them more	1.102	under Part 85 of this chapter.
4 4	definite and certain.		
1.784	Specifications as to pleadings, briefs, and other documents; subscription		SERVICES AND FACILITIES REPORTS
	and verification.	1.801	Reports regarding telegraph carrier
1.735	Formal complaints not stating a	1.802	services. Reports relating to continuing au-
	cause of action; defective plead-	1.002	thority to supplement facilities or
	ings.		to provide temporary or emergency
	APPLICATIONS		service.
1.741	Scope.	1.803	Reports relating to reduction in
1.742	Place of filing, fees, and number of copies.	1.804	temporary experimental service. Reports regarding domestic telegraph
1.748	Who may sign applications.	1.805	speed of service. Reports relating to service by car-
1.744	Amendments.		riers engaged in public radio serv-
1.745	Additional statements.		ice operations.
1.746	Defective applications.		MISCELLANEOUS REPORTS
1.747	Inconsistent or conflicting applications.	1.811	
1.748	Dismissal of applications.	1.611	Reports regarding amendments to charters, by-laws and partnership
1.749	Action on applications under dele-		agreements of carriers engaged in
	gated authority.		domestic public radio services.
	7	^	
	7	.,	

Sec.			Action on Applications
1.812	Reports regarding premature destruction of records.	Sec. 1.971	Grants without a hearing.
1.813	Reports of negotiations regarding	1.973	Designation for hearing.
1.814	foreign communication matters. Reports regarding free service ren-	REPORT	s To Be Filed With the Commission
	dered the Government for na- tional defense.	1.981	Reports, annual and semiannual.
		FORFET:	tures Against Ships and Ship Masters
Subpart	F—Safety and Special Radio Services Applications and Proceedings	1.991	Forfeitures against ships and ship masters.
1.901	GENERAL Scope.	Subpart	G—Schedule of Fees for Applications
1.001	GENERAL FILING REQUIREMENTS	•	Filed With the Commission
1.911	Applications required.		GENERAL INFORMATION
1.912	Where applications are to be filed.	1.1101	Authority.
1.913	Who may sign applications.	1.1103 1.1104	•
1.91 4 1.915	Full disclosures. Shared use of broadcast antenna	1.1105	
1.010	structure.		SCHEDULE OF FRES
1.916	Repetitious applications.	1 1111	
1.918	Amendment of applications.	1.1111	Schedule of fees for Radio Broad- cast Services.
APPLICA	ATION FORMS AND PARTICULAR FILING REQUIREMENTS	1.1113	Schedule of fees for Common Carrier Services.
1.921	Procedure for obtaining a radio sta- tion authorization and for com-	1.1115	Schedule of fees for Safety and Special Radio Services.
1.922	mencement of operation. Forms to be used.	1.1117	
1.923	Waiver of construction permit re-		operator examinations and licensing.
	quirement.	1.1119	Experimental Radio Services (other
1.924	Assignment or transfer of control, voluntary and involuntary.		than Broadcast).
1.925	Application for special temporary authorization.		Subpart H—Ex Parte Presentations
1.926	Application for renewal of license.		GENERAL
1.927	Application for ship radio inspection or periodical survey of ships sub-	1.1201	Definitions.
	ject to compulsory radio require-	1.1203 1.1205	Restricted adjudicative proceedings.
	ments.	1.1200	Decision-making Commission per- sonnel (restricted adjudicative
1.928	Procedure with respect to applica-		proceedings).
	tions for ship radio inspection or periodical survey.	1.1207	Restricted rule making proceedings.
1.92 9	Application for exemption from compulsory ship radio requirements.	1.1209	Decision-making Commission personnel (restricted rule making proceedings).
1.930	Application for temporary waiver of annual inspection.		PROHIBITED PRESENTATIONS
1.931	Application for extension of con- struction permit.	1.1221	Presentations prohibited in re- stricted proceedings which have
1.932	Time in which station must be	1.1223	been designated for hearing. Presentations prohibited in re-
1.933	placed in operation. Installation or removal of apparatus.	1.1220	stricted application proceedings
1.934	Procedure with respect to amateur		prior to their designation for hear-
	radio operator license.	1 1005	ing.
API	PLICATION PROCESSING PROCEDURES	1.1225 1.1227	Solicitation of ex parte presentations. Permissible ex parte communica-
1.951	How applications are distributed.	1.1221	tions.
1.952	How file numbers are assigned.	ΤDo	OCEDURES FOR HANDLING EX PARTE
1.953	How applications are processed.	FR	COMMUNICATIONS
1.955	Frequency coordination, Canada. Rented communications equipment.	1.1241	Written ex parte communications.
1.9 56 1.958	Defective applications.	1.1243	Oral ex parte communications.
1.958	Resubmitted applications.	1.1245	Disclosure of information concerning
1.961	Dismissal of applications.		ex parte presentations.
1.962	Public notice of acceptance for		SANCTIONS
	filing; petitions to deny applica-	1.1251	Sanctions.
	tions of specified categories.	1.1201	Dationons.

APPENDIX A—A PLAN OF COOPERATIVE PRO-CEDURE IN MATTERS AND CASES UNDER THE PROVISIONS OF SECTION 410 OF THE COM-MUNICATIONS ACT OF 1934

Appendix B—Interpretations of Fee Rules and Procedures.

AUTHORITY: The provisions of this Part 1 issued under secs. 4, 803, 48 Stat. 1066, 1062, as amended; 47 U.S.C. 154, 303, unless otherwise noted. Implement, 5 U.S.C. 552.

Subpart A—General Rules of Practice and Procedure

SOURCE: The provisions of this Subpart A appear at 28 F.R. 12415, Nov. 22, 1963, unless otherwise noted.

GENERAL

§ 1.1 Proceedings before the Commission.

The Commission may on its own motion or petition of any interested party hold such proceedings as it may deem necessary from time to time in connection with the investigation of any matter which it has power to investigate under the law, or for the purpose of obtaining information necessary or helpful in the determination of its policies, the carrying out of its duties or the formulation or amendment of its rules and regulations. For such purposes it may subpena witnesses and require the production of evidence. Procedures to be followed by the Commission shall, unless specifically prescribed in this part, be such as in the opinion of the Commission will best serve the purposes of such proceeding.

(Sec. 403, 48 Stat. 1094; 47 U.S.C. 403)

§ 1.2 Declaratory rulings.

The Commission may, in accordance with section 5(d) of the Administrative Procedure Act, on motion or on its own motion issue a declaratory ruling terminating a controversy or removing uncertainty.

(5 U.S.C. 554)

§ 1.3 Suspension, amendment, or waiver of rules.

The provisions of this chapter may be suspended, revoked, amended, or waived for good cause shown, in whole or in part, at any time by the Commission, subject to the provisions of the Administrative Procedure Act and the provisions of this chapter. Any provision of the rules may be waived by the Commission on its own

motion or on petition if good cause therefor is shown.

CROSS REFERENCE: See Subpart C of this part for practice and procedure involving rule making.

§ 1.4 Computation of time.

(a) It is frequently necessary under Commission procedures to compute the terminal date of a period of time where the period begins with the occurrence of an act, event, or default and terminates a specified number of days thereafter. Unless otherwise provided by statute, the first day to be counted in computing the terminal date is the day after the day on which the act, event, or default occurs. The last day of such period of time is included in the computation and any action required must be taken on or before that day.

(b) When a Commission Decision, Order, or Memorandum is involved, the first day to be counted is the day after the day on which the complete text of the document is released by the Commission. When any other Commission action is involved, the first day to be counted is the day after the day on which the "Public Notice" of the action is re-

leased by the Commission.

(c) All petitions, pleadings, tariffs, or other documents filed with the Commission must be tendered for filing in complete form before 5:00 p.m. Any such document lodged with the Commission in complete form after 5:00 p.m. shall be deemed to be tendered for filing as of the next succeeding business day.

- (d) For purposes of this section, the term "holiday" shall include Saturdays, Sundays, legal holidays or half holidays in the District of Columbia, and any other day on which the Commission's offices are closed prior to 5:00 p.m. The term "business day" shall include all other days.
- (e) For purposes of this section, the term "filing period" means the number of days allowed or prescribed by statute, rule, order, notice, or other Commission action for filing any document with the Commission; the term does not include any additional days allowed for filing any document under paragraph (f), (g), or (i) of this section. The term "filing date" means the date upon which the document must be filed after all computations authorized by this section have been made.
- (f) If the filing period is less than seven days, intermediate holidays shall



be excluded in determining the filing

(g) Where service of a document is required by statute or by the provisions of this chapter, where the document is in fact served by mail (see § 1.47(f)), and where the filing period for a response thereto is 10 days or less, an additional 3 days, excluding holidays, will be allowed for filing the response. This paragraph shall not apply to documents which are filed pursuant to the provisions of § 1.89, § 1.120(d), § 1.315(b), or § 1.316.

(h) If both paragraphs (f) and (g) of

(h) If both paragraphs (f) and (g) of this section are applicable, make the computations called for by paragraph (f) before making those called for by

paragraph (g).

(i) If the filing date falls on a holiday, the document shall be filed on the next business day.

[28 F.R. 12415, Nov. 22, 1963, as amended at 33 F.R. 10571, July 25, 1968]

§ 1.5 Mailing address furnished by licensee.

(a) Each licensee shall furnish the Commission with an address to be used by the Commission in serving documents or directing correspondence to that licensee. Unless any licensee advises the Commission to the contrary, the address contained in the licensee's most recent application will be used by the Commission for this purpose.

(b) The licensee is responsible for making any arrangements which may be necessary in his particular circumstances to assure that Commission documents or correspondence delivered to this address will promptly reach him or some person authorized by him to act

in his behalf.

§ 1.6 Availability of station logs and records for Commission inspection.

(a) Station records and logs shall be made available for inspection or duplication at the request of the Commission or its representative. Such logs or records may be removed from the licensee's possession by a Commission representative or, upon request, shall be mailed by the licensee to the Commission by either registered mail, return receipt requested, or certified mail, return receipt requested. The return receipt shall be retained by the licensee as part of the station records until such records or logs are returned to the licensee. A receipt shall be furnished when the logs or records are removed from the licensee's possession by a Commission representative and this receipt shall be retained by the licensee as part of the station records until such records or logs are returned to the licensee. When the Commission has no further need for such records or logs, they shall be returned to the licensee. The provisions of this rule shall apply solely to those station logs and records which are required to be maintained by the provisions of this chapter.

(b) Where records or logs are maintained as the official records of a recognized law enforcement agency and the removal of the records from the possession of the law enforcement agency will hinder its law enforcement activities, such records will not be removed pursuant to this section if the chief of the law enforcement agency promptly certifies in writing to the Federal Communications Commission that removal of the logs or records will hinder law enforcement activities of the agency, stating insofar as feasible the basis for his decision and the date when it can reasonably be expected that such records will be released to the Federal Communications Commission.

§ 1.8 Withdrawal of papers.

The granting of a request to dismiss or withdraw an application or a pleading does not authorize the removal of such application or pleading from the Commission's records.

§ 1.10 Transcript of testimony; copies of documents submitted.

In any matter pending before the Commission, any person submitting data or evidence, whether acting under compulsion or voluntarily, shall have the right to retain a copy thereof, or to procure a copy of any document submitted by him, or of any transcript made of his testimony, upon payment of the charges therefor to the person furnishing the same, which person may be designated by the Commission. The Commission itself shall not be responsible for furnishing the copies.

[29 F.R. 14406, Oct. 20, 1964]

§ 1.12 Notice to attorneys of Commission documents.

In any matter pending before the Commission in which an attorney has appeared for, submitted a document on behalf of or been otherwise designated by a person, any notice or other written communication pertaining to that matter issued by the Commission and which is required or permitted to be furnished to the person will be communicated to the attorney, or to one of such attorneys if more than one is designated. If direct communication with the party is appropriate, a copy of such communication will be mailed to the attorney.

[29 F.R. 14406, Oct. 20, 1964]

§ 1.14 Citation of Commission documents.

The appropriate reference to the FCC Reports, Second Series, shall be included as part of the citation to any document which has been printed in the Reports (e.g., 1 FCC 2d 1 (1965)). See §§ 0.413 and 0.416 of this chapter.

[33 F.R. 11821, Aug. 21, 1968]

PARTIES, PRACTITIONERS, AND WITNESSES 1

§ 1.21 Parties.

- (a) Any party may appear before the Commission and be heard in person or by attorney.
- (b) The appropriate Bureau Chief(s) of the Commission shall be deemed to be a party to every adjudicatory proceeding (as defined in the Administrative Procedure Act) without the necessity of being so named in the order designating the proceeding for hearing.
- (c) When, in any proceeding, a pleading is filed on behalf of either the General Counsel or the Chief Engineer, he shall thereafter be deemed a party to the proceeding.

§ 1.22 Authority for representation.

Any person, in a representative capacity, transacting business with the Commission, may be required to show his authority to act in such capacity.

§ 1.23 Persons who may be admitted to practice.

(a) Any person who is a member in good standing of the bar of the Supreme Court of the United States or of the highest court of any state, territory, or of the District of Columbia, and is not under any order of any court suspending, enjoining, restraining, disbarring, or otherwise restricting him in the practice of law, may represent others before the Commission.

(b) When such member of the bar acting in a representative capacity appears in person or signs a paper in practice before the Commission, his personal appearance or signature shall constitute a representation to the Commission that, under the provisions of this chapter and the law, he is authorized and qualified to represent the particular party in whose behalf he acts. Further proof of authority to act in a representative capacity may be required.

§ 1.24° Censure, suspension, or disbarment of attorneys.

- (a) The Commission may censure, suspend, or disbar any person who has practiced, is practicing or holding himself out as entitled to practice before it if it finds that such person:
- (1) Does not possess the qualifications required by § 1.23:
- (2) Has failed to conform to standards of ethical conduct required of practitioners at the bar of any court of which he is a member;
- (3) Is lacking in character or professional integrity; and/or
- (4) Displays toward the Commission or any of its hearing officers conduct which, if displayed toward any court of the United States or any of its Territories or the District of Columbia, would be cause for censure, suspension, or disparment.
- (b) Before any member of the bar of the Commission shall be censured, suspended, or disbarred, charges shall be preferred by the Commission against such practitioner and he shall be afforded an opportunity to be heard thereon.

§ 1.25 Former Commissioners and employees.

- (a) No Commissioner shall, for a period of one year following the termination of his services as a Commissioner, represent any person before the Commission in a professional capacity, except that this restriction shall not apply to any former Commissioner who has served the full term for which he was appointed. See 47 U.S.C. 154(b).
- (b) No member, officer, or employee of the Commission (1) whose active service with the Commission has terminated but who is receiving pay while on annual leave not taken prior to separation from



¹ Centerhead amended, 29 F.R. 12775, Sept. 10. 1964.

such active service, or (2) who is in any other leave status, shall appear as attorney or participate in the preparation or handling of any matter before, or to be submitted to, the Commission.

- (c) No former member, officer, or employee of the Commission shall act as agent or attorney for any one other than the United States in connection with any particular Commission matter involving a specific party or parties in which the United States is a party or has a direct and substantial interest and in which he participated personally and substantially as a member, officer, or employee of the Commission, through decision, approval, disapproval, recommendation, the rendering of service, investigation, or otherwise. See 18 U.S.C. 207(a).
- (d) No former member, officer, or employee of the Commission shall, within one year after his employment has ceased, appear personally before the Commission as agent or attorney for any one other than the United States in connection with any particular Commission matter involving a specific party or parties in which the United States is a party or has a direct and substantial interest and which was under his official responsibility as an officer or employee of the Commission at any time within a period of one year prior to termination of his employment. See 18 U.S.C. 207(b).
- (e) Notwithstanding the provisions of paragraphs (c) and (d) of this section, a former member, officer, or employee of the Commission who possesses outstanding scientific or technological qualifications may act as attorney or agent or appear personally in connection with a particular matter in his scientific or technological field, upon certification by the Commission, published in the FED-ERAL REGISTER, that the national interest would be served by such action or appearance. See 18 U.S.C. This exception does not apply to persons barred from representing others before the Commission under paragraphs (a) or (b) of this section.

[28 F.R. 13785, Dec. 18, 1963]

§ 1.26 Appearances.

Rules relating to appearances are set forth in §§ 1.87, 1.91, 1.221, and 1.703.

§ 1.27 Witnesses; right to counsel.

Any individual compelled to appear in person in any Commission proceeding may be accompanied, represented, and advised by counsel as provided in this section. (Regulations as to persons seeking voluntarily to appear and give evidence are set forth in § 1.225.)

- (a) Counsel may advise his client in confidence, either upon his own initiative or that of the witness, before, during, and after the conclusion of the proceeding.
- (b) Counsel for the witness will be permitted to make objections on the record, and to state briefly the basis for such objections, in connection with any examination of his client.
- (c) At the conclusion of the examination of his client, counsel may ask clarifying questions if in the judgment of the presiding officer such questioning is necessary or desirable in order to avoid ambiguity or incompleteness in the responses previously given.
- (d) Except as provided by paragraph (c) of this section, counsel for the witness may not examine or cross-examine any witness, or offer documentary evidence, unless authorized by the Commission to do so.
- (5 U.S.C. 555) [29 F.R. 12775, Sept. 10, 1964]

PLEADINGS, BRIEFS, AND OTHER PAPERS

§ 1.41 Informal requests for Commission action.

Except where formal procedures are required under the provisions of this chapter, requests for action may be submitted informally. Requests should set forth clearly and concisely the facts relied upon, the relief sought, the statutory and/or regulatory provisions (if any) pursuant to which the request is filed and under which relief is sought, and the interest of the person submitting the request.

§ 1.42 Applications, reports, complaints; cross-reference.

- (a) Rules governing applications and reports are contained in Subparts D, E, and F of this part.
- (b) Special rules governing complaints against common carriers arising under the Communications Act are set forth in Subpart E of this part.

§ 1.43 Requests for stay; crossreference.

General rules relating to requests for stay of any order or decision are set forth in §§ 1.41, 1.44(e), 1.45 (d) and (e), and 1.298(a). See also §§ 1.102, 1.106(n), and 1.115(h).

§ 1.44 Separate pleadings for different requests.

- (a) Requests requiring action by the Commission shall not be combined in a pleading with requests for action by a hearing examiner or by any person or persons acting pursuant to delegated authority.
- (b) Requests requiring action by a hearing examiner shall not be combined in a pleading with requests for action by the Commission or by any person or persons acting pursuant to delegated authority.
- (c) Requests requiring action by any person or persons pursuant to delegated authority shall not be combined in a pleading with requests for action by any other person or persons acting pursuant to delegated authority.
- (d) Pleadings which combine requests in a manner prohibited by paragraph (a), (b), or (c) of this section may be returned without consideration to the person who filed the pleading.
- (e) Any request to stay the effectiveness of any decision or order of the Commission shall be filed as a separate pleading. Any such request which is not filed as a separate pleading will not be considered by the Commission.

Note: Matters which are acted on pursuant to delegated authority are set forth in Subpart B of Part 0 of this chapter. Matters acted on by the hearing examiner are set forth in § 0.341.

\$1.45 Pleadings; filing periods.

Except as otherwise provided in this chapter, pleadings in Commission proceedings shall be filed in accordance with the provisions of this section.

(a) Oppositions. Oppositions to any motion, petition, or request may be filed within 10 days after the original pleading is filed.

(b) Replies. The person who filed the original pleading may reply to oppositions within 5 days after the time for filing oppositions has expired. The reply shall be limited to matters raised in the oppositions, and the response to all such matters shall be set forth in a single pleading; separate replies to individual oppositions shall not be filed.

- (c) Additional pleadings. Additional pleadings may be filed only if specifically requested or authorized by the Commission.
- (d) Requests for temporary relief; shorter filing periods. Oppositions to a request for stay of any order or to a request for other temporary relief shall be filed within 7 days after the request is filed. Replies to oppositions should not be filed and will not be considered. The provisions of § 1.4(g) shall not apply in computing the filing date for oppositions to a request for stay or for other temporary relief.
- (e) Ex parte disposition of certain pleadings. As a matter of discretion, the Commission may rule ex parte upon requests for continuances and extensions of time, requests for permission to file pleadings in excess of the length prescribed in this chapter, and requests for temporay relief, without waiting for the filing of oppositions or replies.

Note: Where specific provisions conflict with the provisions of this section, the specific provisions are controlling. See, in particular, §§ 1.294(c) and 1.298(a).

[28 F.R. 12415, Nov. 22, 1963, as amended at 33 F.R. 7153, May 15, 1968]

§ 1.46 Motions for extension of time.

Extensions of time for filing any pleading, brief, or other paper may be granted upon motion for good cause shown, unless the time for filing is limited by statute.

§ 1.47 Service of documents and proof of service.

- (a) Where the Commission or any person is required by statute or by the provisions of this chapter to serve any document upon any person, service shall (in the absence of specific provisions in this chapter to the contrary) be made in accordance with the provisions of this section.
- (b) Where any person is required to serve any document filed with the Commission, service shall be made by that person or by his representative on or before the day on which the document is filed: Provided, however, That formal complaints, including supplemental, cross, and amended complaints, filed under section 208 of the Communications Act will be served by the Commission.
- (c) Commission counsel who formally participate in any proceeding shall be

served in the same manner as other persons who participate in that proceeding. The filing of a document with the Commission does not constitute service upon Commission counsel.

(d) Documents may be served upon a party, his attorney, or other duly constituted agent by delivering a copy or by mailing a copy to the last known address. When a party is represented by an attorney of record in a formal proceeding, service shall be made upon such

attorney.

(e) Delivery of a copy pursuant to this section means handing it to the party, his attorney, or other duly constituted agent; or leaving it with the clerk or other person in charge of the office of the person being served; or, if there is no one in charge of such office, leaving it in a conspicuous place therein; or, if such office is closed or the person to be served has no office, leaving it at his dwelling house or usual place of abode with some person of suitable age and discretion then residing therein.

(f) Service by mail is complete upon mailing. If the person upon whom service by mail is being made resides 500 miles or more from the person effecting service, such mailing must be by airmail.

(g) Proof of service, as provided in this section, shall be filed before action is taken. The proof of service shall show the time and manner of service, and may be by written acknowledgment of service, by certificate of the person effecting the service, or by other proof satisfactory to the Commission. Failure to make proof of service will not affect the validity of the service. The Commission may allow the proof to be amended or supplied at any time, unless to do so would result in material prejudice to a party.

§ 1.48 Length of pleadings.

(a) Affidavits, statements, and other materials which are submitted with and factually support a pleading are not counted in determining the length of the pleading. Other materials submitted with the pleading will be disregarded.

(b) Timely requests by a party for permission to file pleadings in excess of the length prescribed by the provisions of this chapter may be granted upon good cause shown. Where the filing period is 10 days or less, the request shall be made within 2 business days after the period begins to run. Where the filing

period is more than 10 days, the request shall be filed at least 10 days before the filing date. (See § 1.4.) If a timely request is made, the pleading need not be filed earlier than two business days after the Commission acts upon the request.

§ 1.49 Specifications as to pleadings and documents.

All pleadings and documents (except printed briefs) filed in any proceeding shall, unless otherwise specifically provided, be on paper either 8 by 10½ or 14 inches or 8½ by 11, 13 or 14 inches, with left-hand margin not less than 1½ inches wide. This requirement shall not apply to original documents, or admissible copies thereof, offered as exhibits or to specially prepared exhibits. The impression shall be on one side of the paper only and shall be double-spaced. except that long quotations shall be single spaced and indented. All papers, except charts and maps, shall be typewritten or prepared by mechanical processing methods, other than letterpress or printing. The foregoing shall not apply to official publications. All copies must be clearly legible.

§ 1.50 Specifications as to briefs.

Briefs may be printed, typewritten, mimeographed, multigraphed, or multilithed. Printed briefs shall be in 10- or 12-point type, on good unglazed paper, 5% inches wide by 9 inches long, with inside margin not less than 1½ inches wide, and with double spaced text and single spaced quotations. Typewritten, mimeographed, multigraphed, or multilithed briefs shall conform to the specifications for pleadings and documents set forth in § 1.49.

§ 1.51 Number of copies of pleadings, briefs, and other documents.

- (a) Except in rule making proceedings (§ 1.419), or with regard to interlocutory matters acted on by the Review Board, the Chief Hearing Examiner, or the Presiding Officer (§ 1.292), or as otherwise specifically provided in this chapter, an original and 19 copies of all pleadings, briefs, and other documents required or permitted to be filed shall be furnished the Commission.
- (b) When service of any document is made by the Commission, an additional copy of that document shall be filed for each party to the proceeding.

§ 1.52 Subscription and verification.

The original of all petitions, motions, pleadings, briefs, and other documents filed by any party represented by counsel, shall be signed by at least one attorney of record in his individual name. whose address shall be stated. Copies should be conformed. A party who is not represented by an attorney shall sign and verify the document and state his Except when otherwise speaddress. cifically provided by rule or statute, documents signed by the attorney for a party need not be verified or accompanied by affidavit. The signature of an attorney constitutes a certificate by him that he has read the document: that to the best of his knowledge, information. and belief there is good ground to support it: and that it is not interposed for delay. If the original of a document is not signed or is signed with intent to defeat the purpose of this section, it may be stricken as sham and false, and the matter may proceed as though the document had not been filed. An attorney may be subjected to appropriate disciplinary action, pursuant to § 1.24, for a willful violation of this rule or if scandalous or indecent matter is inserted.

GENERAL APPLICATION PROCEDURES

- § 1.61 Procedures for handling applications requiring special aeronautical study.
- (a) All antenna surveys are conducted by the Antenna Survey Branch of the Engineering and Facilities Division, Field Engineering Bureau.
- (b) Each operating bureau or office examines the applications for which it is responsible to ascertain whether or not antenna consideration is required. If such consideration is required, the antenna data is furnished to the Antenna Survey Branch.
- (c) The Antenna Survey Branch then ascertains whether applicant is required to submit a "Notice of Proposed Construction or Alteration" [FAA Form 7460-1] to the Federal Aviation Administration.
- (d) If FAA Form 7460-1 is not required, the application and appropriate antenna painting and lighting specifications are returned to the originating bureau or office for such further action as is necessary.

- (e) If FAA Form 7460-1 is required, the originating bureau or office will be so advised. Unless the application includes a statement that FAA Form 7460-1 has been submitted to the Federal Aviation Administration, the originating bureau or office will notify the applicant to do so.
- (f) Upon receipt of a report from the Federal Aviation Administration approving a proposed antenna, the Antenna Survey Branch prescribes antenna tower painting and lighting specifications or other conditions in accordance with the provisions of Part 17 of this chapter and forwards this information to the originating bureau or office. If the proposed tower is disapproved, a report of the disapproval is forwarded to the originating bureau or office.
- (g) Where one or more antenna farm areas have been designated for a community or communities (see § 17.9 of this chapter), an application for a construction permit proposing the erection of an antenna structure over 1,000 feet in height above ground to serve such community or communities will not be accepted for filing unless:
- (1) It is proposed to locate the antenna structure in a designated antenna farm area, or
- (2) It is accompanied by a statement from the Federal Aviation Administration that the proposed structure will not constitute a menace to air navigation, or
- (3) It is accompanied by a request for waiver setting forth reasons sufficient, if true, to justify such a waiver.

NOTE: By Commission Order (FCC 65-455), 30 F.R. 7419, June 5, 1965, the Commission issued the following policy statement concerning the height of radio and television antenna towers:

"We have concluded that this objective can best be achieved by adopting the following policy: Applications for antenna towers higher than 2,000 feet above ground will be presumed to be inconsistent with the public interest, and the applicant will have a burden of overcoming that strong presumption. The applicant must accompany its application with a detailed showing directed to meeting this burden. Only in the exceptional case, where the Commission concludes that a clear and compelling showing has been made that there are public interest reasons requiring a tower higher than 2,000 feet above ground, and after the parties have complied with applicable FAA procedures, and full Commission coordination with FAA on the question of menace to air navigation, will a grant be made. Applicants and parties in interest will, of course, be afforded their statutory hearing rights."

[28 F.R. 12415, Nov. 22, 1963, as amended at 82 F.R. 8813, June 21, 1967; 32 F.R. 20860, Dec. 28, 1967; 34 F.R. 6481, Apr. 15, 1969]

§ 1.62 Operation pending action on renewal application.

(a) (1) Where there is pending before the Commission at the time of expiration of license any proper and timely application for renewal of license with respect to any activity of a continuing nature, in accordance with the provisions of section 9(b) of the Administrative Procedure Act, such license shall continue in effect without further action by the Commission until such time as the Commission shall make a final determination with respect to the renewal application. No operation by any licensee under this section shall be construed as a finding by the Commission that the operation will serve the public interest, convenience, or necessity, nor shall such operation in any way affect or limit the action of the Commission with respect to any pending application or proceeding.

(2) A licensee operating by virtue of this paragraph shall, after the date of expiration specified in the license, post, in addition to the original license, any acknowledgment received from the Commission that the renewal application has been accepted for filing or a signed copy of the application for renewal of license which has been submitted by the licensee, or in services other than broadcast and common carrier, a statement certifying that the licensee has mailed or filed a renewal application, specifying the date of mailing or filing.

(b) Where there is pending before the Commission at the time of expiration of license any proper and timely application for renewal or extension of the term of a license with respect to any activity not of a continuing nature, the Commission may in its discretion grant a temporary extension of such license pending determination of such application. No such temporary extension shall be construed as a finding by the Commission that the operation of any radio station thereunder will serve the public interest, convenience, or necessity beyond the express terms of such temporary extension of license, nor shall such temporary extension in any way affect or limit the action of the Commission with respect to any pending application or proceeding.

(c) Except where an instrument of authorization clearly states on its face that it relates to an activity not of a continuing nature, or where the non-continuing nature is otherwise clearly apparent upon the face of the authorization, all licenses issued by the Commission shall be deemed to be related to an activity of a continuing nature.

(5 U.S.C. 558)

§ 1.65 Substantial and significant changes in information furnished by applicants to the Commission.

Each applicant is responsible for the continuing accuracy and completeness of information furnished in a pending application or in Commission proceedings involving a pending application. Whenever the information furnished in the pending application is no longer substantially accurate and complete in all significant respects, the applicant shall as promptly as possible and in any event within 30 days, unless good cause is shown, amend or request the amendment of his application so as to furnish such additional or corrected information as may be appropriate. Whenever there has been a substantial change as to any other matter which may be of decisional significance in a Commission proceeding involving the pending application, the applicant shall as promptly as possible and in any event within 30 days, unless good cause is shown, submit a statement furnishing such additional or corrected information as may be appropriate, which shall be served upon parties of record in accordance with § 1.47. Where the matter is before any court for review, statements and requests to amend shall in addition be served upon the Commission's General Counsel. For the purposes of this section, an application is "pending" before the Commission from the time it is accepted for filing by the Commission until a Commission grant or denial of the application is no longer subject to reconsideration by the Commission or to review by any court. [29 F.R. 15518, Nov. 19, 1964]

§ 1.68 Action on application for license to cover construction permit.

(a) An application for license by the lawful holder of a construction permit will be granted without hearing where the Commission, upon examination of such application, finds that all the terms, conditions, and obligations set forth in the application and permit have been fully met, and that no cause or circumstance arising or first coming to the knowledge of the Commission since the granting of the permit would, in the judgment of the Commission, make the operation of such station against the public interest.

(b) In the event the Commission is unable to make the findings in paragraph (a) of this section, the Commission will designate the application for hearing upon specified issues.

(Sec. 319, 48 Stat. 1089, as amended; 47 U.S.C. 319)

- § 1.70 Procedures for handling applications involving the use of certain lands and reservations under the jurisdiction of the U.S. Government.
- (a) Any application proposing new or modified transmitting facilities to be located on land under the jurisdiction of the U.S. Forest Service or the Bureau of Land Management shall include a statement that the facilities will be so located. The applicant shall certify that a proper application for a land use permit has been filed with the agency involved on the form prescribed by that agency. Except for applications filed pursuant to Part 73 of this chapter, the applicant shall certify on his application with the Commission that notification has been received from the land agency indicating that, subject to the Commission's determination concerning electromagnetic compatibility, the site in question is available. With respect to applications filed pursuant to Part 73 of this chapter, final consideration of the application will be withheld pending receipt of a certification from the applicant concerning availability of the proposed site. All certifications concerning site availability shall include the date of the notification from the land agency and shall identify the land office from which the notification was sent.
- (b) Following notification to the applicant that, subject to the Commission's determination concerning electromagnetic compatibility, the proposed antenna site is available, the appropriate Government agency will, consistent with agreement between the Commission and the Director of Telecommunications Management, proceed as follows:
- If there are no existing radio users of, or applicants for, the site, notify the Commission to this effect;
- (2) If there are existing user(s) or applicant(s), send a notice containing the

- relevant technical parameters of the proposed new or modified transmitting facilities to all existing users of and applicants for the site in question and simultaneously furnish the Commission with a list of all such users and applicants together with a copy of the aforementioned notice.
- (c) Any existing user of the site, or applicant therefor, may, within 30 days after the date of the notice sent by the appropriate land use agency, file comments concerning the proposed new or modified installation with the Commission, with copies to the applicant and to the land use agency. These comments shall include relevant technical data and an engineering analysis establishing and evaluating the proximity interference expected in either or both directions, together with an explanation of any technical measures that may be taken to eliminate or minimize the expected interference.
- (d) Within 30 days from the last day for filing comments the applicant shall furnish to the Commission a competent engineering analysis of, and comments on, each objection received, together with a clear indication of the measures that the applicant is prepared to take to eliminate or minimize the expected interference.
- (e) If no objections based on electromagnetic compatibility problems are received within 30 days after the notice has been sent by the appropriate land use agency, the application will be processed by the Commission in the normal manner. Should data be received indicating technical objections to the proposed installation or modification, the Commission will determine, through appropriate procedures, the validity of such objections. These procedures may include issuance of a special temporary authorization for a relatively short period of time to test the technical feasibility of the proposed operation. Commission will then determine whether the application should be granted, and, if so, what operating conditions should be imposed. In general, the responsibility for correcting proximity interference will be upon the applicant. Any user affected, however, will be expected to extend all reasonable cooperation in reaching a satisfactory solution. some cases, this may involve adjustments on the part of existing users.
- (f) Upon issuance of a station authorization, which may be restricted to re-

flect conditions imposed by the Commission, the applicant may then obtain a final site permit from the U.S. Forest Service, U.S. Department of Agriculture, or the Bureau of Land Management, U.S. Department of the Interior.

[32 F.R. 2890, Feb. 15, 1967]

§ 1.77 Detailed application procedures; cross references.

The application procedures set forth in §§ 1.61-1.68 are general in nature. More detailed procedures are set forth in this chapter as follows:

(a) Rules governing applications for authorizations in the Broadcast Radio Services are set forth in Subpart D of this part.

(b) Rules governing applications for authorizations in the Common Carrier Radio Services are set forth in Subpart E of this part.

(c) Rules governing applications for authorizations in the Safety and Special Radio Services are set forth in Subpart F of this part.

(d) Rules governing applications for authorizations in the Experimental Radio Services (other than broadcast) are set forth in Part 5 of this chapter.

(e) Rules governing applications for authorizations in the Domestic Public Radio Services are set forth in Part 21 of this chapter.

(f) Rules governing applications for authorizations in the Industrial, Scientific, and Medical Service are set forth in Part 18 of this chapter.

(g) Rules governing applications for type approval and type acceptance of equipment are set forth in Part 2, Subpart F, of this chapter.

(h) Rules governing applications for operator licenses and permits are set forth in §§ 1.83 and 1.84 of this chapter.

MISCELLANEOUS PROCEEDINGS

- § 1.80 Forfeiture proceedings (excluding those pertaining to broadcast licensees and permittees or ships and ship masters).
- (a) Basis. This section is based on section 510 of the Communications Act of 1934, as amended, which provides for the payment of monetary forfeitures to the United States by certain persons who are shown to have willfully or repeatedly committed certain offenses listed therein.
- (b) Applicability. These forfeiture provisions shall apply to the licensee of

any radio station (and, in certain indicated categories, the person operating such station) other than broadcast stations licensed under part 73 or part 74 of this chapter and stations governed by the provisions of Part II or Part III of Title III, or section 507, of the Communications Act of 1934, as amended. For purposes of this section, the term "radio operator" means any person who transmits over a radio station whether or not such person holds a radio operator license issued by the Commission.

- (c) Categories of offenses creating liability. The categories of offenses for which forfeitures shall attach hereunder are set forth as follows (categories (2), (3), (5), and (6) apply both to the station licensee and the radio operator; the remaining categories apply only to the station licensee):
- (1) Operation of a radio station by any person not holding a valid radio operator license or permit of the class prescribed by the provisions in this chapter for the operation of such station.
- (2) Operation of a radio station without identifying such station at the times and in the manner prescribed by the provisions in this chapter for such station.
- (3) Transmission of a false call sign or of a false distress call or message in violation of the provisions in this chapter.
- (4) Operation of a radio station on a frequency not authorized by the Commission for use by such station, including operation with a frequency deviation beyond frequency tolerances authorized by the provisions in this chapter.
- (5) Transmission of unauthorized communications on any frequency designated by the provisions in this chapter as a distress frequency or a calling frequency.
- (6) Operation of a radio station so as to interfere, in violation of the provisions in this chapter, with any distress call or distress communication.
- (7) Failure to attenuate spurious emissions of a radio station to the extent required by the provisions in this chapter applicable to such station.
- (8) Operation of a radio station with power in excess of that authorized by the Commission for the particular station.
- (9) Use of a radio station to render a communication service, including the transmission of communications not permitted by the provisions in this chapter.

which is not authorized by the Commission for the particular station.

- (10) Operation of a radio station with a type of emission not authorized by the Commission for the particular station.
- (11) Operation of a radio station with transmitting equipment not authorized by the Commission for the particular station.
- (12) Failure to respond to a written official communication from the Commission.
- (d) Time limitation on forfeiture liability. A station licensee or radio operator is not subject to forfeiture liability for any offense which occurred more than 90 days prior to the day on which a notice of apparent liability (see paragraph (f) of this section) describing that offense is mailed. Such prior offenses may, however, be cited in a notice of apparent liability for the purposes of establishing the willfulness or the repetitiveness of offenses committed within such 90 day period.
- (e) Forfeiture amounts and limitations. For each separate offense hereunder, the licensee of the radio station involved shall, in addition to any other penalty prescribed by law, forfeit to the United States a sum not to exceed \$100. For each separate offense coming within the provisions of subparagraphs (2), (3), (5), or (6) of paragraph (c) of this section, the radio operator, if he be a different person from the station licensee, shall, in addition to any other penalty prescribed by law, forfeit to the United States a sum not to exceed \$100. However, in the event of multiple offenses within 90 days prior to the date of notice of apparent liability, the following limitations on total monetary liability shall govern. For all offenses, regardless of the total number, coming within any one category of offense listed in paragraph (c) of this section, the maximum forfeiture liability shall be \$100. Furthermore, for all offenses, regardless of the total number, coming within more than one of the categories of offenses listed in paragraph (c) of this section, the maximum forfeiture liability for a radio station licensee shall be \$500, and the maximum forfeiture liability for a radio operator shall be \$400.
- (f) Notice of apparent liability. Whenever it appears that a station licensee and/or a radio operator willfully or repeatedly has committed an offense or offenses within those categories listed

- in paragraph (c) of this section, a written notice of apparent liability will be sent to him by certified mail to his last known address. See § 1.5 of this chapter. This notice will set forth the facts which indicate apparent liability; will identify the offenses listed in paragraph (c) of this section which are involved and the provisions of the Commission's rules and of the license which appear to have been violated; will state the amount of the forfeiture imposed; and will summarize the courses of action available to the person involved under the provisions of paragraphs (g) and (h) of this section.
- (g) Response to a notice of apparent liability. The station licensee or radio operator involved shall respond to a notice of apparent liability within 30 days after the notice is mailed, or within such longer period of time as may be specified in the notice. Any of the following actions by the station licensee or radio operator shall constitute a response meeting the requirements of this paragraph:
- (1) Payment of the forfeiture in the amount specified in the notice of apparent liability. The forfeiture should be paid by check or money order drawn to the order of the Treasurer of the United States and should be mailed to the Federal Communications Commission, Washington, D.C., 20554. The Commission does not accept responsibility for cash payments sent through the mails.
- (2) Submission of a written statement denying liability for the forfeiture in whole or in part and/or requesting (even if liability is admitted) that the forfeiture be canceled or reduced because of extenuating circumstances connected with the offense. Allegations or requests made in any such statement must be supported by detailed factual data and reasons. The statement should be mailed to the Federal Communications Commission, Washington, D.C., 20554.
- (3) Submission of a written request for a personal interview with an official of the Commission. The request should be submitted to the Federal Communications Commission, Washington, D.C., 20554. Submission of such a request does not preclude submission of the statement described in subparagraph (2) of this paragraph; either or both may be submitted.
- (h) Personal interview. (1) If a personal interview is requested, it will be held at the field office or monitoring station of the Commission nearest to

the place of residence of the station licensee or radio operator involved. See § 0.121 of this chapter.

(2) The interview will be held within 30 days after the request is filed with the Commission, and may be scheduled by the Commission for any mutually

period.

(3) In the personal interview, the station licensee or radio operator will be afforded an opportunity to state his position with regard to forfeiture liability and the amount of the forfeiture imposed, and to request that the forfeiture be cancelled or reduced.

convenient time within that 30 day

(4) A summary of the interview will be prepared by the interviewing official and transmitted by him to the Washington offices of the Commission. The interviewing official is not authorized to impose, cancel, or reduce any forfeiture.

(5) The failure of the station licensee or operator to appear at the time and place appointed for the interview shall constitute a waiver of his right to such

an interview.

- (i) Commission action after written statement or interview. After the submission of a written statement and/or after a personal interview, as prescribed in paragraphs (g) and (h) of this section, the Commission will consider all relevant information available to it. Based on such considerations, the Commission will (1) cancel the forfeiture, or (2) offer to reduce the amount of the forfeiture, or (3) require the forfeiture to be paid in full. A notice of such Commission action, stating the amount of the forfeiture (if any), and the date by which it must be paid, will thereupon be mailed to the station licensee or radio operator involved. The forfeiture in the amount stated shall be paid by check or money order drawn to the order of the Treasurer of the United States and shall be mailed to the Federal Communications Commission, Washington, D.C., 20554. Commission does not accept responsibility for cash payments sent through the mails.
- (j) Judicial enforcement of forfeitures imposed by the Commission. If a station licensee or radio operator fails to respond to the notice of apparent liability as required by paragraph (g) of this section, or if he fails to pay the forfeiture imposed by the Commission under paragraph (i) of this section, the case will, without further notice, be referred by the Commission to the Department of

Justice for prosecution in the appropriate Federal District Court to recover the amount of the forfeiture initially imposed.

(Sec. 1, 76 Stat. 68; 47 U.S.C. 510) [28 F.R. 12415, Nov. 26, 1963, as amended at 30 F.R. 9767, Aug. 5, 1965]

§ 1.81 Other forfeiture procedures; cross references.

(a) Rules governing the forfeiture liability of broadcast station licensees and permittees are set forth in § 1.621.

(b) Rules governing the forfeiture liability of ships and ship masters are set

forth in § 1.991.

§ 1.83 Application for radio operator license.

(a) Commercial radio operator license. Applications for commercial radio operator licenses shall be filed as follows:

(1) Restricted radiotelephone operator permit. Application for a restricted radiotelephone operator permit shall be filed on FCC Form 753-1, entitled "Application for Restricted Radiotelephone Operator Permit by Declaration".

(2) All others. Application for a new, renewed, replacement, or duplicate commercial radio operator license, for a verification card, or for a verification of operator license FCC Form 759 (for additional posting) shall be filed on FCC Form 756, entitled "Application for Ra-

dio Operator License".

(3) Waiver of nationality requirement. Applicants desiring waiver of the nationality requirement pursuant to section 303(1) of the Communications Act shall submit a request for waiver with the application form required by this paragraph. The request for waiver shall be filed on FCC Form 755, entitled "Supplemental Application for Commercial Radio Operator License".

(b) Amateur radio operator license. Application for amateur radio operator license is included with the application for station license. See § 1.934.

§ 1.84 Procedure with respect to commercial radio operator license applications.

(a) Upon acceptance of an application for a license submitted in accordance with Part 13 of this chapter, an examination, if required, is conducted. If applicant is found qualified in all respects, the license will be issued. When doubts as to nationality, character, or physical condition arise, the application is referred to the Commission's Field

Offices Division, Field Engineering Bureau, Washington, D.C., 20555, for consideration. If it appears that further information is required to determine the applicant's qualifications or that a grant of the application will not serve the public interest, the applicant will be notified in writing and given an opportunity to furnish such written showings as the Commission may request and as the applicant may desire to submit. If, from the information furnished, it does not appear that the applicant is qualified or that the public interest would be served by a grant of the application, the applicant will be advised thereof in writing and given the opportunity to request, within the period of time to be specified in such writing, that the application be set for hearing. In the case of failure timely to request such hearing, the application will be denied.

(b) Where an examination is not required, the application will be handled with respect to other matters in accordance with the procedure in paragraph (a) of this section.

(c) Applications for renewal of license are handled in accordance with the procedure contained in paragraph (a) of this section, except that no examination is required unless the circumstances as set forth in § 13.28 of this chapter exist, in which case a renewal examination will be required.

(d) Failure to prosecute an application, or failure to respond to official correspondence or request for additional information, will be cause for dismissal. Such dismissal will be without prejudice where an application has not yet been designated for hearing; such dismissal may be made with prejudice after an application has been designated for hearing.

(Sec. 303(1), 48 Stat. 1082, as amended; 47 U.S.C. 303(1)) [29 F.R. 15965, Nov. 24, 1964]

§ 1.85 Suspension of operator licenses.

Whenever grounds exist for suspension of an operator license, as provided in section 303(m) of the Communications Act, the Chief of the Safety and Special Radio Services Bureau, with respect to amateur operator licenses, or the Chief of the Field Engineering Bureau, with respect to commercial operator licenses, may issue an order suspending the operator license. No order of suspension of any operator's license shall take effect until 15 days' notice in writing of the cause for the proposed suspension has

been given to the operator licensee, who may make written application to the Commission at any time within said 15 days for a hearing upon such order. The notice to the operator licensee shall not be effective until actually received by him, and from that time he shall have 15 days in which to mail the said application. In the event that physical conditions prevent mailing of the application before the expiration of the 15-day period, the application shall then be mailed as soon as possible thereafter. accompanied by a satisfactory explanation of the delay. Upon receipt by the Commission of such application for hearing said order of suspension shall be designated for hearing by the Chief, Safety and Special Radio Services Bureau, or the chief, Field Engineering Bureau, as the case may be, and said order of suspension shall be held in abevance until the conclusion of the hearing. Upon the conclusion of said hearing, the Commission may affirm, modify, or revoke said order of suspension. If the license is ordered suspended, the operator shall send his operator license to the office of the Commission in Washington, D.C., on or before the effective date of the order, or, if the effective date has passed at the time notice is received, the license shall be sent to the Commission forthwith.

(Sec. 303(m), 48 Stat. 1082, as amended; 47 U.S.C. 303(m))

§ 1.87 Modification of license or construction permit on motion of the Commission.

(a) Whenever it appears that a station license or construction permit should be modified, the Commission will notify the licensee or permittee in writing of the proposed action and the grounds and reasons therefor and direct him to show cause why an order modifying the license or construction permit in the manner proposed by the Commission should not be issued.

(b) Any order to show cause issued in accordance with paragraph (a) of this section will notify the licensee or permittee that he may request, within a period of time to be stated in the order to show cause, that a hearing be held on the proposed modification. In case of timely request, a hearing will be held on the proposed modification, in no event less than 30 days after the receipt of the order to show cause, unless the Commis-

sion finds that safety of life or property require the fixing of a shorter period.

(c) In order to avail himself of the right to request a hearing and of the opportunity to appear and give evidence upon the matters specified in the order to show cause, the licensee or permittee. in person or by his attorney, shall, within such period of time as may be specified in the order to show cause, file with the Commission a written statement stating that he requests a hearing and will appear at the hearing and present evidence on the matter specified in the order to show cause. Such written statement must contain a detailed response to the matter specified in the order to show cause, and the permittee or licensee shall be limited in the hearing to matters fairly encompassed within the issues raised by the response.

(d) The right to request a hearing shall, unless good cause is shown in a petition to be filed not later than 5 days before the lapse of the time specified in paragraph (c) of this section, be deemed

waived:

(1) In case of failure to timely file a written statement as required by paragraph (c) of this section:

(2) In case of filing the written statement provided for in paragraph (c) of this section but failure to appear at the hearing, either in person or by counsel.

(e) Where the right to request a hearing is waived and no written statement has been filed within the period of time specified in the order to show cause, the licensee or permittee will be deemed to consent to the modification as proposed in the order to show cause and a final decision will be issued by the Commission

accordingly.

(f) Where the right to request a hearing has been waived, a written statement may be filed within the period of time to be specified in the order to show cause, showing with particularity why the license or construction permit should not be modified or not so modified as proposed in the order to show cause. this case, the Commission may, depending upon the facts alleged and proof offered, either call upon the submitting party to furnish additional information under oath, designate the proceeding for hearing, or issue without further proceedings an order modifying the construction permit or license as proposed in the order to show cause or in said written statement. The order to show cause will advise the person against whom it is directed of procedure set forth in this

paragraph.

(g) Any order of modification issued pursuant to this section shall include a statement of the findings and the grounds and reasons therefor, shall specify the effective date of the order, and shall be served on the licensee or permittee.

(Sec. 12, 66 Stat. 717; 47 U.S.C. 316)

§ 1.89 Notice of violations.

(a) Except in cases of wilfulness or those in which public health, interest, or safety requires otherwise, any licensee who appears to have violated any provision of the Communications Act or any provision of this chapter will, before revocation, suspension, or cease and desist proceedings are instituted, be served with a written notice calling these facts to his attention and requesting a statement concerning the matter. FCC Form 793 may be used for this purpose.

(b) Within 10 days from receipt of notice or such other period as may be specified, the licensee shall send a written answer, in duplicate, direct to the office of the Commission originating the official notice. If an answer cannot be sent nor an acknowledgment made within such 10-day period by reason of illness or other unavoidable circumstances, acknowledgment and answer shall be made at the earliest practicable date with a satisfactory explanation of the delay.

(c) The answer to each notice shall be complete in itself and shall not be abbreviated by reference to other communications or answers to other notices. In every instance the answer shall contain a statement of action taken to correct the condition or omission complained of and to preclude its recurrence. In addition:

(1) If the notice relates to violations that may be due to the physical or electrical characteristics of transmitting apparatus and any new apparatus is to be installed, the answer shall state the date such apparatus was ordered, the name of the manufacturer, and the promised date of delivery. If the installation of such apparatus requires a construction permit, the file number of the application shall be given, or if a file number has not been assigned by the Commission. such identification shall be given as will permit ready identification of the application.

(2) If the notice of violation relates to lack of attention to or improper operation of the transmitter, the name and license number of the operator in charge shall be given.

§ 1.91 Revocation and/or cease and desist proceedings; hearings.

- (a) If it appears that a station license or construction permit should be revoked and/or that a cease and desist order should be issued, the Commission will issue an order directing the person to show cause why an order of revocation and/or a cease and desist order, as the facts may warrant, should not be issued.
- (b) An order to show cause why an order of revocation and/or a cease and desist order should not be issued will contain a statement of the matters with respect to which the Commission is inquiring and will call upon the person to whom it is directed (the respondent) to appear before the Commission at a hearing, at a time and place stated in the order, but not less than thirty days after the receipt of such order, and give evidence upon the matters specified in the order to show cause. However, if safety of life or property is involved, the order to show cause may specify a hearing date less than thirty days from the receipt of such order.
- (c) To avail himself of such opportunity for hearing, the respondent, personally or by his attorney, shall file with the Commission, within thirty days of the service of the order or such shorter period as may be specified therein, a written appearance stating that he will appear at the hearing and present evidence on the matters specified in the order. The Commission in its discretion may accept a late appearance. However, an appearance tendered after the specified time has expired will not be accepted unless accompanied by a petition stating with particularity the facts and reasons relied on to justify such late filing. Such petition for acceptance of late appearance will be granted only if the Commission determines that the facts and reasons stated therein constitute good cause for failure to file on time.
- (d) Hearings on the matters specified in such orders to show cause shall accord with the practice and procedure prescribed in this subpart and Subpart B of this part, with the following excep-

- tions: (1) In all such revocation and/or cease and desist hearings, the burden of proceeding with the introduction of evidence and the burden of proof shall be upon the Commission; and (2) the Commission may specify in a show cause order, when the circumstances of the proceeding require expedition, a time less than that prescribed in §§ 1.276 and 1.277 within which the initial decision in the proceeding shall become effective. exceptions to such initial decision must be filed, parties must file requests for oral argument, and parties must file notice of intention to participate in oral argument.
- (e) Correction of or promise to correct the conditions or matters complained of in a show cause order shall not preclude the issuance of a cease and desist order. Corrections or promises to correct the conditions or matters complained of, and the past record of the licensee, may, however, be considered in determining whether a revocation and/or a cease and desist order should be issued.
- (f) Any order of revocation and/or cease and desist order issued after hearing pursuant to this section shall include a statement of findings and the grounds therefor, shall specify the effective date of the order, and shall be served on the person to whom such order is directed. (Sec. 312, 48 Stat. 1086, as amended; 47 U.S.C. 312)

§ 1.92 Revocation and/or cease and desist proceedings; after waiver of hearing.

- (a) After the issuance of an order to show cause, pursuant to § 1.91, calling upon a person to appear at a hearing before the Commission, the occurrence of any one of the following events or circumstances will constitute a waiver of such hearing and the proceeding thereafter will be conducted in accordance with the provisions of this section.
- (1) The respondent fails to file a timely written appearance as prescribed in § 1.91(c) indicating that he will appear at a hearing and present evidence on the matters specified in the order.
- (2) The respondent, having filed a timely written appearance as prescribed in § 1.91(c), fails in fact to appear in person or by his attorney at the time and place of the duly scheduled hearing.
- (3) The respondent files with the Commission, within the time specified for a written appearance in § 1.91(c).

a written statement expressly waiving his rights to a hearing.

(b) When a hearing is waived under the provisions of paragraph (a) (1) or (3) of this section, a written statement signed by the respondent denying or seeking to mitigate or justify the circumstances or conduct complained of in the order to show cause may be submitted within the time specified in § 1.91(c). The Commission in its discretion may accept a late statement. However, a statement tendered after the specified time has expired will not be accepted unless accompanied by a petition stating with particularity the facts and reasons relied on to justify such late filing. Such petitions for acceptance of a late statement will be granted only if the Commission determines that the facts and reasons stated therein constitute good cause for failure to file on time.

(c) Whenever a hearing is waived by the occurence of any of the events or circumstances listed in paragraph (a) of this section, the Chief Hearing Examiner (or the presiding officer if one has been designated) shall, at the earliest practicable date, issue an order reciting the events or circumstances constituting a waiver of hearing, terminating the hearing proceeding, and certifying the case to the Commission. Such order shall be served upon the respondent.

(d) After a hearing proceeding has been terminated pursuant to paragraph (c) of this section, the Commission will act upon the matters specified in the order to show cause in the regular course The Commission will deof business. termine on the basis of all the information available to it from any source. including such further proceedings as may be warranted, if a revocation order and/or a cease and desist order should issue, and if so, will issue such order. Otherwise, the Commission will issue an order dismissing the proceeding. orders specified in this paragraph will include a statement of the findings of the Commission and the grounds and reasons therefor, will specify the effective date thereof, and will be served upon the respondent.

(e) Corrections or promise to correct the conditions or matters complained of in a show cause order shall not preclude the issuance of a cease and desist order. Corrections or promises to correct the conditions or matters complained of, and the past record of the licensee, may, however, be considered in determining whether a revocation and/or a cease and desist order should be issued.

(Sec. 312, 48 Stat. 1086, as amended; 47 U.S.C. 312) [28 F.R. 12415, Nov. 22, 1963, as amended at 29 F.R. 6443, May 16, 1964]

RECONSIDERATION AND REVIEW OF ACTIONS TAKEN PURSUANT TO DELEGATED AU-THORITY: RECONSIDERATION OF ACTIONS TAKEN BY THE COMMISSION

§ 1.101 General provisions.

Under section 5(d) of the Communications Act of 1934, as amended, the Commission is authorized, by rule or order, to delegate certain of its functions to a panel of commissioners, an individual commissioner, an employee board, or an individual employee. Section 0.201(a) of this chapter describes in general terms the basic categories of delegations which are made by the Commission. Subpart B of Part 0 of this chapter sets forth all delegations which have been made by rule. Sections 1.102-1.120 set forth procedural rules governing reconsideration and review of actions taken pursuant to authority delegated under section 5(d) of the Communications Act, and reconsideration of actions taken by the Commission. As used in §§ 1.102–1.117, the term "designated authority" means any person, panel, or board which has been authorized by rule or order to exercise authority under section 5(d) of the Communications Act.

§ 1.102 Effective dates of actions taken pursuant to delegated authority.

(a) Final actions following review of an initial decision. (1) Final decisions of the Review Board, a commissioner, or panel of commissioners following review of an initial decision shall be effective 40 days after public release of the full text of such final decision.

(2) If a petition for reconsideration of such final decision is filed, the effect of the decision is stayed until 40 days after release of the final order disposing of the petition.

(3) If an application for review of such final decision is filed, or if the Commission on its own motion orders the record of the proceeding before it for review, the effect of the decision is stayed until the Commission's review of the proceeding has been completed.

(b) Non-hearing and interlocutory actions. (1) Non-hearing or interlocutory actions taken pursuant to delegated authority shall, unless otherwise ordered by the designated authority, be effective upon release of the document containing the full text of such action or in the event such a document is not released, upon release of a public notice announcing the action in question.

(2) If a petition for reconsideration of a non-hearing action is filed, the designated authority may in its discretion stay the effect of its action pending disposition of the petition for reconsideration. Petitions for reconsideration of interlocutory actions will not be entertained.

(3) If an application for review of a non-hearing or interlocutory action is filed, or if the Commission reviews the action on its own motion, the Commission may in its discretion stay the effect of any such action until its review of the matters at issue has been completed.

§ 1.104 Preserving the right of review; deferred consideration of application for review.

- (a) The provisions of this section apply to all final actions taken pursuant to delegated authority, including final decisions of the Review Board following review of an initial decision and final actions taken by members of the Commission's staff on non-hearing matters. They do not apply to interlocutory actions of the Review Board or the Chief Hearing Examiner in hearing proceedings. (See §§ 1.106(a) and 1.115(e).)
- (b) Any person desiring Commission consideration of a final action taken pursuant to delegated authority shall file either a petition for reconsideration or an application for review (but not both) within 30 days from the date of release of the document containing the full text of such action, or in case such document is not released, after release of a public notice announcing the action in question. The petition for reconsideration will in all cases be acted upon by the designated authority. The application for review will in all cases be acted upon by the Commission.
- (c) If in any matter one party files a petition for reconsideration and a second party files an application for review, the Commission will withhold action on the application for review until final action has been taken on the petition for reconsideration.
- (d) Any person who has filed a petition for reconsideration may file an application for review within 30 days after

final action on his petition. If a petition for reconsideration has been filed, any person who has filed an application for review may, within 30 days after final action on the petition for reconsideration, amend or withdraw his application for review or substitute an amended application therefor.

§ 1.106 Petitions for reconsideration of final action taken by the Commission en banc or by a designated authority pursuant to a delegation.

- (a) Petitions requesting reconsideration of a final action taken pursuant to delegated authority will be acted on by the designated authority. Petitions requesting reconsideration of a final Commission action will be acted on by the Commission. Petitions requesting reconsideration of an interlocutory ruling made by the Commission, the Review Board or the Chief Hearing Examiner will not be entertained. See § 1.115(e) (2). For purposes of reconsideration, an order disposing of a petition to deny is not an interlocutory ruling. For provisions pertaining to reconsideration of an order designating a case for hearing, see § 1.111.
- (b) Except where the Commission has denied an application for review without specifying reasons therefor, any party to the proceeding, or any other person aggrieved or whose interests are adversely affected by any action taken by the Commission or by the designated authority, may file a petition requesting reconsideration of the action taken. If the petition is filed by a person who is not a party to the proceeding, it shall state with particularity the manner in which he is aggrieved or his interests are adversely affected by the action taken, and shall show good reason why it was not possible for him to participate in the earlier stages of the proceeding.
- (c) A petition for reconsideration which relies on facts which have not previously been presented to the Commission or to the designated authority as the case may be, will be granted only under the following circumstances:
- The facts relied on relate to events which have occurred or circumstances which have changed since the last opportunity to present such matters;
- (2) The facts relied on were unknown to petitioner until after his last opportunity to present such matters, and he could not through the exercise of ordinary diligence have learned of the facts

in question prior to such opportunity; or

(3) The Commission or the designated authority determines that consideration of the facts relied on is required in the public interest.

(d) (1) The petition for reconsideration shall, where appropriate, cite the findings of fact and/or conclusions of law which petitioner believes to be erroneous, and shall state with particularity the respects in which he believes such findings and conclusions should be changed. The petition may request that additional findings of fact and conclusions of law be made.

(2) The petition shall state with particularity the respects in which petitioner believes the action taken by the designated authority should be changed.

(3) The petition shall state specifically the form of relief sought and, subject to this requirement, may contain alternative requests.

- (e) Where a petition for reconsideration is based upon a claim of electrical interference, under appropriate rules in this chapter, to an existing station or a station for which a construction permit is outstanding, such petition, in addition to meeting the other requirements of this section, must be accompanied by an affidavit of a qualified radio engineer. Such affidavit shall show, either by following the procedures set forth in this chapter for determining interference in the absence of measurements, or by actual measurements made in accordance with the methods prescribed in this chapter, that electrical interference will be caused to the station within its normally protected contour.
- (f) The petition for reconsideration and any supplement thereto shall be filed within 30 days from the date of release of the document containing the full text of the action taken or, in case such document is not released, after release of a public notice announcing the action in question, and shall be served upon parties to the proceeding. The petition for reconsideration shall not exceed 25 double spaced typewritten pages. supplement or addition to a petition for reconsideration which has not been acted upon by the Commission or by the designated authority, filed after expiration of the 30 day period, will be considered except upon leave granted upon a separate pleading for leave to file, which shall state the grounds therefor.
- (g) Oppositions to a petition for reconsideration shall be filled within 10

days after the petition is filed, and shall be served upon petitioner and parties to the proceeding. Oppositions shall not exceed 25 double spaced typewritten pages.

(h) Petitioner may reply to oppositions within seven days after the last day for filing oppositions, and any such reply shall be served upon parties to the proceeding. Replies shall not exceed 10 double spaced typewritten pages, and shall be limited to matters raised in the opposition.

(i) Petitions for reconsideration, oppositions, and replies shall conform to the requirements of §§ 1.49, 1.51, and 1.52 and shall be submitted to the Secretary, Federal Communications Commission, Washington, D.C., 20554.

(j) The Commission or designated authority may grant the petition for reconsideration in whole or in part or may deny the petition. Its order will contain a concise statement of the reasons for the action taken. Where the petition for reconsideration relates to an instrument of authorization granted without hearing, the Commission or designated authority will take such action within 90 days after the petition is filed.

(k) (1) If the Commission or designated authority grants the petition for reconsideration in whole or in part, it may, in its order, rule on the merits of the petition.

(2) In the alternative, the Commission or designated authority may, in its order granting the petition, order such further procedure as may be useful to it in reaching a decision on the merits of the petition. In that event, its ruling on the merits will be deferred pending completion of such procedure.

(3) In ruling on the merits of the petition, the Commission or designated authority may affirm, reverse, modify or set aside its original action, or may remand the proceeding for such further action, including rehearing, as may be appropriate. Any order disposing of a petition for reconsideration which reverses, changes or modifies the original order is subject to the same provisions with respect to reconsideration as the A petition for reconoriginal order. sideration of an order which disposed of a petition for reconsideration and which did not reverse, change, or modify the original order may be dismissed by the staff as repetitious.

(1) No evidence other than newly discovered evidence, evidence which has

become available only since the original taking of evidence, or evidence which the Commission or the designated authority believes should have been taken in the original proceeding shall be taken on any rehearing ordered pursuant to the provisions of this section.

(m) The filing of a petition for reconsideration is not a condition precedent to judicial review of any action taken by the Commission or by the designated authority, except where the person seeking such review was not a party to the proceeding resulting in the action, or relies on questions of fact or law upon which the Commission or designated authority has been afforded no opportunity to pass. (See § 1.115(c).) Persons in those categories who meet the requirements of this section may qualify to seek judicial review by filing a petition for reconsideration.

(n) Without special order of the Commission, the filing of a petition for reconsideration shall not excuse any person from complying with or obeying any decision, order, or requirement of the Commission, or operate in any manner to stay or postpone the enforcement thereof. However, upon good cause shown, the Commission will stay the effectiveness of its order or requirement pending a decision on the petition for reconsideration. (This paragraph applies only to actions of the Commission en banc. For provisions applicable to actions under delegated authority, see § 1.102.)

Note: Petitions for reconsideration of Commission action in rule making proceedings conducted under section 4 of the Administrative Procedure Act need not be served on participants in the proceeding. When such petitions are filed in proper form, public notice of their filing will be given. Oppositions to such petitions may be filed within 10 days after such public notice is given and need be served only on the person who filed the petition. Replies to such oppositions need be served only on the person who filed the opposition.

(Sec. 405, 48 Stat. 1095, as amended; 47 U.S.C. 405) [28 F.R. 12415, Nov. 22, 1963, as amended at 30 F.R. 4479, Apr. 7, 1965; 30 F.R. 8837, July 14, 1965; 31 F.R. 3074, Feb. 24, 1966]

§ 1.108 Reconsideration on Commission's own motion.

The Commission may, on its own motion, set aside any action made or taken by it within 30 days after release of the document containing the full text of such

action or, in case such a document is not released, after release of a "Public Notice" announcing the action in question.

§ 1.110 Partial grants; rejection and designation for hearing.

Where the Commission without a hearing grants any application in part, or with any privileges, terms, or conditions other than those requested, or subject to any interference that may result to a station if designated application or applications are subsequently granted, the action of the Commission shall be considered as a grant of such application unless the applicant shall, within 30 days from the date on which such grant is made or from its effective date if a later date is specified, file with the Commission a written request rejecting the grant as made. Upon receipt of such request, the Commission will vacate its original action upon the application and set the application for hearing in the same manner as other applications are set for hearing.

§ 1.111 Petitions for reconsideration of an order designating a case for hearing.

- (a) A petition for reconsideration of an order designating a case for hearing will be entertained only in the following circumstances:
- (1) Where the petition relates to an adverse ruling with respect to the petitioner's participation in the proceeding:
- (2) Where the petition is filed by an applicant and asserts that his application should have been granted without hearing.
- (b) A petition for reconsideration under paragraph (a) of this section will be acted upon by the Commission. All questions raised in the petition and relating to the designation order will be considered.

(Sec. 405, 48 Stat. 1095, as amended; 47 U.S.C. 405) [30 F.R. 8838, July 14, 1965]

§ 1.113 Action modified or set aside by person, panel, or board.

- (a) Within 30 days after public notice has been given of any action taken pursuant to delegated authority, the person, panel, or board taking the action may modify or set it aside on its own motion.
- (b) Within 60 days after notice of any sanction imposed under delegated authority has been served on the person affected, the person, panel, or board which imposed the sanction may modify or set it aside on its own motion.

(c) Petitions for reconsideration and applications for review shall be directed to the actions as thus modified, and the time for filing such pleadings shall be computed from the date upon which public notice of the modified action is given or notice of the modified sanction is served on the person affected.

§ 1.115 Application for review of action taken pursuant to delegated authority.

(a) Any person aggrieved by any action taken pursuant to delegated authority may file an application requesting review of that action by the Commission. Any person filing an application for review who has not previously participated in the proceeding shall include with his application a statement describing with particularity the manner in which he is aggrieved by the action taken and showing good reason why it was not possible for him to participate in the earlier stages of the proceeding. Any application for review which fails to make an adequate showing in this respect will be dismissed.

(b) (1) The application for review shall concisely and plainly state the questions presented for review with reference, where appropriate, to the findings of fact or conclusions of law.

(2) The application for review shall specify with particularity, from among the following, the factor(s) which warrant Commission consideration of the questions presented:

(i) The action taken pursuant to delegated authority is in conflict with statute, regulation, case precedent, or established Commission policy.

(ii) The action involves a question of law or policy which has not previously been resolved by the Commission.

(iii) The action involves application of a precedent or policy which should be overturned or revised.

(iv) An erroneous finding as to an important or material question of fact.

(v) Prejudicial procedural error.

(3) The application for review shall state with particularity the respects in which the action taken by the designated authority should be changed.

(4) The application for review shall state the form of relief sought and, subject to this requirement, may contain alternative requests.

Note: If the Commission grants an application for review of a final decision of the Review Board, it will, as the usual practice, permit the parties to file briefs and present oral argument. The Commission will rarely dispose of the merits of a case upon the basis of the application for review and related pleadings. Thus, except where the matter is interlocutory in nature, the application for review should be prepared with the understanding that its purpose is not to obtain a Commission decision on the merits of the issues but rather to convince the Commission to review those issues.

(c) No application for review will be granted if it relies on questions of fact or law upon which the designated authority has been afforded no opportunity to pass.

Note: Subject to the requirements of § 1.106, new questions of fact or law may be presented to the designated authority in a petition for reconsideration.

- (d) Except as provided in paragraph (e) of this section, the application for review and any supplement thereto shall be filed within 30 days from the date of release of the document containing the full text of such action, or in case such a document is not released, after release of a public notice announcing the action in question. Oppositions to the application shall be filed within 15 days after the application is filed. Replies to oppositions shall be filed within 10 days after the opposition is filed and shall be limited to matters raised in the opposition.
- (e) (1) Applications for review of interlocutory rulings made by the Review Board or the Chief Hearing Examiner shall be filed within five days after the order is released or the ruling is made. Oppositions to such applications shall be filed within five days after the application is filed. Replies to such oppositions shall be filed within five days after the opposition is filed and shall be limited to matters raised in the opposition.
- (2) The failure to file an application for review of an interlocutory ruling, or the denial of such application by the Commission, shall not preclude any party entitled to file exceptions to the initial decision from requesting review of the ruling at the time when exceptions are filed. Such requests will be considered in the same manner as exceptions are considered.

Note: Unless the ruling complained of is fundamental and affects the conduct of the entire proceeding, requests for review should be deferred until the time when exceptions are filed.

(f) Applications for review and oppositions shall conform to the requirements of §§ 1.49, 1.51, and 1.52, and shall be submitted to the Secretary, Federal Communications Commission, Washington, D.C., 20554. The application for review shall be served upon the parties to the proceeding. Oppositions to the application for review shall be served on the person seeking review and parties to the proceeding. Applications for review and oppositions shall not exceed 25 double spaced typewritten pages and, in the case of interlocutory matters, shall not exceed 10 double spaced typewritten pages. Replies to oppositions shall not exceed 10 double spaced typewritten pages.

(g) The Commission may grant the application for review in whole or in part, or may deny the application, without specifying reasons for the action taken. No petition requesting reconsideration of an order which denies an application for review without specifying reasons therefor will be entertained.

(h) If the Commission grants the application for review, it may order such further procedure, including briefs and oral argument, as it may deem useful, or may stay the effect of the action taken pursuant to delegated authority pending its further consideration of the matters at issue. (See § 1.102.)

(i) With or without any such additional procedure, the Commission may either affirm, reverse, modify, or set aside the action taken, or remand the proceeding to the designated authority for reconsideration in accordance with its instructions. If an evidentiary hearing has been held, the Commission may remand the proceeding to the person(s) who conducted the hearing for rehearing on such issues and in accordance with such instructions as may be appropriate. An order of the Commission which reverses or modifies the action taken pursuant to delegated authority, or remands the matter for further proceedings, is subject to the same provisions with respect to reconsideration as an original action of the Commission.

(j) No evidence other than newly discovered evidence, evidence which has become available only since the original taking of evidence, or evidence which the Commission believes should have been taken in the original proceeding shall be taken on any rehearing ordered pursuant to the provisions of this section.

(k) The filing of an application for

review shall be a condition precedent to judicial review of any action taken pursuant to delegated authority.

(Sec. 5, 48 Stat. 1068, as amended; 47 U.S.C. 155)

§ 1.117 Review on motion of the Com-

(a) Within 40 days after public notice is given of any action taken pursuant to delegated authority, the Commission may on its own motion order the record of the proceeding before it for review.

(b) If the Commission reviews the proceeding on its own motion, it may order such further procedure as may be useful to it in its review of the action taken pursuant to delegated authority.

(c) With or without such further procedure, the Commission may either affirm, reverse, modify, or set aside the action taken, or remand the proceeding to the designated authority for reconsideration in accordance with its instructions. If an evidentiary hearing has been held, the Commission may remand the proceeding to the person(s) who conducted the hearing for rehearing on such issues and in accordance with such instructions as may be appropriate. order of the Commission which reverses or modifies the action taken pursuant to delegated authority, or remands the matter for further proceedings, is subject to the same provisions with respect to reconsideration as an original action of the Commission.

§ 1.120 Protests of grants without hearing.

(a) The provisions of this section shall not be applicable to any application:
(1) Filed on or after December 12, 1960;
(2) filed before December 12, 1960, but substantially amended (as defined in the applicable provisions of this chapter) on or after that date; or (3) filed before December 12, 1960, and not thereafter substantially amended, but with respect to which the rules in this chapter provide an opportunity for petitions to deny to be filed under section 309 of the Communications Act, as amended. See §§ 1.580 and 1.962.

(b) Where any instrument of authorization for a radio station, other than a license pursuant to a construction permit, has been granted without a hearing, any party in interest may file a protest directed to such grant and request a hearing on the application

granted. Such protest shall be signed by the protestant and subscribed to under oath. Such protest must be filed with the Commission within 30 days after release of the document containing the full text of such action, or in case such a document is not released, after release of a "Public Notice" announcing the action in question and must separately set forth:

- (1) Such allegations of fact as will show the protestant to be a party in interest, i.e., a person aggrieved or whose interests are adversely affected by the Commission's authorization, protest of which is sought. Each such allegation of fact shall be separately stated.
- (2) Facts indicating the reasons why the grant was improperly made or would otherwise not be in the public interest. Each such reason shall be separately stated, and facts in support thereof shall be specified in detail and shall not include general non-specific conclusory arguments and allegations.
- (3) The specific issues upon which protestant wishes a hearing to be held, which issues must relate directly to a matter specified with particularity as part of subparagraph (2) of this paragraph.
- (c) Arguments and citations of authority may be set forth in a brief accompanying the protest but must be excluded from the protest itself.
- (d) Oppositions to protests and briefs in support thereof shall contain all material, including that pertinent to the determination referred to in paragraph (i) of this section, deemed appropriate to the Commission's resolution of the protest. Such oppositions and supporting briefs must be filed within 10 days after the filing of such protest, and any replies to such oppositions must be filed within 5 days after the filing of the oppositions.
- (e) Protests, oppositions, and replies shall be filed with the Commission in original and 14 copies and shall be accompanied by proof of service upon the grantee or the protestant, as the case may be, and/or their respective attorneys.
- (f) The Commission may upon consideration of a protest direct either the protestant or grantee or both to submit further statements of fact under oath relating to the matters raised in the protest.
- (g) Within 30 days from the date of the filing of the protest, the Commission will enter findings as to whether such

protest meets the requirements set forth in paragraphs (b) (1) and (2) of this section. If the Commission finds that one of these requirements is not met, it will dismiss the protest. If the Commission finds that these requirements are met, it will designate the application in question for hearing. As to issues which the Commission believes present no grounds for setting aside the grant, even if the facts alleged were to be proven, the Commission may designate such issues for oral argument only. The other issues will be designated for evidentiary hearing except that the Commission may redraft the issues in accordance with the facts or substantive matters alleged in the protest and may also specify such additional issues as it deems desirable. In any evidentiary hearing subsequently held upon issues specified by the Commission, upon its own initiative or adopted by it, both the burden of proceeding with the introduction of evidence and the burden of proof shall be upon the grantee. With respect to issues resulting from facts set forth in the protest and not adopted or specified by the Commission on its own motion, both the burden of proceeding with the introduction of evidence and the burden of proof shall be upon the protestant.

- (h) The procedure in such protest hearing shall be governed by the provisions of Subpart B of this part, except as otherwise provided in this section.
- (i) Pending hearing and decision, the effective date of the Commission's action to which protest is made shall be postponed to the effective date of the Commission's decision after hearing, unless the authorization involved is necessary to the maintenance or conduct of an existing service or unless the Commission affirmatively finds that the public interest requires that the grant remain in effect, in which event the Commission shall authorize the applicant to utilize the facilities or authorization in question pending the Commission's decision after hearing.

(Sec. 7, 66 Stat. 715, as amended. See, in particular, sec. 4 (a) and (d), 74 Stat. 889, 892; 47 U.S.C. 309) [28 F.R. 12415, Nov. 22, 1963, as amended at 28 F.R. 14503, Dec. 31, 1963]

Subpart B—Hearing Proceedings

Source: The provisions of this Subpart B appear at 28 F.R. 12425, Nov. 22, 1963, unless otherwise noted.



GENERAL

§ 1.201 Scope.

This subpart shall be applicable to the following cases which have been designated for hearing:

- (a) Adjudication (as defined by the Administrative Procedure Act); and
- (b) Rule making proceedings which are required by law to be made on the record after opportunity for a Commission hearing.

NOTE: For special provisions relating to consideration of standard broadcast applications in the light of the 1950 NARBA and the U.S./Mexican Agreement, see § 1.570.

§ 1.202 Official reporter; transcript.

The Commission will designate from time to time an official reporter for the recording and transcribing of hearing proceedings. The transcript of the testimony taken, or argument had, at any hearing will not be furnished by the Commission, but will be open to inspection under § 0.453(a) (1) of this chapter. Copies of such transcript, if desired, may be obtained from the official reporter upon payment of the charges therefor. (5 U.S.C. 556) [32 F.R. 20861, Dec. 28, 1967]

§ 1.203 The record.

The transcript of testimony and exhibits, together with all papers and requests filed in the proceeding, shall constitute the exclusive record for decision. Where any decision rests on official notice of a material fact not appearing in the record, any party shall on timely request be afforded an opportunity to show the contrary.

(5 U.S.C. 556)

§ 1.204 Pleadings; definition.

As used in this subpart, the term "pleading" means any written notice, motion, petition, request, opposition, reply, brief, proposed findings, exceptions, memorandum of law, or other paper filed with the Commission in a hearing proceeding. It does not include exhibits or documents offered in evidence. See § 1.356.

[29 F.R. 8219, June 30, 1964]

§ 1.205 Continuances and extensions.

Continuances of any proceeding or hearing and extensions of time for making any filing or performing any act required or allowed to be done within a specified time may be granted by the Commission or the presiding officer upon motion for good cause shown, unless the time for performance or filing is limited by statute.

- § 1.207 Interlocutory matters, reconsideration and review; cross references.
- (a) Rules governing interlocutory pleadings in hearing proceedings are set forth in §§ 1.291–1.298.
- (b) Rules governing appeal from, and reconsideration of, interlocutory rulings made by the presiding officer are set forth as §§ 1.301 and 1.303.
- (c) Rules governing the reconsideration and review of actions taken pursuant to delegated authority, and the reconsideration of actions taken by the Commission, are set forth in §§ 1.101–1.120. [28 F.R. 12425, Nov. 22, 1968, as amended at

[28 F.R. 12425, Nov. 22, 1963, as amended at 29 F.R. 6443, May 16, 1964]

§ 1.209 Identification of responsible officer in caption to pleading.

Each pleading filed in a hearing proceeding shall indicate in its caption whether it is to be acted upon by the Commission, the Review Board, the Chief Hearing Examiner, or the presiding officer. If it is to be acted upon by the presiding officer, he shall be identified by name.

[29 F.R. 8219, June 30, 1964]

§ 1.211 Service.

Except as otherwise expressly provided in this chapter, all pleadings filed in a hearing proceeding shall be served upon all other counsel in the proceeding or, if a party is not represented by counsel, then upon such party. All such papers shall be accompanied by proof of service. For provisions governing the manner of service, see § 1.47.

[29 F.R. 8219, June 30, 1964]

Participants and Issues 1

- § 1.221 Notice of hearing; appearances.
- (a) Upon designation of an application for hearing, the Commission issues an order containing the following:
- (1) A statement as to the reasons for the Commission's action.
- (2) A statement as to the matters of fact and law involved, and the issues upon which the application will be heard.
- (3) A statement as to the time, place, and nature of the hearing. (If the time



¹ Centerhead amended, 33 F.R. 463, Jan. 12, 1968.

and place are not specified, the order will indicate that the time and place will be specified at a later date.)

(4) A statement as to the legal authority and jurisdiction under which the

hearing is to be held.

(b) The order designating an application for hearing is mailed to the applicant by the Secretary of the Commission and is published in the FEDERAL REGISTER. Reasonable notice of hearing will be given to the parties in all proceedings; and, whenever possible, the Commission will give at least 60 days

notice of comparative hearings.

(c) In order to avail himself of the opportunity to be heard, the applicant, in person or by his attorney, shall, within 20 days of the mailing of the notice of designation for hearing by the Secretary, file with the Commission, in triplicate, a written appearance stating that he will appear on the date fixed for hearing and present evidence on the issues specified in the order. Where an applicant fails to file such a written appearance within the time specified, or has not filed prior to the expiration of that time a petition to dismiss without prejudice, or a petition to accept, for good cause shown, such written appearance beyond expiration of said 20 days, the application will be dismissed with prejudice for failure to prosecute.

(d) The Commission will on its own motion name as parties to the hearing any person found to be a party in in-

terest.

- (e) In order to avail himself of the opportunity to be heard, any person named as a party pursuant to paragraph (d) of this section shall, within 20 days of the mailing of the notice of his designation as a party, file with the Commission, in person or by attorney, a written appearance in triplicate, stating that he will appear at the hearing. Any person so named who fails to file this written statement within the time specified, shall, unless good cause for such failure is shown, forfeit his hearing rights.
- (5 U.S.C. 554. Sec. 309, 48 Stat. 1085, as amended; 47 U.S.C. 309)

§ 1.223 Petitions to intervene.

(a) Where, in cases involving applications for construction permits and station licenses, or modifications or renewals thereof, the Commission has failed to notify and name as a party to the hearing any person who qualifies as a party in

interest, such person may acquire the status of a party by filing, under oath and not more than 30 days after the publication in the FEDERAL REGISTER of the hearing issues or any substantial amendment thereto, a petition for intervention showing the basis of its interest. Where such person's interest is based upon a claim that a grant of the application would cause objectionable interference under applicable provisions of this chapter to such person as a licensee or permittee of an existing or authorized station, the petition to intervene must be accompanied by an affidavit of a qualified radio engineer which shall show. either by following the procedures prescribed in this chapter for determining interference in the absence of measurements or by actual measurements made in accordance with the methods prescribed in this chapter, the extent of such interference. Where the person's status as a party in interest is established, the petition to intervene will be granted.

(b) Any other person desiring to participate as a party in any hearing may file a petition for leave to intervene not later than 30 days after the publication in the FEDERAL REGISTER of the hearing issues or any substantial amendment thereto. The petition must set forth the interest of petitioner in the proceedings, must show how such petitioner's participation will assist the Commission in the determination of the issues in question, and must be accompanied by the affidavit of a person with knowledge as to the facts set forth in the petition. The presiding officer, in his discretion, may grant or deny such petition or may permit intervention by such persons limited to particular issues or to a particular stage of the proceeding.

(c) The granting of any petition to intervene shall not have the effect of changing or enlarging the issues specified in the Commission's notice of hearing unless the Commission shall on

motion amend the same.

(d) Any person desiring to file a petition for leave to intervene later than 30 days after the publication in the Federal Register of the hearing issues or any substantial amendment thereto shall set forth the interest of petitioner in the proceedings, show how such petitioner's participation will assist the Commission in the determination of the issues in question, and set forth reasons why it was not possible to file a petition within the time prescribed by paragraphs (a)

and (b) of this section. Such petition shall be accompanied by the affidavit of a person with knowledge of the facts set forth in the petition, and where petitioner claims that a grant of the application would cause objectionable interference under applicable provisions of this chapter, the petition for leave to intervene must be accompanied by the affidavit of a qualified radio engineer showing the extent of such alleged interference according to the methods prescribed in paragraph (a) of this section. If, in the opinion of the presiding officer, good cause is shown for the delay in filing, he may in his discretion grant such petition or may permit intervention limited to particular issues or to a particular stage of the proceeding.

(Sec. 309, 48 Stat. 1085, as amended; 47 U.S.C 309) [28 F.R. 12425, Nov. 22, 1963, as amended at 29 F.R. 7821, June 19, 1964]

§ 1.225 Participation by non-parties; consideration of communications.

- (a) Any person who wishes to appear and give evidence on any matter, and who so advises the Secretary, will be notified by the Secretary if that matter is designated for hearing. In the case of requests bearing more than one signature, notice of hearing will be given to the person first signing unless the request indicates that such notice should be sent to someone other than such person.
- (b) No person shall be precluded from giving any relevant, material, and competent testimony at a hearing because he lacks a sufficient interest to justify his intervention as a party in the matter.
- (c) When a hearing is held, no communication will be considered in determining the merits of any matter unless it has been received into evidence. The admissibility of any communication shall be governed by the applicable rules of evidence, and no communication shall be admissible on the basis of a stipulation unless Commission counsel as well as counsel for all of the parties shall join in such stipulation.

§ 1.227 Consolidations.

- (a) The Commission, upon motion or upon its own motion, will, where such action will best conduce to the proper dispatch of business and to the ends of justice, consolidate for hearing:
- Any cases which involve the same applicant or involve substantially the same issues, or

- (2) Any applications which present conflicting claims.
- (b) (1) In broadcast cases, except as provided in subparagraph (5) of this paragraph, no application will be consolidated for hearing with a previously filed application or applications unless such application, or such application as amended, if amended so as to require a new file number, is substantially complete and tendered for filing by whichever date is earlier: (i) The close of business on the day preceding the day the previously filed application or one of the previously filed applications is designated for hearing; or (ii) the close of business on the day preceding the day designated by public notice published in the Federal Register as the day any one of the previously filed applications is available and ready for processing.

Note: Subdivision (ii) of this subparagraph applies only to standard broadcast applications for new stations or for major changes in the facilities of authorized stations. See also §§ 1.571 (c) and (h) and 1.591(a).

- (2) In non-broadcast cases other than common carrier cases, any application that is mutually exclusive with another application or applications already designated for hearing will be consolidated for hearing with such other application or applications only if the later application in question has been filed within 5 days after public notice has been given in the Federal Register of the Commission's order which first designated for hearing the prior application or applications with which such application is in conflict.
- (3) In common carrier cases, except those involving Domestic Public Radio Services, any application that is mutually exclusive with another previously filed application will be considered with such prior filed application only if the later filed application is substantially complete and tendered for filing prior to the close of business on the day preceding the day the earlier filed application is designated for hearing. In the Domestic Public Radio Services no application will be consolidated for hearing as mutually exclusive with a previously filed application or applications unless such application, or such application amended so as to constitute a major change therein as defined in § 21.33 of this chapter, is substantially complete and tendered for filing by whichever date is earlier: (i) The close of business

- 1 business day preceding the day on which the Commission designates the earlier filed application for hearing; or (ii) within 60 days after the date of the public notice listing the first prior filed application (with which subsequent applications are in conflict) as having been accepted for filing. An application which is subsequently amended by a major change will, for the purpose of this section, be considered to be a newly filed application. Where major changes which do not relate to the mutually exclusive aspect of a proceeding are warranted, or in the case of multiple mutually exclusive issues where the warranted major changes serve to resolve one or more of the issues but do not relate to the mutually exclusive aspect of the proceeding, such changes or amendments will not serve to alter the existing mutually exclusive status so long as new conflicts are not created.
- (4) Any mutually exclusive application filed after the date prescribed in subparagraphs (1), (2), or (3) of this paragraph will be dismissed without prejudice and will be eligible for refiling only after a final decision is rendered by the Commission with respect to the prior application or applications or after such application or applications are dismissed or removed from the hearing docket.
- (5) An application which is mutually exclusive with an application for renewal of license of a broadcast station will be designated for comparative hearing with such license renewal application if it is substantially complete and tendered for filing no later than the date prescribed in § 1.516(e).

[28 F.R. 12425, Nov. 22, 1963, as amended at 33 F.R. 9658, July 3, 1968; 34 F.R. 7966, May 21, 1969]

§ 1.229 Motions to enlarge, change, or delete issues.

- (a) A motion to enlarge, change or delete the issues may be filed by any party to a hearing.
- (b) Such motions must be filed with the Commission not later than 15 days after the issues in the hearing have first been published in the FEDERAL REGISTER. Any person desiring to file a motion to enlarge, change, or delete the issues after the expiration of such 15 days must set forth the reason why it was not possible to file the petition within the prescribed 15 days. Unless good cause is shown for delay in filing, the motion will not be granted.

(c) Such motions, oppositions thereto. and replies to oppositions shall contain specific allegations of fact sufficient to support the action requested. Such allegations of fact, except for those of which official notice may be taken, shall be supported by affidavits of a person or having personal knowledge persons thereof.

PRESIDING OFFICER

AUTHORITY: \$\$ 1.241, 1.243, and 1.245 issued under (5 U.S.C. 556)

§ 1.241 Designation of presiding officer.

- (a) Hearings will be conducted by the Commission, by one or more commissioners, or by an examiner designated pursuant to section 11 of the Administrative Procedure Act. If a presiding officer becomes unavailable to the Commission prior to the taking of testimony, another presiding officer will be designated.
- (b) Unless the Commission determines that due and timely execution of its functions requires otherwise, presiding officers shall be designated, and notice thereof released to the public, at least 10 days prior to the date set for hearing.

§ 1.243 Authority of presiding officer.

From the time he is designated to preside until issuance of his decision or the transfer of the proceeding to the Commission or to another presiding officer, the presiding officer shall have such authority as is vested in him by law and by the provisions of this chapter, including authority to:

- (a) Administer oaths and affirmations:
 - (b) Issue subpenas;
 - (c) Examine witnesses;
- (d) Rule upon questions of evidence;(e) Take or cause depositions to be taken;
- (f) Regulate the course of the hearing. maintain decorum, and exclude from the hearing any person engaging in contemptuous conduct or otherwise disrupting the proceedings;

(g) Require the filing of memoranda of law and the presentation of oral argument with respect to any question of law upon which he is required to rule during the course of the hearing:

- (h) Hold conferences for the settlement or simplification of the issues by consent of the parties;
- (i) Dispose of procedural requests or similar matters, as provided for in § 0.341 of this chapter;



(j) Take actions and make decisions or recommend decisions in conformity with the Administrative Procedure Act.

§ 1.245 Disqualification of presiding officer.

(a) In the event that a presiding officer deems himself disqualified and desires to withdraw from the case, he shall notify the Commission of his withdrawal at least 7 days prior to the date set for hearing.

(b) Any party may request the presiding officer to withdraw on the grounds of personal bias or other disqualification.

(1) The person seeking disqualification shall file with the presiding officer an affidavit setting forth in detail the facts alleged to constitute grounds for disqualification. Such affidavit shall be filed not later than 5 days before the commencement of the hearing unless, for good cause shown, additional time is necessary.

(2) The presiding officer may file a response to the affidavit; and if he believes himself not disqualified, shall so rule and proceed with the hearing.

- (3) The person seeking disqualification may except to a ruling of non-disqualification and, in that event, shall do so at the time the ruling is made. Unless exception is taken to the ruling at this time, the right to request withdrawal of the presiding officer shall be deemed waived.
- (4) If exception to the ruling is taken, the presiding officer shall certify the question, together with the affidavit and any response filed in connection therewith, to the Commission. The hearing shall be suspended pending a ruling on the question by the Commission.

(5) The Commission may rule on the question without hearing, or it may require testimony or argument on the is-

sues raised.

(6) The affidavit, response, testimony or argument thereon, and the Commission's decision shall be part of the record in the case.

PREHEARING PROCEDURES

§ 1.246 Admission of facts and genuineness of documents.

(a) Within 20 days after the time for filing a notice of appearance has expired; or within 20 days after the release of an order adding parties to the proceeding (see §§ 1.223 and 1.227) or changing the issues (see § 1.229); or within such shorter or longer time as the presiding

officer may allow on motion or notice, a party may serve upon any other party a written request for the admission by the latter of the genuineness of any relevant documents identified in and exhibited by a clear copy with the request or of the truth of any relevant matters of fact set forth in the request.

(b) Each of the matters of which an admission is requested shall be deemed admitted unless, within a period designated in the request, not less than 10 days after service thereof, or within such shorter or longer time as the presiding officer may allow on motion or notice, the party to whom the request is directed serves upon the party requesting the admission either (1) a sworn statement denying specifically the matters of which an admission is requested or setting forth in detail the reasons why he cannot truthfully admit or deny those matters, or (2) written objections on the ground that some or all of the requested admissions are privileged or irrelevant or that the request is otherwise improper in whole or in part. If written objections to a part of the request are made, the remainder of the request shall be answered within the period designated in the request. A denial shall fairly meet the substance of the requested admission, and when good faith requires that a party deny only a part or a qualification of a matter of which an admission is requested, he shall specify so much of it as is true and deny only the remainder.

(c) A copy of the request and of any answer shall be served by the party filing on all other parties to the proceeding

and upon the presiding officer.

(d) Written objections to the requested admission may be ruled upon by the presiding officer without additional pleadings. Rulings may be appealed only if the written objections are overruled.

[33 F.R. 463, Jan. 12, 1968]

§ 1.248 Prehearing conferences; hearing conferences.

(a) The Commission, on its own initiative or at the request of any party, may direct the parties or their attorneys to appear at a specified time and place for a conference prior to a hearing, or to submit suggestions in writing, for the purpose of considering, among other things, the matters set forth in paragraph (c) of this section.

(b) The presiding officer (or the Commission or a panel of commissioners

in a case over which it presides), on his own initiative or at the request of any party, may direct the parties or their attorneys to appear at a specified time and place for a conference prior to or during the course of a hearing, or to submit suggestions in writing, for the purpose of considering any of the matters set forth in paragraph (c) of this section.

- (c) In conferences held, or in suggestions submitted, pursuant to paragraphs (a) and (b) of this section, the following matters, among others, may be considered:
- (1) The necessity or desirability of simplification, clarification, amplification, or limitation of the issues;
- (2) The admission of facts and of the genuineness of documents (see § 1.246), and the possibility of stipulating with respect to facts;
 - (3) The procedure at the hearing:
- (4) The limitation of the number of witnesses;
- (5) In cases arising under Title II of the Communications Act, the necessity or desirability of amending the pleadings and offers of settlement or proposals of adjustment; and
- (6) In cases involving comparative broadcast applications:
- (i) Narrowing the issues or the areas of inquiry and proof at the hearing;
 - (ii) [Reserved]
- (iii) Reports and letters relating to surveys or contacts;
- (iv) Assumptions regarding the avallability of equipment;
 - (v) Network programming;
- (vi) Assumptions regarding the availability of networks proposed;
 - (vii) Offers of letters in general;
- (viii) The method of handling evidence relating to the past cooperation of existing stations owned and/or operated by the applicants with organizations in the area:
- (ix) Proof of contracts, agreements, or understandings reduced to writing;
 - (x) Stipulations:
 - (xi) Need for depositions;
 - (xii) The numbering of exihibits:
- (xiii) The order or offer of proof with relationship to docket number;
- (xiv) The date for the formal hearing; and
- (xv) Such other matters as may expedite the conduct of the hearing.
- (d) At the prehearing conferences prescribed by this section, the parties in any broadcast proceeding shall be pre-

pared to discuss the advisability of reducing any or all phases of their affirmative direct cases to written form. Where it appears that it will contribute significantly to the disposition of the proceeding for the parties to submit any portion of their cases in writing, it is the policy of the Commission to encourage them to do so. However, the phase or phases of the proceeding to be submitted in writing, the dates for the exchange of the written material, and other procedural limitations upon the effect of adopting the written case procedure (such as. whether material ruled out as incompetent may be restored by competent oral testimony) is to be left to agreement of the parties as approved by the presiding officer.

(e) An official transcript of all conferences shall be made.

[28 F.R. 12425, Nov. 22, 1963, as amended at 33 F.R. 463, Jan. 12, 1968]

§ 1.249 Prehearing statement.

Immediately upon convening the formal hearing in any proceeding, the presiding officer shall enter upon the record a statement reciting all actions taken at the prehearing conferences, and incorporating into the record all of the stipulations and agreements of the parties which are approved by him, and any special rules which he may deem necessary to govern the course of the proceeding.

[28 F.R. 12425, Nov. 22, 1963. Redesignated at 33 F.R. 463, Jan. 12, 1968]

HEARING AND INTERMEDIATE DECISION

§ 1.250 Discovery and preservation of evidence; cross-reference.

For provisions relating to prehearing discovery and preservation of admissible evidence, see §§ 1.311—1.325.

[33 F.R. 463, Jan. 12, 1968]

§ 1.253 Time and place of hearing.

- (a) The Commission will specify the day on which and the place at which any hearing is to commence.
- (b) The presiding officer will specify the days on which subsequent hearing sessions are to be held.
- (c) If the Commission specifies that a hearing is to commence in the District of Columbia, it shall be moved therefrom only by order of the Commission.
- (d) If the Commission specifies that a hearing is to commence at a field location, all appropriate proceedings will be completed at such location before the

hearing is moved therefrom. When such proceedings are completed, the presiding officer may move the hearing from the field location specified to another appropriate field location or to the District of Columbia.

§ 1.254 Nature of the hearing; burden of proof.

Any hearing upon an application shall be a full hearing in which the applicant and all other parties in interest shall be permitted to participate but in which both the burden of proceeding with the introduction of evidence upon any issue specified by the Commission, as well as the burden of proof upon all such issues, shall be upon the applicant except as otherwise provided in the order of desig-

(Sec. 309, 48 Stat. 1085, as amended; 47 U.S.C. 809)

§ 1.255 Order of procedure.

- (a) At hearings on a formal complaint or petition or in a proceeding for any instrument of authorization which empowered Commission is issue, the complainant, petitioner, applicant, as the case may be, shall, unless the Commission otherwise orders, and close. At hearings protests, the protestant opens and closes the proceedings in case the issues are not specifically adopted by the Commission; otherwise the grantee does so. At hearings on orders to show cause. to cease and desist, to revoke or modify a station license under sections 312 and 316 of the Communications Act. or other like proceedings instituted by the Commission, the Commission shall open and close.
- (b) At all hearings under Title II of the Communications Act, other than hearings on formal complaints, petitions, or applications, the respondent shall open and close unless otherwise specified by the Commission.
- (c) In all other cases, the Commission or presiding officer shall designate the order of presentation. Intervenors shall follow the party in whose behalf intervention is made, and in all cases where the intervention is not in support of an original party, the Commission or presiding officer shall designate at what stage such intervenors shall be heard.

[28 F.R. 12425, Nov. 22, 1963, as amended at 33 F.R. 463, Jan. 12, 1968]

§ 1.258 Closing of the hearing.

The record of hearing shall be closed by an announcement to that effect at the hearing by the presiding officer when the taking of testimony has been concluded. In the discretion of the presiding officer. the record may be closed as of a future specified date in order to permit the admission into the record of exhibits to be prepared: Provided, The parties to the proceeding stipulate on the record that they waive the opportunity to crossexamine or present evidence with respect to such exhibits. The record in any hearing which has been adjourned may not be closed by such officer prior to the day on which the hearing is to resume, except upon 10 days' notice to all parties to the proceeding.

§ 1.260 Certification of transcript.

After the close of the hearing, the complete transcript of testimony, together with all exhibits, shall be certified as to identity by the presiding officer and filed in the office of the Secretary of the Commission. Notice of such certification shall be served on all parties to the proceedings.

§ 1.261 Corrections to transcript.

Within 10 days after the date of notice of certification of the transcript, any party to the proceeding may file with the presiding officer a motion requesting the correction of the transcript, which motion shall be accompanied by proof of service thereof upon all other parties to the proceedings. Within 5 days after the filing of such a motion, other parties may file a pleading in support of or in opposition to such motion. Thereafter, the presiding officer shall, by order, specify the corrections to be made in the transcript, and a copy of the order shall be served upon all parties and made a part of the record. The presiding officer, on his own initiative, may specify corrections to be made in the transcript on 5 days' notice.

§ 1.263 Proposed findings and conclusions.

(a) Each party to the proceeding may file proposed findings of fact and conclusions, briefs, or memoranda of law: Provided, however, That the presiding officer may direct any party other than Commission counsel to file proposed findings of fact and conclusions, briefs, or memoranda of law. Such proposed findings of fact, conclusions, briefs, and memoranda of law shall be filed within 20 days after the record is closed, unless additional time is allowed.

- (b) All pleadings and other papers filed pursuant to this section shall be accompanied by proof of service thereof upon all other counsel in the proceeding; if a party is not represented by counsel, proof of service upon such party shall be made.
- (c) In the absence of a showing of good cause therefor, the failure to file proposed findings of fact, conclusions, briefs, or memoranda of law, when directed to do so, may be deemed a waiver of the right to participate further in the proceeding.

(5 U.S.C. 557)

§ 1.264 Contents of findings of fact and conclusions.

Proposed findings of fact shall be set forth in serially numbered paragraphs and shall set out in detail and with particularity all basic evidentiary facts developed on the record (with appropriate citations to the transcript of record or exhibit relied on for each evidentiary fact) supporting the conclusions proposed by the party filing same. Proposed conclusions shall be separately stated. Proposed findings of fact and conclusions submitted by a person other than an applicant may be limited to those issues in connection with the hearing which affect the interests of such person.

(5 U.S.C. 557)

§ 1.267 Initial and recommended decisions.

- (a) Except as provided in § 1.274, the presiding officer shall prepare an initial (or recommended) decision which shall be transmitted to the Secretary of the Commission who shall make it public immediately and file it in the docket of the case.
- (b) Each initial and recommended decision shall contain findings of fact and conclusions, as well as the reasons or basis therefor, upon all the material issues of fact, law, or discretion presented on the record; each initial decision shall also contain the appropriate rule or order, and the sanction, relief or denial thereof; and each recommended decision shall contain recommendations as to what disposition of the case should be made by the Commission. Each initial

decision will show the date upon which it will become effective in accordance with the rules in this part in the absence of exceptions, appeal, or review.

(c) The authority of the presiding officer over the proceedings shall cease when he has filed his initial or recommended decision, or, if it is a case in which he is to file no decision, when he has certified the case to the Commission for decision after specifying corrections to the transcript in accordance with § 1.261.

(Sec. 409, 48 Stat. 1096, as amended; 47 U.S.C. 409. 5 U.S.C. 557)

REVIEW PROCEEDINGS

§ 1.271 Delegation of review function.

The Commission may direct, by order or rule, that its review function in a case or category of cases be performed by a commissioner, a panel of commissioners, or by the Review Board, in which event the commissioner, panel, or board shall exercise the authority and perform the functions which would otherwise have been performed by the Commission under §§ 1.273–1.282.

Note: Parties to any proceeding designated for hearing prior to August 31, 1961, are entitled to file exceptions to initial decisions with the Commission and to oral argument before the Commission en banc.

(Sec. 5, 48 Stat. 1068, as amended; 47 U.S.C. 155)

§ 1.273 Waiver of initial or recommended decision.

At the conclusion of the hearing or within 20 days thereafter, all parties to the proceeding may agree to waive an initial or recommended decision, and may request that the Commission issue a final decision or order in the case. If the Commission has directed that its review function in the case be performed by a commissioner, a panel of commissioners, or by the Review Board, the request shall be directed to the appro-The Commispriate review authority. sion or such review authority may in its discretion grant the request, in whole or in part, if such action will best conduce to the proper dispatch of business and to the ends of justice.

§ 1.274 Certification of the record to the Commission for initial or final decision

(a) Where the presiding officer is available to the Commission, and where the Commission finds upon the record

that due and timely execution of its functions imperatively and unavoidably so requires, the Commission may direct that the record in a pending proceeding be certified to it for initial or final decision. Unless the Commission finds that due and timely execution of its functions imperatively and unavoidably requires that no recommended decision be issued, the presiding officer will prepare and file a recommended decision, which will be released with the Commission's initial or final decision.

(b) Where the presiding officer becomes unavailable to the Commission after the taking of testimony has been concluded, the Commission may direct that the record in a pending proceeding be certified to it for initial or final decision. In that event, the record shall be certified to the Commission by the Chief Hearing Examiner.

(c) (1) Where the presiding officer becomes unavailable to the Commission after the taking of evidence has commenced but before it has been concluded, the Commission may order a rehearing before another presiding officer designated in accordance with § 1.241.

(2) Upon a finding that due and timely execution of its functions imperatively and unavoidably so requires, the Commission may (as an alternative) order that the hearing be continued by another presiding officer designated in accordance with § 1.241 or by the Commission itself. In that event, the officer continuing the hearing shall, upon completion of the hearing, certify the proceeding to the Commission for an initial or final decision. Unless the Commission finds upon the record that due and timely execution of its functions imperatively and unavoidably requires that no recommended decision be issued, the officer continuing the hearing shall prepare and file a recommended decision to be released with the Commission's initial or final decision. If all the parties expressly consent, and if the Commission does not order otherwise, the officer continuing the hearing may prepare an initial decision.

(Sec. 409, 48 Stat. 1096, as amended; 47 U.S.C. 409)

§ 1.276 Appeal and review of initial decision.

(a) Within 30 days after the date on which public release of the full text of an initial decision is made, or such other time as the Commission may specify, any of the parties may appeal to the Commission by filing exceptions to the initial decision; and such decision shall not become effective and shall then be reviewed by the Commission, whether or not such exceptions may thereafter be withdrawn. The time for filing such exceptions may be extended for good cause shown.

(b) The Commission may on its own initiative provide, by order adopted not later than 20 days after the time for filing exceptions expires, that an initial decision shall not become final, and that it shall be further reviewed or considered by the Commission.

(c) In any case in which an initial decision is subject to review in accordance with paragraph (a) or (b) of this section, the Commission may, on its own initiative or upon appropriate requests by a party, take any one or more of the following actions:

(1) Hear oral argument on the exceptions;

(2) Require the filing of briefs;

(3) Prior to or after oral argument or the filing of exceptions or briefs, reopen the record and/or remand the proceedings to the presiding officer to take further testimony or evidence;

(4) Prior to or after oral argument or the filing of exceptions or briefs, remand the proceedings to the presiding officer to make further findings or conclusions; and

(5) Prior to or after oral argument or the filing of exceptions or briefs, issue, or cause to be issued by the presiding officer, a supplemental initial decision.

(d) No initial decision shall become effective before 50 days after public release of the full text thereof is made unless otherwise ordered by the Commission. The timely filing of exceptions, the further review or consideration of an initial decision on the Commission's initiative, or the taking of action by the Commission under paragraph (c) of this section shall stay the effectiveness of the initial decision until the Commission's review thereof has been completed. the effective date of an initial decision falls within any further time allowed for the filing of exceptions, it shall be postponed automatically until 30 days after time for filing exceptions has expired.

(e) If no exceptions are filed, and the Commission has not ordered the review of an initial decision on its initiative, or has not taken action under paragraph

- (c) of this section, the initial decision shall become effective, an appropriate notation to that effect shall be entered in the docket of the case, and a "Public Notice" thereof shall be given by the Commission. The provisions of § 1.108 shall not apply to such public notices.
- (f) When any party fails to file exceptions within the specified time to an initial decision which proposes to deny its application, such party shall be deemed to have no interest in further prosecution of its application, and its application may be dismissed with prejudice for failure to prosecute.

(Sec. 40, 48 Stat. 1096, as amended; 47 U.S.C. 409)

§ 1.277 Exceptions; oral arguments.

- (a) Each exception to an initial decision or to any part of the record or proceeding in any case, including rulings upon motions or objections, shall point out with particularity alleged errors in the decision or ruling and shall contain specific references to the page or pages of the transcript of hearing, exhibit, or order if any on which the exception is based. Any objection not saved by exception filed pursuant to this section is waived. The exceptions should be concise and they will not be accepted if they contain argumentative matters or discussions of law. Lengthy excerpts of testimony, when desired, shall not be contained in the exceptions but shall be set forth in an appendix.
- (b) Within the period of time allowed in § 1.276(a) for the filing of exceptions, any party may file a statement in support of an initial decision in whole or in part, which shall be similar in form to a statement of exceptions.
- (c) Exceptions or supporting statements may be accompanied by a separate brief or memorandum of law in support thereof. Except by special permission, such brief or memorandum of law will not be accepted if it exceeds 50 double spaced typewritten pages in length. Within 10 days, or such other time as the Commission may specify. after the time for filing exceptions has expired, any other party may file a reply brief to which the same limitation in length applies. If exceptions have been filed, any party may request oral argu-ment not later than five days after the time for filing replies to the exceptions has expired. The Commission in its discretion will, by order, grant or deny the request for oral argument. Within five

days after release of the Commission's order designating an initial decision for oral argument, as provided in paragraph (d) of this section, any party who wishes to participate in oral argument shall file written notice of intention to appear and participate in oral argument; and failure to file written notice shall constitute a waiver of the opportunity to participate.

(d) Each order scheduling a case for oral argument will contain the allotment of time for each party for oral argument before the Commission. The Commission will grant, in its discretion, upon good cause shown, an extension of such time upon petition by a party, which petition must be filed within 5 days after issuance of said order for oral argument.

- (e) Within 10 days after a transcript of oral argument has been filed in the office of the Secretary of the Commission, any party who participated in the oral argument may file with the Commission a motion requesting correction of the transcript, which motion shall be accompanied by proof of service thereof upon all other parties who participated in the oral argument. Within 5 days after the filing of such a motion, other parties may file a pleading in support of or in opposition to such motion. Thereafter, the officer who presided at the oral argument shall, by order, specify the corrections to be made in the transcript. and a copy of the order shall be served upon all parties to the proceeding. The officer who presided at the oral argument may, on his own initiative, by order, specify corrections to be made in the transcript on 5 days notice of the proposed corrections to all parties who participated in the oral argument.
- (f) Any commissioner or member of the Review Board who is not present at oral argument and who is otherwise authorized to participate in a final decision may participate in making that decision after reading the transcript of oral argument.

(Sec. 409, 48 Stat. 1096, as amended; 47 U.S.C. 409)

§ 1.279 Limitation of matters to be reviewed.

Upon review of any initial decision, the Commission may, in its discretion, limit the issues to be reviewed to those findings and conclusions to which exceptions have been filed, or to those findings and conclusions specified in the Commission's

order of review issued pursuant to § 1.276(b).

§ 1.282 Final decision of the Commission.

- (a) After opportunity has been afforded for the filing of proposed findings of fact and conclusions, exceptions, supporting statements, briefs, and for the holding of oral argument as provided in this subpart, the Commission will issue a final decision in each case in which an initial decision has not become final.
 - (b) The final decision shall contain:
- (1) Findings of fact and conclusions, as well as the reasons or basis therefor, upon all the material issues of fact, law or discretion presented on the record;
- (2) Ruling on each relevant and material exception filed; and
- (3) The appropriate rule or order and the sanction, relief or denial thereof. (5 U.S.C. 557)

INTERLOCUTORY ACTIONS IN HEARING PROCEEDINGS

§ 1.291 General provisions.

- (a) (1) The Commission acts on petitions to amend, modify, enlarge or delete the issues in hearing proceedings which involve rule making matters exclusively. It also acts on interlocutory pleadings filed in matters or proceedings which are before the Commission.
- (2) The Review Board acts on petitions to amend, modify, enlarge, or delete the issues in cases of adjudication (including mixed adjudicative and rule making proceedings) and upon joint requests for approval of agreements filed pursuant to § 1.525. It also acts on interlocutory pleadings filed in matters on proceedings which are before the Board.
- (3) The Chief Hearing Examiner acts on those interlocutory matters listed in § 0.351 of this chapter.
- (4) All other interlocutory matters in hearing proceedings are acted on by the presiding officer. See §§ 0.218 and 0.341 of this chapter.
- (5) Each interlocutory pleading shall indicate in its caption whether the pleading is to be acted upon by the Commission, the Review Board, the Chief Hearing Examiner, or the presiding officer. If the pleading is to be acted upon by the presiding officer, he shall be identified by name.
- (b) All interlocutory pleadings shall be submitted in accordance with the

- provisions of §§ 1.4, 1.44, 1.47, 1.48, 1.49, and 1.52.
- (c) (1) Procedural rules governing interlocutory pleadings are set forth in §§ 1.292-1.298.
- (2) Rules governing appeal from, and reconsideration of, interlocutory rulings made by the presiding officer are set forth in §§ 1.301 and 1.303.
- (3) Rules governing the review of interlocutory rulings made by the Review Board or the Chief Hearing Examiner are set forth in §§ 1.101, 1.102(b), 1.115, and 1.117. Petitions requesting reconsideration of an interlocutory ruling made by the Commission, the Review Board, or the Chief Hearing Examiner will not be entertained. See, however, § 1.113.
- (d) No initial decision shall become effective under § 1.276(e) until all interlocutory matters pending before the Review Board or the Commission in the proceeding at the time the initial decision is issued have been disposed of and the time allowed for appeal from interlocutory rulings of the presiding officer or the Review Board has expired.

[29 F.R. 6443, May 16, 1964, as amended at 29 F.R. 12773, Sept. 10, 1964]

§ 1.292 Number of copies.

- (a) An original and 14 copies of each interlocutory pleading to be acted upon by the Review Board, the Chief Hearing Examiner, or the presiding officer shall be filed.
- (b) An original and 19 copies of each interlocutory pleading to be acted upon by the Commission shall be filed.

 [29 F.R. 6444, May 16, 1964]

§ 1.294 Oppositions and replies.

- (a) Any party to a hearing may file an opposition to an interlocutory request filed in that proceeding.
- (b) Except as provided in paragraph (c) of this section, oppositions shall be filed within 4 days after the original pleading is filed, and replies to oppositions will not be entertained. See, however, § 1.732.
- (c) Oppositions to pleadings in the following categories shall be filed within 10 days after the pleading is filed. Replies to such oppositions shall be filed within 5 days after the opposition is filed, and shall be limited to matters raised in the opposition.
- (1) Petitions to amend, modify, enlarge, or delete the issues upon which the hearing was ordered.



(2) Petitions to intervene.

(3) Petitions by adverse parties requesting dismissal of an application.

(4) Joint requests for approval of agreements filed pursuant to § 1.525.

(d) Additional pleadings may be filed only if specifically requested or authorized by the person(s) who is to make the ruling.

[29 F.R. 6444, May 16, 1964]

§ 1.296 Service.

No pleading filed pursuant to § 1.292 or § 1.294 will be considered unless it is accompanied by proof of service upon the parties to the proceeding.

§ 1.297 Oral argument.

Oral argument with respect to any contested interlocutory matter will be held when, in the opinion of the person(s) who is to make the ruling, the ends of justice will be best served thereby. Timely notice will be given of the date, time, and place of any such oral argument.

[29 F.R. 6444, May 16, 1964]

§ 1.298 Rulings; time for action.

- (a) Unless it is found that irreparable injury would thereby be caused one of the parties, or that the public interest requires otherwise, or unless all parties have consented to the contrary, consideration of interlocutory requests will be withheld until the time for filing oppositions (and replies, if replies are allowed) has expired. As a matter of discretion, however, requests for continuances and extensions of time, requests for permission to file pleadings in excess of the length prescribed in this chapter, and requests for temporary relief may be ruled upon ex parte without waiting for the filing of responsive pleadings.
- (b) Interlocutory matters will be disposed of by written order, which will be released promptly. The order upon contested matters shall contain a statement of the reasons for the ruling therein, unless such order is self-explanatory or is merely an affirmance of a prior denial in which reasons have been given.
- (c) Matters raised orally on the record of a hearing proceeding may, however, be disposed of orally by the presiding officer.

[28 F.R. 12425, Nov. 22, 1963, as amended at 29 F.R. 6444, May 16, 1964]

APPEAL AND RECONSIDERATION OF PRESID-ING OFFICER'S RULING

§ 1.301 Appeal from the presiding officer's adverse ruling: effective date.

- (a) Any party to a hearing proceeding may file an appeal from an adverse ruling of the presiding officer. If a commissioner or panel of commissioners is presiding, the appeal will be acted upon by the Commission. The Commission also acts on appeals from the rulings of a hearing examiner in proceedings which involve rule making matters exclusively. In all other proceedings in which a hearing examiner is presiding, appeals from his rulings will be acted upon by the Review Board.
- (b) The appeal shall be filed within five days after the order is released or the ruling is made. Oppositions to such appeals shall be filed within five days after the appeal is filed. Replies to oppositions shall be filed within five days after the opposition is filed and shall be limited to matters raised in the opposi-Appeals, oppositions, and replies tion. shall be submitted to the Secretary, Federal Communications Commission, Washington, D.C., 20554, and shall be served upon parties to the proceeding. Appeals and oppositions shall not exceed 15 double spaced typewritten pages. Replies shall not exceed 10 double spaced typewritten pages.
- (c) Appeals, oppositions, and replies shall conform to the requirements of §§ 1.49, 1.51, and 1.52.
- (d) The failure to file an appeal shall not preclude parties adversely affected from filing exceptions to rulings of the presiding officer.
- (e) Unless the presiding officer orders otherwise, interlocutory rulings made by him shall be effective when the order is released or the ruling is made. If an appeal is filed, the Review Board or the Commission, as the case may be, may in its discretion stay the effect of the ruling until its review of the matters at issue has been completed.

NOTE: Unless the ruling complained of is fundamental and affects the conduct of the entire case, appeals should be deferred and raised as exceptions.

[28 F.R. 12425, Nov. 22, 1963, as amended at 29 F.R. 6444, May 16, 1964]

§ 1.303 Reconsideration of presiding officer's adverse ruling.

Any party may, after obtaining the oral consent of the presiding officer, file

. Digitized by Google

a petition requesting the presiding officer to reconsider an oral ruling or a written order. Such petitions will be considered in accordance with the provisions of §§ 1.292-1.298.

THE DISCOVERY AND PRESERVATION OF EVIDENCE

AUTHORITY: §§ 1.311 to 1.325 issued under sec. 409, 48 Stat. 1096; 47 U.S.C. 409.

Source: §§ 1.311 to 1.325 appear at 33 F.R. 463, Jan. 12, 1968, unless otherwise noted.

§ 1.311 General.

Sections 1.311-1.325 provide for taking the deposition of any person (including a party), for interrogatories to parties, and for orders to parties relating to the production of documents and things and for entry upon real property. These procedures may be used for the discovery of relevant facts, for the production and preservation of evidence for use at the hearing, or for both purposes.

- (a) Applicability. For purposes of discovery, these proecdures may be used in any case of adjudication (as defined in the Administrative Procedure which has been designated for hearing. For the preservation of evidence, they may be used in any case which has been designated for hearing and is conducted under the provisions of this subpart (see § 1.201).
- (b) Scope of examination. Persons. and parties may be examined regarding any matter, not privileged, which is relevant to the hearing issues, including the existence, description, nature, custody, condition and location of any books, documents, or other tangible things and the identity and location of persons having knowledge of relevant facts. It is not ground for objection to use of these procedures that the testimony will be inadmissible at the hearing if the testimony sought appears reasonably calculated to lead to the discovery of admissible evidence. The use of these procedures against the Commission is subject to the following additional limitations:
- The informer's privilege shall encompass information which may lead to the disclosure of an informer's identity.
- (2) Commission personnel may not be questioned by deposition for the purposes of discovery except on special order of the Commission, but may be questioned by written interrogatories under § 1.323. Interrogatories shall be served on the ap-

propriate Bureau Chief (see § 1.21(b)). They will be answered and signed by those personnel with knowledge of the facts. The answers will be served by the Secretary of the Commission upon parties to the proceeding.

(3) The inspection of Commission records is governed by the Public Information Act of 1966 and by §§ 0.451-0.461 of this chapter. The production of Commission records for inspection and copying is subject to these provisions and is considered under criteria set forth therein. See, in particular, § 0.461(e). Commission personnel may be questioned by written interrogatories regarding the existence, general nature and description. custody, condition and location of Commission records, but may not be questioned concerning their contents unless the records are available (or are made available) for inspection under §§ 0.451-0.461. See § 0.451(b) (4) of this chapter.

(4) Subject to subparagraphs (1) through (3) of this paragraph, Commission personnel may be questioned generally by written interrogatories regarding the existence, description, nature, custody, condition and location of relevant documents and things and regarding the identity and location of persons having knowledge of relevant facts, and may otherwise only be examined regarding facts of the case as to which they have

direct personal knowledge.

(c) Schedule for use of the procedures. (1) Except as provided in this paragraph or as otherwise ordered by the presiding officer, the use of these procedures shall be completed prior to the initial pre-hearing conference. The presiding officer may at any time order the parties or their attorneys to appear at a conference to consider the proper use of these procedures or the time to be allowed for such

- (2) After the record has been closed. the parties may be permitted to take the testimony of witnesses, by deposition, for use in the event of further hearing proceedings, on motion, and solely to avoid a failure of justice.
- (d) Who shall act. Actions provided for in §§ 1.311-1.325 will, in most cases, be taken by the officer designated to preside at the hearing (see § 1.241). If the proceeding, or a particular matter to which the action relates, is before the Commission, the Review Board, a commissioner or panel of commissioners, or the Chief Hearing Examiner, the action

will be taken by such officer or body. The term "presiding officer", as used in §§ 1.311-1.325 shall be understood to refer to the appropriate officer or body. See §§ 0.341, 0.351, 0.365, and 1.271 of this

chapter.

(e) Stipulations regarding the taking of depositions. If all of the parties so stipulate in writing and if there is no interference to the conduct of the proceeding, depositions may be taken before any person, at any time (subject to the limitation below) or place, upon any notice and in any manner, and when so taken may be used like other depositions. An original and one copy of the stipulation shall be filed with the Secretary of the Commission, and a copy of the stipulation shall be served on the presiding officer, at least 3 days before the scheduled taking of the deposition.

§ 1.313 Protective orders.

The use of the procedures set forth in §§ 1.311-1.325 is subject to control by the presiding officer, who may issue any order consistent with the provisions of those sections which is appropriate and just for the purpose of protecting parties and deponents or of providing for the proper conduct of the proceeding. The order may specify any measures, including the following to assure proper conduct of the proceeding or to protect any party or deponent from annoyance, expense, embarrassment or oppression:

(a) That depositions shall not be taken or that interrogatories shall not

be answered.

(b) That certain matters shall not be inquired into.

- (c) That the scope of the examination or interrogatories shall be limited to certain matters.
- (d) That depositions may be taken only at some designated time or place, or before an officer, other than that stated in the notice.
- (e) That depositions may be taken only by written interrogatories or only upon oral examination.
- (f) That, after being sealed, the deposition shall be opened only by order of the presiding officer.
- § 1.315 Depositions upon oral examination—notice and preliminary procedure.
- (a) Notice. A party to a hearing proceeding desiring to take the deposition of any person upon oral examination shall give a minimum of 21 days notice in

writing to every other party, to the person to be examined, and to the presiding officer. An original and three copies of the notice shall be filed with the Secretary of the Commission. Related pleadings shall be served and filed in the same manner. The notice shall contain the following information:

(1) The name and address of each person to be examined, if known, and if the name is not known, a general description sufficient to identify him or the particular class or group to which he

belongs.

(2) The time and place for taking the deposition of each person to be examined, and the name or descriptive title and address of the officer before whom the deposition is to be taken.

(3) The matters upon which each person will be examined. See § 1.319.

(b) Responsive pleadings. (1) Within 7 days after service of the notice to take depositions, a motion opposing the taking of depositions may be filed by any party to the proceeding or by the person to be examined. See § 1.319(a).

(2) Within 14 days after service of the notice to take depositions, a response to the opposition motion may be filed by

any party to the proceeding.

(3) Additional pleadings should not be

filed and will not be considered.

(4) The computation of time provisions set forth in § 1.4(g) shall not apply to pleadings filed under the provisions of this paragraph.

(c) Protective order. On an opposition motion filed under paragraph (b) of this section, or on his own motion, the presiding officer may issue a protective order. See § 1.313. A protective order issued by the presiding officer on his own motion may be issued at any time prior to the date specified in the notice for the taking of depositions.

(d) Authority to take depositions. (1) If an opposition motion is not filed within 7 days after service of the notice to take depositions, and if the presiding officer does not on his own motion issue a protective order prior to the time specified in the notice for the taking of depositions, the depositions described in the notice may be taken. An order for the taking of depositions is not required.

(2) If an opposition motion is filed, the depositions described in the notice shall not be taken until the presiding officer has acted on that motion. If the presiding officer authorizes the taking of depositions, he may specify a time, place

or officer for taking them different from that specified in the notice to take

depositions.

(3) If the presiding officer issues a protective order, the depositions described in the notice may be taken (if at all) only in accordance with the provisions of that order.

[33 F.R. 10571, July 25, 1968]

§ 1.316 Depositions upon written interrogatories—notice and preliminary procedure.

- (a) Service of interrogatories; notice. A party to the hearing proceeding desiring to take the deposition of any person upon written interrogatories shall serve the interrogatories upon every other party and shall give a minimum of 35 days notice in writing to every other party and to the person to be examined. An original and three copies of the interrogatories and the notice (and of all related pleadings) shall be filed with the Secretary of the Commission. A copy of the interrogatories and the notice (and of all related pleadings) shall be served on the presiding officer. The notice shall contain the following information:
- (1) The name and address of each person to be examined, if known, and if the name is not known, a general description sufficient to identify him or the particular class or group to which he belongs.
- (2) The time and place for taking the deposition of each person to be examined, and the name or descriptive title and address of the officer before whom the deposition is to be taken.
- (3) The matters upon which each person will be examined. See § 1.319.
- (b) Additional interrogatories. Within 7 days after the filing and service of the original interrogatories, any other party to the proceeding may, in the same manner, file and serve additional interrogatories to be asked of the same witness at the same time and place, with notice to the witness of any additional matters upon which he will be examined.
- (c) Cross interrogatories. Within 14 days after the filing and service of the original interrogatories, any party to the proceeding may, in the same manner, file and serve cross interrogatories, which shall be limited to matters raised in the original or in the additional interrogatories.
- (d) Responsive pleadings. (1) Within 21 days after service of the original in-

terrogatories, any party to the proceeding may move to limit or suppress any original, additional or cross interrogatory, and the person to be examined may file a motion opposing the taking of depositions. See § 1.319(a).

(2) Within 28 days after service of the original interrogatories, a response to a motion to limit or suppress any interrogatory or to a motion opposing the taking of depositions may be filed by any party to the proceeding.

(3) Additional pleadings should not be

filed and will not be considered.

- (e) Protective order. On a motion to limit or suppress or an opposition motion filed under paragraph (d) of this section, or on his own motion, the presiding officer may issue a protective order. See § 1.313. A protective order issued by the presiding officer on his own motion may be issued at any time prior to the date specified in the notice for the taking of depositions.
- (f) Authority to take depositions. (1) If an opposition motion is not filed within 21 days after service of the notice to take depositions, and if the presiding officer does not on his own motion issue a protective order prior to the time specified in the notice for the taking of depositions, the depositions described in the notice may be taken. An order for the taking of depositions is not required.
- (2) If an opposition motion is filed, the depositions described in the notice shall not be taken until the presiding officer has acted on that motion. If the presiding officer authorizes the taking of depositions, he may specify a time, place or officer for taking them different from that specified in the notice to take depositions.
- (3) If the presiding officer issues a protective order, the depositions described in the notice may be taken (if at all) only in accordance with the provisions of that order.

Note: The computation of time provisions of $\S 1.4(g)$ shall not apply to interrogatories and pleadings filed under the provisions of this section.

[33 F.R. 10571, July 25, 1968]

§ 1.318 The taking of depositions.

(a) Persons before whom depositions may be taken. Depositions shall be taken before any judge of any court of the United States; any U.S. Commissioner; any clerk of a district court; any chan-

cellor, justice or judge of a supreme or superior court; the mayor or chief magistrate of a city; any judge of a county court, or court of common pleas of any of the United States; any notary public; not being of counsel or attorney to any party, nor interested in the event of the proceeding; or presiding officers, as provided in § 1.243.

(b) Attendance of witnesses. The attendance of witnesses at the taking of depositions may be compelled by the use of subpena as provided in §§ 1.331-340.

(c) Oath; transcript. The officer before whom the deposition is to be taken shall administer an oath or affirmation to the witness and shall personally, or by someone acting under his direction and in his presence, record the testimony of the witness. The testimony shall be taken stenographically and transcribed, unless the parties agree otherwise.

(d) Examination. (1) In the taking of depositions upon oral examination, the parties may proceed with examination and cross-examination of deponents as permitted at the hearing. In lieu of participating in the oral examination, parties served with the notice to take depositions may transmit written interrogatories to the officer designated in the notice, who shall propound them to the witness and record the answers verbatim.

(2) In the taking of depositions upon written interrogatories, the party who served the original interrogatories shall transmit copies of all interrogatories to the officer designated in the notice, who shall propound them to the witness and record the answers verbatim.

(e) Submission of deposition to witness; changes; signing. When the testimony is fully transcribed, the deposition of each witness shall be submitted to him for examination and shall be read to or by him, unless such examination and reading are waiver by the witness and by the parties. Any changes in form or substance which the witness desires to make shall be entered upon the deposition by the officer with a statement of the reasons given by the witness for making them. The deposition shall then be signed by the witness, unless the parties by stipulation waive the signing, or the witness is ill, cannot be found, or refuses to sign. If the deposition is not signed by the witness, the officer shall sign it and state on the record the fact of the waiver, the illness or absence of the witness, or of his refusal to sign, together with the reason (if any) given therefor; and the deposition may then be used as fully as though signed, unless upon a motion to suppress, the presiding officer holds that the reason given for the refusal to sign requires rejection of the deposition in whole or in part.

(f) Certification of deposition and filing by officer; copies. The officer shall certify on the deposition that the witness was duly sworn by him, that the deposition is a true record of the testimony given by the witness, and that said officer is not of counsel or attorney to either of the parties, nor interested in the event of the proceeding or investigation. He shall then securely seal the deposition in an envelope endorsed with the title of the action and marked "Deposition of (here insert name of witness)" and shall promptly send the original and two copies of the deposition and of all exhibits, together with the notice and any interrogatories received by nim, by certifled mail to the Secretary of the Commission.

§ 1.319 Objections to the taking of depositions.

(a) Objections to be made by motion prior to the taking of depositions. If there is objection to the substance of any interrogatory or to examination on any matter clearly covered by the notice to take depositions, the objection shall be made in a motion opposing the taking of depositions or in a motion to limit or suppress the interrogatory as provided in §§ 1.315 (b) and 1.316(d) and shall not be made at the taking of the deposition.

(b) Objections to be made at the taking of depositions. Errors and irregularities occurring at the oral examination in the manner of taking the deposition, in the form of the questions or answers. in the oath or affirmation, or in the conduct of parties, and errors of any kind which might be obviated, removed, or cured if promptly presented, are waived unless reasonable objection thereto is made at the taking of the deposition. If such objection is made, counsel shall, if possible, agree upon the measures required to obviate, remove, or cure such errors. The measures agreed upon shall be taken. If agreement cannot be reached, the objection shall be noted on the deposition by the officer taking it, and the testimony objected to shall be taken subject to the objection.

(c) Additional objections which may be made at the taking of depositions. Objection may be made at the taking of

depositions on the ground of relevancy or privilege, if the notice to take depositions does not clearly indicate that the witness is to be examined on the matters to which the objection relates. See paragraph (a) of this section. Objection may also be made on the ground that the examination is being conducted in such manner as to unreasonably annoy, embarrass, or oppress a deponent or party.

(1) When there is objection to a line of questioning, as permitted by this paragraph, counsel shall, if possible, reach agreement among themselves regarding the proper limits of the examination.

- (2) If counsel cannot agree on the proper limits of the examination the taking of depositions shall continue on matters not objected to and counsel shall, within 24 hours, either jointly or individually, telegraph statements of their positions to the presiding officer, together with the telephone numbers at which they and the officer taking the depositions can be reached, or shall otherwise jointly confer with the presiding officer. If individual statements are submitted, copies shall be provided to all counsel participating in the taking of depositions.
- (3) The presiding officer shall promptly rule upon the question presented or take such other action as may be appropriate under § 1.313, and shall give notice of his ruling, by telephone, to counsel who submitted statements and to the officer taking the depositions. The presiding officer shall thereafter reduce his ruling to writing.
- (4) The taking of depositions shall continue in accordance with the presiding officer's ruling. Such rulings are not subject to appeal.

§ 1.321 Use of depositions at the hearing.

- (a) No inference concerning the admissibility of a deposition in evidence shall be drawn because of favorable action on the notice to take depositions.
- (b) Except as provided in this paragraph and in § 1.319, objection may be made at the hearing to receiving in evidence any deposition or part thereof for any reason which would require the exclusion of the evidence if the witness were then present and testifying.
- (1) Objections to the competency of a witness, or the competency, relevancy or materiality of testimony are waived by failure to make them before or during the taking of depositions if (and only if)

the ground of the objection is one which might have been obviated or removed if presented at that time.

(2) Objection on the ground of privilege is waived by failure to make it before or during the taking of depositions

- (c) A party shall not be deemed to make a person his own witness for any purpose by taking his deposition. The introduction in evidence of the deposition or any part thereof for any purpose other than that of contradicting or impeaching the deponent makes the deponent the witness of the party introducing the deposition, but this shall not apply to the use by an adverse party of a deposition as described in subparagraph (2) of paragraph (d) of this section. At the hearing any party may rebut any relevant evidence contained in a deposition whether introduced by him or by any other party.
- (d) At the hearing (or in a pleading), any part or all of a deposition, so far as admissible, may be used against any party who was present or represented at the taking of the deposition or who had due notice thereof, in accordance with any one of the following provisions:

(1) Any deposition may be used by any party for the purpose of contradicting or impeaching the testimony of deponent

as a witness.

(2) The deposition of a party or of any one who at the time of taking the deposition was an officer, director, or managing agent of a public or private corporation, partnership or association which is a party may be used by an

adverse party for any purpose.

- (3) The deposition of a witness. whether or not a party, may be used by any party for any purpose if the presiding officer finds: (i) That the witness is dead; or (ii) that the witness is out of the United States, unless it appears that the absence of the witness was procured by the party offering the deposition; or (iii) that the witness is unable to attend or testify because of age, sickness, infirmity, or imprisonment; or (iv) upon application and notice, that such exceptional circumstances exist as to make it desirable, in the interest of justice and with due regard to the importance of presenting the testimony of witnesses orally in open hearing, to allow the deposition to be used.
- (4) If only part of a deposition is offered in evidence by a party, an adverse party may require him to introduce all

of it which is relevant to the part introduced, and any party may introduce any

other parts.

(5) Substitution of parties does not affect the right to use depositions previously taken; and, when an action in any hearing has been dismissed and another action involving the same subject matter is afterward brought between the same parties or their representatives or successors in interest, all depositions lawfully taken and duly filed in the former action may be used in the latter as if originally taken therefor.

§ 1.323 Interrogatories to parties.

- (a) Interrogatories. Any party may serve upon any other party written interrogatories to be answered in writing by the party served or, if the party served is a public or private corporation or a partnership or association, by any officer or agent, who shall furnish such information as is available to the party. A copy of the interrogatories shall be served upon all parties to the proceeding. An original and three copies of the interand all rogatories, answers, pleadings shall be filed with the Secretary of the Commission. A copy of the interrogatories, answers and all related pleadings shall be served on the presiding officer.
- (1) Except as otherwise provided in a protective order, the number of interrogatories or sets of interrogatories is not limited.
- (2) Except as provided in such an order, interrogatories may be served after a deposition has been taken, and a deposition may be sought after interrogatories have been answered.
- (b) Answers and objections. Each interrogatory shall be answered separately and fully in writing under oath or affirmation, unless it is objected to, in which event the reasons for objection shall be stated in lieu of an answer. The answers shall be signed by the person making them, and the objections by the attorney making them. The party upon whom the interrogatories were served shall serve a copy of the answers and objections upon all parties to the proceeding within 14 days after service of the interrogatories, or within such shorter or longer period as the presiding officer may allow. Answers may be used in the same manner as depositions of a party (see § 1.321(d)).
- (c) Motion to compel an answer. Any party to the proceeding may, within 7 days, move for an order with respect to

any objection or other failure to answer an interrogatory. For purposes of this paragraph, an evasive or incomplete answer is a failure to answer; and if the motion is based on the assertion that the answer is evasive or incomplete, it shall contain a statement as to the scope and detail of an answer which would be considered responsive and complete. The party upon whom the interrogatories were served may file a response within 7 days after the motion is filed, to which he may append an answer or an amended answer. Additional pleadings should not be submitted and will not be considered.

- (d) Action by the presiding officer. If the presiding officer determines that an objection is not justified, he shall order that the answer be served. If an interrogatory has not been answered, the presiding officer may rule that the right to object has been waived and may order that an answer be served. If an answer does not comply fully with the require-ments of this section, the presiding officer may order that an amended answer be served, may specify the scope and detail of the matters to be covered by the amended answer, and may specify any appropriate procedural consequences (including adverse findings of fact and dismissal with prejudice) which will follow from the failure to make a full and responsive answer. If a full and responsive answer is not made, the presiding officer may issue an order invoking any of the procedural consequences specified in the order to compel an answer.
- (e) Appeal. An order to compel an answer is not subject to appeal. An order invoking adverse procedural consequences may be appealed to the Review Board under § 1.301.

[33 F.R. 10572, July 25, 1968]

§ 1.325 Discovery and production of documents and things for inspection, copying, or photographing.

(a) Upon motion of any party showing good cause therefor and upon notice to all other parties, and subject to the provisions of § 1.313, the presiding officer may (1) order any party except the Commission (see paragraph (b) of this section) to produce and permit the inspection and copying or photographing, by or on behalf of the moving party, of any designated documents, papers, books, accounts, letters, photographs, objects, or tangible things, not privileged, which constitute or contain evidence relating

to any of the matters within the scope of the examination permitted by § 1.311(b) and which are in his possession, custody, or control; or (2) order any party to permit entry upon designated land or other property in his possession or control for the purpose of inspecting, measuring, surveying, or photographing the property or any designated object or operation thereon within the scope of the examination permitted by § 1.311(b). The order shall specify the time, place, and manner of making the inspection and taking the copies and photographs and may prescribe such terms and conditions as are just.

(b) Any party seeking the production of Commission records which are not routinely available for inspection (see §§ 0.456, 0.457, and 0.459 of this chapter) may file a request for inspection under § 0.461 of this chapter. See, in particular, § 0.461(e).

SUBPENAS

AUTHORITY: §§ 1.331 to 1.340 issued under sec. 409, 48 Stat. 1096; 47 U.S.C. 409.

§ 1.331 Who may sign and issue.

Subpenas requiring the attendance and testimony of witnesses, and subpenas requiring the production of any books, papers, schedules of charges, contracts, agreements, and documents relating to any matter under investigation or hearing, may be signed and issued as follows:

- (a) Hearings before the Commission en banc, an individual commissioner, or a panel of commissioners: By any commissioner participating in the conduct of the hearing.
- (b) Hearings before a hearing examiner: By the hearing examiner or, in his absence, by the Chief Hearing Examiner.

§ 1.333 Requests for issuance of subpena.

- (a) Unless submitted on the record while a hearing is in progress, requests for a subpena ad testificandum shall be submitted in writing.
- (b) Requests for a subpena duces tecum shall be submitted in writing, duly subscribed and verified, and shall specify with particularity the books, papers, and documents desired and the facts expected to be proved thereby.
- (c) All requests for subpenas shall be supported by a showing of the general relevance and materiality of the evidence sought.

- (d) Requests for subpenas shall be submitted in triplicate, but need not be served on the parties to the proceeding.
- (e) Requests for issuance of a subpena ad testificandum to enforce a notice to take depositions shall be submitted in writing. Such requests may be submitted with the notice or at a later date. The request shall not be granted until the period for the filing of motions opposing the taking of depositions has expired or, if a motion has been filed, until that motion has been acted on. Regardless of the time when the subpena request is submitted, it need not be accompanied by a showing that relevant and material evidence will be adduced, but merely that the person will be examined regarding a nonprivileged matter which is relevant to the hearing issues. The subpena request may ask that a subpena duces tecum be contemporaneously commanding the person to whom it is directed to produce designated books. papers, documents, or tangible things which constitute or contain evidence relating to any of the matters within the scope of the examination permitted by § 1.311(b) but in that event the subpena request will be subject to the provisions of § 1.313 and paragraph (b) of this section.
- (f) Requests for issuance of a subpena duces tecum to enforce an order for the production of documents and things for inspection and copying under § 1.325 may be submitted with the motion requesting the issuance of such an order. Regardless of the time when the subpena request is submitted, it need not be accompanied by a showing that relevant and material evidence will be adduced, but merely that the documents and things to be examined contain non-privileged matter which is relevant to the subject matter of the proceeding.

[28 F.R. 12425, Nov. 22, 1963, as amended at 33 F.R. 466, Jan. 12, 1968]

§ 1.334 Motions to quash.

Any person against whom a subpena is directed may file a motion to quash or limit the subpena, setting forth the reasons why the subpena should not be complied with or why it should be limited in scope.

§ 1.335 Rulings.

Prompt notice, including a brief statement of the reasons therefor, will be given of the denial, in whole or in part, of a request for subpena or of a motion to quash.

§ 1.336 Service of subpenss.

- (a) A subpena may be served by a United States marshal or his deputy, by Commission personnel, or by any person who is not a party to the proceeding and is not less than 18 years of
- (b) Service of a subpena upon the person named therein shall be made by exhibiting the original subpena to him, by reading the original subpena to him if he is unable to read, by delivering the duplicate subpena to him, and by tendering to him the fees for one day's attendance at the proceeding to which he is summoned and the mileage allowed by law. If the subpena is issued on behalf of the United States or an officer or agency thereof, attendance fees and mileage need not be tendered.

§ 1.337 Return of service.

- (a) If service of the subpena is made by a person other than a United States marshal or his deputy such person shall make affidavit thereof, stating the date, time, and manner of service.
- (b) In case of failure to make service, the reasons for the failure shall be stated on the original subpena by the person who attempted to make service.
- (c) The original subpena, bearing or accompanied by the required return affidavit or statement, shall be returned forthwith to the Secretary of the Commission or, if so directed on the subpena, to the official before whom the person named in the subpena is required to appear.

§ 1.338 Subpena forms.

- (a) Subpena forms, marked "Original", "Duplicate", and "Triplicate", and bearing the Commission's seal, may be obtained from the Commission's Dockets Division. These forms are to be completed and submitted with any request for issuance of a subpena.
- (b) If the request for issuance of a subpena is granted, the "Original" and "Duplicate" copies of the subpena are returned to the person who submitted the request. The "Triplicate" copy is retained for the Commission's files.
- (c) The "Original" copy of the subpena includes a form for proof of service. This form is to be executed by the person who effects service and returned by him to the Secretary of the Commis-

sion or, if so directed on the subpena, to the official before whom the person named in the subpena is required to appear.

(d) The "Duplicate" copy of the subpena shall be served upon the person named therein and retained by him. This copy should be presented in support of any claim for witness fees or mileage allowances for testimony on behalf of the Commission.

§ 1.339 Witness fees.

Witnesses who are subpensed and respond thereto are entitled to the same fees, including mileage, as are paid for like service in the courts of the United States. Fees shall be paid by the party at whose instance the testimony is taken.

§ 1.340 Attendance of witness; disobedience.

The attendance of witnesses and the production of documentary evidence may be required from any place in the United States at any designated place of hearing. In case of disobedience to a subpena, the Commission or any party to a proceeding before the Commission may invoke the aid of any court of the United States in requiring the attendance and testimony of witnesses and the production of documentary evidence.

EVIDENCE

§ 1.351 Rules of evidence.

Except as otherwise provided in this subpart, the rules of evidence governing civil proceedings in matters not involving trial by jury in the courts of the United States shall govern formal hearings. Such rules may be relaxed if the ends of justice will be better served by so doing.

§ 1.352 Cumulative evidence.

The introduction of cumulative evidence shall be avoided, and the number of witnesses that may be heard in behalf of a party on any issue may be limited.

§ 1.353 Further evidence during hearing.

At any stage of a hearing, the presiding officer may call for further evidence upon any issue and may require such evidence to be submitted by any party to the proceeding.

§ 1.354 Documents containing matter not material.

If material and relevant matter offered in evidence is embraced in a document

containing other matter not material or relevant, and not intended to be put in evidence, such document will not be received, but the party offering the same shall present to other counsel, and to the presiding officer, the original document, together with true copies of such material and relevant matter taken therefrom, as it is desired to introduce. Upon presentation of such matter, material and relevant, in proper form, it may be received in evidence, and become a part of the record. Other counsel will be afforded an opportunity to introduce in evidence, in like manner, other portions of such document if found to be material and relevant.

§ 1.355 Documents in foreign language.

Every document, exhibit, or other paper written in a language other than English, which shall be filed in any proceeding, or in response to any order, shall be filed in the language in which it is written together with an English translation thereof duly verified under oath to be a true translation. Each copy of every such document, exhibit, or other paper filed shall be accompanied by a separate copy of the translation.

§ 1.356 Copies of exhibits.

No document or exhibit, or part thereof, shall be received as, or admitted in, evidence unless offered in duplicate. In addition, when exhibits of a documentary character are to be offered in evidence, copies shall be furnished to other counsel unless the presiding officer otherwise directs.

§ 1.357 Mechanical reproductions as evidence.

Unless offered for the sole purpose of attempting to prove or demonstrate sound effect, mechanical or physical reproductions of sound waves shall not be admitted in evidence. Any party desiring to offer any matter alleged to be contained therein or thereupon shall have such matter typewritten on paper of the size prescribed by § 1.49, and the same shall be identified and offered in duplicate in the same manner as other exhibits.

§ 1.358 Tariffs as evidence.

In case any matter contained in a tariff schedule on file with the Commission is offered in evidence, such tariff schedule need not be produced or marked for identification, but the matter so offered shall be specified with particularity (tariff and page number) in such manner as to be readily identified, and may be received in evidence by reference subject to check with the original tariff schedules on file.

§ 1.359 Proof of official record; authentication of copy.

An official record or entry therein, when admissible for any purpose, may be evidenced by an official publication thereof or by a copy attested by the officer having legal custody of the record, or by his deputy, and accompanied with a certificate that such officer has the custody. If the office in which the record is kept is within the United States or within a territory or insular possession subject to the dominion of the United States, the certificate may be made by the judge of a court of record of the district or political subdivision in which the record is kept, authenticated by the seal of the court, or may be made by any public officer having a seal of office having official duties in the district or political subdivision in which the record is kept, authenticated by the seal of his office. If the office in which the record is kept is in a foreign state or country, the certificate may be made by a secretary of embassy or legation, consul general, consul, vice consul, or consular agent, or by any officer in the foreign service of the United States stationed in the foreign state or country in which the record is kept, and authenticated by the seal of his office.

§ 1.360 Proof of lack of record.

The absence of an official record or entry of a specified tenor in an official record may be evidenced by a written statement signed by an officer, or by his deputy, who would have custody of the official record, if it existed, that after diligent search no record or entry of a specified tenor is found to exist in the records of his office, accompanied by a certificate as provided in § 1.359. Such statement and certificate are admissible as evidence that the records of his office contain no such record or entry.

§ 1.361 Other proof of official record.

Sections 1.359 and 1.360 do not prevent the proof of official records or of entry or lack of entry therein by any method authorized by any applicable statute or by the rules of evidence at common law.

§ 1.362 Production of statements.

After a witness is called and has given direct testimony in a hearing, and before he is excused, any party may move for the production of any statement of such witness, or part thereof, pertaining to his direct testimony, in possession of the party calling the witness, if such statement has ben reduced to writing and signed or otherwise approved or adopted by the witness. Such motion shall be directed to the presiding officer. If the party declines to furnish the statement, the testimony of the witness pertaining to the requested statement shall be stricken.

[33 F.R. 466, Jan. 12, 1968]

Subpart C-Rule Making Proceedings

AUTHORITY: The provisions of this Subpart C also issued under 5 U.S.C. 553)

SOURCE: The provisions of this Subpart C appear at 28 F.R. 12432, Nov. 22, 1963, unless otherwise noted.

PETITIONS AND RELATED PLEADINGS

§ 1.401 Petitions for rule making.

(a) Any interested person may petition for the issuance, amendment or repeal of a rule or regulation.

(b) The petition for rule making shall conform to the requirements of §§ 1.49, 1.52, and 1.419(b), and should be submitted or addressed to the Secretary, Federal Communications Commission, Washington, D.C., 20554.

(c) The petition shall set forth the text or substance of the proposed rule, amendment, or rule to be repealed, together with all facts, views, arguments and data deemed to support the action requested, and shall indicate how the interests of petitioner will be affected.

[28 F.R. 12432, Nov. 22, 1963, as amended at 28 F.R. 14503, Dec. 31, 1963]

§ 1.403 Notice and availability.

All petitions for rule making meeting the requirements of § 1.401 will be given a file number, and promptly thereafter, a "Public Notice" will be given (by means of a Commission release entitled "Petitions for Rule Making Filed") as to the petitioner, file number, nature of the proposal and date of filing. Petitions are available for public inspection at the Commission's Docket Reference Room in Washington, D.C.

§ 1.405 Responses to petitions; replies.

(a) Any interested person may file a statement in support of or in opposition

to a petition for rule making prior to Commission action on the petition but not later than 30 days after "Public Notice", as provided for in § 1.403, is given of the filing of such a petition. Such a statement shall be accompanied by proof of service upon the petitioner on or prior to the date of filing in conformity with § 1.47 and shall conform in other aspects with the requirements of §§ 1.49, 1.52, and 1.419(b).

(b) Any interested person may file a reply to statements in support of or in opposition to a petition for rule making prior to Commission action on the petition but not later than 15 days after the filing of such a statement. Such a reply shall be accompanied by proof of service upon the party or parties filing the statement or statements to which the reply is directed on or prior to the date of filing in conformity with § 1.47 and shall conform in other aspects with the requirements of §§ 1.49, 1.52, and 1.419(b).

(c) No additional pleadings may be filed unless specifically requested by the Commission or authorized by it.

[28 F.R. 12413, Nov. 22, 1963, as amended at 28 F.R. 14503, Dec. 31, 1963]

§ 1.407 Action on petitions.

If the Commission determines that the petition discloses sufficient reasons in support of the action requested to justify the institution of a rule making proceeding, and notice and public procedure thereon are required or deemed desirable by the Commission, an appropriate notice of proposed rule making will be issued. In those cases where notice and public procedure thereon are not required, the Commission may issue a final order amending the rules. other cases the petition for rule making will be denied and the petitioner will be notified of the Commission's action with the grounds therefor.

RULE MAKING PROCEEDINGS

§ 1.411 Commencement of rule making proceedings.

Rule making proceedings are commenced by the Commission, either on its own motion or on the basis of a petition for rule making. See §§ 1.401–1.407.

§ 1.412 Notice of proposed rule making.

(a) Except as provided in paragraphs (b) and (c) of this section, prior notice of proposed rule making will be given.

(1) Notice is ordinarily given by publication of a "Notice of Proposed Rule

Making" in the Federal Register. The text of the Notice is in that event also released by the Commission, and is available to interested persons in the Office of Information.

(2) If all persons subject to the proposed rules are named, the proposal may (in lieu of publication) be personally

served upon those persons.

(3) If all persons subject to the proposed rules are named and have actual notice of the proposal as a matter of law, further prior notice of proposed rule making is not required.

(b) Rule changes (including adoption, amendment, or repeal of a rule or rules) relating to the following matters will ordinarily be adopted without prior notice:

 Any military, naval, or foreign affairs function of the United States.

- (2) Any matter relating to Commission management or personnel or to public property, loans, grants, benefits, or contracts.
 - (3) Interpretative rules.

(4) General statements of policy.

(5) Rules of Commission organization,

procedure, or practice.

- (c) Rule changes may in addition be adopted without prior notice in any situation in which the Commission for good cause finds that notice and public procedure are impracticable, unnecessary, or contrary to the public interest. The finding of good cause and a statement of the basis for that finding are in such situations published with the rule changes.
- (d) In addition to the notice provisions of paragraph (a) of this section, the Commission, before prescribing any requirements as to accounts, records, or memoranda to be kept by carriers, will notify the appropriate State agencies having jurisdiction over any carrier involved of the proposed requirements.

[28 F.R. 12432, Nov. 22, 1963, as amended at 32 F.R. 20861, Dec. 28, 1967]

§ 1.413 Content of notice.

A notice of the proposed issuance, amendment, or repeal of a rule will include the following:

- (a) A statement of the time, nature and place of any public rule making proceeding to be held.
- (b) Reference to the authority under which the issuance, amendment or repeal of a rule is proposed.
- (c) Either the terms or substance of the proposed rule or a description of the subjects and issues involved.

- (d) The docket number assigned to the proceeding.
- (e) A statement of the time for filing comments and replies thereto.

§ 1.415 Comments and replies.

- (a) After notice of proposed rule making is issued, the Commission will afford interested persons an opportunity to participate in the rule making proceeding through submission of written data, views, or arguments, with or without opportunity to present the same orally in any manner.
- (b) A reasonable time will be provided for submission of comments in support of or in opposition to proposed rules, and the time provided will be specified in the notice of proposed rule making.
- (c) A reasonable time will be provided for filing comments in reply to the original comments, and the time provided will be specified in the notice of proposed rule making.
- (d) No additional comments may be filed unless specifically requested or authorized by the Commission.

§ 1.417 Statutory requirement for hearing.

When rules are required by law to be made on the record after opportunity for a Commission hearing, the requirements of sections 7 and 8 of the Administrative Procedure Act and applicable provisions of Subparts A and B of this part will govern in place of §§ 1.415 and 1.419.

§ 1.419 Form of comments and replies; number of copies.

- (a) Comments, replies, and other documents filed in a rule making proceeding shall conform to the requirements of § 1.49.
- (b) An original and 14 copies of all comments, replies, pleadings, briefs, and other documents filed in a rule making proceeding shall be furnished the Commission.
- (c) Any person desiring to file identical documents in more than one docketed rule making proceeding shall furnish the Commission two additional copies of any such document for each additional docket. This requirement does not apply if the proceedings have been consolidated.

§ 1.421 Further notice of rule making.

In any rule making proceeding where the Commission deems it warranted, a further notice of proposed rule making will be issued with opportunity for parties of record and other interested persons to submit comments in conformity with §§ 1.415 and 1.419.

§ 1.423 Oral argument and other proceedings.

In any rule making proceeding where the Commission determines that an oral argument, hearing or any other type of proceeding is warranted, notice of the time, place and nature of such proceeding will be published in the Federal Register and will be mailed to all parties to the proceeding.

§ 1.425 Commission action.

The Commission will consider all relevant comments and material of record before taking final action in a rule making proceeding and will issue a decision incorporating its finding and a brief statement of the reasons therefor.

§ 1.427 Effective date of rules.

- (a) Any rule issued by the Commission will be made effective not less than 30 days from the time it is published in the Federal Register except as otherwise specified in paragraphs (b) and (c) of this section.
- (b) For good cause found and published with the rule, any rule issued by the Commission may be made effective within less than 30 days from the time it is published in the FEDERAL REGISTER. Rules involving any military, naval or foreign affairs function of the United States; matters relating to agency management or personnel, public property, loans, grants, benefits or contracts; rules granting or recognizing exemption or relieving restriction; rules of organization, procedure or practice; or interpretative rules: and statements of policy may be made effective without regard to the 30-day requirement.
- (c) In cases of alterations by the Commission in the required manner or form of keeping accounts by carriers, notice will be served upon affected carriers not less than 6 months prior to the effective date of such alterations.

Subpart D—Broadcast Applications and Proceedings

SOURCE: The provisions of this Subpart D appear at 28 F.R. 12434, Nov. 22, 1963, unless otherwise noted.

GENERAL

§ 1.501 Scope.

This subpart is applicable to all broadcast services listed in Parts 73 and 74 of this chapter. For additional information relative to applications, see the respective rules relating to each service.

§ 1.502 Emergency Broadcast System Authorizations.

- (a) The Commission issues National Defense Emergency Authorizations (NDEA's) which require stations to operate in a manner consistent with the needs of national security and defense.
- (b) Standard broadcast NDEA's permit such stations to operate, in accordance with the terms of their licenses, during an Emergency Action Condition and during the experimental period of the stations concerned, as provided by § 73.10 of this chapter: Provided, That no interference shall be caused to other stations maintaining a regular operating schedule within the experimental period unless the licensees of such other stations have previously consented thereto.
 - (c) [Reserved]
- (d) Remote pickup broadcast station NDEA's permit such stations to operate in a prescribed manner, on their licensed frequency and with normal power, in the State and local remote pickup broadcast intercommunication networks for intercommunication, cue and control, and program purposes during or after an emergency.
- (e) Other NDEA's which may be issued will be on such terms as may be designated therein.
- (f) All NDEA's are issued for periods of time covered by the station license of the station concerned, subject, however, to being changed or canceled at an earlier date in the discretion of the Commission without the necessity of a hearing.
- (g) Unless canceled, National Defense Emergency renewal authorizations will be issued together with the station's renewal license.

[28 F.R. 12434, Nov. 22, 1963, as amended at 33 F.R. 7491, May 21, 1968]

GENERAL FILING REQUIREMENTS

§ 1.511 Applications required.

(a) Except as provided in paragraph (b) of this section, construction permits as defined in section 3(dd) of the Communications Act of 1934, as amended; station licenses as defined in section 3(bb) of the Communications Act; modifications of construction permits or licenses; renewals of licenses; transfers; and assignments of construction permits

or station licenses, or any rights thereunder, shall be granted only upon written and subscribed application. A separate application shall be filed for each instrument of authorization requested, except as may otherwise be provided in this part.

(b) In cases of (1) emergency found by the Commission involving danger to life or property or due to damage to equipment, or (2) during a national emergency proclaimed by the President or declared by the Congress and during the continuance of any war in which the United States is engaged, and when such action is necessary for the national defense or security or otherwise in furtherance of the war effort, the Commission may grant construction permits and station licenses, or modifications or renewals thereof, without the filing of a formal application; but no authorization so granted shall continue in effect beyond the period of the emergency or war requiring it.

(c) Each individual request submitted under the provisions of paragraph (b) of this section shall contain, as a minimum requirement, the following in-

formation:

(1) Name and address of applicant.

(2) Location of proposed installation or operation.

(3) Official call letters of any valid station authorization already held by applicant and the station location.

(4) Type of service desired (not required for renewal, nor for modification unless class of station is to be modified).

(5) Frequency assignment, authorized transmitter power(s), and authorized class(es) of emission desired (not required for renewal; required for modification only to the extent such information may be involved).

(6) Equipment to be used, specifying the manufacturer and type or model number (not required for renewal; required for modification only to the extent such information may be involved).

- (7) Statements to the extent necessary for the Commission to determine whether or not the granting of the desired authorization will be in accordance with the citizenship eligibility requirements of section 310 of the Communications Act.
- (8) Statement of facts which, in the opinion of the applicant, constitute an emergency to be found by the Commission for the purpose of this section, including the estimated duration of the

emergency; or which, if during an emergency or war declared by the President or Congress, necessitate such action, without formal application, for the national defense or security or in furtherance of the war effort.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

§ 1.512 Where to file; number of copies.

All applications for authorizations required by § 1.511 shall be filed at the Commission's main office in Washington, D.C. The number of copies required for each application is set forth in the FCC Form which is to be used in filing such application.

[29 F.R. 12371, Aug. 28, 1964]

§ 1.513 Who may sign applications.

(a) Except as provided in § 1.511(b) or in paragraph (b) of this section, applications, amendments thereto, and related statements of fact required by the Commission shall be personally signed by the applicant, if the applicant is an individual: by one of the partners, if the applicant is a partnership; by an officer. if the applicant is a corporation: or by a member who is an officer, if the applicant is an unincorporated association. Applications, amendments, and related statements of fact filed on behalf of eligible government entities, such as states and territories of the United States and political subdivisions thereof. the District of Columbia, and units of local government, including incorporated municipalities, shall be signed by such duly elected or appointed officials as may be competent to do so under the laws of the applicable jurisdiction.

(b) Applications, amendments thereto, and related statements of fact required by the Commission may be signed by the applicant's attorney in case of the applicant's physical disability or of his absence from the United States. The attorney shall in that event separately set forth the reason why the application is not signed by the applicant. In addition, if any matter is stated on the basis of the attorney's belief only (rather than his knowledge), he shall separately set forth his reasons for believing that such

statements are true.

(c) Only the original of applications, amendments, or related statements of fact need be signed; copies may be conformed.

(d) Applications, amendments, and related statements of fact need not be submitted under oath. Willful false statements made therein, however, are punishable by fine and imprisonment, U.S. Code, Title 18, section 1001, and by appropriate administrative sanctions, including revocation of station license pursuant to section 312(a)(1) of the Communications Act of 1934, as amended. (Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

§ 1.514 Content of applications.

- (a) Each application shall include all information called for by the particular form on which the application is required to be filed, unless the information called for is inapplicable, in which case this fact shall be indicated.
- (b) The Commission may require an applicant to submit such documents and written statements of fact as in its judgment may be necessary. The Commission may also, upon its own motion or upon motion of any party to a proceeding, order the applicant to amend his application so as to make the same more definite and certain.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

§ 1.516 Specification of facilities.

- (a) An application for facilities in the standard, FM, or television broadcast services shall be limited to one frequency, or channel assignment, and no application will be accepted for filing if it requests alternate frequency or channel assignments.
- (b) An application for facilities in the experimental and auxiliary broadcast services may request the assignment of more than one frequency if consistent with applicable rules in Part 74 of this chapter. Such applications must specify the frequency or frequencies requested and may not request alternate frequencies.
- (c) An application for construction permit for a new broadcast station, the facilities for which are specified in an outstanding construction permit, will not be accepted for filing.
- (d) An application for facilities in the international broadcast service may be filed without a request for specific frequency, as the Commission will assign frequencies from time to time in accordance with §§ 73.702 and 73.711 of this chapter.

(e) (1) Except as provided in subparagraph (2) of this paragraph an application for a construction permit for a new broadcast station or for modification of construction permit or license of a previously authorized broadcast station will not be accepted for filing if it is mutually exclusive with an application for renewal of license of an existing broadcast station unless it is tendered for filing by the end of the first day of the last full calendar month of the expiring license term: Provided, That if the license renewal application is not timely filed as prescribed in § 1.539(a), the deadline for filing applications mutually exclusive therewith is the 60th day after the Commission gives public notice that it has accepted the late-filed renewal application for filing: And provided, further, That if any deadline prescribed in this subparagraph falls on a nonbusiness day. the cutoff shall be the close of business of the first full busines day thereafter.

NOTE: The dates when the licenses of standard, FM, noncommercial FM, television and TV translator broadcast stations regularly expire are listed in §§ 73.34, 73.218, 73.518, 73.630, and 74.15, respectively, of this chapter.

(2) Applications mutually exclusive with a pending application for renewal of license of a broadcast station for a new term commencing on or before June 1, 1969, will not be accepted for filing after June 25, 1969.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308) [28 F.R. 12434, Nov. 22, 1963, as amended at 34 F.R. 7966, May 21, 1969]

§ 1.518 Inconsistent or conflicting applications.

While an application is pending and undecided, no subsequent inconsistent or conflicting application may be filed by the same applicant, his successor or assignee, or on behalf or for the benefit of the same applicant, his successor or assignee.

§ 1.519 Repetitious applications.

(a) Where the Commission has denied an application for a new station or for any modification of services or facilities, or dismissed such application with prejudice, no like application involving service of the same kind to substantially the same area by substantially the same applicant, or his successor or assignee, or on behalf or for the benefit of the original parties in interest, may be filed within 12 months from the effective date

of the Commission's action: Provided, however, That applicants whose applications have been denied in a comparative hearing for a particular television facility allocated in the television allocation table may immediately reapply for another available television channel.

(b) Where an appeal has been taken from the action of the Commission in denying a particular application, another application for the same class of broadcast station and for the same area, in whole or in part, filed by the same applicant, or his successor or assignee, or on behalf or for the benefit of the original parties in interest, will not be considered until final disposition of such appeal.

§ 1.520 Multiple applications.

Where there is one application for new or additional facilities pending, no other application for new or additional facilities, for a station of the same class to serve the same community, may be filed by the same applicant, or his successor or assignee, or on behalf or for the benefit of the original parties in interest. Multiple applications may not be filed simultaneously.

§ 1.522 Amendment of applications.

(a) Subject to the provisions of §§ 1.525 and 1.580, any application may be amended as a matter of right prior to the adoption date of an order designating such application for hearing, merely by filing the appropriate number of copies of the amendments in question duly executed in accordance with § 1.513. If a petition to deny (or to designate for hearing) has been filed, the amendment shall be served on the petitioner. See § 1.571(j) for the effect of certain amendments to standard broadcast applications.

(b) Requests to amend an application after it has been designated for hearing will be considered only upon written petition properly served upon the parties of record in accordance with § 1.47 and, where applicable, compliance with the provisions of § 1.525, and will be granted only for good cause shown. In the case of requests to amend the engineering proposal in standard broadcast applications (other than to make changes with respect to the type of equipment specified), good cause will be considered to have been shown only if, in addition to the usual good cause considerations, it is demonstrated that (1) the amendment is

necessitated by events which the applicant could not reasonably have foreseen (e.g., notification of a new foreign station or loss of transmitter site by condemnation); (2) the amendment could not reasonably have been made prior to designation for hearing; and (3) the amendment does not require an enlargement of issues or the addition of new parties to the proceeding.

(c) Notwithstanding the provisions of paragraph (b) of this section, and subject to compliance with the provisions of § 1.525, a petition for leave to amend may be granted provided it is requested that the application as amended be removed from the hearing docket and returned to the processing line. See § 1.571(i).

[28 F.R. 12434, Nov. 22, 1963, as amended at 31 F.R. 14394, Nov. 9, 1966]

§ 1.525 Agreements between parties for amendment or dismissal of, or failure to prosecute broadcast applications.

(a) Whenever applicants for a construction permit for a broadcast station enter into an agreement to procure the removal of a conflict between applications pending before the Commission by withdrawal or amendment of an application or by its dismissal pursuant to § 1.568. all parties thereto shall, within 5 days after entering into the agreement, file with the Commission a joint request for approval of such agreement. The joint request shall be accompanied by a copy of the agreement and an affidavit of each party to the agreement setting forth in full all relevant facts including, but not limited to: (1) The exact nature of any consideration (including an agreement for merger of interests) promised or paid; (2) information as to who initiated the negotiations; (3) summary of the history of the negotiations; (4) the reasons why it is considered that the arrangement is in the public interest; and (5) a statement fully explaining and justifying any consideration paid or promised. The affidavit of any applicant to whom consideration is paid or promised shall, in addition, include an itemized accounting of the expenses incurred in connection with preparing, filing and advocating his application, and such factual information as the parties rely upon for the requisite showing that such reported expenses represent legitimate prudent outlays. No such agreement

between applicants shall become effective or be carried out unless and until the Commission has approved it, or until the time for Commission review of the agreement has expired.

- (b) (1) Whenever two or more conflicting applications for construction permits for broadcast stations pending before the Commission involve a determination of fair, efficient and equitable distribution of service pursuant to section 307(b) of the Communications Act. and an agreement is entered into to procure the withdrawal (by amendment to specify a different community or by dismissal pursuant to § 1.568) of the only application or applications seeking the same facilities for one of the communities involved, all parties thereto shall file the joint request and affidavits specified in paragraph (a) of this section. If upon examination of the proposed agreement the Commission finds that withdrawal of one of the applications would unduly impede achievement of a fair, efficient and equitable distribution of radio service among the several States and communities, then the Commission shall order that further opportunity be afforded for other persons to apply for the facilities specified in the application or applications to be withdrawn before acting upon the pending request for approval of the agreement.
- (2) Upon release of an order under subparagraph (1) of this paragraph, any party proposing to withdraw its application shall cause to be published a notice of such proposed withdrawal at least twice a week for 2 consecutive weeks within the 3-week period immediately following release of the Commission's order, in a daily newspaper of general circulation published in the community in which it was proposed to locate the station: Provided, however, That if there is no such daily newspaper published in the community, the notice shall be published as follows:
- (i) If one or more weekly newspapers of general circulation are published in the community in which the station was proposed to be located, notice shall be published in such a weekly newspaper once a week for 3 consecutive weeks within the 4-week period immediately following the release of the Commission's order.
- (ii) If no weekly newspaper of general circulation is published in the community in which the station was proposed to be located, notice shall be published at

- least twice a week for 2 consecutive weeks within the 3-week period immediately following the release of the Commission's order in the daily newspaper having the greatest general circulation in the Community in which the station was proposed to be located.
- (3) The notice shall set forth the name of the applicant; the location, frequency and power of the facilities proposed in the application; the location of the station or stations proposed in the applications with which it is in conflict; the fact that the applicant proposes to withdraw the application; and the date upon which the last day of publication shall take place.
- (4) Such notice shall additionally include a statement that new applications for a broadcast station on the same frequency, in the same community, with substantially the same engineering characteristics and proposing to serve substantially the same service area as the application sought to be withdrawn, timely filed pursuant to the Commission's rules, or filed, in any event, within 30 days from the last date of publication of the notice (notwithstanding any provisions of this chapter normally requiring earlier filing of a competing application), will be entitled to comparative consideration with other pending mutually exclusive applications.
- (5) Within 7 days of the last day of publication of the notice, the applicant proposing to withdraw shall file a statement in triplicate with the Commission, setting forth the dates on which the notice was published, the text of the notice and the newspaper in which the notice was published.
- (6) Where the Commission orders that further opportunity be afforded for other persons to apply for the facilities sought to be withdrawn, no application of any party to the agreement will be acted upon by the Commission less than 30 days from the last day of publication of the notice specified in subparagraph (2) of this paragraph. Any applications for a broadcast station on the same frequency in the same community, with substantially the same engineering characteristics and proposing to serve substantially the same service area as the application sought to be withdrawn, filed within the 30-day period following the last date of publication of the notice (notwithstanding any provisions of this chapter normally requiring earlier filing of a competing application) or otherwise

timely filed pursuant to the provisions of this chapter will be entitled to comparative consideration with other pending mutually exclusive applications. If the application of any party to which the new application may be in conflict has been designated for hearing, any such new application will be entitled to consolidation in the proceeding.

(c) (1) Except where a joint request is filed pursuant to paragraph (a) of this section, any applicant filing (i) amendment pursuant to § 1.522(a) or a request for dismissal pursuant § 1.568(a) which would remove a conflict with another pending application; (ii) a petition for leave to amend pursuant to § 1.522 (b) or (c) which would permit a grant of the amended application or an application theretofore in conflict with the amended application; or (iii) request for dismissal pursuant to § 1.568(c), shall file with it an affidavit as to whether or not consideration (including an agreement for merger of interests) has been promised to or received by such applicant, directly or indirectly, in connection with the amendment, petition or request. Upon the filing of a petition for leave to amend or to dismiss an application for broadcast facilities which has been designated for hearing or upon the dismissal of such application on the Commission's own motion pursuant to § 1.568(b), each applicant or party remaining in hearing, as to whom a conflict would be removed by the amendment or dismissal shall submit for inclusion in the record of that proceeding an affidavit stating whether or not he has directly or indirectly paid or promised consideration (including an agreement for merger of interests) in connection with the removal of such conflict.

(2) Where an affidavit filed pursuant to this paragraph states that consideration has been paid or promised, the affidavit shall set forth in full all relevant facts, including, but not limited to, the material listed in paragraph (a) of this section for inclusion in affidavits.

(d) (1) Affidavits filed pursuant to this section shall be executed by the applicant, permittee or licensee, if an individual; a partner having personal knowledge of the facts, if a partnership; or an officer having personal knowledge of the facts, if a corporation or association.

(2) Requests and affidavits which relate to an application which has not been designated for hearing shall bear the file number of such application. If the affiant is also an applicant, the affidavit shall also bear the file number of affiant's pending application(s). Affidavits which relate to an application which is designated for hearing shall bear the file number of that application and the hearing docket number.

(e) For the purposes of this section an application shall be deemed to be "pending" before the Commission and a party shall be considered to have the status of an "applicant" from the time an application is filed with the Commission until an order of the Commission granting or denying it is no longer subject to reconsideration by the Commission or to review by any court.

(Sec. 5(a), 74 Stat. 892; 47 U.S.C. 311(c).) [28 F.R. 12434, Nov. 22, 1963, as amended at 30 F.R. 6780, May 19, 1965]

§ 1.526 Records to be maintained locally for public inspection by applicants, permittees, and licensees.

(a) Records to be maintained. Every applicant for a construction permit for a new station in the broadcast services shall maintain for public inspection a file for such station containing the material in subparagraph (1) of this paragraph, and every permittee or licensee of a station in the broadcast services shall maintain for public inspection a file for such station containing the material in subparagraphs (1), (2), (3), and (4) of this paragraph: Provided, however, That the foregoing requirements shall not apply to applicants for or permittees or licensees of television broadcast translator stations. The material to be contained in the file is asfollows:

(1) A copy of every application tendered for filing by the applicant for such. station after May 13, 1965, pursuant tothe provisions of this part, with respect to which local public notice is required to be given under the provisions of § 1.580 or § 1.594; and all exhibits, letters and other documents tendered for filing aspart thereof, all amendments thereto, copies of all documents incorporated therein by reference, all correspondence between the Commission and the applicant pertaining to the application after it has been tendered for filing, and copies of Initial Decisions and Final Decisions hearing cases pertaining thereto, which according to the provisions of §§ 0.451-0.461 of this chapter are open for

public inspection at the offices of the Commission. Information incorporated by reference which is already in the local file need not be duplicated if the entry making the reference sufficiently identifies the information so that it may be found in the file, and if there has been no change in the document since the date of filing and the applicant, after making the reference, so states. If petitions to deny are filed against the application, and have been duly served on the applicant, a statement that such a petition has been filed shall appear in the local file together with the name and address of the party filing the petition.

Note: Applications tendered for filing on or before May 13, 1965, which are subsequently designated for hearing after May 13, 1965, with local notice being given pursuant to the provisions of § 1.594, and material related to such applications, need not be placed in the file required to be kept by this section. Applications tendered for filing after May 13, 1965, which contain major amendments to applications tendered for filling on or before May 13, 1965, with local notice of the amending application being given pursuant to the provisions of § 1.580, need not be placed in the file required to be kept by this section.

(2) A copy of every application tendered for filing by the licensee or permittee for such station after May 13. 1965, pursuant to the provisions of this part, which is not included in subparagraph (1) of this paragraph and which involves changes in program service, which requests an extension of time in which to complete construction of a new station, or which requests consent to involuntary assignment or transfer, or to voluntary assignment or transfer not resulting in a substantial change in ownership or control and which may be applied for on FCC Form 316; and copies of all exhibits, letters, and other documents filed as part thereof, all amendments thereto, all correspondence between the Commission and the applicant pertaining to the application after it has been tendered for filing, and copies of all documents incorporated therein by reference, which according to the provisions of §§ 0.451-0.461 of this chapter are open for public inspection at the offices of the Commission. Information incorporated by reference which is already in the local file need not be duplicated if the entry making the reference sufficiently identifies the information so that

it may be found in the file, and there has been no change in the document since the date of filing and the licensee, after making the reference so states. If petitions to deny are filed against the application, and have been duly served on the applicant, a statement that such a petition has been filed shall appear in the local file together with the name and address of the party filing the petition.

(3) A copy of every ownership report or supplemental ownership report filed by the licensee or permittee for such station after May 13, 1965, pursuant to the provisions of this part; and copies of all exhibits, letters, and other documents filed as part thereof, all amendments thereto, all correspondence between the permittee or licensee and the Commission pertaining to the reports after they have been filed, and all documents incorporated therein by reference, including contracts listed in such reports in accordance with the provisions of § 1.615 (a) (4) (i) and which according to the provisions of §§ 0.451-0.461 of this chapter are open for public inspection at the offices of the Commission. Information incorporated by reference which is already in the local file need not be duplicated if the entry making the reference sufficiently identifies the information so that it may be found in the file, and if there has been no change in the document since the date of filing and the licensee or permittee, after making the reference, so states.

(4) Such records as are required to be kept by §§ 73.120(d), 73.290(d), 73.590(d), and 73.657(d) of this chapter, concerning broadcasts by candidates for public office.

NOTE: The engineering section of applications mentioned in subparagraphs (1) and (2) of this paragraph, and material related to the engineering section, need not be kept in the file required to be maintained by this paragraph. If such engineering section contains service contour maps submitted with that section, copies of such maps, and information (State, county, city, street address, or other identifying information) showing main studio and transmitter location shall be kept in the file.

(b) Responsibility in case of assignment or transfer. (1) In cases involving applications for consent to assignment of broadcast station construction permits or licenses, with respect to which public notice is required to be given under the provisions of § 1.580 or § 1.594, the file mentioned in paragraph (a) of

this section shall be maintained by the assignor. If the assignment is consented to by the Commission and consummated, the assignee shall maintain the file commencing with the date on which notice of the consummation of the assignment is filed with the Commission. The file maintained by the assignee shall cover the period both before and after the time when the notice of consummation of assignment was filed. The assignee is responsible for obtaining copies of the necessary documents from the assignor or from the Commission files.

(2) In cases involving applications for consent to transfer of control of a permittee or licensee of a broadcast station, the file mentioned in paragraph (a) of this section shall be maintained by the

permittee or licensee.

- (c) Station to which records pertain. The file need contain only applications, ownership reports, and related material that concern the station for which the file is kept. Applicants, permittees, and licensees need not keep in the file copies of such applications, reports, and material which pertain to other stations with regard to which they may be applicants, permittees, or licensees, except to the extent that such information is reflected in the materials required to be kept under the provisions of this section.
- (d) Location of records. The file shall be maintained at the main studio of the station, or at any accessible place (such as a public registry for documents or an attorney's office) in the community to which the station is or is proposed to be licensed, and shall be available for public inspection at any time during regular business hours.
- (e) Period of retention. The records specified in paragraph (a) (4) of this section shall be retained for the periods specified in §§ 73.120(d), 73.290(d), 73.590(d), and 73.657(d) of this chapter (2 years). The records specified in paragraph (a) (1), (2) and (3) of this section shall be retained as follows:
- (1) The applicant for a construction permit for a new station shall maintain such a file so long as the application is pending before the Commission or any proceeding involving that application is pending before the courts. (If the application is granted, subparagraph (2) of this paragraph shall apply.)
 - (2) The permittee or licensee shall

maintain such a file so long as an authorization to operate the station is outstanding.

[30 F.R. 4546, Apr. 8, 1965, as amended at 30 F.R. 13541, Oct. 23, 1965; 31 F.R. 14000, Nov. 2, 1966; 32 F.R. 20861, Dec. 28, 1967]

Application Forms and Particular Filing Requirements

§ 1.531 Formal and informal applications.

- (a) "Formal application" means any request for authorization where an FCC Form for such request is prescribed. "Informal application" means all other requests for authorization. Informal applications may be in letter form, but all such applications should contain a caption clearly indicating the nature of the request submitted therein.
- (b) An informal application requesting modification of an outstanding authorization must comply with the requirements as to signing specified in §§ 1.511 and 1.513.

§ 1.533 Application forms for authority to construct a new station or make changes in an existing station.

- (a) Applications for new facilities or modification of existing facilities shall be made on the following forms:
- (1) FCC Form 301 "Application for Authority to Construct a New Broadcast Station or Make Changes in an Existing Broadcast Station."
- (2) FCC Form 309 "Application for Authority to Construct or Make Changes in an Existing International, Experimental Television, Experimental Facsimile, or a Developmental Broadcast Station."
- (3) FCC Form 313 "Application for Authorization in the Auxiliary Broadcast Services."
- (4) FCC Form 318 "Request for Subsidiary Communications Authorizations." For use by existing FM broadcast licensees applying for permit to establish an SCA service, modification of SCA, or renewal of SCA.
- (5) FCC Form 330 P "Application for Authority to Construct or Make Changes in an Instructional Fixed Station."
- (6) FCC Form 340 "Application for Authority to Construct or make Changes in a Noncommercial Educational TV, FM, or Standard Broadcast Station."
- (7) FCC Form 343 "Application for Authority to Construct or make Changes in a Television Broadcast Booster Station."

- (8) FCC Form 346 "Application for Authority to Construct or make Changes in a Television Broadcast Translator Station."
- (b) Applications for construction permit or modification thereof involving the installation of new transmitting apparatus should be filed at least 60 days prior to the contemplated construction.

(Sec. 308, 319, 48 Stat. 1084, 1089, as amended; 47 U.S.C. 308, 319)

- § 1.534 Application for extension of construction permit or for construction permit to replace expired construction permit.
- (a) Application for extension of time within which to construct a station shall be filed on FCC Form 701. The application shall be filed at least 30 days prior to the expiration date of the construction permit if the facts supporting such application for extension are known to the applicant in time to permit such filing. In other cases, an application will be accepted upon a showing satisfactory to the Commission of sufficient reasons for filing within less than 30 days prior to the expiration date. Such applications will be granted upon a specific and detailed showing that the failure to complete was due to causes not under the control of the grantee, or upon a specific and detailed showing of other matters sufficient to justify the extension.
- (b) Application to replace an expired construction permit shall be made on FCC Form 321 "Application for Construction Permit to Replace Expired Permit." Such application must be filed within 30 days of the expiration date of the authorization sought to be replaced. (Sec. 319, 48 Stat. 1089, as amended; 47 U.S.C. 319)
- § 1.536 Application for license to cover construction permit.
- (a) The application for station license shall be filed by permittee prior to service or program tests.
- (b) The following application forms shall be used:
- (1) FCC Form 302 "Application for New Broadcast Station License."
- (2) FCC Form 310 "Application for an International, Experimental Television, Experimental Facsimile, or a Developmental Broadcast Station License."
- (3) FCC Form 313 "Application for Authorization in the Auxiliary Broadcast Services."

- (4) FCC Form 318 "Request for Subsidiary Communications Authorization."
- (5) FCC Form 341 "Application for Noncommercial Educational TV, FM, or Standard Broadcast Station."
- (6) FCC Form 344 "Application for Television Broadcast Booster Station License."
- (7) FCC Form 347 "Application for Television Broadcast Translator Station License."

(Sec. 319, 48 Stat. 1089, as amended; 47 U.S.C. 319)

§ 1.537 Application for license to use former main transmitter or antenna as an auxiliary.

The following application forms shall be used when no new construction is involved:

- (a) FCC Form 302 "Application for New Broadcast Station License."
- (b) FCC Form 341 "Application for Noncommercial Educational TV, FM, or Standard Broadcast Station."

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

- § 1.538 Application for modification of license.
- (a) An application for modification of license may be filed for:
- (1) Change in name of licensee where no change in ownership or control is involved:
- (2) Change in station location involving no change in transmitter location;
- (3) Change in main studio location of a television station to or from a location outside the principal community;
- (4) Change in studio location of a standard or FM station to a location outside the city limits other than the associated transmitter site;
- (5) Change of hours of operation of a standard broadcast station.
- (b) The application forms set forth in § 1.533 shall be used.
- (Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)
- § 1.539 Application for renewal of license.
- (a) Unless otherwise directed by the Commission, an application for renewal of license shall be filed at least 90 days prior to the expiration date of the license sought to be renewed, except that applications for renewal of license of an experimental or developmental broadcast station shall be filed at least 60 days prior to the expiration date of the license sought to be renewed.

- (b) No application for renewal of license of any broadcast station will be considered unless there is on file with the Commission the information, if any, currently required by §§ 1.611-1.615, inclusive, for the particular class of station. The renewal application shall include a reference by date and file number to such information on file.
- (c) Whenever the Commission regards an application for a renewal of license as essential to the proper conduct of a hearing or investigation, and specifically directs that it be filed by a date certain, such application shall be filed within the time thus specified. If the licensee fails to file such application within the prescribed time, the hearing or investigation shall proceed as if such renewal application had been received.
- (d) The following application forms shall be used:
- (1) FCC Form 303 "Application for Renewal of Broadcast Station License."
- (2) FCC Form 311 "Application for Renewal of an International, Experimental Television, Experimental Facsimile, or a Developmental Broadcast Station License." To be used for all applications for renewal of licenses of Experimental Television, Experimental Facsimile, and Developmental Broadcast stations.
- (3) FCC Form 313 "Application for Authorization in the Auxiliary Broadcast Services." To be used for all applications for renewal of regular licenses of auxiliary broadcasting stations.
- (4) FCC Form 318 "Request for Subsidiary Communications Authorizations."
- (5) FCC Form 342 "Application for Renewal of Noncommercial Educational TV, FM, or Standard Broadcast Station License."
- (6) FCC Form 345 "Application for Renewal of Television Broadcast Booster Station License."
- (7) FCC Form 348 "Application for Renewal of Television Broadcast Translator Station License."
- (Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

§ 1.540 Application for voluntary assignment or transfer of control.

(a) Application for consent to the assignment of construction permit or license, or for consent to the transfer of control of a corporation holding such a construction permit or license, shall be filed with the Commission on FCC Form

- 314 (Assignment of License), FCC Form 315 (Transfer of Control), or FCC Form 316 (Short Form). Such application should be filed with the Commission at least 45 days prior to the contemplated effective date of assignment or transfer of control.
- (b) The following assignment or transfer applications may be filed on FCC Form 316.
- (1) Assignment from an individual or individuals (including partnerships) to a corporation owned and controlled by such individuals or partnerships without any substantial change in their relative interests:
- (2) Assignment from a corporation to its individual stockholders without effecting any substantial change in the disposition of their interests;
- (3) Assignment or transfer by which certain stockholders retire and the interest transferred is not a controlling one:
- (4) Corporate reorganization which involves no substantial change in the beneficial ownership of the corporation;
- (5) Assignment or transfer from a corporation to a wholly owned subsidiary thereof or vice versa, or where there is an assignment from a corporation to a corporation owned or controlled by the assignor stockholders without substantial change in their interests; or
- (6) Assignment of less than a controlling interest in a partnership. (Sec. 310, 48 Stat. 1086, as amended; 47 U.S.C. 310)

§ 1.541 Application for involuntary assignment of license or transfer of control.

- (a) The Commission shall be notified in writing promptly of the death or legal disability of an individual permittee or licensee, a member of a partnership, or a person directly or indirectly in control of a corporation which is a permittee or licensee.
- (b) Within 30 days after the occurrence of such death or legal disability, an application on FCC Form 316 shall be filed requesting consent to involuntary assignment of such permit or license or for involuntary transfer of control of such corporation to a person or entity legally qualified to succeed to the foregoing interests under the laws of the place having jurisdiction over the estate involved.

(Sec. 310, 48 Stat. 1086, as amended; 47 U.S.C. 310)



§ 1.542 Application for temporary authorization.

- (a) The specific circumstances in which temporary authority will be granted are set out in Parts 2, 73, and 74 of this chapter.
- (b) Temporary authority may be granted to a licensee or permittee of a broadcast station to operate such station for a period not to exceed 90 days upon request therefor. Any such request should be filed with the Commission at least 10 days prior to the date of the proposed operation, and should be accompanied by a statement giving full particulars as to the purpose for which the request is made. Any temporary authority issued under this section may be cancelled by the Commission without further notice or hearing.
- (c) No request by a standard broadcast station for temporary authority to extend its hours of operation beyond those authorized by its regular authorization will be accepted or granted by the Commission.
- (d) An informal application may be used provided such application is signed in accordance with the provisions of § 1.513.
- (e) Request for temporary operation necessitated by equipment damage or failure may be made without regard to the procedural requirements of this section.
- (Sec. 309, 48 Stat. 1085, as amended; 47 U.S.C. 309)
- § 1.543 Application for renewal or modification of special service authorization.
- (a) No special service authorization will be issued after February 3, 1958: Provided, however, Consideration will be given to renewal or modification of a special service authorization outstanding on February 3, 1958 providing a satisfactory showing has been made in regard to the following, among others:
- (1) That the requested operation may not be granted on a regular basis under the existing rules governing the operation of standard broadcast stations;
- (2) That experimental operation is not involved as provided for by § 73.32 of this chapter; and
- (3) That public interest, convenience, and necessity will be served by the authorization requested.
- [28 F.R. 12434, Nov. 22, 1963, as amended at 29 F.R. 19256, Dec. 31, 1964]

§ 1.544 Application for standard broadcast station experimental operation.

Special experimental authorization may be issued, in accordance with § 73.32 of this chapter, to the licensee of a standard broadcast station in addition to the regular license. An informal application should be used in applying for such authorization.

(Sec. 303(g), 48 Stat. 1083, as amended; 47 U.S.C. 303(g))

§ 1.545 Application for permit to deliver programs to foreign stations.

Application under section 325(b) of the Communications Act for authority to locate, use, or maintain a broadcast studio in connection with a foreign station consistently received in the United States, should be made on FCC Form 308 "Application for permit to deliver programs to foreign broadcast stations": Provided. That an informal application (in letter form) may be used by applicants holding a valid radio or TV broadcast station license or construction permit. Informal applications must, however, contain a description of the nature and character of the programing proposed, together with other information requested on Page 4 of Form 308.

[32 F.R. 2771, Feb. 10, 1967]

§ 1.546 Application to determine operating power by direct measurement of antenna power.

Application to determine operating power of standard broadcast stations by direct measurement of antenna power shall be made on FCC Form 302 "Application for New Broadcast Station License."

§ 1.547 Application for permission to use lesser grade operators.

(a) Application for temporary permission to operate standard and FM broadcast stations with licensed operators of a lesser grade than normally required by the Commission's rules shall be submitted to the Engineer in Charge of the radio district in which the station is located. Such permission will be granted for periods not to exceed 60 days if a proper showing is made, as set forth in this section, and may be renewed upon request only upon the making of an adequate similar showing. A request for extension of the permission previously granted may be granted upon a showing setting forth what continuing efforts have been made to obtain



licensed operators of a grade normally required. The Engineer in Charge may terminate this permission in the absence of a satisfactory showing in the written report that adequate efforts have been made to obtain such operators, or for other good reason in the judgment of the Engineer in Charge.

(b) Such application or report is not required to be submitted on any numbered or prescribed form. However, the request or report shall be in writing. signed by the licensee, if the licensee is an individual; by a partner, if the licensee is a partnership; or by an officer of the corporation, if the licensee is a corporation.

(c) A specific request for permission to use operators of lesser grade than required by the Commission's rules shall include the following information:

(1) Call letters of the station;

(2) Name of licensee;

(3) The number of persons holding radiotelephone first class operator licenses that will be employed as full-time operators at the station (this does not include part-time employees and persons only available on call in case of emergencies);

(4) A showing that at least one first class operator will be employed full time at the station and will be available on call at all times in the event of equip-

ment failure:

(5) A statement that the additional licensed radiotelephone first class operators required for maintaining the normal schedule of operation could not be obtained for employment at the station;

(6) In the event an operator of the required grade was rejected by the station, a statement should be submitted by the station showing the reason for the

rejection; and

(7) A showing that all known sources of broadcast operators within a reasonable distance have been exhausted. Names and addresses of sources contacted and the date of such contact shall

be stated. (d) The chief operator holding a radiotelephone first class operator license at a station to which temporary permission has been granted shall mail to the Engineer in Charge of the area from whom permission is received, within 3 days after employment of a lesser grade operator, a written certification setting forth the name and operator license number of the lesser grade operator employed and stating that the operator has the ability to perform the normal operation of the station.

(Sec. 318, 48 Stat. 1089, as amended; 47 U.S.C. 318)

§ 1.548 Application to operate by remote control.

Application by an existing licensee orpermittee for a permit to operate a standard or FM broadcast station by remotecontrol shall be made on FCC Form "Request for Modification-301-A Broadcast Station Authorization (Remote Control)."

(Sec. 318, 48 Stat. 1089, as amended; 47 U.S.C. 318)

§ 1.549 Requests for extensions of authority to operate without certain indicating instruments.

Requests for extension of authority tooperate without a frequency monitor, a modulation monitor, a plate ammeter or voltmeter, a base current meter or common point meter, or a transmission line meter for FM and television stations, should be made by informal application to the Engineer in Charge of the radiodistrict in which the station is located. Such requests must contain information as to when and what steps were taken torepair or replace the defective instrument.

§ 1.550 Requests for new or modified call sign assignments.

(a) Requests for new or modified call sign assignments for standard, FM or television broadcasting stations shall bemade by letter to the Secretary, Federal Communications Commission, Washington, D.C., 20554.

(b) (1) No request for a new call sign. assignment shall be made by an applicant for a new station until after the-Commission has granted the construction

permit.

- (2) An applicant for transfer or assignment of an existing station may request a new call sign assignment at the time the application for assignment of license or transfer of control is filed, or at any time thereafter. In making such a request he shall comply with the provisions of this section.
- (c) Each individual request submitted under the provisions of paragraphs (a) and (b) of this section shall include the following:
- (1) A statement that a copy of therequest has been served upon each broadcast station licensed to operate, or whose-

construction has been authorized, in communities wholly or partially within a 35-mile radius of the main post office of the community in which the applicant is authorized to operate, and a list of the call signs and locations of all stations upon which copies of the request have been served.

- (2) Subject to the other requirements of this paragraph, as many as five call signs, listed in descending order of preference, may be included in a single request.
- (d) (1) No request for call signs subject to the provisions of this section will be acted upon by the Commission earlier than 30 days following issuance of public notice of the receipt of such request. Applicants for new or modified call signs are therefore, advised to take no action in reliance on securing said call sign until notified by the Commission that said request has been granted.
- (2) Objections to the assignment of the requested call signs may be filed within the 30-day period following issuance of public notice of the receipt of such request. Objections filed after the 30-day period will be considered only if, in the judgment of the Commission, good -cause has been shown for failure to file within the time specified. The person who filed the original pleading may reply to said objections within 10 days after the time for filing objections has expired, or if the objections are filed late, within 10 days after the late pleading is filed. No further pleadings will be entertained.
- (e) The requirements of paragraphs (b), (c), and (d) of this section do not apply to international broadcasting stations and stations in the experimental, auxiliary, and special broadcasting services.
- (f) If an applicant for a new station does not file a request for assignment of a specific call sign, the Commission will assign an appropriate call sign to the station.

[29 F.R. 15576, Nov. 20, 1964, as amended at 31 F.R. 9798, July 20, 1966]

APPLICATION PROCESSING PROCEDURES

AUTHORITY: \$\$ 1.561 to 1.587 issued under secs. 308, 309, 48 Stat. 1084, 1085, as amended; 47 U.S.C. 308, 309.

§ 1.561 Staff consideration of applications which receive action by the Commission.

Upon acceptance of an application, the complete file is reviewed by the staff and, except where the application is

acted upon by the staff pursuant to delegation of authority, a report containing the recommendations of the staff and any other documents required is prepared and placed on the Commission's agenda.

§ 1.562 Staff consideration of applications which do not require action by the Commission.

Those applications which do not require action by the Commission but which, pursuant to the delegations of authority set forth in Subpart B of Part 0 of this chapter, may be acted upon by Chief, Broadcast Bureau are forwarded to the Broadcast Bureau for necessary action. If the application is granted, the license division issues the formal authorization. In any case where it is recommended that the application be set for hearing, where a novel question of policy is presented, or where the Chief, Broadcast Bureau desires instructions from the Commission, the matter is placed on the Commission agenda.

(Sec. 5, 48 Stat. 1068, as amended; 47 U.S.C. 155)

§ 1.564 Acceptance of applications.

- (a) Applications which are tendered for filing in Washington, D.C., are dated upon receipt and then forwarded to the Broadcast Bureau, where an administrative examination is made to ascertain whether the applications are complete. Applications found to be complete or substantially complete are accepted for filing and are given a file number. In case of minor defects as to completeness, the applicant will be required to supply the missing information. Applications which are not substantially complete will be returned to the applicant.
- (b) Acceptance of an application for filing merely means that it has been the subject of a preliminary review by the Commission's administrative staff as to completeness. Such acceptance will not preclude the subsequent dismissal of the application if it is found to be patently not in accordance with the Commission's rules.
- (c) At regular intervals the Commission will issue a "Public Notice" listing all applications and major amendments thereto which have been accepted for filing.

[28 F.R. 12434, Nov. 22, 1963, as amended at 29 F.R. 12371, Aug. 28, 1964]

§ 1.566 Defective applications.

(a) Applications which are determined to be patently not in accordwith the Commission's rules. regulations. or other requirements. unless accompanied bу an appropriate request for waiver, will be con-sidered defective and will not be ac-cepted for filing or if inadvertently accepted for filing will be dismissed. Requests for waiver shall show the nature of the waiver or exception desired and shall set forth the reasons in support thereof.

(b) If an applicant is requested by the Commission to file any additional documents or information not included in the prescribed application form, a failure to comply with such request will be deemed to render the application defective, and such application will be dismissed.

§ 1.568 Dismissal of applications.

(a) Subject to the provisions of § 1.525, any application may, upon request of the applicant, be dismissed without prejudice as a matter of right prior to the designation of such application for hearing. An applicant's request for the return of an application that has been accepted for filing will be regarded as a request for dismissal.

(b) Failure to prosecute an application, or failure to respond to official correspondence or request for additional information, will be cause for dismissal. Subject to the provisions of § 1.525, such dismissal will be without prejudice where an application has not yet been designated for hearing, but may be made with prejudice after designation for hearing.

(c) Requests to dismiss an application without prejudice after it has been designated for hearing will be considered only upon written petition properly served upon all parties of record and, where applicable, compliance with the provisions of § 1.525. Such requests shall be granted only upon a showing that the request is based on circumstances wholly beyond the applicant's control which preclude further prosecution of his application.

[28 F.R. 12434, Nov. 22, 1963, as amended at 29 F.R. 6444, May 16, 1964]

§ 1.569 Applications for frequencies adjacent to Class I—A channels.

Notwithstanding the provisions of any other rules of the Commission, all ap-

plications (regardless of when they were or may be filed) for frequencies located within 30 kc/s of a Class I-A channel listed in § 73.25(a) of this chapter will be subject to the provisions of this section. The provisions of paragraph (a) of this section apply to the frequencies listed therein, which are within 30 kc/s of a Class I-A channel on which an unlimited time Class II assignment is specifically provided for in § 73.22 or § 73.25 (a) of this chapter, and which are not within 30 kc/s of the remaining Class I-A channels (except that the frequency 1230 kc/s is considered to be within 30 kc/s of duplicated I-A channels only). The provisions of paragraph (b) of this section apply to the frequencies listed in that paragraph, which are within 30 kc/s of unduplicated Class I-A channels and also, in each case, within 30 kc/s of Class I-A channels on which an unlimited time Class II assignment is specifically provided for in § 73.22 or § 73.25 (a) of this chapter. For the purposes of paragraph (b)(2) (i), (ii) and (iii) of this section, the frequency 750 kc/s is regarded as an unduplicated Class I-A clear channel. The provisions of paragraph (c) of this section apply to the three frequencies listed therein, which are within 30 kc/s of unduplicated clear channels only.

(a) (1) The provisions of this paragraph apply to the following frequencies: 910, 920, 990, 1000, 1080, 1090, 1110, 1230, and 1240 kc/s; and 740 kc/s except with respect to the Class I-A clear channel 750 kc/s, in which case the provisions of paragraph (b) of this section apply.

(2) Where it appears that the facilities requested in any application for one of the designated frequencies (other than an application by an existing Class IV station to increase daytime power on 1230 or 1240 kc/s) involves undue risk of interference or prohibited overlap with a possible new Class II-A assignment specified in § 73.22 of this chapter or a new unlimited time Class II assignment at Anchorage, Alaska, or San Diego, California, specified in § 73.25(a) of this chapter, such application will not be accepted for filing or, if filed prior to the effective date of this section, will not be acted upon, until the location and operating facilities of such new Class II station are established. An application for one of the designated frequencies shall be deemed to involve undue risk of interference with a possible new Class II assignment unless it is demonstrated

that no interference would be caused to specified II-A assignments within 30 kc/s, assuming such facilities to be located at the nearest point on the boundary of the nearest state specified by the Clear Channel Decision released September 14, 1961, and assuming such II-A facility radiates at least 1238 mv/m omnidirectionally; and, in the case of frequencies within 30 kc/s of 750 kc/s or 760 kc/s the proposed facility would not cause interference to Class II assignments at San Diego, California, or Anchorage, Alaska, specified in § 73.25 (a) of this chapter.

(3) Assignments of new Class II facilities provided for in §§ 73.22 and 73.25(a) of this chapter will be made without regard to the pendency of applications on adjacent frequencies (i.e., the ten frequencies designated in subparagraph (1) of this paragraph and the additional 20 frequencies which are within 30 kc/s of both duplicated and unduplicated Class I-A channels). Any hearing which may be held on an application for an adjacent frequency will not be comparative with respect to the Class II-A facility. and any issues pertaining to the mutual impact of the Class II-A and adjacent channel proposals will be confined to the question of whether, with the Class II station operating as proposed, the public interest would be served by a grant of the adjacent channel application.

(b) (1) The provisions of this paragraph apply to the following frequencies: 680, 690, 710, 730, 790, 800, 810, 850, 860, 900, 1010, 1050, 1060, 1070, 1130, 1140, 1150, 1170, 1190, and 1220 kc/s; and 740 kc/s with respect to the Class I-A clear channel 750 kc/s.

(2) Applications for new stations on, change of existing stations to, or for any major change in operation of stations presently operating on the designated frequencies will be accepted for filing and acted upon in normal course provided they are accompanied by appropriate exhibits and necessary supporting data to show clearly the following with respect to all Class I-A channels within 30 kc/s of the designated frequency:

(i) The proposed transmitter site is located inside the area encompassed by a 500 mile extension of the 0.5 mv/m-50 percent nighttime contour of Class I-A stations on unduplicated channels.

(ii) No interference or prohibited overlap would be caused to Class I-A stations on unduplicated I-A channels, assuming such stations operate with power increased to 750 kw with their present antenna systems and radiation patterns.

(iii) No interference or prohibited overlap would be caused to an assumed Class II—A station on an unduplicated channel, radiating at least 1238 mv/m omnidirectionally from the nearest point on the boundary described in subdivision (i) of this subparagraph.

(iv) No interference or prohibited overlap would be caused to presently specified Class II-A assignments, assuming such facilities to be located at the nearest point on the boundary of the nearest state specified by the Clear Channel Decision released September 14. 1961, and assuming such II-A facility radiates at least 1238 mv/m omnidirectionally; and, in the case of frequencies within 30 kc/s of 750 kc/s or 760 kc/s, the proposed facility would not cause interference to Class II assignments at San Diego, California, or Anchorage, Alaska, specified in § 73.25(a) of this chapter.

(3) Applications of the type specified in subparagraph (2) of this paragraph filed before October 30, 1961, will be studied to determine whether they comply with each of the four standards set forth in subparagraph (2) of this paragraph. Those applications which are found to comply fully with the standards will be processed and acted upon in normal course. Applications which fail to comply with one or more of the standards will be retained in the pending file without further processing or consideration.

(4) Applications for other changes in facilities on the designated frequencies will be processed and acted upon in normal course.

(c) (1) The provisions of this paragraph apply to all applications for the following frequencies: 610, 620, 630 kc/s.

(2) Applications for new stations on, change of existing stations to, or for any major change in operation of stations presently operating on the designated frequencies will be accepted for filing and acted upon in normal course provided they are accompanied by appropriate exhibits and necessary supporting data to show clearly the following with respect to all Class I—A channels within 30 kc/s of the designated frequency:

(i) The proposed transmitter site is located inside the area encompassed by a 500 mile extension of the 0.5 mv/m-50 percent nighttime contour of Class I-A stations on unduplicated channels.

(ii) No interference or prohibited overlap would be caused to Class I-A stations on unduplicated I-A channels, assuming such stations operate with power increased to 750 kw, with their present antenna systems and radiation patterns.

(iii) No interference or prohibited overlap would be caused to an assumed Class II-A station on an unduplicated channel, radiating at least 1238 my/m omnidirectionally from the nearest point on the boundary described in subdivision

(i) of this subparagraph.

- (3) Applications of the type specified in subparagraph (2) of this paragraph filed before October 30, 1961, will be studied to determine whether they comply with each of the three standards set forth in subparagraph (2) of this paragraph. Those applications which are found to comply fully with the standards will be processed and acted upon in normal course. Applications which fail to comply with one or more of the standards will be retained in the pending file without further processing or consideration.
- (4) Applications for other changes in facilities on the designated frequencies will be processed and acted upon in normal course.
- (d) (1) Applications previously accepted for filing which must be held without action pursuant to paragraph (a), (b), or (c) of this section, will not be designated for hearing unless they conflict with applications which may be acted upon in normal course.
- (2) If the decision in a hearing looks toward grant of an application which may not be acted upon, pursuant to paragraph (a), (b), or (c) of this section, such application and all applications conflicting with it will be held without final action to the extent required by paragraph (a), (b), or (c) of this section. (Sec. 307, 48 Stat. 1083; 47 U.S.C. 307) [28 F.R. 12434, Nov. 22, 1963, as amended at 29 F.R. 11360, Aug. 6, 1964; 30 F.R. 9687, Aug. 4, 1965]
- § 1.570 Standard broadcast applications involving other North American countries.
- (a) Applications involving conflicts with the U.S./Mexican Agreement or with countries which have ratified NARBA. Except for applications falling within the provisions of paragraph (b) of this section, no application will be accepted for filing if authorization of the

- facilities requested in such application would be inconsistent with the provisions of the North American Regional Broadcasting Agreement (NARBA), or the Agreement Between the United States of America and the United Mexican States Concerning Radio Broadcasting in the Standard Broadcast Band (the U.S./Mexican Agreement). Any such application which has heretofore been accepted for filing or which is inadvertently accepted for filing will be dismissed.
- (b) Applications involving conflicts only with respect to Haiti or countries which have signed but not ratified Applications (regardless of when they were or may be filed) for facilities which would be inconsistent with NARBA only with respect to a country which has signed but not completed formal ratification of that agreement, or which would cause objectionable interference (under the standards set forth in NARBA) to a duly notified Haitian station, will be placed or retained in the pending file without further action, except where they conflict with other applications which do not involve international problems. In the latter situation, the various conflicting applications will be designated for hearing in a consolidated proceeding. Where an application inconsistent with international relationships as specified in this paragraph is designated for hearing, the following procedures will govern:

(1) Where it is found that, of the applications involved in a consolidated hearing proceeding, all are inconsistent with international relationships as specified in this paragraph, all will be removed from hearing status and returned to the pending file.

(2) Where, of the applications involved in a consolidated hearing proceeding, one or more but not all are inconsistent with international relationships as specified in this paragraph the hearing issues will include an issue as to such inconsistency (if necessary the hearing issues will be enlarged, and if closed the hearing record will be reopened, to include this matter). initial decision and the final decision will contain findings and conclusions as to this issue, but neither the presiding officer nor the Commission will in their decisions take into account such issue in determining whether the public interest would be served by grant of any of the various

applications. In the decision in such a proceeding, applications will be:

(i) Granted, where they are not inconsistent with international relationships and the public interest will be served thereby.

(ii) Denied, if denial is required because of grant of other applications or for other reasons independent of the consistency issue: or

(iii) Placed in the pending file without removal from hearing status if grant of the application would be in the public interest except for inconsistency with international relationships as specified in this paragraph, or where denial would be only on the basis of comparative consideration with an application which is being placed in the pending file because of such inconsistency.

(3) Where an application inconsistent with international relationships is designated for hearing because of conflict with another application not involving such inconsistency, and the conflict is later removed by amendment or dismissal of the latter application, the inconsistent application will be removed from hearing status and returned to the pending file.

Note 1: Upon ratification by Canada, Cuba and the United States, NARBA entered into force April 19, 1960; the Dominican Republic deposited its ratification on May 4, 1961. When the other signatory power, The Bahama Islands and Jamaica, ratifies the agreement, or when Halti (not a signatory power) formally adheres thereto, the Commission upon notification thereof will give public notice of such occurrence. Applications involving conflicts with respect to such country will thereupon automatically be removed from the provisions of paragraph (b) of this section and will fall within paragraph (a) of this section.

If The Bahama Islands and Jamaica completes formal ratification of NARBA, and at that time Haiti has not yet formalized its adherence to the agreement, the Commission will give consideration to whether applications involving conflicts with Haitian stations should continue to be handled as provided in paragraph (b) of this section, or whether, in view of the then pertaining relationship with Haiti in this area, they should be handled as provided in paragraph (a) of this section, or should be handled otherwise. Applicants for facilities involving conflicts with duly notified Haitian stations should take note of these possibilities.

Nore 2: For the purpose of this section, an application is not regarded as inconsistent with the provisions of NARBA if it is for Class IV facilities operating with more than 250 watts but no more than 1 kw power, to be located in those portions of the United

States where such facilities are not precluded under note 1 to § 73.21(c) of this chapter, and where such facilities would not cause objectionable interference (under the standards set forth in NARBA) to a duly notified station in any other NARBA signatory country or in Haiti.

NOTE 3: As to the use in hearings of groundwave field intensity measurements involving foreign countries, see the note to \$73.183(b) of this chapter.

- (c) Amendment of application designated for hearing. When, in the case of any application which has been designated for hearing on issues not including an issue as to consistency with international relationships and as to which no final decision has been rendered, action under this section becomes appropriate because of inconsistency with international relationships, the applicant involved shall, notwithstanding the provisions of §§ 1.522 and 1.571, be permitted to amend its application to achieve consistency with such relationships. In such cases the provisions of $\S 1.605(c)$ will apply.
- (d) Applications not involving conflict with NARBA or U.S./Mexican Agreement. As a matter of general practice, applications which are consistent with NARBA and the U.S./Mexican Agreement and which would not involve objectionable interference to a duly notifled Haitian assignment, will be considered and acted upon by the Commission in accordance with its established procedure. In particular cases, involving applications of this character but in which special international considerations require that a different procedure be followed, the applicant involved will be formally advised to this effect.

[28 F.R. 12434, Nov. 22, 1963, as amended at 32 F.R. 20861, Dec. 28, 1967]

§ 1.571 Processing of standard broadcast applications.

- (a) Applications for standard broadcast facilities are divided into three groups.
- (1) In the first group are applications for new stations (except applications for new Class II—A stations) or for major changes in the facilities of authorized stations, i.e., any changes in frequency, power, hours of operation, or station location: *Provided, however*, That the Commission may, within 15 days after the tender for filing of any application for other modification of facilities, advise the applicant that such application

is considered to be one for a major change and therefore is subject to the provisions of § 1.580.

- (2) The second group consists of applications for licenses and all other changes in the facilities of authorized stations.
- (3) The third group consists of applications for new Class II-A stations.
- (b) If an application is amended so as to effect a major change as defined in paragraph (a) (1) of this section or so as to result in a transfer of control or assignment which, in the case of an authorized station, would require the filing of an application therefor on FCC Form 314 or 315 (see § 1.540), § 1.580 will apply to such amended application.
- (c) Applications for new stations (except new Class II-A stations) or for major changes in the facilities of authorized stations are processed as nearly as possible in the order in which they are filed. Such applications will be placed in the processing line in numerical sequence, and are drawn by the staff for study, the lowest file number first. Thus, the file number determines the order in which the staff's work is begun on a particular application. There are two exceptions thereto: the Broadcast Bureau is authorized to (1) group together for processing applications which involve interference conflicts where it appears that the applications must be designated for hearing in a consolidated proceeding; and (2) to group together for processing and simultaneous consideration, without designation for hearing, all applications filed by existing Class IV stations requesting an increase in daytime power which involve interlinking interference problems only, regardless of their respective dates of filing. In order that those applications which are entitled to be grouped for processing may be fixed prior to the time processing of the earliest filed application is begun, the Commission will periodically publish in the FEDERAL REG-ISTER a Public Notice listing applications which are near the top of the processing line and announcing a date (not less than 30 days after publication) on which the listed applications will be considered available and ready for processing and by which all applications excepting those specified in exception (2) in this paragraph must be filed if they are to be grouped with any of the listed applications: Provided, That applications which are mutually exclusive with applications for renewal of license of stand-

ard broadcast stations will not be so listed in such a public notice, but will be treated as available and ready for processing upon timely filing as provided in § 1.516(e).

(d) Applications for new Class II-A stations are placed at the head of the processing line and processed as quickly as possible. Action on such applications may be at any time more than 30 days after public notice is given of acceptance of the application for filing.

(e) The processing and consideration of applications for new stations or major changes on those frequencies specified in § 1.569 are subject to certain restric-

tions, as set forth therein.

(f) Applications other than those for new stations or for major changes in the facilities of authorized stations are not placed on the processing line but are processed as nearly as possible in the order in which they are filed.

(g) Applications for modifications of license to change hours of operation of a class IV station, to decrease hours of operation of any other class of station, or to change station location involving no change in transmitter site will be considered without reference to the processing line.

(h) If, upon examination, the Commission finds that the public interest, convenience, and necessity will be served by the granting of an application, the same will be granted. If, on the other hand, the Commission is unable to make such a finding and it appears that a hearing may be required, the procedure set forth in § 1.593 will be followed.

- (i) When an application which has been designated for hearing has been removed from the hearing docket, the application will be returned to its proper position (as determined by the file number) in the processing line. Whether or not a new file number will be assigned will be determined pursuant to paragraph (j) of this section after the application has been removed from the hearing docket.
- (j) (1) A new file number will be assigned to an application for a new station, or for major changes in the facilities of authorized stations, when it is amended to change frequency, to increase power, to increase hours of operation, or to change station location. Any other amendment modifying the engineering proposal, except an amendment respecting the type of equipment specified, will also result in the assign-

ment of a new file number unless such amendment is accompanied by a complete engineering study showing that the amendment would not involve new or increased interference problems with existing stations or other applications pending at the time the amendment is filed. If, after submission and acceptance of such an engineering amendment, subsequent examination indicates new or increased interference problems with either existing stations or other applications pending at the time the amendment was received in the Commission, the application will then be assigned a new file number and placed in the processing line according to the numerical sequence of the new file number.

- (2) A new file number will be assigned to an application for a new station when it is amended to specify a change in ownership as a result of which one or more parties with an ownership interest in the original application do not have, on a collective basis, a 50 percent or more ownership interest in the amended application.
- (3) An application for changes in the facility of an existing station will continue to carry the same file number although an assignment of license or transfer of control of said licensee (permittee)-applicant has been consented to by the Commission, provided the application for changes in facility (FCC Form 301) is amended jointly by the assignor and assignee or transferor and transferee, upon consummation of the assignment or transfer, to reflect the ownership changes and to include the financial and programming proposals of the new licensee (permittee)-applicant.
- (k) When an application is reached for processing, and it is necessary to address a letter to the applicant asking further information, the application will not be processed until the information requested is received, and the application will be placed in the pending file to await the applicant's response.
- (1) When an application is placed in the pending file, the applicant will be notified of the reason for such action.

Note 1: No application tendered for filing after July 13, 1964, will be accepted for filing unless it complies fully with the provisions of new § 73.24(b) and new § 73.37 of this chapter, contained in the Commission's report and order, FCC 64-609 in Docket 15084, adopted July 1, 1964. No application ac-

cepted for filing after July 13, 1964, will be granted prior to August 13, 1964.

- NOTE 2: Pending the Commission's restudy of the rules pertaining to allocation of standard broadcast facilities, requests for authorizations will be considered as set forth in paragraphs (a) and (b) of this note, notwithstanding any provisions of this chapter to the contrary.
- (a) Applications for new standard broadcast stations or for major changes in the facilities of existing stations, if otherwise acceptable under Commission rules, will be accepted for filing only in the following categories:
- (1) Applications mutually exclusive with the renewal of license applications of existing stations.
- .(2) Applications, filed timely in response to, an appropriate cutoff notice, which involve a direct conflict necessitating a hearing with applications tendered for filing on or before July 19, 1968, and subsequently accepted for filing.
- (3) Applications for major changes necessitated by causes beyond the control of an applicant, such as loss of a transmitter site, etc.
- (4) Applications by existing Class IV stations to increase daytime power, or to increase nighttime power to 250 watts, subject to the provisions of section 316 of the Communications Act of 1934, as amended, to existing international understandings to which the United States is a party, and to the restriction on successive applications contained in § 73.37(d).
- (5) Applications for new or changed facilities in the State of Alaska, where the proposed operation: (i) Would not involve overlap of groundwave signal-intensity contours with any existing station prohibited by § 73.37; (ii) if for nighttime facilities, would not cause objectionable skywave interference to any existing station under the provisions of § 73.182(o); and (iii) if for nighttime facilities, not more than 10 percent of the population within the normally protected nighttime contour of the proposed operation would suffer interference, or, if it would, that the proposed operation would be the first nighttime standard broadcast facility in the community or would bring a first nighttime primary standard broadcast service to at least 25 percent of the proposed nighttime primary service area or at least 25 percent of the population within that area.
- (b) Applications now pending and those accepted under (a), above, will be processed and acted upon under rules in force prior to the effective date of this note. Those tendered for filing after July 19, 1968, which do not meet the interim criteria, will be returned.

[28 F.R. 12434, Nov. 22, 1963, as amended at 29 F.R. 9499, July 11, 1964; 33 F.R. 10344, July 19, 1968; 34 F.R. 7966, May 21, 1969; 34 F.R. 14376, Sept. 13, 1969]

§ 1.572 Processing of television broadcast applications.

(a) Applications for television broadcast stations are divided into two groups.

(1) In the first group are applications for new stations or for major changes in the facilities of authorized stations. A major change is, in the case of stations authorized under Part 73 of this chapter, any change in frequency or station location, or any change in power or antenna location or height above average terrain (or combination thereof) which would result in a change of 50 percent or more of the area within the Grade B contour of the station; in the case of television translator stations authorized under Part 74 of this chapter it is any change in frequency (output channel), primary station (input channel) or principal community or communities: *Provided*, however, That, the Commission may, of facilities, advise the applicant that such application is considered to be one for a major change and therefore subject to the provisions of § 1.580.

(2) The second group consists of applications for licenses to cover construction permits and all other changes in the facilities of authorized stations.

(b) If an application is amended so as to effect a major change as defined in paragraph (a) (1) of this section or so as to result in an assignment or transfer of control which, in the case of an authorized station, would require the filing of an application therefor on FCC Form 314 or 315 (see § 1.540), § 1.580 will apply to such amended application.

(c) Applications for television stations will be processed as nearly as possible in

the order in which they are filed.

(d) Regardless of the number of applications filed for channels in a city or the number of assignments available in that city, those applications which are mutually exclusive, i.e., which request the same channel, will be designated for All other applications for channels will, if the applicants are duly qualified, receive grants. For example, if Channels 6, 13, 47, and 53 have been assigned to City X and there are pending two applications for Channel 6 and one application for each of the remaining channels, the latter three applications will be considered for grants without hearing and the two mutually exclusive applications requesting Channel 6 will be designated for hearing. If there are two pending applications for Channel 6

and two applications for Channel 13, separate hearings will be held.

(e) Where applications are mutually exclusive because the distance between their respective proposed transmitter sites is contrary to the station separation requirements set forth in § 73.610 of this chapter, said applications will be processed and designated for hearing at the time the application with the lower file number is reached for processing. In order to be considered mutually exclusive with a lower file number application, the higher file number application must have been accepted for filing at least one day before the lower file number application has been acted upon by the Commission.

[28 F.R. 12434, Nov. 22, 1963, as amended at 32 F.R. 11214, Aug. 2, 1967]

§ 1.573 Processing of FM and noncommercial educational FM broadcast applications.

(a) Applications for FM broadcast stations are divided into two groups.

- In the first group are applications for new stations or for major changes in the facilities of authorized stations, i.e., (i) any change in frequency, station location or class of station, and (ii) any change in power, antenna height above average terrain and/or antenna location. if the change or combination of changes results in a change of 50 percent or more in the area within the station's 1 my/m contour: Provided, however, That the Commission may, within 15 days after the tender for filing of any other application for modification of facilities, advise the applicant that such application is considered to be one for a major change and therefore is subject to the provisions of § 1.580.
- (2) The second group consists of applications for licenses and all other changes in the facilities of authorized stations.
- (b) If an application is amended so as to effect a major change as defined in paragraph (a) (1) of this section or so as to result in an assignment or transfer of control which, in the case of an authorized station, would require the filing of an application therefor on FCC Form 314 or 315 (see § 1.540), § 1.580 will apply to such amended application.
- (c) Except as provided in the note to this section, if, upon examination, the Commission finds that the public interest, convenience and necessity will be served by the granting of an application

for FM broadcast facilities (Class A, Class B, Class C or noncommercial educational), the same will be granted. If, on the other hand, the Commission is unable to make such a finding and it appears that a hearing may be required, the procedure set forth in § 1.593 will be followed.

Note: During further consideration of the matters and issues in Docket No. 14185 (pertaining to the revision of the FM broadcast rules), applications for noncommercial educational FM broadcast authorizations will be subject to the following procedures, notwithstanding any provision of the FM Broadcast rules or of this section to the contrary:

(a) Noncommercial educational stations. With respect to grant and (after Aug. 8, 1962) acceptance of applications for construction permits for new or changed facilities on the channels reserved in § 73.501 of this chapter for educational use, the following restrictions will apply:

(1) Applications for facilities on channels 218, 219, and 220 must meet the following

criteria:

(1) The facilities requested must not exceed the maximum facilities specified in § 73.211 of this chapter (as amended July 25, 1963, effective Sept. 11, 1963) for Class B or Class C commercial stations, depending on the zone in which the requested facilities would be located.

(ii) The requested facilities must be located, with respect to any assignment on FM Channels 221, 222, and 223 specified in the table of assignments in § 73.202 of this chapter at no less than the minimum mileage separations specified for stations of their class in §§ 73.207 and 73.504 of this chapter.

(iii) Where the application is for change in transmitter site, the move must not shorten the separation between the station and other cochannel and adjacent-channel stations, if the result would be a spacing less than that specified in § 73.207 of this chapter (as amended July 25, 1963, effective

Sept. 11, 1963).

(2) No application for facilities on any channel specified in § 73.501 of this chapter will be granted or accepted if the facilities requested would cause objectionable interference within the 1 mv/m contour of any cochannel or adjacent-channel station, or receive interference within the proposed 1 mv/m contour. The following standards shall be used to determine the existence of objectionable interference:

(i) The distance to the 1 my/m contour shall be determined by use of Figure 1 of § 78.333 of this chapter (as amended July 25,

1962, effective Sept. 10, 1962).

(ii) The distance to the applicable interference contour shall be determined by the F(50, 10) curve published with the Commission's Order, FCC 61-1447, adopted December 6, 1961, setting forth the interim procedure for processing FM applications.

(iii) Objectionable interference will be considered to exist where, on the basis of the curves referred to in this subparagraph, the undesired signal of a cochannel signal exceeds one-tenth of the desired signal, the undesired signal of a station 200 kc/s removed exceeds one-half of the desired signal, the undesired signal of a station 400 kc/s removed exceeds 10 times the desired signal, or the undesired signal of a station 600 kc/s removed exceeds 100 times the desired signal.

(3) Directional antennas: No application for construction permit for a new station, change in channel, or change in existing facilities on the same channel will be granted or accepted for filing where it proposes a directional antenna with a maximum-to-

minimum ratio of more than 15 db.

(b) Maximum and minimum facilities for stations on noncommercial channels. No provisions of these rules as to minimum facilities shall apply to noncommercial educational stations operating on the channels specified in § 73.501 of this chapter, or to grant or acceptance of applications by any existing station for increase in facilities on its present channel; and no provisions as to maximum facilities shall apply to noncommercial educational stations on Channels 201 to 217, inclusive, set forth in § 73.501 of this chapter. [28 F.R. 12434, Nov. 22, 1963, as amended at 32 F.R. 20861, Dec. 28, 1967]

§ 1.574 Processing of international broadcast applications.

- (a) Applications for international broadcast facilities are divided into two groups.
- (1) In the first group are applications for new stations for new or additional target zones, or for major changes in the facilities of authorized stations, i.e., a substantial change (other than local) in transmitter location or a significant change in the delivered median field intensity at the target zone: Provided, however, That the Commission may, within 15 days after the tender for filing of any other application for modification, advise the applicant that such application is considered to be one for a major change and therefore is subject to the provisions of § 1.580.
- (2) The second group consists of applications for licenses and all other changes in the facilities of authorized stations.
- (b) If an application is amended so as to effect a major change as defined in paragraph (a) (1) of this section, or so as to result in an assignment or transfer of control which, in the case of an authorized station, would require the filing of an application therefor on FCC Form 314 or 315 (see § 1.540), § 1.580 will apply to such amended application.

(c) Applications for international broadcast stations will be processed as nearly as possible in the order in which they are filed.

Note: Pending rule making pertaining to the authorization of international broadcast stations, no application seeking authority to construct a new international broadcast station or seeking authority to operate a greater number of frequency hours than authorized on April 25, 1963, will be granted, and, after April 25, 1963, no such application will be accepted for filing.

§ 1.578 Amendments to applications for renewal, assignment and/or transfer of control.

- (a) Any amendment to an application for renewal of any instrument of authorization shall be considered to be a minor amendment, except that any amendment which seeks to change the proposals contained therein relating to future programming of a station shall be considered to be a major amendment: Provided, however, That the Commission may, within 15 days after tender for filing of any other amendment, advise the applicant that the amendment is considered to be a major amendment and therefore is subject to the provisions of § 1.580.
- (b) Any amendment to an application for assignment of construction permit or license, or consent to the transfer of control of a corporation holding such a construction permit or license, shall be considered to be a minor amendment. except that any amendment which seeks a change in the ownership interest of the proposed assignee or transferee which would result in a change in control, or any amendment which would require the filing of FCC Forms 314 or 315 (see § 1.540) if the changes sought were made in an original application for assignment or transfer of control, shall be considered to be a major amendment: Provided, however, That the Commission may, within 15 days after the tender for filing of any other amendment, advise the applicant that the amendment is considered to be a major amendment and therefore is subject to the provisions of § 1.580.

§ 1.580 Local notice of the filing of broadcast applications, and timely filing of petitions to deny them.

(a) All applications for instruments of authorization in the broadcast service (and major amendments thereto, as indicated in §§ 1.571, 1.572, 1.573, 1.574,

- and 1.578) are subject to the provisions of this section, except applications for:
- (1) A minor change in the facilities of an authorized station, as indicated in §§ 1.571-1.574.
- (2) Consent to an involuntary assignment or transfer under section 310(b) of the Communications Act of 1934, as amended, or to a voluntary assignment or transfer thereunder which does not result in a change of control and which may be applied for on FCC Form 316 pursuant to the provisions of § 1.540(b).
- (3) A license under section 319(c) of the Communications Act or, pending application for or grant of such license, any special or temporary authorization to permit interim operation to facilitate completion of authorized construction or to provide substantially the same service as would be authorized by such license.
- (4) Extension of time to complete construction of authorized facilities.
- (5) An authorization of facilities for remote pickup or studio links for use in the operation of a broadcast station.
- (6) Authorization pursuant to section 325(b) of the Communications Act where the programs to be transmitted are special events not of a continuing nature.
- (7) An authorization under any of the proviso clauses of section 308(a) of the Communications Act.
- (b) No application subject to the provisions of this section (whether as originally filed or as amended) will be acted upon by the Commission less than 30 days following issuance of public notice of the acceptance for filing of such application or amendment: Provided, however, That the Commission, notwith-standing the requirements of this sec-tion, may, if the grant of such application is otherwise authorized by law and if it finds that there are extraordinary circumstances requiring emergency operations in the public interest and that delay in the institution of such emergency operations would seriously prejudice the public interest, grant a temporary authorization accompanied by a statement of its reasons therefor, to permit such emergency operations for a period not exceeding 90 days, and upon making like findings may extend such temporary authorization for one additional period not to exceed 90 days.
- (c) Except as otherwise provided in paragraph (e) of this section, an applicant filing any application or an amendment thereto which is subject to the provisions of this section (except for appli-

cations for stations in the international broadcast service and for television translator stations) shall cause to be published a notice of such filing as follows: Notice shall be published at least twice a week for 2 consecutive weeks within the 3-week period immediately following the tendering for filing of such application or amendment, or at least twice a week for 2 consecutive weeks within the 3-week period immediately following notification by the Commission pursuant to § 1.571, § 1.572, § 1.573, or § 1.578, in a daily newspaper of general circulation published in the community in which the station is located or proposed to be located: Provided, however, That if there is no such daily newspaper published in the community, the notice shall be published as follows:

(1) If one or more weekly newspapers of general circulation are published in the community in which the station is located or proposed to be located, notice shall be published in such a weekly newspaper once a week for 3 consecutive weeks within the 4-week period immediately following the tendering for filing of such application or amendment, or once a week for 3 consecutive weeks within the 4-week period immediately following notification by the Commission pursuant to § 1.571, § 1.572, § 1.573, or § 1.578.

(2) If no weekly newspaper of general circulation is published in the community in which the station is located or proposed to be located, notice shall be published at least twice a week for 2 consecutive weeks within the 3-week period immediately following the tendering for filing of such application or amendment, or at least twice a week for the 2 consecutive weeks within the 3following immediately period notification by the Commission pursuant to § 1.571, § 1.572, § 1.573, or § 1.578, in the daily newspaper having the greatest general circulation in the community in which the station is located or proposed to be located:

And provided further: That in the case of an application for a permit pursuant to section 325(b) of the Communications Act, the notice shall be published at least twice a week for 2 consecutive weeks within the 3-week period immediately following the tendering for filling of such application, or at least twice a week for 2 consecutive weeks within the 3-week period immediately following notification by the Commission pursuant to § 1.571, § 1.572, § 1.573, or § 1.578, in a

daily newspaper of general circulation in the largest city in the principal area to be served in the United States by the foreign radio broadcast station:

And provided further. That in the case of an application for change in the location of a station, the notice shall be published both in the community in which the station is located and in the community in which the station is proposed to be located.

And provided further, That in the case of applications for the renewal of station licenses, but not amendments thereof, notice shall be published prior to filing as prescribed in paragraph (m) of this section, instead of after filing, as prescribed in this paragraph.

(d) If the application seeks modification, assignment or transfer of an operating broadcast station (except for applications for stations in the international broadcast service and for television translator stations), or is an amendment of an application for renewal of a broadcast station license, the applicant shall, in addition to publishing a notice of such filing as provided in paragraph (c) of this section, cause the same notice to be broadcast over that station at least once daily on 4 days in the second week immediately following the tendering for filing of such application, or in the second week immediately following notification by the Commission pursuant to § 1.571, § 1.572, § 1.573, or § 1.578. In the case of applications for the renewal of broadcast station licenses, but not amendment thereof, notice shall be broadcast at least once daily on 4 days of any single week starting not more than 45 days prior to the due date for filing the renewal application. In the case of television broadcast stations and noncommercial educational television broadcast stations, such notice shall be broadcast orally with camera focused on the announcer. The notice required by this paragraph shall be broadcast during the following periods:

(1) For television broadcast stations, between 7:00 p.m. and 10:00 p.m.

(2) For standard and FM broadcast stations, between 7:00 a.m. and 10:00 a.m., but if such stations do not operate during those hours, then between 6:00 p.m. and 9:00 p.m.

(3) For noncommercial educational television broadcast stations, between 7:00 p.m. and 10:00 p.m., but if the period of broadcast of notice falls within

a portion of the year during which such stations do not broadcast, then such stations need not comply with the provisions of this paragraph.

- (4) For noncommercial educational FM broadcast stations and standard broadcast stations operating as educational stations, between 3:00 p.m. and 10:00 p.m., but if the period of broadcast of notice falls within a portion of the year during which such stations do not broadcast, then such stations need not comply with the provisions of this paragraph.
- (e) If the station in question is the only operating station in its broadcast service which is located in the community involved, or if it is a noncommercial educational station (FM or television) or a standard broadcast station operating as a noncommercial educational station. publication of the notice in a newspaper. as provided in paragraph (c) of this section, is not required, and publication by broadcast over that station as provided in paragraph (d), shall be deemed sufficient to meet the requirements of paragraphs (c) and (d) of this section: Provided, however, That noncommercial educational FM broadcast stations, noncommercial educational television broadcast stations, and standard broadcast stations operating as noncommercial educational stations which do not broadcast during the portion of the year in which the period of broadcast of notice falls must comply with the provisions of paragraph (c) of this section.
- (f) The notice required by paragraphs (c) and (d) of this section shall contain the following information, except as otherwise provided in paragraph (m) of this section in the case of license renewal applications:
- (1) The name of the applicant, if the applicant is an individual; the names of all partners, if the applicant is a partnership; or the names of all officers and directors and of those persons holding 10 percent or more of the capital stock or other ownership interest if the applicant is a corporation or an unincorporated association. (In the case of applications for assignment or transfer of control, information should be included for all parties to the application.)
- (2) The purpose for which the application was filed (i.e., construction permit, modification, transfer or assignment of control, renewal, etc.).

- (3) The date when the application or amendment was tendered for filing with the Commission.
- (4) The call letters, if any, of the station, and the frequency or channel on which the station is operating or proposes to operate.
- (5) In the case of an application for construction permit for a new station, the facilities sought, including type and class of station, power, location of studios, transmitter site and antenna height.
- (6) In the case of an application for modification of a construction permit or license, the exact nature of the modification sought.
- (7) In the case of an amendment to an application, the exact nature of the amendment.
- (8) In the case of applications for a permit pursuant to section 325(b) of the Communications Act, the call letters and location of the foreign radio broadcast station, the frequency or channel on which it operates, and a description of the programs to be transmitted over the station.
 - (9) [Reserved]
- (10) A statement that a copy of the application, amendment(s), and related material are on file for public inspection at a stated address in the community in which the main studio is maintained or is proposed to be located. See § 1.526.
- (g) An applicant filing any application or an amendment thereto for a television broadcast translator station which is subject to this section shall cause to be published a notice of such filing at least once during the two week period immediately following the tendering for filing of such application or major amendment, or, when an applicant is specifically advised by the Commission that public notice is required in a particular case pursuant to § 1.572, such notice shall be published at least once during the two week period immediately following Commission notification, in a daily, weekly or bi-weekly publication having general circulation in the community or area to be served: Provided. however, That, if there is no publication of general circulation in the community or area to be served, the applicant shall determine an appropriate means of providing the required notice to the general public, such as posting in the local post office or other public place. The notice shall state:

(1) The name of the applicant, the community or area to be served, and the transmitter site.

(2) The purpose for which the application was filed (whether the application is for a new translator station, for authority to make changes in an existing translator station, for assignment or transfer of control, renewal, etc.).

(3) The date when the application or amendment was filed with the

Commission.

(4) The output channel or channels on which the station is operating or proposes to operate and the power used or proposed to be used.

(5) In the case of an application for changes in authorized facilities, the na-

ture of the changes sought.

(6) In the case of a major amendment to an application, the nature of the amendment.

- (7) A statement that the station engages in or intends to engage in rebroadcasting, and the call letters, location and channel of operation of each station whose signals it is rebroadcasting or intends to rebroadcast.
- (h) Within 7 days of the last day of publication or broadcast of the notice required by paragraphs (c), (d), or (g) of this section, the applicant shall file a statement in triplicate with the Commission, setting forth the dates on which the notice was published, the newspaper in which the notice was published, the text of the notice, and/or, where applicable, the dates and times that the notice was broadcast and the text thereof. When public notice is given by other means, as provided in paragraph (g) of this section, the applicant shall file, within 7 days of the giving of such notice, the text of the notice, the means by which it was accomplished, and the date thereof.
- (i) Any party in interest may file with the Commission a petition to deny any such application (whether as originally filed or amended) no later than 30 days after issuance of a public notice of the acceptance for filing of any such application or amendment thereto: Provided, however, That in the case of applications for standard broadcast facilities, petitions to deny may be filed at any time prior to the day of Commission grant thereof without hearing or the day of formal designation thereof for hearing; but where the Commission issues a public notice pursuant to the provisions of § 1.571(c) listing standard broadcast ap-

plications as available and ready for processing, no petitions to deny any such listed application will be accepted after the "cutoff" date specified in the public notice: And provided further, That in the case of applications for renewal of license, petitions to deny may be filed at any time up to the last day for filing mutually exclusive applications under § 1.516(e). Petitions to deny shall contain specific allegations of fact sufficient to show that the petitioner is a party, in interest and that a grant of the application would be prima facie inconsistent with the public interest, convenience, and necessity. Such allegations of fact shall, except for those of which official notice may be taken, be supported by affidavit of a person or persons with personal knowledge thereof.

(j) The applicant may file an opposition to any petition to deny, and the petitioner a reply to such opposition (see § 1.45) in which allegations of fact or denials thereof shall be supported by affidavit of a person or persons with per-

sonal knowledge thereof.

(k) (1) The provisions of sections 309 and 311(a) of the Communications Act, as amended by Public Law 86-752, shall not be applicable to applications for assignments of licenses or construction permits or for transfers of control of corporate licensees or permittees which are pending before the Commission and which have not been designated for hearing prior to December 12, 1960.

- (2) Any other applications in the categories subject to this section which are pending before the Commission and which have not been designated for hearing before December 12, 1960, will not be acted on by the Commission until at least 30 days after that date. Any party in interest may file a petition to deny such applications pursuant to the provisions of this section. The provisions of section 309 of the Communications Act, as in effect prior to December 12, 1960, shall not be applicable to such applications. However, such applications shall not be subject to the provisions of paragraph (c) of this section.
 - (l) [Reserved]
- (m) (1) Paragraphs (a) through (k) of this section apply to applications for the renewal of station licenses except that:
- (i) Notices required under paragraphs (c), (d), and (g) of this section shall be given to the public, in the prescribed manner, and at the prescribed times dur-

ing the prescribed number of weeks, but during the 6-week period preceding the date specified in § 1.539(a) for the timely filing of license renewal applications, instead of after the application is filed.

(ii) The information contained in the public notice prescribed in paragraphs (f) and (g) of this section shall reflect the prospective (rather than the previous) filing of the license renewal application.

(iii) Notices for stations subject to paragraphs (c) and (d) of this section shall include the following statement, in addition to the information required under paragraph (f) (1) and (4) of this section.

The application of this station for renewal of its license to operate in the public interest is required to be filed with the Federal Communications Commission no later than (insert here the date prescribed in § 1.539(a)). Members of the public who desire to bring to the Commission's attention facts concerning the operation of this station should write to the Federal Communications Commission, Washington, D.C. 20554, not later than (insert here the date 30 days after the last day for timely filing of the license renewal application). Letters should set out in detail the specific facts which the writer wishes the Commission to consider in passing on the application.

A copy of the license renewal application and related material will, upon filing with the Commission, be available for public inspection at (state here the address where station records are made available for public inspection as required by § 1.526(d)) between the hours of _____ and ____ (Regular

business hours.)

(iv) The statement containing the information prescribed in paragraph (h) of this section shall be filed with the license renewal application.

(2) Paragraphs (a) through (k) of this section apply, without change, to major amendments to license renewal applications, to which § 1.578(a) applies. (Sec. 5(a), 74 Stat. 892; 47 U.S.C. 311) [28 F.R. 12434, Nov. 22, 1963, as amended at 30 F.R. 4547, Apr. 8, 1965; 30 F.R. 6781, May 19, 1965; 34 F.R. 7966, May 21, 1969]

§ 1.587 Procedure for filing informal objections.

Before Commission action on any application for an instrument of authorization, other than a license pursuant to a construction permit, any person may file informal objections to the grant. Such objections shall be signed by the objector. The limitation on pleadings and time for filing pleadings provided for

in § 1.45 shall not be applicable to any objections duly filed under this section.

ACTION ON APPLICATIONS

- § 1.591 Grants without hearing of authorizations other than licenses pursuant to construction permits.
- (a) In the case of any application for an instrument of authorization other than a license pursuant to a construction permit, the Commission will make the grant if it finds (on the basis of the application, the pleadings filed, or other matters which it may officially notice) that the application presents no substantial and material question of fact and meets the following requirements:
- (1) There is not pending a mutually exclusive application filed in accordance with paragraph (b) of this section;
- (2) The applicant is legally, technically, financially, and otherwise qualified;
- (3) The applicant is not in violation of provisions of law or this chapter or established policies of the Commission;
- (4) A grant of the application would otherwise serve the public interest, convenience, and necessity.
- (b) In making its determinations pursuant to the provisions of paragraph (a) of this section, the Commission will not consider any other application, or any other application if amended so as to require a new file number, as being mutually exclusive or in conflict with the application under consideration unless such other application was substantially complete and tendered for filing by whichever date is earlier: (1) The close of business on the day preceding the day on which the Commission takes action with respect to the application under consideration; or (2) the close of business on the day preceding the day designated by public notice in the FEDERAL REGISTER as the day the application under consideration is available and ready for processing; or (3) the date prescribed in § 1.516(e) in the case of applications which are mutually exclusive with applications for renewal of license of broadcast stations.

Note: As used in paragraph (b) (1) of this section, the "day on which the Commission takes action" is the day on which the Commission votes to grant the prior-filed application or to designate it for hearing or, in the case of action by delegated authority, the day on which the person acting under such delegated authority signs the document by

which such action is taken. The "release date" of the document and the "effective date" of the action (see § 1.102) are significant in other respects but are not considered in making the determination under paragraph (b) (1) of this section as to whether an application was filed by "the close of business on the day preceding the day on which the Commission takes action."

NOTE: Paragraph (b)(2) of this section applies only to standard broadcast applications for new stations or for major changes in the facilities of authorized stations. See also §§ 1.227(b)(1) and 1.571 (c) and (h).

(c) If a petition to deny the application has been filed in accordance with § 1.580 and the Commission makes the grant in accordance with paragraph (a) of this section, the Commission will deny the petition and issue a concise statement setting forth the reasons for denial and disposing of all substantial issues raised by the petition.

(Sec. 309, 48 Stat. 1085, as amended; 47 U.S.C. 309) [28 F.R. 12434, Nov. 22, 1963, as amended at 34 F.R. 7966, May 21, 1969; 34 F.R. 14469, Sept. 17, 1969]

§ 1.592 Conditional grant.

- (a) Where a grant of an application would preclude the grant of any application or applications mutually exclusive with it, the Commission may, if the public interest will be served thereby, make a conditional grant of one of the applications and designate all of the mutually exclusive applications for hearing. Such conditional grant will be made upon the express condition that such grant is subject to being withdrawn if, at the hearing, it is shown that public interest will be better served by a grant of one of the other applications. Such conditional grants will be issued only where it appears:
- (1) That some or all of the applications were not filed in good faith but were filed for the purpose of delaying or hindering the grant of another application; or
- (2) That public interest requires the prompt establishment of broadcast service in a particular community or area; or
- (3) That a grant of one or more applications would be in the public interest, and that a delay in making a grant to any applicant until after the conclusion of a hearing on all applications might jeopardize the rights of the United States under the provisions of international agreement to the use of the frequency in question; or

(4) That a grant of one application would be in the public interest, and that it appears from an examination of the remaining applications that they cannot be granted because they are in violation of provisions of the Communications Act, other statutes, or the provisions of this chapter.

(b) When two or more applications for the same television assignment have been designated for hearing, the Commission may, if the public interest will be served thereby, make a conditional grant to a group composed of any two or more of the competing applicants, such grant to terminate when the successful applicant commences operation under the terms of a regular authorization. No conditional grant will be made unless all of the competing applicants have been afforded a reasonable opportunity to participate in the group seeking the conditional grant. In its application, the group shall include a special showing as to the need for the service pending operation by the successful applicant under the terms of a regular authorization; the effect, if any, of a grant on the position of any applicant. which is not a member of the group; and any other factors which are deemed pertinent to the public interest judgment.

§ 1.593 Designation for hearing.

If the Commission is unable, in the case of any application for an instrument of authorization, to make the findings specified in § 1.591(a), it will formally designate the application for hearing on the grounds or reasons then obtaining and will forthwith notify the applicant and all known parties in interest of such action and the grounds and reasons therefor, specifying with particularity the matters and things in issue but not including issues or requirements phrased generally.

(Sec. 309, 48 Stat. 1085, as amended: 47 U.S.C. 309)

§ 1.594 Local notice of designation for hearing.

(a) Except as otherwise provided in paragraph (c) of this section, when an application subject to the provisions of § 1.580 (except for applications for stations in the international broadcast service and for television translator stations) is designated for hearing, the applicant shall cause to be published a notice of such designation as follows: Notice shall be published at least twice a week, for 2:

consecutive weeks within the 3-week period immediately following release of the Commission's order specifying the time and place of the commencement of the hearing, in a daily newspaper of general circulation published in the community in which the station is located or proposed to be located: *Provided, however*, That if there is no such daily newspaper published in the community, the notice shall be published as follows:

(1) If one or more weekly newspapers of general circulation are published in the community in which the station is located or proposed to be located, notice shall be published in such a weekly newspaper once a week for 3 consecutive weeks within the 4-week period immediately following the release of the Commission's order specifying the time and place of the commencement of the hearing;

(2) If no weekly newspaper of general circulation is published in the community in which the station is located or proposed to be located, notice shall be published at least twice a week for 2 consecutive weeks within the 3-week period immediately following the release of the Commission's order specifying the time and place of the commencement of the hearing in the daily newspaper having the greatest general circulation in the community in which the station is located or proposed to be located:

And provided further, That in the case of an application for a permit pursuant to section 325(b) of the Communications Act, the notice shall be published at least twice a week for 2 consecutive weeks within the 3-week period immediately following release of the Commission's order specifying the time and place of the commencement of the hearing in a daily newspaper of general circulation in the largest city in the principal area to be served in the United States by the foreign radio broadcast station:

And provided further, That in the case of an application for change in the location of a station, the notice shall be published both in the community in which the station is located and in the community in which the station is proposed to be located.

(b) When an application which is subject to the provisions of § 1.580 and which seeks modification, assignment, transfer, or renewal of an operating broadcast station is designated for hearing (except for applications for stations in the international broadcast service

and for television translator stations), the applicant shall, in addition to publishing a notice of such designation as provided in paragraph (a) of this section, cause the same notice to be broadcast over that station at least once daily on 4 days in the second week immediately following the release of the Commission's order specifying the time and place of the commencement of the hearing. In the case of television broadcast stations and noncommercial educational television broadcast stations, such notice shall be broadcast orally with camera focused on the announcer. The notice required by this paragraph shall be broadcast during the following periods:

(1) For television broadcast stations, between 7:00 p.m. and 10:00 p.m.

(2) For standard and FM broadcast stations, between 7:00 a.m. and 10:00 a.m., but if such stations do not operate during those hours, then between 6:00 p.m. and 9:00 p.m.

(3) For noncommercial educational television broadcast stations, between 7:00 p.m. and 10:00 p.m., but if the period of broadcast of notice falls within a portion of the year during which such stations do not broadcast, then such stations need not comply with the provisions of this paragraph.

(4) For noncommercial educational FM broadcast stations and standard broadcast stations operating as educational stations, between 3:00 p.m. and 10:00 p.m., but if the period of broadcast of notice falls within a portion of the year during which such stations do not broadcast, then such stations need not comply with the provisions of this

paragraph. (c) If the station in question is the only operating station in its broadcast service which is located in the community involved, or if it is a noncommercial educational station (FM or television) or a standard broadcast station operating as a noncommercial educational station, publication of the notice in a newspaper, as provided in paragraph (a) of this section, is not required, and publication by broadcast over that station as provided in paragraph (b) shall be deemed sufficient to meet the requirements of paragraphs (a) and (b) of this section: Provided, however, That noncommercial educational FM broadcast stations, noncommercial educational television broadcast stations, and standard broadcast stations operating as noncommercial educational stations which do not broadcast during the portion of the year in which the period of broadcast of notice falls must comply with the provisions of paragraph (a) of this section.

- (d) The notice required by paragraphs(a) and (b) of this section shall set forth:
- (1) The name of the applicant or applicants designated for hearing.
- (2) The call letters, if any, of the station or stations involved, and the frequencies or channels on which the station or stations are operating or proposed to operate.
 - (3) The time and place of the hearing.
- (4) The issues in the hearing as listed in the Commission's order of designation for hearing.
- (5) A statement that a copy of the application, amendment(s), and related material are on file for public inspection at a stated address in the community in which the main studio is maintained or is proposed to be located. See § 1.526.
- (e) When an application for renewal of license is designated for hearing, the notice shall contain the following additional statements:
- (1) Immediately preceding the listing of the issues in the hearing:

The application of this station for a renewal of its license to operate this station in the public interest was tendered for filing with the Federal Communications Commission on

After considering this application, the Commission has determined that it is necessary to hold a hearing to decide the following questions:

(2) Immediately following the listing of the issues in the hearing:

The hearing will be held at _____, commencing at ____, on ____, 19__. Members of the public who desire to give evidence concerning the foregoing issues should write to the Federal Communications Commission, Washington, D.C., 20554, not later than _____. Letters should set forth in detail the specific facts concerning which the writer wishes to give evidence. If the Commission believes that the evidence is legally competent, material, and relevant to the issues, it will contact the person in question.

NOTE: In subparagraph (2) of this paragraph, the applicant shall insert, as the date on or before which members of the public who desire to give evidence should write to the Commission, the date 30 days after the date of release of the Commission's order specifying the time and place of the commencement of the hearing.

- (f) When an application for a television broadcast translator station which is subject to the provisions of § 1.580 is designated for hearing, the applicant shall cause to be published a notice of such designation as follows: Notice shall be published at least once during the 2-week period immediately following release of the Commission's order specifying the time and place of the commencement of the hearing in a daily. weekly or biweekly publication having general circulation in the community or area to be served: Provided. however. That, if there is no publication of general circulation in the community or area to be served, the applicant shall determine an appropriate means of providing the required notice to the general public, such as posting in the local post office or other public place. The notice shall state:
- (1) The name of the applicant or applicants designated for hearing.
- (2) The call letters, if any, of the station or stations involved, the output channel or channels of such stations, and the call letters, channel and location of the station or stations being or proposed to be rebroadcast.
 - (3) The time and place of the hearing.
- (4) The issues in the hearing as listed in the Commission's order of designation for hearing.
- (5) If the application is for renewal of license, the notice shall contain, in addition to the information required by subparagraphs (1) through (4) of this paragraph, the statements required by paragraph (e) of this section.
- (g) Within 7 days of the last day of publication or broadcast of the notice required by paragraphs (a) and (b) of this section, the applicant shall file a statement in triplicate with the Commission. setting forth the dates on which the notice was published, the newspaper in which the notice was published, the text of the notice, and/or, where applicable, the date and time the notice was broadcast and the text thereof. When public notice is given by other means, as provided in paragraph (f) of this section, the applicant shall file, within 7 days of the giving of such notice, the text of the notice, the means by which it was accomplished, and the date thereof.
- (h) The failure to comply with the provisions of this section is cause for dismissal of an application with prejudice. However, upon a finding that applicant

has complied (or proposes to comply) with the provisions of section 311(a) (2) of the Communications Act of 1934, as amended, and that the public interest, convenience and necessity will be served thereby, the presiding officer may authorize an applicant, upon a showing of special circumstances, to publish notice in a manner other than that prescribed by this section; may accept publication of notice which does not conform strictly in all respects with the provisions of this section; or may extend the time for publishing notice.

(Sec. 5(a), 74 Stat. 892; 47 U.S.C. 311) [28 F.R. 12434, Nov. 22, 1963, as amended at 29 F.R. 6444, May 16, 1964; 30 F.R. 4547, Apr. 8, 1965; 30 F.R. 6781, May 19, 1965]

§ 1.597 Procedures on transfer and assignment applications.

- (a) If, upon the examination, pursuant to sections 309(a) and 310(b) of the Communications Act of 1934, as amended, of an application for Commission consent to an assignment of a broadcast construction permit or license or for a transfer of control of a corporate permittee or licensee, it appears that the station involved has been operated by the proposed assignor or transferor for less than three successive years, the application will be designated for hearing on appropriate issues pursuant to section 309(b) of the Communications Act of 1934, as amended, unless the Commission is able to find that:
- (1) The application involves a translator station only, or an FM station operated for at least three years together with a Subsidiary Communications Authorization held for a lesser period; or
- (2) The application involves a proforma assignment or transfer of control;
- (3) The assignor or transferor has made an affirmative factual showing, supported by affidavits of a person or persons with personal knowledge thereof, which establishes that (due to unavailability of capital, to death or disability of station principals, or to other changed circumstances affecting the licensee or permittee occurring subsequent to the acquisition of the license or permit) Commission consent to the proposed assignment or transfer of control will serve the public interest, convenience and necessity.
- (b) The commencement date of the three-year period set forth in paragraph

(a) of this section shall be determined as follows:

(1) Where the authorizations involved in the application consist of a license and a construction permit authorizing a major change in the facilities of the licensed station (as defined in §§ 1.571. 1.572, and 1.573), the three-year period shall commence with the date of the Commission's grant of the construction permit for the modification. However, when operating authority has been issued to cover the construction permit for a major change in facility, the commencement date for calculating the length of time the station has been operated for purposes of this section shall then revert to the date the licensee received its original operating authority. A grant of authority for minor modifications in authorized facilities shall have no effect upon the calculation of this time period.

(2) Where the authorization involved in the application consists of a permit authorizing the construction of a new facility, or of a license covering such permit, the three-year period shall commence with the date of issuance of ini-

tial operating authority.

(3) Where the operating station involved in the application was obtained by means of an assignment or transfer of control (other than pro forma), the three-year period shall commence with the date of grant by the Commission of the application for said assignment or transfer of control. If the station was put in operation after such assignment or transfer, subparagraphs (1) and (2)

of this paragraph shall apply.

- (4) Where an application is filled for Commission consent to a transfer of control of a corporation holding multiple licenses and/or construction permits, the commencement date applicable to the last-acquired station shall apply to all the stations involved in the transfer, except where the application involves an FM station operated for less than three years and an AM station operated for more than three years, both serving substantially the same area. Said exception shall apply to the same circumstances where assignment applications are involved.
- (c) In determining whether a broadcast interest has been held for three years, the Commission will calculate the period between the date of acquisition (as specified in paragraph (b) of this section) and the date the application for

transfer or assignment is tendered for

filing with the Commission.

(d) With respect to applications filed after the three-year period, the Chief of the Broadcast Bureau is directed (1) to examine carefully such applications, on a case-to-case basis, to determine whether any characteristics of trafficking remain; and (2) if so, to seek additional information, by letter inquiries to the applicants, such as that which will be required to be developed and tested in the hearing process with respect to stations held less than three years.

(e) (1) As used in paragraphs (e) and

(f) of this section:

(i) "Unbuilt station" refers to a standard, FM or television broadcast station for which a construction permit is outstanding, and, regardless of the stage of physical completion, for which Program Test Authority has not been issued.

(ii) "Seller" includes the assignor(s) of a construction permit for an unbuilt station, the transferor(s) of control of the holder of such construction permit, and any principal of such assignor(s) or transferor(s) who retains an interest in the permittee or acquires or reacquires such interest within 1 year after the issuance of Program Test Authority.

(2) The Commission will not consent to the assignment or transfer of control of the construction permit of an unbuilt station if the agreements or understandings between the parties provide for, or permit, payment to the seller of a sum in excess of the aggregate amount clearly shown to have been legitimately and prudently expended, and to be expended, by the seller solely for preparing, filing, and advocating the grant of the construction permit for the station, and for other steps reasonably necessary toward placing the station in operation.

(3) (i) Applications for consent to the assignment of construction permit or transfer of control shall, in the case of unbuilt stations, be accompanied by declarations both by the assignor (or and by the assignee transferor) transferee) that-except as clearly disclosed in detail in the applicationsthere are no agreements, arrangements. or understandings for reimbursement of the seller's expenses or other payments to the seller, for the seller's retention of any interest in the station, for options or any other means by which the seller may acquire such an interest, or for any other actual or potential benefit to the seller in the form of loans, the subsequent

repurchase of the seller's retained interest, or otherwise.

(ii) When the seller is to receive reimbursement of his expenses, the applications of the parties shall include an itemized accounting of such expenses, together with such factual information as the parties rely upon for the requisite showing that those expenses represent legitimate and prudent outlays made solely for the purposes allowable under subparagraph (2) of this paragraph.

(f) (1) Whenever an agreement for the assignment of the construction permit of an unbuilt station or for the transfer of control of the permittee of an unbuilt station, or any arrangement or understanding incidental thereto, provides for the retention by the seller of any interest in the station, or for any other actual or potential benefit to the seller in the form of loans or otherwise, the question is raised as to whether the transaction involves actual or potential gain to the seller over and above the legitimate and prudent out-of-pocket expenses allowable under paragraph (e)(2) of this section. In such cases the Commission will designate the assignment or transfer applications for evidentiary hearing: Provided, That a hearing is not mandatory in cases coming within subparagraph (2) of this paragraph.

(2) It is not intended to forbid the seller to retain an equity interest in an unbuilt station which he is transferring or assigning if the seller obligates himself, for the period ending 1 year after the issuance of Program Test Authority, to provide that part of the total capital made available to the station, up to the end of that period, which is proportionate to the seller's equity share in the permittee, taking into account equity capital, loan capital, and guarantees of interest and amortization payments for the transfer or assignment. This condition will be satisfied:

(i) In the case of equity capital: By paid-in cash capital contributions proportionate to the seller's equity share;

(ii) In cases where any person who has an equity interest in the permittee provides loan capital: By the seller's provision of that part of the total loan capital provided by equity holders which is proportionate to the seller's equity share; and

(iii) In cases where any person cosigns or otherwise guarantees payments under notes given for loan capital pro-

vided by nonequity holders: By similar guarantees by the seller covering that part of such payments as is proportionate to the seller's equity share: Provided, That this condition shall not be deemed to be met if the guarantees given by persons other than the seller cover, individually or collectively, a larger portion of such payments than the ratio of the combined equities of persons other than the seller to the total equity.

(3) In cases which are subject to the requirements of subparagraph (2) (i),

(ii), and (iii) of this paragraph:

(i) The assignee's (or transferee's) application shall include a showing of the anticipated capital needs of the station through the first year of its operation and the seller's financial capacity to comply with the above requirements, in the light of such anticipated capital needs.

(ii) The Commission will determine from its review of the applications whether a hearing is necessary to ensure compliance with the above requirements.

(iii) Compliance with the above requirements will be subject to review by the Commission at any time, either when considering subsequently filed applications or whenever the Commission may otherwise find it desirable.

(iv) Within 30 days after any time when a seller is required to provide equity or loan capital or execute guarantees, the permittee shall furnish the Commission a written report containing sufficient details as to the sources and amounts of equity capital paid in, loan capital made available, or guarantees obtained as to enable the Commission to ascertain compliance with the above requirements.

(v) No steps shall be taken by the permittee to effectuate arrangements for the provision of equity or loan capital from sources not previously identified and disclosed to the Commission, until 30 days after the permittee has filed with the Commission a report of such arrangements and of provisions made for the seller's compliance with the above requirement.

(vi) Subdivisions (iv) and (v) of this subparagraph shall cease to apply 1 year after the issuance of Program Test Authority.

(4) Applications subject to this paragraph (f) will, in any event, be designated for evidentiary hearing in any case where the agreements, arrangements or understandings with the seller provide for the seller's option to acquire

equity in the station or to increase equity interests he retains at the time of the assignment or transfer of control. An evidentiary hearing will similarly be held in any case in which the assignee(s), transferee(s) or any of their principals, or any person in privity therewith, has an option to purchase all or part of the seller's retained or subsequently acquired equity interests in the station.

(Sec. 310, 48 Stat. 1086, as amended: 47 U.S.C. 310) [28 F.R. 12434, Nov. 22, 1963, as amended at 34 F.R. 5103, Mar. 12, 1969]

§ 1.598 Period of construction.

Each construction permit will specify a maximum of 60 days from the date of granting thereof as the time within which construction of the station shall begin and a maximum of 6 months thereafter as the time within which construction shall be completed and the station ready for operation, unless otherwise determined by the Commission upon proper showing in any particular case.

(Sec. 319, 48 Stat. 1089, as amended: 47 U.S.C. 319)

§ 1.599 Forfeiture of construction permit.

A construction permit shall be automatically forfeited if the station is not ready for operation within the time specified therein or within such further time as the Commission may have allowed for completion, and a notation of the forfeiture of any construction permit under this provision will be placed in the records of the Commission as of the expiration date.

(Sec. 319, 48 Stat. 1089, as amended; 47 U.S.C. 319)

§ 1.601 License, simultaneous modification and renewal.

When an application is granted by the Commission necessitating the issuance of a modified license less than 60 days prior to the expiration date of the license sought to be modified, and an application for renewal of said license is granted subsequent or prior thereto (but within 30 days of expiration of the present license), the modified license as well as the renewal license shall be issued to conform to the combined action of the Commission.

§ 1.603 Special waiver procedure relative to applications.

(a) In the case of any broadcast applications designated for hearing, the parties may request the Commission to grant or deny an application upon the

basis of the information contained in the applications and other papers specified in paragraph (b) of this section, without the presentation of oral testimony. Any party desiring to follow this procedure should execute and file with the Commission a waiver in accordance with paragraph (e) of this section, and serve copies on all other parties, or a joint waiver may be filed by all the parties. Upon the receipt of waivers from all parties to a proceeding, the Commission will decide whether the case is an appropriate one for determination without the presentation of oral testimony. If it is determined by the Commission that, notwithstanding the waivers, the presentation of oral testimony is necessary, the parties will be so notified and the case will be retained on the hearing docket. If the Commission concludes that the case can appropriately be decided without the presentation of oral testimony, the record will be considered as closed as of the date the waivers of all parties were first on file with the Commission.

(b) In all cases considered in accordance with this procedure, the Commission will decide the case on the basis of the information contained in the applications and in any other papers pertaining to the applicants or applications which are open to public inspection and which were on file with the Commission when the record was closed. The Commission may call upon any party to furnish any additional information which the Commission deems necessary to a proper decision. Such information shall be served upon all parties. The waiver previously executed by the parties shall be considered in effect unless within 10 days of the service of such information the waiver is withdrawn.

(c) Any decision by the Commission rendered pursuant to this section will be in the nature of a final decision, unless otherwise ordered by the Commission.

(d) By agreeing to the waiver procedure prescribed in this section, no party shall be deemed to waive the right to petition for reconsideration or rehearing, or to appeal to the courts from any adverse final decision of the Commission.

(e) The waiver provided for by this section shall be in the following form:

WAIVER

Name of ap	plicant	
Call letters		
Docket No.		

The undersigned hereby requests the Commission to consider its application and grant or deny it in accordance with the procedure prescribed in § 1.603 of the Commission's rules and regulations. It is understood that all the terms and provisions of _____are incorporated in this waiver.

§ 1.605 Retention of applications in hearing status after designation for hearing.

(a) After an application for a broadcast facility is designated for hearing, it will be retained in hearing status upon the dismissal or amendment and removal from hearing of any other application or applications with which it has been consolidated for hearing.

(b) Where any applicants for a broadcast facility file a request pursuant to § 1.525(a) for approval of an agreement to remove a conflict between their applications, the applications will be retained in hearing status pending such proceedings on the joint request as may be ordered and such action thereon as may be taken.

(1) If further hearing is not required on issues other than those arising out of the agreement, the proceeding shall be terminated and appropriate disposition shall be made of the applications.

- (2) Review Board action pursuant to this paragraph shall become final unless any of the parties files an application for review within 10 days after public release of the document containing the full text of that action or unless the Commission, by order issued within 20 days after the time for filing an application for review expires, stays the Board's action and reviews that action on its own motion.
- (3) Where further hearing is required on issues unrelated to the agreement, the presiding officer shall continue to conduct the hearing on such other issues pending final action on the agreement, but the record in the proceeding shall not be closed until such final action on the agreement has been taken.
- (4) In any case where a conflict between applications will be removed by an agreement for an engineering amendment to an application, the amended application shall be removed from hearing status upon final approval of the agreement and acceptance of the amendment.
- (c) An application for a broadcast facility which has been designated for hearing and which is amended so as to eliminate the need for hearing or further hearing on the issues specified, other

than as provided for in paragraph (b) of this section, will be removed from hearing status.

FORMS AND INFORMATION TO BE FILED WITH THE COMMISSION

§ 1.611 Financial report.

Each licensee or permittee of a commercially operated standard, FM, television, or international broadcast station (as defined in Part 73 of this chapter) shall file with the Commission on or before April 1 of each year, on FCC Form 324, an annual financial report.

§ 1.613 Filing of contracts.

Each licensee or permittee of a standard, FM, television, or international broadcast station (as defined in Part 73 of this chapter), whether operating or intending to operate on a commercial or noncommercial basis, shall file with the Commission copies of the following contracts, instruments, and documents together with amendments, supplements, and cancellations, within 30 days of execution thereof. The substance of oral contracts shall be reported in writing.

(a) Contracts relating to network service: All network affiliation contracts, agreements or understandings between a station and a national, regional or other network shall be reduced to writing and filed. Each such filing on or after May 1. 1969 initially shall consist of a written instrument containing all the terms and conditions of such contract, agreement, or understanding without reference to any other paper or document by incorporation or otherwise. Subsequent filings may simply set forth renewal, extension, amendment, or change, as the case may be, of a particular contract previously filed in accordance herewith. Transcription agreements or contracts for the supplying of film for television stations which specify option time must be filed. This section does not require the filing of transcription agreements or contracts for the supplying of film for television stations which do not specify option time, nor contracts granting the right to broadcast music such as ASCAP, BMI, or SESAC agreements.

(b) Contracts relating to ownership or control: Contracts, instruments, or documents relating to the present or future ownership or control of the licensee or permittee or of the licensee's or permittee's stock, rights, or interests therein, or relating to changes in such

ownership or control. This paragraph shall include but is not limited to the following:

- (1) Articles of partnership, association, and incorporation, and changes in such instruments;
- (2) Bylaws, and any instruments effecting changes in such bylaws;
- (3) Any agreement, document, or instrument (i) providing for the assignment of a license or permit or (ii) affecting, directly or indirectly, the ownership or voting rights of the licensee's or permittee's stock (common or preferred, voting or nonvoting), such as: (a) Agreements for transfer of stock; (b) instruments for the issuance of new stock; or (c) agreements for the acquisition of licensee's or permittee's stock by the issuing licensee or permittee corporation. Pledges, trust agreements, options to purchase stock and other executory agreements are required to be filed: Provided. however, That trust agreements are not required to be filed, unless requested specifically by the Commission; in lieu of the trust agreement, the licensee or permittee may submit the following information concerning the trust: (1) Name of trust; (2) duration of trust; (3) number of shares of stock owned; (4) name of beneficial owner of stock; (5) name of record owner of stock; (6) name of the party or parties who have the power to vote or control the vote of the shares: and (7) any conditions on the powers of voting the stock or any unusual characteristics of the trust.
- (4) Proxies with respect to the licensee's or permittee's stock running for a period in excess of one year; and all proxies, whether or not running for a period of one year, given without full and detailed instructions binding the nominee to act in a specified manner. With respect to proxies given without full and detailed instructions, a statement showing the number of such proxies, by whom given and received, and the percentage of outstanding stock represented by each proxy shall be submitted by the licensee or permittee within 30 days after the stockholders' meeting in which the stock covered by such proxies has been voted; Provided, however, That when the licensee or permittee is a corporation having more than 50 stockholders, such complete information need be filed only with respect to proxies given by stockholders who are officers or directors, or who have 1 percent or more of the corporation's voting stock; in cases where

the licensee or permittee is a corporation having more than 50 stockholders and the stockholders giving the proxies are neither officers or directors nor hold 1 percent or more of the corporation's stock, the only information required to be filed is the name of any person voting 1 percent or more of the stock by proxy, the number of shares voted by proxy by such person, and the total number of shares voted at the particular stockholders' meeting in which the shares were voted by proxy;

(5) Mortgage or loan agreements containing provisions restricting the licensee's or permittee's freedom of operation, such as those affecting voting rights, specifying or limiting the amount of dividends payable, the purchase of new equipment, the maintenance of current assets, etc.; or

(6) Any agreement reflecting a change in the officers, directors, or stockholders of a corporation, other than the licensee or permittee, having an interest, direct or indirect, in the licensee or permittee as specified by § 1.615.

(c) Contracts relating to the sale of broadcast time to "time brokers" for

(d) Contracts relating to Subsidiary Communications Authorization Operation, except contracts granting licensees or permittees engaged in SCA the right to broadcast copyright music.

(e) Time sales contracts: Time sales contracts with the same sponsor for 4 or more hours per day, except where the length of the events (such as athletic contests, musical programs, and special events) broadcast pursuant to the contract is not under control of the station.

act is not under control of the station.

(f) Contracts relating to personnel:

(1) The following contracts, agreements, or understandings shall be filed: management consultant agreements with independent contractors; contracts relating to the utilization in a management capacity of any person other than an officer, director, or regular employee of the licensee or permittee station; management contracts with any persons, whether or not officers, directors, or regular employees, which provide for both a percentage of profits and a sharing in losses; or any similar agreements.

(2) The following contracts, agreements, or understandings need not be filed: agreements with persons regularly employed as general or station managers or salesmen; contracts with pro-

gram managers or program personnel; contracts with chief engineers or other engineering personnel except those contracts required to be filed under the provisions of §§ 73.93(c), 73.265(c), and 73.565(c) of this chapter; contracts with attorneys, accountants, or consulting radio engineers; contracts with performers; contracts with station representatives; contracts with labor unions; or any similar agreements.

[28 F.R. 12434, Nov. 22, 1963, as amended at 33 F.R. 9083, June 20, 1968; 34 F.R. 5949, Mar. 29, 1969]

§ 1.615 Ownership reports.

(a) Each licensee of a TV, FM, or standard broadcast station (as defined in Part 73 of this chapter), other than noncommercial educational shall file an Ownership Report (FCC Form 323) at the time the application for renewal of station license is required to be filed: Provided, however, That licensees owning more than one TV. FM. or standard broadcast station need file only one Ownership Report at 3-year intervals. Ownership Reports shall give the following information as of a date not more than 30 days prior to the filing of the Ownership Report:

(1) In the case of an individual, the

name of such individual;

(2) In the case of a partnership, the names of the partners and the interest of each partner;

Note: Any change in partners or in their rights will require prior consent of the Commission upon an application for consent to assignment of license or permit. If such change involves less than a controlling interest, the application for Commission consent to such change may be made upon FCC Form 316 (Short Form).

(3) In the case of a corporation, association, trust, estate, or receivership, the data applicable to each:

(i) The name, residence, citizenship, and stock-holdings of officers, directors. stockholders, trustees, executors, administrators, receivers, and members of any association;

(ii) Full information as to family relationship or business association between two or more officials and/or stockholders, trustees, executors, administrators, receivers, and members of any association;

(iii) Capitalization with a description of the classes and voting power of stock authorized by the corporate charter or other appropriate legal instrument and the number of shares of each class issued and outstanding; and

(iv) Full information on FCC Form 323 with respect to the interest and identity of any person having any direct, indirect, fiduciary, or beneficiary interest in the licensee or any of its stock;

For example:

- (a) Where A is the beneficial owner or votes stock held by B, the same information should be furnished for A as is required for B.
- (b) Where X corporation controls the licensee, or holds 25 percent or more of the number of issued and outstanding shares of either voting or non-voting stock of the licensee, the same information should be furnished with respect to X corporation (its capitalization, officers, directors, and stockholders and the amount of stock [by class] in X held by each) as is required in the case of the licensee, together with full information as to the identity and citizenship of the person authorized to vote licensee's stock, in case of voting stock.
- (c) The same information should be furnished as to Y corporation if it controls X corporation or holds 25 percent or more of the number of issued and outstanding shares of either voting or non-voting stock of X, and as to Z corporation if it controls Y corporation or holds 25 percent or more of the number of issued and outstanding shares of either voting or non-voting stock of Y and so on back to natural persons.
 - (4) In the case of all licensees:
- (i) A list of all contracts still in effect required to be filed with the Commission by § 1.613 showing the date of execution and expiration of each contract; and (ii) Any interest which the licensee may have in any other broadcast station.
- (b) A permittee shall file an Ownership Report (FCC Form 323) within 30 days of the date of grant by the Commission of an application for original construction permit. The Ownership Report of the permittee shall give the information required by the applicable portions of paragraph (a) of this section.
- (c) A supplemental Ownership Report (FCC Form 323) shall be filed by each licensee or permittee within 30 days after any change occurs in the information required by the Ownership Report from that previously reported. Such report shall include without limitation:
- Any change in capitalization or organization;
- (2) Any change in officers and directors;
- (3) Any transaction affecting the ownership, direct or indirect, or voting

- rights of licensee's or permittee's stock, such as:
 - (i) A transfer of stock:
- (ii) Issuance of new stock or disposition of treasury stock; or
- (iii) Acquisition of licensee's or permittee's stock by the issuing corporation; or
- (4) Any change in the officers, directors, or stockholders of a corporation other than the licensee or permittee such as X, Y, or Z corporation described in the example in paragraph (a) (3) of this section.

Note: Before any change is made in the organization, capitalization, officers, directors, or stockholders of a corporation other than licensee or permittee, which results in a change in the control of the licensee or permittee, prior Commission consent must be received under § 310(b) of the Communications Act and § 1.540. A transfer of control takes place when an individual, or group in privity, gains or loses affirmative or negative (50 percent) control. See instructions on FCC Form 323 "Ownership Report".

- (d) Exceptions: Where information is required under paragraph (a), (b), or (c) of this section with respect to a corporation or association having more than 50 stockholders or members, such information need be filed only with respect to stockholders or members who are officers or directors of the corporation or association, or to other stockholders or members who have 1 percent or more of either the voting or nonvoting stock of the corporation or voting rights in the association (regardless of whether the stockholder or member is an investment company as defined in Note 4 to §§ 73.35, 73.240, and 73.636 of this chapter): Provided, however, That such information with respect to stock held by stockbrokers need be filed only if the stock is held by the stockbroker in its name (either for itself or for customers) for a period exceeding 30 days.
- (e) The provisions of this paragraph apply to all licensees and permittees of noncommercial educational TV, FM, or standard broadcast stations.
- (1) Each licensee covered by this paragraph shall file an Ownership Report (FCC Form 323E): Provided, however, That licensees owning more than one noncommercial educational TV, FM, or standard broadcast station need file only one Ownership Report at 3-year intervals. Ownership Reports shall give

the following information as of a date not more than 30 days prior to the filing

of the Ownership Report:

(i) The following information as to all officers, members of governing board, and holders of 1 percent or more ownership interest (if any): Name, residence, office held, citizenship, principal profession or occupation, and by whom appointed or elected.

(ii) Full information of FCC Form 323E with respect to the interest and identity of any individual, organization, corporation, association, or any other entity which has direct or indirect control over the licensee or permittee.

(iii) A list of all contracts still in effect required by § 1.613 to be filed with the Commission, showing the date of execution and expiration of each contract.

- (iv) Any interest which the licensee or permittee or any of its officers, members of the governing board, and holders of 1 percent or more ownership interest (if any) hold in any other broadcast station.
- (2) A permittee shall file an Ownership Report (FCC Form 323E) within 30 days of the date of grant by the Commission of an application for original construction permit. The Ownership Report of the permittee shall give the information required by the applicable portions of this paragraph.
- (3) A supplemental Ownership Report (FCC Form 323E) shall be filed by each licensee or permittee within 30 days after any change occurs in the information required by the Ownership Report from that previously reported. Such report should include, without limitation:
 - (i) Any change in organization;
- (ii) Any change in officers or directors;
- (iii) Any transaction affecting the ownership (direct or indirect) or voting rights with respect to the licensee or permittee (or with respect to any stock interest therein).
- (f) A copy of all ownership and supplemental ownership reports and related material filed pursuant to this section shall be maintained and made available for public inspection locally as required by § 1.526.

[28 F.R. 12434, Nov. 22, 1963, as amended at 30 F.R. 4547, Apr. 8, 1965; 33 F.R. 9084, June 20, 1968]

FORFEITURES RELATING TO BROADCAST LICENSEES AND PERMITTEES

§ 1.621 Forfeitures relating to broadcast licensees and permittees.

- (a) Whenever the Commission finds that grounds exist to support a suit for collection of forfeiture provided by section 503(b) of the Communications Act of 1934, as amended, a written notice of apparent liability shall be issued by the Commission and shall be sent by the Commission by registered or certified mail to the last known address of the licensee or permittee. The notification shall specify the date or dates, facts, and the nature of the alleged act or acts, omission or omissions with which the licensee or permittee is charged, and shall specifically identify the particular provision or provisions of the law, rule, or regulation or the license, permit, or cease and desist order involved and shall set forth the amount of the forfeiture. The notification shall inform the licensee or permittee that:
- (1) He may admit liability by paying the amount specified therein;
- (2) He has a right under section 503(b) (2) of the Act to show in writing why he should not be held liable; and
- (3) If he admits liability but considers the amount of the forfeiture excessive, he may submit in writing the reasons therefor.
- (b) Payment must be made, or a written statement in duplicate in response to a notification of apparent liability must be submitted, within 30 days from the receipt of the notification or the attempted delivery thereof. After consideration of the statement, an order shall be entered declaring non-liability or establishing the amount of the forfeiture. If the licensee or permittee fails to take any action in respect to a notification of apparent liability for forfeiture, an order shall be entered establishing the forfeiture as the amount set forth in the notice of apparent liability. Orders of forfeiture shall also advise the party or parties of the Commission's authority under section 504(b) of the Act to remit or mitigate such forfeitures upon application therefor. The Commission shall serve orders of forfeiture or orders of nonliability upon the licensee or permittee involved.
- (c) Orders of forfeiture may be satisfied by payment, within 30 days from the date of receipt of the order, of the amount specified therein. Applications

for mitigation or remission shall be filed within 30 days from the date of receipt of the order of forfeiture. The application must state the facts relied upon and must be in duplicate. After considering the application, an order remitting the entire amount, mitigating the forfeiture, or denying relief shall be served on the licensee or permittee. Such orders may be satisfied by payment within 30 days from the date of receipt of the notification of the amount specified therein.

- (d) If the licensee or permittee fails to take any action in respect to an order imposing or mitigating a forfeiture or denying relief, the case may be referred by the Commission to the Attorney General of the United States for appropriate civil action to recover the forfeiture in accordance with the provisions of section 504(a) of the Act.
- (e) Payment of forfeitures shall be made by check or similar means drawn to the order of the Treasurer of the United States and mailed to the Commission.
- (f) Factual material contained in statements or applications submitted by the licensee or permittee in accordance with the provisions of this section shall, except for material of which official notice may be taken, be supported by affidavit of a person or persons with personal knowledge thereof.

(Sec. 503, 48 Stat. 1101, as amended; 47 U.S.C. 503)

Subpart E—Complaints, Applications, Tariffs, and Reports Involving Common Carriers

SOURCE: The provisions of this Subpart E appear at 28 F.R. 12450, Nov. 22, 1963, unless otherwise noted.

GENERAL

§ 1.701 Show cause orders.

- (a) The Commission may commence any proceeding within its jurisdiction against any common carrier by serving upon the carrier an order to show cause. The order shall contain a statement of the particulars and matters concerning which the Commission is inquiring and the reasons for such action, and will call upon the carrier to appear before the Commission at a place and time therein stated and give evidence upon the matters specified in the order.
- (b) Any carrier upon whom an order has been served under this section shall file its answer within the time specified in the order. Such answer shall specifi-

cally and completely respond to all allegations and matters contained in the show cause order.

(c) All papers filed by a carrier in a proceeding under this section shall conform with the specifications of §§ 1.49 and 1.50 and the subscription and verification requirements of § 1.52. An original and 14 copies of all such papers shall be filed.

§ 1.703 Appearances.

- (a) Hearings. Except as otherwise required by § 1.221 regarding application proceedings, by § 1.91 regarding proceedings instituted under section 312 of the Communications Act of 1934, as amended, or by Commission order in any proceeding, no written statement indicating intent to appear need be filed in advance of actual appearance at any hearing by any person or his attorney.
- (b) Oral arguments. Within 5 days after release of an order designating an initial decision for oral argument or within such other time as may be specifled in the order, any party who wishes to participate in the oral argument shall file a written statement indicating that he will appear and participate. Within such time as may be specified in an order designating any other matter for oral argument, any person wishing to participate in the oral argument shall file a written statement to that effect setting forth the reasons for his interest in the matter. The Commission will advise him whether he may participate. (See § 1.277 for penalties for failure to file appearance statements in proceedings involving oral arguments on initial decisions.)
- (c) Commission counsel. The requirement of paragraph (b) of this section shall not apply to counsel representing the Commission or the Chief of the Common Carrier Bureau.

COMPLAINTS

§ 1.711 Formal or informal complaints.

Complaints filed against carriers under section 208 of the Communications Act may be either formal or informal.

§ 1.713 Satisfaction of complaints; damages.

If a carrier satisfies any complaint brought to its attention by the Commission, a statement must be filed with the Commission, in duplicate, setting forth when and how the complainant has been satisfied: *Provided*, *however*, That no complaint seeking damages as a result of

alleged unjust or unreasonable charges, practices, classifications, or regulations contained in an effective tariff schedule on file with the Commission shall be satisfied except after appropriate authorization by the Commission.

INFORMAL COMPLAINTS

§ 1.716 Form.

An informal complaint shall be in writing and shall contain: (a) The name and address of the complainant, (b) the name of the carrier against which the complaint is made, and (c) a complete statement of the facts tending to show that such carrier did or omitted to do anything in contravention of the Communications Act.

§ 1.717 Procedure.

Upon receipt of any informal complaint, the Commission will forward a copy to the carrier complained of or take the question up by correspondence with The carrier will also be the carrier. called upon, within such time as may be prescribed, either to satisfy the complaint or advise the Commission of its refusal or inability to do so. If the carrier satisfles the complaint, it shall so notify the Commission in accordance with the provisions of § 1.713. The Commission will forward a copy of the carrier's notice of satisfaction to the complainant. If the carrier refuses or is unable to satisfy the complaint, it shall so notify the Commission, in duplicate, and the Commission will forward a copy of such notice to the complainant, with a statement of the procedure to be followed to further prosecute the complaint.

§ 1.718 Unsatisfied informal complaints; formal complaints relating back to the filing dates of informal complaints.

When an informal complaint has not been satisfied pursuant to § 1.717, the complainant may file a formal complaint in the form specified in § 1.721. filing will be deemed to relate back to the filing date of the informal complaint: Provided, That the formal complaint: (a) Is filed within 6 months from the date of the Commission's statement accompanying a copy of the carrier's notice of refusal or inability to satisfy, (b) makes reference to the date of the informal complaint, and (c) is based on the same cause of action as the informal complaint. If no formal complaint is filed within the 6-month period, the

complainant will be deemed to have abandoned the unsatisfied informal complaint, and such complaint will be deemed dismissed.

FORMAL COMPLAINTS

§ 1.721 Form.

- (a) A formal complaint shall contain the name of each complainant and defendant, the address of each complainant, and the name and address of his attorney, if represented by attorney, and shall be subscribed and verified by the complainant.
- (b) The following form may be used in cases to which it is applicable, with such alterations as the circumstances may render necessary.

COMPLAINT

BEFORE THE FEDERAL COMMUNICATIONS
COMMISSION, WASHINGTON, D. C.

Docket No. _____ (To be inserted by the Secretary of the Commission)

Complainant v.

Defendant

The complainant (here insert full name of each complainant and if a corporation the corporate title of such complainant) shows:

- (1) That (here state occupation and post office address of each complainant).
- (2) That (here insert the full name, occupation, and post office address of each defendant).
- (3) That (here insert fully and clearly the specific act or thing complained of, together with such facts as are necessary to give a full understanding of the situation).

Wherefore, complainant asks (here state specifically the relief desired).

Dated at _____ this ____ day of

(Name of each complainant)

(Name and address of attorney, if any)

Form of Verification

being first duly sworn, on oath, deposes, and says: That he is the complainant (or one of the complainants) in the above-entitled matter; that he has read the within and foregoing complaint and knows the contents thereof, and that the matter and things therein stated are true of his own knowledge, save and except those matters therein stated on information and belief, and as to those he believes them to be true.



Subscribed and sworn to before me this ____ day of _____ 19__.

(Notary public or other proper officer)

§ 1.722 Statement of issues and facts.

A formal complaint shall be so drawn as to advise the Commission and the defendant fully wherein the provisions of the Communications Act, or an order, rule, or regulation of the Commission have been violated; as to the facts claimed to constitute such violation, including such data as will identify, with reasonable certainty, the communications, transmissions, or other services complained of (as well as any other appropriate facts elicited by § 1.723); and as to the relief sought.

§ 1.723 Damages; allegations with certainty.

- (a) In case recovery of damages is sought, the complaint shall contain appropriate allegations showing such data as will serve to identify, with reasonable certainty, the communications, transmissions, or other services for which recovery is sought, and shall state:
- (1) That the complainant makes claim for damages;
- (2) The name and address of each individual claimant asking damages;
- (3) The name and address of the defendant against which claim is made;
- (4) The communications, transmissions, or other services rendered, the charge applied thereto, the date when charges were paid, by whom paid, and by whom borne;
- (5) The period of time within which, or the specific dates when the communications, transmissions, or other services were rendered;
- (6) The points of origin and reception of the communications or transmissions, and if the damages sought to be recovered are for services other than communications or transmissions, then the allegations of the complaint shall state the nature and extent of such services, the date or dates when rendered, when paid for, and by whom borne;

(7) The nature and amount of injury sustained by each claimant;

- (8) Separately, the damages with respect to each communication, transmission, or other service for which recovery is sought;
- (9) If damages are sought on behalf of others than the complainant, in what capacity or by what authority complaint is made in their behalf; and

- (10) That suit has not been filed in any court on the basis of the same cause of action.
- (b) Damages will not be awarded upon a complaint unless specifically requested. Damages may be awarded, however, upon a supplemental complaint based upon the finding of the Commission in the original proceeding.

§ 1.724 Specific tariff schedule references.

The several charges, classifications, regulations, or practices complained of should be set out by specific reference to the tariff schedules in which they appear, whenever that is possible.

§ 1.725 Joinder of complainants and causes of action.

- (a) Two or more complainants may join in one complaint if their respective causes of action are against the same defendant and concern substantially the same alleged violation of the Communications Act and substantially the same facts.
- (b) Two or more grounds of complaint involving the same principle, subject, or statement of facts may be included in one complaint, but should be separately stated and numbered.

§ 1.726 Discrimination, preference, or prejudice.

When unjust or unreasonable discrimination or undue or unreasonable preference, advantage, prejudice, or disadvantage is alleged, the complaint shall clearly specify the particular person, company or other entity, locality, or description of traffic affected thereby, and the particular discrimination, preference, advantage, prejudice, or disadvantage relied upon as constituting a violation of the Communications Act.

§ 1.727 Supplemental complaints.

- (a) Filing. There may be filed with the Commission a supplemental complaint setting forth transactions, occurrences or events which have happened since the filing of the original complaint and which relate to the original cause of action.
- (b) Seeking damages. If recovery of damages or overcharges is sought by supplemental complaint, it must be filed with the Commission within the statutory periods of limitations as to actions contained in section 415 of the Communications Act.

§ 1.728 Cross complaints.

A cross complaint, seeking any relief within the jurisdiction of the Commission against any carrier which is a party (complainant or defendant) to the proceeding, may be filed by a defendant with its answer. A cross complaint will be accepted for filing and will be served by the Commission in the manner provided in § 1.729 for serving complaints. For the purpose of this subpart, the term "cross complaint" shall include counterclaim.

§ 1.729 Copies; service.

(a) An original and 19 copies of all pleadings and briefs filed in any formal complaint proceeding shall be furnished the Commission. When service is to be made by the Commission, one extra copy shall be furnished for each party to the proceeding.

(b) The Commission will serve a copy of any formal complaint filed with it (and any supplemental, amended, or cross complaint) together with a notice of the filing of the complaint. Such notice shall call upon the carrier to satisfy the complaint in accordance with § 1.713 or answer the same in writing within the time specified in said notice.

(c) All subsequent pleadings and brief filed in any formal complaint proceeding shall be served by the filing party on all other parties to the proceeding in accordance with the requirements of § 1.47. Proof of such service shall also be made in accordance with the requirements of said section.

[28 F.R. 12450, Nov. 22, 1963, as amended at 29 F.R. 8219, June 30, 1964]

§ 1.730 Answers to complaints, supplemental complaints, amended complaints, and cross complaints.

Any carrier upon whom a copy of a formal complaint, supplemental complaint, amended complaint, or cross complaint is served under this subpart shall serve an answer within the time specified by the Commission in its notice of The answer shall advise complaint. the parties and the Commission fully completely of the nature and the defense, and shall respond specifically to all material allegations of the complaint. Collateral or immaterial issues shall be avoided in answers and every effort should be made to narrow the Matters alleged as affirmative defenses shall be separately stated and numbered. Any defendant failing to file and serve an answer within the time and in the manner prescribed may be deemed in default.

§ 1.731 Motions to dismiss complaints or to make them more definite and certain.

(a) A defendant may serve with his answer a motion to dismiss a complaint because of lack of legal sufficiency appearing on the face of such complaint.

(b) Within 10 days after service of a complaint by the Commission, a defendant may file a motion that the allegations in the complaint be made more definite and certain, such motion to point out the defects complained of and details desired. If such motion is granted by the Commission, it will order the complainant to file an amended complaint within such time as may be specified in the order.

§ 1.732 Replies to answers or amended answers; motions to make answers more definite and certain.

Within 10 days after service of an answer or an amended answer, a complainant may serve a reply which shall be responsive to matters contained in such answer or amended answer and shall not contain new matters. Failure to reply will not be deemed as admission of any allegations contained in such answer or amended answer. A complainant may also serve with his reply a motion that the answer be made more definite and certain, such motion to point out the defects complained of and the details desired. such motion is granted by the Commission, it will order the defendant to file an amended answer within such time as may be specified in the order.

§ 1.733 Oppositions to motions to dismiss complaints or to make them more definite and certain.

Within 10 days after service of a motion to dismiss a complaint or to make it more definite and certain, a complainant may serve an opposition to such motion.

§ 1.734 Specifications as to pleadings, briefs, and other documents; subscription and verification.

All papers filed in any formal complaint proceeding must be drawn in conformity with the requirements of §§ 1.49, 1.50, and 1.52.

§ 1.735 Formal complaints not stating a cause of action; defective pleadings.

- (a) Any document purporting to be a formal complaint which does not state a cause of action under the Communications Act will be dismissed. In such case any amendment to such document will be considered a new filing which must be made within the statutory periods of limitations of actions contained in section 415 of the Communications Act, if recovery of damages or overcharges is sought.
- (b) Any pleading filed in a formal complaint proceeding not in conformity with the requirements of the applicable rules in this part (other than the matter covered in paragraph (a) of this section) may be deemed defective. In such case the Commission will:
- (1) Request that specified defects be corrected and that corrected pleadings be filed and served within a prescribed time as a condition to being treated as timely filed: and
- (2) Notify all persons known to the Commission to have been served with any defective pleading of the action taken under this paragraph.

APPLICATIONS

§ 1.741 Scope.

The general rules relating to applicacontained in §§ 1.742 through 1.748 apply to all applications filed by carriers except those filed by public correspondence radio stations pursuant to Parts 21, 81, 83, 85, and 87 of this chapter, and those filed by common carriers pursuant to Part 25 of this chapter. Part 21 contains general rules applicable to applications filed pursuant thereto. For general rules applicable to applications filed pursuant to Parts 81, 83, 85, and 87, see such parts and Subpart F of this part. For rules applicable to applications filed pursuant to Part 25, see said part.

§ 1.742 Place of filing, fees, and number of copies.

All applications shall be tendered for filing at the Commission's main office in Washington, D.C. The applications will be dated by the Mail and Files Division upon receipt and then forwarded to the Common Carrier Bureau. The number of copies required for each application and the nonrefundable fees (see Subpart G) which must accompany each application in order to qualify it for accept-

ance for filing and consideration are set forth in the rules in this chapter relating to various types of applications. However, if any application is not of the types covered by this chapter, an original and two copies of each such application shall be submitted, accompanied by a nonrefundable fee of \$10.

[29 F.R. 12371, Aug. 28, 1964]

§ 1.743 Who may sign applications.

- (a) Except as provided in paragraph (b) of this section, applications, amendments thereto, and related statements of fact required by the Commission shall be personally signed by the applicant. if the applicant is an individual; by one of the partners, if the applicant is a partnership; by an officer or duly authorized employee, if the applicant is a corporation; or by a member who is an officer, if the applicant is an unincorporated association. Applications, amendments, and related statements of fact filed on behalf of eligible government entities, such as states and territories of the United States and political subdivisions thereof, the District of Columbia, and units of local government, including incorporated municipalities. shall be signed by such duly elected or appointed officials as may be competent to do so under the laws of the applicable jurisdiction.
- (b) Applications, amendments thereto, and related statements of fact required by the Commission may be signed by the applicant's attorney in case of the applicant's physical disability, or in case the applicant does not reside in any of the contiguous 48 States of the United States or in the District of Columbia. The attorney shall in that event separately set forth the reason why the application is not signed by the applicant. In addition, if any matter is stated on the basis of the attorney's belief only (rather than his knowledge), he shall separately set forth his reasons for believing that such statements are true.
- (c) Only the original of applications, amendments, or related statements of fact need be signed; copies may be conformed.
- (d) Applications, amendments, and related statements of fact need not be signed under oath. Willful false statements made therein, however, are punishable by fine and imprisonment, U.S. Code, Title 18, section 1001, and by appropriate administrative sanctions, in-

cluding revocation of station license pursuant to section 312(a) (1) of the Communications Act of 1934, as amended.

§ 1.744 Amendments.

(a) Any application not designated for hearing may be amended at any time by the filing of signed amendments in the same manner, and with the same number of copies, as was the initial application. If a petition to deny (or to designate for hearing) has been filed, the amendment shall be served on the petitioner.

(b) After any application is designated for hearing, requests to amend such application may be granted by the presiding officer upon good cause shown by petition, which petition shall be properly served upon all other parties to

the proceeding.

(c) The applicant may at any time be ordered to amend his application so as to make it more definite and certain. Such order may be issued upon motion of the Commission (or the presiding officer, if the application has been designated for hearing) or upon petition of any interested person, which petition shall be properly served upon the applicant and, if the application has been designated for hearing, upon all parties to the hearing.

[29 F.R. 6444, May 16, 1964 and 31 F.R. 14394, Nov. 9, 1966]

§ 1.745 Additional statements.

The applicant may be required to submit such additional documents and written statements of fact, signed and verified (or affirmed), as in the judgment of the Commission (or the presiding officer, if the application has been designated for hearing) may be necessary. Any additional documents and written statements of fact required in connection with applications under Title II of the Communications Act need not be verified (or affirmed).

[29 F.R. 6444, May 16, 1964]

§ 1.746 Defective applications.

(a) Applications not in accordance with the applicable rules in this chapter may be deemed defective and returned by the Commission without acceptance of such applications for filing and consideration. Such applications will be accepted for filing and consideration if accompanied by petition showing good cause for waiver of the rule with which the application does not conform.

(b) The assignment of a file number, if any, to an application is for the ad-

ministrative convenience of the Commission and does not indicate the acceptance of the application for filing and consideration.

§ 1.747 Inconsistent or conflicting applications.

When an application is pending or undecided, no inconsistent or conflicting application filed by the same applicant, his successor or assignee, or on behalf or for the benefit of said applicant, his successor, or assignee, will be considered by the Commission.

§ 1.748 Dismissal of applications.

(a) Before designation for hearing. Any application not designated for hearing may be dismissed without prejudice at any time upon request of the applicant. An applicant's request for the return of an application that has been accepted for filing and consideration, but not designated for hearing, will be deemed a request for dismissal without prejudice. The Commission may dismiss an application without prejudice before it has been designated for hearing when the applicant fails to comply or justify noncompliance with Commission requests for additional information in connection with such application.

(b) After designation for hearing. A request to dismiss an application without prejudice after it has been designated for hearing shall be made by petition properly served upon all parties to the hearing and will be granted only for good cause shown. An application may be dismissed with prejudice after it has been designated for hearing when the applicant:

(1) Fails to comply with the requirements of § 1.221(c);

(2) Otherwise fails to prosecute his application; or

(3) Fails to comply or justify non-compliance with Commission requests for additional information in connection with such application.

[28 F.R. 12450, Nov. 22, 1963, as amended at 29 F.R. 6445, May 16, 1964]

§ 1.749 Action on applications under delegated authority.

Certain applications do not require action by the Commission but, pursuant to the delegated authority contained in Subpart B of Part 0 of this chapter, may be acted upon by the Telegraph Committee, the Telephone Committee, or the Chief of the Common Carrier Bureau, respectively, subject to reconsideration by the Commission.

SPECIFIC TYPES OF APPLICATIONS UNDER TITLE II OF COMMUNICATIONS ACT

§ 1.761 Cross reference.

Specific types of applications under Title III of the Communications Act involving public correspondence radio stations are specified in Parts 21, 23, 81, 83, and 87 of this chapter.

§ 1.762 Interlocking directorates.

Applications under section 212 of the Communications Act for authority to hold the position of officer or director of more than one carrier subject to the act or for a finding that two or more carriers are commonly owned shall be made in the form and manner, with the number of copies and accompanied by the fees set forth in Part 62 of this chapter. The Commission shall be informed of any change in status of any person authorized to hold the position of officer or director of more than one carrier, as required by Part 62 of this chapter.

§ 1.763 Construction, extension, acquisition or operation of lines.

(a) Applications under section 214 of the Communications Act for authority to construct a new line, extend any line, acquire or operate any line or extension thereof, or to engage in transmission over or by means of such additional or extended line, to furnish temporary or emergency service, or to supplement existing facilities shall be made in the form and manner, with the number of copies and accompanied by the fees specified in Part 63 of this chapter.

(b) In cases under this section requiring a certificate, notice is given to and a copy of the application is filed with the Secretary of the Army, the Secretary of the Navy, and the Governor of each State involved. Hearing is held if any of these persons desires to be heard or if the Commission determines that a hearing should be held. Copies of applications for certificates are filed with the regulatory agencies of the States involved.

§ 1.764 Discontinuance, reduction, or impairment of service.

(a) Applications under section 214 of the Communications Act for authority to discontinue, reduce, or impair service to a community or part of a community or for the temporary, emergency, or partial discontinuance, reduction, or impairment of service shall be made in the form and manner, with the number of copies and accompanied by the fees specified in Part 63 of this chapter. Posted and published notice shall be given the public as required by Part 63 of this chapter.

(b) In cases under this section requiring a certificate, notice is given to and a copy of the application is filed with the Secretary of the Army, the Secretary of the Navy, and the Governor of each State involved. Hearing is held if any of these persons desires to be heard or if the Commission determines that a hearing should be held. Copies of all formal applications under this section requesting authorizations (including certificates) are filed with the Secretary of the Army, the Secretary of the Navy, and the Governor of each State involved. Copies of all applications under this section requesting authorizations (including certificates) are filed with the regulatory agencies of the States involved.

§ 1.765 Consolidation or acquisition of telephone companies.

Applications under section 221(a) of the Communications Act for authority to consolidate or acquire telephone companies shall be made in the form and manner, with the number of copies and accompanied by the fees shown in Part 66 of this chapter.

§ 1.766 Consolidation of domestic telegraph carriers.

(a) Applications under section 222 of the Communications Act by two or more domestic telegraph carriers for authority to effect a consolidation or merger or any domestic telegraph bу to acquire all or any part of the domestic telegraph properties, domestic telegraph facilities, or domestic telegraph operations of any carrier shall contain such information as is necessary for the Commission to act upon such application under the provisions of section 222 of the act. Each such application shall be accompanied by a non-refundable fee of \$10.00

(b) These applications are acted upon by the Commission after public hearing. Reasonable notice in writing of the public hearing and an opportunity to be heard is given by the Commission to the Governor of each of the States in which any of the physical property involved in such proposed consolidation or merger is situated, to the Secretary of State, the Secretary of the Army, the Attorney General of the United States, the Secretary of the Navy, representatives of employees where represented by bargaining representatives known to the Commission, and to such other persons as the Commission may deem advisable,

§ 1.767 Cable landing licenses.

- (a) Applications for cable landing licenses under 47 U.S.C. 34-39 and Executive Order No. 10530, dated May 10, 1954, should be filed in duplicate and in accordance with the provisions of that Executive Order. These applications should contain the name and address of the applicant; the corporate structure and citizenship of officers if a corporation; a description of the submarine cable, including the type and number of channels and the capacity thereof; the location of points on the shore of the United States and in foreign countries where cable will land (including a map); the proposed use, need, and desirability of the cable; and such other information as may be necessary to enable the Commission to act thereon. A separate application shall be filed with respect to each individual cable system for which a license is requested, or for which modification or amendment of a previous license is requested, and each such application shall be accompanied by a nonrefundable fee of \$100.
- (b) These applications are acted upon by the Commission after obtaining the approval of the Secretary of State and such assistance from any executive department or establishment of the Government as it may require.
- (c) Original files relating to submarine cable landing licenses and applications for licenses since June 30, 1934, are kept by the Commission. Such applications for licenses (including all documents and exhibits filed with and made a part thereof, with the exception of any maps showing the exact location of the submarine cable or cables to be licensed) and the licenses issued pursuant thereto, with the exception of such maps, shall, unless otherwise ordered by the Commission, be open to public inspection in the offices of the Commission in Washington, D.C.
- (d) Original files relating to licenses and applications for licenses for the landing operation of cables prior to June 30, 1934, were kept by the Department of State, and such files prior to 1930 have been transferred to the Executive and

Foreign Affairs Branch of the General Records Office of the National Archives. Requests for inspection of these files should, however, be addressed to the Federal Communications Commission washington, D.C., 20554; and the Commission will obtain such files for a temporary period in order to permit inspection at the offices of the Commission.

TARIFFS

§ 1.771 Filing.

Schedules of charges, and classifications, practices, and regulations affecting such charges, required under section 203 of the Communications Act shall be constructed, filed, and posted in accordance with and subject to the requirements of Part 61 of this chapter.

§ 1.772 Application for special tariff permission.

Applications under section 203 of the Communications Act for special tariff permission shall be made in the form and manner, with the number of copies and accompanied by the fees shown in Part 61 of this chapter.

§ 1.773 Petitions for suspension of tariff schedules.

- (a) Content. A petition for suspension of a new tariff schedule or any provision thereof shall indicate the schedule affected by its Federal Communications Commission number and give specific reference to the items against which protest is made, together with a statement indicating in what respects the protested tariff schedule is considered unlawful. No petition shall include a prayer that it also be considered a formal complaint. Any such formal complaint shall be filed as a separate pleading as provided in § 1.721.
- (b) When filed. A petition for suspension shall be filed with the Commission and served upon the publishing carrier and the Chief, Common Carrier Bureau at least 14 days before the effective date of the tariff schedule. In case of emergency and within the time limits herein provided, a telegraphic request for suspension may be sent to the Commission setting forth succinctly the substance of the matters required by paragraph (a) of this section. A copy of any such telegraphic request shall be sent simultaneously to the publishing carrier and the Chief, Common Carrier Bureau and forthwith confirmed by petition filed and served in accordance with this section.

(c) Reply. A publishing carrier may reply to a petition for suspension, but any such reply shall be filed with the Commission and served simultaneously upon petitioner and the Chief, Common Carrier Bureau within 3 days after service of the petition for suspension.

(d) Copies; service. An original and 14 copies of each petition or reply must be filed with the Commission, and served in accordance with paragraphs (b) and

(c) of this section.

[31 F.R. 6868, May 10, 1966]

CONTRACTS, REPORTS, AND REQUESTS REQUIRED TO BE FILED BY CARRIERS

§ 1.781 Requests for extension of filing time.

Requests for extension of time within which to file contracts, reports, and requests referred to in §§ 1.783 through 1.814 shall be made in writing and may be granted for good cause shown.

CONTRACTS

§ 1.783 Filing.

Copies of carrier contracts, agreements, concessions, licenses, authorizations or other arrangements, shall be filed as required by Part 43 of this chapter.

FINANCIAL AND ACCOUNTING REPORTS AND REQUESTS

§ 1.785 Annual financial reports.

- (a) Annual financial reports shall be filed by carriers and affiliates as required by Part 43 of this chapter on the following forms:
- (1) Form H (holding companies who do not report to the Commission in the manner prescribed in paragraph (b) of this section).
- (2) Form L (licensees in the domestic public land mobile radio services who do not report to the Commission on Annual Report Form M).
 - (3) Form M (telephone companies).
- (4) Form O (wire-telegraph and ocean-cable carriers).
 - (5) Form R (radiotelegraph carriers).
- (6) Form P (licensees in domestic public point-to-point microwave radio service who are miscellaneous common carriers as defined in § 21.1 of this chapter).
- (b) Verified copies of annual reports filed with the Securities and Exchange Commission on its Form 10-K, Form 1-MD, or such other form as may be prescribed by that Commission for filing

of equivalent information, shall be filed annually with this Commission by each person directly or indirectly controlling any communications common carrier in accordance with Part 43 of this chapter.

(c) Carriers having separate departments or divisions for carrier and non-carrier operations shall file separate supplemental annual reports with respect to such carrier and non-carrier operations in accordance with Part 43 of this chapter.

[28 F.R. 12450, Nov. 22, 1963, as amended at 31 F.R. 747, Jan. 20, 1966; 34 F.R. 12137, July 19, 1969]

§ 1.786 Monthly financial reports.

Monthly reports of revenues, expenses, and other items shall be filed by carriers as required by Part 43 of this chapter on the following forms:

FCC Form 901-Telephone.

FCC Form 903—Radiotelegraph and Oceancable.

FCC Form 905-Wire-telegraph.

§ 1.787 Reports of proposed changes in depreciation rates.

Carriers shall file reports regarding proposed changes in depreciation rates as required by Part 43 of this chapter.

§ 1.788 Reports regarding pensions and benefits.

Carriers shall file reports regarding pensions and benefits as required by Part 43 of this chapter.

§ 1.789 Reports regarding division of international telegraph communication charges.

Carriers engaging in international telegraph communication shall file reports in regard to the division of communication charges as required by Part 43 of this chapter.

§ 1.790 Reports relating to traffic by international carriers.

Carriers shall file periodic reports regarding overseas point-to-point traffic and marine telegraph traffic as required by Part 43 of this chapter.

[29 F.R. 14667, Oct. 28, 1964]

§ 1.791 Reports and requests to be filed under Part 31 of this chapter.

Reports and requests shall be filed either periodically, upon the happening of specified events, or for specific approval by class A and class B telephone companies in accordance with and subject to the provisions of Part 31 of this chapter.

§ 1.792 Reports and requests to be filed under Part 33 of this chapter.

Reports and requests shall be filed either periodically, upon the happening of specified events, or for specific approval, by class C telephone companies in accordance with and subject to the provisions of Part 33 of this chapter.

§ 1.793 Reports and requests to be filed under Part 34 of this chapter.

Reports and requests shall be filed either periodically, upon the happening of specified events, or for specific approval, by radiotelegraph carriers in accordance with and subject to the provisions of Part 34 of this chapter.

§ 1.794 Reports and requests to be filed under Part 35 of this chapter.

Reports and requests shall be filed either periodically, upon the happening of specified events, or for specific approval, by wire-telegraph and oceancable carriers in accordance with and subject to the provisions of Part 35 of this chapter.

SERVICES AND FACILITIES REPORTS

§ 1.801 Reports regarding telegraph carrier services.

Telegraph carriers shall file descriptions of their services as required by Part 43 of this chapter.

§ 1.802 Reports relating to continuing authority to supplement facilities or to provide temporary or emergency service.

Carriers receiving authority under Part 63 of this chapter shall file quarterly or semiannual reports as required therein.

§ 1.803 Reports relating to reduction in temporary experimental service.

As required in Part 63 of this chapter, carriers shall report reductions in service which had previously been expanded on an experimental basis for a temporary period.

§ 1.804 Reports regarding domestic telegraph speed of service.

The Western Union Telegraph Company shall furnish monthly reports under Subpart B of Part 64 of this chapter in regard to origin to destination speed of service on F.C.C. Form 338-B and any additional recurring monthly speed of service reports prepared by the telegraph company, together with copies of related instructions issued by the

company to its field offices, in accordance with Part 64 of this chapter.

§ 1.805 Reports relating to service by carriers engaged in public radio service operations.

Monthly and quarterly reports must be filed with the Commission in connection with certain fixed public radio service operations. No form is prescribed. A complete description of the contents of these reports is contained in Part 23 of this chapter.

MISCELLANEOUS REPORTS

§ 1.811 Reports regarding amendments to charters, by-laws and partnership agreements of carriers engaged in domestic public radio services.

Amendments to such documents shall be reported and filed in accordance with Part 21 of this chapter.

§ 1.812 Reports regarding premature destruction of records.

Pursuant to the requirements of Part 42 of this chapter, carriers shall file reports relating to the premature destruction of records.

§ 1.813 Reports of negotiations regarding foreign communication matters.

Pursuant to the requirements of Part 43 of this chapter, carriers engaging or participating in foreign communications shall file monthly reports covering negotiations conducted.

§ 1.814 Reports regarding free servicerendered the Government for national defense.

Carriers rendering free service in connection with the national defense to any agency of the United States Government shall file reports in accordance with Part 2 of this chapter.

Subpart F—Safety and Special Radio Services Applications and Proceedings ¹

SOURCE: The provisions of this Subpart F appear at 28 F.R. 12454, Nov. 22, 1963, unless otherwise noted.

GENERAL

§ 1.901 Scope.

This subpart is applicable to all services listed in Parts 81-99 of this chapter, except that rules involving common carriers concerning complaints, tariffs,

¹ Amended, 29 F.R. 9387, July 9, 1964.

applications and reports required under Title II of the Communications Act are set forth in subpart E of this part. (For additional information relative to applications, see the rules in this chapter relating to each of the respective services.) In case of any conflict or inconsistency between the rules set forth in this subpart and the rules for the specific services enumerated in this section, the former shall govern.

GENERAL FILING REQUIREMENTS § 1.911 Applications required.

- (a) Except as provided in paragraph (b) of this section, station licenses as defined in section 3(bb) of the Communications Act; operator licenses or modifications of renewals thereof; assignments of station licenses or any rights thereunder; and consent to transfer control of a corporation holding a license, shall be granted only upon written and signed application.
- (b) In cases (1) of an emergency found by the Commission involving danger to life or property or due to damage to equipment, or (2) of a national emergency proclaimed by the President or declared by the Congress and during the continuance of any war in which the United States is engaged, when such action is necessary for the national defense or security or otherwise in furtherance of the war effort, or (3) of emergency where the Commission finds, in these services, that it would not be feasible to secure renewal applications from existing licensees or otherwise to follow normal licensing procedure, the Commission may grant station licenses, or modifications or renewals thereof, without the filing of a formal application; but no such authorization shall be granted for or continue in effect beyond the period of the emergency or war requiring it. The procedure to be followed for requests submitted under the provisions of this paragraph is the same as for obtaining special temporary authority § 1.925.
- (c) In case of vessels at sea, the Commission may issue by cable, telegraph, or radio a permit for the operation of a station until the vessel returns to a port of the continental United States.
- (d) Canadian licensees desiring to operate in the United States under the terms of Articles 2 and 3 of the Convention between the United States and Canada concerning Operation of Certain Radio Equipment or Stations (which en-

tered into force May 15, 1952) shall make application upon FCC Form 410, which shall be filed with the Secretary, Federal Communications Commission, Washington, D.C., 20554. Forms may be obtained from the FCC Secretary, any field office of the Commission, or from the Controller of Telecommunications, Department of Transport, Ottawa, Canada.

(e) An alien amateur desiring to operate in the United States under provisions of sections 303(1)(2) and 310(a) of the Communications Act of 1934, as amended, and under the terms of a biateral agreement in force between his country and the United States concluded pursuant to the provisions of Public Law 88–313, shall make application on FCC Form 610–A, which shall be filed with the Secretary, Federal Communications Commission, Washington, D.C., 20554. Forms may be obtained from the Secretary, any field office of the Commission and, in some instances, from United States missions abroad.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308) [28 F.R. 12454, Nov. 22, 1963, as amended at 30 F.R. 2706, Mar. 3, 1965; 34 F.R. 19419, Dec. 9, 1969]

§ 1.912 Where applications are to be filed.

(a) Applications for any class of amateur operator license requiring examination under Part 97 of this chapter shall be filed in the nearest FCC field office, listed in § 0.121 of this chapter.

(b) All applications (except those for renewal of station license) for authority to establish or operate stations (other than ship stations) covered by Part 85 of this chapter, "Public Fixed Stations and Stations of the Maritime Services in Alaska", including correspondence relating thereto, shall be filed in triplicate with the Commission's Engineer in Charge at Seattle, Washington.

(c) Formal applications for ship station licenses for use of radiotelephone or radar transmitting apparatus or both, and applications for modification of such licenses, shall, when accompanied by requests for interim ship station licenses, be filed in accordance with § 83.35 of this chapter and presented in person by applicants or their agents at the nearest field office of the Commission as shown in § 0.121 (a) and (b) of this chapter, or at the Commission's



¹ Form filed as part of original document.

main office in Washington, D.C.: Provided, That, as an alternative procedure, an applicant, in Alaska, for such a ship station license may submit an application by mail to the Commission's Field Engineering Office at Anchorage, Alaska, when accompanied by a written request for an interim ship station license. Applications for renewal of ship station licenses are not subject to the provisions of this paragraph.

- (d) All formal applications for Class B, C, or D station licenses in the Citizens Radio Service, ship station license (FCC Forms 502 and 405-B), and aircraft station license (FCC Forms 404 and 405-B) shall be mailed to, or filed in person at the Commission's office at 334 York Street, Gettysburg, Pa., 17325. Any special requests or applications for special temporary authority concerning a Class B, C, or D station and all applications for Class A station licenses shall be filed in accordance with paragraph (e) of this section.
- (e) All other applications shall be filed with the Commission's offices in Washington as follows:
- (1) Applications submitted by mail shall be addressed to:
- Federal Communications Commission, Washington, D.C. 20554.
- (2) Hand-carried applications accompanied by fees shall be delivered to:
- Mail and Files Division, Office of Executive Director, 1919 M Street NW., Washington, D.C.
- (3) Hand-carried applications not accompanied by fees shall be delivered to:
 Office of the Secretary, 1919 M Street NW., Washington, D.C.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308) [28 F.R. 12454, Nov. 22, 1963, as amended at 31 F.R. 6113, Apr. 21, 1966; 32 F.R. 20861, Dec. 28, 1967]

§ 1.913 Who may sign applications.

(a) Except as provided in paragraph (b) of this section, applications, amendments thereto, and related statements of fact required by the Commission shall be personally signed by the applicant, if the applicant is an individual; by one of the partners, if the applicant is a partnership; by an officer, if the applicant is a corporation; or by a member who is an officer, if the applicant is an unincorporated association. Applications, amendments, and related statements of fact

filed on behalf of eligible government entities, such as states and territories of the United States and political subdivisions thereof, the District of Columbia, and units of local government, including incorporated municipalities, shall be signed by such duly elected or appointed officials as may be competent to do so under the laws of the applicable jurisdiction.

- (b) Applications, amendments thereto, and related statements of fact required by the Commission may be signed by the applicant's attorney in case of the applicant's physical disability or of his absence from the United States. The attorney shall in that event separately set forth the reason why the application is not signed by the applicant. In addition, if any matter is stated on the basis of the attorney's belief only (rather than his knowledge), he shall separately set forth his reasons for believing that such statements are true.
- (c) Only the original of applications, amendments, or related statements of fact need be signed; copies may be conformed.
- (d) Applications, amendments, and related statements of fact need not be signed under oath. Willful false statements made therein, however, are punishable by fine and imprisonment, U.S. Code, Title 18, section 1001, and by appropriate administrative sanctions, including revocation of station license pursuant to section 312(a)(1) of the Communications Act of 1934, as amended.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

§ 1.914 Full disclosures.

Each application shall contain full and complete disclosures with regard to the real party or parties in interest and as to all matters and things required to be disclosed by the application forms.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

§ 1.915 Shared use of broadcast antenna structure.

Applicants who propose to share the use of an antenna structure used by a standard, FM, or TV broadcast station shall submit the following information as a part of the application:

(a) A scale sketch of the antenna system showing the position of the proposed antenna on the tower structure and its

relation to any required obstruction lights and other antennas on the tower; and

(b) A diagram which will clearly indicate the proposed method of mounting the transmission feed lines and how these lines will bridge antenna base insulators if employed by the broadcast station. (Sec. 308, 48 Stat. 1084, as amended; 47

Ù.S.C. 308)

§ 1.916 Repetitious applications.

Where the Commission has, for any reason, denied an application for a new station or for any modification of services or facilities, dismissed such application with prejudice, or revoked the license for a radio station in the Safety and Special Radio Services, the Commission will not consider a like or new application involving service of the same kind to substantially the same area by substantially the same applicant, its successor or assignee, or on behalf of or for the benefit of the original parties in interest, until after the lapse of 12 months from the effective date of the The Commission Commission's order. may, for good cause shown, waive the requirements of this section.

Amendment of applications.

- (a) Any amendment to an application shall be signed and submitted in the same manner and with the same number of copies as was the original application.
- (b) Any application may be amended as a matter of right prior to the designation of such application for hearing merely by filing the appropriate number of copies of the amendments in question duly executed. If a petition to deny (or to designate for hearing) has been filed. the amendment shall be served on the petitioner.
- (c) The Commission (or the presiding officer, if the application has been designated for hearing) may, upon its own motion or upon motion of any party to a proceeding, order the applicant to amend his application so as to make the same more definite and certain, and may require an applicant to submit such documents and written statements of fact as in its judgment may be necessary.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308) [28 F.R. 12454, Nov. 22, 1963, as amended at 29 F.R. 6445, May 16, 1964; 31 F.R. 14394, Nov. 9, 1966]

APPLICATION FORMS AND PARTICULAR FILING REQUIREMENTS

- § 1.921 Procedure for obtaining a radio station authorization and for commencement of operation.
- (a) Persons desiring to install and operate radio transmitting equipment should first submit an application for a radio station authorization in accordance with the rules for the particular service. A list of all application forms used by Safety and Special Radio Services Bureau is contained in § 1.922. Each form contains appropriate instructions concerning the number of required copies, where it may be filed, and the services in which it is intended to be used.
- (b) Each application shall include all information called for by the particular form on which the application is required to be filed unless the information called for is inapplicable, in which case that fact shall be indicated.
- (c) In some cases equipment and service tests are required before an authorized station may be placed in regular operation. Reference should be made to the specific service regarding these provisions.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

§ 1.922 Forms to be used.

NOTE: FCC Form 407 may be used in applying for station license in Maritime Mobile and Alaska Public Fixed Services until March 2, 1970, when use of Form 503 will be required, 34 F.R. 19419, Dec. 9, 1969.

FCC Form. Title

400

Application for Radio Station Authorization in the Safety and Special Radio Services.

400-10 Instructions for completion of FOC Form 400.

Application for Microwave Station 402 Authorization in the Safety and Special Radio Services.

402-10 Instructions for Completion of FCC Form 402.

402-A Annual Report of Licensees of Microwave and Other Fixed Stations When Such Facilities Are Used Cooperatively With Other Persons.

Application for Aircraft Radio Station 404 License.

405-A Application for Renewal of Radio License (Short Form).

405-B License Expiration Notice and/or Renewal Application.

Application for Ground Station Au-406 thorization in the Aviation Services. FCC
Form
410 Registration

Registration of Canadian Radio Station Licensee and Application for Permit to Operate.

453-B Certificate of Special Temporary Authorization for Operation of Radio Station on Board New Aircraft.

Title

480 Application for Civil Air Patrol Radio Station Authorization.

481 Application for Authority to Operate a Station in the Radio Amateur Civil Emergency Service.

482 Certification of Civil Defense Radio Officer.

501 Application for Ship Radio Station License.

502 Application for Ship Radiotelephone and/or Radionavigation Station License.

503 Application for Land Radio Station License in the Maritime Mobile or Alaska Public Fixed Service under Parts 81, 83 and 85.

505 Application for Class B, C, or D Station License in the Citizens Radio Service.

525 Application for Disaster Communications Radio Station Construction Permit and License.

610 Application for amateur radio station and/or operator license.

610-A Application of Alien Amateur Radio Licensee for Permit to Operate in the United States.

610-B Application for Amateur Club or Military Recreation Station.

701 Application for Additional Time to Construct Radio Station.

702 Application for Consent to Assignment of Radio Station Construction
Permit or License (For Stations in
Services Other Than Broadcast).

708 Application for Consent to Transfer of Control of Corporation Holding Construction permit or Station License (For Station in Services Other Than Broadcast).

714 Supplement to Application for New or Modified Radio Station Authorization (concerning antenna structure notification to FAA).

820 Application for Exemption from Ship Radio Station Requirements.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308) [28 F.R. 12454, Nov. 22, 1963, as amended at 28 F.R. 14503, Dec. 31, 1963; 29 F.R. 9387, July 9, 1964; 29 F.R. 13816, Oct. 7, 1964; 30 F.R. 2446, Feb. 25, 1965; 30 F.R. 2706, Mar. 3, 1965; 30 F.R. 6779, May 19, 1965; 30 F.R. 12779, Oct. 7, 1965; 31 F.R. 6113, Apr. 21, 1966; 33 F.R. 10345, July 19, 1968; 34 F.R. 12218, July 24, 1969; 34 F.R. 19419, Dec. 9, 1969]

§ 1.923 Waiver of construction permit requirement.

(a) Effective July 14, 1969, a construction permit shall not be required for any

station in the Safety and Special Radio Services.

(b) Licensees must apply for modification of station license for any antenna change which would be inconsistent with the terms of the station authorization or for which notification is required to be given to the Federal Aviation Administration by that agency. Part 17 of this chapter describes the notification criteria.

[34 F.R. 9283, June 12, 1969]

§ 1.924 Assignment or transfer of control, voluntary and involuntary.

(a) (1) Radio station licenses are not transferable; however, except for those set forth in subparagraph (2) of this paragraph, they may be assigned. Licenses must be assigned whenever there is a change of ownership of an authorized radio station as, for example, if the radio communication equipment is sold with a business. The new owner must apply for assignment to him of the existing authorization in accordance with the rules under which the station is authorized.

(2) Licenses for stations in the Amateur, Aviation (aircraft), Citizens, and Maritime (ship) Radio Services cannot be assigned. Whenever there is a change of ownership of one of these latter stations, the new owner must apply for a new license. Upon receipt of the new license, the former license must be surrendered for cancellation.

(b) (1) Application for consent to voluntary assignment of a construction permit or license, or for consent to voluntary transfer of control of a corporation holding a construction permit or license, shall be filed with the Commission at least 60 days prior to the contemplated effective date of assignment or transfer of control.

(2) The following application forms should be used:

(i) FCC Form 400: for assignment of station authorization in services under Parts 89, 91, and 93 of this chapter, except as provided in subdivision (ii) of this subparagraph. Attached thereto shall be a signed letter from proposed assignor stating his desire to assign his current authorization in accordance with the rules governing the particular service involved.

(ii) FCC Form 402: for assignment of an authorization for fixed stations in the Safety and Special Radio Services using frequencies above 952 Mc/s (so-called microwave stations). Attached thereto shall be a signed letter from proposed assignor stating his desire to assign his current authorization in accordance with the rules governing the particular service involved.

(iii) FCC Form 406: for assignment of ground station authorizations in the Aviation Services, except as provided in subdivision (ii) of this subparagraph.

(iv) [Reserved]

(v) FCC Form 702: for assignment of licenses or construction permits of all

other types.

(vi) FCC Form 703: for consent to transfer control of a corporation holding any type of license or construction permit.

- (c) (1) In the event of the death or legal disability of a permittee or licensee, a member of a partnership, or a person directly or indirectly in control of a corporation which is a permittee or licensee, the Commission shall be notified in writing promptly of the occurrence of such
- death or legal disability.

 (2) Within 30 days after the occurrence of such death or legal disability (except in the case of a ship or amateur station), application shall be filed for consent to involuntary assignment of such permit or license, or for involuntary transfer of control of such corporation, to a person or entity legally qualified to succeed to the foregoing interests under the laws of the place having jurisdiction over the estate involved. The procedure and forms to be followed are the same as those specified in paragraph (b) of this section.
- (3) In the case of stations in the Amateur, Aviation (aircraft), Citizens, and Maritime (ship) Radio Services, involuntary assignment of licenses will not be made; such licenses shall be surrendered for cancellation upon the death or legal disability of the licensee.

(Sec. 310, 48 Stat. 1086, as amended; 47 U.S.C. 310) [28 F.R. 12454, Nov. 22, 1963, as amended at 31 F.R. 2600, Feb. 10, 1966]

§ 1.925 Application for special temporary authorization.

(a) Special temporary authority may be granted to install and operate new equipment, or to operate a licensed station in a manner and to an extent or for service other or beyond that authorized in an existing license, upon proper application therefor. No such request will be considered unless full particulars as to the purpose for which the request is made are stated and unless the request is received by the Commission at least 10 days prior to the date of proposed operation. A request received within less than 10 days may be accepted upon due showing of sufficient reasons.

- (b) Requests for such temporary authorization may be filed in letter form, properly signed; however, in cases of emergency involving danger to life or property or due to damage to equipment, such request may be made by telephone or telegraph, provided written request, properly signed, is submitted within 10 days from the date of such request.
- (c) The purchasers of a new aircraft with factory-installed radio equipment may operate the radio station on the aircraft for a period of 30 days under Special Temporary Authority evidenced by a copy of a certificate (FCC Form 453B) executed by the manufacturer, dealer, or distributor, the original of which has been mailed to the Commission with the formal application for station license.

§ 1.926 Application for renewal of license.

- (a) Application for renewal of station license shall be submitted on FCC Form 405-A (except as noted in paragraph (b) of this section).
- (b) (1) Applications for renewal of an amateur operator license, an amateur station license (with the exception of an amateur club or military recreation station license), or a combined amateur operator station license shall be filed on FCC Form 610.
- (2) Applications for renewal of aircraft radio station licenses, shall be submitted on FCC Form 405-B.
- (3) Application for renewal of authorization to operate an amateur station in the Radio Amateur Civil Emergency Service (RACES) shall be filed on FCC Form 481-1 and shall be submitted concurrently with the application for renewal of the basic amateur radio station license.
- (4) Applications for renewal of an amateur club or military recreation station license shall be filed on FCC Form 610-B.

(5)-(8) [Reserved]

(9) Applications for renewal of licenses for radiotelephone stations required by Title III, Part II of the Com-



munications Act of 1934, as amended, or the Safety of Life at Sea Convention and for renewal of all ship licenses which include radiotelegraph shall be filed on Form 405-A. An application for renewal of ship radiotelephone station license (other than those listed in the preceding sentence) and/or ship radionavigation station license shall be filed on FCC Form 405-B.

(10) Application for renewal of Class B, Class C, or Class D station license in the Citizens Radio Service shall be submitted on FCC Form 505.

(11) Application for renewal of Class A station license in the Citizens Radio Service shall be submitted on FCC Form

(c) All applications for renewal of license must be made during the license term and should be filed within 90 days but not later than 30 days prior to the end of the license term. In any case in which the license has, in accordance with the provisions of this chapter, made timely and sufficient application for renewal of license, no license with reference to any activity of a continuing nature shall expire until such application shall have been finally determined.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308) [28 F.R. 12454, Nov. 22, 1963, as amended at 28 F.R. 14503, Dec. 31, 1963; 29 F.R. 3229, Mar. 11, 1964; 30 F.R. 9315, July 27, 1965; 31 F.R. 6113, Apr. 21, 1966; 34 F.R. 18802, Nov. 25, 1969]

- § 1.927 Application for ship radio inspection or periodical survey of ships subject to compulsory radio requirements.
- (a) Applications for ship radio inspection and certification of the ship radio license in accordance with the requirements of section 362(b) of the Communications Act, and/or issuance of a Safety Convention certificate in accordance with the terms of Regulations 12 and 13, Chapter 1 of the Safety Convention, should be submitted on FCC Form 801 entitled "Application for Ship Radio Inspection". This form should be forwarded to the Engineer in Charge of the radio district office nearest the desired port of inspection (see § 0.121 of this chapter).
- (b) Applications for periodical survey as required by Article 11 of the Great Lakes Agreement, and certification prescribed by Articles 12 and 13 thereof, should be submitted on FCC Form 809 "Application for Periodical Survey

(Great Lakes Agreement)." This form should be forwarded to the Engineer in Charge of the radio district office nearest the desired place of survey (see § 0.121 of this chapter).

(c) Applications for inspection of ship radio equipment and apparatus, for the purposes of Part II of Title III of the Communications Act of 1934, as amended, or the Great Lakes Agreement. on a Sunday or national holiday, or during other than the established working hours on any other day, should be submitted on FCC Form 808 entitled "Application for and Certificate of Overtime Service Involving Inspection of Ship Radio Equipment." This form should be forwarded to the Engineer in Charge of the radio district office nearest the desired port of inspection (see § 0.121 of this chapter).

(d) Application for periodical inspection and certification of vessels subject to Part III of Title III of the Communications Act pursuant to section 385 thereof should be submitted on FCC Form 812 entitled "Application for Periodical Inspection (Communications Act, Title III, Part III)." This form should be forwarded to the Engineer in Charge of the radio district office nearest the desired port of inspection (see § 0.121 of this chapter).

(Sec. 10(b), 50 Stat. 196, as amended, 47 U.S.C. 360; and sec. 1, 70 Stat. 1047, 47 U.S.C. 385) [28 F.R. 12454, Nov. 22, 1963, as amended at 30 F.R. 6779, May 19, 1965]

§ 1.928 Procedure with respect to applications for ship radio inspection or periodical survey.

After the following applications are accepted for filing, the Engineer in Charge of the radio district office in which the application is submitted makes the necessary examination and issues the appropriate certification:

(a) Application for ship radio inspection and certification of the ship radio license, pursuant to the requirements of section 362 (b) of the Communications Act:

(b) Application for a Safety Convention certificate in accordance with the terms of Regulations 12 and 13, Chapter 1 of the Safety Convention.

- (c) Application for periodical survey as required by Article 11 of the Great Lakes Agreement and certification prescribed by Articles 12 and 13 thereof;
- (d) Application for periodical inspection and certification of vessels subject

to Part III of Title III of the Communications Act, pursuant to section 385 thereof.

(Sec. 10(b), 50 Stat. 196, as amended, 47 U.S.C. 360; and sec. 1, 70 Stat. 1047, 47 U.S.C. 385) [28 F.R. 12454, Nov. 22, 1963, as amended at 30 F.R. 6779, May 19, 1965]

§ 1.929 Application for exemption from compulsory ship radio requirements.

Applications for exemption, filed under the provisions of sections 352 (b) or (c) and 383 of the Communications Act; Regulation 5, Chapter IV of the Safety Convention; and Article 6 of the Great Lakes Radio Agreement, shall be submitted on FCC Form 820 entitled "Application for Exemption from Ship Radio Station Requirements".

(Sec. 10(b), 50 Stat. 192, as amended, 47 U.S.C. 352; and sec. 170 Stat. 1047, 47 U.S.C 383) [30 F.R. 6779, May 19, 1965]

§ 1.930 Application for temporary waiver of annual inspection.

Informal application for temporary waiver of the annual inspection required under section 362(b) of the Communications Act, as provided in that section, shall be filed by the vessel owner, the vessel's operating agency, the ship station licensee, or the master of the vessel with the Commission's Engineer in Charge of the radio district office nearest the port where the ship is located.

(Sec. 10(b), 50 Stat. 196, as amended; 47 U.S.C. 360)

§ 1.931 Application for extension of construction permit.

(a) A construction permit shall be automatically forfeited if the station is not ready for operation within the time specified therein or within such further time as the Commission may have allowed for completion, and a notation of the forfeiture of any construction permit under this provision will be placed in the records of the Commission as of the expiration date.

(b) Application for extension of time within which to construct a station in the Public Safety, Industrial, and Land Transportation Radio Services shall be submitted on FCC Form 400 or on FCC Form 402, as appropriate; in the Aviation Services, on FCC Form 406, except Civil Air Patrol applications which shall use FCC Form 480; in the case of Class A stations in the Citizens Radio Service, on FCC Form 400; and in all other services, on FCC Form 701. Such applica-

tion shall be filed at least 30 days prior to the expiration date of the construction permit if the facts supporting such application for extension are known to the applicant in time to permit such filing. In other cases such applications will be accepted upon a showing satisfactory to the Commission of sufficient reasons for filing within less than 30 days prior to the expiration date. Such applications will be granted upon a specific and detailed showing that the failure to complete was due to causes not under the control of the grantee, or upon a specific and detailed showing of other matters sufficient to justify the extension.

(Sec. 319, 48 Stat. 1089, as amended; 47 U.S.C. 319)

§ 1.932 Time in which station must be placed in operation.

In those cases in which a license is issued initially in lieu of a construction permit, if the station authorized is not placed in operation within eight months from the date of grant, the authorization shall be invalid and must be returned to the Commission for cancellation.

§ 1.933 Installation or removal of apparatus.

(a) In the Public Safety, Industrial, and Land Transportation Radio Services, replacement of transmitting equipment may be made without prior authorization: Provided, The replacement transmitter appears in the Commission's "Radio Equipment List, Part C" as designated for use in the Public Safety, Industrial, and Land Transportation Radio Services, and the substitute equipment employs the same type of emission and does not exceed the power limitation as set forth in the station authorization.

(b) In the Citizens Radio Service, replacement of transmitting equipment may be made without prior authorization: Provided, The replacement transmitter appears in the Commission's "Radio Equipment List, Part C" as designated for use in the Citizens Radio Service or, in the case of a Class C or Class D station using crystal control, the substitute equipment is crystal controlled: Provided, further, That the substitute equipment employs the same type of emission and does not exceed the frequency tolerance and power limitations prescribed for the particular class of station involved.

Digitized by Google

§ 1.934 Procedure with respect to amateur radio operator license.

After an application for an amateur radio operator license is accepted and an ·examination is conducted in accordance with § 97.27 of this chapter, the examination is graded by the office supervising the examination. The results of the examination are forwarded to Washington, and if the applicant is successful, a license is issued by the Safety and Special Radio Services Bureau.

Application Processing Procedures

§ 1.951 How applications are distributed.

- (a) Amateur and Citizens Division: Amateur, Disaster, RACES, and Citizens.
- (b) Aviation and Marine Division: (1) Aviation Radio Services applications: Air Carrier Aircraft, Private Aircraft, Airdrome Control, Aeronautical En Route, Aeronautical Fixed, Operational Fixed (Aviation), Aeronautical Utility Mobile, Radionavigation (Aviation), Flight Test, Flying School, Aeronautical Public Service, Civil Air Patrol, Advisory, Aeronautical Aeronautical Metropolitan, Aeronautical Search and

Rescue Mobile, and Aeronautical Multi-

- com. (2) Marine Radio Services applications: Public Coast Stations, Limited Coast Stations, Stations on Land in the Maritime Radiodetermination Service, Fixed Stations associated with the Maritime Mobile Service, Stations operated in the Land Mobile Service for maritime purposes, Stations on Shipboard in the Maritime Services, and Public Fixed Stations in Alaska.
- (c) Industrial and Public Safety Facilities Division:
- Industrial Radio Services applications: Business, Forest Products, Industrial Radiolocation, Manufacturers, Motion Picture, Petroleum, Power, Relay Press, Special Industrial, and Telephone Maintenance.
- (2) Land Transportation Radio Services applications: Motor Carrier, Railroad, Taxicab, and Automobile Emergency.
- (3) Public Safety Radio Services applications: Fire, Forestry-Conservation, Highway Maintenance, Local Government, Police, Special Emergency, and State Guard.
- (Secs. 4, 5, 48 Stat. 1066, 1068, as amended; 47 U.S.C. 154, 155) [31 F.R. 6832, May 7, 1966]

§ 1.952 How file numbers are assigned.

- (a) File numbers are assigned to certain categories of applications by the various Divisions of the Safety and Special Radio Services Bureau. Applications for fixed stations using frequencies above 952 Mc/s are further designated by the addition of a letter X at the end of the file number.
- (b) File number symbols and service or class of station designators:

AMATEUR AND DISASTER SERVICES

Y-Amateur.

D-Disaster.

R-RACES.

AVIATION SERVICES

-Aeronautical and fixed group.

AM-Aircraft group.

AA—Aviation auxiliary group.

AR—Aviation radionavigation land.

AC-Civil Air Patrol.

CITIZENS SERVICE

CA-Class A. CB-Class B.

CC-Class C.

CD-Class D.

INDUSTRIES SERVICES

IB-Business

IF-Forest products.

IR—Industrial radiolocation. IX—Manufacturers.

IM-Motion picture.

IP-Petroleum.

IW—Power.
IY—Relay press.
IS—Special industrial.

IT-Telephone maintenance.

LAND TRANSPORTATION SERVICES

LA-Automobile emergency.

LI-Interurban passenger.

LJ-Interurban property.

LR-Railroad.

LX-Taxicab.

LU-Urban passenger.

LV-Urban property.

MARINE SERVICES

MK-Alaskan group.

M-Coastal group.

MA-Marine auxiliary group.

MR-Marine radiolocation land

PUBLIC SAFETY SERVICES

PF-Fire.

PO—Forestry conservation.

PH-Highway maintenance.

PL—Local government.

PP-Police.

PS-Special emergency.

PG-State Guard.

(c) Application or authorization designator symbols:

P—Construction Permit.

MP-Modified CP.

MP/L-Modified CP and License.

MP/ML-Modified CP and Modified License.

AP-Assignment of Permit.

L-License.

ML-Modified License.

AL-Assignment of License.

P/L—Combination CP and License.

R-Renewed License.

TC-Transfer of Control.

§ 1.953 How applications are processed.

- (a) Applications are processed in sequence according to date of filing. Applications which are in accordance with the provisions of this chapter and established policies of the Commission may be processed to completion in accordance with the applicable delegations of authority as set forth in Part 0 of this chapter;
- (b) Applications are presented to the Commission in cases where:
- (1) Applicant requests reconsideration of action taken by the staff under such delegations of authority;
- (2) Requests are made for waiver of, or exception to, a rule for a period in excess of 180 days, but not including certain categories of requests for waivers in excess of 180 days pertaining to ship stations covered by § 0.331(b) (13) of this chapter;
 - (3) [Reserved]
- (4) The staff is not authorized to dismiss an application consistent with the provisions of this chapter and is unable to reach the positive public interest findings prescribed by § 1.971(a); or
- (5) A petition has been filed to deny an application of the categories listed in § 1.962.

[28 F.R. 12454, Nov. 22, 1963, as amended at 34 F.R. 19419, Dec. 9, 1969]

§ 1.955 Frequency coordination, Canada.

(a) As a result of mutual agreements, the Commission has, since May 1950, exchanged comments with the Canadian Department of Transport regarding proposed assignments in certain frequency bands for stations north of "Line A". Line A is described as follows: Begins at Aberdeen, Washington, running by great circle arc to the intersection of 48° N. and 120° W., thence along parallel 48° N. to the intersection of 95° W., thence by great circle arc through the southernmost point of Duluth, Minn.,

thence by great circle arc to 45° N. 85° W., thence southward along the meridian 85° W., to its intersection with parallel 41° N., thence along parallel 41° N. to its intersection with meridian 82° W., thence by great circle arc through the southernmost point of Bangor, Maine, thence by great circle arc through the southernmost point of Searsport, Maine, at which point it terminates.

(b) The frequency bands are as follows:

Mc/s	Mc/s
30.56-32.00	75.40-76.00
33.00-34.00	150.80-174.00
35.00-36.00	450.00-464.725
37.00-38.00	465.275-470.00
39.00-40.00	942.00-960.00
42.00-46.60	1850.0-2200.0
47.00-49.60	8700.0-4200.0
72.00-74.60	5925.0-7125.0

Gc/s 10.55-13.25

(c) Due, however, to the nature of the service, proposed assignments on the following specific frequencies are not coordinated:

Mc/s	Mc/s
156.3	156.7
156.35	156.8
156.4	156.9
156.45	156.95
156.5	157.0 and 161.6
156.55	157.05
156.6	157.1
156.65	157.1 5

(Sec. 4, 48 Stat. 1066, as amended; 47 U.S.C. 154. Interprets or applies sec. 303, 48 Stat. 1082, as amended; 47 U.S.C. 303) [28 F.R. 12454, Nov. 22, 1963, as amended at 29 F.R. 2700, Feb. 26, 1964]

§ 1.956 Rented communications equipment.

- (a) Applications for authorization in the Safety and Special Radio Services which indicate that the equipment therefor will be obtained pursuant to lease-maintenance arrangements with the American Telephone and Telegraph Company or its subsidiaries will not be granted.
- (b) For the purposes of this section, subsidiaries of A. T. & T. include the following:

Bell Telephone Co. of Nevada Citizen Telephone Co., Inc. Illinois Bell Telephone Co. Indiana Bell Telephone Co. Michigan Bell Telephone Co. New England Telephone and Telegraph Co. New Jersey Bell Telephone Co. New York Telephone Co. Northwestern Bell Telephone Co. Southern Bell Telephone and Telepgraph Co. Southwestern Bell Telephone Co.

The Bell Telephone Co. of Pennsylvania

The Chesapeake and Potomac Telephone Co.
The Chesapeake and Potomac Telephone Co.
of Maryland

The Chesapeake and Potomac Telephone Co. of Virginia

The Chesapeake and Potomac Telephone Co. of West Virginia

The Cincinnati and Suburban Bell Telephone

The Diamond State Telephone Co.

The Mountain States Telephone and Telegraph Co.

The Ohio Bell Telephone Co.

The Pacific Telephone and Telegraph Co.
The Southern New England Telephone Co.
Wisconsin Telephone Co.

Note: Pending final action in Docket No. 12722, the terms of this section are not intended to encompass in a negative or affirmative manner, applications involving telephone company lease-maintenance arrangements which have been found or may be found, by any jurisdiction, to be "the furnishing of common carrier communications services" and/or if the charges therefor are or may become "subject to public regulation." See Pars. 24 and 25, First Report and Order, Docket No. 12722.

§ 1.958 Defective applications.

(a) Applications which are incomplete with respect to completeness of answers, supplementary statements, execution, or other matters of a formal character shall be deemed to be defective and may be returned to the applicant with a brief statement as to such defects.

(b) Applications will also be deemed to be defective and may be returned to the applicant in the following cases:

(1) Statutory disqualification of applicant, e. g., aliens under section 310 of the Communications Act;

(2) Proposed use or purpose of station would be unlawful;

(3) Requested frequency is not allocated for assignment for the service proposed.

(c) Applications which are not in accordance with the provisions of this chapter or other requirements of the Commission will be considered defective and may be dismissed unless accompanied either by (1) a petition to amend any rule or regulation with which the application is in conflict, or (2) a request of the applicant for waiver of, or exception to, any rule, regulation, or requirement with which the application is in conflict. Such request shall show the nature of the waiver or exception

desired and set forth the reasons in support thereof. Applications may be dismissed, if the accompanying petition for waiver or amendment of rules does not set forth reasons which, sufficient if true, would justify a waiver or change of the rules.

(d) If an applicant is requested by the Commission to file any additional documents or information not included in the prescribed application form, failure to comply with such request will be deemed to render the application defective, and such application may be dismissed.

(Sec. 308, 48 Stat. 1084, as amended; 47 U.S.C. 308)

§ 1.959 Resubmitted applications.

Any application which has been returned to the applicant for correction will be processed in original order of receipt when resubmitted if it is received within 30 days (45 days outside continental United States) from the date on which it was returned to the applicant. If the application is not resubmitted within the prescribed time, it will be treated as a new application and considered at the time other applications received on the same date are considered.

§ 1.961 Dismissal of applications.

(a) Any application may, upon written request signed by the applicant or his attorney, be dismissed without prejudice as a matter of right prior to the designation of such application for hearing.

(b) Failure to prosecute an application, or failure to respond to official correspondence or request for additional information, will be cause for dismissal. Such dismissal will be without prejudice where an application has not yet been designated for hearing; such dismissal may be with prejudice after an application has been designated for hearing.

(c) Requests to dismiss an application without prejudice after it has been designated for hearing will be considered only upon written petition properly served upon all parties of record and will be granted only for good cause shown. Such petition must be accompanied by a written and signed statement of a person with knowledge of the facts as to whether or not consideration has been promised to or received by petitioner, directly or indirectly, in connection with the filing of such petition for dismissal of the application.

Digitized by Google

- § 1.962 Public notice of acceptance for filing; petitions to deny applications of specified categories.
- (a) Except as qualified in paragraph (b) of this section, the provisions of this section shall apply to all applications for authorizations, and substantial amendments thereof, for the categories of stations and services listed in this paragraph which are filed with the Commission on or after December 12, 1960, and to such applications which were filed prior to December 12, 1960, but are amended substantially on or after that date.
- (1) Fixed point-to-point stations using frequencies above 890 Mc/s (exclusive of control, relay, and repeater stations used as integral parts of mobile radio systems).
- (2) Industrial radiopositioning stations for which frequencies are assigned on an exclusive basis.
 - (3) Aeronautical enroute stations.
 - (4) Aeronautical advisory stations.
 - (5) Airdrome control stations.
 - (6) Aeronautical fixed stations.
- (7) Public coast stations, excluding those located in Alaska which will not render service for hire.
- (b) The provisions of this section are not applicable to applications for the type of authorizations listed in this paragraph.
- (1) A minor change in the facilities of an authorized station or a minor amendment of an application on file.
- (2) Consent to an involuntary assignment or transfer under section 310(b) of the Communications Act or to a voluntary assignment or transfer thereunder which does not involve a substantial change in ownership or control.
- (3) A license under section 319(c) of the Communications Act or, pending application for or grant of such license, any special or temporary authorization to permit interim operation to facilitate completion of authorized construction or to provide substantially the same service as would be authorized by such licensee.
- (4) Extension of time to complete construction of authorized facilities.
- (5) A special temporary authorization not to exceed 30 days where the applicant does not contemplate the filing of an application for regular operation, or not to exceed 60 days pending or after the filing of an application for regular operation.

- (6) An authorization under any of the proviso clauses of section 308(a) of the Communications Act.
- (c) For the purposes of this section, a substantial amendment of an application on file and applications for a substantial change in the facilities of an authorized station shall be:
- (1) Any addition or change in frequency (except deletion of a frequency);
 - (2) Any change in antenna azimuth:
- (3) Any change in antenna beam width;
- (4) Any change in antenna location greater than 5 seconds:
- (5) Any change in antenna location of less than 5 seconds but also involving a requirement for special aeronautical study:
 - (6) Any change in emission;
 - (7) Any increase in antenna height:
- (8) Any increase in authorized power in excess of a 2 to 1 ratio;
- (9) Any increase in emission bandwidth.
- (d) All amendments of an application on file and all changes requested in the facilities of an authorized station other than those amendments and modifications listed in paragraph (c) of this section shall be considered minor.
- (e) The Commission will issue at regular intervals a "Public Notice" listing all applications subject to this section which have been accepted for filing. Such "Public Notice" will re-list any application which has been amended substantially since its previous listing. For the purposes of this section, "accepted for filing" means that an application has been received at the Commission. Such acceptance for filing shall not preclude the subsequent dismissal of an application, pursuant to the provisions of this chapter, as being defective.
- (f) No application subject to the provisions of this section, as originally filed or substantially amended, will be granted by the Commission prior to the thirty-first day following the issuance of public notice of the acceptance for filing of such application or of any substantial amendment thereof: Provided, however, That the Commission, notwithstanding the requirements of this paragraph, may, if the grant of such application is otherwise authorized by law and if it finds that there are extraordinary circumstances requiring emergency operations in the public interest and that delay in institu-

tion of such emergency operations would seriously prejudice the public interest, grant a temporary authorization, accompanied by a statement of its reasons therefor, to permit such emergency operation for a period not exceeding 90 days, and upon making like findings may extend such temporary authorization for one additional period not to exceed 90 days.

(g) Any party in interest may file with the Commission a petition to deny any application, whether as filed originally or as subsequently amended, subject to the provisions of this section, at any time prior to the day the Commission grants such application or formally designates such application for hearing. A petitioner shall serve a copy of such petition on the applicant. A petition shall contain specific allegations of fact sufficient to show that the petitioner is a party in interest and that a grant of the application would be prima facie inconsistent with the public interest, convenience, and necessity. Such allegations of fact except for those of which official notice may be taken, shall be supported by affidavit of a person or persons with personal knowledge thereof.

(h) The applicant may file an opposition to any petition to deny and the petitioner may file a reply thereto (see § 1.45) in which allegations of fact or denials thereof, except for those of which official notice may be taken, shall be supported by affidavit of a person or persons with personal knowledge thereof. The applicant shall serve a copy of his opposition on the petitioner, and the petitioner shall serve a copy of his reply on the applicant.

(Sec. 309, 48 Stat. 1085, as amended; 47 U.S.C. 309) [28 F.R. 12454, Nov. 22, 1963, as amended at 29 F.R. 7822, June 19, 1964; 30 F.R. 4479, Apr. 7, 1965]

ACTION ON APPLICATIONS

AUTHORITY: §§ 1.971 and 1.973 issued under sec. 309, 48 Stat. 1085, as amended; 47 U.S.C. 309.

§ 1.971 Grants without a hearing.

(a) The Commission will grant without a hearing an application for a station authorization if it is proper upon its face and if the Commission finds from an examination of such application and supporting data, any pleading filed, or other matters which it may officially notice, that:

- (1) There are no substantial and material questions of fact;
- (2) The applicant is legally, technically, financially, and otherwise qualified:
- (3) A grant of the application would not involve modification, revocation, or non-renewal of any existing license or outstanding construction permit:
- (4) A grant of the application would not preclude the grant of any mutually exclusive application; and
- (5) A grant of the application would serve the public interest, convenience, and necessity.
- (b) If a petition to deny an application has been filed pursuant to § 1.962 and the Commission grants such application pursuant to paragraph (a) of this section, the Commission will deny the petition and issue a concise statement of the reason for such denial and disposing of all substantial issues raised by the petition.

§ 1.973 Designation for hearing.

- (a) If the Commission is unable to make findings prescribed by 1.971(a) with reference to an application filed prior to December 12, 1960, and not amended substantially on or after that date, the Commission, by letter, will notify the applicant and other known parties in interest of the grounds and reasons for its inability to make such findings and of all the objections made to such application. Following such notice, the applicant will be given an opportunity to reply. If the Commission after considering such reply is still unable to make the findings prescribed by § 1.971(a), it will formally designate the application for hearing on the grounds or reasons then obtaining and shall notify the applicant and all other known parties in interest of such action.
- (b) If the Commission is unable to make the findings prescribed in § 1.971 (a) with reference to any application filed on or after December 12, 1960, or any application filed prior to December 12, 1960, but amended substantially on or after that date, it will formally designate the application for hearing on the grounds or reasons then obtaining and will notify forthwith the applicant and all other known parties in interest of such action.
- (c) Orders designating applications for hearing will specify with particularity the matters and things in issue and

will not include issues or requirements

phrased generally.

(d) Parties in interest, if any, who are not notified by the Commission of its action in designating a particular application for hearing may acquire the status of a party to the proceeding by filing a petition for intervention showing the basis of their interest not more than 30 days after publication in the Federal Register of the hearing issues or any substantial amendment thereto.

(e) Any hearing subsequently held upon such applications shall be a full hearing in which the applicant and all other parties in interest shall be permitted to participate. The burden of proceeding with the introduction of evidence and burden of proof shall be upon the applicant, except that with respect to any issue presented by a petition to deny or a petition to enlarge the issues, such burdens shall be as determined by the Commission.

[28 F.R. 12454, Nov. 22, 1963, as amended at 30 F.R. 3223, Mar. 9, 1965]

Reports To Be Filed With the Commission

§ 1.981 Reports, annual and semiannual.

- (a) Licensees of stations authorized for developmental operation shall submit a report on the results of the developmental program. The report shall be filed with and made a part of each application for renewal of authorization.
- (b) The report shall include comprehensive and detailed information on the following:
 - (1) The final objective.
 - (2) Results of operation to date.
 - (3) Analysis of the results obtained.
 - (4) Copies of any published reports.
- (5) Need for continuation of the program.
- (6) Number of hours of operation on each frequency.
- (c) Where required by the particular service rules, licensees who have entered into agreements with other persons for the cooperative use of radio station facilities must submit annually an audited financial statement reflecting the non-profit cost-sharing nature of the arrangement to the Commission's offices in Washington, D. C., no later than three months after the close of the licensee's fiscal year.

FORFEITURES AGAINST SHIPS AND SHIP MASTERS

§ 1.991 Forfeitures against ships and shipmasters.

(a) Whenever it appears that a shipowner or a shipmaster has become subject to forfeitures under sections 364. 386, or 507 of the Communications Act of 1934. as amended, a written notice of apparent liability will be sent to the shipowner and/or to the shipmaster to the last known address. This notice will set forth the facts which indicate apparent liability: will identify the offenses which are involved and the provisions of the statute, treaty, or rule which appear to have been violated; will state the amount of the forfeiture incurred; and will summarize the courses of action available to the person involved under the provisions of paragraph (b) of this section.

(b) Response to a notice of apparent liability: The shipowner and/or shipmaster involved shall respond to a notice of apparent liability within 30 days after the notice is mailed, or within such longer period of time as may be specified in the notice. Any of the following actions by the shipowner or shipmaster shall constitute a response meeting the require-

ments of this paragraph:

(1) Payment of the forfeiture in the amount specified in the notice of apparent liability. The forfeiture should be paid by check or money order drawn to the order of the Treasurer of the United States and should be mailed to the Federal Communications Commission, Washington, D.C., 20554. The Commission does not accept responsibility for cash payments sent through the mails.

(2) Submission of a written statement denying liability for the forfeiture in whole or in part and/or requesting (even if liability is admitted) that the forfeiture be canceled or reduced because of extenuating circumstances connected with the offense. Allegations or requests made in any such statement must be supported by detailed facts and reasons. The statement should be mailed to the Federal Communications Commission, Washington, D.C., 20554.

(c) Commission action after written statement: After the submission of a written statement, as prescribed in paragraph (b) of this section, the Commission will consider all relevant information available to it. Based on such considerations, the Commission will (1) cancel the forfeiture, (2) offer to reduce

the amount of the forfeiture, or (3) require the forfeiture to be paid in full. Notice of such Commission action, stating the amount of the forfeiture (if any) and the date by which it must be paid, will thereupon be mailed to the ship-owner and/or the shipmaster involved. The forfeiture in the amount stated shall be paid by check or money order drawn to the order of the Treasurer of the United States and shall be mailed to the Federal Communications Commission, Washington, D.C., 20554. The Commission does not accept responsibility for cash payments sent through the mails.

(d) Judicial enforcement of forfeitures imposed by the Commission: If a shipowner or shipmaster falls to respond to the notice of apparent liability as required by paragraph (b) of this section, or if he falls to pay the forfeiture imposed by the Commission under paragraph (c) of this section, the case will, without further notice, be referred by the Commission to the Attorney General of the United States for prosecution in the appropriate Federal District Court to recover the amount of the forfeiture initially imposed.

Note: See §§ 1.101-1.117 for reconsideration and review procedures.
[31 F.R. 2550, Feb. 9, 1966]

Subpart G—Schedule of Fees for Applications Filed With the Commission

AUTHORITY: The provisions of this Subpart G also issued under sec. 501, 65 Stat. 290; 31 U.S.C. 483a.

GENERAL INFORMATION

§ 1.1101 Authority.

Authority for this Subpart is contained in Title V of the Independent Offices Appropriation Act of 1952 (5 U.S.C. 140) which provides that any service rendered by a Federal agency to or for any person shall be performed on a self-sustaining basis to the fullest extent possible. Title V further provides that the head of each Federal agency is authorized by regulation to prescribe such fees as he shall determine to be fair and equitable.

[28 F.R. 12461, Nov. 22, 1963]

§ 1.1103 Payment of fees.

(a) Each application, filed on or after January 1, 1964, for which a fee is prescribed in this subpart, must be accompanied by a remittance in the full

amount of the fee. In no case will an application be accepted for filing or processed prior to payment of the full amount specified. Applications for which no remittance is received, or for which an insufficient amount is received, may be returned to the applicant.

(b) Fee payments accompanying applications received in the Commission's Offices in Washington, D.C., or in any of the Commission's field offices, should be in the form of a check or money order payable to the Federal Communications Commission. The Commission will not be responsible for cash sent through the mails. All fees collected will be paid into the United States Treasury as miscellaneous receipts in accordance with the provisions of Title V of the Independent Offices Appropriations Act of 1952 (31 U.S.C. 483a).

(c) Receipts will be furnished upon request in the case of payments made in person, but no receipts will be issued for payments sent through the mails.

(d) Except as provided in §§ 1.1104 and 1.1105, all fees will be charged irrespective of the Commission's disposition of the application. Applications returned to applicants for additional information or corrections will not require an additional fee when resubmitted.

[28 F.R. 12461, Nov. 22, 1963, as amended at 30 F.R. 14806, Nov. 30, 1965]

§ 1.1104 Return or refund of fees.

(a) The full amount of any fee submitted will be returned or refunded, as appropriate, in the following instances:

(1) Where no fee is required for the application filed.

(2) Where the application is filed by an applicant who cannot fulfill a prescribed age requirement.

(3) Where the application is filed for renewal without reexamination of an amateur or commercial radio operator license after the grace period has expired,

(4) Where the applicant is precluded from obtaining a license by the provisions of section 303(1) or 310(a) of the Communications Act.

(5) Where circumstances beyond the control of the applicant, arising after the application is filed, would render a grant useless.

(b) Payments in excess of an applicable fee will be refunded only if the overpayment exceeds \$2.

[80 F.R. 14806, Nov. 30, 1965]

§ 1.1105 General exceptions.

(a) No fee is required for an application filed for the sole purpose of amending an authorization or pending application (if a fee is otherwise required) so as to comply with new or additional requirements of the Commission's rules or the rules of another Federal government agency affecting the authorization or pending application: however, if the applicant also requests an additional modification or the renewal of his authorization, the appropriate modification or renewal fee must accompany the application. Fee exemptions arising out of this general exception will be announced to the public in the orders

amending the rules or in other appropriate Commission notices.

(b) No fee is required for an application filed by an alien pursuant to a reciprocal radio licensing agreement.

(c) No fee is required for any application or request for an STA or waiver. [30 F.R. 14806, Nov. 30, 1965]

SCHEDULE OF FEES

§ 1.1111 Schedule of fees for Radio Broadcast Services.

(a) Except as provided in paragraph (b) of this section, applications filed in the Radio Broadcast Services shall be accompanied by the fees prescribed below:

	AM	FM	TV	Translator	Auxiliary
Application for construction permit for new station	\$75 75	\$75 75	\$150 150	\$10 210	1 \$30 1 10
than one broadcast station license is involved, the application must be accompanied by the total amount of the fees prescribed for each license so involved). Applications filed on FCC Form 316 (where more than one broadcast station license is involved, the application must be	75	75	150	10	1 30
accompanied by the total amount of the fees prescribed for each license so involved)	30	30	30	No fee	No fee
FCC Form 321	30	30	30	10	1 30
Application for modification other than a major change	30 30	30	30	No fee	1 10
All other applications in the broadcast services (excluding television translator applications not specified above)	1 4 30				
Application for Subscription Television Authorization	-		150		·

¹ With respect to applications for remote pickup broadcast stations authorized under Subpart D of Part 74 of this chapter, one fee will cover the base station (if any) and all the remote pickup mobile stations of a main station, provided the applications therefor are filed at the same time.

2 For determining when a translator application is required to be accompanied by a fee under this section (though not for other purposes in the translator or other broadcast services), "major change" is defined to include only a change in the output frequency of the translator.

3 In all services

In all services 4 For each application.

(b) Fees are not required in the fol-

- lowing instances: (1) Applications filed by tax exempt organizations for the operation of stations providing noncommercial educational broadcast services, whether or not such stations operate on frequencies allocated for noncommercial educational use.
- (2) Applications in the AM service requesting only authority to determine antenna power by direct measurement.
- (3) Applications filed for covering licenses in the Auxiliary Broadcast Services.

[30 F.R. 14806, Nov. 30, 1965, as amended at 34 F.R. 14375, Sept. 13, 1969]

§ 1.1113 Schedule of fees for Common Carrier Services.*

Applications filed for Common Carrier Services shall be accompanied by the fees prescribed below:

DOMESTIC PUBLIC LAND MOBILE RADIO SERVICE 1

Application for initial construction permit or for relocation of a base station (including authority for mobile units, blanket dispatch station authority, and standby transmitters without independent radiating systems * 4)

Application for initial construction permit or for relocation of a dispatch station, control station or repeater station 4_____

* See footnotes to § 1.1113 on following

875

page.

Application for modification of construction permit or license for base station, dispatch station, control station or repeater station at an existing station location		30 10 5
Application for renewal of license for dispatch station, control station or repeater station	All other applications for construction permits or modification of construction permits (no fee required when filed as part of a modification application requiring a \$30 fee)Application for license for operation of an STL station at temporary-fixed	30
Application for license for operation of a rural subscriber station at tempo-	locationsApplication for license for operation of	80
rary-fixed locations 10	a mobile television pickup station	80
Application for license or modification of license for individual subscriber	Application for modification of license_ Application for renewal of license	10 5
stations5	INTERNATIONAL FIXED PUBLIC RADIOCOMMUN	NI-
Application for renewal of license 5	CATION SERVICES	
Application for construction permit or for modification of construction permit to add or change point(s) of communication or to increase service to existing points of communication or for relocation of facilities	Application for construction permit for a replacement transmitter(s) at an authorized station (no fee will be charged for application for modi- fication of license to delete trans- mitter(s) being replaced if both	100
filed as part of a modification application requiring a \$30 fee) 10		50
17n Abia comice costs to more abias of a final	Application for change of location of	100
¹ In this service each transmitter at a fixed location is a separate station notwithstand-	Application for modification of li-	.00
ing the inclusion of more than one such sta-	A	10 75
tion on a single authorization or under a single call sign. ² When included as part of a base station application, a request for blanket dispatch	International Control Station: Application for an initial construction permit for a new station or an additional transmitter(s) at an au-	
station authority made pursuant to the pro- visions of § 21.519(a) of this chapter does not	thorized station	30
require an individual application or fee. A request for such dispatch station authority filed separately from a base station construction permit application requires an application for modification of license and an appropriate fee. 3 An application for a standby transmitter	Application for construction permit for a replacement transmitter(s) at an authorized station (no fee will be charged for application for modifi- cation of license to delete trans- mitter being replaced if both appli- cations are filed simultaneously)	10
having its own independent radiating system	Application for change of location of an authorized station	30
requires the same fee as a base station application.	Application for modification of license_	10
4 No additional fee will be charged for applications for license to cover a construction		10
plications for license to cover a construction permit unless there is a modification or vari-	OTHER RADIO APPLICATIONS	
ation of outstanding authority involved. In	Application for assignment of an au- thorization or transfer of control (a	
that event the appropriate fee for modification is applicable.	separate \$10 fee is required for each	
⁵ This fee applies to any request for dispatch station authority not made pursuant	All other Common Carrier Radio appli-	10
to § 21.519(a) of this chapter.		10

COMMON CARRIER NONRADIO APPLICATIONS
Applications by Communications Com-
mon Carriers for Authorization to
Own Stock in the Communications
Satellite Corp\$10
Section 214 Applications by Telephone
Co 50
Section 214 Applications by Telegraph
Co 10
Cable Landing License Applications 100
Section 221 Applications 50
Interlocking Directorate Applications 10
Tariff applications to change charges or
regulations on less than statutory
notice10
All Other Common Carrier nonradio
Applications 10
[30 F.R. 14807, Nov. 30, 1965]
§ 1.1115 Schedule of fees for Safety and Special Radio Services.

and Special Radio Services.

(a) Except as provided in paragraph

(b) of this section, all formal applications filed in the Safety and Special Radio Services shall be accompanied by the fees prescribed below:

Applications in the Amateur Radio Service:
For initial license, including new class

For initial authorization_____
For modification of authorization____
Applications for Common Carrier Public
Coast Stations in the Maritime
Radio Services:

(b) Fees are not required in the following instances:

(1) Applications filed in the Police, Fire, Forestry-Conservation, Highway Maintenance, Local Government, and State Guard Radio Services.

(2) Applications filed by governmental entities in any of the Safety and Special Radio Services.

(3) Applications filed by the following in the Special Emergency Radio Service: Hospitals, Disaster Relief Organizations, Beach Patrols, and School Buses, and non-profit Ambulance Operators and Rescue Organizations.

(4) Applications filed in the Disaster

Communications Service.

(5) Applications for ship inspections pursuant to the Great Lakes Agreement, the Safety of Life at Sea Convention, and Parts II and III, Title III, of the Communications Act of 1934, as amended.

(6) Applications for Novice Class license in the Amateur Radio Service, applications for amateur stations under military auspices, and applications filed in the Radio Amateur Civil Emergency Service (RACES).

(7) Operational Fixed Microwave Applications filed for Closed Circuit Educa-

tional Television Service.

(8) Applications for Civil Air Patrol Stations, Aeronautical Radionavigation Stations and for Aeronautical Search and Rescue Stations.

(9) Applications for license to cover

construction permit.

[28 F.R. 12462, Nov. 22, as amended at 30 F.R. 14807, Nov. 30, 1965; 31 F.R. 6114, Apr. 21, 1966; 33 F.R. 4739, Mar. 20, 1968; 34 F.R. 9283, June 12, 1969]

§ 1.1117 Schedule of fees for commercial radio operator examinations and licensing.

(a) Except as provided in paragraphs
(b) and (c) of this section, applications filed for commercial radio operator examinations and licensing shall be accompanied by the fees prescribed below:

Applications for new operator license: First-class license, either radiotelephone or radiotelegraph_____ Second-class license, either radiotelephone or radiotelegraph.... 4 Third-class permit, either radiotele-phone or radiotelegraph..... Restricted radiotelephone permit____ Application for renewal of operator 2 Application for endorsement of operator license_____ 2 Application for duplicate license or for replacement license___ Application for provisional certificate for a radiotelephone third-class operator permit endorsed for broadcast use_____

(b) No fee need accompany an application for a verification card (FCC Form 758-F) or for a verified statement (FCC Form 759).

30

50

(c) Whenever an application requests both an operator license and an endorsement the required fee will be the fee prescribed for the license document involved.

[30 F.R. 14808, Nov. 30, 1965, as amended at 33 F.R. 913, Jan. 25, 1968]

§ 1.1119 Experimental Radio Services (other than Broadcast).

Fees are not required in the case of applications filed in the Experimental Radio Services (other than Broadcast).
[28 F.R. 14262, Nov. 22, 1963]

Subpart H—Ex Parte Presentations

AUTHORITY: The provisions of this Subpart H issued under secs. 4, 303, 409, 48 Stat. 1066, 1082, 1096, as amended; 47 U.S.C. 154, 303, 409.

SOURCE: The provisions of this Subpart H appear at 30 F.R. 9271, July 24, 1965, unless otherwise noted.

GENERAL

§ 1.1201 Definitions.

- (a) Restricted proceeding. A proceeding in which the restrictions set forth in this subpart apply. See §§ 1.1203 and 1.1207.
- (b) Commission personnel. All members, officers, and employees of the Commission.
- (c) Decision-making Commission personnel. Those Commission personnel listed in §§ 1.1205 and 1.1209.
- (d) Non-decision-making Commission personnel. All Commission personnel other than decision-making Commission personnel.
- (e) Interested person. Any person having a direct or indirect interest in the outcome of a restricted proceeding, including the following:
- (1) Parties to the restricted proceeding.
- (2) Any other person who might be aggrieved or adversely affected by the outcome of the restricted proceeding. See sections 402(b)(6) and 405 of the Communications Act of 1934, as amended.
- (3) Agents for persons who might be aggrieved or adversely affected by the outcome of the restricted proceeding, including attorneys and consulting engineers.
- (f) Presentation. Any communication going to the merits or outcome of any aspect of a restricted proceeding.

- Note: The term "presentation" is narrower than the general term "communication". For a discussion of the term "presentation", see 1 F.C.C. 2d (1965), at paragraphs 19-25.
- (g) Ex parte presentation. (1) Any written presentation, made to decision-making personnel by any other person, which is not served on the parties to the proceeding. See §§ 1.47 and 1.211.
- (2) Any oral presentation, made to decision-making Commission personnel by any other person, without advance notice to the parties to the proceeding and opportunity for them to be present.

§ 1.1203 Restricted adjudicative proceedings.

- (a) All adjudicative proceedings, including the following, are "restricted" from the time they are designated for hearing until they are removed from hearing status, or have been decided by the Commission, and are no longer subject to reconsideration by the Commission or to review by any court:
- (1) Any proceeding involving the issuance, renewal, modification or assignment of any instrument authorizing the construction or operation of radio facilities under Title III of the Communications Act of 1934, as amended.
- (2) Any proceeding involving the transfer of control of a corporate licensee or permittee under section 310 of the Communications Act.
- (3) Any revocation and/or cease and desist proceeding under section 312 of the Communications Act, unless and until the hearing is waived pursuant to the provisions of § 1.92.
- (4) Any proceeding involving the issuance or suspension of an operator license or permit under section 303(1) or (m) of the Communications Act.
- (5) Any proceeding conducted pursuant to the provisions of sections 206, 207, 212, 214(a) or 221(a) of the Communications Act.
- (6) Any proceeding conducted pursuant to the provisions of section 201(c) (6), (7) or (9), or section 304(f), of the Communications Satellite Act of 1962.
- (b) In the following circumstances, in addition, proceedings involving applications filed under section 308 of the Communications Act are "restricted", as to interested persons, prior to their designation for hearing:
- (1) Application proceedings are restricted as to interested persons from the day on which a petition to deny is filed.

(See section 309(d) of the Communications Act.) If the petition is denied, the proceeding is restricted until the order disposing of the petition is no longer subject to reconsideration by the Commission or to review by any court. If the proceeding is designated for hearing, paragraph (a) of this section applies.

(2) Application proceedings are restricted as to interested persons from the day on which public notice of the filing of a mutually exclusive application is given. Prior to the day on which public notice is given, such proceedings are restricted as to any interested person having actual knowledge that a mutually exclusive application has been filed. If action is taken by the applicants (or any of them) looking toward removal of the conflict between the applications, the restrictions continue until such action has been approved by the Commission and the Commission's order with respect thereto is no longer subject to reconsideration by the Commission or to review by any court. See § 1.525. If the proceeding is designated for hearing, paragraph (a) of this section applies.

(i) Except as provided in subdivision (ii) of this subparagraph, the "Public Notices" issued at regular intervals listing all applications and major amends thereto which have been tendered (or, in non-broadcast services, accepted) for filing shall constitute public notice of the filing of a mutually exclusive application. See §§ 1.564(c), 1.962(e) and

21.27(b) of this chapter.

(ii) Where there is doubt as to whether two applications are in fact mutually exclusive, or where the conflict between the applications does not clearly appear from such regularly issued "Public Notices", the Commission will endeavor to issue specific public notices stating that there is a possibility of conflict between the applications. In such circumstances, the specific public notice, rather than the regularly issued "Public Notices" of applications tendered (or accepted) for filing, shall constitute public notice for purposes of this section. (Such public notices are based or. a preliminary review of the applications by the administrative staff and are accorded no significance in determining whether the applications should be designated for hearing.)

[30 F.R. 9271, July 24, 1965, as amended at 34 F.R. 8242, May 28, 1969]

§ 1.1205 Decision-making Commission personnel (restricted adjudicative proceedings).

The following categories of persons are designated as decision-making Commission personnel in restricted adjudicative proceeding:

(a) The Commissioners and their per-

sonal office staffs.

(b) The Chief of the Office of Opinions and Review and his staff.

(c) The Review Board and its staff.

(d) The Chief Hearing Examiner, the hearing examiners, and the staff of the Office of Hearing Examiners.

(e) The General Counsel and his staff.

(f) The Chief Engineer and his staff.

§ 1.1207 Restricted rule making proceedings.

Rule making proceedings which are required by statute to be decided on the record after opportunity for hearing, including the following, are "restricted" from the time they are designated for hearing until they are removed from hearing status, or have been decided by the Commission, and are no longer subject to reconsideration by the Commission or to review by any court:

(a) Any proceeding conducted pursuant to the provisions of sections 201(a), 204, 205, 213(a), 214(d), 221(c), or 222

of the Communications Act.

(b) Any proceeding involving the establishment of "charges, classifications, practices, regulations, and other terms and conditions", or the allocation of available facilities and stations among users, conducted pursuant to the provisions of section 201(c) (2) of the Communications Satellite Act of 1962

(c) Any rate making proceeding conducted pursuant to the provisions of section 201(c) (5) of the Communications

Satellite Act of 1962.

[30 F.R. 9271, July 24, 1965, as amended at 34 F.R. 8243, May 28, 1969]

§ 1.1209 Decision-making Commission personnel (restricted rule making proceedings).

The following categories of persons are designated as decision-making Commission personnel in restricted rule making proceedings:

(a) The Commissioners and their per-

sonal office staffs.

(b) The Chief of the Office of Opinions and Review and his staff.

(c) The Chief Hearing Examiner, the hearing examiners, and the staff of the Office of Hearing Examiners.

- (d) The Chief of the Common Carrier Bureau and his staff.
 - (e) The General Counsel and his staff.
 - (f) The Chief Engineer and his staff.
- (g) The Chief of the CATV Task Force and his staff when participating in proceedings involving service by common carriers to community antenna television systems.

[30 F.R. 9271, July 24, 1965, as amended at 32 F.R. 4358, Mar. 22, 1967]

PROHIBITED PRESENTATIONS

§ 1.1221 Presentations prohibited in restricted proceedings which have been designated for hearing.

Except as provided in § 1.1227, the following presentations are prohibited in restricted proceedings which have been designated for hearing:

- (a) Oral presentations. Persons outside the Commission and non-decision-making Commission personnel shall not, directly or indirectly, make or attempt to make any oral ex parte presentation.
- (b) Written presentations. Interested persons and non-decision-making Commission personnel shall not, directly or indirectly, make or attempt to make any written ex parte presentation.

§ 1.1223 Presentations prohibited in restricted application proceedings prior to their designation for hearing.

As provided in § 1.1203(b), certain application proceedings are "restricted" following the submission of a petition to deny or public notice of the filling of a mutually exclusive application. Except as provided in § 1.1227, no interested person shall, directly or indirectly, make or attempt to make any oral or written ex parte presentation to decision-making Commission personnel concerning such a proceeding. Nor, in the absence of public notice, shall such an ex parte presentation be made, directly or indirectly, by an interested person having actual knowledge that a mutually exclusive application has been filed.

§ 1.1225 Solicitation of ex parte presentations.

- (a) No person shall solicit or encourage others to make any presentation which he is himself prohibited from making under the provisions of this subpart.
- (b) Except as provided in § 1.1227, decision-making personnel shall not solicit or encourage ex parte presentations from any person, and shall not entertain ex

parte presentations which are made to them.

§ 1.1227 Permissible ex parte communications.

The following communications shall not be considered to be ex parte presentations prohibited by the provisions of this subpart:

- (a) Ex parte communications authorized by statute or by the Commission's rules. (See, for example, § 1.333(d).) However, pleadings which are required to be served but which may be ruled on ex parte do not fall within this category. (See, for example, §§ 1.296 and 1.298(a); such pleadings may not be submitted ex parte.)
- (b) Such ex parte communications initiated by the staff of the Common Carrier Bureau or the CATV Task Force as may be necessary for the adduction of record evidence in restricted rule making proceedings.
- (c) Any communication made by or to the General Counsel or his staff concerning judicial review of any matter which has been decided by the Commission.
- (d) Any communication from an agency of the Federal Government involving classified security information.
- (e) Any request for information solely with respect to the status of a restricted proceeding. (Interested persons, however, are prohibited from directly or indirectly soliciting ex parte status inquiries.)
- (f) Any communication between decision-making and non-decision-making Commission personnel prior to the designation of a restricted proceeding for hearing.

[30 F.R. 9271, July 24, 1965, as amended at 32 F.R. 4358, Mar. 22, 1967]

PROCEDURES FOR HANDLING EX PARTE COMMUNICATIONS

§ 1.1241 Written ex parte communications.

- (a) To the extent possible, written exparte communications not authorized under § 1.1227 will be forwarded to the Executive Director rather than to the addressee.
- (b) Unauthorized written ex parte communications which reach decision-making Commission personnel will be forwarded by them to the Executive Director. If the circumstances in which an unauthorized written ex parte presentation is made are not apparent from the presentation itself, a statement de-

scribing those circumstances shall be submitted to the Executive Director with the presentation.

(c) Unauthorized written ex parte communications, and all statements and correspondence pertaining thereto, will be placed in a public file, which will be associated with, but not made a part of, the file or record of the restricted proceeding to which the communication pertains. Prior to designation for hearing, no such communication, statement or correspondence shall be considered. without disclosure, in determining the merits of a restricted proceeding. After designation for hearing, no such materials shall be considered in determining the merits of a restricted proceeding unless they are made a part of the record of that proceeding.

(d) (1) If, in the judgment of the Executive Director, any unauthorized written ex parte presentation forwarded to him is prohibited by § 1.1221(b) or § 1.1223; or if in his judgment the presentation was solicited or encouraged in violation of § 1.1225; or if in his judgment the presentation should, for any other reason, be brought specifically to the attention of the parties, the Executive Director will serve copies of the presentation, together with copies of any statement describing the circumstances in which it was made, upon the parties to the proceeding.

(2) If the written presentation is voluminous or the parties numerous, or if other circumstances satisfy the Executive Director that service of the presentation would be unduly burdensome, he may (in lieu of service) notify the parties to the proceeding that the presentation has been made and that it is available for public inspection.

(e) A copy of any statement describing the circumstances in which any unauthorized written ex parte presentation was submitted will be forwarded to the person who submitted the presentation. Within 10 days after the statement is mailed to him, the person who submitted the presentation may himself file with the Executive Director a notarized statement with regard to these circumstances, which the Executive Director (if he deems service appropriate) may serve upon parties to the proceeding.

§ 1.1243 Oral ex parte communications.

(a) If an unauthorized oral ex parte presentation is made to decision-making Commission personnel, they will advise the person making the presentation that it is prohibited and terminate the discussion of such matters.

- (b) If an unauthorized ex parte presentation has in fact been made, the person to whom the presentation was made shall forward to the Executive Director a statement containing such of the following information as is known to him:
- (1) The name of the restricted proceeding.
- (2) The name and address of the person making the presentation and his relationship (if any) to the parties to the proceeding or their attorneys.
- (3) The date and time of the presentation, its duration, and the circumstances (telephone, personal interview, casual meeting, etc.) under which it was made.
- (4) A brief statement as to the substance of the matters discussed.
- (5) Whether the person making the presentation persisted in doing so after having been advised that the presentation is prohibited.
- (6) The date and time at which the statement was prepared.
- (c) All statements submitted to the Executive Director pursuant to the provisions of this section, and all correspondence pertaining thereto, will be placed in a public file, which will be associated with, but not made part of, the file or record of the restricted proceeding the presentation pertains. which Prior to designation for hearing, no such presentation, statement or correspondence shall be considered, without disclosure, in determining the merits of a restricted proceeding. After designation for hearing, no such materials shall be considered in determining the merits of a restricted proceeding unless they are made a part of the record of that proceeding.
- (d) All statements submitted to the Executive Director pursuant to the provisions of this section shall be served by the Executive Director on the parties to the proceeding. If the parties are numerous, or if other circumstances satisfy the Executive Director that service of the statement would be unduly burdensome, he may (in lieu of service) notify the parties to the proceeding that the presentation has been made and that a statement with respect to it is available for public inspection.
- (e) The Executive Director will forward to the person who made the presentation a copy of the statement prepared by the person to whom the presen-

tation was made. Within 10 days after that statement is mailed to him, the person who made the presentation may himself file with the Executive Director a notarized statement with respect to the substance of the presentation and the circumstances in which it was made, which the Executive Director (if he deems service appropriate) may serve upon parties to the proceeding.

§ 1.1245 Disclosure of information concerning ex parte presentations.

Any party to a restricted proceeding who has substantial reason to believe that an unauthorized ex parte presentation has been solicited, attempted or made, or who has information regarding such a presentation, shall promptly advise the Executive Director in writing of all the facts and circumstances concerning that presentation which are known to him.

SANCTIONS

§ 1.1251 Sanctions.

(a) Parties. Upon notice and hearing, any party to a restricted proceeding who directly or indirectly makes any unauthorized ex parte presentation, who encourages or solicits others to make any such presentation, or who fails to advise the Executive Director of the facts and circumstances concerning an unauthorized ex parte presentation (see § 1.1245), may be disqualified from further participation in that proceeding. Such alternative or additional sanctions as may be appropriate may be imposed.

(b) Commission personnel. Violations of the provisions of this subpart by Commission personnel will be disposed of in accordance with the procedures set forth in Administrative Order No. 10 and the penalties therein specified.

(c) Other persons. Such sanctions as may be appropriate under the circumstances will be imposed upon other persons who violate the provisions of this subpart.

APPENDIX A—A PLAN OF COOPERATIVE PROCE-DURE IN MATTERS AND CASES UNDER THE PROVISIONS OF SECTION 410 OF THE COM-MUNICATIONS ACT OF 1934

(Approved by the Federal Communications Commission October 25, 1938, and approved by the National Association of Railroad and Utilities Commissioners on November 17, 1938.)

PRELIMINARY STATEMENT CONCERNING THE PURPOSE AND EFFECT OF THE PLAN

Sec. 410 of the Communications Act of 1934 authorizes cooperation between the Federal

Communications Commission, hereinafter called the Federal Commission, and the State commissions of the several States, in the administration of said Act. Subsection (a) authorizes the reference of any matter arising in the administration of said Act to a board to be composed of a member or members from each of the States in which the wire. or radio communication affected by or involved in the proceeding takes place, or is proposed. Subsection (b) authorizes conferences by the Federal Commission with State commissions regarding the relationship between rate structures, accounts, charges, practices, classifications, and regulations of carriers subject to the jurisdiction of such State commissions and of said Federal Commission and joint hearings with State commissions in connection with any matter with respect to which the Federal Commission is authorized to act.

Obviously, it is impossible to determine in advance what matters should be the subject of a conference, what matters should be referred to a board, and what matters should be heard at a joint hearing of State commissions and the Federal Commission. It is understood, therefore, that the Federal Commission or any State commission will freely suggest cooperation with respect to any proceedings or matter affecting any carrier subject to the jurisdiction of said Federal Commission and of a State commission, and concerning which it is believed that cooperation will be in the public interest.

To enable this to be done, whenever a proceeding shall be instituted before any commission, Federal or State, in which another commission is believed to be interested, notice should be promptly given each such interested commission by the commission before which the proceeding has been instituted. Inasmuch, however, as failure to give notice as contemplated by the provisions of this plan will sometimes occur purely through inadvertence, any such failure should not operate to deter any commission from suggesting that any such proceeding be made the subject matter of cooperative action, if cooperation therein is deemed desirable.

It is understood that each commission whether or not represented in the National Association of Railroad and Utilities Commissioners, must determine its own course of action with respect to any proceeding in the light of the law under which, at any given time, it is called upon to act, and must be guided by its own views of public policy; and that no action taken by such Association can in any respect prejudice such freedom of action. The approval by the Association of this plan of cooperative procedure, which was jointly prepared by the Association's standing Committee on Cooperation between Federal and State commissions and said Federal Commission, is accordingly recommendatory only; but such plan is designed to be, and it is believed that it will be, a helpful step in the promotion of cooperative relations between the State commissions and said Federal Commission.

NOTICE OF INSTITUTION OF PROCEEDING

Whenever there shall be instituted before the Federal Commission any proceeding involving the rates of any telephone or telegraph carrier, the State commissions of the States affected thereby will be notified immediately thereof by the Federal Commission, and each notice given a State commission will advise such commission that, if it deems the proceeding one which should be considered under the cooperative provisions of the Act, it should either directly or through the National Association of Railroad and Utilities Commissioners, notify the Federal Commission as to the nature of its interest in said matter and request a conference, the creation of a joint board, or a joint hearing as may be desired, indicating its preference and the reasons therefor. Upon receipt of such request the Federal Commission will consider the same and may confer with the commission making the request and with other interested commission. or with representatives of the National Association of Railroad and Utilities Commissioners, in such manner as may be most suitable; and if cooperation shall appear to be practicable and desirable, shall so advise each interested State commission, directly, when such cooperation will be by joint conference or by reference to a joint board appointed under said Sec. 410 (a), and, as hereinafter provided, when such cooperation will be by a joint hearing under said Sec. 410 (b).

Each State commission should in like manner notify the Fedreal Commission of any proceeding instituted before it involving the toll telephone rates or the telegraph rates of any carrier subject to the jurisdiction of the Federal Commission.

PROCEDURE GOVERNING JOINT CONFERENCES

The Federal Commission, in accordance with the indicated procedure, will confer with any State commission regarding any matter relating to the regulation of public utilities subject to the jurisdiction of either commission. The commission desiring a conference upon any such matter should notify the other without delay, and thereupon the Federal Commission will promptly arrange for a conference in which all interested State commissions will be invited to be present.

PROCEDURE GOVERNING MATTERS REFERRED TO A BOARD

Whenever the Federal Commission, either upon its own motion or upon the suggestion of a State commission, or at the request of any interested party, shall determine that it is desirable to refer a matter arising in the administration of the Communications Act of 1934 to a board to be composed of a member or members from the State or States affected or to be affected by such matter, the procedure shall be as follows:

The Federal Commission will send a request to each interested State commission to nominate a specified number of members to serve on such board.

The representation of each State concerned shall be equal, unless one or more of the States affected chooses to waive such right of equal representation. When the member or members of any board have been nominated and appointed, in accordance with the provisions of the Communications Act of 1934, the Federal Commission will make an order referring the particular matter to such board, and such order shall fix the time and place of hearing, define the force and effect the action of the board shall have, and the manner in which its proceedings shall be conducted. The rules of practice and procedure, as from time to time adopted or prescribed by the Federal Commission, shall govern such board, as far as applicable.

PROCEDURE GOVERNING JOINT HEARINGS

Whenever the Federal Commission, either upon its own motion or upon suggestions made by or on behalf of any interested State commission or commissions, shall determine that a joint hearing under said Sec. 410 (b) is desirable in connection with any matter pending before said Federal Commission, the procedure shall be as follows:

(a) The Federal Commission will notify the general solicitor of the National Association of Railroad and Utilities Commissioners that said Association, or, if not more than eight States are within the territory affected by the proceeding, the State commissions interested, are invited to name Cooperating Commissioners to sit with the Federal Commission for the hearing and consideration of said proceeding.

(b) Upon receipt of any notice from said Federal Commission inviting cooperation, if not more than eight States are involved, the general solicitor shall at once advise the State commissions of said States, they being represented in the membership of the association. of the receipt of such notice, and shall request each such commission to give advice to him in writing, before a date to be indicated by him in his communication requesting such advice (1) whether such commission will cooperate in said proceeding, (2) if it will, by what commissioner it will be represented therein.

(c) Upon the basis of replies received, the general solicitor shall advise the Federal Commission what States, if any, are desirous of making the proceeding cooperative and by what commissioners they will be represented, and he shall give like advice to each State commission interested therein.

(d) If more than eight States are interested in the proceeding, because within territory for which rates will be under consideration therein, the general solicitor shall advise the president of the association that the association is invited to name a cooperating committee of State commissioners repre-

senting the States interested in said proceeding.

The president of the association shall thereupon advise the general solicitor in writing (1) whether the invitation is accepted on behalf of the association, and (2) the names of commissioners selected to sit as a cooperating committee. The president of the association shall have the authority to accept or to decline said invitation for the association, and to determine the number of commissioners who shall be named on the cooperating committee, provided that his action shall be concurred in by the chairman of the association's executive commit-In the event of any failure of the president of the association and chairman of its executive committee to agree, the second vice president of the association (or the chairman of its committee on cooperation between State and Federal commissions, if there shall be no second vice president) shall be consulted, and the majority opinion of the three shall prevail. Consultations and expressions of opinion may be by mail or telegraph.

- (e) If any proceeding, involving more than eight States, is pending before the Federal Commission, in which cooperation has not been invited by that Commission, which the association's president and the first and second vice presidents, or any two of them, consider should be made a cooperating proceeding, they may instruct the general solicitor to suggest to the Federal Commission that the proceeding be made a cooperative proceeding; and any State commission considering that said proceeding should be made cooperative may request the president of the association or the chairman of its executive committee to make such suggestion after consideration with the executive officers above named. If said Federal Commission shall assent to the suggestion, made as aforesaid, the president of the association shall have the same authority to proceed, and shall proceed in the appointment of a cooperating committee, as is provided in other cases involving more than eight States, wherein the Federal Commission has invited cooperation, and the invitation has been accepted.
- (1) Whenever any case is pending before the Federal Commission involving eight States or less, which a commission of any of said States considers should be made cooperative, such commission, either directly or through the general solicitor of the association, may suggest to the Federal Commission that the proceeding be made cooperative. If said Federal Commission accedes to such suggestion, it will notify the general solicitor of the association to that effect and thereupon the general solicitor shall proceed as is provided in such case when the invitation has been made by the Federal Commission without State commission suggestion.

APPOINTMENT OF COOPERATING COMMISSIONERS BY THE PRESIDENT

In the appointment of any cooperating committee, the president of the association shall make appointments only from commissions of the States interested in the particular proceeding in which the committee is to serve. He shall exercise his best judgment to select cooperating commissioners who are especially qualified to serve upon cooperating committees by reason of their ability and fitness; and in no case shall he appoint a commissioner upon a cooperating committee until he shall have been advised by such commissioner that it will be practicable for him to attend the hearings in the proceeding in which the committee is to serve, including the arguments therein, and the cooperative conferences, which may be held following the submission of the proceeding, to an extent that will reasonably enable him to be informed upon the issues in the proceeding and to form a reasonable judgment in the matters to be determined.

TENURE OF COOPERATORS

- (a) No State commissioner shall sit in a cooperative proceeding under this plan except a commissioner who has been selected by his commission to represent it in a proceeding involving eight States or less, or has been selected by the president of the association to sit in a case involving more than eight States, in the manner hereinbefore provided.
- (b) A commissioner who has been selected, as hereinbefore provided, to serve as a member of a cooperating committee in any proceeding, shall without further appointment, and without regard to the duration of time involved, continue to serve in said proceeding until the final disposition thereof, including hearings and conferences after any order or reopening, provided that he shall continue to be a State commissioner.
- (c) No member of a cooperating committee shall have any right or authority to designate another commissioner to serve in his place at any hearing or conference in any proceeding in which he has been appointed to serve.
- (d) Should a vacancy occur upon any cooperating committee, in a proceeding involving more than eight States, by reason of the death of any cooperating commissioner, or of his ceasing to be a State commissioner, or of other inability to serve, it shall be the duty of the president of the association to fill the vacancy by appointment, if, after communication with the chairman of the cooperating committee, it be deemed necessary to fill such vacancy.
- (e) In the event of any such vacancy occurring upon a cooperating committee involving not more than eight States, the vacancy shall be filled by the commission from which the vacancy occurs.

COOPERATING COMMITTEE TO DETERMINE RE-SPECTING ANY REPORT OF STATEMENT OF ITS ATTITUDE

(a) Whenever a cooperating committee shall have concluded its work, or shall deem such course advisable, the committee shall consider whether it is necessary and desirable to make a report to the interested State commissions, and, if it shall determine to make a report, it shall cause the same to be distributed through the secretary of the association, or through the general solicitor to all interested commissions.

(b) If a report of the Federal Commission will accompany any order to be made in said proceeding, the Federal Commission will state therein the concurrence or nonconcurrence of said cooperating committee in the decision or order of said Federal Commission.

CONSTRUCTION HEREOF IN CERTAIN RESPECTS EXPRESSLY PROVIDED

It is understood and provided that no State or States shall be deprived of the right of participation and cooperation as hereinbefore provided because of nonmembership in the association. With respect to any such State or States, all negotiations herein specified to be carried on between the Federal Commission and any officer of such association shall be conducted by the Federal Commission directly with the chairman of the commission of such State or States.

[28 F.R. 12462, Nov. 22, 1963, as amended at 29 F.R. 4801, Apr. 4, 1964]

APPENDIX B—INTERPRETATIONS OF FEE RULES AND PROCEDURES

Implementation of Commission's the schedule of filing fees has resulted in a number of questions requiring interpretations of the rules and administrative procedures applicable to the fee schedule. In view of the fact that many of these interpretations should serve as useful guidance to the general public, periodic releases of such interpretations are being made. The questions and interpretations which have been released are set forth below. Further information relative to the fee schedule may be obtained by reference to the folder on fees maintained in the Broadcast and Docket Reference Room of the Commission's offices at Washington, D.C. 64-1. Question. An application is original-

64-1. Question. An application is originally submitted before the effective date of the fees but is returned to the applicant as incomplete. If the corrected application is resubmitted after the effective date of the fee schedule, will the prescribed fee be required?

Interpretation. Applications which were originally submitted before the effective date of the fee schedule and returned for corrections etc., will not be required to be accompanied by the prescribed fee when resubmitted subsequent to the effective date of the fee schedule. However, if the resubmission is made on a new form, the original application must be enclosed to verify that

the original application was submitted before the effective date of the fee schedule.

64-2. Question. Will refunds be made to the applicant or to the attorney, manufacturer, or other representative who might have submitted payment?

Interpretation. All refunds will be made payable to the applicant, irrespective of the fact that payment of the fee may have been received from an attorney, manufacturer, or other representative.

64-3. Question. Will applications receive a preliminary screening to determine whether the appropriate fee has been paid, or will this determination be made only after the application has awaited its turn in the processing line?

Interpretation. Applications will be screened upon receipt in the processing division to determine whether the appropriate fee has been paid. The application will not take its place on the processing line until the full amount of the prescribed fee has been paid.

64-4. Question. When will fee credit be accorded to an applicant?

Interpretation. Fee credit will be accorded only in those instances where the application is returned for additional information or corrections, e.g., the application is undated or unsigned, applicable questions are unanswered, inconsistency in spelling of names, necessary frequency coordination committee letter has been omitted, etc. (However, the fact that a fee credit will be allowed upon the resubmission of an application which has been returned as incomplete will not be construed to mean that the original application was accepted for filing.) No credit will be accorded an applicant whose application has been dismissed, e.g., the applicant is not eligible for a license in the service in which he has applied, or the applicant has requested dismissal. See § 1.1103 (d) of the rules.

64-5. Question. An application for a radio license is returned to the applicant for correction after the fee has been paid. Applicant decides to abandon his application, and he either gives or sells his returned application bearing the fee payment stamp to a friend. The friend then applies for the same type of radio license enclosing the first applicant's returned application as credit towards the prescribed fee. Will the Commission accept this as payment?

Interpretation. The fee credit accorded to an applicant in connection with the resubmission of an application is a credit extended to the applicant of record and is not transferable to another party. Therefore, the application bearing the fee payment stamp is not acceptable as payment in this instance, and the second applicant must submit the appropriate fee with his application.

64-6. Question. Who must pay the fee required to accompany applications in the Broadcast Services for assignment of license or transfer of control?

Interpretation. The rules require only that such applications be accompanied by the prescribed fee. The parties involved in the assignment or transfer must determine amongst themselves how the fee should be paid. If both parties to the application are exempt under § 1.1111(b) of the rules, no fee will be required. However, where only one party may be exempt under § 1.1111(b), the application must be accompanied by the full amount of the prescribed fee.

64-7. Question. A licensee in the Industrial Radio services holds a license for a base mobile system in a particular area. He subsequently applies for an additional base station in the same area to communicate with the same mobile units. Commission procedures require that he also file one application for the existing base station and one for a license to cover the existing group of mobile units. Will the applicant be required to pay fees for each of the three applications which are required?

Interpretation. The applicant will be required to submit the prescribed fee with each of the three applications. By applying for an additional base station, the applicant is concurrently modifying his existing base station license and applying for a license to cover his group of mobile stations (where more than one base station serves a group of mobile stations, the mobile stations must be covered by a separate license).

64-8. Question. Will a fee be required to accompany an application to modify an aircraft station license to add crash beacon equipment?

Interpretation. Under §§ 1.1115(a) and 87.53(a) of the Commission's rules, all applications for modification of an aircraft station license must be accompanied by the prescribed fee. However, under the Commission's rules, an initial application for an aircraft station license can include crash beacon equipment even though the applicant may not plan to install the beacon equipment at the time the license is issued.

64-9. Question. Will the holder of a Novice, Technician, or Conditional Class Amateur license who is required to appear for an additional examination pursuant to § 97.35 of the rules be required to pay a fee?

Interpretation. No fee will be required with respect to examinations held pursuant to § 97.35 of the rules because no application is involved. Furthermore, in no event will the holder of a Novice Class license be required to pay an application filing fee. See §§ 1.1115(b) (6) and 97.55(b) of the rules, wherein Novice Class licensees are exempted from the payment of fees.

64-10. Question. Will fees be required for amendments to applications?

Interpretation. No fee will be required for an amendment to an application provided that the amendment does not constitute, in effect, a new application. However, if an application, as amended, would require a higher fee than originally paid, an amount equal to the difference must accompany the amendment. (E.g., an applicant files for a base mobile license in the Business Radio Service and pays the prescribed \$10 fee. He then submits an amendment to his application to request, instead, an authorization for an operational fixed microwave radio station in the same radio service. Therefore, he must accompany his amendment with a \$20 fee, the difference between the original fee paid and the \$30 fee prescribed for applications for operational fixed microwave station authorizations.)

64-11. Question. When will an amendment be considered as a new application for purposes of the fee schedule?

Interpretation. In determining when an amendment will be considered a new application for purposes of the fee schedule, the following guide lines will be used:

following guide lines will be used:
a. In the Common Carrier Services and the
Public Safety, Industrial, and Land Transportation Services, an amendment will be
considered a new application if it changes
the original application to another radio
service, e.g., from the Rural Radio Service to
the Domestic Public Land Mobile Radio
Service, or from the Special Industrial Radio
Service to the Business Radio Service.

b. In the Citizens Radio Service, an amendment will be considered a new application if it changes an application from either a Class B, C, or D station to an application for a Class A station, or, conversely, if it changes an application for a Class A station to an application for either a Class B, C, or D station. Amendments involving only changes between Class B, C, or D stations will not be considered as new applications and no fees will be required.

c. In the Aviation and Marine Radio Services, an amendment will be considered a new application if it changes the application to a different class station requiring a different application form.

d. In the Broadcast Services, an amendment to an application for renewal, assignment and/or transfer of control will be considered as a new application if it amounts to a major amendment as set forth in § 1.578(a) and (b) of the Commission's rules.

An amendment to a standard broadcast application will be considered as a new application if it requires the original application to receive a new file number pursuant to § 1.571(j) of the Commission's rules.

An amendment to a television broadcast application will be considered as a new application if it effects a major change as defined in § 1.572(a) (1) of the Commission's rules.

An amendment to an FM broadcast application will be considered as a new application if it effects a major change as defined in § 1.573(a) (1) of the Commission's rules.

64-12. [Reserved]

64-13. Question. What fees will apply in the various services with respect to the filing of an application for additional time to construct a radio station?

Interpretation. Applications for additional time to construct a radio station fall within the category "all other applications" as used in the fee rules applicable to the Broadcast, Safety & Special, and Common Carrier Services, respectively. Thus, such applications must be accompanied by a fee of \$10 in the Safety and Special and Common Carrier Services, and a fee of \$30 in the Broadcast Services (under § 1.1111, no fee is required if the application is filed with respect to a television translator station).

64-14. Question. Applications filed Civil Air Patrol stations are exempt from the payment of fees under §§ 1.1115(b)(8) and 87.53(b) of the rules. Does this exemption include applications filed in the Citizens Radio Service by Civil Air Patrol units?

Interpretation. The exemptions for Civil Air Patrol stations contained in §§ 1.1115(b) (8) and 87.53(b) of the rules relate to applications filed for Civil Air Patrol stations under Subpart 0 of Part 87 of the rules, and do not relate to applications which might be filed by Civil Air Patrol units under other provisions of the Commission's rules. should be emphasized that an exemption which is provided for a specific class of applicant filing in a particular service will not be accorded to that applicant if he should choose to apply in a service other than the one in which the exemption is provided.

64-15. Question. A Justice of the Peace applies for a Citizens license and requests that he be exempted from the payment of the prescribed fee because the license will be used in conjunction with his duties as a Justice of the Peace. Should a fee be re-

Interpretation. If the applicant is to be exempted, he must come within the provisions of §§ 1.1115(b)(2) and 95.23(b) which exempt applications filed in the Safety and Special Radio Services by governmental en-To come within this exemption, a Justice of the Peace must show that the cost of his equipment, as well as any fee required, would be borne by the governmental entity from which he derives his authority, and that the station will be used solely in conjunction with his duties as a Justice of the

64-16. Question. Will an application filed in the Amateur Radio Service for a special call sign as well as for an initial license or the renewal of an existing license require the payment of two fees, or will the \$20 fee for special call sign suffice for both requests?

Interpretation. An application for an Amateur special call sign is a separate and distinct request and the \$20 fee prescribed will not cover an additional request for either an initial license or the renewal or modification of an existing license. In such cases both fees will be required. However, the \$20 fee will cover the incidental modification of an existing license to indicate a new call sign.

64-17. Question. Are fees required to accompany applications filed for remote pickup base and mobile stations participating in the Emergency Broadcast System?

Interpretation. Certain remote pickup base and mobile stations licensed in the Auxiliary Broadcast Services and participating in the Emergency Broadcast System are installed and operated by the applicant at the request and expense of the Office of Civil Defense, Department of Defense. Applicants for such stations would, therefore, be reimbursed by the Office of Civil Defense for any fees paid in obtaining authorization for the construction and operation of these stations. Accordingly, no fees are required if the station has been installed and operated at the request and expense of the Office of Civil Defense.

64-18. Question. Will a fee be required to accompany an application submitted by an amateur radio club located in a Veterans Administration Hospital?

Interpretation. If it may be shown that the amateur club is operated at the expense of the Veterans Administration, no fee will be required. An exemption from the payment of fees is provided in §§ 1.1115(b) (6) and 97.55(b) of the rules for applications filed for amateur stations under military auspices. In adopting these rules, the Commission stated that the charging of a fee for such applications would be inappropriate because the cost of establishing such stations, including the payment of any fees, is borne by the Armed Service involved. An amateur club operated at the expense of the Veterans Administration would, therefore, be so closely related to a military amateur club as to fall within the exemption provided in §§ 1.1115(b) (6) and 97.55(b).

64-19. [Reserved]

[29 F.R. 4801, Apr. 4, 1964, as amended at 29 F.R. 12516, Sept. 2, 1964; 30 F.R. 14808. Nov. 30, 1965

PART 2—FREQUENCY ALLOCATIONS AND RADIO TREATY MATTERS: **GENERAL RULES AND REGULA-TIONS**

Subpart A-Definitions

Sec.

2.1 Definitions.

Subpart B-Allocation, Assignment, and Use of Radio Frequencies

International regulations in force. 2.100

2.101 Nomenclature of frequencies.

2.102 Assignment of frequencies.

2.103 Government use of non-Government frequencies.

2.104 Radio astronomy notification.

Application and format of the Table 2.105 of Frequency Allocations.

2.106 Table of Frequency Allocations.

Subpart C-Emissions

2.201 Emission, modulation, and transmission characteristics.

2.202 Bandwidths.

Subpart D—Call Signs and Other Forms of Identifying Radio Transmissions

Sec. 2.301 Station identification requirement.

2.302 Call signs.

2.303 Other forms of identification of stations.

Subpart E—Distress, Disaster, and Emergency Communications

2.401 Distress messages.

2.402 Control of distress traffic.

2.403 Retransmission of distress message

2.404 Resumption of operation after distress

2.405 Operation during emergency.

2.406 National defense; free service.

2.407 National defense; emergency authorization.

Subpart F—Equipment Type Approval and Type Acceptance

GENERAL

2.551 Program defined.

2.553 Radio equipment list.

2.555 Making available type approved or type accepted equipment for testing or inspection.

2.557 Limitation on availability of equipment files for public reference.

TYPE APPROVAL

2.561 Type approval.

2.563 Limitations on type approval.

2.565 Withdrawal or refusal of type approval.

2.567 Identification and changes in type approved equipment.

TYPE ACCEPTANCE

2.571 Type acceptance.

2.573 Limitation on type acceptance.

2.575 Withdrawal or refusal or type acceptance.

2.577 General information required for type acceptance.
2.579 Measurement data required for type

2.579 Measurement data required for type acceptance.

2.582 Measurement procedure for type ac-

2.582 Measurement procedure for type acceptance.

2.584 Identification and changes in type accepted equipment.

TECHNICAL INFORMATION FOR APPLICATION REFERENCE

2.591 Submission of technical information for application reference.

2.593 Identification and changes in equipment information filed for application reference.

Subpart G—Treaties and Other International Agreements

2.601 General.

2.602 Citation abbreviations used in this subpart.

2.603 Treaties and other international agreements relating to radio.

Subpart H—Prohibition Against Eavesdropping Sec.

2.701 Prohibition against use of a radio device for eavesdropping.

AUTHORITY: The provisions of this Part 2 issued under sec. 4, 303, 48 Stat. 1066, 1082, as amended; 47 U.S.C. 154, 303, unless otherwise noted.

SOURCE: The provisions of this Part 2 appear at 28 F.R. 12465, Nov. 22, 1963, unless otherwise noted.

Subpart A—Definitions

§ 2.1 Definitions.

may also participate.

The following definitions are issued: Active satellite. An earth satellite carrying a station intended to transmit or re-transmit radiocommunication signals.

Aeronautical advisory station. An aeronautical station used for advisory and civil defense communications primarily with private aircraft stations.

Aeronautical fixed service. A fixed service intended for the transmission of information relating to air navigation, preparation for and safety of flight.

Aeronautical fixed station. A station in the aeronautical fixed service.

Aeronautical mobile service. A mobile service between aeronautical stations and aircraft stations, or between aircraft stations, in which survival craft stations

Aeronautical multicom land station. An aeronautical station operating in the aeronautical multicom service.

Aeronautical multicom mobile station. A mobile station operating in the aeronautical multicom service.

Aeronautical multicom service. A mobile service not open to public correspondence, used to provide communications essential to conduct of activities being performed by or directed from private aircraft.

Aeronautical radionavigation service. A radionavigation service intended for the benefit of aircraft.

Aeronautical search and rescue station. A land or mobile station in the aeronautical mobile service used for communication with aircraft and other aeronautical search and rescue stations pertaining to search and rescue activities with aircraft.

Aeronautical station. A land station in the aeronautical mobile service. In certain instances an aeronautical station may be placed on board a ship or an earth satellite.

Aeronautical telemetering land station. A telemetering land station used in the flight testing of manned or unmanned aircraft, missiles, or major components thereof

Aeronautical telemetering mobile station. A telemetering mobile station used in the flight testing of manned or unmanned aircraft, missiles, or major components thereof.

Aeronautical utility land station. A land station located at airdrome control towers and used for control of ground vehicles and aircraft on the ground at airdromes.

Aeronautical utility mobile station. A mobile station used for communication, at airdromes, with the aeronautical utility land station, ground vehicles, and aircraft on the ground.

Aircarrier aircraft station. An aircraft station aboard an aircraft engaged in, or essential to, transportation of passengers or cargo for hire.

Aircraft station. A mobile station in the aeronautical mobile service on board an aircraft or an air-space vehicle.

Airdrome control station. An aeronautical station providing communication between an airdrome control tower and aircraft.

Amateur service. A service of self-training, intercommunication and technical investigations carried on by amateurs, that is, by duly authorized persons interested in radio technique solely with a personal aim and without pecuniary interest.

Amateur station. A station in the amateur service.

Aural broadcast STL station. A fixed station utilizing telephony for the transmission of aural program material between a studio and the transmitter of a broadcasting station other than an international broadcasting station, for simultaneous or delayed broadcast.

Aural broadcast intercity relay station. A fixed station utilizing telephony for the transmission of aural program material between broadcasting stations other than international broadcasting stations, for simultaneous or delayed broadcast.

Authorized frequency. The frequency assigned to a station by the Commission and specified in the instrument of authorization.

Authorized power. The power assigned to a radio station by the Commission and specified in the instrument of authorization. The authorized power

does not necessarily correspond to the power used by the Commission for purposes of its Master Frequency Record (MFR) and notification to the International Telecommunication Union.

Aviation instructional station. A land or mobile station in the aeronautical mobile service used for radiocommunications pertaining to instructions to students or pilots while actually operating aircraft or engaged in soaring activities.

Aviation services. Aviation services are primarily for the safe, expeditious and economical operation of aircraft. They include the aeronautical fixed service, aeronautical mobile service, aeronautical radionavigation service, and secondarily, the handling of public correspondence to and from aircraft.

Baseband. In the process of modulation, the baseband is the frequency band occupied by the aggregate of the modulating signals when first used to modulate a carrier.

Base station. A land station in the land mobile service carrying on a service with land mobile stations.

Broadcasting service. A radiocommunication service in which the transmissions are intended for direct reception by the general public. This service may include sound transmissions, television transmissions or other types of transmissions.

Broadcasting station. A station in the broadcasting service.

Carrier. In a frequency stabilized system, the sinusoidal component of a modulated wave whose frequency is independent of the modulating wave: or the output of a transmitter when the modulating wave is made zero: or a wave generated at a point in the transmitting system and subsequently modulated by the signal: or a wave generated locally at the receiving terminal which, when combined with the side bands in a suitable detector, produces the modulating wave

Carrier frequency. The frequency of the carrier.

Citizens radio service. A radiocommunication service of fixed, land, and mobile stations intended for personal or business radiocommunications, radiocommuni

Civil Air Patrol land station. A land station used exclusively for communi-

cations of the Civil Air Patrol.

Civil Air Patrol mobile station. A mobile station used exclusively for communications of the Civil Air Patrol.

Coast station. A land station in the maritime mobile service.

Common carrier fixed station. A fixed station open to public correspondence.

Common carrier land station. A land station open to public correspondence.

Common carrier mobile station. A mobile station open to public correspondence.

Communication-satellite earth station. An earth station in the communication-satellite service.

Communication-satellite service. A space service:

- —between earth stations, when using active or passive satellites for the exchange of communications of the fixed or mobile service, or
- —between an earth station and stations on active satellites for the exchange of communications of the mobile service, with a view to their re-transmission to or from stations in the mobile service.

Communication-satellite space station. A space station in the communication-satellite service, on an earth satellite.

Community antenna relay service. A fixed service, the stations of which are used for the transmission of television and related audio signals, and signals of standard and FM broadcasting stations, to a terminal point from which the signals are distributed to the public by cable.

Community antenna relay station. A fixed station in the community antenna relay service.

Deep space. Space at distances from the earth equal to or greater than the distance between the earth and the moon.

Disaster communications service. A service of fixed, land, and mobile stations licensed or authorized to provide essential communications incident to or in connection with disaster or other incidents which involve loss of communications facilities normally available or which require the temporary establishment of communications facilities beyond those normally available.

Domestic fixed public service. A fixed service, the stations of which are open to public correspondence, for radiocommunications originating and terminating solely at points all of which lie within:

(a) the State of Alaska, or (b) the State of Hawaii, or (c) the contiguous 48 states and the District of Columbia, or (d) a single possession of the United States. Generally, in cases where service is afforded on frequencies above 72 Mc/s, radiocommunications between the contiguous 48 States (including the District of Columbia) and Canada or Mexico, or radiocommunications between the State of Alaska and Canada, are deemed to be in the domestic fixed public service.

Domestic fixed public station. A fixed station in the domestic fixed public service.

Domestic public radiocommunication services. The land mobile and domestic fixed public services the stations of which are open to public correspondence.

Duplex operation. Operating method in which transmission is possible simultaneously in both directions.

Earth station. A station in the space service located either on the earth's surface, including on board a ship, or on board an aircraft.

Environmental communications. Communications in the maritime mobile service for the broadcast of information pertaining to the environmental conditions in which vessels operate, i.e., weather, sea conditions, time signals of a grade adequate for practical navigation, notices to mariners and hazards to navigation.

Experimental station. A station utilizing radio waves in experiments with a view to the development of science or technique. This definition does not include amateur stations.

Facsimile. A system of telecommunication for the transmission of fixed images, with or without half-tones, with a view to their reproduction in a permanent form.

Facsimile broadcasting station. A broadcasting station utilizing facsimile primarily.

Fixed earth station. An earth station intended to be used at a specified fixed point.

Fixed public control service. A fixed service carried on for the purpose of transmitting intelligence between transmitting or receiving stations in the public radiocommunication services and the message centers or control points associated therewith.

Fixed service. A service of radiocommunication between specified fixed points.

Fixed station. A station in the fixed service.

Flight test station. An aeronautical station used for the transmission of essential communications in connection with the testing of aircraft or major components of aircraft: Provided, however, flight test stations, when operating on the frequency 3281 kc/s, are desig-nated as land stations, only with respect to operation on the frequency 3281 kc/s.

FM broadcasting station. A broadcasting station utilizing telephony by means of frequency modulation, and when authorized under a Subsidiary Communications Authorization (SCA). utilizing F9 emissions.

Gc/s (gigacycle per second). A gigacycle per second (Gc/s) means one

thousand megacycles.

Harmful interference. Any emission, radiation or induction which endangers the functioning of a radionavigation service or of other safety services or seriously degrades, obstructs, or repeatedly interrupts a radiocommunication service operating in accordance with this chapter.

Hertz. A unit of frequency equivalent to one cycle per second. The terms Hertz (Hz) and cycle(s) per second (c/s) are synonymous and may be used inter-

changeably.

Industrial radio services. Any service of radiocommunication essential to, operated by, and for the sole use of, those enterprises which for purposes of safety or other necessity require radiocommunication in order to function efficiently. the radio transmitting facilities of which are defined as fixed, land, mobile or radiolocation stations.

Industrial, scientific andmedical equipment (ISM equipment). **Devices** which use radio waves for industrial, scientific, medical, or any other purposes including the transfer of energy by radio and which are neither used nor intended to be used for radiocommunication.

Instructional television fixed station. A fixed station operated by an educational organization and used primarily for the transmission of visual and aural instructional, cultural and other types of educational material to one or more fixed

receiving locations.

Instrument landing system. A radionavigation system which provides aircraft with horizontal and vertical guidance just before and during landing and, at certain fixed points, indicates the distance to the reference point of landing.

Instrument landing system glide path. A system of vertical guidance embodied in the instrument landing system which indicates the vertical deviation of the aircraft from its optimum path of descent.

Instrument landing system localizer, system of horizontal guidance embodied in the instrument landing system which indicates the horizontal deviation of the aircraft from its optimum path of descent along the axis of the runway.

International broadcasting station. A broadcasting station employing frequencies allocated to the broadcasting service between 5950 kc/s and 26100 kc/s, whose transmissions are intended to be received directly by the general public in foreign countries.

International control station. A fixed station in the fixed public control service associated directly with the international fixed public radiocommunication service.

International fixed public radio serv-A fixed service, the stations of ice. which are open to public correspondence and which, in general, is intended to provide radiocommunication between any one of the contiguous 48 states (including the District of Columbia) and the State of Alaska, or the State of Hawaii, or any U.S. possession or any foreign point; or between any U.S. possession and any other point; or between the State of Alaska and any other point; or between the State of Hawaii and any other point. In addition, radiocommunications within the contiguous 48 states (including the District of Columbia) in connection with the relaying of international traffic between stations which provide the above service, are also deemed to be in the international fixed public radiocommunication service; provided, however, that communications solely between Alaska, or any one of the contiguous 48 states (including the District of Columbia), and either Canada or Mexico are not deemed to be in the international fixed public radiocommunication service when such radiocommunications are transmitted on frequencies above 72 Mc/s.

International fixed public station. A fixed station in the international fixed

public radio service.

Interzone station. A fixed station in the public safety (police) radio service using radiotelegraphy (A1 emission) for communication with zone stations within the zone and with interzone stations in other zones.

Ionospheric scatter. The propagation of radio waves by scattering as a result of irregularities or discontinuities in the ionization of the ionosphere.

Kc/s (kilocycle per second). A kilocycle per second (kc/s) means one thou-

sand cycles per second.

Land mobile service. A mobile service between base stations and land mobile stations, or between land mobile stations.

Land mobile station. A mobile station in the land mobile service capable of surface movement within the geographical limits of a country or continent.

Land station. A station in the mobile service not intended to be used while in motion.

Land transportation radio service. Any private service of radiocommunication essential to the conduct of certain land transportation activities and operated for the use of persons engaged in those activities, the transmitting facilities of which are defined as fixed, land, mobile or radiolocation stations.

Localizer station. A radionavigation land station in the aeronautical radionavigation service which provides signals for the lateral guidance of aircraft with respect to a runway centerline.

Loran station. A long distance radionavigation land station transmitting synchronized pulses. Hyperbolic lines of position are determined by the measurement of the difference in the time of arrival of these pulses.

Marine radiobeacon station. A radionavigation land station, the emissions of which are intended to enable a ship station to determine its bearing or its direction in relation to the marine radiobeacon station.

Maritime mobile service. A mobile service between coast stations and ship stations, or between ship stations, in which survival craft stations may also participate.

Maritime radionavigation service. A radionavigation service intended for the benefit of ships.

Marker beacon. A transmitter in the aeronautical radionavigation service which radiates vertically a distinctive pattern for providing position information to aircraft.

Mc/s (megacycle per second). A megacycle per second (Mc/s) means one thousand kilocycles.

Meteorological aids service. A radiocommunication service used for meteorological, including hydrological, observations and exploration. Meteorological-satellite earth station. An earth station in the meteorological-satellite service.

Meteorological-satellite service. A space service in which the results of meteorological observations, made by instruments on earth satellites, are transmitted to earth stations by space stations on these satellites.

Meteorological-satellite space station. A space station in the meteorological-satellite service, on an earth satellite.

Mobile earth station. An earth station intended to be used while in motion or during halts at unspecified points.

Mobile, except television pickup, station. Any mobile station other than a television pickup station.

Mobile service. A service of radio communication between mobile and land stations, or between mobile stations.

Mobile station. A station in the mobile service intended to be used while in motion or during halts at unspecified points.

Modulation. The process of producing a wave some characteristic of which varies as a function of the instantaneous value of another wave, called the modulating wave.

Omni-directional range station. A radionavigation land station in the aeronautical radionavigation service providing direct indication of the bearing (omni-bearing) of that station from an aircraft.

Operational fixed station. A fixed station, not open to public correspondence, operated by and for the sole use of those agencies operating their own radiocommunication facilities in the public safety, industrial, land transportation, marine, or aviation service.

Operational land station. A land station, excluding aeronautical stations, not open to public correspondence, operated by and for the sole use of those agencies operating their own radiocommunication facilities in the public safety, industrial, land transportation, marine, or aviation services.

Operational mobile station. A mobile station, excluding aircraft stations, not open to public correspondence, operated by and for the sole use of those agencies operating their own radiocommunication facilities in the public safety, industrial, land transportation, marine, or aviation services.

Passive satellite. An earth satellite intended to transmit radiocommunication signals by reflection.

Port operations. Communications in or near a port, or in locks or waterways, between coast stations and ship stations. or between ship stations, in which messages are restricted to those relating to the operational handling, the movement and the safety of ships and, in emergency, to the safety of persons. Messages which are of a public correspondence nature shall be excluded.

Primary standard of frequency. The primary standard of frequency for radio frequency measurements shall be the national standard of frequency maintained by the National Bureau of Standards, Department of Commerce, Washington, D.C. The operating frequency of all radio stations will be determined by comparison with this standard or the standard signals of station WWV of the National Bureau of Standards.

Private aircraft station. An aircraft station on board an aircraft not operated as an air carrier.

Public correspondence. Any telecommunication which the offices and stations must, by reason of their being at the disposal of the public, accept for transmission.

Public safety radio service. Any service of radiocommunication essential either to the discharge of non-Federal governmental functions or the alleviation of an emergency endangering life or property, the radio transmitting facilities of which are defined as fixed, land, mobile, or radiolocation stations.

Racon. A radionavigation system transmitting, automatically or in response to a predetermined received signal, a pulsed radio signal with specific characteristics.

Racon station. A radionavigation land station which employs a racon.

Radar. A radiodetermination system based on comparison of reference signals with radio signals reflected, or retransmitted, from the position to be determined.

Radio. A general term applied to the use of radio waves.

Radio altimeter. A radionavigation equipment, on board an aircraft, which makes use of the reflection of radio waves from the ground to determine the height of the aircraft above the ground.

Radio astronomy. Astronomy based on the reception of radio waves of cosmic origin.

Radio astronomy service A service involving the use of radio astronomy.

Radio astronomy station. A station in the radio astronomy service.

Radiobeacon station. A station in the radionavigation service the emissions of which are intended to enable a mobile station to determine its bearing or direction in relation to the radiobeacon station.

Radiocommunication. Telecommunication by means of radio waves.

Radiodetermination. The determination of position, or the obtaining of information relating to position, by means of the propagation properties of radio waves.

Radiodetermination service. A service involving the use of radiodetermination.

Radiodetermination station. A station in the radiodetermination service.

Radio direction-finding. Radiodetermination using the reception of radio waves for the purpose of determining the direction of a station or object.

Radio direction-finding station. A radiodetermination station using radio direction-finding.

Radiolocation. Radiodetermination used for purposes other than those of radionavigation.

Radiolocation land station. A station in the radiolocation service not intended to be used while in motion.

Radiolocation mobile station. A station in the radiolocation service intended to be used while in motion or during halts at unspecified points.

Radiolocation service. A radiodetermination service involving the use of radiolocation.

Radionavigation. Radiodetermination used for the purposes of navigation, including obstruction warning.

Radionavigation land station. A station in the radionavigation service not intended to be used while in motion.

Radionavigation land test station (MTF). A radionavigation land station (Maintenance Test Facility) in the aeronautical radionavigation service which is used as a radionavigation calibration station for the transmission of essential information in connection with the testing and calibration of aircraft navigational aids, receiving equipment, and interrogators at predetermined surface locations. The primary purpose of this facility is to permit maintenance testing by aircraft radio service personnel

Radionavigation land test station (OTF). A radionavigation land station (Operational Test Facility) in the aero-



nautical radionavigation service which is used as a radionavigation calibration station for the transmission of essential information in connection with the testing and calibration of aircraft navigational aids, receiving equipment, and interrogators at predetermined surface locations. The primary purpose of this facility is to permit the pilot to check a radionavigation system aboard the aircraft prior to takeoff.

Radionavigation mobile station. A station in the radionavigation service intended to be used while in motion or during halts at unspecified points.

Radionavigation-satellite earth station. An earth station in the radionavigation-satellite service.

Radionavigation-satellite service. A service using space stations on earth satellites for the purpose of radionavigation, including, in certain cases, transmission or re-transmission of supplementary information necessary for the operation of the radionavigation system.

Radionavigation-satellite space station. A space station in the radionavigation-satellite service, on an earth satellite.

Radionavigation service. A radiodetermination service involving the use of radionavigation.

Radio range station. A radionavigation land station in the aeronautical radionavigation service providing radial equisignal zones.

Radiosonde. An automatic radio transmitter in the meteorological aids service usually carried on an aircraft, free balloon, kite or parachute, and which transmits meteorological data.

Radiosonde station. A station in the meteorological aids service employing a radiosonde.

Radio waves (or Hertzian waves). Electromagnetic waves of frequencies lower than 3,000 Gc/s [3,000,000 Mc/s], propagated in space without artificial guide.

Remote pickup broadcast base station. A base station licensed for communicating with remote pickup broadcast mobile stations.

Remote pickup broadcast mobile station. A land mobile station licensed for the transmission of program material and related communications from the scene of events which occur outside a studio to broadcasting station, and for communicating with other remote pickup broadcast base and mobile stations.

Safety service. A radiocommunication service used permanently or temporarily for the safeguarding of human life and property.

Ship station. A mobile station in the maritime mobile service located on board a vessel, other than a survival craft, which is not permanently moored.

Simplex operation. Operating method in which transmission is made possible alternately in each direction, for example, by means of manual control.

Spacecraft. Any type of space vehicle including an earth satellite or a deep-space probe, whether manned or unmanned.

Space research earth station. An earth station in the space research service.

Space research service. A space service in which spacecraft or other objects in space are used for scientific or technological research purposes.

Space research space station. A space station in the space research service.

Space service. A radiocommunication service:

- between earth stations and space stations,
- -or between space stations,
- or between earth stations when the signals are re-transmitted by space stations, or transmitted by reflection from objects in space, excluding reflection or scattering by the ionosphere or within the earth's atmosphere.

Space station. A station in the space service located on an object which is beyond, is intended to go beyond, or has been beyond, the major portion of the earth's atmosphere.

Space telecommand. The use of radiocommunication for the transmission of signals to a space station to initiate, modify or terminate functions of the equipment on a space object, including the space station.

Space telemetering. The use of telemetering for the transmission from a space station of results of measurements made in a spacecraft, including those relating to the functioning of the spacecraft.

Space tracking. Determination of the orbit, velocity or instantaneous position of an object in space by means of radio determination, excluding primary radar, for the purpose of following the movement of the object.

Standard broadcasting station. A broadcasting station operated on a frequency in the band 535-1605 kilocycles.

Standard frequency service. A radiocommunication service for scientific technical and other purposes, providing the transmission of specified frequencies of stated high precision, intended for general reception.

Stationary satellite. A satellite, the circular orbit of which lies in the plane of the earth's equator and which turns about the polar axis of the earth in the same direction and with the same period as those of the earth's rotation.

Survival craft station. A mobile station in the maritime or aeronautical mobile service intended solely for survival purposes and located on any lifeboat, liferaft, or other survival equipment.

Telecommunication. Any transmission, emission or reception of signs, signals, writing, images, and sounds or intelligence of any nature by wire, radio, optical or other electromagnetic systems.

Telegraphy. A system of telecommunication which is concerned in any process providing transmission and reproduction at a distance of documentary matter, such as written or printed matter or fixed images, or the reproduction at a distance of any kind of information in such form. The foregoing definition appears in the International Telecommunication Convention, but, for the purposes of the Commission's rules, telegraphy shall mean, unless otherwise specified, "A system of telecommunication for the transmission of written matter by the use of a signal code."

Telemetering. The use of telecommunication for automatically indicating or recording measurements at a distance from the measuring instrument.

Telemetering fixed station. A fixed station, the emissions of which are used for telemetering.

Telemetering land station. A land station, the emissions of which are used for telemetering.

Telemetering mobile station. A mobile station, the emissions of which are used for telemetering.

Telephony. A system of telecommunication set up for the transmission of speech or, in some cases, other sounds.

Television. A system of telecommunication for transmission of transient images of fixed or moving objects.

Television broadcasting station. A broadcasting station utilizing both television and telephony to provide combi-

nation and simultaneous visual and aural programs intended to be received directly by the general public.

Television inter-city relay station. A fixed station used for intercity transmission of television program material and related communications for use by television broadcast stations.

Television pickup station. A land mobile station used for the transmission of television program material and related communications from the scenes of events occurring at points removed from television broadcast station studios to television broadcast stations.

Television STL station (studio-transmitter link). A fixed station used for the transmission of television program material and related communications from a studio to the transmitter of a television broadcast station.

Terrestrial service. Any radio service defined in this Part, other than a space service or the radio astronomy service.

Terrestrial station. A station in a terrestrial service.

Tropospheric scatter. The propagation of radio waves by scattering as a result of irregularities or discontinuities in the physical properties of the troposphere.

Zone station. A fixed station in the public safety (police) radio service using radiotelegraph (A1 emission) for communication with other similar stations in the same zone and with an interzone station.

(Sec. 308, 48 Stat., as amended, 1084; 47 U.S.C. 308) [28 F.R. 12465, Nov. 22, 1963, as amended at 30 F.R. 7155, May 28, 1965; 30 F.R. 8679, July 9, 1965; 30 F.R. 8839, July 14, 1965; 30 F.R. 13370, Oct. 21, 1965; 31 F.R. 14395, Nov. 9, 1966; 33 F.R. 10856, July 31, 1968; 34 F.R. 5379, Mar. 19, 1969; 34 F.R. 5656, Mar. 26, 1969]

Subpart B—Allocation, Assignment, and Use of Radio Frequencies

§ 2.100 International regulations in force.

The Radio Regulations (Geneva, 1959), which became effective internationally on May 1, 1961, were incorporated to the extent practicable in Subparts A and B of this part and became effective nationally on December 1, 1961. The Radio Regulations were subsequently revised, in part, by the Extraordinary Administrative Radio Conference (Geneva, 1963) which specified January 1, 1965 as the effective date of the revision. The partial revision has

also been incorporated to the extent practicable in Subparts A and B of this part and is applicable nationally, effective July 1, 1965.

[30 F.R. 7156, May 28, 1965]

§ 2.101 Nomenclature of frequencies.

Band No.	Frequency subdivision				Frequency range
4 5 6 7 8	LF (lo MF (n HF (h	very low w frequencedium figh frequency	ency) frequen dency).		Below 30 kc/s. 30 to 300 kc/s. 300 to 3000 kc/s. 3 to 30 Mc/s. 30 to 300 Mc/s.
9		(ultra	high	íre-	300 to 3000 Mc/s.
10	SHF	(super	high	fre-	3 to 30 Gc/s.
11	e quen EHF quen	extreme	ly high	íre-	30 to 300 Gc/s.

§ 2.102 Assignment of frequencies.

- (a) Except as otherwise provided in this section, the assignment of frequencies and bands of frequencies to all stations and classe: of stations and the licensing and authorizing of the use of all such frequencies between 10 kc/s and 90 Gc/s, and the actual use of such frequencies for radiocommunication or for any other purpose, including the transfer of energy by radio, shall be in accordance with the Table of Frequency Allocations in § 2.106.
- (b) On the condition that harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations the following exceptions to paragraph (a) of this section may be authorized:
- (1) In individual cases the Commission may, without rule-making proceedings, authorize on a temporary basis only, the use of frequencies not in accordance with the Table of Frequency Allocations for projects of short duration or emergencies where the Commission finds that important or exceptional circumstances require such utilization: Provided, That such authorizations are not intended to develop a service to be operated on frequencies other than those allocated such service.
- (2) A station for the development of techniques or equipment to be employed by services or classes of stations set forth in columns 8 and 9 of the Table of Frequency Allocations may be authorized the use of frequencies allocated to those services or classes of stations.
- (3) Experimental stations engaged solely in scientific or technical radio experiments not related to an existing or proposed service nor intended to de-

velop a proposed service or specific use of radio may be authorized the use of any frequency which is in a band allocated to the fixed, land mobile or broadcasting services or to one of these services shared with another service.

- (4) Experimental stations engaged in communications essential to research projects when existing communication facilities are inadequate may be authorized the use of any frequency which is in a band allocated to the fixed, land mobile or broadcasting services or to one of those services shared with another service.
- (5) Experimental stations engaged solely in ionospheric sounding by means of the technique of sweeping a band of frequencies may be authorized the use of any band or bands or frequencies not allocated, on an exclusive or shared basis, to the radio astronomy service.
- (6) Experimental stations to be operated pursuant to a contractual agreement with the United States Government and intended for the sole and express purpose of developing equipment or a technique to be employed by stations belonging to and operated by the United States may be authorized the use of any frequency which is not in a band allocated, on an exclusive or shared basis, to the radio astronomy service.
- (7) Experimental stations intended for the sole and express purpose of developing equipment or a technique to be employed by stations under the jurisdiction of a foreign government may be authorized the use of any frequency which is not in a band allocated to the amateur service or the radio astronomy service.
- (8) In the event a band is reallocated so as to delete its availability for use by a particular service, the Commission may provide for the further interim use of the band by stations in that service for a temporary, specific period of time.
- (c) The use of frequencies in the bands above 25 Mc/s allocated exclusively to Government stations and the use of frequencies below 25 Mc/s which may not be in accordance with § 2.106 may be authorized to non-Government stations in those instances where the Commission finds, after consultations with the Government appropriate agency agencies, that such assignment is necessary for intercommunication with Government stations or where such use by non-Government stations is required for coordination with Government activities.



(d) Aircraft stations may communicate with stations of the maritime mobile service. They shall then conform to those provisions of the Radio Regulations which relate to the maritime mobile service. For this purpose aircraft stations should use the frequencies allocated to the maritime mobile service. However, having regard to interference which may be caused by aircraft stations at high altitudes, maritime mobile frequencies in the bands above 30 Mc/s shall not be used by aircraft stations in any specific area without the prior agreement of all administrations of the area in which interference is likely to be caused. In particular, aircraft stations operating in Region 1 should not use frequencies in the bands above 30 Mc/s allocated to the maritime mobile service by virtue of any agreement between administrations in that Region.

(e) Non-Government services operating on frequencies in the band 25-50 Mc/s must recognize that it is shared with various services of other countries: that harmful interference may be caused by skywave signals received from distant stations of all services of the United States and other countries radiating power on frequencies in this band; and that no protection from such harmful interference generally can be expected. Persons desiring to avoid such harmful interference should consider operation on available frequencies higher in the radio spectrum not generally subject to this type of difficulty.

(f) The stations of a service shall use frequencies so separated from the limits of a band allocated to that service as not to cause harmful interference to the service to which the frequency bands immediately adjoining are allocated.

(g) In the bands above 25 Mc/s which are allocated to the non-Government land mobile service, fixed stations may be authorized on the following conditions:

(1) That such stations are authorized in the service shown in Column 11 of the Table of Frequency Allocations in the band in question:

(2) That harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations.

[28 F.R. 12465, Nov. 22, 1963, as amended at 30 F.R. 7156, May 28, 1965; 34 F.R. 12138, July 19, 1969]

§ 2.103 Government use of non-Government frequencies.

- (a) Government stations may be authorized to use non-Government frequencies in the bands above 25 Mc/s if the Commission finds that such use is necessary for coordination of Government and non-Government activities: Provided, however, That:
- (1) Government operation on non-Government frequencies shall conform with the conditions agreed upon by the Commission and the Office of Emergency Planning (the more important of which are contained in subparagraphs (2), (3), and (4) of this paragraph), a complete list of which is available for public examination at each of the Commission's Field Engineering and Monitoring Bureau Field Offices:

(2) Such operations shall be in accordance with Commission rules governing the service to which the frequencies involved are allocated;

(3) Such operations shall not cause harmful interference to non-Government stations and, should harmful interference result, that the interfering Government operation shall immediately terminate: and

(4) Government operation has been certified as necessary by the non-Government licensees involved and this certification has been furnished, in writing, to the Government agency with which communication is required.

§ 2.104 Radio astronomy station notification.

- (a) Pursuant to No. 639AC, Article 9A and Section F of Appendix 1A to the international Radio Regulations (as revised, Geneva, 1963), operators of radio astronomy stations desiring international recognition of their use of specific radio astronomy frequencies or bands of frequencies for reception, should file the following information with the Commission for inclusion in the Master International Frequency Register:
- (1) The center of the frequency band observed, in kc/s up to 30,000 kc/s inclusive, and in Mc/s above 30,000 kc/s.
- (2) Date of putting into use (actual or foreseen, as appropriate).
- (3) Name and location of the station, including geographical co-ordinates in degrees and minutes.
- (4) Width of frequency band observed by the station.

- (5) Antenna type and dimensions, effective area and angular coverage in azimuth and elevation.
- (6) Maximum hours of reception (G.M.T.) of the frequency band shown in subparagraph (1) of this paragraph.
- (7) Overall receiving system noise temperature (°K).
- (8) Class of observations to be taken on the frequency band shown in sub-paragraph (1) of this paragraph. Class A observations are those in which the sensitivity of the equipment is not a primary factor. Class B observations are those of such a nature that they can be made only with advanced low-noise receivers using the best techniques.
- (b) Observations being conducted on frequencies or frequency bands not allocated to the radio astronomy service should be reported as in paragraph (a) of this section for information purposes. Information in this category will not be submitted for entry in the Master International Frequency Register and protection from interference will not be afforded such operations by stations in other services.

[30 F.R. 7156, May 28, 1965]

- § 2.105 Application and format of the Table of Frequency Allocations.
- (a) In the Table of Frequency Allocations below 25 Mc/s, the authority extended to stations in the fixed service, unless otherwise specified, extends only to those stations in the following categories of service:
 - (1) Aeronautical fixed.
- (2) Fixed (in U.S. possessions and Alaska).
 - (3) International fixed public.
- (b) In the Table of Frequency Allocations between 5,000 and 25,000 kc/s, the authority extended to stations in the mobile service unless otherwise specified, extends only to those stations in the following categories of service:
 - (1) Aeronautical mobile.
 - (2) Maritime mobile.
- (c) Columns 1, 2, 3, and 4 of the Table of Frequency Allocations indicate the allocation applicable on a worldwide basis or in Region 2 in accordance with the Geneva Radio Regulations.
- (d) In Column 6 (above 25 Mc/s) the letter G means Federal Government radio stations, i.e., those belonging to and op-

- erated by the United States. The symbol NG means other than Federal Government radio stations, i.e., those whose frequencies are assigned by the Commission.
- (e) Column 10 lists certain frequencies which have national or international significance and are used for specific standardized purposes. These frequencies may be made available for assignment to stations which conform to the nature of service or station listed in Column 11 opposite the assignable frequency. The assignment and use of the frequencies listed in Column 10 is limited to those stations which, by definition, are included in the services and classes of stations (Column 8 or 9) to which the frequency band (Column 7) is allocated.
- (f) In Column 11 "SERVICES" are in large block print and "stations" in small print.
- (g) Where, in Columns 2, 4, and 8, a pand is indicated as allocated to more than one service, such services are listed in the following order:
- (1) Services, the names of which are printed in all capital letters (example: FIXED); these services are called "primary" services;
- (2) Services, the names of which are italicized and are printed with the first letter capitalized and all others small type (example: *Radionavigation*); these services are called "permitted" services;
- (3) Services, the names of which are printed as in subparagraph (2) of this paragraph, but not italicized (example: Mobile); these are called "secondary" services, which are on a noninterference basis to the primary and permitted services.

NOTE 1. Geneva Radio Regulation No. 138: Permitted and primary services have equal rights, except that, in the preparation of frequency plans, the primary service, as compared with the permitted service, shall have prior choice of frequencies.

Note 2. Geneva Radio Regulation No. 139: Stations of a secondary service: (a) Shall not cause harmful interference to stations of primary or permitted services to which frequencies are already assigned or to which frequencies may be assigned at a later date; (b) cannot claim protection from harmful interference from stations of a primary or permitted service to which frequencies are already assigned or may be assigned at a later date; (c) can claim protection, however, from harmful interference from stations of the same or other secondary serverors.

ice(s) to which frequencies may be assigned at a later date.

(h) The following symbols are used to designate footnotes in the Table of Frequency Allocations.

(1) Any footnote consisting of three digits or three digits and a one or two

letter suffix, e.g., (170) or (215A), denotes a paragraph in the Geneva (1959) Radio Regulations as amended by the Space Conference (Geneva, 1963). Where such a footnote is applicable, without modification, to the national Table of Frequency Allocations, the sym-

§ 2.106 Table of Frequency Allocations.

	Worldwide		Region 2	United S	tates
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	5	6.
Below 10 (157)	(Not allocated).			Below 10	(Not allocated).
10-14	RADIONAVIGA- TION. Radiolocation.				
14-19. 95 (159)	FIXED. MARITIME MOBILE. (158)				
19. 95–20 05 (159)	STANDARD FRE- QUENCY. (160)				
20. 05-70 (159) (161)	FIXED. MARITIME MOBILE. (158)				
70-90		70-90 (164)	FIXED, MARITIME MOBILE. (158) MARITIME RADIONAVIGA- TION. (162) Radiolocation.		
90-110		90-110 (166) (167)	RADIONAVIGA- TION. FIXED. Maritime mobile. (158)		
110-130		110-130 (164) (167) (168)	FIXED. MARITIME MOBILE. MARITIME RADIO- NAVIGATION. (162) Radiolocation.		
130-160		130–160 (167)	FIXED. MARITIME MOBILE.		
160-200		160-200	FIXED. (179)		

See footnotes at end of table.

bol appears in the national table as well as in Column 1, 2, 3 or 4.

(2) Any footnote consisting of the letters US followed by one or more digits, e.g., US1, denotes a stipulation the application of which is a matter of agreement between the Commission and other

appropriate Government agencies.

(3) Any footnote consisting of the letters NG followed by one or more digits, e.g., NG1, is a stipulation applicable to non-Government stations.

[28 F.R. 12466, Nov. 22, 1963, as amended at 30 F.R. 7156, May 28, 1965]

	<u> </u>	1		
Band (kHz)	Service	Class of station	Fre- quency (kHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
Below 10	(Not allocated).			
10-14	RADIONAVIGA- TION.	Radionavigation land. Radionavigation mobile.		RADIONAVIGA- TION.
14-19.95	FIXED.	Fixed.		INTERNATIONAL FIXED PUBLIC
19. 95-20. 05	STANDARD FRE- QUENCY.	Standard frequency.	20	Standard frequency.
20. 05-59	FIXED.	Fixed.		INTERNATIONAL FIXED PUBLIC
59-61	STANDARD FRE- QUENCY.	Standard frequency.	60	Standard frequency.
61-70	FIXED.	Fixed.		INTERNATIONAL FIXED PUBLIC
70-90 (US103)	FIXED. Radiolocation.	Fixed. Radiolocation land. Radiolocation mobile.		INTERNATIONAL FIXED PUBLIC RADIOLOCATION
90-110 (166) (US18) (US104)	RADIONAVIGA- TION.	Radionavigation land. Radionavigation mobile.		RADIONAVIGA- TION.
110-130 (US103)	FIXED. MARITIME MOBILE. Radiolocation.	Coast. Fixed. Radiolocation land. Radiolocation mobile. Ship.		FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC MARITIME MOBILE. RADIOLOCATION
130-160	FIXED. MARITIME MOBILE.	Coast. Fixed. Ship.		FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC MARITIME MOBILE.
160-200	FIXED.	Fixed.		FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC

Title 47—Chapter I

	Worldwide		Region 2	United	States
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	5	6
200-285		200-285	AERONAUTICAL RADIO- NAVIGATION. Aeronautical mobile.		
285-315	MARITIME RADIO- NAVIGATION (radiobeacons). Aeronautical radio- navigation.				
315-325		315–325	MARITIME RADIO- NAVIGATION (radiobeacons). Aeronautical radio- navigation.		
325-405	AERONAUTICAL RADIO- NAVIGATION. Aeronautical mobile. (181)				
405–415 (182)		405-415	MARITIME RADIO- NAVIGATION (radio direction- finding). Aeronautical radio- navigation. Aeronautical mobile.		
415-490	MARITIME MOBILE. (185) (186)				
490-510 (187)	MOBILE (distress and calling).				
510-525		510-525	MOBILE. Aeronautical radio- navigation. (188)		
525-535		525-535	MOBILE. Broadcasting. (191) Aeronautical radio- navigation. (188)		
535-1605	BROADCASTING.				

See footnotes at end of table.

T. 10							
Federal Communications Commission							
Band (kHz)	Service	Class of station	Frequency (kHz)	$\mathbf{Nature} egin{cases} \mathbf{OF} \ \mathbf{SERV-} \\ \mathbf{ICES} \\ \mathbf{of} \ \mathbf{stations} \end{cases}$			
7	8	9	10	11			
200-285	AERONAUTICAL RADIO- NAVIGATION. (US18) Aeronautical mobile.	Aeronautical. Aircraft. Radionavigation land.		AERONAUTICAL MOBILE. AERONAUTICAL RADIO- NAVIGATION.			
285-325 (US16)	MARITIME RADIO- NAVIGATION. (US18)	Radionavigation land.		MARITIME RADIO- NAVIGATION,			
325-405	AERONAUTICAL RADIO- NAVIGATION. (US18) Aeronautical mobile.	Aeronautical. Aircraft. Radionavigation land.		AERONAUTICAL MOBILE. AERONAUTICAL RADIO- NAVIGATION.			
405-415 (US18)	MARITIME RADIO- NAVIGATION (radio direction- finding). Aeronautical radio- navigation. Aeronautical mobile. (US19)	Aeronautical. Aircraft. Radionavigation land. Radionavigation mobile.	410	Radio direction- finding.			
415-490	MARITIME MOBILE.	Coast. Ship.		MARITIME MOBILE (telegraphy).			
490-510	MOBILE.	Coast. Mobile.	500	Distress and calling.			
510-535 (US14)							
535-1605 (US15) (NG16)	BROADCASTING.			Standard broadcast- ing.			

Title 47—Chapter I

Wo	rldwide		Region 2	United 8	States
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	5	6
1605–1800		1605–1800	FIXED. MOBILE. AERONAUTICAL RADIONAVIGA- TION. Radiolocation.		
1800-2000		1800-2000 (198)	AMATEUR. FIXED. MOBILE except aeronautical mobile. RADIONAVIGA- TION.		
2000-2065		2000-2065	FIXED. MOBILE.		
		_			

See footnotes at end of table.

	Federal Communications Commission							
Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV- ICES of stations				
7	8	9	10	11				
1605-1715 (US97)	AERONAUTICAL RADIONAVIGA- TION. FIXED. LAND MOBILE. MARITIME MOBILE. RADIOLOCA- TION.	Base. Mobile. Fixed. Land mobile. Radionavigation land.	1638 1708	AERONAUTICAL FIXED. AERONAUTICAL RADIONAVIGA- TION. FIXED (in Alaska). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. PUBLIC SAFETY. RADIOLOCATION. Remote pickup broadcast base. Remote pickup broadcast mobile. Radionavigation land. Do.				
1715–1750	FIXED. LAND MOBILE. MARITIME MOBILE. RADIOLOCATION.	Base. Mobile. Fixed. Land mobile. Ship.		AERONAUTICAL FIXED. FIXED (in Alaska). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. PUBLIC SAFETY. RADIOLOCATION.				
1750-1800 (NG14)	FIXED. MOBILE. RADIOLOCATION.	Fixed. Land. Mobile.		DISASTER. RADIOLOCATION.				
1800-2000 (198) (US18)	RADIONAVIGA- TION. Amateur. (NG15)	Amateur. Loran.		AMATEUR. Loran.				
2000-2035	MARITIME MOBILE. (NG19)	Coast. Ship.		MARITIME MOBILE.				
2035-2065	MARITIME MOBILE.	Coast.		Coast (telegraphy).				

209

Worldwide			Region 2		United States	
Band (kHz)	Service	Band	(kHz)	Service	Band (kHz)	Allocation
1	2	:	3	4	5	6
2065-2107		200	85-2107	MARITIME MOBILE. (200)		
2107-2170		210	07-2170	FIXED. MOBILE.		
2170-2194	MOBILE (distress and calling). (201)					
2194-2300		219		FIXED. MOBILE.		
				MOBILE.		
2300-2495		230	0-2495	FIXED. MOBILE. BROADCASTING. (202)		
2495-2505		249	05-2505 (203) (204)	STANDARD FREQUENCY.		

	Federal	Communications Com	nission	
Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV-ICES of stations
7	8	9	10	11
2065-2068. 5 (200)	MARITIME MOBILE.	Coast. Ship.		MARITIME MOBILE (telephony).
2068. 5- 2078. 5	MARITIME MOBILE.	Ship.		Ship (wideband tele- graphy, facsimile, and special trans- mission systems).
2078. 5- 2089. 5 (200)	MARITIME MOBILE.	Coast. Ship.		MARITIME MOBILE (tele- phony).
2089. 5- 2092. 5	MARITIME MOBILE.	Ship.		Ship (calling, telegraphy).
2092, 5-2107 (200)	MARITIME MOBILE.	Coast. Ship.		MARITIME MOBILE.
2107-2170	FIXED. LAND MOBILE. MARITIME MOBILE. (NG19)	Base. Coast. Fixed. Land mobile. Ship.		AERONAUTICAL FIXED. FIXED (in Alaska). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. PUBLIC SAFETY.
2170-2194 (201)	MOBILE.	Aircraft. Coast. Ship. Survival craft.	2182	AERONAUTICAL MOBILE (telephony). MARITIME MOBILE (telephony). (NG22) Distress and calling frequency.
2194-2495	FIXED. LAND MOBILE, MARITIME MOBILE. (NG19)	Base. Coast. Fixed. Land mobile. Ship.		AERONAUTICAL FIXED. FIXED (in Alaska). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. PUBLIC SAFETY.
2495-2505 (US68)	STANDARD FREQUENCY. Radio astronomy. (US74)	Radio astronomy. Standard frequency.	2500	RADIO ASTRONOMY. Standard frequency.
·				

Worldwide			Region 2	United States	
Band (kHz)	Service	Bạnd (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	5	6
2505-2625		2505-2625	FIXED. ² MOBILE.		
2625-2850		2625-2850	FIXED. MOBILE.		
2850-3025	AERONAUTICAL MOBILE. (R)				
3025-3155	AERONAUTICAL MOBILE. (OR)				
3155-3200	FIXED. MOBILE except aeronautical mobile. (R)				
3200-3230	FIXED. MOBILE except aeronautical mobile. (R) BROADCASTING. (202)				
3230-3400	FIXED. MOBILE except aeronautical mobile. BROAD CASTING. (202)				

	Federal Communications Commission					
Band (kHz)		Class of station	Frequency (kHz)	Nature OF SERV- ICES of stations		
7	8	9	10	11		
2505-2850 (N G 20)	FIXED. LAND MOBILE. MARITIME MOBILE.	Base. Coast. Fixed. Land mobile. Ship.	2638 2738 2804 2808 2812	AERONAUTICAL FIXED (in Alaska). FIXED (in Alaska). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. Intership (telephony). (NG44) Do. PUBLIC SAFETY. Zone and interzone police. Do. Do.		
2850-3025 (NG61)	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.		
3025-3155	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.		
3155-3200	FIXED. LAND MOBILE. MARITIME MOBILE.	Base. Coast. Fixed. Land mobile. Ship.		AERONAUTICAL FIXED. FIXED (in Alaska and Puerto Rico). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. PUBLIC SAFETY.		
3200-3250 (N G 20)	FIXED. LAND MOBILE. MARITIME MOBILE.	Base. Coast. Fixed. Land mobile. Ship.		AERONAUTICAL FIXED. FIXED (in Alaska). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. PUBLIC SAFETY.		
3230-3240 (US105)	FIXED. LAND MOBILE. MARITIME MOBILE.	Base. Coast. Fixed. Land mobile. Ship.		AERONAUTICAL FIXED. FIXED (in Alaska). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. PUBLIC SAFETY.		

	Worldwide		Region 2		United States	
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation	
1	2	3	4	5	6	
3400-3500	AERONAUTICAL MOBILE. (R)					
3500-4000		3500-4000	AMATEUR. FIXED. MOBILE except aeronautical mobile. (R)			
4000-4063	FIXED.					
4063-4438	MARITIME MOBILE. (208) (209)					

	Federal	Communications Com	mission	
Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
3240-3400 (US105)	FIXED. LAND MOBILE. MARITIME MOBILE. MOBILE.	Base. Coast. Fixed. Land. Land mobile. Mobile. Ship.		AVIATION (flight test and aeronautica fixed only). FIXED (in Alaska). INDUSTRIAL. INTERNATIONAL FIXED PUBLIC. MARITIME MOBILE. PUBLIC SAFETY.
3400-3500 (NG61)	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
3500-4000	AMATEUR.	Amateur.		AMATEUR.
4000-4063	FIXED.	Fixed.	xed.	
4063-4139.5	MARITIME MOBILE.	Ship.		Ship (telephony).
4139.5- 4142.5 (US82)	MARITIME MOBILE.	Ship. Coast.		Ship. Coast. (telephony, simplex).
4142.5- 4162.5	MARITIME MOBILE.	Ship.		Ship (wideband teleg- raphy, facsimile, and special trans- mission systems).
4162. 5-4166	MARITIME MOBILE.	Ship. Buoy. Interrogating Coast.		Ship. Buoy. Inter- rogating Coast. (Oceanographic data transmission).
4166- 4172. 25	MARITIME MOBILE.	Ship.		Ship (narrow-band direct-printing telegraph and data transmission systems).
4172. 25-4178	MARITIME MOBILE.	Ship.		Ship (high traffic, telegraphy).
4178-4187	MARITIME MOBILE.	Ship.		Ship (calling, teleg- raphy).
4187-4231	MARITIME MOBILE.	Ship.		Ship (low traffic, telegraphy).
4231-4361	MARITIME MOBILE.	Coast.		Coast (wide-band and manual telegraphy, facsimile, special and data transmission systems and direct-printing telegraph systems).
4361-4438	MARITIME MOBILE.	Coast.		Coast (telephony).

Worldwide			Region 2	United States	
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	5	6
4438-4650		4438-4650	FIXED. MOBILE except aeronautical mobile. (R)		
4650-4700	AERONAUTICAL MOBILE. (R)				
4700-4750	AERONAUTICAL MOBILE. (OR)				
4750-4850		4750-4850	FIXED. BROADCASTING. (202)		
4850-4995	FIXED. LAND MOBILE. BROADCASTING. (202)				
4995-5005	STANDARD FREQUENCY. (204) (210)				
5005-5060	FIXED. BROADCASTING. (202)				
5060-5250	FIXED.				
5250-5450		5250-5450	FIXED. LAND MOBILE.		
5450-5480		5450-5480	AERONAUTICAL MOBILE. (R)		
5480-5680	AERONAUTICAL MOBILE. (R)				
5680-5730	AERONAUTICAL MOBILE. (OR)				
5730-5950	FIXED.				
					

4850-4995 4995-5005 (US68)	Service 8	Class of station	Fre- quency	OF SERV-
4438-4650 4650-4700 (NG60) (NG61) 4700-4750 4750-4850 4850-4995 (US68)	8		(kHz)	Nature ICES of stations
4650-4700 (NG60) (NG61) 4700-4750 4750-4850 4850-4995		9	10	11
(NG60) (NG61) 4700-4750 4750-4850 4850-4995 4995-5005 (US68)	FIXED. MOBILE.	Base. Fixed. Mobile.	Fixed.	
4750-4850 4850-4995 4995-5005 (US68)	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
4850-4995 4995-5005 (US68)	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
4995-5005 (US68)	FIXED.	Fixed.		AERONAUTICAL FIXED. FIXED (in Alaska). INTERNATIONAI FIXED PUBLIC.
(US68)	FIXED.	Fixed.		AERONAUTICAL FIXED. FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC.
	STANDARD FREQUENCY. Radio astronomy. (US74)	Radio astronomy. Standard frequency.	5000	RADIO ASTRONOMY. Standard frequency.
5005-5450	FIXED.	Fixed.		AERONAUTICAL FIXED. FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC. Zone and interzon police.
5450-5680 (NG61)	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
5680-5730	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
5730-5950	FIXED.	Fixed.		AERONAUTICAL FIXED. FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC.

Worldwide			Region 2	United	United States	
Band Hz)	Service	Band (kHz)	Service	Band (kHz)	Allocation	
1	2	3	4	5	6	
5950-6200	BROADCASTING.					
6200-6525 (211)	MARITIME MOBILE.					
6525-6685 6685-6765	AERONAUTICAL MOBILE. (R) AERONAUTICAL MOBILE. (OR)					
6765-7000	FIXED.					
7000-7100	AMATEUR.					
7100-7300		7100-7300	AMATEUR.			
7300-8195	FIXED.					

	Federal	Communications Com	mission	
Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
5950-6200 (NG25)	BROADCASTING.	International broadcasting.		International broadcasting.
6200-6210. 4	MARITIME MOBILE.	Ship.		Ship (telephony).
6210. 4- 6216. 5 (US82)	MARITIME MOBILE.	Ship. Coast.		Ship. Coast. (tele- phony, simplex).
6216. 5- 6244. 5	MARITIME MOBILE.	Ship.		Ship (wide-band tele- graphy, fascimile, and special transmis- sion systems).
6244. 5-6248	MARITIME MOBILE.	Ship. Buoy. Interrogating Coast.		Ship. Buoy. Interrogating Coast. (Oceanographic data transmission).
6248-6258. 25	MARITIME MOBILE.	Ship.		Ship (narrow-band direct-printing tele- graph and data transmission systems).
6258. 25-6267	MARITIME MOBILE.	Ship.		Ship (high traffic, telegraphy).
6267-6280.5	MARITIME MOBILE.	Ship.		Ship (calling, teleg- raphy).
6280. 5- 6345. 5	MARITIME MOBILE.	Ship.		Ship (low traffic, telegraphy).
6345. 5-6514	MARITIME MOBILE.	Coast.		Coast (wide-band and manual telegraphy, facsimile, special and data transmission systems and direct- printing telegraph systems). (NG27)
6514-6525	MARITIME MOBILE.	Coast.		Coast (telephony).
6525-6685 (NG61)	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
6685-6765	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
6765-7000	FIXED.	Fixed.		AERONAUTICAL FIXED. FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC.
7000-7300	AMATEUR.	Amateur.		AMATEUR.
7300-8195	FIXED.	Fixed.		AERONAUTICAL FIXED. FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC. Zone and interzone police.

Title 47—Chapter I

•	Worldwide		Region 2	United States	
	1		1		<u> </u>
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	. 5	6
8195-8815	MARITIME MOBILE. (213)				
8815-8965	AERONAUTICAL MOBILE. (R)				
8965-9040	AERONAUTICAL MOBILE. (OR)				
9040-9500	FIXED.				

Federal Communications Commission

Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
8195-8281. 2	MARITIME MOBILE.	Ship.		Ship (telephony).
8281, 2-8288 (US82)	MARITIME MOBILE.	Ship. Coast.		Ship. Coast. (tele- phony, simplex).
8288-8328	MARITIME MOBILE.	Ship.		Ship (wide-band tele- graphy, facsimile, and special trans- mission systems).
8328-8331.5	MARITIME MOBILE.	Ship. Buoy. Interrogating Coast.	Buoy. Interrogating	
8331. 5- 8341. 75	MARITIME MOBILE.	Ship.		Ship (narrow-band direct-printing tele- graph and data transmission systems).
8341.75-8356	MARITIME MOBILE.	Ship.		Ship (high traffic, telegraphy).
8356-8374	MARITIME MOBILE.	Ship.		Ship (calling, teleg- raphy).
8374-8459. 5	MARITIME MOBILE.	Ship.		Ship (low traffic, telegraphy).
8459. 5- 8728. 5	MARITIME MOBILE.	Coast.		Coast (wide-band and manual telegraphy, facsimile, special and data transmission systems and direct- printing telegraph systems).
8728. 5-8815	MARITIME MOBILE.	Coast.		Coast (telephony).
8815-8965	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		
8965-9040	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
9040-9500	FIXED.	Fixed.		AERONAUTICAL FIXED. FIXED (in Alaska). INTERNATIONAL FIXED PUBLIC.

Title 47---Chapter I

Worldwide			Region 2	United 8	United States	
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation	
1	2	3	4	5	6	
9500-9775	BROADCASTING.					
9775-9995	FIXED.					
9995-10005	STANDARD FREQUENCY. (204) (214) (215)					
10005-10100	AERONAUTICAL MOBILE. (R)					
10100-11175	FIXED.					
11175-11275	AERONAUTICAL MOBILE. (OR)					
11275-11400	AERONAUTICAL MOBILE. (R)					
11400-11700	FIXED. (216)					
11700-11975	BROADCASTING.	,				
11975-12330	FIXED.					
12330-13200	MARITIME MOBILE. (213)					

	Federal	Communications Comm	nission	
Band (kHz)	Service	Class of station Frequency (kHz)		Nature OF SERV- ICES of stations
7	8	9	10	11
9500-9775	BROADCASTING.	International broad- casting.		International broad- casting.
9775-9995	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC
9995-10005 (215) (US83)	STANDARD FREQUENCY. Radio astronomy. (US74)	Radio astronomy. Standard frequency.	10000	RADIO ASTRONO- MY. Standard frequency.
10005-10100 (NG61)	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
10100-11175	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC
11175-11275	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
11275-11400 (NG61)	AERONAUTICAL MOBILE. (R)	Aeronautical, Aircraft.		AERONAUTICAL MOBILE.
11400-11700	FIXED.	Fixed.	Fixed.	
11700-11975	BROADCASTING.	International broad- casting.		International broad- casting.
11975-12330	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC
12330-12421	MARITIME MOBILE.	Ship.		Ship (telephony).
12421- 12431. 5 (US82)	MARITIME MOBILE.	Ship. Coast.		Ship. Coast. (telephony, simplex).
12431.5- 12479.5	MARITIME MOBILE.	Ship.		Ship (wideband teleg raphy, facsimile, and special trans- mission systems).
12479. 5- 12483	MARITIME MOBILE.	Ship. Buoy. Interrogating Coast.		Ship. Buoy. Inter- rogating Coast. (Oceanographic data transmission).
12483- 12503, 25	MARITIME MOBILE.	Ship.		Ship (narrow-band direct-printing telegraph and data transmission systems).
12503.25- 12534	MARITIME MOBILE.	Ship.		Ship (high traffic, telegraphy).

Worldwide		Region 2		United States	
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	5	6
13200-13260	AERONAUTICAL MOBILE. (OR)				
13260-13360	AERONAUTICAL MOBILE. (R)				
13360-14000 (217)	FIXED.				
14000-14350	AMATEUR. (218)				
14350-14990	FIXED.				
14990-15010	STANDARD FRE- QUENCY. (204) (219)				
15010-15100	AERONAUTICAL MOBILE. (OR)		· · · · · · · · · · · · · · · · · · ·		
15100-15450	BROADCASTING.				

	Federal Communications Commission						
Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV- ICES of stations			
7	8	9	10	11			
12534-12561	MARITIME MOBILE.	Ship.		Ship (calling, telegra- phy).			
12561-12689	MARITIME MOBILE.	Ship.		Ship (low traffic, telegraphy).			
12689- 13107. 5	MARITIME MOBILE.	Coast.		Coast (wide-band and manual telegraphy, facsimile, special and data transmission systems and direct- printing telegraph systems).			
13107.5- 13200	MARITIME MOBILE.	Coast.		Coast (telephony).			
13200-13260	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.			
13260-13360 (NG61)	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.			
13360-14000	FIXED.	Fixed.	13560	AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC. Industrial, scientific and medical equip- ment.			
14000-14350	AMATEUR.	Amateur.		AMATEUR.			
14350-14990	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.			
14990-15010 (US68)	STANDARD FREQUENCY. Radio astronomy. (US74.)	Radio astronomy. Standard frequency.		RADIO ASTRON- OMY. Standard frequency.			
15010-15100	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.			
15100-15450	BROADCASTING.	International broad- casting.		International broad- casting.			

	Worldwide		Region 2	United 8	States
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	5	6
15450-15762	FIXED.				
15762-15768	FIXED. Space Research. (215A)				(US100)
15768-16460	FIXED.				
16460-17360	MARITIME MOBILE. (213)				
17360-17700	FIXED.				
17700-17900	BROADCASTING.				
17900-17970	AERONAUTICAL MOBILE. (R)				

	Federal	Communications Comm	nission	
Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV-ICES of stations
7	8	9	10	11
15450-16460	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.
16460-16565	MARITIME MOBILE.	Ship.		Ship (telephony).
16565-16576 (US82)	MARITIME MOBILE.	Ship. Coast.		Ship. Coast. (telephony, simplex).
16576- 16636. 5	MARITIME MOBILE.	Ship.		Ship (wideband teleg- raphy, facsimile, and special trans- mission systems).
16636.5- 16640	MARITIME MOBILE.	Ship. Buoy. Interrogating Coast.		Ship. Buoy. Interrogating Coast. (Oceanographic data transmission).
16640- 16660.5	MARITIME MOBILE.	Ship.		Ship (narrow-band direct-printing telegraph and data transmission systems).
16660. 5- 16712	MARITIME MOBILE.	Ship.		Ship (high traffic, telegraphy).
16712-16748	MARITIME MOBILE.	Ship.		Ship (calling, telegraphy).
16748- 16917. 5	MARITIME MOBILE.	Ship		Ship (low traffic, telegraphy).
16917.5- 17255	MARITIME MOBILE.	Coast.		Coast (wide-band and manual telegraphy, facsimile, special and data transmis- sion systems and direct-printing tele- graph systems).
17255-17360	MARITIME MOBILE.	Coast.		Coast (telephony).
17360-17700	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.
17700-17900	BROADCASTING.	International broad- casting.		International broad- casting.
17900-17970 (NG61)	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
	•	•	. –	

Worldwide		Region 2		United 8	United States	
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation	
1	2	3	4	5	6	
17970-18036	AERONAUTICAL * MOBILE. (OR)					
18030-18036	FIXED. Space Research. (215A)				(US100)	
18036-19990	FIXED.					
19990-20010	STANDARD FREQUENCY. (204) (220) (221) (221A)					
20010-21000	FIXED.					
21000-21450	AMATEUR.			-		
21450-21750	BROADCASTING.					
21750-21850	FIXED.					
21850-22000	AERONAUTICAL FIXED. AERONAUTICAL MOBILE. (R)					
22000-22720	MARITIME MOBILE.					

	Federal	Communications Comm	ission .	
Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
17970-18030	AERONAUTICAL MOBILE. (OR)	Aeronautical. Aircraft.		AERONAUTICAL MOBILE.
18030-19990	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.
19990-20000 (US68)	STANDARD FREQUENCY. Radio astronomy. (US74)	Radio astronomy. Standard frequency.	20000	Standard frequency.
20000-20010 (US84)	STANDARD FREQUENCY. Space research.	Earth. Space. Standard frequency.	20000	Standard frequency.
20010-21000	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.
21000-21450	AMATEUR.	Amateur.		AMATEUR.
21450-21750	BROADCASTING.	International broadcasting.		International broadcasting.
21750-21850	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.
21850-22000	AERONAUTICAL FIXED. AERONAUTICAL MOBILE. (R)	Aeronautical. Aeronautical fixed. Aircraft.		AERONAUTICAL FIXED. AERONAUTICAL MOBILE.
22000- 22094.5	MARITIME MOBILE.	Ship.		Ship (telephony).
22094.5- 22112 (US82)	MARITIME MOBILE.	Ship. Coast.		Ship. Coast. (telephony, simplex).

***************************************	Worldwide		Region 2	United States	
Band (kHz)	Service	Band (kHz)	Service	Band (kHz)	Allocation
1	2	3	4	5	6
22 720-23200	FIXED.				
23200-23350	AERONAUTICAL.				
	AERONAUTICAL FIXED. AERONAUTICAL MOBILE. (OR)				
23350-24990 (222)	FIXED. LAND MOBILE.				

	Federal	Communications Comm	nission	
Band (kHz)	Service	Class of station	Frequency (kHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
22112- 22160. 5	MARITIME MOBILE.	Ship.		Ship (wideband teleg- raphy, facsimile, and special transmission systems).
22160.5- 22164	MARITIME MOBILE.	Ship. Buoy. Interrogating Coast.		Ship. Buoy. Interrogating Coast. (Oceanographic datatransmission).
22164- 22184. 5	MARITIME MOBILE.	Ship.		Ship (narrow-band direct-printing tele- graph and data trans mission systems).
22184. 5- 22222. 5	MARITIME MOBILE.	Ship.		Ship (high traffic, telegraphy).
22222, 5- 22267, 5	MARITIME MOBILE.	Ship.		Ship (calling, teleg- raphy).
22267. 5- 22374	MARITIME MOBILE.	Ship.		Ship (low traffic, telegraphy).
22374- 22624.5	MARITIME MOBILE.	Coast.		Coast (wide-band and manual telegraphy, facsimile, special and data transmission systems and direct- printing telegraph systems).
22624.5- 22720	MARITIME MOBILE.	Coast.		Coast (telephony).
22720-23200	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.
23200-23350	AERONAUTICAL FIXED. AERONAUTICAL MOBILE. (OR)	Aeronautical. Aeronautical fixed. Aircraft.		AERONAUTICAL FIXED. AERONAUTICAL MOBILE.
23350-24990	FIXED.	Fixed.		AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.

Worldwide			Region 2	United S	states
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
24. 99–25. 0 1	STANDARD FREQUENCY. (204) (223)			24. 99-25. 01	G, NG. (US68)
25. 01-25. 07	FIXED. MOBILE except aeronautical mobile.			25. 01-25. 33	NG.
25. 07-25. 11	MARITIME MOBILE. (224)				
25. 11-25. 6	FIXED. MOBILE except aeronautical mobile.			25. 33-25. 6	G.
25. 6-26. 1	BROADCASTING.			25. 6-26. 1	G, NG.
26, 1-27, 5 (225) (226)	FIXED. MOBILE except aeronautical mobile.			26. 1-26. 48	NG.
•				26. 48-26. 95 (US10)	G.
				26. 95-27. 54	NG.
		·			
27.5-28		27.5-28	METEOROLOGI- CAL AIDS. FIXED. MOBILE.	27. 54-28	G.
28-29.7	AMATEUR.	·		28-29.7	AMA- TEUR. (US1)

	Communications Comm		
Service	Class of station	Frequency (MHz)	Nature OF SERVICES of stations
8	9	10	11
STANDARD FREQUENCY. Radio astronomy. (US74)	Radio astronomy. Standard frequency.	25. 0	RADIO ASTRONOMY. Standard frequency.
LAND MOBILE. (NG50)	Land mobile. 2		INDUSTRIAL.
BROADCASTING.	International broadcasting.		International broadcasting.
LAND MOBILE.	Base. Land mobile.	26. 11- 26. 47 (NG32)	Remote pickup broad- cast base; remote pickup broadcast mobile.
		26. 62	Civil air patrol land; civil air patrol mobile.
FIXED.	Fixed.	26. 955	INTERNATIONAL FIXED PUBLIC.
CITIZENS. (US1)	Fixed. Land. Mobile.	27.12	Industrial, scientific and medical equip- ment.
FIXED. MOBILE.	Fixed. Land. Mobile.		
LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.
AMATEUR.	Amateur.		AMATEUR.
	STANDARD FREQUENCY. Radio astronomy. (US74) LAND MOBILE. (NG50) BROADCASTING. LAND MOBILE. CITIZENS. (US1) FIXED. MOBILE.	STANDARD FREQUENCY. Radio astronomy. (US74) LAND MOBILE. (NG50) BROADCASTING. Land mobile. Base. Land mobile. FIXED. CITIZENS. (US1) Fixed. Land. Mobile. Fixed. Land. Mobile. Fixed. Land. Mobile. LAND MOBILE. Base. Land mobile.	Service Class of station quency (MHz) 8 9 10 STANDARD Radio astronomy. Standard frequency. 25. 0 LAND MOBILE. Base. 25. 02-25. 32 (NG32) BROADCASTING. International broadcasting. LAND MOBILE. Base. 26. 47 (NG32) CITIZENS. (US1) Fixed. 26. 955 CITIZENS. (US1) Fixed. Land. Mobile. Fixed. Land. Mobile. Land mobile.

	Worldwide		Region 2		United States	
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation	
1	2	3	4	5	6	
29. 7–30. 005	FIXED. (228) (229) (231) (232) MOBILE.			29.7-29.89	NG.	
				29. 89-29. 91 (232)	G.	
			·	29. 91-30	NG.	
30. 005- 30. 01	FIXED. (228) (229) (231) MOBILE. SPACE RESEARCH. SPACE (Satellite identification).			30-30. 56	G. (US94)	
30. 01–37. 75	FIXED. (228) (229) (230) (231) MOBILE.			30. 56-32	NG.	
				32-33 (231)	G.	
				33-34	NG.	
				34-35 (231)	G.	
				35-36	NG.	
				36-37 (231)	G.	

	Federal	Communications Com	mission	
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
29. 7-29. 8	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.
29. 8-29. 89	FIXED. (232)	Fixed.	29. 81- 29. 88 (N G31)	AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.
29, 91-30	FIXED. (232)	Fixed.	29. 92- 29. 99 (N G31)	AERONAUTICAL FIXED. INTERNATIONAL FIXED PUBLIC.
30. 56-32	LAND MOBILE.	Base, Land mobile,		INDUSTRIAL. LAND TRANS- PORTATION. PUBLIC SAFETY.
33-33.01	LAND MOBILE.	Base.		LAND TRANS-
33. 01-33. 11	LAND MOBILE.	Base. Land mobile.		PORTATION. PUBLIC SAFETY.
33, 11-33, 41	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.
33. 41-34	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY.
35–35, 2	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.
35. 2-35. 68	LAND MOBILE.	Base. Land mobile.		DOMESTIC PUBLIC.
35, 68–36	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.

Worldwide		Region 2		United States	
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
				37-38	NG.
37. 75–38. 25	FIXED. (228) (229) (231) MOBILE. Radio astronomy.				
38. 25-41	FIXED. (228) (229)			38–39	G. (US81)
(235) (236)	FIXED. (228) (229) (230) (231) MOBILE.			39-40	NG. (US94)
				40-42 (236)	G. (US94)
41-50		41-50	FIXED. (228) (230) (237) MOBILE.	42-46. 6	NG.
				46.6-47	G.
				47-49.6	NG.
				49.6-50	G.
50-54		50-54	AMATEUR.	50-54	AMA- TEUR. (US1)

	Federa	Communications Com	mission	
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature SERVICES of stations
7	8	9	10	11
37-37.01	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.
37. 01-37. 43 (NG 59)	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY.
37. 43-37. 89	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.
37. 89–38	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY.
39-40	LAND MOBILE.	Base. Land mobile.	_	PUBLIC SAFETY.
			40.68	Industrial, scientific and medical equip- ment.
42-42. 95	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY.
42. 95-43. 2	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.
43. 2-43. 68	LAND MOBILE.	Base. Land mobile.		DOMESTIC PUBLIC.
43. 68-44. 61	LAND MOBILE.	Base. Land mobile.		LAND TRANSPORTA- TION.
44. 61–46. 6	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY.
47-47. 43	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY.
47. 43-47. 69	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY. INDUSTRIAL.
47. 69-49. 6	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.
50-51	AMATEUR.	Amateur.	-	AMATEUR.

Worldwide			Region 2	United 8	tates
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
54-68		54-68	FIXED. (228) (237) MOBILE. BROADCASTING.	54-72 (US95)	NG.
68-73. 0		68-73.0	FIXED. MOBILE. BROADCASTING.	72-73	NG.
				(US20)	
73–74. 6		73-74.6	RADIO ASTRONOMY. (258A) (253B)	73-74.6	G, NG. (US21) (US100)
74.6-75.4		74. 6-75. 4 (259)	AERONAUTICAL RADIONAVIGA- TION.	74. 6-75. 4	G, NG.
75. 4-88		75. 4-88	FIXED. MOBILE. BROADCASTING.	75. 4-76 (US20)	NG.
				76-88 (US23)	NG.
88-100		88-100	BROADCASTING.	88-108 (US23)	NG.
100-108		100-108	BROADCASTING.	(US93)	
108-117. 975	AERONAUTICAL RADIONAVIGA- TION.			108-117, 975 (US93)	G, NG.

	Federal	Communications Comm	ission	
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
54-72	BROADCASTING.	Television broadcast- ing.	55. 25 59. 75 61. 25 65. 75 67. 25 71. 75	Video Sound Channel 2. Video Sound Channel 3. Video Sound Channel 4.
72-73 (NG56)	FIXED. (NG1) (NG3) (NG49)	Operational fixed.	72.02- 72.98 (NG33)	Operational fixed.
73-74.6	RADIO ASTRON- OMY. (US74)	Radio astronomy.		RADIO ASTRON- OMY.
74.6-75.4	AERONAUTICAL RADIONAVIGA- TION.	Aeronautical radio- navigation.	75	Marker beacon.
75. 4-76 (NG 56)	FIXED. (NG1) (NG3) (NG49)	Operational fixed.	75. 42- 75. 98 (N G33)	Operational fixed.
76-88 (NG21)	BROADCASTING.	Television broadcast- ing.	77. 25 81. 75 83. 25 87. 75	Video Sound Channel 5. Video Channel 6.
88-108 (N G21)	BROADCASTING.	FM broadcasting. (NG2).	88. 1- 107. 9 (NG36)	FM Channel 201- FM Channel 300.
108-117. 975	AERONAUTICAL RADIONAVIGA- TION.	Radionavigation land.	108. 1 108. 2 108. 3 108. 4	Localizer. Omni-directional range. Localizer. Omni-directional
			108. 5 108. 6	range. Localizer. Omni-directional range.
			108. 7 108. 8	Localizer. Omni-directional range. Localizer.
			109. 0 109. 1 109. 2	Omni-directional range. Localizer. Omni directional
			109. 3 109. 4	range. Localizer. Omni-directional range.
			109. 5 109. 6	Localizer. Omni-directional range.
			109.7 109.8	Localizer. Omni-directional range.
			109. 9 110. 0	Localizer. Omni-directional range. Localizer.

Wo	Worldwide		Region 2		s
Band (MHz)	Service	Band(MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
			:		
!					
117. 975-132	AERONAUTICAL MOBILE. (R) (273) (273A)			117. 975-121. 975 (273) (US26) (US28) (US28) (US85)	G, NG.
				121. 975-123. 075 (US29) (US3:) (US31) (US80) (US86) (US102)	NG.
				123,075-123,575 (US32) (US33)	G, NG.

	Federal Com	munications Commiss	ion	
Band (MHz)	Service	Class of station	Fre- quency (MHz)	Nature SERVICES of stations
7	8	9	10	11
			110. 2	Omni-directional
			110. 3 110. 4	range. Localizer. Omni-directional range.
			110. 5 110. 6	Localizer. Omni-directional range.
			110.7 110.8	Localizer. Omni-directional range.
			110.9 111.0	Localizer. Omni-directional range.
			111.1 111.2	Localizer. Omni-directional range.
			111.3 111.4	Localizer. Omni-directional range.
			111.5 111.6	Localizer. Omni-directional range.
			111.7 111.8	Localizer. Omni-directional range.
			111.9	Localizer.
			112.0- 117.9 (NG35)	Omni-directional range.
117. 975- 121. 975	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.	118- 121. 4 (NG34)	Airdrome control.
			121.5	AERONAUTICAL MOBILE.
			121. 6- 121. 90 (N G 34)	Aeronautical utility land; aeronautical utility mobile.
			121. 95	Flight test.
121. 975- 123. 075	AERONAUTICAL MOBILE.	Aeronautical. Aircraft.	122. 0- 123. 05 (NG34)	Private aircraft.
123.075- 123.575	AERONAUTICAL MOBILE.	Aeronautical. Aircraft.	123.1	Aeronautical search and rescue mobile.
			123, 15 123, 2 123, 25 123, 3	Flight test. Do. Do. Flight test; aviation
:			123. 35 123. 4 123. 45	instructional. Flight test.

Worldwide		Region 2		United States	
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
				123.575-128.825 (US26) (US85)	G, NG.
				128. 825-132 (US2) (US85)	NG.
132-136		132-136	FIXED. MOBILE. (273A) (276).	132-136 (US2) (US26) (US85)	G, NG. (US3)
136–137		136-137 (281A) (281B)	SPACE RE- SEARCH (Telem- etering and tracking).	136-137	G, NG. (U8100)
137-138 (281 E)	METEOROLOGI- CAL-SATELLITE. SPACE RESEARCH (Telemetering and tracking). (281F) SPACE (Telemeter- ing and tracking).			137-138	G, NG. (US100)
138-143. 6		138-143. 6	FIXED. MOBILE. Radiolocation.	138-144 (US10)	G.
143.6- 143.65		143.6-143.65	FIXED. MOBILE. SPACE RE- SEARCH (Telemetering and tracking). Radiolocation.		
143.65-144		143. 65–144	FIXED. MOBILE. Radiolocation.		
144-146	AMATEUR. (284A).			144-148	AMA- TEUR. (284A) (US1)
146-148		146-148	AMATEUR.	il	
148-149.9		148-149. 9 (285A)	FIXED. MOBILE.	148-149.9 (US10) (US86)	G.
(285A) 149. 9- 150. 05 (285B)	RADIONAVIGA- TION-SATEL- LITE.	(280A.)		149. 9-150. 05	G, NG. (US100)
150. 05-174		150. 05-174	FIXED. MOBILE.	150. 05-150. 8	G.
				150.8-157.0375	NG.

	Federal	Communications Com	mission	
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
			123. 5 123. 55	Flight test; aviation instructional. Flight test.
123.575- 128.825	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.	123.6- 128.8 (NG34)	AERONAUTICAL MOBILE.
128. 825-132	AERONAUTICAL MOBILE. (R)	Aeronautical. Aircraft.	128. 85- 132 (NG34)	AERONAUTICAL MOBILE.
132-136	AERONAUTICAL MOBILE (R).	Aeronautical. Aircraft.	132.05- 135.95 (NG34)	AERONAUTICAL MOBILE.
136-137	SPACE RE- SEARCH.	Space.		Telemetering; track- ing.
137-138	METEOROLOGI- CAL-SATEL- LITE. SPACE (Telemeter- ing and tracking).	Space.		METEOROLOGI- CAL-SATELLITE. SPACE.
			143.90	Civil air patrol land;
144-148	AMATEUR.	Amateur.		mobile. AMATEUR.
			148. 15 148. 25	Civil sir patrol land; civil air patrol mobile. Earth (telecommand).
149. 9-150. 05	RADIONAVIGA- TION-SATEL- LITE.	Space.		RADIONAVIGA- TION-SATEL- LITE.
150. 8–150. 98	LAND MOBILE.	Base. Land mobile.		LAND TRANSPORTA- TION. (NG51).

Title 47—Chapter I

Worldwide		R	Region 2		States
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
(285A)		(285A)		(US86)	
					(U877)
(287)					(US106) (US107)

	Federal Communications Commission					
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature SERVICES of stations		
7	8	9	10	11		
150. 98- 151. 49	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY (NG51).		
151. 49-152	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.		
152-152. 255	LAND MOBILE.	Base. Land mobile.		DOMESTIC -PUB- LIC.		
152, 255- 152, 465	LAND MOBILE.	Base. Land mobile.		LAND TRANS- PORTATION. (NG38)		
152. 465- 152. 495	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.		
152, 495- 152, 855	LAND MOBILE.	Base. Land mobile.		DOMESTIC PUB- LIC. (NG4)		
152, 855- 153, 7325	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL. (NG4)		
153. 7325- 154. 46	LAND MOBILE.	Base. Land mobile.	154, 2	PUBLIC SAFETY. (NG26) (NG57) Earth (telecom- mand).		
154. 46- 154. 6275	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL. (NG57).		
154. 6275- 156. 25	LAND MOBILE.	Base. Land mobile.		PUBLIC SAFETY.		
156. 250— 157. 0375	MARITIME MOBILE. (NG5)	Coast. Ship.	156. 275 156. 300 156. 325 156. 350 156. 455 156. 450 156. 455 156. 500 156. 525 156. 550 156. 575 156. 625 156. 700 156. 725 156. 870 156. 870 156. 875 156. 950 156. 950 156. 950 156. 975 157. 090	MARITIME MOBILE. Do. MARITIME MOBILE. (NG24) MARITIME MOBILE. Do. MARITIME MOBILE. (NG24) MOBILE. (NG24) MOBILE. (NG24) MOBILE. (NG24)		

Worldwide			Region 2	United 8	United States	
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation	
1	2	3	4	5	6	
				157. 0375- 157. 1875	G.	
				157. 1875- 162. 0125	NG. (US77)	
				162, 0125- 173, 2	G	
				173.2	G. (US8) (US11) (US13)	
				173. 2- 173. 4	NG.	
				173.4- 174.0	G.	

	Federa	l Communications Com	mission		
Band (MHz) Service		Class of station	Frequency (MHz)	Nature OF SERV- ICES of stations	
7	8	9	10	11	
157. 1875- 157. 450 MARITIME MOBILE. (N G5)		Ship.	157, 200 157, 225 157, 250 157, 275 157, 300 157, 325 157, 350 157, 375 157, 400 157, 425	MARITIME MOBILE. Do. Do. Do. Do. Do. Do. Do. Do. Do. Do	
157. 45- 157. 725	LAND MOBILE.	Base. Land mobile.		LAND TRANS- PORTATION. (NG5) (NG38)	
157.725- 157.755	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.	
157. 755- 158. 115	LAND MOBILE.	Base. Land mobile.		DOMESTIC PUB- LIC.	
158. 115- 158. 475	LAND MOBILE.	Base. Land mobile.		INDUSTRIAL.	
158. 475– 158. 715	LAND MOBILE.	Base. Land mobile.		DOMESTIC PUB	
158. 715- 159. 48	LADN MOBILE.	Base. Land mobile.		PUBLIC SAFETY.	
159. 48- 161. 575	LAND MOBILE.	Base. Land mobile.		LAND TRANSPOR TATION. (NG6 (NG26) (NG28)	
161. 575- 161. 625 (US77)	MARITIME MO- BILE.	Coast.	161. 6	MARITIME MO BILE. (NG6 (NG17) (NG26)	
161. 625- 161. 775	LAND MOBILE.	Base. Land mobile.		Remote pickup broad cast base; remot pickup broadcas mobile. (NG6	
161. 775- 162. 0125 MARITIME MOBILE. (N G5).		Coast.	161, 800 161, 825 161, 850 161, 875 161, 900 161, 925 161, 950 161, 975 162, 000	(NG26) Coast (NG26). Do. Do. Coast. Do. Do. Do. Do. Do. Do. Do. Do. Do.	
			166. 25 170. 15	PUBLIC SAFETY. Remote pickup. Do.	
			170, 425 170, 475 170, 575 171, 425 171, 575 171, 575 172, 225 172, 275 172, 375	PUBLIC SAFETY. Do. Do. Do. Do. Do. Do. Do. D	
173. 2-173. 4	FIXED. LAND MOBILE.	Base. Fixed. Land mobile.		INDUSTRIAL.	

	Worldwide		Region 2	United 8	States
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
174-216		174-216 (294)	FIXED. MOBILE. BROADCASTING.	174-216	NG.
216-220		216-220	FIXED.	216-220	G. (US5)
			RADIOLOCATION.		
220-225		220-225	AMATEUR. RADIOLOCATION.	220-225	G, NG. (U834)
225-235		225-235	FIXED. MOBILE.	225–328, 6 (310)	G.
235-267 (309)	FIXED. MOBILE.			(ŬŜiħ) (US98)	
267-272	FIXED. MOBILE. Space (telemetering). (309A) (309B)				
272-273	FIXED. MOBILE. SPACE (Telemeter- ing). (309A)			:	
273–328, 6 (310)	FIXED. MOBILE.				
328. 6-335. 4 (310)	AERONAUTICAL RADIONAVIGA- TION. (311)			328. 6-335. 4 (310)	G, NG.
335. 4–399. 9	FIXED. MOBILE.			335, 4-399, 9	G.

	Federal (Communications Comm	ission	
Band (MHz)		Class of station	Frequency (MHz)	Nature OF SERVICES of stations
7	. 8	9	10	11
174-216	BROADCASTING.	Television broadcast- ing.	175. 25 179. 75 181. 25 185. 75 187. 25 191. 75 193. 25 197. 75 199. 25 203. 75 205. 25 209. 75 211. 25 215. 75	Video Sound Channel 7. Video Channel 8. Video Channel 9. Video Channel 10. Video Channel 11. Video Channel 11. Video Channel 12. Video Channel 12. Sound Channel 13. Sound Channel 13.
			217. 425 217. 475 217. 525 217. 550 217. 675 217. 675 219. 325 219. 375 219. 450 219. 475 219. 525 219. 575	Telemetering land; te emetering mobile. Do. Do. Do. Do. Do. Do. Do. Do. Do. Do
220-225	Amateur. (NG13)	Amateur.		AMATEUR.
			243	Survival craft and equipment.
328. 6-335. 4	AERONAUTICAL RADIONAVIGA- TION. (311)	Radionavigation land.		Glide path.

Worldwide		Region 2		United S	tates
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
399.9-400.05	RADIONAVIGA- TION- SATELLITE, (311A)			399. 9–40 0. 05	G, NG. (US100)
400. 05-401	METEOROLOGI- CAL AIDS. METEOROLOGI- CAL- SATELLITE. (Maintenance telemetering). SPACE RE- SEARCH (Telem- etering and tracking).			400, 05-401	G, NG.
401-402	METEOROLOGI- CAL AIDS. SPACE (Telemeter- ing). (315A) Fixed. Mobile except aero- nautical mobile.			401-402	G, NG.
402-406 (317)	METEOROLOGI- CAL AIDS. Fixed. Mobile except aero-			402-404	G, NG.
	nautical mobile.			404-406	G, NG.
406-420 (317)	FIXED. MOBILE except aeronautical mobile.			406-420	G. (US13)
420-450		420-450 (318) (319A)	RADIOLOCATION. Amateur.	420-450 (US6) (US35) (US87)	G, NG.
450-460 (318) (319A)	FIXED. MOBILE.			450-470 (US6) (US87)	NG.

	Federal	Communications Com	mission		
Band (MHz) Service		Class of station	Frequency (MHz)	Nature OF SERV-ICES of stations	
7	8	9	10	11	
399, 9–400, 05	RADIONAVIGA- TION- SATELLITE.	Space.		RADIONAVIGA- TION- SATELLITE.	
400. 05-401	METEOROLOGI- CAL AIDS. (US70) SPACE RE- SEARCH (Telemetering and tracking).	Radiosonde. Space.		Radiosonde. Space.	
401-402	METEOROLOGI- CAL AIDS. (US70) SPACE (Telemeter- ing). (315A)	Radiosonde. Space.		Radiosonde. Space.	
402-404	METEOROLOGI- CAL AIDS. (US70)	Radiosonde.		Radiosonde.	
404-406	METEOROLOGI- CAL AIDS. (US70) Radio astronomy. (US74)	Radio astronomy. Radiosonde.		RADIO ASTRON- OMY. Radiosonde.	
420-450	Amateur. (US7)	Amateur.		AMATEUR.	
450-451	LAND MOBILE.	Base. Land mobile.		Remote pickup broad cast base; remote pickup broadcast mobile.	
451-454	LAND MOBILE.	Base. Land Mobile.		PUBLIC SAFETY. INDUSTRIAL. LAND TRANS- PORTATION.	
454-455	LAND MOBILE.	Base. Land mobile.		DOMESTIC PUBLIC. (NG1:	
455-456	LAND MOBILE.	Base. Land mobile.		Remote pickup broad cast base; remote pickup broadcast mobile.	
456-459	LAND MOBILE.	Mobile.		PUBLIC SAFETY INDUSTRIAL. LAND TRANS-PORTATION.	
459-460	LAND MOBILE.	Base. Land mobile.		DOMESTIC PUBLIC. (NG12	

	Worldwide		Region 2	United S	itates
Band (MHz)	Service	Band (MIIz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
460-470	FIXED. MOBILE. Meteorological- Satellite. (318A)		·		(US100)
470-890		470-890 (332)	BROADCASTING.	470-890 (NG30) (NG43) (US88)	NG. (US100)
890-942		890-942 (339A) (340)	FIXED. RADIOLOCATION.	890-942 (340)	G. (US36)
942-960		942-960 (339A)	FIXED.	942-960	NG.
960-1215	AERONAUTICAL RADIONAVIGA- TION. (341)			960-1215	G, NG.
1215-1300	RADIOLOCATION.			1215-1300	G, NG. (US37)

	Federal	Communications Commi	ssion		
Band (MIIz)	Service	Class of station	Fre- quency (MHz)	Nature SERVICES of stations	
7	8	9	10	11	
460–462, 5375	LAND MOBILE.	Base. Land Mobile.		PUBLIC SAFETY. INDUSTRIAL. LAND TRANS- PORTATION.	
462, 5375– 462, 7375	LAND MOBILE.	Base. Land Mobile.		CITIZENS RADIO.	
462. 7375- 465. 0125	LAND MOBILE.	Base. Land Mobile.		PUBLIC SAFETY. INDUSTRIAL. LAND TRANS- PORTATION.	
465. 0125- 467. 5375	LAND MOBILE.	Mobile.		PUBLIC SAFETY. INDUSTRIAL. LAND TRANS-PORTATION.	
467. 5375- 467. 7375	LAND MOBILE.	Mobile.		CITIZENS RADIO	
467. 7375-470	LAND MOBILE.	Mobile.		PUBLIC SAFETY. INDUSTRIAL. LAND TRANS- PORTATION.	
470-890	BROADCASTING.	Television broadcasting.			
			915	Industrial, scientific, and medical equip- ment.	
942-952 (NG8) (NG9) (NG58)	FIXED. (NG40)	Aural broadcast STL. International aero- nautical fixed (Alaska, Hawaii, and U.S. posses- sions only). International fixed public (Alaska, Hawaii, and U.S. possessions only).			
952-960 (N G10)	FIXED.	International fixed public (Puerto Rico and Virgin Islands only). International con- trol. Operational fixed.			
960-1215	AERONAUTICAL RADIONAVIGA- TION. (341)				
1215-1300	Amateur.	Amateur.		AMATEUR.	

Worldwide		Region 2		United States	
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6
1300-1350	AERONAUTICAL RADIONAVIGA- TION. (346) Radiolocation.			1300-1350	G, NG. (US38)
1350-1400		1350-1400	RADIOLOCA- TION.	1350-1400	G.
1400-1427	RADIO ASTRONOMY.		·	1400-1427	G, NG.
1427-1429	FIXED. MOBILE except aeronautical mobile. SPACE (Telecommand).			1427-1429	G, NG. (US60)
1429-1435		1429-1435	FIXED. MOBILE.	1429-1435	G.
1435-1525		1435-1525	MOBILE. Fixed.	1435-1525	G, NG. (US78)
1525-1535		1525-1535	SPACE. (Telemetering.) (350A) Fixed. Mobile. (350D)	1525-1535	G, NG. (350A) (US89) (US100)
1535-1540	SPACE. (Telemetering). (350A)			1535-1540 (350A)	G, NG.
1540-1660	AERONAUTICAL RADIONAVIGA- TION. (352A) (352B)			1540-1660	G, NG. (352A) (352B) (US39)
1660-1664. 4 (354A)	METEOROLOGI- CAL AIDS. METEOROLOGI- CAL-SATEL- LITE. (324A)			1660-1670	G, NG. (324A) (US74) (US99) (US100) (US101)
1664. 4- 1668. 4 (353A) (354A)	METEOROLOGI- CAL AIDS. METEOROLOGI- CAL-SATEL- LITE. (324A) Radio astronomy.				
1668. 4-1670 (354 A)	METEOROLOGI- CAL AIDS. METEOROLOGI- CAL-SATEL- LITE. (324A)				
1670–1690	METEOROLOGI- CAL AIDS, FIXED, MOBILE except aeronautical mobile.			1670–1690	G, NG. (US99)
1690-1700 (354A)		1690-1700	METEOROLOGI- CAL AIDS. METEOROLOGI- CAL-SATEL- LITE. (324A)	1690-1700	G, NG. (324A) (US99) (US100)
1700- 1710		1700-1710 (355A)	SPACE RE- SEARCII (Telem- etering and track- ing).	1700-1710	G, NG. (US100)
1710-1770		1710-1770	FIXED. MOBILE.	1710-1850	G.

	Federal	Communications Com	mission	
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
1300-1350	AERONAUTICAL RADIONAVIGA- TION. (346)			
1400-1427	RADIO ASTRON- OMY. (US74)			
1427-1429	SPACE.	Earth.		Earth (telecommand).
1435-1525	MOBILE.	Aeronautical tele- metering.		AVIATION.
1525-1535	MOBILE. SPACE.	Aeronautical tele- metering. Space.		AVIATION. Space (telemetering).
1535-1540	SPACE. (Telemetering).	Space.		Space (telemetering).
1540-1660	AERONAUTICAL RADIONAVIGA- TION.			
1660–1670	METEOROLOGI- CAL AIDS, METEOROLOGI- CAL-SATEL- LITE, RADIO ASTRON- OMY.	Radio astronomy. Radiosonde. Space.		
1670–1690	METEOROLOGI- ICAL AIDS.	Radiosonde.		
1690-1700	METEOROLOGI- CAL AIDS. METEOROLOGI- CAL-SATEL- LITE.	Radiosonde, Space.		
1700-1710	SPACE RE- SEARCH (Telem- etering and track- ing).	Space.		Space (telemetering and tracking).
	l	1	1	1

	Worldwide		Region 2		States
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocations
1	2	3	4	5	6
1770-1790		1770-1790	FIXED, MOBILE, Meteorological- Satellite, (356AA)		(US100)
1790-2290		1790-2290 (356A)	FIXED. MOBILE.	1850-2200	NG. (US96)
2290-2300		2290-2300 (356B)	SPACE RE- SEARCH (Telem- etering and track- ing in deep space).	2200-2290 2290-2300	G. G, NG. (US100)
2300-2450 (357)		2300-2450	RADIOLOCATION. Amateur. Fixed. Mobile.	2300-2450	G, NG. (US40)
2450-2550 (357)		2450-2550	FIXED. MOBILE. RADIOLOCATION.	2450-2500	NG. (US41)
2550-2690 (365)	FIXED. MOBILE.			2500-2690	NG.
2690-2700 (364A) (365)	RADIO ASTRONO- MY.			2690-2700	G, NG. (US74) (US100)

	Federal (Communications Commi	ssion	
Band (MHz)	Service	Class of station	Fre- quency (MHz)	Nature SERVICES of stations
7	8	9	10	11
1850-1990 (NG8)	FIXED.	International control. Operational fixed.		
1990-2110 (NG11)	FIXED. MOBILE.	Television pickup. Television STL.		
2110-2130 (NG10) (NG23) (US90)	FIXED.	Domestic fixed public.		
2130-2150 (NG10) (NG23)	FIXED.	Operational fixed. International control.		
2150-2160 (N G23) (N G45)	FIXED.			
2160-2180 (NG10) (NG23)	FIXED.	Domestic fixed public.		
2180-2200 (NG10) (NG23)	FIXED.	Operational fixed. International control.		
2290-2300	SPACE RE- SEARCH (Telem- etering and track- ing in deep space).	Space.		Space (telemetering and tracking in deep space).
2300-2450 (357)	Amateur.	Amateur.		
2450-2500 (357)	FIXED. MOBILE. Radiolocation.		2450	Industrial, scientific and medical equip- ment.
2500-2690 (N G8) (N G47)	FIXED.	International control. Operational fixed. Instructional television fixed.		
2690-2700	RADIO ASTRONO-	Radio astronomy.		1

	Worldwide	Region 2		United States		
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation	
1	2	3	4	5	6	
2700-2900 (366)	AERONAUTICAL RADIONAVIGA- TION. (346). Radiolocation.			2700-2900	(346) (366) (US42) (US43)	
2900-3100	RADIONAVIGATION. (367). Radiologation.			2900-3100	G, NG.	
3100-3300 (369)	RADIOLOCATION.			3100-3300	G. (369) (US45) (US46)	
3300-3406		3300-3400	RADIO- LOCATION. Amateur.	3300-3500	G, NG. (US61)	
3400-3500		3400-3500	RADIO- LOCATION. COMMUNICA- TION- SATELLITE (satellite to earth). (374A) Amateur.			
3500–3700		3500–3700	FIXED. MOBILE. RADIO- LOCATION. COMMUNICA- TION- SATELLITE (satellite to earth). (374A)	3500-3700	G.	
3700-4200		3700-4200	FIXED. MOBILE. COMMUNICA- TION- SATELLITE (satellite to earth). (374A)	3700-4200		
4200-4400 (352A)	AERONAUTICAL RADIO- NAVIGATION.			4200-4400	G, NG. (352A) (US47)	
4400-4700	FIXED. MOBILE. COMMUNICA- TION- SATELLITE (earth to satellite). (392A)			4400-4990	G.	
4700-4990 (365)	FIXED. MOBILE.					
4990-5000		4990-5000 (383A)	RADIO ASTRONOMY.	4990-5000	G, NG. (US74) (US100)	
5000-5250 (352A) (352B)	AERONAUTICAL RADIONAVI- GATION.			5000-5250	G, NG. (352A) (352B)	
5250-5255	RADIOLOCA- TION. Space research.			5250-5350	G.	
5255-5350	RADIOLOCA- TION.	11				

	Federal	Communications Com	mission	
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature OF SERV ICES of stations
7	8	9	10	11
2900-3100	MARITIME RADIONAVIGA- TION Radiolocation. (US44)			
3300–3500	Amateur.	Amateur.		AMATEUR.
3700-4200	COMMUNICA- TION- SATELLITE. (374A) (US91) FIXED. (NG)	Common carrier fixed. Space.		COMMUNICA- TION- SATELLITE. DOMESTIC PUBLIC. (NG41 (NG55)
4200-4400	AERONAUTICAL RADIO- NAVIGATION.	Altimeter.		
4990-5000	RADIO ASTRONOMY.	Radio astronomy.	-	
5000-5250	AERONAUTICAL RADIONAVI- GATION.			

	Worldwide		Region 2	United S	itates
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation
1	2	3	4	5	6 .
5350-5460	AERONAUTICAL RADIONAVIGA- TION. (385) Radiolocation.			5350-5460	G, NG.
5460-5470	RADIONAVIGA- TION. (385) Radiolocation.			5460-5470	G, NG.
5470-5650 (387)	MARITIME RADIONAVI- GATION, Radiolocation.			5470-5600	G, NG.
				5600-5650	G, NG.
5650-5670	RADIOLOCATION.			5650-5925	G, NG
	Amateur.			(391)	(US52)
5670-5725 (389A)	RADIOLOCATION. Amateur. Space research (deep space).				(US100)
5725-5925		5725-5925	RADIOLOCA- TION.	_	
		(391)	Amateur.		
5925-6425	FIXED. MOBILE. COMMUNICA- TION-SATEL- LITE (earth to satellite). (392A)			5925-6425	(392A)
6425-7250 (392B) (392F) (393)	FIXED. MOBILE.			6425-7125	NG.
				7125-7250	G.

	Federal	Communications Comm	rission	
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature SERVICES of stations
7	8	9	10	11
5350-5460	AERONAUTICAL RADIONAVIGA- TION. (385) Radiolocation. (US48)			
5460-5470	RADIONAVIGA- TION. (385) (US65) Radiolocation. (US49)			
5470-5600	MARITIME RADIONAVI- GATION. (US65) Radiolocation. (US50)			
5600-5650	MARITIME RADIONAVI- GATION. (US65) METEOROLOGI- CAL AIDS. (387) Radiolocation. (US51)			
5650-5925	Amateur.	Amateur.		
			5800	Industrial scientific & medical equipment
5925-6425	COMMUNICA- TION-SATEL- LITE. (US91) FIXED. (NG)	Common carrier fixed. Fixed earth.		COMMUNICA- TION-SATEL- LITE, DOMESTIC PUB- LIC. (NG41) (NG55)
6425-6525 (N G46)	MOBILE.	Common carrier land. Common carrier mobile.		
6525-6575 (NG46)	MOBILE.	Operational land. Operational mobile.		
6575-6875 (NG8)	FIXED.	International control. Operational fixed.		
6875-7125 (NG11)	FIXED. MOBILE.	Television pickup. Television STL.		

Worldwide			Region 2	United 8	United States		
Band (MHz)	Service	Band (MHz)	Service	Band (MHz)	Allocation		
1	2	3	4	5	6		
7250-7300 (374A) (392C) (392D) (392G)	COMMUNICA- TION-SATEL- LITE (satellite to earth).			7250-7300	(374A) (392D) (US100)		
7300-7750 (392F)	FIXED. MOBILE. COMMUNICA- TION-SATEL- LITE (satellite to earth). (374A) (392D)			7300-7750	(374A) (392D)		
7750-7900	FIXED. MOBILE.			7750-7900	G.		
7900-7975	FIXED. MOBILE. COMMUNICA- TION-SATEL- LITE (earth to satellite). (392A)			7900-7975	(392A)		
7975-8025 (392A) (392C) (392H)	COMMUNICA- TION-SATEL- LITE (earth to satellite).			7975-8025	(392A) (US100)		
8025-8400	FIXED. MOBILE. COMMUNICA- TION-SATEL- LITE (earth to satellite). (392A)			8025-8400	(392A)		
8400-8500		8400-8500	SPACE RE- SEARCH. (394C)	8400-8500	G, NG. (US62) (US100)		
8500-8750	RADIOLOCATION.			8500-9000	G. (US53)		
8750-8850	RADIOLOCATION. AERONAUTICAL RADIONAVIGA- TION. (396)						
8850-9000	RADIOLOCATION.						
9000-9200	AERONATUICAL RADIONAVIGA- TION. (346) Radiolocation.			9000-9200	G, NG.		
9200-9300	RADIOLOCATION.			9200-9300	G. (US45)		
9300-9500 (399)	RADIONAVIGA- TION. Radiolocation.			9300-9500	G, NG.		
9500-9800	RADIOLOCATION.						
9800-10000 (401A)	RADIOLOCATION. Fixed.			9500-10000	G. (401A)		
10000-10500 (401A)	RADIOLOCATION. Amateur.			10000-10500	G, NG. (401A) (US58)		

	Federal (Communications Commi	ission	
Band (MHz)	Service	Class of station	Frequency (MHz)	Nature OF SERV-ICES of stations
7	8	9	10	11
7250-7300	COMMUNICA- TION-SATEL- LITE. (US91)	Space.		COMMUNICA- TION-SATEL- LITE,
7300-7750	COMMUNICA- TION-SATEL- LITE. (US91) METEOROLOGI- CAL-SATEL- LITE. (G/NG) (US92) [FIXED (G). MOBILE (G)].	Space.		COMMUNICA- TION-SATEL- LITE. METEOROLOGI- CAL-SATEL- LITE.
7900-7975	COMMUNICA- TION-SATEL- LITE. (US91) [FIXED (G). MOBILE (G)].	Earth.		COMMUNICA- TION-SATEL- LITE.
7975-8025	COMMUNICA- TION-SATEL- LITE. (US91)	Earth.		COMMUNICA- TION-SATEL- LITE.
8025-8400	COMMUNICA- TION-SATEL- LITE. (US91) [FIXED (G). MOBILE (G)].	Earth.		COMMUNICA- TION-SATEL- LITE.
8400-8500	SPACE RE- SEARCH. Fixed. Mobile.	Fixed. Mobile. Space.		
			8800	Airborne doppler radar.
9000-9200	AERONAUTICAL RADIONAVIGA- TION. (346) (US54) Radiolocation. (US55)			
9300-9500	RADIONAVIGA- TION. (US66) (US71) Meteorological aids. (US67) Radiolocation. (US56)			
10000-10500	Amateur. Radiolocation. (NG42)	Amateur. Radiolocation land. Radiolocation mobile.		AMATEUR. RADIOLOCATION

Worldwide			Region 2	United States		
Band (GHz)	Service	Band (GHz)	Service	Band (GHz)	Allocation	
	2	3	4	5	6	
10. 5–10. 55		10. 5-10. 55	RADIOLOCA- TION. (404).	10. 5-10. 55	G, NG. (US59)	
10. 55–10. 68	FIXED. MOBILE. Radiolocation.			10. 55-10. 68	NG.	
10. 68-10. 7 (405B)	RADIO AS- TRONOMY.			10. 68-10. 7	G, NG. (US74) (US100)	
10. 7-11. 7	FIXED. MOBILE.			10.7-11.7	NG.	
11. 7-12. 7	FIXED. MOBILE except aeronautical mobile. BROADCAST- ING.			11. 7-13. 25	NG.	
	MOBILE.					
13. 25-13. 4	AERONAUTICAL RADIONAVIGA- TION. (406)			13. 25–13. 4	G, NG.	
13.4-14.0	RADIOLOCATION.			13.4-14.0	G.	
14. 0-14. 3	RADIONAVIGA- TION.			14, 0-14, 3	G, NG.	
14. 3-14. 4	RADIONAVIGA- TION-SATEL- LITE.			14.3-14.4	G, NG.	
14. 4-15. 25	FIXED. MOBILE.			14, 4-15, 25	G.	
15. 25-15. 35 (409A)	SPACE RESEARCH.			15, 25-15, 35	G, NG. (US100)	
15. 35-15. 4 (409C)	RADIO ASTRONOMY.			15. 35–15. 4	G, NG. (US74) (US100)	

	Federal	Communications Comm	ission	
Band (GHz) Service		Service Class of station		Nature OF SERVICES of stations
7	8	9	10	11
10. 5–10. 55	RADIOLOCA- TION.	Radiolocation land. Radiolocation mobile.		RADIOLOCATION.
10. 55-10. 68 (N G46)	MOBILE.	Operational land. Operational mobile.		
10. 68-10. 7	RADIO AS- TRONOMY.			
10.7-11.7	FIXED.	Common carrier fixed.		DOMESTIC PUBLIC. (NG41)
11.7-12.2	MOBILE.	Common carrier land. Common carrier mobile (except aeronautical mobile).		
12.2-12.7 (NG8) (NG52)	FIXED.	International control. Operational fixed.		
12.7-12.95	FIXED. MOBILE.	Community Antenna Relay. Television Intercity Relay. Television Pickup. (NG53) Television STL.		
12. 95-13. 2 (N G11)	FIXED. MOBILE.	Television Pickup. Television STL.		
13. 2-13. 25	FIXED. MOBILE.			
13. 25-13. 4	AERONAUTICAL RADIONAVIGA- TION. (406)			Airborne doppler radar.
14.0-14.3	RADIONAVIGA- TION.			
14. 3-14. 4	RADIONAVIGA- TION-SATEL- LITE.	Earth. Space.		RADIONAVIGA- TION-SATEL- LITE.
15. 25-15. 35	SPACE RESEARCH.	Space.		
15. 35-15. 4	RADIO ASTRONOMY,			
	l—	ļ 	l	

§ 2.106

Title 47—Chapter I

Worldwide			Region 2	United States		
Band (GHz)	Service	Band (GHz)	Service	Band (GHz)	Allocation	
1 .	2	3	4	5	6	
15. 4-15. 7	AERONAUTICAL RADIONAVIGA- TION. (352A) (352B)			15. 4-15. 7	G, NG. (352A) (352B)	
15.7-17.7	RADIOLOCATION.			15. 7-17. 7	G.	
17. 7-19. 3	FIXED. MOBILE.			17. 7-19. 3	NG.	
19. 3-19. 4 (409 D)	RADIO ASTRON- OMY.			19. 3-19. 4	G, NG. (US74) (US100)	
19. 4-21. 0	FIXED. MOBILE.			19. 4-19. 7	NG.	
				19. 7-21. 0	G.	
21. 0-22. 0	AMATEUR.			21. 0-22. 0	AMA- TEUR.	
22. 0-23. 0 (410)	FIXED. MOBILE.			22. 0-23. 0	G. (410)	
23. 0-24. 25	RADIOLOCATION.			23. 0-24. 25	G.	
24. 25-25. 25	RADIONAVIGA- TION. (411)			24. 25-25. 25	G. NG. (US72)	
25. 25-31. 0	FIXED. MOBILE.			25. 25-27. 525 27. 525-31. 3	G. NG.	
31. 0-31. 3 (412H)	FIXED. MOBILE. Space research.			2.1.020 01.0	(US100)	
31. 3-31. 5 (412A)	RADIO ASTRON- OMY.			31. 3–31. 5	G, NG. (US74) (US100)	
31. 5-31. 8		31. 5-31. 8 (405C)	SPACE RE- SEARCH.	31. 5-31. 8	G, NG. (US100)	
31.8-32.3 (412B)	RADIONAVIGA- TION. Space research.			31. 8-33. 4	G, NG. (US100)	
32, 3–33 . 0	RADIONAVIGA- TION.					
33. 0–33. 4		33. 0-33. 4	RADIONAVIGA- TION. (412F)			
33. 4-34. 2 (412G)	RADIOLOCATION.			33. 4-38. 6	G. (US100)	
34. 2-35. 2 (412C) (412D)	RADIOLOCATION. Space research.					
35. 2-36. 0	RADIOLOCATION.			-11 1		
36. 0-40. 0 (412E)	FIXED. MOBILE.			38.6-40.0	NG.	
				30.0-10.0	.10.	
Above 40.0	Not allocated.			40.0-88.0	G, NG.	
				88. 0-90. 0	G, NG. (US74)	
				Above 90.0	G, NG.	

	Federal	Communications Com	mission	
Band (GHz)	Service	Class of station	Frequency (GHz)	Nature OF SERV- ICES of stations
7	8	9	10	11
15. 4–15. 7	AERONAUTICAL RADIONAVIGA- TION.			
17.7-19.3	FIXED. MOBILE.		-	
19. 3–19. 4	RADIO ASTRON- OMY.			
19. 4-19. 7	FIXED. MOBILE.			
21. 0-22. 0	AMATEUR.	Amateur.		AMATEUR.
			22. 125	Industrial, scientific and medical equip- ment.
24, 25–25, 25	RADIONAVIGA- TION. (411)			
27. 525-31. 3	FIXED. MOBILE.	·		
31. 3-31. 5	RADIO ASTRON- OMY.		-	
31. 5-31. 8	SPACE RE- SEARCH.			
31. 8–33. 4	RADIONAVIGATION. (US69)			
38. 6–40. 0	FIXED. MOBILE.		_	
40. 0-88. 0		Amateur. Experimental.		
88. 0-90. 0	RADIO ASTRON- OMY.			
A bove 90. 0		Amateur. Experimental.		

FOOTNOTES

GENEVA FOOTNOTES

- (157) Administrations authorizing the use of frequencies below 10 kHz for special national purposes shall ensure that no harmful interference is caused thereby to the services to which the bands above 10 kHz are allocated (see also Article 14, No. 699).
- (158) Limited to coast telegraph stations (A1 and F1 only). Exceptionally, the use of class A7J emission is permissible subject to the necessary bandwidth not exceeding that normally used for class A1 and F1 emissions in the hands concerned.
- (159) The stations of services to which the bands between 14 and 70 kHz are allocated may transmit standard frequency and time signals, Such stations shall be afforded protection from harmful interference. In Albania, Bulgaria, Hungary, Poland, Roumania, Czechoslovakia, and the U.S.R., the frequencies 25 kHz and 50 kHz will be used for this purpose under the same conditions.
 - (160) The standard frequency is 20 kHz.
- (161) In the U.S.S.R., frequencies in the band 60-80 kHz may be used for industrial, scientific and medical purposes subject to the condition that interference is not caused to stations of services to which this band is allocated.
 - (162) Limited to continuous wave systems.
- (164) The establishment and operation of maritime radionavigation stations shall be subject to agreements between administrations whose services, operating in accordance with the Table, may be affected. However, the fixed, maritime mobile and radiolocation services shall not cause harmful interference to maritime radionavigation stations established under such agreements.
- (166) The development and operation of long distance radionavigation systems are authorized in this band, which will become exclusively allocated, wholly or in part, to the radionavigation service for the use of any one such system as soon as it is internationally adopted. Other considerations being equal, preference should be given to the system requiring the mininum band-width for world-wide service and causing the least harmful interference to other services. If a pulse radionavigation system is employed, the pulse emissions shall nevertheless be confined within the band 90-110 kHz and shall not cause harmful interference outside the band to stations operating in accordance with the Regulations. In Regions 1 and 3, during the period prior to the international adoption of any long distance radionavigation system, the operation of specific radionavigation stations shall be subject to agreements between administrations whose services, operating in accordance with the Table, may be affected. Once established under such agreements, radionavigation stations shall be protected from harmful interference.

- (167) Only classes A1 or F1, A4 or F4 emissions are authorized in the band 90–160 kHz for stations of the fixed and maritime mobile services. Exceptionally, class A7J emission is also authorized in the band 90–160 kHz for stations of the fixed and maritime mobile
- (168) Aeronautical stations may use frequencies in the bands 110-112 kHz, 115-126 kHz and 129-130 kHz on a permitted basis for high-speed communications to aircraft.
- (179) In northern areas which are subject to auroral disturbances the aeronautical fixed service is the primary service.
- (181) Norwegian fixed stations situated in northern areas subject to auroral disturbances are allowed to continue operation on two frequencies in the band 385-395 kHz for transmissions chiefly composed of weather messages.
- (182) The frequency 410 kHz is designated for the maritime radionavigation service (radio direction-finding). Other allocated services in the band 405-415 kHz shall not cause harmful interference to radio direction-finding. In the band 405-415 kHz no frequency shall be assigned to coast stations.
- (185) In the European Maritime Area, subject to the conditions specified in the Final Acts of the European Maritime Conference (Copenhagen 1948), and any subsequent revision of that agreement, the administrations concerned may keep in the bands 415–485 kHz and 515–525 kHz such of the following broadcasting stations as will not cause harmful interference to the maritime mobile service: Hamar, Innsbruck, Oestersund, Oulu.
- (186) Limited to radiotelegraphy. (187) The frequency 500 kHz is the international distress and calling frequency for radiotelegraphy. The conditions for its use are prescribed in Article 32.
- (188) In operating stations of the aeronautical radionavigation service, the administrations concerned shall take all the technical steps necessary to avoid harmful interference to the maritime mobile service.
- (191) The carrier power of broadcasting stations in this band shall not exceed 250 watts.
- (198) In Region 2 the Loran system has priority. Other services to which the band is allocated may use any frequency in this band provided that they do not cause harmful interference to the Loran system.

In Region 3 the Loran system in any particular area operates either on 1850 or 1950 kHz, the bands occupied being 1825–1875 kHz and 1925–1975 kHz respectively. Other services to which the band 1800–2000 kHz is allocated may use any frequency therein on condition that no harmful interference is caused to the Loran system operating on 1850 or 1950 kHz.

(200) In Region 2, except in Greenland, coast stations and ship stations using radiotelephony shall be limited to class A3A or A3J emission and to a peak envelope power not exceeding 1kw. Preferably, the following carrier frequencies should be used: 2065.0 kHz, 2079 kHz, 2082.5 kHz, 2086.0 kHz, 2099.0 kHz, 2100.0 kHz, 2103.5 kHz.

(201) The frequency 2182 kHz is the international distress and calling frequency for radiotelephony. The conditions for the use of this frequency are prescribed in Article 35.

(202) For the conditions of use of this band by the broadcasting service see Nos. 135, 136, and 423 to 428.

(203) Standard frequency is 2500 kHz.

(204) The standard frequency guardbands at 2.5 MHz, 5 MHz, 10 MHz, 15 MHz, 20 MHz, and 25 MHz may be used by the radio astronomy service. The radio astronomy service shall be protected from harmful interference from services operating in other bands in accordance with the provisions of these Regulations, only to the extent that these services are protected from each other.

(208) In the U.S.S.R., in the bands 4063-4133 kHz and 4408-4438 kHz, fixed stations of limited power may operate provided that, in order to minimize the possibility of causing harmful interference to the maritime mobile service, they are situated at least 600 km from the coast. A limited power station is one whose power and antenna characteristics are so adjusted that the field strength established at any point in any direction does not exceed that obtainable with a nondirective antenna and a peak envelope power of 1 kw.

(209) On condition that harmful interference is not caused to the maritime mobile service, the frequencies between 4063 and 4438 kHz may be used exceptionally by fixed stations communicating only within the boundary of the country in which they are located, with a mean power not exceeding 50 watts; however, in Regions 2 and 3, between 4238 and 4368 kHz, a mean power not exceeding 500 watts may be used by such fixed stations.

(210) The standard frequency is 5000 kHz.

(211) On condition that harmful interference is not caused to the maritime mobile service, the frequencies between 6200 and 6525 kHz may be used exceptionally by fixed stations, communicating only within the boundary of the country in which they are located, with a mean power not exceeding 50 watts. At the time of notification of these frequencies, the attention of the International Frequency Registration Board will be drawn to the above conditions.

(213) Between 8615 and 8815 kHz, 12,925 and 13,200 kHz, and between 17,160 and 17,360 kHz, the U.S.S.R. will meet their special requirements for the fixed service with due regard to technical provisions (power, location, antenna, etc.) with a view to minimizing the possibility of harmful interference to the maritime mobile service. Coast stations in the maritime mobile service will also have due regard to technical provisions (power, location, antenna, etc.) with a view

to minimizing the possibility of harmful interference to the fixed service in the U.S.S.R. The International Frequency Registration Board will be consulted regarding this subject.

(214) The standard frequency is 10,000 kHz.

(215) The band 10,003-10,005 kHz is also allocated, on a secondary basis, to the space research service.

(215A) In Bulgaria, Cuba, Hungary, Poland, Roumania, Czechoslovakia and the U.S.S.R., the space research service is a primary service in the bands 15,762–15,768 kHz and 18,030–18,036 kHz.

(216) In the U.S.S.R., the band 11,400-11,450 kHz is also allocated to the aeronautical mobile (OR) service.

(217) The frequency 13,560 kHz is designated for industrial, scientific and medical purposes. Emissions must be confined within the limits of ± 0.05 percent of that frequency. Radiocommunication services operating within those limits must accept any harmful interference that may be experienced from the operation of industrial, scientific and medical equipment.

(218) In the U.S.S.R., the band 14,250-14,350 kHz is also allocated to the fixed service.

(219) The Standard frequency is 15,000 kHz.

(220) The standard frequency is 20,000 kHz.

(221) The band 19,990-20,010 kHz is also allocated, on a secondary basis, to the space research service.

(221A) The frequency 20,007 kHz may also be used, in emergency, in the search for, and rescue of, astronauts and space vehicles. Emissions must be confined in a band of ±3 kHz about this frequency.

(222) Intership radiotelegraphy may be used in the maritime mobile service between the frequencies 23,350 and 24,000 kHz.

(223) The standard frequency is 25,000 kHz.

(224) Limited to ship stations employing A1 or F1 emissions.

(225) The frequency 27,120 kHz is designated for industrial, scientific and medical purposes. Emissions must be confined within the limits of ± 0.6 percent of that frequency. Radiocommunication services operating within those limits must accept any harmful interference that may be experienced from the operation of industrial, scientific and medical equipment.

(226) In Region 2, Australia and New Zealand, the amateur service may operate between the frequencies 26,960 and 27,230 kHz.

(228) Stations designed to use ionospheric scatter may operate only subject to agreements between administrations concerned and those whose services, operating in accordance with the Table, may be affected.

(229) Systems designed to use ionospheric scatter or other fixed service systems designed to operate over distances exceeding 800 km shall confine their emissions to the following bands:

Region 1	Region 2	Region 3	
32. 6-33	32. 6-33	32. 6-33	
36. 2-36. 6	34. 6-35	34. 6-35	
39 -39. 4	36. 4-36. 8	36. 4-36. 8	

and shall have priority in Region 2 in the bands shown above for such use in that Region.

(230) In the case of the bands referred to in No. 229, which are limited in a particular Region the provisions of No. 117 shall apply and administrations shall avoid beaming such transmissions toward another Region unless specifically coordinated otherwise.

(231) Ionospheric scatter stations, existing on January 1, 1960, and not causing harmful interference to the other services to which the band is allocated, may continue to operate on frequencies now assigned until reaccommodated.

(232) Conventional (F2) long distance fixed service use of the band 29.7-30 MHz is not excluded in Region 2 provided that such use is coordinated between the administrations concerned.

(235) The band 39.986-40.002 MHz is also allocated, on a secondary basis, to the space research service.

(236) The frequency 40.68 MHz is designated for industrial, scientific and medical purposes. Emissions must be confined within the limits of ± 0.05 percent of that frequency. Radiocommunication services operating within those limits must accept any harmful interference that may be experienced from the operation of industrial, scientific and medical equipment.

(237) Systems designed to use ionospheric scatter which may cause harmful interference to the broadcasting service are prohibited.

(253A) In Region 2, fixed, mobile and broadcasting service operations previously authorized in the band 73-74.6 MHz may continue to operate on a non-interference basis to the radio astronomy service.

(253B) In Cuba, the band 73-74.6 MHz is also allocated to the fixed, mobile and broadcasting services.

(259) The frequency 75 MHz is assigned to aeronautical marker beacons. Administrations shall refrain from assigning frequencies close to the limits of the guard-band to stations of other services which, because of their power or geographical position, might cause harmful interference to marker beacons.

(273) The frequency 121.5 MHz is the aeronautical emergency frequency in this band; mobile stations of the maritime mobile service may communicate on this fre-

quency for safety purposes with stations of the aeronautical mobile service.

(273A) In the band 117.975-132 MHz and in the band 132-136 MHz where the aeronautical mobile (R) service is authorized, the use and development, for this service, of systems using space communication techniques may be authorized but limited initially to satellite relay stations of the aeronautical mobile (R) service. Such use and development shall be subject to coordination between administrations concerned and those having services operating in accordance with the Table, which may be affected.

(276) In Region 2 in the band 132-135 MHz, the aeronautical mobile (R) service shall operate on a primary basis subject to coordination between administrations concerned and those having services operating in accordance with the Table, which may be affected.

(281A) For the use of the band 136-137 MHz, see Recommendation No. 7A.

(281B) In Region 2, the band 136-137 MHz is also allocated to the fixed and mobile services until January 1, 1969. Thereafter, in Cuba, the band will continue to be allocated also to the fixed and mobile services.

(281E) In Regions 2 and 3, the band 137–138 MHz is also allocated to the fixed and mobile services until January 1, 1969. Thereafter, in Cuba, Malaysia, Pakistan and the Philippines, the band 137–138 MHz will continue to be allocated also to the fixed and mobile services.

(281F) The band 137-138 MHz will be used mainly for research concerning the establishment, technical improvement, and maintenance of operational space systems.

(284A) In the band 144-146 MHz, artificial satellites may be used by the amateur service.

(285A) The frequencies 148.25 ± 15 kHz and 154.2 ± 15 kHz may be used for space telecommand, subject to agreement among the administrations concerned and those having services operating in accordance with the Table, which may be affected.

(285B) Stations operating in the fixed and mobile services may continue to use this band until January 1, 1969. This cessation date shall not apply in Austria, Bulgaria, Cuba, Hungary, Iran, Kuwait, Morocco, Pakistan, the Netherlands, Poland, the United Arab Republic, Yugoslavia and Roumania where the fixed and mobile services will continue to have equal primary status with the radionavigation-satellite service. (See Recommendation No. 6A.)

(287) The frequency 156.8 MHz is the international safety and calling frequency for the maritime mobile VHF radiotelephone service. Administrations shall ensure that a guardband on each side of the frequency 156.8 MHz is provided. The conditions for the use of this frequency are contained in Article 35.

In the bands 156.025-157.425 MHz, 160.625-160.975 MHz, and 161.475-162.025 MHz, each

administration shall give priority to the maritime mobile service on only such frequencies as are assigned to stations of the maritime mobile service by that administration (see Article 35).

Any use of frequencies in these bands by stations of other services to which they are allocated should be avoided in areas where such use might cause harmful interference to the maritime mobile VHF radiotelephone service.

However, the frequency bands in which priority is given to the maritime mobile service may be used for radiotelephone communications on inland waterways, subject to agreements between interested and affected administrations and taking into account current frequency usage and existing agreements.

(294) The band 183.1-184.1 MHz is also allocated, and on a secondary basis, to the space research service.

(309A) Space stations employing frequencies in the band 267-273 MHz for telemetering purposes may also transmit tracking signals in the band.

(309B) In the band 267-272 MHz individual administrations may use space telemetering in their countries on a primary basis, subject to the agreement of the administrations concerned and those having services operating in accordance with the Table, which may be affected.

(310) Radio astronomy observations on the Deuterium line (322–329 MHz) are carried out in a number of countries under national arrangements. Administrations should bear in mind the needs of the radio astronomy service in their future planning of this band.

(311) Limited to Instrument Landing Systems (glide path).

(311A) Stations operating in the fixed and mobile services may continue to use this band until January 1, 1969. This cessation date shall not apply in Bulgaria, Cuba, Greece, Hungary, Iran, Kuwait, Lebanon, Morocco, the United Arab Republic and Yugoslavia where the fixed and mobile services will continue to have equal status with the radionavigation-satellite service. (See Recommendation No. 6A.)

(315A) Space stations employing frequencies between 401-402 MHz for telemetering purposes may also transmit tracking signals in this band.

(317) The band 404-410 MHz in Region 2 and the band 406-410 MHz in Regions 1 and 3 are also allocated to the radio astronomy service. An appropriate continuous band within these limits shall be designated on a national or area basis. In making assignments to stations of other services to which these bands are allocated, administrations are urged to take all practicable steps to protect radio astronomy observations from harmful interference.

(318) Radio altimeters may also be used, temporarily, in the band 420-460 MHz until

they are able to operate in a band allocated to the aeronautical radionavigation service or until they are no longer required.

(318A) In Bulgaria, Cuba, Hungary, Poland, Roumania, Czechoslovakia and the U.S.S.R., the band 460-470 MHz may be used, on a primary basis, by the meteorological-satellite service subject to agreement among administrations concerned and those having services, or intending to introduce services, operating in accordance with the Table, which may be affected.

(319A) The band 449.75-450.25 MHz may be used for space telecommand, subject to agreement among the administrations concerned and those having services operating in accordance with the Table, which may be affected.

(324A) It is intended that meteorologicalsatellite space stations operating in this band shall transmit to selected earth stations. The location of such earth stations is subject to agreement among administrations concerned and those having services operating in accordance with the Table, which may be affected.

(332) In Region 1, except the African Broadcasting Area, the band 606-614 MHz, and in Region 3, the band 610-614 MHz may be used by the radio astronomy service. Administrations shall avoid using the band concerned for the broadcasting service as long as possible, and thereafter, as far as practicable, shall avoid the use of such effective radiated powers as will cause harmful interference to radio astronomy observations.

In Region 2, the band 608-614 MHz is reserved exclusively for the radio astronomy service until the first Administrative Radio Conference after January 1, 1974, which is competent to review this provision; however, this provision does not apply to Cuba.

(339A) Specific portions of the frequency band 900-960 MHz may also be used, on a secondary basis, for experimental purposes in connection with space research.

(340) In Region 2, the frequency of 915 MHz is designated for industrial, scientific and medical purposes. Emissions must be confined within the limits of ± 25 MHz of that frequency. Radiocommunication services operating within those limits must accept any harmful interference that may be experienced from the operation of industrial, scientific and medical equipment.

(341) The band 960-1215 MHz is reserved on a world-wide basis for the use and development of airborne electronic aids to air navigation and any directly associated groundbased facilities.

(346) The use of the bands 1300-1350 MHz, 2700-2900 MHz and 9000-9200 MHz by the aeronautical radionavigation service is restricted to ground-based radars and, in the future, to associated airborne transponders which transmit only on frequencies in these bands and only when actuated by radars operating in the same band.

(350A) Space stations employing frequencies in the band 1525-1540 MHz for telemetering purposes may also transmit tracking signals in the band.

(350D) In Cuba, the band 1525-1535 MHz is also allocated, on a primary basis, to the mobile service.

(352A) The bands 1540-1660 MHz, 4200-4400 MHz, 5000-5250 MHz and 15.4-15.7 GHz are reserved, on a world-wide basis, for the use and development of airborne electronic aids to air navigation and any directly associated ground-based or satellite-borne facilities.

(352B) The bands 1540-1660 MHz, 5000-5250 MHz and 15.4-15.7 GHz are also allocated to the aeronautical mobile (R) service for the use and development of systems using space communication techniques. Such use and development is subject to agreement and coordination between administrations concerned and those having services operating in accordance with the Table, which may be affected.

(353A) In view of the successful detection of two spectral lines in the region of 1665 MHz and 1667 MHz by astronomers, administrations are urged to give all practicable protection in the band 1664.4–1668.4 MHz for future research in radio astronomy.

(354A) In Algeria, Bulgaria, Cuba, Hungary, Kuwait, Lebanon, Morocco, Pakistan, Poland, the United Arab Republic, Yugoslavia, Roumania, Czechoslovakia and the U.S.S.R., the bands 1660–1670 MHz and 1690–1700 MHz are also allocated to the fixed service and the mobile, except aeronautical mobile. service.

(355A) In Cuba, the band 1700-1710 MHz is also allocated to the fixed and mobile services

(356AA) In Bulgaria, Cuba, Hungary, Poland, Roumania, Czechoslovakia, and the U.S.S.R., the meteorological-satellite service, in the band 1770-1790 MHz, shall be on a primary basis, subject to coordination with the administration concerned and those having services operating in accordance with the Table, which may be affected by the siting of earth stations.

(356A) The band 2110-2120 MHz may be used for tele-command in conjunction with spacecraft engaged in deep space research, subject to agreement between the administrations concerned and those having services operating in accordance with the Table, which may be affected.

(356B) In Cuba, the band 2290-2300 MHz is also allocated to the fixed and mobile services.

(357) The frequency 2450 MHz is designated for industrial, scientific and medical purposes except in Albania, Bulgaria, Hungary, Poland, Roumania, Czechoslovakia, and the U.S.S.R., where the frequency 2375 MHz is used. Emissions must be confined within ±50 MHz of the frequencies designated. Radiocommunication services operating within

these limits must accept any harmful interference that may be experienced from the operation of industrial, scientific and medical equipment.

(364A) In Algeria, Bulgaria, Cuba, Hungary, India, Israel, Kuwait, Lebanon, Morocco, Pakistan, the Philippines, Poland, the United Arab Republic, Yugoslavia, Roumania, Czechslovakia and the U.S.S.R., the band 2690–2700 MHz is also allocated to the fixed and mobile services.

(365) In making assignments to stations in the fixed and mobile services, administrations are urged to take all practicable steps to protect radio astronomy observations from harmful interference.

(366) In the band 2700-2900 MHz groundbased radars used for meteorological purposes are authorized to operate on the basis of equality with stations of the aeronautical radionavigation service.

(367) The use of the band 2900-3100 MHz by the aeronautical radionavigation service is limited to ground-based radars.

(369) In the band 3100-3300 MHz, existing racons and shipborne radars in merchant ships may operate within the band 3100-3266 MHz.

(374A) This band may also be used for the transmission of tracking and telemetering signals associated with communicationsatellite space stations operating in the same hand.

(383A) In Cuba, the band 4990-5000 MHz is also allocated to the fixed and mobile services, and the provisions of No. 365 apply.

(385) The use of the band 5350-5470 MHz by the aeronautical radionavigation service is limited to airborne radars and associated airborne beacons.

(387) Between 5600 and 5650 MHz, ground-based radars used for meteorological purposes are authorized to operate on the basis of equality with stations of the maritime radionavigation service.

(389A) In Bulgaria, Cuba, Hungary, Poland, Roumania, Czechoslovakia and the U.S.S.R., the space research service is a primary service in the band 5670-5725 MHz.

(391) The frequency 5800 MHz is designated for industrial, scientific and medical purposes. Emissions must be confined within the limits of ± 75 MHz of that frequency. Radiocommunication services operating within those limits must accept any harmful interference that may be experienced from the operation of industrial, scientific, and medical equipment.

(392A) This band may also be used for the transmission of telecommand signals associated with communication-satellite earth stations operating in the same band.

(392B) The band 7120-7130 MHz may be used for telecommand in association with space services, subject to agreement between the administrations concerned and those having services operating in accordance with the Table, which may be affected.



- (392C) Stations of the fixed and mobile services, previously authorized in the bands 7250-7300 MHz and 7975-8025 MHz, may continue to operate until January 1, 1969. This provision does not apply to the countries listed in Nos. 392G and 392H.
- (392D) As an exception, passive communication-satellite systems also may be accommodated in the band 7250-7750 MHz, subject to:
- (a) Agreement between administrations concerned and those whose services, operating in accordance with the Table, may be affected:
- (b) The coordination procedure laid down in Article 9 and 9A.

Such systems shall not cause any more interference at active earth station receivers than would be caused by fixed or mobile services. Power-flux density limitations at the earth's surface after reflection from the passive communication-satellites shall not exceed those prescribed in these Regulations for active communication-satellite systems.

The maximum effective power radiated in any direction in the horizontal plane by earth stations of passive satellite systems shall not exceed +55 dbW, not taking the site shielding factor into account. If the distance between a transmitting station of a passive system and the territory of another administration exceeds 400 km, this limitation may be increased in that direction by 2 db for each 100 km in excess of 400 km up to a maximum of 65 dbW.

- (392F) In the bands 7200-7250 MHz and 7300-7750 MHz, the meteorological-satellite service may use a band up to 100 MHz in width on a primary basis. These bands may also be used for the transmission of tracking and telemetering signals associated with meteorological-satellite space stations operating in the same band.
- (392G) In Algeria, Austria, Bulgaria, Cyprus, Cuba, Ethiopia, Finland, Hungary, Japan, Kuwait, Lebannon, Liberia, Malaysia, Morocco, the Philippines, Poland, the United Arab Republic, Yugoslavia, Roumania, Sweden, Switzerland, Czechoslovakia, and the U.S.S.R., the band 7250–7300 MHz is also allocated to the fixed and mobile services.
- (392H) In Algeria, Bulgaria, Cuba, Ethiopia, Finland, Hungary, Japan, Kuwait, Lebanon, Morocco, Poland, the United Arab Republic, Yugoslavia, Roumania, Sweden, Switzerland, Czechoslovakia, and the U.S.S.R., the band 7975–8025 MHz is also allocated to the fixed and mobile services.
- (393) In Italy, the band 6450-6575 MHz is also allocated to the radiolocation service.
- (394C) In Cuba, the band 8400-8500 MHz is also allocated to the fixed and mobile services.
- (396) The use of the band 8750-8850 MHz by the aeronautical radionavigation service is limited to airborne Doppler navigation aids on a centre frequency of 8800 MHz.

- (399) The use of band 9300-9500 MHz by the aeronautical radionavigation service is limited to airborne weather radars, and ground-based radars. In this band ground-based radars used for meterorological purposes have priority over other radiolocation devices.
- (401A) The band 9975-10,025 MHz may be used by weather radar on meteorological-satellites.
- (404) Limited to continuous wave systems. (405B) In Algeria, Bulgaria, Cuba, Hungary, Japan, Kuwait, Lebanon, Pakistan, Poland, the United Arab Republic, Yugoslavia, Roumania, Czechoslovakia, and the U.S.S.R., the band 10.68-10.7 GHz is also allocated to the fixed and mobile services.
- (406) Limited to Doppler navigation aids. (409A) In Algeria, Buigaria, Cuba, Hungary, Kuwait, Lebanon, Morocco, Pakistan, Poland, the United Arab Republic, Yugoslavia, Roumania, Czechoslovakia, and the U.S.S.R., the band 15.25–15.35 GHz is also allocated to the fixed and mobile services.
- (409C) In Algeria, Bulgaria, Cuba, Hungary, Kuwait, Lebanon, Morocco, Pakistan, Poland, the United Arab Republic, Yugoslavia, Roumania, Czechoslovakia, and the U.S.S.R., the band 15.35-15.4 GHz is also allocated to the fixed and mobile services.
- (409D) In Bulgaria, Cuba, Hungary, Kuwait, Lebanon, Poland, the United Arab Republic, Roumania, Czechoslovakia, and the U.S.S.R., the band 19.3–19.4 GHz is also allocated to the fixed and mobile services.
- (410) The frequency 22.125 GHz is designated for industrial, scientific and medical purposes. Emissions must be confined within the limits of ± 125 MHz of that frequency. Radiocommunication services operating within those limits must accept any harmful interference that may be experienced from the operation of industrial, scientific, and medical equipment.
- (411) In the band 24.25-25.25 GHz, ground-based radionavigation aids are not permitted except where they operate in cooperation with airborne or shipborne radionavigation devices.
- (412A) In Bulgaria, Cuba, Hungary, Poland, the United Arab Republic, Roumania, Czechoslovakia, and the U.S.S.R., the band 31.3–31.5 GHz is also allocated to the fixed and mobile services.
- (412B) In Bulgaria, Cuba, Hungary, Poland, Yugoslavia, Roumania, Czechoslovakia, and the U.S.S.R., the space research service is a primary service in the band 31.8-32.3 GHz.
- (412C) In Bulgaria, Cuba, Hungary, Poland, Roumania, Czechoslovakia, and the U.S.S.R., the space research service is a primary service in the band 34.2–35.2 GHz.
- (412D) The band 34.4-34.5 GHz may be used by weather radar devices on meteorological satellites for the detection of cloud.
- (412E) In Bulgaria, Cuba, Hungary, Poland, Yugoslavia, Roumania, Czechoslovakia, and the U.S.S.R., the band 36.5–37.5 GHz is also allocated to the radio astronomy service.

(412F) In Cuba and India, the band 33-33.4 GHz is also allocated to the radio astronomy service.

(412G) In Bulgaria, Cuba, Hungary, Poland, Yugoslavia, Roumania, Czechoslovakia, and the U.S.S.R., the band 33.4-34 GHz is also allocated to the radio astronomy service.

(412H) In Bulgaria, Cuba, Hungary, Poland, Roumania, Czechoslovakia, and the U.S.S.R., the space research service is a primary service in the band 31-31.3 GHz.

NG FOOTNOTES

NG1 Operational fixed stations may be authorized to use frequencies in this band in accordance with columns 10 and 11 of the Table of Frequency Allocations, on the condition that harmful interference will not be caused to the reception of television stations on Channels 4 or 5. In any area in the continental United States (excluding Alaska), the aviation service and marine service may each be authorized to use four of the frequencies in the bands 72–73 and 75.4–76 MHz listed in column 10 for operational fixed stations in these services.

NG2 Facsimile broadcasting stations may be authorized in the band 88-108 MHz.

NG3 Fixed stations in the domestic fixed public service and control stations in the domestic public radio services may be authorized to use any of the frequencies in the band 72-73 and 75.4-76 MHz indicated in column 10 on the conditions that (a) harmful interference will not be caused to the reception of television stations on Channels 4 or 5 and (b) that harmful interference will not be caused to operational fixed stations.

NG4 The use of the frequencies in the band 152.84–153.38 MHz may be authorized, in any area, to remote pickup broadcast base and mobile stations on the condition that harmful interference will not be caused to stations operating in accordance with the Table of Frequency Allocations.

NG5 Licensees in the public safety radio services holding a valid authorization on June 30, 1958, to operate in the frequency band 156.27-157.47 MHz or on the frequencies 161.85, 161.91 or 161.97 MHz may, upon proper application, continue to be authorized for such operation, including expansion of existing systems, until such time as harmful interference is caused to the operation of any authorized station other than those licensed in the public safety radio service.

NG6 Stations in the public safety radio services authorized as of June 30, 1958, to use frequencies in the band 159.51–161.79 MHz in areas other than Puerto Rico and the Virgin Islands may continue such operation, including expansion of existing systems, on the condition that harmful interference will not be caused to stations in the services to which these bands are allocated. In Puerto Rico and the Virgin Islands this authority is limited to frequencies in the band 160.05–

161.37 MHz. No new public radio service system will be authorized to operate on these frequencies.

NG8 Frequencies in this band will be selected for assignment in such a manner that, on an engineering basis, the lowest frequency in the band is assigned which will not cause harmful interference to stations in that area already assigned frequencies in accordance with the Table of Frequency Allocation.

NG9 Aural brodacast intercity relay stations may be authorized to use the band 942-952 MHz on the condition that harmful interference will not be caused to other classes of stations operating in accordance with the Table of Frequency Allocations.

NG10 Frequencies in this band will be selected for assignment in such a manner that, on an engineering basis, the highest frequency in the band is assigned which will not cause harmful interference to stations in that area already assigned frequencies in accordance with the Table of Frequency Allocations.

NG11 Television inter-city relay stations may be authorized to use frequencies in this band on the condition that harmful interference will not be caused to stations operating in accordance with the Table of Frequency Allocations.

NG12 Frequencies in the bands 454.40–455 MHz and 459.40–460 MHz may be assigned to domestic public land and mobile stations to provide a two-way air-ground public radiotelephone service.

NG13 In those portions of the States of Texas and New Mexico in the area bounded on the south by parallel 31°53' N., on the east by longitude 105°40' W., on the north by parallel 33°24' N. and on the west by longitude 106°40' W., the frequency band 220-225 MHz is not available for use by amateur stations engaged in normal amateur operation between the hours of 0500 and 1800 local time Monday through Friday, inclusive, of each week. However, the entire frequency band 220-225 MHz shall be available in all areas to those amateur stations authorized to operate in an organized civil defense network during all periods when civil defense emergencies exist and, in addition, special arrangements for civil defense drills between the hours and within the area set forth above may be made upon mutual agreement between the Federal Communications Commission Engineer in Charge at Dallas, Tex., and the Area Frequency Coordinator at White Sands, N. Mex., if it apepars necessary to conduct such drills. Such arrangements shall specify dates and times, and will depend upon the degree of use of the frequency band at White Sands at any particular time.

NG14 Stations in the radiolocation service shall not cause harmful interference to stations in the disaster communications service between sunset and sunrise or at any time during an actual or imminent disaster. Conversely, stations in the disaster communications service shall not cause harmful in-



terference to stations in the radiolocation service between sunrise and sunset except during an actual or imminent disaster.

NG15(a) The Amateur Service may use the sections of the band 1800-2000 kHz which are not required for LORAN-A in accordance with the following conditions:

(1) The use of these frequencies by the Amateur Service shall not be a bar to the

expansion of the radionavigation (LORAN-A) service;

- (2) The Amateur Service shall not cause harmful interference to the radio navigation (LORAN-A) service;
- (3) Only types A1 and amplitude modulated double and single sideband telephony emission shall be employed;
 - (4) Amateur operation shall be limited to:

	Maximum DC plate input power in watts							
Area	1800-1825 kHz	1825-1850 kHz	1850-1875 kHz	1875-1900 kHz	1900-1925 kHz	1925-1950 kHz	1950–1975 kHz	1975-2000 kHz
	Day/ Night	Day/ Night	Day/ Night	Day/ Night	Day! Night	Day/ Night	Day/ Night	Day/ Night
Alabama	500/100	100/25	0	0	0	0	100/25	500/100
Alaska	200/50	0	0	200/50	0	0 / 00	0 (100	1000,000
Arizona Arkansas	0 1000/200	0 200/50	0 100/25	0	0	200/50 100/25	500/100 100/25	1000/200 500/100
California	0	200/00	0 20	ŏ	100/25	200/50	200/50	500/100
Colorado	200/50	0	0	o	0	200/50	200/50	1000/200
Connecticut	500/100	100/25	0	0	0	0	0	100/05
Delaware District of Columbia	500/100 500/100	100/25 100/25	0 0	0	0	0	0	100/25 100/25
Florida	500/100	100/25	ŏ	ŏ	ŏ	ŏ	100/25	500/100
Georgia	500/100	100/25	ŏ.	ŏ	ŏ	ŏ	0	200/50
Iawaii	0	0	Ō	0	200/50	100/25	100/25	500/100
daho.	100/25	0 200/50	0 100/25	100/25	100/25	100/25	100/25	500/100
llinoisndiana	1000/200 1000/200	200/50 500/100	100/25	0	0	0	0	200/50 200/50
owa	1000/200	200/50	200/50	ŏ	Ŏ	100/25	100/25	500/10
Cansas	500/100	100/25	100/25	0	0	100/25	200/50	1000/20
Kentucky	1000/200	500/100	100/25	0	0	0	0	200/50
Jouisiana	500/100 500/100	$\frac{100/25}{100/25}$	0	0	0	0	100/25	500/10
Maryland	500/100	100/25	ŏ	ŏ	ŏ	ŏ	ŏ	100/25
Aassachusetts	500/100	100/25	ŏ	ŏ	ŏ	ŏ	ŏ	0,20
dichigan	1000/200	500/100	100/25	0	0	0	0	100/25
Minnesota	500/100	100/25	100/25	100/25	100/25	100/25	100/25	500/10
Mississippi	500/100 1000/200	100/25 200/50	0 100/25	0	0	100/25	100/25 100/25	500/10 500/10
Montana	1000/25	200/30	0 20	100/25	100/25	100/25	100/25	500/10
Vebraska	500/100	100/25	100/25	0,20	0	200/50	200/50	1000/200
Yevada	0	0	Ō.	Ō	100/25	200/50	200/50	1000/20
New Hampshire	500/100	100/25	0	0	0	0	0	0
New Jersey New Mexico	500/100 100/25	100/25	0	0	0	100/25	0 500/100	0 1000/20
New York	500/100	100/25	ŏ	ŏ	ŏ	0	0	0
North Carolina	500/100	100/25	Ŏ	Ŏ	Ŏ	Ŏ	Ŏ	100/25
North Dakota	500/100	100/25	100/25	100/25	100/25	200/50	200/50	1000/20
Ohio	1000/200 500/100	500/100 100/25	100/25 100/25	Ü	0	0 100/25	0 200/50	100/25 1000/20
Oklahoma Oregon	0	0	0	0	200/50	100/25	100/25	500/10
ennsylvania	500/100	100/25	ŏ	ŏ	200/00	100/00	0,20	000, 10
Rhode Island	500/100	100/25	0	0	Ó	0	Ó	Ō
outh Carolina	500/100	100/25	0	0	0	0,0	0	200/50
outh Dakota Cennessee	500/100 1000/200	100/25 500/100	100/25 100/25	100/25	100/25	200/50	200/50	1000/20
emessee	200/50	0 100	0	ň	ŏ	ŏ	100/25	200/50 500/10
Jtah	100/25	0	ŏ	100/25	100/25	200/50	200/50	1000/20
ermont	500/100	100/25	Ō	Ŏ.	Ŏ,	Ŏ,	Ŏ,	O
rginia	500/100	100/25	0	0	0 (50	0	0	100/25
Vašhington Vest Virginia	1000/200	0 500/100	0 100/25	0	200/50	0	0	500/10 100/25
Visconsin	1000/200	200/50	200/50	ŏ	ŏ	ŏ	ŏ	200/50
Vyoming	200/50	0	o'	100/25	100/25	200/50	200/50	1000/200
Puerto Rico	500/100	100/25	0	0	0	0	0	200/5
irgin Islands Wan Island	500/100 500/100	$\frac{100/25}{100/25}$	0 0	0	0	0	0 100/25	200/50 500/10
errana Bank	500/100	100/25	ŏ	Ö	Ö	Ö	100/25	500/10
Roncador Kev	500/100	100/25	0	0	0	Ó	100/25	500/10
Navassa Island Baker, Canton, Ender-	500/100	100/25	Ō	Ō	Ö	Ō	0′	200/50
saker, Canton, Ender-	100/95	0	0	100/95	100/95	0	0	100/05
bury, Howland Juam, Johnston, Midway.	100/25 0	ő	0	100/25	100/25 100/25	Ů	Ů	100/25 100/25
American Samoa	200/50	Ö	ŏ	200/50	200/50	ŏ	ŏ	200/50
Vake	100/25	0	Ō	100/25	0	0	Ŏ	0
Palmyra, Jarvis	oʻ	0	0	0	200/50	0	0	200/50

(b) This footnote shall be considered as temporary in the sense that it shall remain subject to cancellation or to revisions, in whole or in part, by order of the Commission without hearing whenever the Commission shall deem such cancellation or revision to be necessary or desirable in the light of the priority within this band of the LORAN-A system of radionavigation.

NG16 For conditions which apply to the use of this band, refer to the North American Regional Broadcasting Agreement.

NG17 Stations in the Land Transportation Radio Services authorized as of May 15, 1958, to operate on the frequency 161.61 MHz may, upon proper application, continue to be authorized for such operation, including expansion of existing systems, on the condition that harmful interference will not be caused to the operation of any authorized station in the maritime mobile service. No new Land Transportation Radio Service system will be authorized to operate on 161.61 MHz.

NG19 Fixed stations associated with the maritime mobile service may be authorized, for purposes of communication with coast stations, to use frequencies assignable to ship stations in this band on the condition that harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations.

NG20 Fixed stations in the public safety radio service may be authorized the use of frequencies in this band which are authorized to base and mobile stations of this service on the condition that harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations.

NG21 In Hawaii, the frequency bands 76-88 MHz and 98-108 MHz are allocated exclusively to the fixed service for use by common carrier fixed stations for inter-island communications only. Spacing between assignable frequencies is not specified.

NG22 The frequency 2182 kHz may be authorized to fixed stations associated with the maritime mobile service for the sole purpose of transmitting distress calls and distress traffic, and urgency and safety signals and messages.

NG23 Frequencies in the band 2110-2200 MHz may also be assigned to stations in the international fixed public radio service located south of 25°30' north latitude in the State of Florida and in U.S. Possessions in the Caribbean area.

NG24 On condition that harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations, land stations authorized and used primarily as coast stations (not open to public correspondence) and associated land mobile stations may be authorized to use, on a secondary basis, the frequencies 156.35, 156.45 and 156.55 MHz: Provided, That, in each case, the frequency assignment will be common to the maritime mobile and land mobile services and that the

maritime mobile service shall be protected from harmful interference from the land mobile service.

NG25 The frequency 6147.5 kHz may be authorized for simplex operation by non-Government coast and ship radiotelephone stations operating in the Mississippi River system on the condition that harmful interference shall not be caused to stations operating in accordance with the Table of Frequency Allocations.

NG26 In Puerto Rico and the Virgin Islands only, the bands 154.04-154.46 MHz and 161.385-161.875 MHz are allocated exclusively to (a) aeronautical fixed service, (b) domestic fixed public service, and (c) international fixed public service and are not available to the mobile services.

NG27 The carrier frequencies 6451.9 and 6455.0 kHz may be authorized to ship telephone stations and coast telephone stations operating in the Mississippi River maritime mobile service system on the condition that harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations and that any interference from such services must be accepted.

NG28 The frequency band 160.86-161.40 MHz is available for assignment to remote pickup base and remote pickup mobile stations in Puerto Rico and the Virgin Islands only on a shared basis with the land transportation radio service.

NG30 Stations in the international fixed public radiocommunication service in Florida, south of 25°30' north latitude, may be authorized to use frequencies in the band 716-890 MHz on the condition that harmful interference will not be caused to the broadcasting service of any country. This is an interim allocation the termination of which will later be specified by the Commission when it is determined that equipments are generally available for use in bands allocated internationally to the fixed service.

NG31 The spacing between frequency assignments in this band shall be 10 kHz. The first and last assignable frequencies are those indicated in column 10.

NG32 The spacing between frequency assignments in this band shall be 20 kHz. The first and last assignable frequencies are those indicated in column 10.

NG33 The spacing between frequency assignments in this band shall be 20 kHz. The first and last assignable frequencies are those indicated in column 10.

NG34 The spacing between frequency assignments in this band shall be $50~\rm kHz$. The first and last assignable frequencies are those indicated in column 10.

NG35 The spacing between frequency assignments in this band shall be 100 kHz. The first and last assignable frequencies are those indicated in column 10.

NG36 The spacing between frequency assignments in this band shall be 200 kHz. The first and last assignable frequencies are those indicated in column 10.

NG38 The frequencies 152.3, 152.36, 152.42, 157.56, 157.62, and 157.68 MHz may be assigned exclusively to stations in the industrial radio service outside standard metropolitan areas of 50,000 or more population.

stations NG40 Non-Government fixed which were authorized on April 16, 1958, to use frequencies in the band 890-942 MHz may, upon the showing that interference is being caused by or to their assignments, be authorized to use frequencies in the band 942-952 MHz provided the bandwidth of emission does not exceed 1100 kHz and provided that an engineering study by the Commission indicates that the proposed frequency assignment for such stations in the band 942-952 MHz is likely to result in the elimination of the interference occurring in the band 890-942 MHz, and will not cause interference to existing operations in the band 942-952 MHz.

NG41 Frequencies in the bands 3700-4200 MHz, 5925-6425 MHz, and 10.7-11.7 GHz may also be assigned to stations in the international fixed public and international control services located in U.S. Possessions in the Caribbean area.

NG42 Non-Government stations in the radiolocation service shall not cause harmful interference to the amateur service.

NG43 Fixed stations in the domestic public radio services in Alaska, south of 56° north latitude and east of 134° west longitude, may be authorized to use frequencies in the band 800-830 MHz, on the condition that harmful interference will not be caused to the broadcasting service of any country.

NG44 Under exceptional circumstances, the frequencies 2638 and 2738 kHz may be authorized to coast stations.

NG45 Authorizations in this frequency band will be granted for omni-directional point-to-point operations only, excluding broadcast and auxiliary broadcast operations.

NG46 Licensees holding a valid authorization, on July 15, 1963, to operate in the frequency bands 6425–6525 MHz, 6525–6575 MHz, 7050–7125 MHz, and 10,550–10,680 MHz may continue to be authorized for such operation until July 15, 1968, on the condition that harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations.

NG47 Pending determination of final action concerning allocation of the 2500–2690 MHz band to the Instructional television service and further Order of the Commission, frequencies in this band are not available for operation of new radio systems comprised of operational fixed or international control stations except for those stations which comply with the technical standards established for the instructional television service. [Reference, Report and Order, FCC 63–722, in Docket No. 14744]

NG49 The frequencies 72.02, 72.04, 72.06, 72.08, 72.10, 72.12, 72.14, 72.16, 72.18, 72.20, 72.22, 72.24, 72.26, 72.28, 72.30, 72.32, 72.34, 72.36, 72.38, 72.40, 72.44, 72.48, 72.52, 72.56, 72.60, 75.44, 75.48, 75.52, 75,56, and 75.60 MHz

may be authorized for low powered (1 watt input) mobile operations in the manufacturers radio service subject to the condition that no interference is caused to the reception of television stations operating on Channels 4 and 5.

NG50 In the band 25.07-25.11 MHz, stations in the Industrial Radio Services shall not cause harmful interference to, and must accept interference from, stations in the Maritime Mobile Service operating in accordance with the International table of frequency allocations.

NG51 In Puerto Rico and the Virgin Islands only, the bands 150.88-150.98 MHz and 150.98-151.49 MHz are allocated exclusively to the Business Radio Service. Stations in the Land Transportation and Public Safety Radio Services in those territories which have been authorized as of January 1, 1965, to use frequencies in the bands 150.8-150.98 MHz and 150.98-151.49 MHz, respectively, may continue to operate on these frequencies until January 1, 1970.

NG52 Stations used to relay television signals to community antenna television systems, which are authorized to operate in the band 12.2–12.7 GHz on November 22, 1965, may continue to be authorized to so operate until February 1, 1971, under the conditions specified in that license.

NG53 In the band 12.7–12.95 GHz, television pickup stations shall not cause harmful interference to community antenna relay, television intercity relay and television STL stations.

NG55 Frequencies in the bands 3700-4200 MHz and 5925-6425 MHz will not be assigned to stations used to relay television signals to community antenna television systems after March 22, 1968: Provided, however, That stations which were authorized to so operate in these bands on March 22, 1968, may continue to be authorized to so operate until February 1, 1971, under the conditions specified in that license, and after February 1, 1971, only upon conditions that: (a) The station is not within fifty (50) miles of the coordinates of the principal city, as set forth in the U.S. Department of Commerce publication "Air Line Distances Between Cities in the United States," of one of the top 25 standard metropolitan statistical areas, as ranked by the U.S. Census Bureau, and (b) the station shall not cause harmful interference to fixed earth stations in the Communication Satellite Service or to fixed common carrier stations in the Domestic Public Service which are used for services other than the relay of television signals to community antenna television systems.

NG56 The frequencies 72.08, 72.24, 72.40, 72.96 and 75.64 MHz may be authorized for low-powered (one watt input) mobile operations in the Citizens Radio Service subject to the condition that interference will not be caused to remote control of industrial equipment operating on the same or adjacent frequencies and to the reception of television stations operating on Channels

4 or 5. TV interference shall be considered to occur whenever reception of regularly used television signals is impaired or destroyed, regardless of the strength of the television signals or the distance to the television station.

NG57 The frequency bands 154.4525–154.46 MHz, 154.46-154.4675 MHz and 154.4675-154.4750 MHz may also be assigned to fixed stations.

NG58 Low-power broadcast auxiliary stations licensed pursuant to the provisions of \$74.437 of this chapter may be authorized to operate in the band 942-952 MHz subject to the condition that no harmful interference is caused to stations operating in accordance with the Table of Frequency Allocations.

NG59 The frequencies 37.60 and 37.84 MHz may be authorized only for use by base, mobile and operational fixed stations participating in an interconnected or coordinated power service utility system. Existing operations not conforming to this limitation must be terminated by September 24, 1970.

NG60 Frequencies in this band may be authorized for communication with helicopters in support of off-shore drilling operations on the condition that harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations.

NG61 Frequencies in this band may be authorized for flight test purposes on the condition that harmful interference will not be caused to services operating in accordance with the Table of Frequency Allocations.

U.S. FOOTNOTES

US1 Pulsed emissions prohibited.

US2 The frequency 132.0 MHz may be authorized to non-Government stations only.

US3 The Government fixed, mobile except aeronautical mobile (R), and radiolocation services may be authorized in the band 132.0-136.0 MHz on condition that harmful interference is not caused to the aeronautical mobile (R) service.

US5 Until January 1, 1970, the frequencies 217.425 through 217.675 MHz and 219.325 through 219.575 MHz, inclusive, may be authorized for use by non-Government telemetering mobile stations aboard aircraft and telemetering land stations, for telemetering to and from aircraft in flight, when an engineering study indicates that harmful interference will not be caused to stations operating in accordance with the Table of Frequency Allocations.

US6 Radio altimeters operating in the band 420-460 MHz under a valid authorization on February 15, 1968, may continue to operate after that date, on the condition that harmful interference is not caused to stations of services operating in accordance with the U.S. Table of Frequency Allocations. Such authorizations may be renewed, but no new authorizations will be granted after February

15, 1968, and all radio altimeter operations in this band shall be discontinued prior to January 1, 1973.

US7 In the band 420-450 MHz and within the following areas, the DC plate power input to the final stage of a transmitter employed in the amateur service shall not exceed 50 watts, unless expressly authorized by the Commission after mutual agreement, on a case-by-case basis, between the Federal Communications Commission Engineer in Charge at the applicable District Office and the Military Area Frequency Coordinator at the applicable military base:

(a) Those portions of Texas and New Mexico bounded on the south by latitude 31°53′ North, on the east by longitude 105°40′ West, on the north by latitude 33°24′ North, and on the west by longitude 106°40′ West:

(b) The entire State of Florida including the Key West area and the areas enclosed within a 200-mile radius of Patrick Air Force Base, Fla. (latitude 28°21' North, longitude 80°43' West), and within a 200-mile radius of Eglin Air Force Base, Fla. (latitude 30°30' North, longitude 86°30' West);

(c) The entire State of Arizona;

(d) Those portions of California and Nevada south of latitude 37°10′ North, and the areas enclosed within a 200-mile radius of the U.S. Naval Missile Center, Point Mugu, Calif. (latitude 34°09′ North, longitude 119°11′ West).

US8 The use of the frequencies 170.475, 171.425, 171.575, and 172.275 MHz east of the Mississippi River, and 170.425, 170.575, 171.475, 172.225, and 172.375 MHz west of the Mississippi River may be authorized to fixed, land and mobile stations operated by non-Federal forest fire-fighting agencies. In addition, land stations and mobile stations operated by non-Federal conservation agencies, for mobile relay operation only, may be authorized to use the frequency 172.275 MHz east of the Mississippi River and the frequency 171.475 MHz west of the Mississippi River. The use of any of the foregoing nine frequencies shall be on the condition that no harmful interference will be caused to Government stations.

US10 The use of frequencies 26.62 MHz (in all areas), 143.90 MHz (in the continental United States excluding Alaska), and 148.15 MHz (in all areas) may be authorized to Civil Air Patrol land stations and Civil Air Patrol mobile stations on the condition that harmful interference will not be caused to Government stations.

US11 The use of the frequencies 166.250 and 170.15 MHz may be authorized to non-Government remote pickup broadcast base and land mobile stations and to non-Government base, fixed and land mobile stations in the public safety radio services (the sum of the bandwidth of emission and tolerance not to exceed 60 kHz) in the continental United States (excluding Alaska) only, except within the area bounded on the west by the Mississippi River, on the north by the parallel of

latitude 37°30′ N., and on the east and south by that arc of the circle with center at Springfield, Ill., and radius equal to the airline distance between Springfield, Ill., and Montgomery, Ala., subtended between the foregoing west and north boundaries, on the condition that harmful interference will not be caused to Government stations present or future in the Government band 162–174 MHz. The use of these frequencies by remote pickup broadcast stations will not be authorized for locations within 150 miles of New York City; and the use of these frequencies by the public safety radio services will not be authorized except for locations within 150 miles of New York City.

US13 For the specific purpose of transmitting hydrological and meteorological data in cooperation with agencies of the Federal Government, the following frequencies may be authorized to non-Government fixed stations on the condition that harmful interference will not be caused to Government stations:

MHz	MHz	MHz	MHz
169.425	170.275	171.125	406.075
169.450	170.300	171.825	406.125
169.475	170.325	171.850	406.175
169.500	171.025	171.875	412.625
169.525	171.050	171.900	412.675
170.225	171.075	171.925	412.725
170.250	171.100	406.025	412.775

Licensees holding a valid authorization on June 11, 1962, to operate on the frequencies 169.575, 170.375, 171.175, 171.975, or 406.050 MHz may continue to be authorized for such operation on the condition that harmful interference will not be caused to Government stations.

US14 This frequency band is not available to non-Government stations except that the frequency 512 kHz is available for use by non-Government ship telegraph stations, as a working frequency. When 500 kHz is being used for distress purposes, ship and coast stations may use 512 kHz for calling.

US15 The use of the frequency 540 kHz is subject to the conditions that no harmful interference is caused to the services operating on 500 kHz, and in the band 510-535 kHz

US16 Airdrome control stations may continue to be authorized to use frequencies in the band 290-325 kHz on the condition that harmful interference will not be caused to stations of services operating in accordance with the Table of Frequency Allocations.

US17 For the radiolocation activities of the petroleum industry only, radiolocation land stations and radiolocation mobile stations making use of SHORAN equipment may be authorized the use of the frequencies 230 MHz, 250 MHz and 310 MHz at locations within 150 miles of the respective ocean shorelines of Alaska and the contiguous 48 states (including the areas in and about bays and sounds), provided that no harmful interference is caused to stations oper-

ating in accordance with the Table of Frequency Allocations and provided that SHORAN operations are coordinated locally in advance with Federal Government authorities making use of frequencies in this band in the same area.

US18 Navigation aids in the United States and possessions between 90 and 110 kHz, 200 and 415 kHz, and 1800 and 2000 kHz, are normally operated by the U.S. Government. However, authorizations may be made by the Commission for non-Government operation in these bands subject to the conclusion of appropriate arrangements between the Commission and the Government agencies concerned and upon special showing of need for service which the Government is not yet prepared to render.

US19 In the band 405-415 kHz, stations in the aeronautical mobile service shall not cause harmful interference to stations in the radionavigation service.

US20 In Alaska, Government stations in the fixed service may be authorized to use frequencies in the band 72-73 and 75.4-76 MHz on the condition that harmful interference will not be caused to the reception of TV Channel 4.

US21 Existing Government operations and non-Government stations authorized in this band as of December 1, 1961, may continue and shall not be required to afford protection to radio astronomy observatories within the United States and its possessions. However, by international agreement, such stations must afford protection to the observatories of other countries.

US23 In Alaska, the frequency bands 76-88 MHz and 88-100 MHz are allocated to Government radio services and the non-Government fixed service.

US25 The use of frequencies in the band 25.85-26.1 MHz may be authorized in any area to non-Government remote pickup broadcast base and mobile stations on the condition that harmful interference is not caused to stations in the broadcasting service.

US26 The bands 117.975-121.425 MHz, 123.575-128.825 MHz and 132.025-136 MHz are for air traffic control communications,

US28 The band 121.65-121.975 MHz is for use by aeronautical utility land and mobile stations, and for air traffic control communications.

US29 The current use of the band 121.975-123.075 MHz by military aircraft is temporary and may continue until they are moved to an appropriate band.

US30 The band 121.975-123.075 MHz is available to FAA aircraft for communications pursuant to flight inspection functions in accordance with the Federal Aviation Act of 1958.

US31 Except as provided below, the band 121.975-123.075 MHz is for use by private aircraft stations.

The frequencies 122.80, 122.85, 122.95, 123.00, and 123.05 MHz may be assigned to aeronautical advisory stations.

The frequency 122.90 MHz may be assigned to aeronautical multicom stations.

Air carrier aircraft stations may use 122.00 MHz for communications with aeronautical stations of the Federal Aviation Administration and 122.8 MHz for communication with other aircraft and with aeronautical advisory stations.

Frequencies in the band 121.975-122.625 MHz may be used by aeronautical stations of the Federal Aviation Administration for communication with private aircraft stations only except that 122.0 MHz may also be used for communication with air carrier aircraft stations concerning weather information. US32 The band 123.075-123.575 MHz is

US32 The band 123.075-123.575 MHz is for (a) non-Government operations in accordance with the Commission's Rules and (b) for FAA communications incident to flight test activities pertinent to aircraft certification.

US33 The band 123.075-123.575 MHz is for use by flight test and aviation instructional stations.

US34 The only non-Government service permitted in the band 220-225 MHz is the amateur service. The amateur service shall not cause harmful interference to the radiolocation service.

US35 Except as provided by footnotes US6 and US87, the only non-Government service permitted in the band 420-450 MHz is the amateur service. The amateur service shall not cause harmful interference to the radio-location service.

US36 Each non-Government station in the fixed service, authorized to operate in the band 890-942 MHz and holding a valid authorization to so operate as of April 16, 1958, may continue to operate in that band for the duration of the term of its authorization in effect as of that date, and shall be protected from harmful interference from the radiolocation service during such term. Renewals of such authorizations, however, shall be contingent upon the condition that each such fixed station (1) accept any harmful interference that may be experienced from the operation of ISM equipments on 915 MHz (2) accept any harmful interference that may be experienced from the radiolocation service and (3) shall not cause harmful interference to the radiolocation service.

US37 The only non-Government service permitted in the band 1215-1300 MHz is the amateur service. The amateur service shall not cause harmful interference to the radiolocation service.

US38 The non-Government use of the band 1300-1350 MHz is limited to the aeronautical radionavigation service.

US39 Within the band 1540-1660 MHz, radio altimeters are permitted to use only the portion 1600-1660 MHz and then only until such time as international standardization of other aeronautical radionavigation systems or devices requires the discontinuance of radio altimeters in this band.

US40 The only non-Government service permitted in the band 2300-2450 MHz is the

amateur service. The amateur service shall not cause harmful interference to the radio-location service.

US41 The Government radiolocation service is permitted in the band 2450-2500 MHz on the condition that harmful interference is not caused to non-Government services.

US42 Temporarily, and until certain operations of the radiolocation service in the band 2700–2900 MHz can be transferred to other appropriate frequency bands, the aeronautical radionavigation and meteorological aids services may, in certain geographical areas, be subject to receiving some degree of interference from the radiolocation service.

US43 Non-Government land based radars in the aeronautical radionavigation service may be authorized in the band 2700-2900 MHz, subject to the conclusion of appropriate arrangements between the Commission and the Government agencies concerned, and upon special showing of need for service which the Government is not yet prepared to render.

US44 The non-Government radiolocation service may be authorized in the band 2900-3100 MHz on the condition that no harmful interference is caused to Government services.

US45 Each non-Government radiolocation land or radiolocation mobile station authorized to operate in the band 3100–3246 MHz, 3266–3300 MHz, or 9200–9300 MHz and holding a valid authorization to so operate as of April 16, 1958, may continue to so operate in that band for the duration of the term of its authorization in effect as of that date. Renewals of such authorizations, however, shall be contingent upon the condition that each such station shall not cause harmful interference to Government services.

US46 Each non-Government radionavigation station authorized to operate a shipborne radar in the band 3100-3246 MHz, and holding a valid authorization to so operate as of April 16, 1958, may continue to operate in that band for the duration of the term of its authorization in effect as of that date. Renewals of such authorizations, however, shall be contingent upon the condition that each such station must accept any harmful interference that might be experienced from Government services in that band.

US47 The band 4200-4400 MHz is reserved exclusively for radio altimeters until such time as international standardization of other aeronautical radionavigation systems or devices requires the discontinuance of radio altimeters in this band.

US48 The non-Government radiolocation service may be authorized in the band 5350-5460 MHz on the condition that it does not cause harmful interference to the aeronautical radionavigation service or to the Government radiolocation service.

US49 The non-Government radiolocation service may be authorized in the band 5460-5470 MHz on the condition that it does not cause harmful interference to the aeronauti-

cal or maritime radionavigation services or to the Government radiolocation service.

US50 The non-Government radiolocation service may be authorized in the band 5470-5600 MHz on the condition that it does not cause harmful interference to the maritime radionavigation service or to the Government radiolocation service.

US51 In the band 5600-5650 MHz, the non-Government radiolocation service shall not cause harmful interference to the Government radiolocation service.

US52 The only non-Government service permitted in the band 5650-5925 MHz is the amateur service. The amateur service shall not cause harmful interference to the radiologation service.

US53 In view of the fact that the band 13.25-13.4 GHz is allocated exclusively to Doppler navigation aids, Government and non-Government airborne Doppler radars in the aeronautical radionavigation service are permitted in the band 8750-8850 MHz only on the condition that they must accept any interference which may be experienced from stations in the radiolocation service in the band 8500-10.000 MHz.

US54 Temporarily, and until certain operations of the radiolocation service in the band 9000-9200 MHz can be transferred to other appropriate frequency bands, the aeronautical radionavigation service may, in certain geographical areas, be subject to receiving some degree of interference from the radiolocation service.

US55 The non-Government radiolocation service may be authorized in the band 9000–9200 MHz on the condition that harmful interference is not caused to the aeronautical radionavigation service or the Government radiolocation service.

US56 The non-Government radiolocation service may be authorized in the band 9300–9500 MHz on the condition that harmful interference is not caused to the Government radiolocation services.

US58 In the band 10,000-10,500 MHz, pulsed emissions are prohibited, except for weather radars on board meteorological-satellites in the band 10,000-10,025 MHz. The amateur service and the non-Government radiolocation service, which shall not cause harmful interference to the Government radiolocation service, are the only non-Government services permitted in this band. The non-Government radiolocation service is limited to survey operations using transmitters with a power not to exceed 1 watt into the antenna.

US59 The band 10,500-10,550 MHz is restricted to systems using type A0 emission with a power not to exceed 40 watts into the antenna.

US60 The use of this band by non-Government services is limited to the space (telecommand) service.

US61 The use of this band by non-Government services is limited to the amateur service.

US62 The use of this band by Government services is limited to the space research service.

US65 The use of the band 5460-5650 MHz by the maritime radionavigation service is limited to shipborne radars.

US66 The use of the band 9300-9500 MHz by the aeronautical radionavigation service is limited to airborne radars and associated airborne beacons.

US67 The use of the band 9300-9500 MHz by the meteorological aids service is limited to ground-based radars. Radiolocation installations will be coordinated with the meteorological aids service and, insofar as practicable, will be adjusted to meet the requirements of the meteorological aids service.

US68 The non-Government use of this band is limited to the radio astronomy service.

US69 In the band 31.8-33.4 GHz, ground-based radionavigation aids are not permitted except where they operate in cooperation with airborne or shipborne radionavigation devices.

US70 The meteorological aids service allocation in the band 400.05-406 MHz does not preclude the operation therein of associated ground transmitters.

US71 In the band 9300-9320 MHz, low-powered maritime radionavigation stations shall be protected from harmful interference caused by the operation of land-based equipment.

US72 In the band 24.25-25.25 GHz, Government radio-location devices (ASDM) are permitted between 24.25-24.47 GHz on a shared basis.

US74 The radio astronomy service shall be protected from extraband radiation only to the extent that such radiation exceeds the level which would be present if the offending station were operating in compliance with the technical standards or criteria applicable to the service in which it operates.

US77 Government stations may also be authorized:

- (a) Port operations used on a simplex basis by coast and ship stations of the frequencies 156.6 and 156.7 MHz;
- (b) Duplex port operations use of the frequency 157.0 MHz for ship stations and 161.6 MHz for coast stations;
- (c) Intership use of 156.3 MHz on a simplex basis.

US78 In the band 1435–1525 MHz, the frequencies between 1435 and 1485 MHz will be assigned primarily for the flight testing of manned aircraft, or major components thereof; the frequencies between 1485 and 1525 MHz will be assigned primarily for the flight testing of unmanned aircraft and missiles or major components thereof. Included as permissible usage for aeronautical telemetering stations in the band 1435–1525 MHz is telemetry associated with launching and re-entry into the earth's atmosphere, as well as any incidental orbiting prior to reentry, of manned or unmanned objects undergoing flight tests.



US80 Government stations may use the frequency 122.9 MHz subject to the following conditions.

- (a) All operation by Government stations shall be restricted to the purpose for which the frequency is authorized to non-Government stations, and shall be in accordance with the appropriate provisions of the Commission's rules and regulations, Part 87, Aviation Services;
- (b) Use of the frequency is required for coordination of activities with Commission licensees operating on this frequency; and
 (c) Government stations will not be au-

(c) Government stations will not be au thorized for operation at fixed locations.

US81 The band 38-38.16 MHz may be used by both Government and non-Government radio astronomy observatories. No new assignments are to be made and Government stations in the band 38-38.16 MHz will be moved to other bands on a case-by-case basis, as required, to protect radio astronomy observations from harmful interference. As an exception, however, low-powered military transportable and mobile stations used for tactical and training purposes will continue to use the band. To the extent practicable, the latter operations will be adjusted to relieve such interference as may be caused to radio astronomy observations. In the event of harmful interference from such local operations, radio astronomy observatories may contact local military commands directly, with a view to effecting relief. A list of military commands, areas of coordination, and points of contact for purposes of relieving interference may be obtained upon request from the Office of Chief Engineer, Federal Communications Commission, Washington, D.C. 20554.

US82 The assignable frequencies in this band may be authorized on a shared non-priority basis to Government and non-Government ship and coast stations (SSB telephony, with peak envelope power not to exceed 1 kw).

US83 Non-Government use of this band is limited to the following 9995-10,003 kHz, radio astronomy service; 10,003-10,005 kHz, radio astronomy and space research services.

US84 The non-Government use of this band is limited to the space research service.
US85 In the bands 117.975-123.075 and 123.575-136 MHz, the use and development, for the aeronautical mobile (R) service, of systems using space communication techniques may be authorized but limited initially to satellite relay stations of the

aeronautical mobile (R) service.

US86 The frequencies 148.25 MHz ±15 KHz and 154.2 MHz ±15 kHz may be used by Government and non-Government stations for space telecommand at specific locations, subject to such conditions as may be imposed on a case-by-case basis. With respect to 154.2 MHz, the commands are to be limited to short duration of the order of three seconds ("Address and execute" commands). Further, on a case-by-case basis and solely to avoid harmful interference to non-Government stations in the land mobile

service, a comparable replacement frequency assignment will be made available below 150.8 MHz, if required.

US87 The frequency 450 MHz, with maximum emission bandwith of 500 kHz, may be used by Government and non-Government stations for space telecommand at specific locations, subject to such conditions as may be applied on a case-by-case basis.

US88 Stations in the broadcasting service will not be authorized in the band 608-614 MHz prior to January 1, 1974. In the interim the band is available for use by the radio astronomy service shall be protected from extraband radiation only to the extent that such radiation exceeds the level which would be present if the offending station were operating in compliance with the technical standards or criteria applicable to the service in which it operates.

US89 The aeronautical telemetering frequencies in the band 1525–1535 MHz will be assigned primarily for the flight testing of unmanned aircraft and missiles or major components thereof. Included as permissible usage for aeronautical telemetering stations in the band 1525–1535 MHz is telemetry associated with launching and reentry into the earth's atmosphere, as well as any incidental orbiting prior to reentry, of manned or unmanned objects undergoing flight tests.

US90 The band 2110–2120 MHz may be

US90 The band 2110-2120 MHz may be used by Government and non-Government stations for space telecommand at specific locations in conjunction with spacecraft engaged in deep space research, subject to such conditions as may be applied on a case-by-case basis.

US91 The ultimate disposition of this band in the communication-satellite service, as between Government and non-Government, is deferred. In the meanwhile the non-Government may exploit the 4 and 6 GHz bands and the Government may exploit the 7 and 8 GHz bands for communication-satellite service systems intended to become operational. Any modification of this policy will be discussed and agreed in the FCC/DTM (IRAC) mechanism prior to the filing of applications with the IRAC for frequency assignments which are not in accordance with the foregoing.

US92 In the band 7300-7750 MHz, the meteorological-satellite service may use a band up to 100 MHz in width. This 100 MHz band may also be used for the transmission of tracking and telemetering signals associated with meteorological-satellite space stations operating in the same band.

US93 In the conterminous United States, the frequency 108.0 MHz may be authorized for use by VOR test facilities, the operation of which is not essential for the safety of life of property, subject to the condition that no interference is caused to the reception of FM broadcasting stations operating in the band 88–108 MHz. In the event that such interference does occur, the licensee or other agency authorized to operate the facility shall discontinue operation on 108 MHz and shall not

resume operation until the interference has been eliminated or the complaint otherwise satisfied. VOR test facilities operating on 108 MHz will not be protected against interference caused by FM broadcasting stations operating in the band 88-108 MHz nor shall the authorization of a VOR test facility on 108 MHz preclude the Commission from authorizing additional FM broadcasting stations.

US94 The bands 30.005-30.015 MHz and 39.986-40.02 MHz are also allocated, on a secondary basis, to the space research service for space to earth transmissions only.

US95 The band 66-72 MHz is not available for television broadcasting at Guam, Mariana Islands. Subject to agreement by the Commission, frequencies within this band may be authorized until July 1, 1970, for use by Government stations in the maritime mobile service in the Mariana Islands and vicinity.

US96 In the band 1990-2110 MHz, the frequencies 2106.4 MHz and 2101.8 MHz may be authorized for Government earth station transmissions in connection with Project Apollo, until December 31, 1970, at the following sites only:

Goldstone, Calif. (35°23'20" N., 116°50'53" W.)

Guam, Mariana Islands (13°18'34" N., 144°-44'10" E.)

Cape Kennedy, Fla. (28°28′54″ N., 80°34′35″ W.)

Kauai, Hawaii (22°07'31" N., 159°40'16" W.) Corpus Christi, Tex. (27°39'19" N., 97°22'49" W.)

Full power operation shall occur only when spacecraft launched as a part of Project Apollo are in actual flight. During such operation, the carrier shall be fully modulated at all times to ensure dispersal of the transmitted power, and transmissions shall not occur using antenna elevation angles of less than 3° above the horizontal plane. Operation at all other times shall be confined to laboratory tests or subdued radiation spacecraft tests, subject to the condition that no harmful interference is caused to TV broadcast auxiliary stations.

US97 The use of the band 1605-1715 kHz by non-Government stations in the aeronautical radionavigation service is limited to the frequencies 1638 and 1708 kHz. Stations in the radiolocation service shall not cause harmful interference to stations in the aeronautical radionavigation service operating on 1638 or 1708 kHz.

US98 The frequency 243 MHz is the frequency in this band for use by Government and non-Government survival craft stations and equipment used for survival purposes.

US99 In the band 1660-1700 MHz, the meteorological aids service (radiosonde) will to the maximum extent practicable confine its operations above the frequency 1670 MHz. Whenever it is necessary to operate radiosondes in the band 1660-1670 MHz within the United States, the radio astronomers will be notified in a timely manner.

US100 In the Additional Protocol to the Final Acts of the Space EARC, Geneva, 1963, a declaration on behalf of the USA states that the USA cannot accept any obligation to observe the exceptions claimed by Cuba in those footnotes to the Table of Frequency Allocations which were adopted by the EARC and which specifically name Cuba.

US101 In the band 1660-1670 MHz, the radio astronomy service must accept such interference as may be received from the meteorological-satellite service.

US102 In Alaska only, the frequency 122.1 MHz may also be used for air carrier air traffic control purposes at locations where other frequencies are not available to air carrier aircraft stations for air traffic control.

US103 Non-Government aeronautical radionavigation stations, intended to provide service for helicopter operations in the New York City area, may be authorized on the following frequencies:

kHz	kHz
70.8375	113.340
84.945	116.1735
85.005	127.5075
85.065	

Such authorizations shall be limited to the specific sites, coverage area and period of time in accordance with formal advice from the Federal Aviation Agency to the Federal Communications Commission that the service is required.

US104 The LORAN radionavigation system has priority in this band in the United States and Possessions.

US105 On the express condition that harmful interference is not caused to stations operating in accordance with the Table of Frequency Allocations, frequencies in the bands 3230-3240 and 3240-3400 kHz may be assigned to radiolocation systems which are also assigned frequencies in the 1605-1800 kHz band, provided the use of frequencies in the bands 3230-3240 and 3240-3400 kHz is necessary for the proper functioning of the particular radiolocation system.

US106 The frequency 156.75 MHz is available for assignment to non-Government and Government stations for environmental communications in accordance with an agreed plan.

US107 The frequency 156.8 MHz is the national distress, safety and calling frequency for the maritime mobile VHF radiotelephone service for use by Government and non-Government ship and coast stations. Guard bands of 156.7625-156.7875 and 156.8125-156.8375 MHz are maintained.

[34 F.R. 19421, Dec. 9, 1969, as amended at 34 F.R. 20199, Dec. 24, 1969]

Subpart C—Emissions

§ 2.201 Emission, modulation, and transmission characteristics.

The following system of designating emission, modulation and transmission characteristics shall be employed.

Digitized by Google

	Gum hai
 (a) Emissions are designated according to their classification and their necessary bandwidth. (b) Emissions are classified and symbolized according to the following characteristics: (1) Type of modulation of main car- 	(4) Telephony (including sound broad-casting)
rier.	(8) Multichannel voice-frequency teleg-
(2) Type of transmission.(3) Supplementary characteristics.	raphy
(c) Types of modulation of main car-	(e) Supplementary characteristics:
rier: (1) Amplitude A (2) Frequency (or Phase) F (3) Pulse P (d) Types of transmission: (1) Absence of any modulation intended to carry information 0 (2) Telegraphy without the use of a modulating audio frequency 1 (3) Telegraphy by the on-off keying of a modulating au dio frequency or audio frequencies, or by the on-off keying of the modulated emission	(1) Double sideband
(special case: an unkeyed modu- lated emission)2	(f) The classification of typical emissions is tabulated as follows:

Type of modulation of main carrier	Type of transmission	Supplementary characteristics	Sym- bol
Amplitude modulation	With no modulation Telegraphy without the use of a modulating audio frequency (by on-off keying). Telegraphy by the on-off keying of an amplitude modulating audio frequency or audio frequencies, or by the on-off keying of the modulated emission (special case: an unkeyed emission amplitude modulated). Telephony	Double sideband	A0 A1 A2 A3 A3A
	Facsimile (with modulation of main carrier either directly or by a frequency modulated subcarrier). Facsimile.	rier. Single sideband, suppressed carrier. Two independent sidebands	A3B A4
	Television	rier. Vestigial sideband Single sideband, reduced car- rier. Two independent sidebands	A5C A7A A9B
Frequency (or Phase) modulation.	Telegraphy by frequency shift keying without the use of a modulating audio frequency: one of two frequencies being emitted at any instant. Telegraphy by the on-off keying of a frequency modulating audio frequency or by the on-off keying of a frequency modulated emission (special case: an unkeyed emission, fre-		F1 F2
	quency modulated). Telephony Facsimile by direct frequency modulation of the carrier. Television Four-frequency diplex telegraphy Cases not covered by the above, in which the main carrier is frequency modulated.		F3 F4 F5 F6 F9

Type of modulation of main carrier	Type of transmission	Supplementary characteristics	Sym- bol
Pulse modulation	A pulsed carrier without any modulation in-		P0
	tended to carry information (e.g. radar). Telegraphy by the ou-off keying of a pulsed carrier without the use of a modulating audio frequency. Telegraphy by the on-off keying of a modulating audio frequency or audio frequencies, or by the on-off keying of a modulated pulsed carrier (special case: an unkeyed modulated		P1D
	pulsed carrier).	Audio frequency or audio frequencies modulating the amplitude of the pulses.	P2D
		Audio frequency or audio fre- quencies modulating the width (or duration) of the pulses.	P2E
		Audio frequency or audio frequencies modulating the phase (or position) of the pulses.	P2F
	Telephony	Amplitude modulated pulses Width (or duration) modu- lated pulses.	P3E
		Phase (or position) modulated pulses.	P3F
		Code modulated pulses (after sampling and quantization).	P3G
	Cases not covered by the above in which the main carrier is pulse modulated.		Po

- (g) Type B emission: As an exception to the above principles, damped waves are symbolized in the Commission's rules and regulations as type B emission.
- (h) Whenever the full designation of an emission is necessary, the symbol for that emission, as given above, shall be preceded by a number indicating in kilocycles per second the necessary bandwidth of the emission. Bandwidths shall generally be expressed to a maximum of three significant figures, the third figure being almost always a nought or a five.

§ 2.202 Bandwidths.

- (a) Occupied bandwidth: The frequency bandwidth such that, below its lower and above its upper frequency limits, the mean powers radiated are each equal to 0.5 percent of the total mean power radiated by a given emission. In some cases, for example multichannel frequency-division systems, the percentage of 0.5 percent may lead to certain difficulties in the practical application of the definitions of occupied and necessary bandwidth; in such cases a different percentage may prove useful.
- (b) Necessary bandwidth: For a given class of emission, the minimum value of the occupied bandwidth sufficient to en-

- sure the transmission of information at the rate and with the quality required for the system employed, under specified conditions. Emissions useful for the good functioning of the receiving equipment as, for example, the emission corresponding to the carrier of reduced carrier systems, shall be included in the necessary bandwidth.
- (c) The necessary bandwidth may be determined by one of the following methods:
- (1) Use of the formulas included in the following Table which also gives examples of necessary bandwidths and designation of corresponding emissions:
- (2) Computation in accordance with Recommendations of the International Radio Consultative Committee (C.C.I.R.);
- (3) Measurement, in cases not covered by subparagraphs (1) or (2) of this paragraph.
- (d) The value so determined should be used when the full designation of an emission is required. However, the necessary bandwidth so determined is not the only characteristic of an emission to be considered in evaluating the interference that may be caused by that emission.

(e) In the formulation of the table, the following terms have been employed:

 B_n =Necessary bandwidth in cycles per second.

B=Telegraph speed in bauds.

N=Maximum possible number of black plus white elements to be transmitted per second, in facsimile and television.

M=Maximum modulation frequency in cycles per second. C=Subcarrier frequency in cycles per second.

D=Half the difference between the maximum and minimum values of the instantaneous frequency. Instantaneous frequency is the rate of change of phase.

t=Pulse duration in seconds.

K=An overall numerical factor which varies according to the emission and which depends upon the allowable signal distortion.

I. AMPLITUDE MODULATION

Description and class of	Necessary bandwidth in cycles	Examples	
emission	per second	Details	Designation of emission
Continuous wave telegraphy, A1.	B _a =BK K=5 for fading circuits. K=3 for nonfading circuits.	Morse code at 25 words per minute, $B=20$, $K=5$. Bandwidth: 100 c/s. Four-channel time-division multiplex, 7-unit code, 42.5 bands per channel, $B=170$, $K=5$. Bandwidth: 850 c/s.	0.1A1 0.85A1
Telegraphy modulated by an audio frequency, A2.	B _n =BK+2M K=5 for fading circuits. K=3 for nonfading circuits.	Morse code at 25 words per minute, $B=20$, $M=1,000$, $K=5$. Bandwidth: 2,100 c/s.	2.1A2
Telephony, A3.	$B_n=M$ for single sideband. $B_n=2M$ for double sideband.	Double sideband telephony $M=3,000$. Bandwidth: 6,000 c/s. Single sideband telephony reduced carrier, $M=3,000$. Bandwidth: 3,000 c/s. Telephony, two independent sidebands, $M=3,000$. Bandwidth: 6,000 c/s.	6A3 3A3A 6A3B
Sound broadcasting A3.	B _n =2M M may vary between 4,000 and 10,000 depending on quality desired.	Speech and music, $M=4,000$. Bandwidth: 8,000 c/s.	8A3
Facsimile, carrier modu- lated by tone and by keying, A4.	B _n =KN+2M K=1.5	The total number of picture elements (black plus white) transmitted per second is equal to the circumference of the cylinder multiplied by the number of lines per unit length and by the speed of rotation of the cylinder in revolutions per second. Diameter of cylinder = 70 mm. Number of lines per mm=5. Speed of rotation=1 r.p.s. N=1,100. M=1,900. Bandwidth: 5,450 c/s.	5.45A4
Television (visual and aural) A5 and F3.	Refer to relevant CCIR docu- ments for the bandwidths of the commonly used tele- vision systems.	Number of lines = 525. Number of lines per second = 15,750. Video bandwidth: 4.2 Mc/s. Total visual bandwidth: 5.75 Mc/s. FM aural bandwidth including guard bands: 250,000 c/s. Total bandwidth: 6 Mc/s.	5750A5C 250F3
Composite transmission, $B_n=2M$ (double sideband)		Television relay, video limited to 4 Mc/s, audio on 6.5 Mc/s FM subcarrier, subcarrier deviation=50 kc/s. M=subcarrier frequency plus its maximum deviation=6.55×104. Bandwidth: 13.1×104 c/s.	13,100A9
Composite transmission, A9.	B _n =2M (double sideband)	Microwave relay system providing 10 telephone channels occupying baseband between 4 and 164 kc/s. M=164,000. Bandwidth: 328,000 c/s.	328A9

Title 47—Chapter I

II. FREQUENCY MODULATION

Description and class of	Necessary bandwidth in cycles	Examples	
emission	per second	Details	Designation of emission
Frequency-shift telegraphy: F1.	$B_n = 2.6D + 0.55B$ for $1.5 \le \frac{2D}{B} \le 5.5$. $B_n = 2.1D + 1.9B$ for $5.5 \le \frac{2D}{B} \le 20$.	Four-channel time-division multiplex with 7-unit code, 42.5 bands per channel, $B=170$, $D=200$; $\frac{2D}{B}=2.35$, therefore the first formula in column 2 applies. Bandwidth: 613 c/s	0.6F1
Commercial telephony: F3.	B _n =2M+2DK K is normally 1 but under certain conditions a higher value may be necessary.	For an average case of commercial telephony, D=15,000 M=3,000. Bandwidth: 36,000 c/s.	36F3
Sound broadcasting: F3.	$B_n=2M+2DK$	D=75,000, $M=15,000$ and assuming $K=1$. Bandwidth: 180,000 c/s	180F3
Facsimile: F4.	$B_n = KN + 2M + 2D$ $K = 1.5$	(See facsimile, amplitude modulation.) Diameter of cylinder = 70 mm. Number of lines per mm = 5. Speed of rotation = 1 r.p.s. $N=1,100$. $M=1,900$. $D=10,000$. Bandwidth: 25,450 c/s.	25.5F4
Four-frequency diplex Telegraphy: F6.	If the channels are not synchronized, $B_n=2.6D+2.75B$ where B is the speed of the higher speed channel. If the channels are synchronized the bandwidth is as for F1, B being the speed of either channel.	Four-frequency diplex system with 400 c/s spacing between frequencies, channels not synchronized, 170 bauds keying in each channel, $D=600$, $B=170$. Bandwidth: 2,027 c/s.	2, 05F6
Composite transmission:	B _n =2M+2DK	Microwave relay system providing 240 telephone channels occupying baseband between 60 and 1050 kilocycles. $M=1.05\times10^4$. $D=2.35\times10^4$; Bandwidth: 6.8×10^6 c/s.	6800F9
Composite transmission: F9.	$B_n = 2M + 2DK$	TV microwave relay, aural program on 7.5 Mc/s subcarrier; subcarrier deviation plus or minus 150 kilocycles. $M=$ subcarrier frequency plus maximum deviation = (7.5 plus 0.15) \times 10. $D=$ 1 \times 10 (visual) plus 0.3 \times 10 (aural). Bandwidth: 17 9 \times 10 c/s.	17, 900F9
Composite transmission: F9.	$B_n = 2M + 2DK$ $K = 1$	Stereophonic FM broadcasting (U.S. system) with multiplexed subsidiary communications subcarrier, M=75,000, D=75,000. Bandwidth: 300,000 c/s.	300F9
	III. Pulse Mo	DDULATION	46
Unmodulated pulse: P0	$B_n = \frac{2K}{t}$	t=3×10 ⁻⁴ , K=6	4000P0
	$B_{s=\frac{1}{t}}$ K depends on the ratio of pulse duration to pulse rise time. Its value usually falls between 1 and 10 and in many cases it does not need to exceed 6.	Bandwidth: 4×10 ⁶ c/s.	
Modulated pulse: P2 or P3.	The bandwidth depends on the particular types of mod- ulation used, many of these being still in the develop- ment stage.		
Composite transmission: P9.	$B_n = \frac{2K}{t}$ $K = 1.6$	Microwave relay, pulse-position mod- ulated by 36 channel baseband; pulse width at half amplitude = 0.4 microseconds. Bandwidth: 8×10 ⁶ c/s.	8000P9

Subpart D—Call Signs and Other Forms of Identifying Radio Transmissions

AUTHORITY: The provisions of this Subpart D issued under secs. 4, 5, 303, 48 Stat., as amended, 1066, 1068, 1082; 47 U.S.C. 154, 155, 303.

SOURCE: The provisions of this Subpart D appear at 34 F.R. 5104, Mar. 12, 1969, unless otherwise noted.

§ 2.301 Station identification requirement.

Each station using radio frequencies shall identify its transmissions according to the procedures prescribed by the rules governing the class of station to which it belongs with a view to the elimination of harmful interference and the general enforcement of applicable radio treaties, conventions, regulations, arrangements, and agreements in force, and the enforcement of the Communica-

tions Act of 1934, as amended, and the Commission's rules.

§ 2.302 Call signs.

The table which follows indicates the composition and blocks of international call signs available for assignment when such call signs are required to be transmitted for station identification by the rules pertaining to particular classes of stations. When stations operating in two or more classes are authorized to the same licensee for the same location, the Commission will assign a separate call sign to each station in a different class. (In addition to the U.S. call sign allocations listed below, call sign blocks AAA through AEZ and ALA through ALZ have been assigned to the Department of the Army; call sign block AFA through AKZ has been assigned to the Department of the Air Force; and call sign block NAA through NZZ has been assigned jointly to the Department of the Navy and the U.S. Coast Guard.)

Class of station	Composition of call sign	Call sign blocks
Coast (Classes I and II) except for coast tele- phone in Alaska.	3 letters	KAA through KZZ. WAA through WZZ.
Coast (Class III) and maritime radiodetermi- nation.	3 letters, 3 digits	KAA200 through KZZ999. WAA200 through WZZ999.
Coast telephone in Alaska	3 letters, 2 digits	KAA20 through KZZ99. WAA20 through WZZ99.
Fixed	• •	WAA20 through WZZ99.
Marine receiver test	3 letters, 3 digits (plus general geo- graphic location when required).	KAA2000 through KZZ999. WAA2000 through WZZ999.
Ship telegraph	4 letters 1	KAAA through KZZZ. WAAA through WZZZ.
Ship telephone	2 letters, 4 digits, or 3 letters, 4 digits 1	WA2000 through WZ9999, or descending from WZZ9999.
Ship telegraph plus telephone	4 letters	KAAA through KZZZ. WAAA through WZZZ.
Ship radar	graph call sign, or, if ship has no telephone or telegraph: 2 letters, 4 digits, or 3 letters, 4 digits	WA2999 through WZ9999, or descending from WZZ9999.
Ship survival craft	Call sign of the parent ship followed by 2 digits.	KAAA20 through KZZZ99. WAAA20 through WZZZ99.
Cable-repair ship marker buoy	by the letters "BT" and the inden- tifying number of the buoy.	•
Marine utility	2 letters, 4 digits	KA2000 through KZ9999.
Shipyard mobile	2 letters, 4 digits	KA2000 through KZ9999.
Aircraft telegraph	5 letters	WAAAA through WZZZZ.
Aircraft telegraph and telephone	5 letters 2	WAAAA through WZZZZ. WAAAA through WZZZZ.
Aircraft telephone	assigned)	KAAAA through KZZZZ. WAAAA through WZZZZ.
Aircraft survival craft	Whenever a call sign 2 is assigned, call sign of the parent aircraft fol- lowed by a single digit other than	
Aeronautical	3 letters, 1 digit 2	KAA2 through KZZ9. WAA2 through WZZ9.
Land mobile (base)	3 letters, 3 digits	WAA 2000 through KZZ 999.
Land mobile (mobile telegraph)	4 letters, 1 digit	KAAA2 through KZZZ9. WAAA2 through WZZZ9.
Land mobile (mobile telephone)	2 letters, 4 digits	KA2000 through KZ9999. WA2000 through WZ9999.
Broadcasting (standard)	4 letters 3 (plus location of station).	KAAA through KZZZ. WAAA through WZZZ.

See footnotes at end of table.

Class of station	Composition of call sign	Call sign blocks
Broadcasting (FM)	4 letters (plus location of station)	KAAA through KZZZ. WAAA through WZZZ.
Broadcasting with suffix "FM"	6 letters * (plus location of station)	KAAA-FM through KZZZ-FM.
		WAAA-FM through WZZZ-FM.
Broadcasting (television)	4 letters (plus location of station)	WAAA through WZZZ.
Broadcasting with suffix "TV"	_ 6 letters * (plus location of station)	KAAA-TV through KZZZ- TV.
		WAAA-TV through WZZZ-TV.
Television broadcast translator	lattama	KØ2AA through K83ZZ.
Disaster station, except U.S. Government	4 letters, 1 digit	KAAA2 through KZZZ9. WAAA2 through WZZZ9.
Experimental (letter "X" follows the digit)		KA2XAA through KZ9XZZ.
Amateur (letter "X" may not follow digit)	_ 1 letter, 1 digit, 2 letters 4	K1AA through KØZZ. W1AA through WØZZ.
Do	1 letter, 1 digit, 3 letters 4	K1AAA through KØZZZ. W1AAA through WØZZZ.
Do	2 letters, 1 digit, 2 letters 4	KA1AA through KZØZZ.
Do	2 letters, 1 digit, 3 letters 4	WAIAA through WZØZZ. KAIAAA through KZØZZZ.
Standard frequency		WA1AAA through WZØZZZ. WWV, WWVB through WWVI
Space station	2 letters, 2 digits	WWVL, WWVS. KA2Ø through KZ99.
Citizens radio		W A 20 through W Z99.
Citizens radio in trust territories	1 letter, 4 digits	K0001 through K9999.

NOTE: The symbol Ø indicates the digit zero.

§ 2.303 Other forms of identification of stations.

The following table indicates forms of identification which may be used in lieu of call signs by the specified classes of stations. Such recognized means of identification may be one or more of the following: name of station, location of sta-

tion, operating agency, official registration number, characteristic signal, characteristic of emission, or other clearly distinguishing form of identification readily recognized internationally. Reference should be made to the appropriate part of the rules for complete information on identification procedures for each service.

Class of station	Identification, other than assigned call sign
Aircraft (U.S. registry) telephone.	Registration number preceded by the type of the aircraft, or the radiotelephony designator of the aircraft operating agency followed by the flight identification number.
Aircraft (foreign registry) tele- phone.	Foreign registry identification consisting of five characters. This may be preceded by the radiotelephony designator of the aircraft operating agency or it may be preceded by the type of the aircraft.
Aeronautical	Name of the city, area, or airdrome served together with such additional identification as may be required.
Aircraft survival craft	Appropriate reference to parent aircraft, e.g., the air car- rier parent aircraft flight number or identification, the aircraft registration number, the name of the aircraft manufacturer, the name of the aircraft owner, or any other pertinent information.
Ship telephone	When an official call sign is not yet assigned: Complete name of the ship and name of licensee. On 156.65 Mc/s: Name of ship.
Public coast (radiotelephone)	The approximate geographic location in a format approved

¹ Ships with transmitter-equipped survival craft shall be assigned four letter call signs.

Ships with transmitter-equipped survival craft shall be assigned four letter call signs.
 See § 2.303.
 A 3 letter call sign now authorized for and in continuous use by a licensee of a standard broadcasting station may continue to be used by that station. The same exception applies also to frequency modulation and television broadcasting stations using 5 letter call signs consisting of 3 letters with the suffix "FM" or "TV".
 Plus other identifying data as may be specified.

Title 47—Chapter I

Class of station Fixed _____ Fixed: Rural subscriber service_ Land mobile: Public safety, forestry conservation, highway maintenance, local government, shipyard, land transportation, and aviation services. Land mobile: Industrial service_

Land mobile: Railroad radio

Land mobile: Broadcasting (re-

Broadcasting (Emergency

Broadcasting (television auxil-

Disaster station____

Amateur (RACES)

and rural radio.

mote pickup).

Broadcast System). Broadcasting (aural STL and

intercity relay).

Broadcasting booster).

iary).

service.

Identification, other than assigned call sign

Geographic location. When an approved method of superimposed identification is used, QTT DE (abbreviated name of company or station).

Assigned telephone number.

Name of station licensee (in abbreviated form if practicable), or location of station, or name of city, area, or facility served. Individual stations may be identified by additional digits following the more general identification.

Mobile unit cochannel with its base station: Unit identifler on file in the base station records. Mobile unit not cochannel with its base station: Unit identifier on file in the base station records and the assigned call sign of either the mobile or base station. Temporary base station: Unit designator in addition to base station identification.

Land mobile: Domestic public Special mobile unit designation assigned by licensee or by assigned telephone number.

Name of railroad, train number, caboose number, engine number, or name of fixed wayside station or such other number or name as may be specified for use of railroad employees to identify a specific fixed point or mobile unit. A railroad's abbreviated name or initial letters may be used where such are in general usage. Unit designators may be used in addition to the station identification to identify an individual unit or transmitter of a base station.

Identification of associated broadcasting station.

State and operational area identification.

Call sign of the broadcasting station with which it is associated.

Call sign of the TV broadcasting station with which it is licensed as an auxiliary, or call sign of the TV broadcasting station whose signals are being relayed, or by network identification.

(television Retransmission of the call sign of the primary station.

By radiotelephony: Name, location, or other designation of station when same as that of an associated station in some other service. Two or more separate units of a station operated at different locations are separately identified by the addition of a unit name, number, or other designation at the end of its authorized means of identification.

Tactical call signs. When two or more separate units of a station authorized to operate in the Radio Amateur Civil Emergency Service are operated independently at different locations, each unit shall separately identify itself by the addition of a unit number at the end of its call sign.

Subpart E—Distress, Disaster, and **Emergency Communications**

§ 2.401 Distress messages.

Each station licensee shall give absolute priority to radiocommunications or signals relating to ships or aircraft in distress; shall cease all sending on frequencies which will interfere with hearing a radiocommunication or signal of distress and except when engaged in answering or aiding the ship or aircraft in distress, shall refrain from sending any radiocommunications or signals until there is assurance that no interference will be caused with the radiocommunications or signals relating thereto; and shall assist the ship or aircraft in distress, so far as possible, by complying with its instructions.

§ 2.402 Control of distress traffic.

The control of distress traffic is the responsibility of the mobile station in distress or of the mobile station which, by the application of the provisions of § 2.403, has sent the distress call. These stations may, however, delegate the control of the distress traffic to another station.

§ 2.403 Retransmission of distress message.

Any station which becomes aware that a mobile station is in distress may transmit the distress message in the following cases:

(a) When the station in distress is not itself in a position to transmit the message.

(b) In the case of mobile stations, when the master or the person in charge of the ship, aircraft, or other vehicles carrying the station which intervenes believes that further help is necessary.

(c) In the case of other stations, when directed to do so by the station in control of distress traffic or when it has reason to believe that a distress call which it has intercepted has not been received by any station in a position to render aid.

§ 2.404 Resumption of operation after distress.

No station having been notified to cease operation shall resume operation on frequency or frequencies which may cause interference until notified by the station issuing the original notice that the station involved will not interfere with distress traffic as it is then being routed or until the receipt of a general notice that the need for handling distress traffic no longer exists.

§ 2.405 Operation during emergency.

The licensee of any station (except amateur, standard broadcast, FM broadcast, noncommercial educational FM broadcast, or television broadcast) may, during a period of emergency in which normal communication facilities disrupted as a result of hurricane, flood, earthquake, or similar disaster, utilize such station for emergency communication service in communicating in a manner other than that specified in the instrument of authorization: Provided: (a) That as soon as possible after the beginning of such emergency use, notice be sent to the Commission at Washington, D.C., and to the Engineer in Charge of the district in which the station is

located, stating the nature of the emergency and the use to which the station is being put, and (b) That the emergency use of the station shall be discontinued as soon as substantially normal communication facilities are again available. and (c) That the Commission at Washington, D.C., and the Engineer in Charge shall be notified immediately when such special use of the station is terminated: Provided further, (d) That in no event shall any station engage in emergency transmission on frequencies other than, or with power in excess of, that specified in the instrument of authorization or as otherwise expressly provided by the Commission, or by law: And provided further, (e) That any such emergency communication undertaken under this section shall terminate upon order of the Commission.

NOTE: Part 3 of this chapter contains provisions governing emergency operation of standard, FM, noncommercial educational FM, and television broadcast stations. Part 12 of this chapter contains such provisions for amateur stations.

[28 F.R. 13785, Dec. 18, 1963]

§ 2.406 National defense; free service.

Any common carrier subject to the Communications Act may render to any agency of the United States Government free service in connection with the preparation for the national defense. Every such carrier rendering any such free service shall make and file, in duplicate, with the Commission, on or before the 31st day of July and on or before the 31st day of January in each year, reports covering the periods of 6 months ending on the 30th day of June and the 31st day of December, respectively, next prior to said dates. These reports shall show the names of the agencies to which free service was rendered pursuant to this rule, the general character of the communications handled for each agency, and the charges in dollars which would have accrued to the carrier for such service rendered to each agency if charges for all such communications had been collected at the published tariff rates.

§ 2.407 National defense; emergency authorization.

The Federal Communications Commission may authorize the licensee of any radio station during a period of national emergency to operate its facilities upon such frequencies, with such power and points of communica-

tion, and in such a manner beyond that specified in the station license as may be requested by the Army, Navy, or Air Force.

Subpart F—Equipment Type Approval and Type Acceptance

GENERAL

§ 2.551 Program defined.

In order to carry out its responunder the Communications sibilities Act and the various treaties and international regulations, it is necessary for the Commission to ascertain that the equipment involved is capable of meeting the technical operating standards set forth in said statutes. treaties and the Commission's rules and regulations. To facilitate such determinations in those services where equipment is generally standardized, to promote the improvement of equipment and to promote the efficient use of the radio spectrum, the Commission has designed two specific procedures for securing advance approval of equipment. These procedures are designated as type approval and type acceptance. Ordinarily, type approval contemplates tests conducted by Commission personnel, while type acceptance is based on data concerning the equipment submitted by the manufacturer or the individual prospective licensee. The procedures described in this subpart are intended to apply to equipment in those services which specifically require either type approval or type acceptance. These procedures may also be applied to equipment components, such as radio frequency power amplifiers, etc., to the extent specified in the rules of the particular service in which such components will be used.

§ 2.553 Radio equipment list.

A list of type approved and type accepted equipment is published periodically by the Commission. Public notice of type acceptance and type approval will be by publication in the equipment list, a copy of which will be furnished each of manufacturer listed equipment. Equipment which was listed prior to May 16, 1955, will be continued on the list unless it is removed by Commission action in accordance with the provisions of § 2.575. Copies of the Radio Equipment List are available for inspection at the Commission offices in Washington, D.C., and at each of its field offices. The

Radio Equipment List is published in three parts:

Part A, Television Broadcast Equipment Part B, Aural Broadcast Equipment Part C, Other than Broadcast Equipment

§ 2.555 Making available type approved or type accepted equipment for testing or inspection.

Upon request by the Commission any manufacturer of equipment which has been type approved or type accepted by the Commission shall cooperate in making available to the Commission models of said type approved or type accepted equipment in order that the equipment may be tested or inspected either at the place of manufacture or at the Commission's laboratory at Laurel, Maryland.

§ 2.557 Limitation on availability of equipment files for public reference.

(a) Files containing information about equipment submitted by manufacturers and other persons pursuant to the rules in this part will not be open to the public.

(b) The Commission will cooperate with a manufacturer's desire to withhold the addition of new equipment to the radio equipment list until a date no earlier than that specified by the manufacturer.

TYPE APPROVAL

§ 2.561 Type approval.

(a) Type approval is normally based on tests performed at the Commission's laboratory at Laurel, Maryland. In certain cases, type approval may be based on tests performed at other locations provided the tests are conducted in accordance with procedures specified by the Commission and by or under the direction and supervision of Commission personnel.

(b) Application for type approval may be in the form of a letter addressed to the Chief, Laboratory Division, Federal Communications Commission, Post Office Box 40, Laurel, Md. 20810. The letter shall specify the part of the rules under which type approval is desired and shall include any information specifically required to be submitted under such part of the rules. In addition, the request shall describe the equipment to be tested and include the size and weight of each component. In most cases, the laboratory will advise the applicant to ship the equipment prepaid to Chief, Laboratory Division, Post Office Box 40, Laurel, Md.

20810, complete with operating instructions and circuit diagrams. Upon completion of the tests, the equipment will be returned to the applicant, shipping

charges collect.

(c) In the event of failure of the equipment to meet the Commission's technical requirements, notice may be given directly by the Chief, Laboratory Division, and arrangements made for modification or adjustment as required. [28 F.R. 12465, Nov. 22, 1963, as amended at 33 F.R. 14167, Sept. 19, 1968]

§ 2.563 Limitations on type approval.

(a) Type approval is limited to a determination that, if the equipment is properly maintained and operated and no unauthorized change whatsoever is made in its construction, it is capable of complying with the technical requirements of the applicable part of the rules. Type approval shall not be construed as a determination with respect to features not covered by the rules of the service under which the equipment is approved.

(b) Type approval shall not be construed to mean that the equipment will continue to be satisfactory as the Commission's technical standards may be changed to conform with progress in

the state of the art.

§ 2.565 Withdrawal or refusal of type approval.

(a) Type approval may be withdrawn, if upon subsequent inspection or operation it is determined that the manufacturer has made unauthorized changes in the equipment or that the equipment does not comply with the technical requirements of the applicable part of the rules. The procedure for withdrawal of type approval in such cases shall be the same as that prescribed by the Commission for revocation of a radio station license pursuant to the provisions of the Communications Act of 1934, as amended.

(b) In the event changes in the Commission's technical standards necessitate the withdrawal of type approval, the procedure to be followed will be set forth in the order finalizing the revised technical standards after appropriate rule

making proceedings.

(c) When type approval has been withdrawn, the manufacturer shall make no further sale of equipment which in any manner indicates that such equipment meets the type approval requirements of the Commission.

(d) When type approval has been withdrawn for unauthorized changes by

the manufacturer, the Commission will consider that fact in determining whether the manufacturer in question is eligible to receive any new type approval.

(e) Any person affected by a refusal to grant type approval may file a petition for reconsideration within 30 days after written notice of such refusal has been issued, as provided in § 1.106 of this chapter.

§ 2.567 Identification and changes in type approved equipment.

(a) Each type of equipment for which type approval is requested shall be identified by a type number assigned by the manufacturer of the equipment. The type number shall consist of a series of not more than a total of seventeen digits, letters, punctuation marks, and spaces. The type number shall be shown on a name plate affixed in a conspicuous place to such equipment. All equipment used pursuant to type approval shall have affixed to it such indication of type approval as may be required in the service rules governing such equipment.

(b) No change whatsoever may be made in the design of type approved equipment without prior authorization from the Commission. When a change is requested, the Commission may authorize the change or require that the modified equipment be identified with a new type number and be resubmitted

for type approval tests.

(c) If the assignment of a different type number is required as a result of equipment modification, a new name plate bearing the new type number shall be affixed to the modified equipment.

(d) Users of type approved equipment shall not modify their own equipment.

TYPE ACCEPTANCE

§ 2.571 Type acceptance.

(a) Type acceptance of equipment is based on representations and test data submitted to the Commission by the manufacturer or prospective licensee. This information may be subject to check by Commission measurements.

(b) A separate request for type acceptance shall be submitted for each different type of equipment. Each request shall be signed by the applicant or by a duly authorized representative who shall certify that the application was prepared by him or at his direction and that to the best of his knowledge and belief the facts set forth in the application and accom-

panying technical data are true and correct. The technical test data required to be submitted shall be certified by the engineer who performed or supervised the tests who shall attach a brief statement of his qualifications.

(c) In the event there is no Commission action on an application for type acceptance within thirty (30) days after the application is filed, or within 30 days after the applicant has submitted additional data at the request of the Commission, the equipment will be deemed approved in accordance with the Commission's type acceptance procedure. The term Commission action as used in this section means either issuance of a public notice, a listing of the accepted equipment on the equipment list or sending a letter or post card to the applicant.

(d) Notice to applicant of type acceptance will be by post card and public notice will be given by listing the accepted equipment on the Commission's equipment list. Notice that an application has been found unacceptable for type acceptance will be by letter to the applicant.

[28 F.R. 12465, Nov. 22, 1963, as amended at 29 F.R. 10585, July 30, 1964]

§ 2.573 Limitation on type acceptance.

- (a) Type acceptance is limited to a finding that, insofar as can be determined from the data submitted, and if the equipment is properly maintained and operated and no change is made in its construction, except as provided for in § 2.584, the equipment complies with current technical standards of the service in which the equipment will be operated. The fact that a particular equipment has been type accepted for licensing purposes shall not be construed as a determination with respect to mechanical features, nor of reliability under service conditions, except as provided for in the rules of the service under which the equipment is accepted.
- (b) Type acceptance shall not be construed to mean that the equipment will continue to be satisfactory as the Commission's technical standards may be changed to conform with progress in the state of the art.
- § 2.575 Withdrawal or refusal of type acceptance.
- (a) Type acceptance may be withdrawn, if upon subsequent inspection

or operation it is determined that the manufacturer has made changes in the equipment other than as provided for in § 2.584, or that the equipment does not comply with the technical requirements of the applicable part of the rules. The procedure for withdrawal of type acceptance shall be the same as that prescribed by the Commission for revocation of a radio station license pursuant to the provisions of the Communications Act of 1934, as amended.

(b) In the event changes in the Commission's technical standards necessitate the withdrawal of type acceptance, the procedure to be followed will be set forth in the order finalizing the revised technical standards after appropriate

rule making proceedings.

(c) When type acceptance has been withdrawn, the manufacturer shall make no further sale of equipment which in any manner indicates that such equipment meets the type acceptance requirements of the Commission.

(d) When type acceptance has been withdrawn for changes made by the manufacturer where such changes are not in accordance with the provisions of § 2.584, the Commission will consider that fact in determining whether the manufacturer in question is eligible to receive any new type acceptance.

(e) Any person affected by a refusal to grant type acceptance may file a petition for reconsideration within 30 days after written notice of such refusal has been issued, as provided in §1.106 of

this chapter.

§ 2.577 General information required for type acceptance.

(a) Each request for type acceptance of equipment shall include the information listed in paragraph (b) of this section and in §§ 2.579 and 2.582. This information is general and is the minimum required for all equipments. In many cases, additional information specific to a particular service is also required. The applicant should carefully read the rules of the service in which the proposed equipment is intended to be operated to make sure that such additional specific information is furnished. If deemed necessary, the Commission may require additional information, test data, or testing in its own Laboratory at Laurel, Maryland, before determining the acceptability of any specific equipment.

- (b) The request shall include the following information insofar as it is applicable to the equipment:
- (1) The type number of the equipment in accordance with § 2.584.
- (2) The service and rule part under which the equipment is intended to be operated.
- (3) Description of equipment: The description of equipment should include the type of emission, frequency range, power rating as defined in the applicable part of the rules, voltages applied to and currents into the several elements of the final amplifier tube or tubes for normal operation, function of each tube, circuit diagrams, instruction books when available, tune-up procedure, a description of the oscillator circuit and any devices installed for the purpose of frequency stabilization. When circuits or devices are employed for limiting modulation or suppression of spurious radiation a description of these should be included. The description should be sufficiently complete to develop all factors that may affect a determination as to whether the equipment will comply with the technical standards of the applicable rule parts.
- (4) Photographs of equipment: Adequately identified photographs of sufficient size and clarity to reveal equipment construction and layout should be furnished. These should include at least one view showing the control panel or panels, including meters and labels for controls and meters, and sufficient views of the internal construction to define component placement and chassis assembly. Insofar as these requirements are met by photographs or drawings contained in instruction manuals supplied with the type acceptance request, additional photographs need be sufficient only to complete the required showing.
- § 2.579 Measurement data required for type acceptance.

Measurements shall be made to establish the following:

- (a) Radio frequency power output at the RF output terminals when the transmitter is adjusted in accordance with the tune-up procedure to give the values of current and voltage on the circuit elements specified in § 2.577(b) (3). Give details of the radio frequency load attached to the output terminals when this test is made.
- (b) Modulation characteristics: (1) Voice modulated communications equip-

- ment: A curve or equivalent data showing the frequency response of the audio modulating circuit over a frequency range of 100 to 5000 cycles shall be submitted.
- (2) Other types of equipment: A curve or equivalent data will be submitted which shows that the equipment will meet the modulation requirements of the rules under which the equipment is to be licensed.
- (3) Equipment which employs modulation limiting: If a modulation limiting device or circuit is incorporated in the equipment a curve showing the percentage of modulation versus the modulation input voltage shall be supplied.
- (c) Bandwidth occupied: The frequency bandwidth such that, below its lower and above its upper frequency limits, the mean powers radiated are each equal to 0.5 percent of the total mean power radiated by a given emission; measured under the following conditions as applicable:
- (1) Telegraph transmitters for manual operation—when keyed at 16 dots per second.
- (2) Other keyed transmitters—when keyed at the maximum machine speed.
- (3) Voice modulated transmitters equipped with a device to prevent overmodulation—when modulated by an input signal 16 db greater than that required to produce 50 percent modulation: Test at 2500 cycles.
- (4) Voice modulated transmitters without a device to prevent overmodulation—when modulated by an input signal large enough to produce at least 85 percent modulation: Test at 2500 cycles.
- (5) Standard broadcast transmitters—when modulated with a frequency of 7500 cycles at 85 percent modulation. FM broadcast transmitters, including TV aural transmitters—when modulated with a frequency of 15 kc/s at 85 percent modulation.
- (6) Transmitters in which the modulating baseband comprises more than three independent channels—when modulated with a test signal consisting of a band of random noise extending continuously from below 20 kc/s to the highest frequency in the baseband. The level of the test signal shall be adjusted to provide RMS modulation which is 22.4 percent of the full rated peak modulation of the transmitter. The test signal shall be applied through any preemphasis networks used in normal service.

- (7) Transmitters in which the modulating baseband comprises not more than three independent channels—when modulated by the full complement of signals for which the transmitter is rated. The level of modulation for each channel should be set to that prescribed in rule parts applicable to the services for which the transmitter is intended. If specific modulation levels are not set forth in the rules, the test levels should provide the manufacturer's maximum rated condition.
- (8) Transmitters designed for other types of modulation—when modulated by an appropriate signal of sufficient amplitude to be representative of the type of service in which used. A description of the input signal used should be supplied.

(d) Spurious emissions from the transmitting equipment for the follow-

ing conditions:

- (1) Radio frequency voltage measurements at the antenna terminals. The generated frequency voltages within the equipment and appearing on a spurious frequency shall be checked at the equipment output terminals when properly loaded with a suitable artificial antenna, Curves or equivalent shall show the magnitude of each harmonic and other spurious emission that can be detected when the equipment is operated with a carrier modulated under the conditions specified in paragraph (c) of this section for determining the band width occupied. The amplitude of spurious emissions which are more than 20 db below the permissible value need not be shown.
- (2) Field strength measurements of spurious radiations. A report of field strength measurements made to detect spurious emissions that may be radiated directly from the cabinet, control circuits, power leads, or intermediate circuit elements under normal conditions of installation and operation shall be made for the following equipments:
- (i) Those in which the spurious emissions are required to be 60 db or more below the carrier level.

(ii) All equipment operating on frequencies higher than 25 Mc/s.

- (iii) Hand carried transmitters, or others, where the antenna is an integral part of and attached directly to the transmitter.
- (iv) Other types of equipment as required, when in the opinion of the

- Commission, there is need for such measurement.
- (e) In all of the measurements set forth in paragraph (d) of this section, the spectrum should be investigated from the lowest radio frequency generated in the equipment up to at least the tenth harmonic of the carrier frequency or to the highest frequency possible in the present state of the art of measuring techniques. Particular attention should be paid to harmonics and subharmonics of the carrier frequency as well as to those frequencies removed from the carrier by multiples of the oscillator frequency. Radiation at the frequencies of multiplier stages should also be checked. The amplitude of spurious emissions which are more than 20 db below the permissible value need not be reported.
- (f) Frequency stability: The frequency stability of transmitting equipment shall be checked with variations in:
- (1) Temperature. Vary the ambient temperature from -30° to $+50^{\circ}$ Centigrade for non-broadcast equipment. Use suitable limits for broadcast equipment. Only the oscillator need be subjected to this test.
- (2) Primary supply voltage. Vary the primary supply voltage from 85 percent to 115 percent of the normal supply voltage at the input to the cable normally provided with the equipment, or at the power supply terminals if cables are not normally provided.

§ 2.582 Measurement procedure for type acceptance.

- (a) The Commission will accept data in full accordance with established standards and measurement procedures as published by engineering societies and associations such as the Electronic Industries Association, the Institute of Electrical and Electronics Engineers, Inc., and the American Standards Asso-Specific reference should be ciation. made to the standards used. In lieu of such standards, the applicant will submit a description of each measurement procedure together with a listing of the actual test equipment used. The Commission will accept measurements made by a qualified engineer where the procedures employed are based upon sound engineering principles.
- (b) For radio frequency equipment other than transmitters, appropriate

tests and performance requirements may be specified in other parts of this chapter.

§ 2.584 Identification and changes in type accepted equipment.

- (a) Each type of equipment for which type acceptance is requested shall be identified by a type number assigned by the manufacturer of the equipment. The type number shall consist of a series of not more than a total of seventeen digits, letters, punctuation marks, and spaces. The type number shall be shown on a name plate affixed in a conspicuous place to such equipment. All equipment used pursuant to type acceptance shall have affixed to it such indication of type acceptance as may be required in the service rules governing such equipment.
- (b) Equipment of the same type is defined for the purposes of type acceptance as being equipment which is electrically and mechanically interchangeable. In addition, transmitters of the same type will have the same basic tube line up, frequency multiplication, basic oscillator circuit, basic modulator circuit, and maximum rated power input.
- (c) Permissive changes may be made in type accepted equipment without prior Commission approval. There are two classifications of permissive changes, neither of which may involve change of type as defined in paragraph (b) of this section. The first classification includes those modifications in type accepted equipment which do not change the equipment characteristics beyond the rated limits established by the manufacturer and accepted by the Commission when obtaining type acceptance. The permissive second classification of changes includes those which bring the performance of the equipment outside the manufacturers' rated limits as originally filed but not below the minimum requirements of the applicable rules. The Commission shall be supplied with complete information and results of tests regarding this class of permissive changes, in accordance with § 2.571(b), prior to the operation of the modified equipment under an authorization of the Commission.
- (d) Changes in type accepted equipment, except permissive changes as set forth in paragraph (c) of this section, shall not be made except under prior authorization of the Commission. When such a change is requested, the Commission may authorize the change or require that the modified equipment be identi-

fied with a new type number and that additional information be submitted for further consideration as to type acceptance.

(e) If the assignment of a different type number is required as a result of equipment modification, a new name plate bearing the new type number shall be affixed to the modified equipment.

(f) Users shall not modify their own equipment except as provided in paragraphs (c) or (d) of this section, as applicable.

TECHNICAL INFORMATION FOR APPLICATION REFERENCE

§ 2.591 Submission of technical information for application reference.

- (a) Applications for station authorizations in some services require a detailed technical description of the equipment proposed to be used. In order to simplify the preparation and processing of applications by eliminating the need for submission of equipment specifications with each application, the Commission will accept for application reference purposes detailed technical specifications of equipment designed for use in these services. Manufacturers desiring to avail themselves of this procedure should submit all information required by the application forms and the rules for the services in which the equipment is to be used. Applications for station authorizations submitted subsequent to such filing may refer to the technical information so filed.
- (b) Receipt by the Commission of data for application reference purposes does not imply that the Commission has made or intends to make any finding regarding the acceptability of the equipment for licensing and such equipment will not be included on the list of equipment acceptable for licensing. Each applicant is expected to exercise appropriate care in the selection of equipment to insure that the unit selected will comply with the rules governing the service in which it is proposed to operate.

[28 F.R. 12465, Nov. 22, 1963, as amended at 29 F.R. 10586, July 30, 1964]

§ 2.593 Identification and changes in equipment information filed for application reference.

(a) Each type of equipment, for which information is filed for application reference purposes, shall be identified by a type number assigned by the

manufacturer of the equipment. The type number shall consist of a series of not more than a total of seventeen digits, letters, punctuation marks, and spaces. The type number shall be shown on a name plate affixed in a conspicuous place to such equipment.

(b) If the assignment of a different type number is required as a result of equipment modification, a new name plate bearing the new type number shall be affixed to the modified equipment.

Subpart G—Treaties and Other International Agreements

§ 2.601 General.

This subpart is corrected to July 1, 1969. The Commission does not distribute copies of these documents. Inquiry may be made to the U.S. Government Printing Office concerning availability for purchase.

[34 F.R. 11302, July 8, 1969]

§ 2.602 Citation abbreviations used in this subpart.

Trenwith—Treaties, Conventions, International Acts, Protocols, and Agreements between the United States of America and Other Powers, 1923–37 (compiled under S. Res. No. 132, 75th Cong., 1st sess.).

LNTS—League of Nations Treaty Series. Stat.—United States Statutes at Large. UST—United States Treaties and Other International Agreements.

TS-Treaty Series.

EAS—Executive Agreement Series.
TIAS—Treaties and Other International Acts

[32 F.R. 10181, July 11, 1967]

§ 2.603 Treaties and other international agreements relating to radio.

(a) The applicable treaties and other international agreements in force relating to radio and to which the United States of America is a party (other than reciprocal operating agreements for radio amateurs) are listed below:

Date	Citations	Subject
	IV Trenwith 4248, 4250 and 4251. TS 724-A.	US-UK (also for Canada and Newfoundland) Bilateral Arrangements providing for the Prevention of Interference by Ships off the Coasts of these Countries with Radio Broadcasting. Effected by exchange of notes Sept. and Oct., 1925. Entered into force Oct. 1, 1925.
		US-Canada Arrangement governing Radio Communications between Private Experimental Stations. Effected by exchange of notes at Washington Oct. 2 and Dec. 29, 1929, and Jan. 12, 1929. Entered into force Jan. 1, 1929. Continued by the arrangement contained in
	15 ///-A.	US-Canada (including Newfoundland) Arrangement relating to Assignment of High Frequencies on the North American Continent. Effected by exchange of notes at Ottawa Feb. 26 and 28, 1929. Entered into force Mar. 1, 1929. (Originally, Cuba was also a party to this arrangement, but by virtue of notice to the Canadian Govern-
		Os-Peru Arrangement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Lima Feb. 16, and May 23, 1934. Entered into force
	EAS 02.	US-Canada Arrangement relative to Radio Communications between Private Experimental Stations and between Amateur Stations. Continues the arrangement contained in TS 767-A. Effected by exchange of notes at Ottawa Apr. 23, and May 2 and 4, 1934. En-
		US-Chile Arrangement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Santiago Aug. 2 and 17, 1934. Entered into force Aug. 17,
	1 5 938.	Inter-American Radio Communications Convention between the United States and Other Powers. Signed at Havana Dec. 13, 1937. (First Inter-American Radio Conference.) Entered into force for the United States July 21, 1938, for Parts I, III and IV; Apr. 17, 1939, for Part II.
	10 313.	Office) terminated for all parties Dec. 20, 1958 (TIAS 4079). Regional Radio Convention between the United States (in behalf of the Canal Zone) and Other Powers. Signed at Guatemala City
	LAS 140.	Dec. 8, 1938. Entered into force Oct. 8, 1939. US-Canada Arrangement governing the Use of Radio for Civil Aeronautical Services. Effected by exchange of notes at Washington Feb. 20, 1939. Entered into force Feb. 20, 1939.
	TIAS 1527.	type Communication Channels. Signed at Moscow May 24, 1946.
1947	61 Stat. (4) 3800	US-Canada Agreement providing for Frequency Modulation Broad- casting in Channels in the Radio Frequency Band 88-108 Mc/s. Effected by exchange of notes at Washington Jan. 8 and Oct. 15, 1947. Entered into force Oct. 15, 1947.

Date	Citations	Subject
947	61 Stat. (4) 3416 TIAS 1676.	US-UN Agreement relative to Headquarters of the United Nations. Signed at Lake Success June 26, 1947. Entered into force Nov. 21, 1947. Supplemented by the agreement contained in TIAS 5961 which was signed Feb. 9, 1966. Os-UK Agreement regarding Standardization of Distance Measuring Equipment. Signed at Washington Oct. 13, 1947. Entered into force Oct. 13, 1947.
.947	61 Stat. (3) 3131	US-UK Agreement regarding Standardization of Distance Measuring Equipment. Signed at Washington Oct. 13, 1947. Entered into force Oct. 13, 1947.
948	9 UST 621 TIAS 4044.	Equipment. Signed at washington Oct. 13, 1947. Entered into force Oct. 13, 1947. Entered into force Oct. 13, 1947. Entered into force Mar. 17, 1958. Modified by the amendments contained in TLAS 6285 and in TLAS 6490 adopted by the IMCO Assembly Sept. 15, 1964, and Sept. 28, 1965, respectively. Inter-American Radio Agreement between the United States and Canada and Other American Republics. Signed at Washington July 9, 1949. (Fourth Inter-American Radio Conference.) Entered
949	3 UST (3) 3064 TIAS 2489.	Inter-American Radio Agreement between the United States and Canada and Other American Republics. Signed at Washington July 9, 1949. (Fourth Inter-American Radio Conference.) Entered into force Apr. 13, 1952, subject to the provisions of Article 13.
		July 9, 1949. (Fourth Inter-American Radio Conference.) Entered into force Apr. 13, 1952, subject to the provisions of Article 13. London Telecommunications Agreement between the United States and Certain British Commonwealth Governments. Signed at London Aug. 12, 1949. Entered into force Feb. 24, 1950. Amended by the agreement contained in TIAS 2705 which was signed Oct. 1, 1952.
		US-Ecuador Arrangement regarding Radio Communications be- tween Amsteur Stations on Behalf of Third Parties. Effected by exchange of notes at Quito Mar. 16 and 17, 1950. Entered into force Mar. 17, 1950.
		US-Liberia Arrangement regarding Radio Communications be- tween Amateur Stations on Behalf of Third Parties. Effected by
1950	11 UST 413 TIAS 4460.	1951. Entered into force Jan. 11, 1951. North American Regional Broadcasting Agreement (NARBA). Signed at Washington Nov. 15, 1950. Entered into force Apr. 19, 1960. Effective between United States, Canada, Cuba, Dominican Republic, and the United Kingdom of Great Britain and Northern Ireland for the Bahama Islands. Ratification on behalf of Jamaica pending.
951	3 UST (3) 3787 TIAS 2508.	Ireland for the Bahama Islands. Ratification on behalf of Jamaica pending. US-Canada Convention relating to the Operation by Citizens of Either Country of Certain Radio Equipment or Stations in the Other Country. Signed at Ottawa Feb. 8, 1951. Entered into force May 15, 1952.
1951 and 1952	3 UST (3) 3892 TLAS 2520.	US-Cuba Arrangement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Havana Sept. 17, 1961, and Feb. 27, 1962. Entered into
1951	3 UST (2) 2860 TIAS 2459.	
1952	3 UST (4) 4926 TIAS 2666.	US-Cuba Agreement concerning the Control of Electromagnetic Radiation. Effected by exchange of notes at Havana Dec. 10 and 18, 1951. Entered into force Dec. 18, 1951. US-Canada Agreement for the Promotion of Safety on the Great Lakes by Means of Radio. The agreement applies to vessels of all countries as provided for in Article 3. Signed at Ottawa Feb. 21, 1952. Entered into force Nov. 13, 1954. US-Canada Agreement relating to the Assignment of Television Frequency Channels along United States-Canadian Border. Effected by exchange of notes at Ottawa Apr. 23 and June 23, 1952. Entered into force June 23, 1952. London Revision (1952) of the London Telecommunications Agreement (1949) between the United States and Certain British Commonwealth Governments, Signed at London Oct. 1, 1952. Entered
1952	3 UST (3) 4443TIAS 2594.	US-Canada Agreement relating to the Assignment of Television Frequency Channels along United States-Canadian Border. Effected by exchange of notes at Ottawa Apr. 23 and June 23, 1952. Entered into force June 23, 1952.
		into force Oct. 1, 1952. This amends the agreement contained in
1953	5 UST (3) 2840 TIAS 3138.	
		Transmitting Equipment. Effected by exchange of notes at Washington Mar. 9 and 17, 1953. Entered into force Mar. 17, 1953. US-Panama Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Panama July 19 and Aug. 1, 1956. Entered into force Sept. 1, 1956.
1956		US-Costa Rica Agreement regarding Radio Communications be- tween Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Washington Aug. 13 and Oct. 19, 1956. Entered into force Oct. 19, 1956.
1956	7 UST 3159 TIAS 3694.	US-Nicaragua Agreement regarding Radio Communications be- tween Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Managua Oct. 8 and 16, 1956. Entered into force Oct. 16, 1956
		US-Mexico Agreement regarding Radio Broadcasting in the Standard Broadcast Band. Signed at Mexico Jan. 29, 1957. Entered into force June 9, 1961. Amended by the protocol contained in TIAS 6210 signed Apr. 13, 1966.
1957	9 UST 1037 TIAS 4079.	Multilateral Declaration between the United States and other Powers terminating Part II (Inter-American Radio Office) of the Inter-American Radio Communications Convention of Dec. 13, 1937 (TS-933). Signed at Washington Dec. 20, 1957. Entered into force Dec. 20, 1957. Additionally, a Contract on the Exchange of Notifications of Radio Broadcasting Frequencies between the Pan American Union, the United States and Other Powers was signed at Washington Dec. 20, 1957. Entered into force Jan. 1, 1958.

Date	Citations	Subject
	TIAS 4089.	US-Mexico Agreement regarding Allocation of Ultra High Frequency Channels to Land Border Television Stations. Effected by exchange of notes at Mexico July 16, 1958. Entered into force July 16, 1958.
1958	10 UST 2423 TIAS 4390.	Telegraph Regulations (Geneva Revision, 1958) Annexed to the International Telecommunication Convention, Signed at Geneva
1959	10 UST 1449TIAS 4295.	Nov. 29, 1958. Entered into force Jan. 1, 1960. US-Mexico Arrangement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Mexico July 31, 1959. Entered into force Aug. 30, 1959.
1999 8ud 1900	TIAS 4442.	Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Tegucigalpa Oct. 26, 1959, and Feb. 17, 1960. and related
		US-Venezuela Arrangement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Caracas Nov. 12, 1959. Entered into force
	TIAS 4893.	International Radio Regulations Annexed to the International Tele- communication Convention. Signed at Geneva Dec. 21, 1959. Entered into force with respect to the United States Oct. 23, 1961. Revised by the Partial Revisions of the Radio Regulations, Geneva 1959, contained in TIAS 5603, TIAS 6332, and TIAS 6590 signed Nov. 8, 1963, Apr. 20, 1968 and Nov. 3, 1967, respectively.
		Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Port-au-Prince Jan. 4 and 6, 1960. Entered into force Feb.
	TIAS 5780.	International Convention for the Safety of Life at Sea and Annexed Regulations. Signed at London June 17, 1960. Entered into force May 26, 1965. Corrections to certain annexes contained in TIAS 6284 strand Reb. 15, 1966.
	TIAS 4596.	US-Paraguay Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Asuncion Aug. 31, and Oct. 6, 1960. Entered into force
	TIAS 6115.	US-Uruguay Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Montevideo Sept. 12, 1961. Entered into force Sept. 26,
1961	12 UST 1695 TIAS 4888.	US-Bolivia Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange
1962	13 UST 411TIAS 5001.	US-El Salvador Arrangement regarding Radio Communications between Amateur Stations on Behalf of Third Parties, Effected by
		exchange of notes at San Salvador Apr. 5, 1962. Entered into force May 5, 1962. US-Mexico Agreement relating to the Assignment of VHF Television Channels along United States-Mexican Border. Effected by exchange of notes at Mexico Apr. 18, 1962. Entered into force Apr. 18, 1962.
		1962. US-Canada Agreement relating to the Coordination and Use of Radio Frequencies above 30 Mc/s. Effected by exchange of notes at Ottawa Oct. 24, 1962. Entered into force Oct. 24, 1962. The tech- nical annex to this agreement was revised by the agreement con- tained in TIAS 5833 signed June 16 and 24, 1965.
		tained in TIAS 8833 signed June 16 and 24, 1965. US-Dominican Republic Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Santo Domingo Apr. 18 and 22, 1963. Entered into force May 22, 1963.
		Partial Revision of the Radio Regulations, Geneva, 1959, Final Acts of the EARC to Allocate Frequency Bands for Space Radiocom- munication Purposes. Signed at Geneva Nov. 8, 1963. Entered into
		US-Colombia Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Bogota Nov. 16 and 29, 1963. Entered into force Dec. 29.
		US-Other Governments Agreement Establishing Interim Arrangements for a Global Commercial Communications Satellite System and Special Agreement. Done at Washington Aug. 20, 1964. Entered into force Aug. 20, 1964. Additionally, a Supplementary Agreement on Arbitration was done at Washington June 4, 1965. Entered into force Nov. 21, 1966.
	TIAS 6285.	Amendments to Articles 17 and 18 of the IMCO Convention (TIAS 4044). Adopted by the IMCO Assembly at London Sept. 15, 1964.
		US-Brazil Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Washington Luna 1 1965. Entered into force June 1 1965
1965	. 16 UST 923 TIAS 5833.	US-Canada Agreement regarding Coordination and Use of Radio Frequencies above 30 Mc/s Revising the Technical Annex to the Agreement of Oct. 24, 1962 (TIAS 5205). Effected by exchange of notes at Ottawa June 16 and 24, 1965. Entered into force June 24, 1965.

Date	Citations	Subject
1965	16 UST 883TIAS 5827.	US-Israel Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Washington July 7, 1965. Entered into force Aug. 6, 1965. Amendment to Article 28 of the IMCO Convention (TIAS 4044). Adopted by the IMCO Assembly at Paris Sept. 28, 1965. Entered into force Nov. 3, 1968.
1965	TIAS 6490.	Amendment to Article 28 of the IMCO Convention (TIAS 4044). Adopted by the IMCO Assembly at Paris Sept. 28, 1965. Entered into force Nov. 3, 1968.
	- 18 UST 575 TIAS 6267.	International Telecommunication Convention. Signed at Montreux Nov. 12, 1965. Entered into force with respect to the United States
	TIAS 5961.	US-UN Agreement regarding Headquarters of the United Nations Supplementing the Agreement of June 28, 1947 (TIAS 1676). Signed at New York Feb. 9, 1966. Entered into force Feb. 9, 1966. Amended
	TIAS 6284.	Proces-Verbal of Rectification to Certain Annexes to the International Convention for the Safety of Life at Sea of June 17, 1960 (TIAS 5780). Done at London Feb. 15, 1966.
1966	18 UST 141 TIAS 6210.	US-Mexico Protocol regarding Radio Broadcasting in the Standard Broadcast Band Amending the Agreement of Jan. 29, 1957 (TIAS 4777). Signed at Mexico Apr. 13, 1966. Entered into Force Jan. 12, 1967.
1966	- 18 UST 2091 T1AS 6332.	Partial Revision of the Radio Regulations, Geneva, 1959, Final Acts of the EARC for the Preparation of a Revised Allotment Plan for the Aeronautical Mobile (R) Service. Signed at Geneva Apr. 29, 1966. Entered into force for the United States Aug. 23, 1967, except for the frequency allotment plan contained in Appendix 27 which shall enter into force Apr. 10, 1970.
1966	17 UST 2319 TIAS 6176.	US-UN Agreement regarding Headquarters of the United Nations Amending the Supplemental Agreement of Feb. 9, 1966 (TIAS 5961). Effected by exchange of notes at New York Dec. 8, 1966. Entered into force Dec. 8, 1966.
1967	- 18 UST 365 TIAS 6244	US-Argentina Agreement regarding Radio Communications between Amateur Stations on Behalf of Third Parties. Effected by exchange of notes at Buenos Aires Mar. 31, 1967. Entered into force Apr. 30, 1967.
	TIAS 6268.	US. Canada Agreement relating to Pre-Sunrise Operation of Certain Standard (AM) Radio Broadcasting Stations. Effected by exchange of notes at Ottawa Mar. 31 and June 12, 1967. Entered into force June 12, 1967. Amended by the agreement contained in TIAS 6626 signed Apr. 18, 1968, and Jun. 31, 1969.
1967	TIAB 0590.	Partial Revision of the Radio Regulations, 1959, Final Acts of the WARC to deal with Matters relating to the Maritime Mobile Service. Signed at Geneva Nov. 3, 1967. Entered into force Apr. 1, 1969.
1968 and 1969	TIAS 6626.	US-Canada Agreement relating to Pre-Sunrise Operation of Certain Standard (AM) Radio Broadcasting Stations Amending the Agree- ment of Mar. 31 and June 12, 1967 (TIAS 6268). Effected by exchange of notes at Ottawa Apr. 18, 1968, and Jan. 31, 1969. Entered into force Jan. 31, 1969.

(b) The applicable agreements in force between the United States and another country relating to the reciprocal granting of authorizations to permit licensed amateur radio operators of either country to operate their stations in the other country are as follows:

Date	Citations	Subject
1964	15 UST 1787 TIAS 5649.	US-Costa Rica Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at San Jose Aug. 17 and 24, 1964. Entered into force Aug. 24, 1964.
1965	16 UST 93 TIAS 5766.	US-Dominican Republic Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Santo Domingo Jan. 28 and Feb. 2, 1965. Entered into force Feb. 2, 1965.
1965	16 UST 165 TIAS 5777.	US-Bolivia Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at La Paz Mar. 16, 1965. Entered into force Apr. 15, 1965.
1965	- 16 UST 181 TIAS 5779.	US-Ecuador Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Quito Mar. 26, 1965. Entered into force Mar. 26, 1965.
1965	- 16 UST 817 TIAS 5815.	US-Portugal Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Lisbon May 17 and 26, 1965. Entered into force May 26, 1965.
1965	16 UST 869 TIAS 5824.	US-Belgium Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Brussels June 15 and 18, 1965. Entered into force June 18, 1965.
1965	16 UST 973 TIAS 5836.	US-Australia Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Canberra June 25, 1965. Entered into force June 25, 1965.
1965	16 UST 1160 TIAS 5860.	US-Feru Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Lima June 28 and Aug. 11, 1965. Entered into force Aug. 11, 1965.
1965	16 UST 1746 TIAS 5900.	US-Luxembourg Agreement regarding Alien Amateur Radio Oper- ators. Effected by exchange of notes at Luxembourg July 7 and 29, 1965. Entered into force July 29, 1965.

Date	Citations	Subject
	TIAS 5856.	US-Sierra Leone Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Freetown Aug. 14 and 16. 1965. Entered into force Aug. 16, 1965.
	TIAS 5899.	US-Colombia Agreement regarding Alien Amateur Radio Operators. Effected by sechange of notes at Bogota Oct. 19 and 28, 1965. En-
	TIAS 5941.	US-UK Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at London Nov. 26, 1965. Entered into force Nov. 26, 1965.
	TIAS 5978.	US-Paraguay Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Asuncion Mar. 18, 1966. Entered
1966	17 UST 719 TIAS 6022.	US-France Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Paris May 5, 1966, with related
		US-India Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at New Delhi May 16 and 25, 1966.
	TIAS 6028.	US-Israel Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Washington June 15, 1966. Entered into force June 15, 1966.
	TIAS 6189.	US-Netherlands Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at The Hague June 22, 1966.
1966	17 UST 1120 TIAS 6068.	US-Federal Republic of Germany Arrangement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Bonn Line 22 and 20, 1968. Entered into force Line 20, 1968.
	TIAS 6061.	US-Kuwait Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Kuwait July 19 and 24, 1966. Entered
	TIAS 6112.	US-Nicaragua Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Managua Sept. 3 and 20, 1966. Entered into force Sept. 20, 1966.
	TIAS 6159.	US-Panama Agreement regarding Alien Amateur Radio Operators. Effected by exchange of note at Panama Nov. 16, 1966. Entered into
1966 and 1967	18 UST 525 TIAS 6259.	US-Honduras Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Tegucigalpa Dec. 29, 1966, Jan. 24
1967	18 UST 554 TIAS 6264.	and Apr. 17, 1967. Entered into force Apr. 17, 1967. US-Switzerland Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Bern Jan. 12 and May 16, 1967. Entered into force May 16, 1967. US-Trinidad and Tobago Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at St. Ann's and Port of Spain Jan. 14 and Mar. 16, 1967. Entered into force Mar. 16, 1967. US-Argentina Agreement regarding Alien Amateur Radio Operators.
1967	18 UST 543TIAS 6261.	US-Trinidad and Tobago Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at St. Ann's and Port of Spain Jan. 14 and Mar. 16, 1967. Entered into force Mar. 16, 1967.
	TIAS 6243.	Effected by exchange of notes at Buenos Aires Mar. 31, 1967. Entered
1967	TIAS 6309.	into force Apr. 30, 1967. US-El Salvador Agreement regarding Alien Amateur Radio Opera- tors. Effected by exchange of notes at San Salvador May 24 and June 5, 1967. Entered into force June 5, 1967.
	TIAS 6273.	US-Norway Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Oslo May 27 and June 1, 1967. Entered into force June 1, 1967.
	TLAS 6281.	US-New Zealand Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Wellington June 21, 1967. Entered into force June 21, 1967.
	TIAS 6348.	US-Venezuela Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Caracas Sept. 18, 1967. Entered into force Oct. 3, 1967.
1967	TIAS 6378.	Done at Vienna Nov. 21, 1967. Entered into force Dec. 21, 1967.
1967	18 UST 2882 TIAS 6380.	into force Oct. 3, 1967. US-Austria Agreement regarding Alien Amateur Radio Operators. Done at Vienna Nov. 21, 1967. Entered into force Dec. 21, 1967. US-Chile Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Washington Nov. 30, 1967. Entered into force Dec. 30, 1967. US-Finland Agreement regarding Alien Amateur Radio Operators.
1967	. 18 UST 3153 TIAS 6406.	tered into force Dec. 30, 1967. US-Finland Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Helsinki Dec. 15 and 27, 1967. Entered into force Dec. 27, 1967. US-Monaco Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Nice and Paris Mar. 29, and Oct. 16, 1968. Extered into force Dec. 1, 1968.
	TIAS 6622.	
1968	TIAS 6494.	US-Guyana Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Georgetown May 6 and 13, 1968. Entered into force May 13, 1968.
1968	TIAS 6553.	Entered into force May 13, 1968. US-Barbados Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Bridgetown Sept. 10 and 12, 1968. Entered into force Sept. 12, 1968.
1968	TIAS 6566.	Entered into force Sept. 12, 1968. US-Ireland Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Dublin Oct. 10, 1968. Entered into force Oct. 10, 1968.
1968	TIAS 6654.	Effected by exchange of notes at Djakarta Dec. 10, 1968. Entered
1969		US-Sweden Agreement regarding Alien Amateur Radio Operators. Effected by exchange of notes at Stockholm May 27, and June 2, 1969. Entered into force June 2, 1969.



(c) With respect to its relations with several countries, the United States is bound by certain superseded treaties and agreements because some of the contracting countries other than the United States did not become a party to subsequent treaties and agreements. These include the following:

Date	Citations	Subject
1912	38 Stat. 1672 TS 581.	International Radiotelegraph Convention. Signed at London July 5, 1912. Entered into force July 1, 1913. Superseded by the International Radiotelegraph Convention and General Regulations, Washington, 1927 (TS 767).
1927	45 Stat. 2760 TS 767.	International Radiotelegraph Convention and General Regulations. Signed at Washington Nov. 25, 1927. Entered into force Jan. 1, 1929. Superseded by the International Telecommunication Convention and General Radio Regulations, Madrid, 1932 (TS 867).
1932	49 Stat. 2391 TS 867.	International Telecommunication Convention. Signed at Madrid Dec. 9, 1932. Entered into force for the United States June 12, 1934. Superseded by the International Telecommunication Convention, Atlantic City, 1947 (TIAS 1901).
1937	54 Stat. 2514 EAS 200.	Inter-American Arrangement concerning Radiocommunications and Annex. Signed at Havana Dec. 13, 1937. (First Inter-American Radio Conference.) Entered into force for the United States July 18, 1938. This arrangement was replaced by the Inter-American Agreement concerning Radiocommunications, Santiago, 1940 (EAS 231).
1938	54 Stat. 1417 TS 948.	General Radio Regulations (Cairo Revision, 1938) Annexed to the International Telecommunication Convention, Madrid, 1932. Signed at Cairo Apr. 8, 1938. Entered into force Sept. 1, 1939. Superseded by the Radio Regulations, Atlantic City, 1947 (TIAS 1901).
1940	55 Stat. 1482 EAS 231.	Inter-American Radiocommunications Agreement between the United States, Canada and Other American Republics. Signed at Santiago, Jan. 26, 1940. (Second Inter-American Radio Conference.) Entered into force with respect to the United States Feb. 25, 1942. Replaced by the Inter-American Radio Agreement, Washington, 1949 (TIAS 2489).
1947	63 Stat. (2) 1399 TIAS 1901.	International Telecommunication Convention and Radio Regulations. Signed at Atlantic City Oct. 2, 1947. Entered into force Jan. 1, 1949. The Convention was superseded by the International Telecommunication Convention, Buenos Aires, 1952 (TIAS 3266). The Radio Regulations were superseded by the International Radio Regulations, Geneva, 1959 (TIAS 4993).
1949	2 UST (1) 17TIAS 2175.	Telegraph Regulations (Paris Revision, 1949) Annexed to the International Telecommunication Convention. Signed at Paris Aug. 5, 1949. Entered into force with respect to the United States Sept. 26, 1950. Superseded by the Telegraph Regulations, Geneva Revision, 1958 (TIAS 4390).
1952	6 UST 1213 TIAS 3266.	International Telecommunication Convention. Signed at Buenos Aires Dec. 22, 1952. Entered into force with respect to the United States June 27, 1955. Superseded by the International Telecommunication Convention, Geneva, 1959 (TIAS 4892).
1959	12 UST 1761TIAS 4892.	International Telecommunication Convention. Signed at Geneva Dec. 21, 1959. Entered into force with respect to the United States Oct. 23, 1961. Superseded by the International Telecommunication Convention, Montreux, 1965.

(d) There are certain treaties and agreements primarily concerned with matters other than the use of radio but which affect the work of the Federal Communications Commission insofar as they involve communications. Among the most important of these are the following which are available from the Secretary General, International Civil Aviation Organization (ICAO), International Aviation Building, 1080 University Street, Montreal, Canada:

Date	Citations	Subject
1944	61 Stat. (2) 1180TIAS 1591.	International Civil Aviation Convention. Signed at Chicago Dec. 7, 1944. Entered into force Apr. 4, 1947. Amended by the protocols contained in TIAS 3756 and TIAS 5170. ICAO Communications Division. Second Session. Montreal.
1949		ICAO Communications Division, Third Session, Montreal.
1951		ICAO Communications Division, Fourth Session, Montreal. ICAO Communications Division, Fifth Session, Montreal.
1954	8 UST 179 TIAS 3756.	Protocol Amending the International Civil Aviation Convention (TIAS 1591). Done at Montreal June 14, 1954. Entered into force Dec. 12, 1956.
1957 1958 1961	13 UST 2105TIAS 5170.	ICAO Communications Division, Sixth Session, Montreal. ICAO Communications Division, Special Session, Montreal. Protocol Amending the International Civil Aviation Convention (TIAS 1591). Done at Montreal June 21, 1961. Entered into force July 17, 1962.
1962 1963 1966		ICAO Communications Division, Seventh Session, Montreal. ICAO Communications Division, Special Session, Montreal. ICAO Communications/Operations (COM/OPS) Divisional Meeting, Montreal.

[32 F.R. 10181, July 11, 1967, as amended at 33 F.R. 10090, July 13, 1968; 34 F.R. 11302, July 8, 1969]

Digitized by Google

Sec.

5.58

Subpart H—Prohibition Against Eavesdropping

Prohibition against use of a § 2.701 radio device for eavesdropping.

- (a) No person shall use, either directly or indirectly, a device required to be licensed by section 301 of the Communications Act of 1934, as amended, for the purpose of overhearing or recording the private conversations of others unless such use is authorized by all of the parties engaging in the conversation.
- (b) Paragraph (a) of this section shall not apply to operations of any law enforcement officers conducted under lawful authority.

[31 F.R. 3400, Mar. 4, 1966]

PART 5-EXPERIMENTAL RADIO SERVICES (OTHER THAN BROAD-CAST)

Subpart A-General

5.1	Basis and purpose.
5.2	Services available.
5.3	Definition of terms.
5.4	General citizenship restrictions.
5.5	Transfer and assignment of station authorization.

Subpart B-Applications and Licenses

5.51	Station authorization required.
5.52	Procedure for obtaining a radio sta-
	tion license.
5.53	Filing of applications.
5.54	Who may sign applications.
5.55	Forms to be used.
5.56	Procedure for obtaining a special temporary authorization.
5.57	Supplementary statements required.

Partial grants. Defective applications. 5.59

Amendment or dismissal of applica-5.60 tions.

5.61 Construction period.

Licenses required for separate experi-5.62 mental projects.

5.63 License period.

5.64 Change in equipment.

5.65 Operation at a temporary location.

5.66 Discontinuance of station operation. 5.67 Policy governing the assignment of

frequencies. 5.68 Cancellation provisions.

Notification to the National Radio Astronomy Observatory. 5.69

Subpart C—Technical Standards

5.101	Frequency stability.
5 102	Tymes of emission

5.103 Emission limitations.

5.104 Modulation requirements.

5.105 Power and antenna height.

5.106 Transmitter control requirements.

5.107 Transmitter measurements.

Subpart D-Operating Requirements

Sec. 5.151 General limitations on use.

5.152 Station identification.

5.153 Suspension of transmission required.

5.154 Mobile installations in vehicles not under the continuous control of the licensee.

5.155 Operator requirements.

5.156 Evidence of operator license.

5.157 Posting station licenses and transmitter identification cards or plates. 5.158

Authorized points of communication. 5.159 Operation during an emergency.

5.160 Inspection of stations.

5.161 Inspection and maintenance of tower marking and lighting and associated

control equipment. Notice of violation. 5.162

5.163 Content of station records.

5.164 Form of station records.

5.165 Retention of station records.

5.166 Adherence to program of research.

Subpart E-Experimental Service (Research)

5.201 Eligibility for license. 5.202

Scope of service. 5.203 Frequencies for Experimental Service (Research).

5.204 Experimental report.

Subpart F---Experimental Service (Developmental)

5.251 Eligibility for license.

5.252 Scope of service.

5.253 Frequencies for Experimental Serv-

ice (Developmental).

5.254 Frequencies for field strength surveys or equipment demonstrations.

Experimental report. 5.255

Subpart G--- [Reserved]

Subpart H-Student Authorizations

5.401 Eligibility for license.

5.402 Filing of applications. 5.403 Waiver of construction permit.

5.404 Waiver of operator requirement.

Power limitation. 5.405

5.406 Frequencies.

5.407 Measurements required.

5.408 Waiver of technical and operating requirements.

5.409 Non-interference condition.

5.410 Record of operation.

5.411 Notification.

AUTHORITY: The provisions of this Part 5 issued under secs. 4, 303, 48 Stat. 1066, 1082, as amended; 47 U.S.C. 154, 303. Interpret or apply sec. 301, 48 Stat. 1081, as amended; 47 **v**.ŝ.č. 301.

Source: The provisions of this Part 5 appear at 28 F.R. 12506, Nov. 22, 1963, unless otherwise noted.

Subpart A-General

§ 5.1 Basis and purpose.

(a) The rules following in this part are promulgated pursuant to the provisions of Title III of the Communications Act of 1934, as amended, which vests authority in the Federal Communications Commission to regulate radio transmissions and to issue licenses for radio stations.

(b) The purpose of this part is to prescribe the manner in which parts of the radio frequency spectrum may be made available for experimentation as defined in this part and provided for.

§ 5.2 Services available.

- (a) Experimental services are classified, according to the type of experimentations permitted, into two categories as follows:
- (1) Experimental Service (Research)
 —Subpart E.
- (2) Experimental Service (Developmental)—Subpart F.

§ 5.3 Definition of terms.

For the purpose of this part, the following definitions shall be applicable. For other definitions, refer to Part 2 of this chapter (Frequency Allocations and Radio Treaty Matters; General Rules and Regulations).

- (a) Authorized frequency. The frequency assigned to a station by the Commission and specified in the instrument of authorization.
- (b) Authorized power. The power assigned to a radio station by the Commission and specified in the instrument of authorization. The authorized power does not necessarily correspond to the power used by the Commission for purposes of its Master Frequency Record (MFR) and notification to the International Telecommunication Union.
- (c) Experimental Service. A service in which Hertzian waves are employed for purposes of experimentation in the radio art or for purposes of providing essential communications for research projects which could not be conducted without the benefit of such communications.
- (d) Experimental Service (Research). An Experimental Service (1) for research in the radio art not related to the development of an established or proposed new service, or (2) for providing essential communications for research projects which could not be conducted without the benefit of such communications.
- (e) Experimental Service (Developmental). An Experimental Radio Service for the development of equipment, engineering or operational data, or tech-

- niques for an existing or proposed radio service.
- (f) Fixed service. A service of radiocommunication between specified fixed points.
- (g) Fixed station. A station in the fixed service.
- (h) Harmful interference. Any radiation or any induction which endangers the functioning of a radionavigation service or of a safety service or obstructs or repeatedly interrupts a radio service operating in accordance with the Table of Frequency Allocations and other provisions of Part 2 of this chapter.
- (i) Landing area. As defined by Title I, section I (22) of the Civil Aeronautics Act of 1938, as amended, landing area means any locality, either of land or water, including airdromes and intermediate landing fields, which is used, or intended to be used, for the landing and take-off of aircraft, whether or not facilities are provided for the shelter, servicing, or repair of aircraft, or for receiving or discharging passengers or cargo.
- (j) Land station. A station in the mobile service not intended for operation while in motion.
- (k) Mobile service. A service of radiocommunication between mobile and land stations, or between mobile stations.
- (1) Mobile station. A station in a mobile service intended to be used while in motion or during halts at unspecified points.
- (m) Mean power of radio transmitter. The power supplied to the antenna during normal operation, averaged over a time sufficiently long compared to the period corresponding to the lowest frequency encountered in actual modulation.
- (n) Peak power of a radio transmitter. The mean power supplied to the antenna during one radio frequency cycle at the highest crest of the modulation envelope, taken under conditions of normal operation.
- (o) Person. An individual, partnership, association, joint stock company, trust, or corporation.
- (p) Public correspondence. Any telecommunication which the offices and stations, by reason of their being at the disposal of the public, must accept for transmission.
- (q) Radio service. An administrative subdivision of the field of radio-communication. In an engineering sense, the subdivisions may be made according to the method of operation, as, for example,

mobile service and fixed service. In a regulatory sense, the subdivisions may be descriptive of particular groups of licensees, as, for example, the groups of persons licensed under this part.

(r) Station authorization. Any construction permit, license, or special temporary authorization issued by the Commission.

§ 5.4 General citizenship restrictions.

A station license may not be granted to or held by:

- (a) Any alien or the representative of any alien.
- (b) Any foreign government or the representative thereof.
- (c) Any corporation organized under the laws of any foreign government.
- (d) Any corporation of which more than one-fifth of the capital stock is owned of record or voted by: Aliens or their representatives; a foreign government or representative thereof; or any corporation organized under the laws of a foreign country.
- (e) Any corporation of which any officer or director is an alien.
- (f) Any corporation directly or indirectly controlled by any other corporation of which any officer or more than one-fourth of the directors are aliens, if the Commission finds that the public interest will be served by the refusal or revocation of such license; or
- (g) Any corporation directly or indirectly controlled by any other corporation of which more than one-fourth of the capital stock is owned of record or voted by: Aliens or their representative; a foreign government or representative; thereof; or any corporation organized under the laws of a foreign country, if the Commission finds that the public interest will be served by the refusal or revocation of such license.

§ 5.5 Transfer and assignment of station authorization.

A station authorization. the frequencies authorized to be used by the grantee of such authorization, and the rights therein granted by such authorization shall not be transferred, assigned, or in any manner either voluntarily or involuntarily disposed of, or indirectly by transfer of control of any corporation holding such authorization, to any person, unless the Commission shall, after securing full information. decide that said transfer is in the public interest, and shall give its consent in writing. Requests for authority to transfer or assign a station authorization shall be submitted on the forms prescribed by § 5.55.

Subpart B—Applications and Licenses

§ 5.51 Station authorization required.

No radio transmitter shall be operated in the Experimental Radio Services except under and in accordance with a proper station authorization granted by the Federal Communications Commission.

§ 5.52 Procedure for obtaining a radio station license.

- (a) The first step toward obtaining a station license is the filing of of an application for a construction permit in accordance with this part. After the construction and installation are completed, an application for station license may be submitted in accordance with § 5.55.
- (b) In the case of complete transmitters which are to be used without modification and the installation of which requires only the interconnection of the transmitter units, the application for license may be submitted simultaneously with the application for construction permit.

(c) When the design and construction of the transmitting equipment is an integral part of the experimental program, the application for license may be filed simultaneously with the application for construction permit.

[28 F.R. 12506, Nov. 22, 1963, as amended at 34 F.R. 3802, Mar. 5, 1969]

Filing of applications.

- (a) To assure that necessary formation is supplied in a consistent manner by all persons, standard forms are prescribed for use in connection with the majority of applications and reports submitted for Commission consideration. Standard numbered forms applicable to the Experimental Services are discussed in § 5.55 and may obtained from the Washington, D.C., Office of the Commission, or any of its engineering field Concerning matters where no offices. standard form is applicable, the informal application procedure outlined in § 5.55 (k) should be followed.
- (b) Any application for radio station authorization and all correspondence relating thereto shall be submitted to the Commisson's office at Washington, D. C.
- (c) Unless otherwise specified, an application shall be filed at least sixty days

prior to the date on which it is desired that Commission action thereon be completed.

- (d) Each application for station authorization shall be specific and complete with regard to station location, proposed equipment, power, antenna height, and operating frequency; and other information required by the application form and this part.
- (e) Applications involving operation at temporary locations:
- (1) When a land station or a fixed station is to remain at a single location for less than six months, the location is considered to be temporary and the procedure outlined in § 5.65 shall apply.
- (2) When a land station or fixed station authorized to operate at temporary locations remains at a single location for more than six months, an application for modification of the station authorization to specify the permanent location shall be filed within thirty days after expiration of the six-month period.
- (f) Unless otherwise specified in a particular case or for a particular form, each application shall be filed in duplicate.

§ 5.54 Who may sign applications.

- (a) Except as provided in paragraph (b) of this section, applications, amendments thereto, and related statements of fact required by the Commission shall be personally signed by the applicant, if the applicant is an individual; by one of the partners, if the applicant is a partnership; by an officer or duly authorized employee, if the applicant is a corporation; or by a member who is an officer, if the applicant is an unincorporated association. Applications, amendments, and related statements of fact filed on behalf of eligible government entities, such as states and territories of the United States and political subdivisions thereof, the District of Columbia, and units of local government, including incorporated municipalities, shall be signed by such duly elected or appointed officials as may be competent to do so under the laws of the applicable jurisdiction.
- (b) Applications, amendments thereto, and related statements of fact required by the Commission may be signed by the applicant's attorney in case of the applicant's physical disability or of his absence from the United States. The attorney shall in that event separately set forth the reason why the application is

not signed by the applicant. In addition, if any matter is stated on the basis of the attorney's belief only (rather than his knowledge), he shall separately set forth his reasons for believing that such statements are true.

(c) Only the original of applications, amendments, or related statements of fact need be signed; copies may be conformed.

(d) Applications, amendments, and related statements of fact need not be submitted under oath. Willful false statements made therein, however, are punishable by fine and imprisonment, U.S. Code, Title 18, section 1001, and by appropriate administrative sanctions, including revocation of station license pursuant to section 312(a) (1) of the Communications Act of 1934, as amended.

§ 5.55 Forms to be used.

- (a) Application for construction permit for land stations and fixed stations. A separate application for construction permit shall be submitted on Form 440 for each base station and each fixed station. Such applications, with respect to proposed antenna structures, must comply with the requirements of Part 17 of this chapter.
 - (b) [Reserved]
- (c) Application for construction permit for mobile station. Application for construction permit for each mobile station comprising any specified number of mobile units to be operated in the same service, including hand-carried or pack-carried units, may be combined into one application and shall be submitted on FCC Form 440.
- (d) Application for station license. Application for station license shall be filed on FCC Form 403 upon completion of construction or installation in accordance with the terms and conditions set forth in the construction permit.
- (e) Application for modification of construction permit. Separate application for modification of construction permit shall be submitted on FCC Form 440 for each station to be located at a fixed point. Application for modification of construction permit for any number of mobile units to be operated in the same service, including hand-carried or pack-carried units, may be combined into one application and shall be submitted on FCC Form 440.
- (f) Application for modification of station license. Application for modi-

fication of station license shall be submitted on FCC Form 403. A blanket application for modification of a group of station licenses of the same class may be submitted in those cases where the modification requested is the same for all stations covered by the application The individual stations covered by such application shall be clearly identified therein.

(g) Application for renewal of station license. (1) Application for renewal of station license shall be submitted on FCC Form 405. A blanket application may be submitted for renewal of a group of station licenses in the same class in those cases where the renewal requested is in exact accordance with the terms of the previous authorizations. The individual stations covered by such applications shall be clearly identified thereon. Unless otherwise directed by the Commission, each application for renewal of license shall be filed at least 60 days prior to the expiration date of the license to be renewed.

(2) If the station license sought to be renewed is used for the purpose of fulfilling the requirements of a contract with an agency of the U.S. Government, the application for renewal shall be accompanied by Form 440A in triplicate, Supplemental Information for Applications in the Experimental Radio Service Involving Government Contracts.

(h) Application for additional time to construct radio station. FCC Form 701 shall be submitted in duplicate, whenever it is necessary to request an extension of the time limit specified on a valid

construction permit.

- (i) Application for consent to assignment of radio station construction permit or license. Application on FCC Form 702 shall be submitted when the legal right to construct or to control the use and operation of a station is to be transferred as a result of a voluntary act (contract or other agreement) or an involuntary act (death or legal disability) of the grantee of a station authorization or by involuntary assignment of the physical property constituting the station under a court decree in bankruptcy proceedings, or other court order, or by operation of law in any other manner. Such application must be accompanied by statements signed by the proposed assignee which supply information required of an original licensee by § 5.57.
- (j) Application for consent to transfer of control of corporation holding construction permit or station license, Ap-

- plication for consent to transfer of control shall be submitted on FCC Form 703 whenever it is proposed to change the control of a corporation holding a station authorization.
- (k) Informal application. (1) An application not submitted on a standard form prescribed by the Commission is considered to be an informal application. Each informal application shall be submitted in duplicate, normally in letter form, and with the original signed in accordance with § 5.54. Each application shall be clear and complete within itself as to the facts presented and the action desired.
- (2) An informal application for authority to operate transmitting equipment will be accepted only under the conditions set forth in § 5.56.

[28 F.R. 12506, Nov. 22, 1963, as amended at 29 F.R. 9388, July 9, 1964; 34 F.R. 3802, Mar. 5, 1969]

- § 5.56 Procedure for obtaining a special temporary authorization.
- (a) The Commission may issue a special temporary authorization under this part in cases where a need is shown for operation of an authorized station for a limited time only, in a manner other than that specified in the existing authorization, but not in conflict with the Commission's rules.
- (b) An application for special temporary authorization may be filed as an informal application in the manner prescribed by § 5.55 (k) and shall contain the following information:
 - (1) Name and address.
 - (2) Need for special action.
 - (3) Type of operation to be conducted.
 - (4) Purpose of operation.
- (5) Time and date of proposed operation.
- (6) Class of station, call sign of station, and nature of service.
 - (7) Location of proposed operation.
- (8) Equipment to be used, including name of manufacturer, model and number of units.
 - (9) Frequency(s) desired.
- (10) Plate power input to final radio frequency stage.
 - (11) Type of emission.
- (12) Overall height of antenna structure above ground.
- (c) No request for special temporary authorization will be considered unless full particulars as to the purpose for which the request is made are stated. [28 F.R. 12506, Nov. 22, 1963, as amended at 34 F.R. 3802, Mar. 5, 1969]

Digitized by Google

§ 5.57 Supplementary statements required.

- (a) Showing. Each applicant for an authorization in the Experimental Radio Service must enclose with his application for construction permit a factual showing in regard to the following:
- (1) That the applicant is a person qualified to carry forward the proposed program of experimentation:
- (2) That the program of experimentation will be conducted by qualified personnel, and the applicant possesses adequate technical facilities to carry forward the program and has made adequate financial appropriations toward this end;
- (3) That the applicant has an organized plan of experimentation leading to a specific objective;
- (4) That the program of experimentation has reasonable promise of contribution to the development, extension, or expansion or utilization of the radio art, or is along lines not already investigated;
- (5) That the applicant has a program of experimentation that has reached a stage in the laboratory where actual transmission by radio is essential to its further progress;
- (6) That the station(s) shall be operated in accordance with the applicable Commission's rules and regulations and only in such a manner and at such times as to preclude harmful interference with established stations or services.
- (b) Confirmation clauses. In addition to the showing required by paragraph (a) of this section, a statement shall be filed with and made a part of each application for construction permit for a station in these services confirming the applicant's understanding:
- (1) That all operations on the frequencies assigned will be on an experimental basis, and conducted in accordance with the provisions of this part and as specified in the station instrument of authorization:
- (2) That the granting of the authority requested shall not be construed as a finding on the part of the Commission:
- (i) That the frequencies authorized are the best suited to the particular purpose to be served by the station;
- (ii) That the applicant is qualified to operate a station in a service on any basis other than experimental;
- (iii) That the applicant will be authorized to operate on any basis other than experimental;

- (iv) That the Commission is obligated by the results of the experimental program to make provision in the serviceallocation plan for the applicant's type of operation.
- (3) That the applicant desires and is willing to conduct and finance the experimental program with full knowledge and understanding of the provisions of this section.
- (c) Applications involving government contracts. The provisions of paragraphs (a) and (b) of this section shall not be applicable to applicants for an authorization in the Experimental Service (Research) to be used for the purpose of fulfilling the requirements of a contract with an agency of the United States Government. In lieu thereof, such applicants shall include as a part of the application, FCC Form 440—A in triplicate, Supplemental Information for Applications in the Experimental Radio Service Involving Government Contracts.
- (d) Applications involving development of equipment for export purposes. In addition to the information required by paragraphs (a) and (b) of this section, in the case of experimentation for the purpose of developing equipment for export purposes to be employed by stations under the jurisdiction of a foreign government, the application shall include:
- (1) The contract number and the name of the foreign government concerned.
- (2) The daily hours of operation and the estimated date of the beginning and end of the specific time period for which the authorization is required.
- (e) Applications involving communications essential to a research project. The provisions of paragraphs (a) and (b) of this section shall not be applicable to applications for an authorization in the Experimental Service (Research) to be used for communications essential to a research project. Applicants in this service shall include as a part of the application for construction permit the following:
- (1) A description of the nature of the research project being conducted.
- (2) A showing that communication facilities are necessary for the research project involved.
- (3) A showing that existing communication facilities are inadequate.

§ 5.58 Partial grants.

Where the Commission without a hearing grants any application in part, or with any privileges, terms, or conditions other than those requested, or subject to any interference that may result to a station if designated application or applications are subsequently granted, the action of the Commission shall be considered as a grant of such application unless the applicant shall, within 30 days from the date on which such grant is made or from its effective date if a later date is specified, file with the Commission a written request rejecting the grant as made. Upon receipt of such request, the Commission will vacate its original action upon the application and set the application for hearing in the same manner as other applications are set for hearing.

§ 5.59 Defective applications.

- (a) Applications which are defective with respect to completeness of answers to required questions, execution or other matters of a purely formal character will not be received for filing by the Commission, and will be returned to the applicant with a brief statement as to the omissions.
- (b) If an applicant is requested by the Commission to file any documents or information not included in the prescribed application form, a failure to comply with such request will constitute a defect in the application.
- (c) Applications which are not in accordance with the Commission's rules, regulations or other requirements will be considered defective unless accompanied either (1) by a petition to amend any rule or regulation with which the application is in conflict, or (2) by a request of the applicant for waiver of, or an exception to, any rule, regulation or requirement with which the application is in conflict. Such request shall show the nature of the waiver or exception desired and set forth the reasons in support thereof.

§ 5.60 Amendment or dismissal of applications.

(a) Any application may be amended or dismissed without prejudice upon request of the applicant prior to the time the application is granted or designated for hearing. Each amendment to, or request for dismissal of an application shall be signed, authenticated, and submitted in the same manner and with the

same number of copies as required for the original application. All subsequent correspondence or other material which the applicant desires to have incorporated as a part of an application already filed shall be submitted in the form of an amendment to the application.

(b) Fallure to prosecute an application, or failure to respond to official correspondence or request for additional information, will be cause for dismissal. Such dismissal will be without prejudice where an application has not yet been designated for hearing; such dismissal may be made with prejudice after an application has been designated for hearing.

§ 5.61 Construction period.

Each radio station construction permit issued by the Commission will specify the date of grant as the earliest date of commencement of construction and installation, and a maximum of eight months thereafter as the time within which construction shall be completed and the station ready for operation, unless otherwise determined by the Commission in any particular case.

§ 5.62 Licenses required for separate experimental projects.

A separate station license will be required for each class of station in the experimental radio services. Application for a class of station embracing widely divergent and unrelated experimentations normally will require a separate license for each phase of the experimental program: Provided, however, That the Commission may, when circumstances warrant, issue a single license embracing the entire project.

§ 5.63 License period.

- (a) The basic license period for stations in the Experimental Radio Services is 2 years.
- (b) A license will not be granted for a period longer than that which is required for completion of the experimental project. If such period is estimated to be less than 2 years, a statement to that effect by the applicant may facilitate a grant of the application. See also § 5.58.
- (c) The expiration dates for licenses in the Experimental Radio Services will be distributed over the 12 calendar months, in accordance with the alphabetical distribution of the names of licensees. Hence, an initial license may be granted for a basic period of 1½ to

2½ years, depending on the date of grant and the alphabetical position of the name of the licensee.

[32 F.R. 1129, Feb. 1, 1967]

§ 5.64 Change in equipment.

- (a) A change may be made in a licensed transmitter without specific authorization from the Commission provided: (1) The change does not result in operation inconsistent with any term of the outstanding authorization for the station involved; and (2) a description of the change is incorporated in the next application for renewal or modification of license.
- (b) Prior authorization from the Commission is required before the following antenna changes may be made at a station at a fixed location:
- (1) Any change which will either increase the height of a structure supporting the radiating portion of the antenna or decrease the height of a lighted antenna structure.
- (2) Any change in the location of an antenna when such relocation involves a change in the geographic coordinates of latitude or longitude by as much as one second, or when such relocation involves a change in street address.

§ 5.65 Operation at a temporary location.

- (a) An application for authority to operate at temporary locations shall specify the general geographical area within which the operation will be confined.
- (b) When a station is authorized to operate at temporary locations, the following notification procedure shall be followed:
- (1) When the station is placed in operation for the first time, the Engineer in Charge of the Radio District(s) involved shall be notified.
- (2) When the station is moved from one location to another, the Engineer in Charge of the Radio District(s) involved shall be notified.

§ 5.66 Discontinuance of station operation.

In case of a permanent discontinuance of operation of a fixed or land station in the experimental services, or in case of permanent discontinuance of operation of all transmitter units listed in the license for a mobile station in the experimental services, the licensee shall forward the station license to the Washington, D. C., office of the Commission

for cancellation. A copy of the request for cancellation of the license shall be forwarded to the Commission's Engineer in Charge of the radio district in which the station is located.

§ 5.67 Policy governing the assignment of frequencies.

- (a) Each frequency or band of frequencies, available for assignment to stations in the experimental services is available on a shared basis only, and will not be assigned for the exclusive use of any one applicant, and such use may also be restricted to one or more specified geographical areas. Normally not more than one frequency in a band of frequencies will be assigned for the use of a single applicant unless a showing is made demonstrating that need for the assignment of additional frequencies is essential to the proposed program of experimentation.
- (b) Frequency assignments will be made only on the condition that harmful interference will not be caused to any station operating in accordance with the Table of Frequency Allocation of Part 2 of this chapter.
- (c) The frequencies available for use in these services are set forth in §§ 5.203 and 5.253.

§ 5.68 Cancellation provisions.

The applicant for a station in the experimental services accepts the license with the express understanding: (a) That the authority to use the frequency or frequencies assigned is granted upon an experimental basis only and does not confer any right to conduct an activity of a continuing nature; and (b) that said grant is subject to change or cancellation by the Commission at any time without hearing if in its discretion the need for such action arises.

§ 5.69 Notification to the National Radio Astronomy Observatory.

In order to minimize possible harmful interference at the National Radio Astronomy Observatory site located at Green Bank, Pocahontas County, West Virginia, and at the Naval Radio Research Observatory site at Sugar Grove, Pendleton County, West Virginia, any applicant for a station authorization other than mobile, temporary base, temporary fixed, Citizens Radio, Civil Air Patrol, or Amateur seeking a station license for a new station, a construction permit to construct a new station or to modify an existing station license in a

manner which would change either the frequency, power, antenna height or directivity, or location of such a station within the area bounded by 39°15' N on the north, 78°30' W on the east, 37°30' N on the south and 80°30' W on the west shall, at the time of filing such application with the Commission, simultaneously notify the Director, National Radio Astronomy Observatory, P.O. Box #2, Green Bank, West Virginia, 24944, in writing, of the technical particulars of the proposed station. Such notification shall include the geographical coordinates of the antenna, antenna height, antenna directivity if any, proposed frequency, type of emission, and power. In addition, the applicant shall indicate in his application to the Commission the date notification was made to the Observatory. After receipt of such applications, the Commission will allow a period of twenty (20) days for comments or objections in response to the notifications indicated. If an objection to the proposed operation is received during the twenty-day period from the National Radio Astronomy Observatory for itself or on behalf of the Naval Radio Research Observatory, the Commission will consider all aspects of the problem and take whatever action is deemed appropriate.

Subpart C—Technical Standards § 5.101 Frequency stability.

(a) The frequencies assigned in accordance with the provisions of §§ 5.203 and 5.253 shall be maintained to within the tolerance set forth in the rules governing the service to which the frequencies are assigned in the Table of Frequency Allocations of Part 2 of this chapter unless otherwise designated in the station authorization.

(b) Less restrictive tolerances than those specified in paragraphs (a) and (b) of this section may be authorized for stations in the experimental services provided the applicant presents satisfactory evidence of the need for such tolerances and that the program of research can and will be conducted without causing harmful interference to any other radio service operating in accordance with the Table of Frequency Allocations set forth in Part 2 of this chapter.

§ 5.102 Types of emission.

(a) Stations in the Experimental Radio Services may be authorized to use

any of the classifications of emissions covered in Part 2 of this chapter.

§ 5.103 Emission limitations.

(a) Each authorization issued to a station operating in these services will show, as the prefix to the emission classification, a figure specifying the maximum authorized bandwidth in kilocycles to be occupied by the emission. The specified band shall contain those frequencies upon which a total of 99 percent of the radiated power appears, extended to include any discrete frequency upon which the power is at least 0.25 percent of the total radiated power. Any radiation in excess of the limits specified in paragraph (b) of this section is considered to be an unauthorized emission.

(b) The mean power of emissions shall be attenuated below the mean output power of the transmitter in accordance with the following schedule:

(1) On any frequency removed from the assigned frequency by more than 50 percent up to and including 100 percent of the authorized bandwidth: at least 25 decibels:

(2) On any frequency removed from the assigned frequency by more than 100 percent up to and including 250 percent of the authorized bandwidth: at least 35 decibels:

(3) On any frequency removed from the assigned frequency by more than 250 percent of the authorized bandwidth: at least 43 plus 10 Log₁₀ (mean output power in watts) decibels or 80 decibels, whichever is the lesser attenuation.

(c) When an unauthorized emission results in harmful interference, the Commission may, at its discretion, require appropriate technical changes in equipment to eliminate such interference.

(d) The Commission may, at its discretion, designate limits other than those set forth in paragraph (b) of this section after a proper showing of the need therefor.

§ 5.104 Modulation requirements.

Modulation requirements of stations in the Experimental Radio Services shall not be in excess of that necessary for the conduct of the authorized program of experimentation nor shall the emissions exceed the limitations imposed by § 5.103, unless specifically authorized by the Commission.

§ 5.105 Power and antenna height.

(a) The effective radiated power and antenna height which may be used by a station in this service shall be no more than the minimum necessary to satisfactorily carry on the authorized program of experimentation. No station shall operate at any time with power in excess of that authorized in the station instrument of authorization.

(b) The maximum power requested by an applicant shall not be in excess of the maximum obtainable power of the transmitter consistent with satisfactory

technical operation.

§ 5.106 Transmitter control requirements.

- (a) Each transmitter shall be so installed and protected that it is not accessible to or capable of operation by persons other than those duly authorized by the licensee.
- (b) A control point is an operating position which meets all of the following conditions:
- (1) The position must be under the control and supervision of the licensee;
- (2) It is a position at which the monitoring facilities required by this section are installed; and
- (3) It is a position at which an operator responsible for the operation of the transmitter is stationed.
- (c) Except where unattended operation is authorized in accordance with § 5.155 (b) (4), each station shall be provided with a control point, the location of which will be specified in the license. It will be assumed that the location of the control point is the same as that of the transmitting equipment unless the application includes a request for a different location. Authority must be obtained from the Commission for the installation of additional control points.
- (d) A dispatch point is a position from which messages may be transmitted under the supervision of a control point operator. Dispatch points may be installed without authorization from the Commission.
- (e) At each control point the following facilities shall be installed:
- (1) A carrier operated device which will provide continuous visual indication when the transmitter is radiating; or, in lieu thereof, a pilot lamp or meter, or other dependable device, which will provide continuous visual indication when the transmitter control circuits have been placed in a condition to produce

radiation: Provided, however, That the provisions of this subparagraph shall not apply to hand-carried or pack-carried transmitters.

(2) Equipment to permit the operator to aurally monitor all transmissions originating at dispatch points under his supervision; and capable of receiving transmissions from stations to which harmful interference might be caused.

(3) Facilities which will permit the operator either to disconnect the dispatch point circuits from the transmitter or to render the transmitter inoperative from any dispatch point under his supervision; and

(4) Facilities which will permit the operator to turn the transmitter carrier

on and off at will.

(5) In cases in which the applicant makes a satisfactory showing that one or more of the requirements set forth in subparagraphs (1) through (4) of this paragraph would be inappropriate and requests that such requirements be eliminated or others substituted, the Commission will consider such requests.

§ 5.107 Transmitter measurements.

- (a) Except as provided in paragraph (c) of this section, the licensee of each station in the Experimental Radio Services shall employ suitable procedures to determine that the carrier frequency, power, and modulation do not exceed the limits prescribed in the station authorization.
- (b) The transmitter operating characteristics shall be checked and necessary adjustments made when the equipment is originally installed, when any changes are made which might result in operation outside the limits specified in the instrument of authorization, and as often as necessary to maintain the transmitter operation within the limits specified in the instrument of authorization.
- (c) Exceptions to paragraphs (a) and (b) of this section may be made: *Provided*, The applicant makes a satisfactory showing that the nature of the proposed program of experimentation precludes compliance therewith.

Subpart D—Operating Requirements

§ 5.151 General limitations on use.

(a) The following transmission limitations are applicable to all classes of stations in the experimental services and are in addition to the specific requirements applicable to each respective class

of service as covered under Subparts E and F of this part:

- (1) Stations may make only such transmissions as are necessary and directly related to the conduct of the licensee's stated program of experimentation as specified in his application for construction permit and license and the related station instrument of authorization, and as governed by the provisions of the rules and regulations contained in this part. All transmissions shall be limited to the minimum practical transmission time.
- (2) When transmitting, the licensee must use every precaution to insure that the radio frequency energy emitted will not cause harmful interference to the services carried on by stations operating in accordance with the Table of Frequency Allocations of Part 2 of this chapter and, further, that the power radiated is reduced to the lowest practical value consistent with the program of experimentation for which the station authorization is granted. If harmful interference to an established radio service develops, the licensee shall cease transmissions and such transmissions shall not be resumed until it is certain that harmful interference will not be caused.
- (b) Unless expressly permitted in the instrument of authorization, experimental stations shall not be used:
- (1) To retransmit signals of any other station, except in conjunction with the authorized program of experimentation.
- (2) To transmit programs intended for public reception or render any communication service.

§ 5.152 Station identification.

Each class of station in the experimental services shall, unless specifically exempted by the terms of the station authorization, transmit its assigned call sign at the end of each complete transmission: Provided, however, That the transmission of the call sign at the end of each transmission is not required for projects requiring continuous, frequent, or extended use of the transmitting apparatus, if, during such periods and in connection with such use, the call sign is transmitted at least once every thirty minutes.

§ 5.153 Suspension of transmission required.

The radiations of the transmitter shall be suspended immediately upon detection or notification of a deviation from the technical requirements of the station authorization until such deviation is corrected, except for transmissions concerning the immediate safety of life or property, in which case the transmissions shall be suspended as soon as the emergency is terminated.

§ 5.154 Mobile installations in vehicles not under the continuous control of the licensee.

A mobile radio station licensed in the experimental services may not be installed or maintained in a vehicle, aircraft, or vessel, which is not at all times controlled exclusively by the licensee, unless precautions have been taken to eliminate effectively the possibility of the licensed transmitter being operated during the period that the vehicle, aircraft, or vessel is not under the control of the licensee.

§ 5.155 Operator requirements.

- (a) All transmitter adjustments which may affect the proper operation of a station shall be made by or under the immediate supervision and responsibility of a person holding a radiotelephone or radiotelegraph first or second class operator license: Provided, however, That only a person holding a radiotelegraph first or second class operator license shall perform such functions at a radiotelegraph station transmitting by any type of the Morse code.
- (b) A person holding a radiotelephone or radiotelegraph first or second class operator license, as may be appropriate for the type of emission being used, shall be on duty and in charge of the transmitter during the normal rendition of service: Provided, however, That if the transmitter is so designed that none of the operations necessary to be performed during the normal rendition of service may cause off-frequency operation or result in any unauthorized radiation, an operator holding any class of commercial radio operator license except Aircraft Radiotelephone Operator Authorization or Temporary Limited Radiotelegraph Second Class Operator License shall be on duty and in charge of the transmitter except:
- (1) Only a person holding a commercial radiotelegraph operator license of any class except Temporary Limited Radiotelegraph Second Class shall operate a station when transmitting radiotelegraphy by any type of Morse code: *Provided, however*, That a person holding a

commercial radiotelephone operator license of any class except Aircraft Radiotelephone Operator Authorization may operate such station when telegraphy is transmitted by automatic means for identification, testing, or actuating an automatic signalling device.

(2) An unlicensed person may operate a mobile station when transmitting radiotelephony on frequencies above 25 Mc/s.

(3) An unlicensed person may operate a mobile station when transmitting radiotelephony on frequencies below 25 Mc/s when such mobile station is under the operational control of a land station of the same licensee.

(4) No person is required to be in attendance at a station when transmitting on frequencies above 50 Mc/s for telemetering purposes or when serving as a relay station for the purpose of retransmitting by self-actuating means signals from another station or stations.

(c) The provisions of this section authorizing unlicensed persons to operate certain stations shall be applicable only to stations located within the United States, its territories or possessions and which communicate exclusively with one or more stations located in the United States, its territories or possessions.

(d) The provisions of this section authorizing unlicensed persons to operate mobile stations shall not be construed to change or diminish in any respect the responsibility of station licensees to have and to maintain control over the stations licensed to them, or for the proper functioning and operation of those stations in accordance with the terms of the licenses of those stations.

§ 5.156 Evidence of operator license.

Whenever a licensed operator is required by § 5.155, the original license or verification card of such operator shall be immediately available at the place where the operator is on duty.

§ 5.157 Posting station licenses and transmitter identification cards or plates.

(a) The current authorization for each station at a fixed location shall be posted in a conspicuous place at the principal control point of the station, and a photocopy of such authorization shall be posted at all other control points listed on the authorization. In addition, an executed Transmitter Identification

Card (FCC Form 452–C) or a plate of metal or other durable substance, legibly indicating the call sign and the licensee's name and address, shall be affixed, readily visible for inspection, to each transmitter operated at a fixed location when such transmitter is not in view of, or is not readily accessible to, the operator at the principal control point.

(b) The current authorization for each mobile station shall be retained as a permanent part of the station records, but need not be posted. In addition, an executed Transmitter Identification Card (FCC Form 452-C) or a plate of metal or other durable substance, legibly indicating the call sign and the licensee's name and address, shall be affixed, readily visible for inspection, to each mobile transmitter: Provided, That, if the transmitter is not in view of the operating position, or is not readily accessible for inspection, then such card or plate shall be affixed to the control equipment at the transmitter operating position or posted adjacent thereto.

§ 5.158 Authorized points of communication.

Stations in the experimental services may communicate only with other stations licensed in the experimental services: Provided, however, That upon a satisfactory showing that the proposed communications are essential to the conduct of the research project, authority may be granted to communicate with stations in other services and U.S. Government stations.

§ 5.159 Operation during an emergency.

(a) The licensee of any station in the Experimental Radio Services may, during a period of emergency in which the normal communication facilities are disrupted as a result of hurricane, flood, earthquake, or similar disaster, utilize such station for emergency communication service, subject to other provisions of this section, by communicating in a manner other than that specified in the station license.

(b) As soon as possible after the beginning of such emergency use, notice shall be sent to the Commission at Washington, D.C., and to the Engineer in Charge of the district in which the station is located, stating the nature of the emergency and the use to which the station is being put.

315

(c) The emergency use of the station shall be discontinued as soon as substantially normal communication facilities

are again available.

(d) The Commission at Washington, D.C., and the Engineer in Charge shall be notified immediately when such special use of the station is terminated.

(e) In no event shall any station engage in emergency transmission on frequencies other than, or with power in excess of, that specified in the instrument of authorization or as otherwise expressly provided by the Commission, or by law.

(f) The Commission may, at any time, order the discontinuance of any such emergency communication undertaken

under this section.

Inspection of stations. § 5.160

All stations and records of stations in the Experimental Services shall be made available for inspection at any time while the station is in operation or shall be made available for inspection upon reasonable request of an authorized representative of the Commission.

Inspection and maintenance of § 5.161 tower marking and lighting and associated control equipment.

The licensee of any radio station which has an antenna structure required to be painted and illuminated pursuant to the provisions of section 303(q) of the Communications Act of 1934, as amended, and Part 17 of this chapter, shall perform the inspections and maintain the tower marking and lighting, and associated control equipment, in accordance requirements of the §§ 17.43 through 17.57 of this chapter.

[34 F.R. 3802, Mar. 5, 1969]

§ 5.162 Notice of violation.

(a) Any licensee who appears to have violated any provision of the Communications Act or any provision of this chapter, shall be served with a written notice calling the facts to his attention and requesting a statement concerning the matter.

(b) Within 10 days from receipt of notice or such other period as may be specified, the licensee shall send a written answer, in duplicate, direct to the office of the Commission originating the official notice. If an answer cannot be sent nor an acknowledgment made within such 10-day period by reason of illness or other unavoidable circumstances, acknowledgment and answer shall be made

at the earliest practicable date with a satisfactory explanation of the delay.

= : :::=

(c) The answer to each notice shall be complete in itself and shall not be abbreviated by reference to other communications or answers to other notices. If the notice relates to violations that may be due to the physical or electrical characteristics of transmitting paratus, the answer shall state fully what steps, if any, have been taken to prevent future violations, and, if any new apparatus is to be installed, the date such apparatus was ordered, the name of the manufacturer, and the promised date of delivery. If the installation of such apparatus requires a construction permit, the file number of the application shall be given, or if a file number has not been assigned by the Commission, such identification shall be given as will permit ready identification of the application. If the notice of violation relates to lack of attention to or improper operation of the transmitter, the name and license number of the operator in charge shall be given.

§ 5.163 Content of station records.

(a) The licensee of each station in the experimental services shall maintain adequate records of the station's operations, including:

(1) Dates and hours of operation.

- (2) All measurements of the frequency(s), including the name of the person making the measurements, the exact frequency measured or the observed deviations from the assigned frequency(s) expressed in cycles, kilocycles or percent plus or minus, and a statement of any corrective action taken.
 - (3) Power.

(4) Types of emission.

(5) Chronological record of experimentation conducted.

(6) The name of the operator on duty.

- (b) For all stations, when service or maintenance duties are performed which may affect their proper operation, the responsible operator shall sign and date an entry in the station record concerned. giving:
- (1) Pertinent details of all duties performed by him or under his supervision:
 - (2) His name and address; and
- (3) The class, serial number and expiration date of his license: Provided, however, That the information called for under subparagraph (2) of this paragraph and this subparagraph, so long as it remains unchanged, is not required to

be repeated in the case of a person who is regularly employed as operator on a full-time basis at the stations.

(c) For stations whose antenna structure is required to be illuminated, a record in accordance with the requirements of § 17.49 of this chapter.

[28 F.R. 12506, Nov. 22, 1963, as amended at 34 F.R. 3802, Mar. 5, 1969]

§ 5.164 Form of station records.

(a) The records shall be kept in an orderly manner, in suitable form, and in such detail that the data required are readily available. Key letters or abbreviations may be used if proper meaning or explanation is set forth in the record.

(b) Each entry in the record shall be signed by a person having actual knowledge of the facts to be recorded.

- (c) No record or portion thereof shall be erased, obliterated, or willfully destroyed within the required retention period. Any necessary correction may be made only by the persons originating the entry, who shall strike out the erroneous portion, initial the correction made, and indicate the date of correction.
- (d) A copy of this part shall be maintained in the records of each fixed or land station licensed under this part.

§ 5.165 Retention of station records.

Records required to be kept by this part shall be retained by the licensee for a period of at least one year.

§ 5.166 Adherence to program of research.

(a) The program of experimentation as stated by an applicant in its application for construction permit or license or in the station instrument of authorization, shall be substantially adhered to unless the licensee is authorized to do otherwise by the Commission.

(b) Where some phases of the experimental program are not covered by the general rules of the Commission or by the rules of this part, the Commission may specify supplemental or additional requirements or conditions in each case as deemed necessary in the public interest, convenience, or necessity.

Subpart E—Experimental Service (Research)

§ 5.201 Eligibility for license.

Authorizations for stations in the Experimental Service (Research) will be issued only to persons qualified to con-

duct experimentations utilizing hertzian waves for scientific or technical radio research not related to an existing service or proposed service, or for communications in connection with research projects when existing communication facilities are inadequate.

§ 5.202 Scope of service.

Stations operating in the Experimental Service (Research) will be permitted to conduct the following types of operations:

- (a) Experimentations in scientific or technical radio research.
- (b) Development of radio technique, equipment or engineering data not relating to an existing or proposed service, including field or factory testing or calibration of equipment.
- (c) Experimentations under contractual agreement with the United States Government, or for export purposes.
- (d) Communications essential to research projects.
- (e) Technical demonstrations of equipment or techniques.

§ 5.203 Frequencies for Experimental Service (Research).

Stations operating in the Experimental Service (Research) may be authorized to use any government or non-government frequency designated in the table of frequency allocations set forth in Part 2 of this chapter as available for assignment to this service: *Provided*, That the need of the specific frequency(s) requested is fully justified by the applicant.

§ 5.204 Experimental report.

- (a) (1) Except in the case of those stations providing essential communications for research projects, a report on the results of the experimental program carried on under this subpart shall be filed with and made a part of each application for renewal of license.
- (2) The licensee shall, upon request, forward experimental reports at such times during the term of the station authorization as the Commission may deem necessary to evaluate the progress of the experimental program.
- (b) An applicant may request that the Commission withhold from the public certain reports and associated material, and the Commission will withhold the same unless the public interest requires otherwise.
- (c) In the case of experimentations which are under contractual agreement

with the United States, such reports shall include the information in paragraph (d) in so far as security regulations

(d) The experimental report shall include comprehensive information on the following items:

- (1) Detailed analysis of the results obtained.
- (2) Report on the experimentation conducted.
- (3) Total number of hours of operation on each frequency assigned.
- (4) Copies of publications covering the experimental work.
- (5) A list of patents issued as a result of the experimental work.
 - (6) Any other pertinent information.

Subpart F-Experimental Service (Developmental)

§ **5.2**51 Eligibility for license.

- (a) Authorizations for stations in the Experimental Service (Developmental) will be issued only to persons qualified to conduct experimentations utilizing hertzian waves for the development of equipment for use in an existing service or for the development of equipment or technical operational data directly related to a use of radio not provided by existing rules.
- (b) Applicants eligible for authorizations in an established service, and seeking to develop operational data or techniques directed toward the improvement or extension of that service, shall conduct such projects under the developmental rules of the established service.

§ 5.252 Scope of service.

Stations operating in the Experimental Service (Developmental) will be permitted to conduct the following types of operations:

(a) Development of radio equipment, operational or engineering data related to an existing or proposed radio service.

- (b) Field strength surveys by persons not eligible for authorization in any other service or the demonstration of equipment by manufacturers to prospective purchasers for proposed stations in existing services. Transmission shall be limited to test messages, essential to the installation, extension or development of a radio communication facility and the procedure set forth in § 5.254 shall apply.
- (c) Testing of equipment in connection with production or type approval of such equipment.

§ 5.253 Frequencies for Experimental Service (Developmental).

- (a) Any frequency allocated in Part 2 of this chapter to a particular service may be assigned for the purposes set forth in § 5.252 provided the proposed operation is in accordance with the rules governing the service involved.
- (b) Frequencies which have been allocated in Part 2 of this chapter to a shared Government and non-Government service for which rules governing non-Government stations have not been promulgated may be assigned for the purpose indicated in column 8 of the table of frequency allocations contained

in § 2.106 of this chapter.

- (c) Frequencies which have been allocated in Part 2 of this chapter to a non-Government service for which rules have not been promulgated may be assigned to the class of stations indicated in column 9 of the table of allocations for the purpose indicated in column 8 of that table.
- (d) Except as provided in paragraph (f) of this section, no developmental authorization for experimentation related to an established service will be made which involves the assignment of frequencies not included in the rules governing such service or the use of allocated frequencies in a manner contrary to the rules governing that service until the Commission has made a determination that such frequency assignment or use is in the public interest. Each application requesting the use of frequencies not allocated to an established service for which a developmental program is planned or which is contrary to the rules governing that service must be accompanied by a petition requesting the amendment of the rules governing the service involved to provide for the proposed operation.
- (e) Except as provided in paragraph (f) of this section, frequencies will not be assigned for the development of a service for which no frequencies have been allocated until the Commission has made a preliminary determination that the public interest, convenience, or necessity would be served by the establishment of the service. Such applications must be accompanied by a petition requesting the allocation of frequencies for the proposed service and setting forth the reasons in support of the petition.

Digitized by Google

(f) Where circumstances require and permit, the Commission may defer action on any petition required by paragraphs (d) and (e) of this section and authorize a grant of limited duration for the sole purpose of developing data which the Commission finds to be necessary to make the determinations with respect to such petitions. Such grants shall in no way be construed as a finding by the Commission with respect to the matters set forth in the petitions or that the operation of any radio station thereunder will serve the public interest, convenience, or necessity beyond the express terms of the particular grant. The terms of such grant, including frequency, power, emission, etc., will be specified in the instrument of authorization.

§ 5.254 Frequencies for field strength surveys or equipment demonstrations.

- (a) Specific frequencies will not normally be designated in a station license issued for the purpose of conducting field strength surveys or equipment demonstrations. Prior to the commencement of a survey, the licensee will request a specific frequency assignment and submit the following information:
 - (1) Time, date and duration of survey.
 - (2) Frequency to be used.
- (3) Location of transmitter and geographical area to be covered.
 - (4) Purpose of survey.
 - (5) Method and equipment to be used.(6) Names and addresses of persons

for whom the survey is conducted.

(b) Upon receipt of authority from the Commission to conduct a particular survey, the licensee shall furnish the Engineer in Charge of the radio district in which the survey is to be conducted, sufficiently in advance to assure receipt before the commencement thereof, the following information: Time, date, duration, frequency, location of transmitter, area to be covered, and purpose of survey.

§ 5.255 Experimental report.

(a) A report on the results of the experimental program carried on under this subpart shall be filed with and made a part of each application for renewal of license: Provided, however, That the licensee shall, upon request, forward experimental reports at such time during the term of the station authorization as the Commission may deem necessary to

evaluate the progress of the experimental program.

- (b) An applicant may request that the Commission withhold from the public certain reports and associated material, and the Commission will withhold the same unless the public interest requires disclosures.
- (c) The experimental report of stations operating in this service for the development of existing services shall include comprehensive information on the following items in the order designated:
- (1) Detailed analysis of the results obtained.
- (2) Report on the experimental work conducted.
- (3) The total number of hours of operation on each frequency.
- (4) Copies of any published reports on the program of experimentation.
- (5) Any other pertinent information that may be useful to the Commission in evaluating the merits of the proposed operations.
- (d) In addition to the information included in paragraph (c) of this section, the experimental report of a station authorized for the development of a new service shall include comprehensive information on the following items:
- Probable public support and methods of its determination.
 - (2) Practicability of service operations.
 - (3) Interference encountered.
- (4) Pertinent information relative to merits of the proposed service.
- (5) Propagation characteristics of frequencies used, particularly with respect to service objective.
- (6) Frequencies believed to be more suitable and reasons therefor.
- (7) Type of signals or communications employed in the experimental work.

Subpart G—[Reserved] Subpart H—Student Authorizations § 5.401 Eligibility for license.

The Commission may issue an authorization under this subpart to students for the purpose of presenting experiments or technical demonstrations for school or school approved projects which require the use of radio for a limited period of time. Such authorizations may, in the discretion of the Commission, be granted to students of seventh grade or higher level.

§ 5.402 Filing of applications.

(a) Notwithstanding the requirements in Subpart B of this part, an application for a student authorization may be filed in letter form, in duplicate, with the original signed. The application shall contain the information set forth in paragraph (b) of this section.

(b) The application shall contain the

following information:

 Name and address of applicant.
 A statement that the applicant is a citizen of the United States.

(3) Applicant's school and grade.

(4) A detailed description in narrative form of the project including the type and purpose of operation.

(5) Place of operation—street address, name of building, or other specific

location.

(6) Date(s) of operation including the exact hours, when known, as well as the duration of each period of operation.

- (7) Equipment to be used. If manufactured, list name of manufacturer and type number. For other equipment, describe in detail and furnish a circuit diagram.
- (8) Frequency(ies) desired and range of frequencies which could be employed.

(9) The method by which the frequency of operation will be determined.

(10) Frequency tolerance.

(11) The means by which this tolerance will be maintained.

(12) DC plate power input to final radio frequency stage. If not known, indicate any known power rating of equipment and state whether this is power output of transmitter or radiated power, and whether average or peak.

(13) Type of emission including a description of the modulation that will be

applied, if modulated.

(14) Description of the antenna to be used, including height above ground.

(c) The application shall be accompanied by a signed statement from the principal of the school, or a member of its faculty, on appropriate letterhead, stating that the project has the approval of the school and indicating the person under whose general supervision the project will be conducted.

§ 5.403 Waiver of construction permit.

Subject to the requirements of §§ 5.401 and 5.402, the provisions contained in section 319 (d) of the Communications Act are waived insofar as such provisions require the issuance of

a construction permit prior to the issuance of the student authorization provided for in this subpart.

§ 5.404 Waiver of operator requirement.

Subject to the requirements of §§ 5.401 and 5.402, the provisions contained in section 318 of the Communications Act are waived insofar as such provisions require that a station authorized under this subpart be operated by a person holding an operator's license: Provided, however, That the Commission, in its discretion, may require that operations authorized pursuant to this subpart be conducted by, or in the presence of, a licensed operator of the appropriate grade.

§ 5.405 Power limitation.

No authorization under this subpart will be issued unless the description of the project shows that the dc plate power input to the final radio frequency stage does not exceed 5 watts: Provided, however, That a greater power may be authorized if a satisfactory showing is made that such greater power is necessary and that appropriate measures will be taken to prevent interference.

§ 5.406 Frequencies.

(a) Frequencies in the following bands are available for assignment in authorizations issued under this subpart:

27.23-27.28 Mc/s. 460-461 Mc/s. 462.525-467.475 Mc/s. 2450-2500 Mc/s.

(b) In each case, the carrier frequency must be far enough inside the band so as to keep the sideband energy within the band limits specified.

§ 5.407 Measurements required.

The frequency of operation must be measured or checked prior to each time of operation.

§ 5.408 Waiver of technical and operating requirements.

Subject to the provisions of §§ 5.405, 5.406, and 5.407, the provisions in Subparts C and D of this part are waived insofar as such provisions require a station authorized under this subpart to observe the technical and operating requirements set forth therein.

§ 5.409 Non-interference condition.

Each authorization issued to a student under this subpart is subject to the condition that no harmful interference, as defined in § 5.4(h), is caused to any authorized station.

§ 5.410 Record of operation.

- (a) The licensee holding an authorization issued under this subpart shall maintain a record of operation containing the following information:
- (1) The date and time of each period of operation.
- (2) The frequency of operation as measured or checked at the beginning of each period of operation.
- (3) A brief description of the experimentation being conducted.
- (b) Each entry shall be signed by the person operating the equipment.
- (c) These records shall be retained for one month after the termination of the authorization.

§ 5.411 Notification.

- (a) The holder of an authorization issued under this subpart shall notify the Engineer in Charge of the district in which the station will be operated in advance of each scheduled operation.
- (b) The notice to the Engineer in Charge shall be in writing and shall contain the following information:
 - (1) Place of operation.
- (2) Date(s) of operation, including exact time if known.
 - (3) Frequency(ies) to be used.
 - (4) Call letters of station.

PART 13—COMMERCIAL RADIO OPERATORS

GENERAL

Dec.	
13.1	Licensed operators required.
13.2	Classes of operator licenses.
13.3	Dual holding of licenses.
13.4	Term of licenses.
13.5	Eligibility for new license.
10 4	Operator license posting of

200

13.7 Operators, place of duty.
13.8 Provisional Radio Operator Certificate.

APPLICATIONS

13.11	Procedi	ıre.		
13.12	Special	provisions,	radiotelegraph	first
	class	_		

13.13 Age limit, restricted radiotelephone operator permit.

EXAMINATIONS

13.21	Examination	elements.
19 22	Exemination	requirements

- Sec. 13.23 Form of writing.
- 13.24 Passing mark.
- 13.25 New class, additional requirements. 13.26 Canceling and issuing new licenses.
- 13.27 Eligibility for re-examination.
- 13.28 Renewal service requirements, renewal examinations, and exceptions.

CODE TESTS

- 13.41 Transmitting speed requirements.
- 13.42 Transmitting test procedure.
- 13.43 Receiving speed requirements.
- 13.44 Receiving test procedure.13.45 Computing words or code groups.

SCOPE OF AUTHORITY

- 13.61 Operating authority.
- 13.62 Special privileges.
- 13.63 Operator's responsibility.
- 13.64 Obedience to lawful orders.
- 13.65 Damage to apparatus.
- 13.66 Unnecessary, unidentified, or superfluous communications.
- 13.67 Obscenity, indecency, profanity.
- 13.68 False signals.
- 13.69 Interference.
- 13.70 Fraudulent licenses.

MISCELLANEOUS

- 13.71 Issue of duplicate or replacement licenses
- 13.72 Exhibiting signed copy of application.
- 13.73 Verification card.
- 13.74 Posting requirements for operator.
- 13.75 Record of service and maintenance duties performed.
- 13.76 Limitation on aircraft pilots.

SERVICE

- 13.91 Endorsement of service record.
- 13.92 Aviation service endorsement.
- 13.93 Service acceptability.
- 13.94 Statement in lieu of service endorsement.

Appendix-Filing fees.

AUTHORITY: The provisions of this Part 13 issued under secs. 4, 303, 48 Stat. 1066, 1082 as amended; 47 U.S.C. 154, 303.

SOURCE: The provisions of this Part 13 appear at 28 F.R. 12515, Nov. 22, 1963, unless otherwise noted.

GENERAL

§ 13.1 Licensed operators required.

Unless otherwise specified by the Commission, the actual operation of any radio station for which a station license is required shall be carried on only by a licensed radio operator of the required class (see § 13.61).

NOTE A: Whenever the term "license" is used generally to denote an authorization from the Commission, it includes "license," "permit" and "authorization".

Note B: Provision is made in Parts 5, 21, 81, 83, 87, 89, 91, 93, and 95 of this chapter for operation of certain radio stations without licensed operators subject to limitations and conditions specified therein.

§ 13.2 Classes of operator licenses.

The classes of commercial radio operator licenses issued by the Commission are classified basically as radiotelegraph and radiotelephone licenses, and are further classified in accordance with international usage as follows:

- (a) General radio operator group:
- (1) General radiotelegraph certificates:
- (i) Radiotelegraph First Class Operator License.
- (ii) Radiotelegraph Second Class Operator License.
- (iii) Temporary Limited Radiotelegraph, Second Class Operator License.

Note: This class of license will be issued until further order of the Commission or until September 15, 1969, whichever occurs first.

- (2) General radiotelephone certificates (classification by international usage):
- (i) Radiotelephone First Class Operator License.
- (ii) Radiotelephone Second Class Operator License.
 - (b) Restricted radio operator group:
- (1) Special radiotelegraph certificate (classification by international usage):
- (i) Radiotelegraph Third Class Operator Permit.
- (2) Restricted radiotelephone certificate (classification by international usage):
- (i) Radiotelephone Third Class Operator Permit.
 - (c) Limited radio operator group:
- (1) Limited radiotelephone operator certificate:
- (i) Restricted Radiotelephone Operator Permit.
- [28 F.R. 12515, Nov. 22, 1963, as amended at 81 F.R. 14591, Nov. 16, 1966]

§ 13.3 Dual holding of licenses.

- (a) Except as provided by paragraph (b) of this section, a person may not hold more than one radiotelegraph operator license or permit and one radiotelephone operator license or permit at the same time.
- (b) A person may at the same time hold (1) both a temporary limited radiotelegraph second-class operator license and a radiotelegraph third-class operator permit, (2) both a provisional certificate

for radiotelephone third-class operator permit endorsed for broadcast use and a radiotelephone third-class operator permit not so endorsed, (3) both a provisional certificate for a radiotelephone third-class operator permit endorsed for broadcast use and a restricted radiotelephone operator permit.

[33 F.R. 913, Jan. 25, 1968]

§ 13.4 Term of license.

- (a) Except as provided in paragraphs (b), (c), and (d) of this section, commercial operator licenses will normally be issued for a term of 5 years from the date of issuance.
- (b) Restricted Radiotelephone Operator Permits issued to U.S. citizens or other U.S. nationals will normally be issued for the lifetime of the operator. The terms of all Restricted Radiotelephone Operator Permits issued prior to November 15, 1953, which were outstanding on that date were extended to encompass the lifetime of such operators.
- (c) A commercial operator license, of any grade, granted to an alien aircraft pilot under a waiver of the U.S. nationality provisions of section 303(1) of the Communications Act, until such time as the question of a national security policy has been determined with respect to such persons will normally be issued for a period not in excess of 1 year from the date of issuance. An operator license issued to an alien shall be valid only if the operator continues to hold an aircraft pilot certificate issued by the Federal Aviation Administration or one of its predecessor agencies and is lawfully in the United States.
- (d) Temporary limited radiotelegraph second-class operator licenses will normally be issued for a term of 2 years. [28 F.R. 12515, Nov. 22, 1963, as amended at 31 F.R. 12480, Sept. 21, 1966; 33 F.R. 698,

§ 13.5 Eligibility for new license.

Jan. 19, 1968]

- (a) Commercial licenses are issued only to citizens and other nationals of the United States except, in the case of aliens who hold aircraft pilot certificates issued by the Federal Aviation Administration or one of its predecessor agencies, the Commission, if it finds that the public interest will be served thereby, may waive the requirement of U.S. nationality.
- (b) Notwithstanding any other provisions of the Commission's rules, no person otherwise eligible shall be deemed to

be eligible to be examined for or to receive a commercial radio operator license of any class, (1) whose commercial radio operator license is under suspension or is involved in a suspension proceeding, (2) who is involved in any pending litigation based on an alleged violation of the Communications Act of 1934, as amended, or (3) who is afflicted with complete deafness or complete muteness or complete inability for any other reason to transmit correctly and to receive correctly by telephone spoken messages in English.

(c) No applicant who is eligible to apply for any commercial radio operator license shall, by reason of any physical handicap, other than as set forth in paragraph (b) of this section, be denied the privilege of applying and being permitted to attempt to prove his qualifications (by examination if examination is required) for such commercial radio operator license in accordance with established procedure; nor, subject to the following conditions, shall such applicant be denied the issuance of any commercial radio operator license for which he is found qualified:

(1) If the applicant is afflicted with uncorrected physical handicap which would clearly prevent the performance of all or any part of the duties of a radio operator, under the license for which application is made, at a station under emergency conditions involving safety of life or property, he may be issued the license for which he is found qualified: Provided. That any license so received, if of the diploma-form (as distinguished from such document of the card-form), shall bear a restrictive endorsement as follows:

This license is not valid for the performance of any operating duties, other than installation, service and maintenance duties, at any station licensed by the Federal Communications Commission which is required, directly or indirectly, by any treaty, statute or rule or regulation pursuant to statute, to be provided for safety purposes.

Provided further, That in the case of a diploma-form license for which no examination in technical radio matters is required, the endorsement will be modified by deleting the reference therein to installation, service, and maintenance duties.

(2) If an applicant afflicted with blindness is afforded a waiver of the written examination requirement and is found qualified for a radiotelephone third class operator permit, he may be issued the permit: *Provided*, That the license so received shall bear an endorsement as follows:

This license is not valid for the operation of any station licensed by the Commission unless the station has been adapted for operation by a blind person and the equipment to be used in such station for that purpose is capable of providing for operation in compliance with the Commission's rules.

(3) In any case where an applicant who normally would receive or has received a commercial radio operator license bearing the endorsement prescribed by subparagraph (1) of this paragraph, indicates his desire to operate a station falling within the prohibitive terms of the endorsement, he may request in writing that such endorsement not be placed upon, or be removed from, his license, and may submit in support of his request any written comment or statement of himself or any interested party.

(4) An applicant who shows that he has theretofore performed satisfactorily (by means of the service record appearing on the appropriate license document of the applicant or such other proof as may be appropriate under the circumstances of the particular case) the duties of a radio operator at a station required. directly or indirectly, by any treaty, statute, or rule or regulation pursuant to statute to be provided for safety purposes, during a period when he was afflicted by uncorrected physical handicaps of the same kind and to the same degree as the physical handicaps shown by his current application (this showing may be made by means of the applicant's written, sworn statement or such other documentary proof as may be appropriate under the circumstances of the particular case), shall not be deemed to be within the provisions of subparagraph (1) of this paragraph.

(d) Even though otherwise eligible to apply for an operator license, no person shall be eligible to apply for a temporary limited radiotelegraph second-class operator license except a person who on or after January 1, 1940, held, but does not hold at the time of filing application, a license which was valid and outstanding on its date of expiration in the following categories:

(1) A radiotelegraph first- or secondclass operator license; (2) A temporary limited radiotelegraph second-class operator license is-

sued after examination;

(3) A temporary limited radiotelegraph second-class operator license issued on the basis of having previously held at any time a radiotelegraph firstor second-class operator license. When a temporary limited radiotelegraph operator license is sought under this third category, the applicant shall show that he has had at least 6 months' satisfactory service in the aggregate as a qualified radiotelegraph operator on board a ship or ships of the United States while holding a temporary limited radiotelegraph second-class operator license previously issued by the Commission.

[28 F.R. 12515, Nov. 22, 1963, as amended at 31 F.R. 12481, Sept. 21, 1966; 32 F.R. 9232, June 29, 1967; 33 F.R. 698, Jan. 19, 1968]

§ 13.6 Operator license, posting of.

The original license of each station operator shall be posted at the place where he is on duty, except as otherwise provided in this part or in the rules governing the class of station concerned.

§ 13.7 Operators, place of duty.

(a) Except as may be provided in the rules governing a particular class of station, one or more licensed radio operators of the grade specified by this part shall be on duty at the place where the transmitting apparatus of each licensed radio station is located and in actual charge thereof whenever it is being operated: Provided, however, That, (1) subject to the provisions of paragraph (b) of this section, where remote control of the transmitting apparatus has been authorized to be used, the Commission may modify the foregoing requirements upon proper application and showing being made so that such operator or operators may be on duty at the control point in lieu of the place where the transmitting apparatus is located; (2) in the case of two or more stations, except amateur and broadcast, licensed in the name of the same person to use frequencies above 30 megacycles only, a licensed radio operator holding a valid radiotelegraph or radiotelephone first- or second-class license who has the station within his effective control may be on duty at any point within the communication range of such stations in lieu of the transmitter location or control point during the actual operation of the transmitting apparatus and shall supervise the emissions of all such stations so as to insure the proper operation in accordance with the station license.

(b) An operator may be on duty at a remote control point in lieu of the location of the transmitting apparatus in accordance with the provisions of paragraph (a) (1) of this section: Provided. That all of the following conditions are met: (1) The transmitter shall be so installed and protected that it is not accessible to other than duly authorized persons; (2) the emissions of the transmitter shall be continuously monitored at the control point by a licensed operator of the grade specified for the class of station involved; (3) provision shall be made so that the transmitter can quickly and without delay be placed in an inoperative condition by the operator at the control point in the event there is a deviation from the terms of the station license; (4) the radiation of the transmitter shall be suspended immediately when there is a deviation from the terms of the station license.

§ 13.8 Provisional Radio Operator Certificate.

(a) In circumstances requiring immediate authority to operate a radio station pending submission of proof of eligibility or of qualifications or pending a determination by the Commission as to these matters, an applicant for a radio operator license may request a Provisional Radio Operator Certificate.

(b) Except as provided by paragraph
 (e) of this section, a request for a Provisional Radio Operator Certificate may be in letter form and shall be in addi-

tion to the formal application.

(c) Except as provdied by paragraph (e) of this section, if the Commission finds that the public interest will be served, it may issue such certificates for a period not to exceed 6 months with such additional limitations as may be indicated.

(d) Except as provided by paragraph (e) of this section, a Provisional Radio Operator Certificate will not be issued if the applicant has not fulfilled examination or service requirements, if any, for the license applied for.

(e) A request for a Provisional Radio Operator Certificate for a radiotelephone third-class operator permit endorsed for broadcast use shall be made on FCC Form 756C, which provides for a certification by the holder of a radiotelephone first-class operator license that he is responsible for the technical maintenance of a radio broadcast station, and that he has instructed the applicant in the operation of a broadcast station and believes him to be capable of performing the duties expected of a person holding a radiotelephone third-class operator permit with broadcast endorsement. If the Commission finds that the public interest will be served, it may issue such certificates under the following conditions:

- (1) The certificate is valid for a period not to exceed 12 months.
 - (2) The certificate is not renewable.
- (3) The certificate may be issued to a person only once.
- (4) Additional limitations may be specified, as necessary.
- (5) The certificate may be issued prior to the fulfillment of examination requirements for the radiotelephone third-class operator permit endorsed for broadcast use.

[33 F.R. 914, Jan. 25, 1968]

APPLICATIONS

§ 13.11 Procedure.

Applications shall be (a) General. governed by applicable rules in force on the date when application is filed (see The application in the pre-§ 13.28). scribed form and including all required subsidiary forms and documents, properly completed and signed, and accompanied by the prescribed fee (see Appendix, Part 13), shall be submitted to the appropriate office as indicated in paragraph (b) of this section. If the application is for renewal of license, it may be filed at any time during the final year of the license term or during a 1-year period of grace after the date of expiration of the license sought to be renewed. During this 1-year period of grace, an expired license is not valid. A renewed license issued upon the basis of an application filed during the grace period will be dated currently and will not be backdated to the date of expiration of the license being renewed. A renewal application shall be accompanied by the license sought to be renewed. If the prescribed service requirements for renewal without examination (see § 13.28) are fulfilled, the renewed license may be issued by mail. If the service record on the reverse side of the license does not fully describe or cover the service desired by the applicant to be considered in connection with license renewal (as might occur in the case of service rendered at U.S. Government stations), the renewal application shall be supported by documentary evidence describing in detail the service performed and showing that the applicant actually performed such service in a satisfactory manner. A separate application must be submitted for each license involved, whether it requests renewal, new license, endorsement, duplicate, or replacement.

(b) Place of filing. (1) An application (FCC Form 753) for restricted radiotelephone operator permit shall be submitted to the Federal Communications Commission, Gettysburg, Pa. 17325, with the fol-

lowing exceptions:

(i) When the applicant is located in Alaska, Hawaii, Puerto Rico, or the Virgin Islands of the United States, the application may be submitted by mail or in person to the nearest engineering field office.

(ii) When the applicant is at any other location and the application is accompanied by a written showing by the applicant of immediate need for a permit for safety purposes and presented in person by the applicant or his agent, the application may be submitted to the nearest engineering field office.

(iii) When accompanied by a request (FCC Form 755) for a waiver of the U.S. nationality requirement, as in the case of an alien applicant who is an aircraft pilot (see § 13.4.(c)), the application shall be submitted in person or by mail to the Federal Communications Commission, Washington, D.C. 20554.

(2) An application (FCC Form 756) for an operator license of any other class. for verification card (FCC Form 758-F) or for a verification of operator license (FCC Form 759) shall be submitted in person or by mail to the field office at which the applicant desires his application to be considered and acted upon, which office will make final arrangements for conducting any required examination. Whenever an examination is to be taken at a designated examination point away from a field office, the application shall be submitted in advance of the examination to the field office having jurisdiction over the area in which the examination is to be taken.

(c) Restricted radiotelephone operator permit. No oral or written examination is required for this permit. If the application is properly completed and signed, and if the applicant is found to

be qualified, the permit may be issued forthwith.

(d) Short term license. A license, other than a Restricted Radiotelephone Operator Permit, issued for a term of less than 5 years (see §13.4(c)), may be extended for a period not exceeding the portion of the 5-year term remaining, without further examination, provided proper application for extension is filed prior to expiration of the license.

[29 F.R. 15761, Nov. 24, 1964, as amended at 31 F.R. 15, Jan. 4, 1966; 32 F.R. 2627, Feb. 8, 1967]

§ 13.12 Special provisions, radiotelegraph first class.

An applicant for a radiotelegraph first-class operator license must be at least 21 years of age at the time the licensed is issued and shall have had an aggregate of one year of satisfactory service as an operator manipulating the key of a manually operated public ship or coast station handling public correspondence by radiotelegraphy.

§ 13.13 Age limit, restricted radiotelephone operator permit.

An applicant for a restricted radiotelephone operator permit must be at least 14 years of age at the time the permit is issued.

Examinations 3 3 2

§ 13.21 Examination elements.

Written examinations will comprise questions from one or more of the following examination elements:

1. Basic law. Provisions of laws, treaties and regulations with which every operator should be familiar.

2. Basic operating practice. Radio operating procedures and practices generally followed or required in communicating by means of radiotelephone stations.

3. Basic radiotelephone. Technical, legal and other matters applicable to the operation of radiotelephone stations other than broadcast.

4. Advanced radiotelephone. Advanced technical, legal and other matters particularly applicable to the operation of the various classes of broadcast stations.

5. Radiotelegraph operating practice. Radio operating procedures and practices generally followed or required in communicating by means of radiotelegraph stations primarily other than in the maritime mobile services of public correspondence.

Technical, Advance radiotelegraph. legal and other matters applicable to the operation of all classes of radiotelegraph stations, including operating procedures and practices in the maritime mobile services of public correspondence, and associated matters such as radio navigational aids, message traffic routing and accounting, etc.

7. Aircraft radiotelegraph. Basic theory and practice in the operation of radio communication and radio navigational systems

in general use on aircraft.

8. Ship radar techniques. Specialized theory and practice applicable to the proper installation, servicing and maintenance of ship radar equipment in general use for marine navigational purposes.

9. Basic broadcast. Basic regulatory matters applicable to the operation of standard, commercial FM, and noncommercial educa-

tional FM broadcast stations.

§ 13.22 Examination requirements.

Applicants for original licenses will be required to pass examinations as follows:

(a) Radiotelephone second-class oper**a**tor license:

(1) Ability to transmit and receive spoken messages in English.

(2) Written examination elements: 1, 2, and 3.

(b) Radiotelephone first-class operator license:

(1) Ability to transmit and receive spoken messages in English.

(2) Written examination elements: 1.

2, 3, and 4. (c) Radiotelegraph second-class operator license:

(1) Ability to transmit and receive spoken messages in English.

(2) Transmitting and receiving code test of twenty (20) words per minute plain language and sixteen (16) code groups per minute.

(3) Written examination elements: 1,

2, 5, and 6.

(d) Temporary limited radiotelegraph second-class operator license: (1) Ability to transmit and receive spoken messages in English.

(2) Transmitting and receiving code test of twenty (20) words per minute plain language and sixteen (16) code groups per minute.

(e) Radiotelegraph first-class operator license:

(1) Ability to transmit and receive spoken messages in English.

(2) Transmitting and receiving code test of twenty-five (25) words per minute plain language and twenty (20) code

groups per minute. (3) Written examination elements: 1, 2, 5, and 6.

(f) Radiotelephone third-class operator permit:

(1) Ability to transmit and receive spoken messages in English.

- (2) Written examination elements: 1 and 2.
- (g) Radiotelegraph third-class operator permit:
- (1) Ability to transmit and receive spoken messages in English.
- (2) Transmitting and receiving code test of twenty (20) words per minute plain language and sixteen (16) code groups per minute.
- (3) Written examination elements: 1, 2, and 5.
- (h) Restricted radiotelephone operator permit:

No oral or written examination is for required this permit. In lieu thereof, applicants will be required to certify in writing to a declaration which states that the applicant has need for the requested permit; can receive and transmit spoken messages in English; can keep at least a rough written log in English or in some other language in general use that can be readily translated into English: is familiar with the provisions of treaties, laws, and rules and regulations governing the authority granted under the requested permit; and understands that it is his responsibility to keep currently familiar with all such provisions.

[28 F.R. 12515, Nov. 22, 1963, as amended at 81 F.R. 12481, Sept. 21, 1966]

§ 13.23 Form of writing.

Written examination shall be in English and shall be written by the applicant in longhand in ink, except that diagrams may be in pencil.

§ 13.24 Passing mark.

A passing mark of 75 percent of a possible 100 percent will be required on each element of a written examination.

§ 13.25 New class, additional requirements.

The holder of a license, who applies for another class of license, will be required to pass only the added examination requirements for the new class of license: Provided, That the holder of a radiotelegraph third-class operator permit who takes an examination for a radiotelegraph second-class operator license more than 1 year after the issuance date of the third-class permit will also be required to pass the code test prescribed therefor: Provided further, That no person holding a new, duplicate, or replacement restricted radiotelephone operator permit issued on the basis of a declaration, or

a renewed restricted radiotelephone operator permit which renews a permit issued upon the basis of a declaration shall, by reason of the declaration or the holding of such permit, be relieved in any respect of qualifying by examination when applying for any other class of license. In addition, no person holding a temporary limited radiotelegraph second-class operator license will, by the holding of such license, be relieved of qualifying by examination when applying for any other class of license for which examination on any subject matter is required.

[31 F.R. 12481, Sept. 21, 1966]

§ 13.26 Canceling and issuing new licenses.

If the holder of a license qualifies for a higher class in the same group, the license held will be canceled upon the issuance of the new license. Similarly, if the holder of a restricted operator permit qualifies for a first- or second-class operator license of the corresponding type, the permit held will be canceled upon issuance of the new license.

§ 13.27 Eligibility for reexamination.

An applicant who fails an examination element will be ineligible for 2 months to take an examination for any class of license requiring that element. Examination elements will be graded in the order listed (see § 13.21), and an applicant may, without further application, be issued the class of license for which he qualifies.

Note: A month after date is the same day of the following month, or if there is no such day, the last day of such month. This principle applies for other periods. For example, in the case of the 2-month period to which this note refers, an applicant examined December 1 may be reexamined February 1, and an applicant examined December 29, 30, or 31 may be reexamined the last day of February while one examined February 28 may be reexamined April 28.

§ 13.28 Renewal service requirements, renewal examinations, and exceptions.

A restricted radiotelephone operator permit normally is issued for the lifetime of the holder and need not be renewed. A temporary limited radio telegraph second-class operator license is not renewable. A license of any other class may be renewed without examination provided that the service record on

the reverse side of the license (see §§ 13.91 to 13.94) shows at least two years of satisfactory service in the aggregate during the license term and while actually employed as a radio operator under that license. If this two-year renewal service requirement is not fulfilled but the service record shows at least one year of satisfactory service in the aggregate during the last three years of the license term and while actually employed as a radio operator under that license, the license may be renewed upon the successful completion of a renewal examination, which may be taken at any time during the final year of the license term or during a one-year period of grace after the date of expiration of the license sought to be renewed. The renewal examination will consist of the highest numbered examination element normally required for a new license of the class sought to be renewed, plus the code test (if any) required for such a new license. If the renewal examination is not successfully completed before expiration of the aforementioned one-year period of grace, the license will not be renewed on any basis.

Note: By order dated and effective April 4, 1951, the Commission temporarily waived the requirement of prior service as a radio operator or examination for renewal of his commercial radio operator license. This corder is applicable to commercial radio operator licenses which expired after June 30, 1950 until further order of the Commission.

CODE TESTS

§ 13.41 Transmitting speed requirements.

An applicant is required to transmit correctly in the International Morse code for 1 minute at the rate of speed prescribed in this part for the class of license desired.

§ 13.42 Transmitting test procedure.

Transmitting tests shall be performed by the use of the conventional Morse key except that a semi-automatic key, if furnished by the applicant, may be used in transmitting code tests of 25 words per minute.

§ 13.43 Receiving speed requirements.

An applicant is required to receive the International Morse Code by ear, and legibly transcribe, consecutive words or code groups for a period of 1 minute with-

out error at the rate of speed specified in the rules for the class of license for which the application is made.

§ 13.44 Receiving test procedure.

Receiving code tests shall be written in longhand either in ink or pencil except that in the case of the 25 words per minute code test a typewriter may be used when furnished by the applicant.

§ 13.45 Computing words or code groups.

Each five characters shall be counted as one word or code group. Punctuation marks or figures count as two characters.

SCOPE OF AUTHORITY

§ 13.61 Operating authority.

The various classes of commercial radio operator licenses issued by the Commission authorize the holders thereof to operate radio stations, except amateur, as follows (See also § 13.62 (c) for additional operating authority with respect to standard and FM broadcast stations):

(a) Radiotelegraph first-class operator

license. Any station except:

(1) Stations transmitting television, or (2) Any of the various classes of broadcast stations other than remote pickup

and ST broadcast stations, or

- (3) On a cargo vessel (other than a vessel operated exclusively on the Great Lakes) required by treaty or statute to be equipped with a radiotelegraph installation, the holder of this class of license may not act as chief or sole operator until he has had at least 6 months' satisfactory service in the aggregate as a qualified radiotelegraph operator in a station on board a ship or ships of the United States.
- (4) On an aircraft employing radiotelegraphy, the holder of this class of license may not operate the radiotelegraph station during the course of normal rendition of service unless he has satisfactorily completed a supplementary examination qualifying him for that duty, or unless he has served satisfactorily as chief or sole radio operator on an aircraft employing radiotelegraphy prior to February 15, 1950. The supplementary examination shall consist of:
- (i) Written examination element: 7.
 (5) At a ship radar station, the holder of this class of license may not supervise or be responsible for the performance of any adjustments or tests during or coincident with the installation, servicing or maintenance of the radar equip-

ment while it is radiating energy unless he has satisfactorily completed a supplementary examination qualifying him for that duty and received a ship radar endorsement on his license certifying to that fact: *Provided*. That nothing in this subparagraph shall be construed to prevent persons holding licenses not so endorsed from making replacements of fuses or of receiving-type tubes. The supplementary examination shall consist of:

(i) Written examination element: 8.

(b) Radiotelegraph second-class operator license. Any station except:

(1) Stations transmitting television, or

(2) Any of the various classes of broadcast stations other than remote pickup

and ST broadcast stations, or

- (3) On a passenger vessel (a ship shall be considered a passenger ship if it carries or is licensed or certificated to carry more than 12 passengers; a cargo ship means any ship not a passenger ship) required by treaty or statute to maintain a continuous radio watch by operators or on a vessel having continuous hours of service for public correspondence, the holder of this class of license may not act as chief operator, or
- (4) On a vessel (other than a vessel operated exclusively on the Great Lakes) required by treaty or statute to be equipped with a radiotelegraph installation, the holder of this class of license may not act as chief or sole operator until he has had at least 6 months' satisfactory service in the aggregate as a qualified radiotelegraph operator in a station on board a ship or ships of the United States.
- (5) On an aircraft employing radiotelegraphy, the holder of this class of license may not operate the radiotelegraph station during the course of normal rendition of service unless he is at least eighteen (18) years of age and has satisfactorily completed a supplementary examination qualifying him for that duty, or unless he has served satisfactorily as chief or sole radio operator on an aircraft employing radiotelegraphy prior to February 15, 1950. The supplementary examination shall consist of:
- (i) Transmitting and receiving code test at twenty-five (25) words per minute plain language and twenty (20) code groups per minute.
 - (ii) Written examination element: 7.
- (6) At a ship radar station, the holder of this class of license may not supervise or be responsible for the performance

of any adjustments or tests during or coincident with the installation, servicing or maintenance of the radar equipment while it is radiating energy unless he has satisfactorily completed a supplementary examination qualifying him for that duty and received a ship radar endorsement on his license certifying to that fact: *Provided*, That nothing in this subparagraph shall be construed to prevent persons holding licenses not so endorsed from making replacements of fuses or of receiving-type tubes. The supplementary examination shall consist of:

(i) Written examination element: 8.

(c) Temporary limited radiotelegraph operator license. Any ship station or ship radar station, subject to the following conditions and limitations:

(1) On a passenger vessel required by treaty or statute to maintain a continuous watch by operators or on a vessel having continuous hours of service for public correspondence, the holder of this class of license may not act as chief operator.

(2) On a vessel (other than a vessel operated exclusively on the Great Lakes) required by treaty or statute to be equipped with a radiotelegraph installation, the holder of this class of license may not act as chief or sole operator until he has had at least 6 months' satisfactory service in the aggregate as a qualified radiotelegraph operator in a ship or ships of the United States.

- (3) At a ship radar station licensed in the ship service, the holder of this class of license may not supervise or be responsible for the performance of any adjustments or tests during or coincident with the installation, servicing or maintenance of the radar equipment while it is radiating energy unless he has satisfactorily completed a supplementary examination qualifying him for that duty and received a ship radar endorsement on his license certifying to that fact: Provided, That nothing in this subparagraph shall be construed to prevent persons holding licenses not so endorsed from making replacements of fuses or of receiving type tubes. The supplementary examination shall consist of:
- (i) Written examination element: 8.
 (d) Radiotelegraph third class operator permit. Any station except:
- (1) Stations transmitting television, or
- (2) Any of the various classes of broadcast stations other than noncom-

mercial educational FM broadcast stations using transmitters with power ratings of 10 watts or less, remote pickup broadcast stations and ST broadcast stations, or

(3) Class I-B coast stations (other than when transmitting manual radio-telegraphy for identification or for testing) at which the power in the antenna of the unmodulated carrier wave is authorized to exceed 250 watts, or

(4) Class II-B or Class III-B coast stations (other than those in Alaska and other than when transmitting manual radiotelegraphy for identification or for testing) at which the power in the antenna of the unmodulated carrier wave is authorized to exceed 250 watts, or

(5) Ship stations or aircraft stations other than those at which the installation is used solely for telephony and at which the power in the antenna of the unmodulated carrier wave is not authorized to exceed 250 watts, or

(6) Ship stations and coast stations open to public correspondence by telegraphy, or

(7) Radiotelegraph stations on board a vessel required by treaty or statute to be equipped with a radio installation, or

(8) Aircraft stations while employing radiotelegraphy:

Provided, That (1) such operator is prohibited from making any adjustments that may result in improper transmitter operation, and (2) the equipment is so designed that the stability of the frequencies of the transmitter is maintained by the transmitter itself within the limits of tolerance specified by the station license, and none of the operations necessary to be performed during the course of normal rendition of the service of the station may cause off-frequency operation or result in any unauthorized radiation, and (3) any needed adjustments of the transmitter that may affect the proper operation of the station are regularly made by or under the immediate supervision and responsibility of a person holding a first- or second-class commercial radio operator license, either radiotelephone or radiotelegraph as may be appropriate for the class of station involved (as determined by the scope of the authority of the respective licenses as set forth in paragraphs (a), (b), (e), and (f) of this section and § 13.62), who shall be responsible for the proper functioning of the station equipment, and (4) in the case of ship radiotelephone or aircraft radiotelephone stations when the power in the antenna of the unmodulated carrier wave is authorized to exceed 100 watts, any needed adjustments of the transmitter that may affect the proper operation of the station are made only by or under the immediate supervision and responsibility of an operator holding a first- or second-class radiotelegraph license, who shall be responsible for the proper functioning of the station equipment.

(e) Radiotelephone first-class operator license. Any station except:

(1) Stations transmitting telegraphy by any type of the Morse Code, or

(2) Ship stations licensed to use telephony and power in excess of 100 watts for communication with Class I-B coast stations.

- (3) At a ship radar station, the holder of this class of license may not supervise or be responsible for the performance of any adjustments or tests during or coincident with the installation, servicing or maintenance of the radar equipment while it is radiating energy unless he has satisfactorily completed a supplementary examination qualifying him for that duty and received a ship radar endorsement on his license certifying to that fact: Provided, That nothing in this subparagraph shall be construed to prevent persons holding licenses not so endorsed from making replacements of fuses or The suppleof receiving-type tubes. mentary examination shall consist of:
 - (i) Written examination element: 8. (f) Radiotelephone second-class oper-

ator license. Any station except:

- (1) Stations transmitting telegraphy by any type of the Morse Code, or
- (2) Standard broadcast stations, or(3) International broadcast stations,
- (4) FM broadcast stations, or
- (5) Non-commercial educational FM broadcast stations with transmitter power rating in excess of 1 kilowatt, or
- (6) Television broadcast stations licensed for commercial operation, or
- (7) Ship stations licensed to use telephony and power in excess of 100 watts for communication with Class I-B coast stations.
- (8) At a ship radar station, the holder of this class of license may not supervise or be responsible for the performance of any adjustments or tests during or coincident with the installation, servicing or maintenance of the radar equipment while it is radiating energy unless he has satisfactorily com-

pleted a supplementary examination qualifying him for that duty and received a ship radar endorsement on his license certifying to that fact: Provided, That nothing in this subparagraph shall be construed to prevent persons holding licenses not so endorsed from making replacements of fuses or of receiving-type tubes. The supplementary examination shall consist of:

- (i) Written examination element: 8. (g) Radiotelephone third-class opera-
- tor permit. Any station except:
 (1) Stations transmitting television,
- (1) Stations transmitting television, or
- (2) Stations transmitting telegraphy by any type of the Morse code, or
- (3) Any of the various classes of broadcast stations other than non-commercial educational FM broadcast stations using transmitters with power ratings of 10 watts or less, remote pickup broadcast stations and ST broadcast stations or
- (4) Class I-B coast stations at which the power in the antenna of the unmodulated carrier wave is authorized to exceed 250 watts, or
- (5) Class II-B or Class III-B coast stations, other than those in Alaska at which the power in the antenna of the unmodulated carrier wave is authorized to exceed 250 watts, or
- (6) Ship stations or aircraft stations other than those at which the installation is used solely for telephony and at which the power in the antenna of the unmodulated carrier wave is not authorized to exceed 250 watts:

Provided, That (1) such operator is prohibited from making any adjustments that may result in improper transmitter operation, and (2) the equipment is so designed that the stability of the frequencies of the transmitter is maintained by the transmitter itself within the limits of tolerance specified by the station license, and none of the operations necessary to be performed during the course of normal rendition of the service of the station may cause off-frequency operation or result in any unauthorized radiation, and (3) any needed adjustments of the transmitter that may affect the proper operation of the station are regularly made by or under the immediate supervision and responsibility of a person holding a first- or second-class commercial radio operator license, either radiotelephone or radiotelegraph as may be appropriate

for the class of station involved (as determined by the scope of the authority of the respective licenses as set forth in paragraphs (a), (b), (e), and (f) of this section and § 13.62), who shall be responsible for the proper functioning of the station equipment, and (4) in the case of ship radiotelephone or aircraft radiotelephone stations when the power in the antenna of the unmodulated carrier wave is authorized to exceed 100 watts, any needed adjustments of the transmitter that may affect the proper operation of the station are made only by or under the immediate supervision and responsibility of an operator holding a first- or secondclass radiotelegraph license, who shall be responsible for the proper functioning of the station equipment.

- (h) Restricted radiotelephone operator permit. Any station except:
- (1) Stations transmitting television, or (2) Stations transmitting telegraphy by any type of the Morse Code, or
- (3) Any of the various classes of broadcast stations other than remote pickup, broadcast STL, and FM intercity relay stations, or
- (4) Ship stations licensed to use telephony for communication with Class I coast stations on frequencies between 4000 kc/s and 30 Mc/s, or
- (5) Radio stations provided on board vessels for safety purposes pursuant to statute or treaty, or
- (6) Coast stations, other than those in Alaska, while employing a frequency below 30 Mc/s, or
- (7) Coast stations at which the power in the antenna of the unmodulated carrier wave is authorized to exceed 250 watts:
- (8) At a ship radar station the holder of this class of license may not supervise or be responsible for the performance of any adjustments or tests during or coincident with the installation, servicing or maintenance of the radar equipment while it is radiating energy: Provided, That nothing in this subparagraph shall be construed to prevent any person holding such a license from making replacements of fuses or of receiving type tubes:

Provided, That, with respect to any station which the holder of this class of license may operate, such operator is prohibited from making any adjustments that may result in improper transmitter operation, and the equip-

ment is so designed that the stability of the frequencies of the transmitter is maintained by the transmitter itself within the limits of tolerance specified by the station license, and none of the operations necessary to be performed during the course of normal rendition of the service of the station may cause offfrequency operation or result in any unauthorized radiation, and any needed adjustments of the transmitter that may affect the proper operation of the station are regularly made by or under the immediate supervision and responsibility of a person holding a first- or second-class commercial radio operator license, either radiotelephone or radiotelegraph, who shall be responsible for the proper functioning of the station equipment.

[28 F.R. 12515, Nov. 22, 1963, as amended at 31 F.R. 12481, Sept. 21, 1966]

§ 13.62 Special privileges.

In addition to the operating authority granted under § 13.61, the following special privileges are granted the holders of commercial radio operator licenses:

(a) [Reserved]

- (b) The holder of any class of radiotelephone operator license, whose license authorizes him to operate a station while transmitting telephony, may operate the same station when transmitting on the same frequencies, any type of telegraphy under the following conditions:
- (1) When transmitting telegraphy by automatic means for identification, for testing, or for actuating an automatic selective signaling device, or
- (2) When properly serving as a relay station and for that purpose retransmitting by automatic means, solely on frequencies above 50 megacycles, the signals of a radiotelegraph station, or
- (3) When transmitting telegraphy as an incidental part of a program intended to be received by the general public, either directly or through the intermediary of a relay station or stations.
- (c) The holder of a commercial radio operator license of any class may operate broadcast stations under the following conditions:
- (1) A standard broadcast station with authorized operating power of 10 kw or less and employing a nondirectional antenna, an FM broadcast station with authorized transmitter output power of 10 kw or less, or a noncommercial educational FM broadcast station with authorized transmitter output power of

more than 1 kw but not in excess of 10 kw: Provided, That adjustments of transmitting equipment by such operators, except when under the immediate supervision of a radiotelephone first-class operator, and except as provided in paragraph (d) of this section, shall be limited to the following:

(i) Those necessary to commence or terminate transmitter emissions as 9

routine matter.

(ii) Those external adjustments that may be required as a result of variations of primary power supply.

(iii) Those external adjustments which may be necessary to insure modulation

within the limits required.

(iv) Those adjustments necessary to effect any changes in operating power which may be required by the station's instrument of authorization.

- (2) A noncommercial educational FM broadcast station with authorized transmitter power output of more than 10 watts but not in excess of 1 kw: Provided, That adjustments of transmitting equipment by such operators, except under the immediate supervision of a radiotelephone first- or second-class operator, shall be limited to those adjustments set forth in subparagraph (1) (i), (ii), and (iii) of this paragraph.
- (3) A noncommercial educational FM broadcast station with authorized transmitter power output of 10 watts or less: Provided, That adjustments of transmitting equipment by such operators, except under the immediate supervision of a radiotelephone first- or second-class operator or a radiotelegraph first- or second-class operator, shall be limited to those adjustments set forth in subparagraph (1), (i), (ii), and (iii) of this paragraph.
- (4) Should the broadcast transmitting apparatus be observed to be operating in a manner inconsistent with the station's instrument of authorization and none of the adjustments specifically described under subparagraph (1) (2) or (3) of this paragraph are effective in bringing it into proper operation, an operator holding a lesser grade license than that which authorizes unlimited adjustment, with respect to the class of broadcast station involved, and not acting under the supervision of a person holding the higher grade license permitting such unlimited adjustment, shall terminate the station's emissions.
- (5) Except in the case of noncommercial educational FM broadcast stations

with authorized transmitter output power of 10 watts or less, the special operating authority granted in this section with respect to broadcast stations is subject to the condition that there shall be in regular full-time employment at the station one or more operators of a class authorized to make or supervise all adjustments, whose primary duty shall be to effect and insure the proper functioning of the transmitting equipment. In the case of a noncommercial educational FM broadcast station with authorized transmitter output power of 10 watts or less such operator(s) shall nevertheless be available on call to make or supervise any needed adjustments.

(d) When an emergency action condition is declared, a person holding any class of radio operator license or permit who is authorized thereunder to perform limited operation of a standard broadcast station may make any adjustments necessary to effect operation in the emergency broadcast system in accordance with the station's National Defense Emergency Authorization: Provided. That the station's responsible first-class radiotelephone operator(s) shall have previously instructed such person in the adjustments to the transmitter which are necessary to accomplish operation in the Emergency Broadcast System.

§ 13.63 Operator's responsibility.

The licensed operator responsible for the maintenance of a transmitter may permit other persons to adjust a transmitter in his presence for the purpose of carrying out tests or making adjustments requiring specialized knowledge or skill, provided that he shall not be relieved thereby from responsibility for the proper operation of the equipment.

§ 13.64 Obedience to lawful orders.

All licensed radio operators shall obey and carry out the lawful orders of the master or person lawfully in charge of the ship or aircraft on which they are employed.

§ 13.65 Damage to apparatus.

No licensed radio operator shall willfully damage, or cause or permit to be damaged, any radio apparatus or installation in any licensed radio station.

§ 13.66 Unnecessary, unidentified, or superfluous communications.

No licensed radio operator shall transmit unnecessary, unidentified, or superfluous radio communications or signals.

§ 13.67 Obscenity, indecency, profanity.

No licensed radio operator or other person shall transmit communications containing obscene, indecent, or profane words, language, or meaning.

§ 13.68 False signals.

No licensed radio operator shall transmit false or deceptive signals or communications by radio, or any call letter or signal which has not been assigned by proper authority to the radio station he is operating.

§ 13.69 Interference.

No licensed radio operator shall willfully or maliciously interfere with or cause interference to any radio communication or signal.

§ 13.70 Fraudulent licenses.

No licensed radio operator or other person shall alter, duplicate, or fraudulently obtain, or assist another to alter, duplicate, or fraudulently obtain an operator license. Nor shall any person use a license issued to another or a license which he knows to have been altered, duplicated, or fraudulently obtained.

MISCELLANEOUS

§ 13.71 Issue of duplicate or replacement licenses.

(a) An operator whose license or permit has been lost, mutilated, or destroyed shall immediately notify the Commission. If the authorization is of the diploma form, a properly executed application for duplicate should be submitted to the office of issue. If the authorization is of the card form (Restricted Radiotelephone Operator Permit), a properly executed application for replacement should be submitted to the Federal Communications Commission, Gettys-burg, Pa., 17325. In either case the application shall embody a statement of the circumstances involved in the loss. mutilation, or destruction of the license or permit. If the authorization has been lost, the applicant must state that reasonable search has been made for it, and, further, that in the event it be found, either the original or the duplicate (or replacement) will be returned for cancellation. If the authorization is of the diploma form, the applicant should also submit documentary evidence of the service that has been obtained under the original authorization, or a statement embodying that information.

(b) The holder of any license or permit whose name is legally changed may make application for a replacement document to indicate the new legal name by submitting a properly executed application accompanied by the license or permit affected. If the authorization is of the diploma form, the application should be submitted to the office where it was issued. If the authorization is of the card form (Restricted Radiotelephone Operator Permit) it should be submitted to the Federal Communications Commission, Gettysburg, Pa., 17325.

[29 F.R. 15761, Nov. 24, 1964]

§ 13.72 Exhibiting signed copy of application.

When a duplicate or replacement operator license or permit has been requested, or request has been made for renewal upon service or for an endorsement or a verification card, the operator shall exhibit in lieu of the original document a signed copy of the application which has been submitted by him.

§ 13.73 Verification card.

The holder of an operator license or permit of the diploma form (as distinguished from such document of the card form) may, by filing a properly executed application accompanied by his license or permit, obtain a verification card (Form 758-F). This card may be carried on the person of the operator in lieu of the original license or permit when operating any station at which posting of an operator license is not required: *Provided*, That the license is readily accessible within a reasonable time for inspection upon demand by an authorized Government representative.

§ 13.74 Posting requirements for operator.

(a) Performing duties other than, or in addition to, service or maintenance, at two or more stations. The holder of any class of radio operator license or permit of the diploma form (as distinguished from the card form) who performs any radio operating duties, as contrasted with but not necessarily exclusive of service or maintenance duties, at two or more stations at which posting of his license or permit is required shall post at one such station his operator license or permit and shall post at all other such stations a duly issued verified statement (Form 759).

(b) Performing service or maintenance duties at one or more stations. The

holder of a radiotelephone or radiotelegraph first- or second-class radio operator license who performs, or supervises. and is responsible for service or maintenance work on any transmitter of any station for which a station license is required, shall post his license at the transmitter involved whenever the transmitter is in actual operation while service or maintenance work is being performed: Provided, That in lieu of posting his license, he may have on his person either his license or a verification card (Form 758-F): And provided further, That if he performs operating duties in addition to service or maintenance duties he shall, in lieu of complying with the foregoing provisions of this paragraph, comply with the posting requirements applicable to persons performing such operating duties, as set forth in paragraph (a) of this section, and in the rules and regulations applicable to each service.

(c) One or more verified statements (Form 759), as necessary, will be issued to the holder of a restricted radio-telephone operator permit (card form license) who because of an operator license posting requirement at one station would not otherwise be able to comply with a license posting requirement or to carry his permit on his person when so required at another station or stations.

§ 13.75 Record of service and maintenance duties performed.

In every case where a station log or service and maintenance records are required to be kept, and where service or maintenance duties are performed which may affect the proper operation of a station, the responsible operator shall sign and date an entry in the log of the station concerned, or in the station maintenance records if no log is required, giving:

(a) Pertinent details of all service and maintenance work performed by him or under his supervision;

(b) His name and address; and

(c) The class, serial number and expiration date of his license:

Provided, That the responsible operator shall not be subject to requirements of paragraphs (b) and (c) of this section in relation to a station, or stations of one licensee at a single location, at which he is regularly employed as an operator on a full time basis and at which his license is properly posted.

§ 13.76 Limitation on aircraft pilots.

Notwithstanding any other provisions of this part, a license issued to an alien aircraft pilot pursuant to section 303(1) of the Communications Act shall be valid only for such operation of radio stations on aircraft as is complementary to his functions and duties as a pilot.

SERVICE

§ 13.91 Endorsement of service record.

A station licensee, or his duly authorized agent, or the master of a vessel acting as the agent of a licensee, shall endorse the service record appearing on said operator license, showing the call letters and types of emission of the station operated, the nature and period of employment, and quality of performance of duty.

§ 13.92 Aviation service endorsement.

If the operator has operated more than three stations in the aviation service, the service may be shown by giving the name of the aviation chain or company in lieu of listing the call letters of the several stations.

§ 13.93 Service acceptability.

Credit will be allowed only for satisfactory service obtained under conditions that required the employment of licensed operators, or when obtained at United States Government stations.

§ 13.94 Statement in lieu of service endorsement.

The holder of a radiotelegraph firstor second-class operator license, or a
temporary limited radiotelegraph second-class operator license desiring an
endorsement to be placed thereon attesting to an aggregate of at least 6 months'
satisfactory service as a qualified operator on a vessel of the United States or
an applicant for a temporary limited
radiotelegraph second-class operator license under § 13.5(d) (3) may, in the
event documentary evidence cannot be
produced, submit to any office of the
Commission a statement under oath accompanied by the license to be endorsed,
embodying the following:

- (a) Names of ships at which employed:
 - (b) Call letters of stations;
 - (c) Types of emission used;
- (d) Type of service performed as follows:
- (1) Manual radiotelegraph operation only; and
 - (2) Transmitter control only; or

- (3) Combination of (1) and (2) running concurrently:
- (e) Whether service was satisfactory or unsatisfactory:
 - (f) Period of employment:
- (g) Name of master, employer, licensee, or his duly authorized agent.
- [28 F.R. 12515, Nov. 22, 1963, as amended at 31 F.R. 12481, Sept. 21, 1966]

APPENDIX

FILING FEES

(Note: The Commission's general fee schedule is set forth in Subpart G, Part I of this Chapter. The text of that portion of the general fee schedule which is pertinent to applications filed for commercial radio operator licenses is reprinted below.)

§ 1.1103 Payment of fees.

(a) Each application, filed on or after January 1, 1964, for which a fee is prescribed in this subpart, must be accompanied by a remittance in the full amount of the fee. In no case will an application be accepted for filing or processed prior to payment of the full amount specified. Applications for which no remittance is received, or for which an insufficient amount is received, may be returned to the applicant.

(b) Fee payments accompanying applications received in the Commission's offices in Washington, D.C., or in any of the Commission's field offices, should be in the form of a check or money order payable to the Federal Communications Commission. The Commission will not be responsible for cash sent through the mails. All fees collected will be paid into the U.S. Treasury as miscellaneous receipts in accordance with the provisions of Title V of the Independent Offices Appropriations Act of 1952 (5 U.S.C. 140).

(c) Receipts will be furnished upon request in the case of payments made in person, but no receipts will be issued for payments sent through the mails.

(d) Except as provided in §§ 1.1104 and 1.1105, all fees will be charged irrespective of the Commission's disposition of the application. Applications returned to applicants for additional information or corrections will not require an additional fee when resubmitted.

§ 1.1104 Return or refund of fees.

- (a) The full amount of any fee submitted will be returned or refunded, as appropriate, in the following instances:
- (1) Where no fee is required for the application filed.
- (2) Where the application is filed by an applicant who cannot fulfill a prescribed age requirement.
- (3) Where the application is filed for renewal without reexamination of an amateur or commercial radio operator license after the grace period has expired.
- (4) Where the applicant is precluded from obtaining a license by the provisions of section 303(1) or 310(a) of the Communications Act.

Sec.

15.70

mission.

- (5) Where circumstances beyond the control of the applicant, arising after the application is filed, would render a grant useless.
- (b) Payments in excess of an applicable fee will be refunded only if the overpayment exceeds \$2.
- § 1.1117 Schedule of fees for commercial radio operator examinations and licensing.
- (a) Except as provided in paragraphs (b) and (c) of this section, applications filed for commercial radio operator examinations and licensing shall be accompanied by the fees prescribed below:

Applications for new operator license: First-class license, either radiotelephone or radiotelegraph___ 85 Second-class license, either radiotelephone or radiotelegraph_____ Third-class permit, either radiotelephone or radiotelegraph_____ Restricted radiotelephone permit__ Application for renewal of operator li-2 cense__ Application for endorsement of operator license __. Application for duplicate license or for replacement license_____ 2 Application for provisional certificate for a radiotelephone third-class operator permit endorsed for broadcast use...

(b) No fee need accompany an application for a verification card (FCC Form 758-F) or for a verified statement (FCC Form 758-b).

(c) Whenever an application requests both an operator license and an endorsement the required fee will be the fee prescribed for the license document involved.

[31 F.R. 15, Jan. 4, 1966, as amended at 33 F.R. 914, Jan. 25, 1968]

PART 15—RADIO FREQUENCY DEVICES Subpart A—General

15.1 Basis of this part. 15.2 Scope of this part. General condition of operation. 15.3 15.4 General definitions. 15.5 Equipment available for inspection. 15.6 Information required by the Commission. 15.7 General requirement for restricted radiation devices. Prohibition against eavesdropping. 15.11 Subpart B-Incidental Radiation Devices 15.31 Operating requirements. Subpart C-Radio Receivers 15.61 Scope of this subpart. 15.63 Radiation interference limits. 15.65 All-channel television broadcast reception. 15.66 Exemption from all-channel requirements. 15.69 Certification of receivers.

Information to be filed with Com-

Sec. 15.71	Identification of ceivers.	ſ	certificated	re-
15.72	Date when certific	at	tion is require	d.
15.75	Measurement pro	ce	dure.	
15.81	Operation of radio	0 1	receivers aboa	rd s

ship.
15.82 Interference from a radio receiver.

Subpart D—Community Antenna Television Systems

15.161 Radiation from a community antenna television system.
 15.162 Demonstration of compliance.

15.163 Interference from a community antenna television system.

15.164 Responsibility for receiver generated interference.

15.165 Measurement of field strength.

Subpart E—Low Power Communication Devices

15.201 Frequencies of operation. 15.202 Radiation limitation below 1600

kc/s.

15.203 Alternative requirement for operation on frequencies between 160 and 190 kc/s.

15.204 Alternative requirement for operation on frequencies between 510 and 1600 kc/s.

15.205 Operation within the frequency band 26.97–27.27 Mc/s.

15.211 Operation above 70 Mc/s.

15.212 Telemetering devices and wireless microphones in the band 88-108 Mc/s.

15.213 Custom built telemetering devices.15.214 Alternative provisions for measuring

devices.

15.220 Eavesdropping prohibited.

15.220 Eavesdropping prohibited.15.221 Class B emission prohibited.

15.222 Interference from low power communication devices.

15.227 Certification requirements. 15.228 Location of certificate.

15.229 Date when certification is required.

15.235 Type approval.

15.236 Identification of type approved devices.

15.237 Changes in type approved equipment.

15.238 Withdrawal of certificate of type approval.

15.251 Certification of measuring device operating pursuant to § 15.214.

15.252 Content of certificate required by § 15.214.

15.253 Report of measurements for a device operating pursuant to § 15.214.

15.254 Identification of a device certificated under § 15.214.

AUTHORITY: The provisions of this Part 15 issued under secs. 4, 303, 48 Stat. 1066, 1082, as amended: 47 U.S.C. 154, 303. unless otherwise noted. Interpret or apply sec. 301, 48 Stat. 1081; 47 U.S.C. 301.

SOURCE: The provisions of this Part 15 appear at 28 F.R. 12521, Nov. 22, 1963, unless otherwise noted.

Subpart A-General

§ 15.1 Basis of this part.

(a) Section 301 of the Communications Act of 1934, as amended, provides for the control by the Federal Government over all the channels of interstate and foreign radio communication and further provides, in part, that no person shall use or operate apparatus for the transmission of energy. communications or signals by radio when the effects of such operation extend beyond state lines or cause interference with the transmission or reception of energy, communications, or signals, of any interstate or foreign character by radio, except under and in accordance with the Communications Act and a license granted under the provisions of that act. Restricted and incidental radiation devices emit radio frequency energy on frequencies within the radio spectrum and constitute a source of harmful interference to authorized radio communication services operating upon the channels of interstate and foreign communication unless precautions are taken which will prevent the creation of any substantial amount of such interference.

(b) Sections 303(s) and 330 of the Communications Act of 1934, as amended, provide that all television broadcast receivers shipped in interstate commerce, or imported from any foreign country, for sale or resale to the public, shall comply with the provisions of this chapter for capability of adequately receiving all television broadcast channels.

§ 15.2 Scope of this part.

(a) This part contains rules that set forth the conditions under which the operation of incidental and restricted radiation devices is considered to fall outside the purview of section 301 of the Communications Act which specifies when a station license is required as a condition for lawful operation.

(b) No incidental or restricted radiation device which fails to conform to the provisions of this part, or which causes harmful interference, may be operated without a station license. Unless such devices may be operated in accordance with the provisions of some other part of this chapter (see particularly Part 95, Citizens Radio Service), persons wishing to operate such devices in a manner inconsistent with this part will be required

to first secure an amendment of the Commission's rules to establish a licensed service providing for such operation and setting forth the technical and other limitations thereof: Provided. That in appropriate circumstances, when such a petition for rule making has been filed. the Commission may consider, prior to final action thereon, applications for Special Temporary Authorizations to operate stations on a developmental basis where it can be shown that such temporary operation would be in aid of a final determination as to whether the proposed rule should be adopted, and that such temporary operation would otherwise be in the public interest; and Provided further, That the Commission will, in exceptional situations, consider individual applications for licenses to operate incidental or restricted radiation devices, not conforming to the provisions of this part, where it can be shown that the proposed operation would be in the public interest, that it is for a unique type of station or for a type of operation which is incapable of establishment as a regular service, and that the proposed operation cannot feasibly be conducted under this part.

(c) This part also contains requirements, technical specifications, and certification procedure for the all-channel reception capability of television broadcast receivers.

§ 15.3 General condition of operation.

Persons operating restricted or incidental radiation devices shall not be deemed to have any vested or recognizable right to the continued use of any given frequency, by virtue of prior registration or certification of equipment. Operation of these devices is subject to the conditions that no harmful interference is caused and that interference must be accepted that may be caused by other incidental or restricted radiation devices, industrial, scientific or medical equipment, or from any authorized radio service.

§ 15.4 General definitions.

(a) Radio frequency energy. Electromagnetic energy at any frequency in the radio spectrum between 10 kc/s and 3,000,000 Mc/s.

(b) Harmful interference. Any emission, radiation or induction which endangers the functioning of a radionavigation service or of other safety services or seriously degrades, obstructs or repeatedly interrupts a radiocommunica-

tion service operating in accordance with

this chapter.

(c) Incidental radiation device. device that radiates radio frequency energy during the course of its operation although the device is not intentionally designed to generate radio frequency energy

- (d) Restricted radiation device. device in which the generation of radio frequency energy is intentionally incorporated into the design and in which the radio frequency energy is conducted along wires or is radiated, exclusive of transmitters which require licensing under other parts of this chapter and exclusive of devices in which the radio frequency energy is used to produce physical, chemical or biological effects in materials and which are regulated under the provisions of Part 18 of this chapter.
- (e) Community antenna television A restricted radiation device sustem. designed and used for the purpose of distributing television signals by means of conducted or guided radio frequency currents to a multiplicity of receivers outside the confines of a single building.

NOTE: The television signals that are distributed are modulated radio frequency signals and may be:

(a) Broadcast signals that have been re-

ceived and amplified,

(b) Broadcast signals that have been received and converted to another frequency,

- (c) Any other modulated radio frequency signals fed into the system.
- (f) Low power communication device. A low power communication device is a restricted radiation device, exclusive of those employing conducted or guided radio frequency techniques, used for the transmission of signs, signals (including control signals), writing, images and sounds or intelligence of any nature by radiation of electromagnetic energy.

Examples: Wireless microphone, phonograph oscillator, radio controlled garage door opener and radio controlled models.

- (g) Television broadcast receiver. Apparatus designed to receive television pictures broadcast simultaneously with sound.
- (h) Noise figure of a television broadcast receiver. The ratio of (1) the total noise power delivered by the receiver into its output termination when the noise temperature of its input termination is standard (290° K) at all fre-

quencies, to (2) the portion thereof engendered by the input termination.

NOTE: For a television broadcast receiver. portion (2) includes only that noise from the input termination which appears in the output via the principal frequency transformation and does not include spurious contributions such as those from imagefrequency transformation.

(i) Peak picture sensitivity for television broadcast receiver. The lowest input signal which results in standard picture test output when the receiver is tuned for maximum picture output.

Note: Standard picture test output for symmetrical sine-wave modulation shall be 20 volts peak-to-peak between the control elements of the picture tube.

(Secs. 1, 2, 76 Stat. 150, 151; 47 U.S.C. 303(s), 330)

§ 15.5 Equipment available for inspection.

Any equipment or device subject to the provisions of this part together with any license, certificate, notice of registration or any technical data required to be kept on file by the operator of the device shall be made available for inspection by Commission representatives upon reasonable request.

§ 15.6 Information required by the Commission.

The owner or operator of any device subject to this part shall promptly furnish to the Commission or its representative such information as may be requested concerning the operation of the device, including a copy of any measurements made for the purpose certification.

§ 15.7 General requirement for restricted radiation devices.

Unless regulated under some other subpart of this part, any apparatus which generates a radio frequency electromagnetic field functionally utilizing a small part of such field in the operation of associated apparatus not physically connected thereand at a distance not greater ${\rm than} {157,000 \over F(kc/s)} {\rm feet} \Big({
m equivalent \ to \ } {\lambda \over 2\pi}$ not be licensed provided:

(a) That such apparatus shall be operated with the minimum power possible to accomplish the desired purpose.

(b) That the best engineering principles shall be utilized in the generation of radio frequency currents so as to guard against interference to established radio services, particularly on the fundamental and harmonic frequencies.

(c) That in any event the total electromagnetic field produced at any point a distance of $\frac{157,000}{F(kc/s)}$ feet (equivalent to

 $\frac{\lambda}{2\pi}$) from the apparatus shall not exceed 15 microvolts per meter.

(d) That the apparatus shall conform to such engineering standards as may from time to time be promulgated by the Commission.

(e) That in the event harmful interference is caused, the operator of the apparatus shall promptly take steps to eliminate the harmful interference.

Note: Radio receivers, community antenna television systems, and low power communication devices are regulated elsewhere in Part 15 and are not regulated by this section.

§ 15.11 Prohibition against eavesdropping.

- (a) No person shall use, either directly or indirectly, a device operated pursuant to the provisions of this part for the purpose of overhearing or recording the private conversations of others unless such use is authorized by all of the parties engaging in the conversation.
- (b) Paragraph (a) of this section shall not apply to operations of any law enforcement officers conducted under lawful authority.

[31 F.R. 3400, Mar. 4, 1966]

Subpart B—Incidental Radiation Devices

§ 15.31 Operating requirements.

An incidental radiation device shall be operated so that the radio frequency energy that is radiated does not cause harmful interference. In the event that harmful interference is caused, the operator of the device shall promptly take steps to eliminate the harmful interference.

Subpart C—Radio Receivers § 15.61 Scope of this subpart.

Radio receivers come within the scope of this subpart insofar as they are restricted radiation devices and generate and radiate radio frequency energy or are designed to receive television pictures broadcast simultaneously with sound. Typically this subpart limits oscillator radiation from superheterodyne receivers. In the case of television broadcast receivers, this subpart also limits the radiation of radio frequency energy which is generated by the sweep circuits, and requires that such receivers be capable of adequately receiving all television broadcast channels.

(Secs. 1, 2, 76 Stat. 150, 151; 47 U.S.C. 303(s), 330)

§ 15.63 Radiation interference limits.

(a) The radiation from all radio receivers that operate (tune) in the range 30 to 890 Mc/s, including frequency modulation broadcast receivers and television broadcast receivers, manufactured after the effective date specified in § 15.72 shall not exceed the following field strength limits at a distance of 100 feet or more from the receiver:

Frequency of radiation (Mc/s)	Field strength (uv/m)
0.45 up to and including 25 Over 25 up to and including 70 Over 70 up to and including 130 130-174	See paragraph (b). 32. 50. 50-150 (linear inter-
174-260260-470	polation). 150. 150-500 (linear in-
470-1000	terpolation). 500 (see paragraph (c) below).

(b) Pending the development of suitable measurement techniques for measuring the actual radiation in the band 0.45 to 25 Mc/s, the interference capabilities of a receiver in this band will be determined by the measurement of radio frequency voltage between each power line and ground at the power terminals of the receiver. This requirement applies only to radio receivers intended to be connected to power lines of public utility systems. For television broadcast receivers the voltage so measured shall not exceed 100 uv at any frequency between 450 kc/s and 25 Mc/ inclusive. For all other receivers the voltage shall not exceed 100 uv at any frequency between 450 kc/s and 9 Mc/s inclusive. 1000 uv for frequencies between 10 Mc/s and 25 Mc/s and linear increase from 100 uv to 1000 uv for frequencies between 9 Mc/s and 10 Mc/s.

(c) For television broadcast receivers the limit 500 μ V/m is temporarily increased to 1,000 μ V/m until January 31, 1970.

[28 F.R. 12520, Nov. 22, 1963, as amended at 34 F.R. 7497, May 9, 1969]

§ 15.65 All-channel television broadcast reception.

(a) Except as provided in § 15.66, all television broadcast receivers manufactured after April 30, 1964, and shipped in interstate commerce or imported from any foreign country into the United States, for sale or resale to the public, shall be capable of adequately receiving all channels allocated by the Commission to the television broadcast service. A television broadcast receiver shall be considered to be capable of adequately receiving all channels if it meets the specifications in this section for noise figure and peak picture sensitivity.

 (b) The noise figure of any television broadcast receiver subject to paragraph
 (a) of this section, for any television channel between 14 and 83 inclusive,

shall not be larger than 18 db.

(c) The peak picture sensitivity of any television broadcast receiver subject to paragraph (a) of this section, averaged for all channels between 14 and 83 inclusive, shall not be more than 8 db larger than the peak picture sensitivity of that receiver averaged for all television channels between 2 and 13 inclusive. (Interpret or apply secs. 1, 2, 76 Stat. 150, 151; 47 U.S.C. 303(s), 330)

§ 15.66 Exemption from all-channel requirement.

Subject to the following conditions, television broadcast receivers manufactured on or before April 30, 1966, which are not capable of receiving all channels allocated by the Commission television broadcasting may be shipped in interstate commerce or imported into the United States:

(a) The television receiver is in transport to a specific educational institution, for use exclusively in that institution, and in connection with an existing or planned in-school educational television instruction program; and

(b) The television receiver is permanently identified (through stenciling, etching, raised lettering or other similarly appropriate means), and the shipping carton is identified, as follows:

FOR IN-SCHOOL USE ONLY

(Interpret or apply secs. 1, 2, 76 Stat. 150, 151; 47 U.S.C. 303(s), 330)

§ 15.69 Certification of receivers.

(a) (1) No receiver manufactured after the effective dates of this subpart (see § 15.72) that operates in the range 30 to 890 Mc/s, including frequency modulation broadcast receivers and television broadcast receivers, shall be operated without a station license unless it has been certificated to demonstrate compliance with the radiation interference limits set forth in § 15.63.

(2) Except as provided in § 15.66, no television broadcast receiver manufactured after April 30, 1964, shall be shipped in interstate commerce or imported from any foreign country into the United States, for sale or resale to the public, unless it has been certificated to comply with the noise figure and peak picture sensitivity requirements set forth in § 15.65. (This provision does not apply to carriers which transport television broadcast receivers without trading in them.)

(b) The owner or operator need not certificate his own receiver as meeting the radiation interference limits of § 15.63 if it has been certificated by the manufacturer or the distributor.

(c) Certification made by the manufacturer or the distributor shall be based on tests made on receivers actually produced for sale. Tests shall be performed on a sufficient number of production units to assure that all production units comply with the requirements of this subpart.

(d) The certificate may be executed by an engineer skilled in making and interpreting such measurements as are required.

(e) The certificate shall contain the following information:

(1) Name of the manufacturer or distributor of the receiver.

(2) Model number.

(3) Brief description of receiver, including tuning range, type of circuit, and purpose for which used (as broadcast, aircraft, etc.).

(4) Brief statement of the measurement procedure used.

Note: If a standard procedure is used, it is sufficient to identify the standard. Other measurement procedures must be described in detail.

- (5) Date the measurements were made.
 - (6) A summary of the data obtained.

Note: For television broadcast receivers the measured noise figure and peak picture sensitivity must be stated.

(7) A statement certifying that, on the basis of measurements made, the radio receiver is capable of complying with the



requirements of this part under normal operation with the usual maintenance.

- (8) The name and address of the certifying engineer, and name and address of his employer, if any.
 - (9) Date of the certificate.
- (f) The certificate shall be retained by the owner, the manufacturer, or the distributor (as may be appropriate) for a period of 5 years, and shall be made available, upon reasonable request, to an authorized Commission representative, or photostat furnished by mail. (See § 15.70 for filing requirement with FCC.)

(Secs. 1, 2, 76 Stat. 150, 151; 47 U.S.C. 303(s), 830)

§ 15.70 Information to be filed with Commission.

- (a) Each manufacturer, distributor or other certifying agency that issues certifications pursuant to this subpart shall file with the Commission a description of its measurement facilities used for certification.
- (b) A copy of each certificate prepared by a manufacturer, distributor or certifying agency shall be filed with the Commission at the time the certificate is prepared.
- (c) This information is not open to public inspection.

§ 15.71 Identification of certificated receivers.

- (a) Each certificated receiver shall be identified by a distinctive seal or label, which may be a part of the name plate and which shall state that the receiver has been certificated for compliance with the requirements of this subpart. The seal or label shall be permanently attached to the receiver and shall be readily visible for inspection by prospective purchasers.
- (b) After January 1, 1964, cartons containing television broadcast receivers shipped in interstate commerce or imported into the United States, for sale or resale to the public, shall clearly indicate either (1) that the receiver within was manufactured on or before April 30, 1964, or (2) that the receiver within meets all FCC requirements pertaining to all-channel reception (see § 15.65).
- (c) For identification of receivers exempted from the all-channel requirement, see § 15.66.

(Interpret or apply secs. 1, 2, 76 Stat. 150, 151; 47 U.S.C. 303(s), 330)

§ 15.72 Date when certification is required.

- (a) For television broadcast receivers.

 (1) VHF television broadcast receivers manufactured after May 1, 1956, shall comply with the certification requirements with respect to radiation of radio frequency energy, except that compliance with the power line interference limits for frequencies between 3 Mc/s and 25 Mc/s is required for such receivers manufactured after December 31, 1957.
- (2) UHF television broadcast receivers manufactured after December 31, 1957, shall comply with the certification requirements with respect to radiation of radio frequency energy.
- (3) Except as provided in § 15.66, all television broadcast receivers manufactured after April 30, 1964, shall comply with the certification requirement with respect to adequacy of all-channel reception.
- (b) For other receivers. All radio receivers other than television broadcast receivers that operate (tune) in the range 30 to 890 Mc/s manufactured after October 1, 1956, shall comply with the certification requirements with respect to radiation of radio frequency energy, except as follows:
- (1) FM broadcast receivers manufactured after December 31, 1956, shall comply with the certification requirements with respect to frequencies above 25 Mc/s. All such receivers manufactured after December 31, 1957, shall comply with the certification requirements with respect to all frequencies.
- (2) The radiation interference limits and the certification requirement with respect thereto shall be met by all pocket type super-regenerative receivers used in the one-way signalling services as defined in Part 21 of this chapter which are manufactured after December 31, 1956.

(Interpret or apply secs. 1, 2, 76 Stat. 150, 151; 47 U.S.C. 303(s), 330) [28 F.R. 12521, Nov. 22, 1963, as amended at 29 F.R. 2558, Feb. 19, 1964]

§ 15.75 Measurement procedure.

(a) Any measurement procedure acceptable to the Commission may be used to show compliance with the requirements of this subpart. A detailed description of the proposed measurement procedure, including a list of the test equipment to be used, shall be submitted to the Commission when requesting a

determination regarding the acceptability of the proposed measurement procedure.

- (b) The following methods of measurement are considered acceptable procedures for certification of receivers pursuant to § 15.69:
- (1) Institute of Electrical and Electronics Engineers Standard 187 (formerly 51 IRE 17S1) for radiation measurements
- (2) Institute of Electrical and Electronics Engineers Standard 213 (formerly 61 IRE 2781) for conducted interference measurements from frequency modulated and television broadcast receivers in the range 300 kc/s to 25 Mc/s.

(3) Institute of Electrical and Electronics Engineers Standard 190 (formerly 60 IRE 17S1) for measurement of noise figure and peak picture sensitivity of a television broadcast receiver.

(4) International Electrotechnical Commission Publication No. 106 (1959) and Supplement 106A (1962) for measurement of radiated interference from broadcast receivers. (A conversion factor of 0.1 (-20 db) shall be applied to the measured values for comparison with the limits of § 15.63.)

Note: This publication and supplement may be purchased from the United States of America Standards Institute (formerly American Standards Association), 10 East 40th Street, New York, N.Y. 10016.

(c) In the case of measurements in the field, radiation in excess of 15 uv/m at any frequency between 450 kc/s and 25 Mc/s at the border of the property and more than 15 feet from any power line crossing this border under the control and exclusive use of the person operating or authorizing the operation of the receiver will be considered an indication of noncompliance with the radiation requirements of this subpart.

[28 F.R. 12521, Nov. 22, 1963, as amended at 30 F.R. 578, Jan. 16, 1965; 31 F.R. 13391, Oct. 15, 1966]

§ 15.81 Operation of radio receivers aboard a ship.

In addition to meeting the requirements of this part, a radio receiver operated aboard a ship shall also meet the requirements of Part 83 of this chapter.

§ 15.82 Interference from a radio receiver.

The operator of a radio receiver, regardless of tuning range, date of manu-

facture, or of certification, which causes harmful interference shall promptly take steps to eliminate the harmful interference

Subpart D—Community Antenna Television Systems

§ 15.161 Radiation from a community antenna television system.

Radiation from a community antenna television system shall be limited as follows:

	Dis-	Radiation limits (uv/m)	
Frequencies (Mc/s)	tance (ft.)	General require- ment	Sparsely inhabited areas i
Up to and including 54	100	15	15
Over 54 up to and includ- ing 132	10	20	400
over 216	10 100	50 15	1,000- 15

¹For the purpose of this section, a sparsely inhabitated area is that area within 1,000 feet of a community antenna television system where television broadcast signals are, in fact, not being received directly from a television broadcast station.

§ 15.162 Demonstration of compliance.

The operator of each CATV system shall be responsible for insuring that each such system is designed, installed and operated in a manner which fully complies with the provisions of this subpart. Each system operator shall be prepared to show, upon reasonable demand by an authorized representative of the Commission, that the system does, in fact, comply with the rules.

§ 15.163 Interference from a community antenna television system.

In the event that the operation of a community antenna television system causes harmful interference to reception of authorized radio stations the operator of the system shall immediately take whatever steps are necessary to remedy the interference.

§ 15.164 Responsibility for receiver generated interference.

Interference originating in a radio receiver shall be the responsibility of the receiver operator in accordance with the provisions of Subpart C of this part: Provided, however, That the op-

erator of the community antenna television system to which the receiver is connected shall be responsible for the suppression of receiver generated interference that is distributed by the system when this interference is conducted into the system at the receiver.

§ 15.165 Measurement of field strength.

Measurements to determine the field strength of radio frequency energy generated by community antenna television systems shall be made in accordance with standard engineering procedures. Measurements made above 25 Mc/s shall include the following:

(a) A field strength meter using a horizontal dipole antenna shall be employed.

(b) Field strength shall be expressed in terms of the RMS value of synchro-

- nizing peak.
 (c) The dipole antenna shall be placed 12 feet above the ground and positioned directly below the system components. Where such placement results in a separation of less than 10 feet between the center of the dipole antenna and the system components, the dipole shall be repositioned to provide a separation of 10 feet.
- (d) The horizontal dipole antenna shall be rotated about a vertical axis and the maximum meter reading shall be used.
- (e) Measurements shall be made where other conductors are 10 or more feet away from the measuring antenna.

Subpart E-Low Power Communication Devices

§ 15.201 Frequencies of operation.

- (a) A low power communication device may be operated on any frequency in the bands 10-490 kc/s, 510-1600 kc/s, and 26.97-27.27 Mc/s.
- (b) Other frequencies above 70 Mc/s may be used for operations of short duration in accordance with the requirements set forth in § 15.211.
- (c) Telemetering devices and wireless microphones may be operated in the band 88-108 Mc/s in accordance with the provisions of § 15.212.
- (d) A low power communication device used for measurement of the characteristics of materials may be operated on frequencies and under the alternative provisions listed in § 15.214.

[28 F.R. 12521, Nov. 22, 1963, as amended at 34 F.R. 17172, Oct. 23, 1969]

§ 15.202 Radiation limitation below 1600 kc/s.

A low power communication device which operates on any frequency between 10 and 490 kc/s or between 510 and 1600 kc/s shall limit the radiation so that the field strength does not exceed the value specified in the following table:

Frequency (kc/s)	Distance (feet)	Field strength (uv/m)
10-490	1,000	2400 F(kc/s)
510-1600	100	24000 F(kc/s)

§ 15.203 Alternative requirement for operation on frequencies between 160 and 190 kc/s.

In lieu of meeting the radiation limitation, stated in § 15.202, a low power communication device operating on a frequency between 160 and 190 kc/s need only meet the following requirements:

- (a) The power input to the final radio frequency stage (exclusive of filament or heater power) does not exceed one
- (b) All emissions below 160 kc/s or above 190 kc/s are suppressed 20 db or more below the unmodulated carrier.
- (c) The total length of the transmission line plus the antenna does not exceed 50 feet.
- § 15.204 Alternative requirement for operation on frequencies between 510 and 1600 kc/s.

In lieu of meeting the radiation limitation stated in § 15.202, a low power communication device operating on a frequency between 510 and 1600 kc/s inclusive need only meet the following requirements:

(a) The power input to the final radio stage (exclusive of filament or heater power) does not exceed 100 milliwatts.

(b) The emissions below 510 kc/s or above 1600 kc/s are suppressed 20 db or more below the unmodulated carrier.

- (c) The total length of the transmission line plus the antenna does not exceed 10 feet.
- (d) Low power communication devices obtaining their power from the lines of public utility systems shall limit the radio frequency voltage appearing on each power line to 200 microvolts or less on any frequency from 510 kc/s to

1600 kc/s. Measurements shall be made from each power line to ground both with the equipment grounded and with the equipment ungrounded.

Note: One method of determining radio frequency voltage on the power line is described in "Military Specification for Interference Measurement" MIL-16910 (SHIPS) dated January 14, 1952, available from the Commanding Officer, Naval Supply Depot, Scotia, New York, 12302. Note that this procedure calls for grounding the equipment under test, whereas the Commission's rules call for measurements both with the equipment grounded and with the equipment ungrounded.

§ 15.205 Operation within the frequency band 26.97-27.27 Mc/s.

A low power communication device may operate within the band 26.97-27.27 Mc/s $(27.12 \text{ Mc/s}\pm150 \text{ kc/s})$ provided it complies with all of the following requirements:

(a) The carrier of the device shall be maintained within the band 26.97-27.27

Mc/s.

(b) All emissions, including modulation products, below 26.97 Mc/s or above 27.27 Mc/s shall be suppressed 20 db or more below the unmodulated carrier.

(c) The power input to the final radio stage (exclusive of filament or heater power) shall not exceed 100 milliwatts.

(d) The antenna shall consist of a single element that does not exceed 5 feet in length.

§ 15.211 Operation above 70 Mc/s.

(a) Except for telemetering devices and wireless microphones operated in accordance with the requirements of §§ 15.212 and 15.213, a low power communication device, manufactured on or after July 15, 1963 may be operated on frequencies above 70 Mc/s, provided it complies with all of the following conditions:

(1) The radiated field on any frequency from 70 Mc/s up to and including 1000 Mc/s does not exceed the limits specified for receivers in § 15.63.

(2) The radiated field on any frequency above 1000 Mc/s does not exceed 500 microvolts per meter at a distance of 100 feet.

(3) The device is provided with means for automatically limiting operation so that the duration of each transmission shall not be greater than one second and the silent period between transmissions shall not be less than 30 seconds.

(4) The device shall be so constructed that there are no external or readily ac-

cessible controls which may be adjusted to permit operation in a manner inconsistent with the provisions of this paragraph.

(5) Radio controls for door openers are exempted from the duty cycle limitation of subparagraph (3) of this paragraph: *Provided*, the transmitter part of the control may be activated only by a switch which turns the transmitter off when released.

(6) Radiation from the transmitter or associated receiver of radio controls for door openers must not fall within any of the following bands:

Mc/s	Mc/s	Gc/s
73 - 75. 4 108 -118 121. 4-121. 6 242. 8-243. 2 265 -285 328. 6-335. 4 404 -406	608- 614 960-1215 1400-1427 1535-1670 2690-2700 4200-4400 4990-5250	10. 68-10. 70 15. 35-15. 4 19. 3 -19. 4 31. 3 -31. 5 88 -90

(b) Except for radio controls for door openers and for telemetering devices and wireless microphones operated in accordance with the requirements of §§ 15.212 and 15.213, a low power communications device, manufactured before July 15, 1963, may be operated on any frequency above 70 Mc/s: Provided, It complies with all of the following conditions:

(1) The radiated field on any frequency from 70 Mc/s up to and including 1000 Mc/s does not exceed the limits specified for receivers in § 15.63.

(2) The radiated field on any frequency above 1000 Mc/s does not exceed 500 microvolts per meter at a distance

of 100 feet.

(3) The device is provided with means for automatically limiting operation to a duration of not more than one second, not to occur more than once in 30 seconds.

[28 F.R. 12521, Nov. 22, 1963, as amended at 30 F.R. 9316, July 27, 1965]

§ 15.212 Telemetering devices and wireless microphones in the band 88-108 Mc/s.

(a) Operation in the band 88-108 Mc/s is limited to low power communication devices employed as telemetering devices or as wireless microphones. This band shall not be used for two way communication.

(b) Users of these devices shall take adequate precautions to insure that

harmful interference is not caused to the reception of transmissions from any FM or television broadcast station or any other class of station licensed by the Commission. In the event that such interference does occur, operation of the telemetering device or wireless microphone shall be promptly suspended and shall not be resumed until the interference has been eliminated. Users of these devices must accept any interference which may be caused by the operation of any licensed station operating in accordance with the terms of its license.

- (c) Emissions from the device shall be confined within a band 200 kc/s wide centered on the operating frequency. Such 200 kc/s band shall lie wholly within the frequency range 88-108 Mc/s.
- (d) The field strength of emissions radiated within the specified 200 kc/s band shall not exceed 50 uv/m at a distance of 50 feet or more from the device.
- (e) The field strength of emissions radiated on any frequency outside the specified 200 kc/s band shall not exceed 40 uv/m at a distance of 10 feet or more from the device.
- (f) Except as provided in § 15.213, no such device shall be operated unless it has been type approved pursuant to § 15.235.
- (g) No antenna other than that furnished by the manufacturer shall be used with any type approved device.

§ 15.213 Custom built telemetering devices.

Custom built telemetering devices used for experimentation by an educational institution need not be type approved, *Provided:*

- (a) The device complies with the technical requirements of § 15.212;
- (b) The device has been certificated pursuant to §§ 15.227 and 15.228; and
- (c) The educational institution notifies the Engineer-in-Charge of the local FCC office, in writing, in advance of operation. The notice shall include:
- (1) The dates and place where the device will be operated;
- (2) The purpose for which the device will be used:
- (3) A description of the device including the operating frequency, RF power output, and antenna; and
- (4) A statement certifying that the device complies with the technical provisions of § 15.212.

§ 15.214 Alternative provisions for measuring devices.

- (a) A low power communication device used for measurement of the characteristics of materials may operate in the frequency bands listed in paragraph (c) pursuant to the provisions in this section.
- (b) A device operated pursuant to the alternative provisions of this section may not be used for voice communications, or the transmission of any other type message.
- (c) The device shall operate within the frequency bands:

MHz	MHz
13.554-13.566	890-940
26.96-27.28	(See note)
40.66-40.70	2400-2500
	5725-5875
	22000-22250

NOTE: The frequency band 890-940 MHz is subject to change pursuant to the reallocation of frequencies that may be made in the band 806-960 MHz in the rule making proceeding in Docket No. 18262.

(d) The maximum level of emission from the device shall not exceed:

Fundamental	Emission (uv/m at 100 feet)		
frequency in the band	On funda- mental frequency	On harmonic frequencies	On other frequencies
13.554-13.566 MHz 26.96-27.28 MHz 40.66-40.70 MHz above 890 MHz		0. 5 1. 0 1. 5 50. 0	0. 5 1. 0 1. 5 15. 0

- (e) The device shall be self-contained with no external or readily accessible controls which may be adjusted to permit operation in a manner inconsistent with the provisions of this section. Any antenna that may be used with the device shall be permanently attached thereto and shall not be readily modifiable by the user.
- (f) The device shall be prototype certificated pursuant to §§ 15.251–15.254 inclusive.

[34 F.R. 17172, Oct. 23, 1969]

§ 15.220 Eavesdropping prohibited.

As provided in § 15.11, the use of a low power communication device for eavesdropping is prohibited.

[31 F.R. 3400, Mar. 4, 1966]

§ 15.221 Class B emission prohibited.

Operation of low power communication devices that produce Class B emissions (damped waves) is prohibited.

§ 15.222 Interference from low power communication devices.

Notwithstanding the other requirements of this part, the operator of a low power communication device, regardless of date of manufacture, which causes harmful interference to an authorized radio service, shall promptly stop operating the device until the harmful interference has been eliminated.

§ 15.227 Certification requirements.

- (a) Except for telemetering devices and wireless microphones which have been type approved pursuant to § 15.235, no low power communication device manufactured after the dates set forth in § 15.229 shall be operated without a station license unless it has been certificated to demonstrate compliance with the requirements in this part.
- (b) The owner or operator need not certificate his own low power communication device, if it has been certificated by the manufacturer or distributor.
- (c) Where certification is based on measurement of a prototype, a sufficient number of units shall be tested to assure that all production units comply with the technical requirements of this subpart.
- (d) The certificate may be executed by a technician skilled in making and interpreting the measurements that are required to assure compliance with the requirements of this part.
- (e) The certificate shall contain the following information:
- (1) The operating conditions under which the device is intended to be used.
- (2) The antenna to be used with the device.
- (3) A statement certifying that the device can be expected to comply with the requirements of this subpart under the operating conditions specified in the certificate.
- (4) The month and year in which the device was manufactured.

§ 15.228 Location of certificate.

The certificate shall be permanently attached to the device and shall be readily visible for inspection.

§ 15.229 Date when certification is required.

All low power communication devices which operate on frequencies of 70 Mc/s or above, manufactured after June 30, 1958, shall comply with the type approval

or certification requirements of this subpart. All low power communication devices which operate on frequencies below 70 Mc/s, manufactured after December 31, 1957, shall comply with the certification requirements of this subpart.

§ 15.235 Type approval.

- (a) A manufacturer of a telemetering device or wireless microphone who desires to obtain type approval for his equipment may request permission to submit such equipment to the Commission for testing by following the procedure set out in Subpart F of Part 2 of this chapter, as modified by this section. The manufacturer shall furnish the following with his request for type approval:
- (1) A report of measurements showing that the equipment is capable of complying with the requirements of § 15.212;
- (2) A statement that at least 10 units are proposed to be manufactured; and
- (3) A statement agreeing to include a reprint of Subparts A and E of this Part 15, current as of date of manufacture, with each unit offered for sale or resale to the public.
- (b) To receive type approval, telemetering devices and wireless microphones must meet the following requirements:
- (1) The device must comply with the technical limitations of § 15.212.
- (2) The design and construction of the equipment must give reasonable assurance of compliance with the requirements of § 15.212 for at least five years under normal operation and with average maintenance.
- (3) The device must be so constructed that the adjustment of any control accessible to the user shall not cause operation in violation of § 15.212.

§ 15.236 Identification of type approved devices.

The Commission will assign a type approval number to each telemetering device or wireless microphone which is type approved. The type approval number and the following statement shall be permanently inscribed upon or permanently attached to each production unit as follows:

FCC Type Approval No. _____

Valid only when operated pursuant to FCC Rules, Part 15, and when used with antenna furnished by manufacturer.



§ 15.237 Changes in type approved equipment.

No changes whatsoever may be made in a type approved telemetering device or wireless microphone, including the antenna, except on specific prior approval by the Commission.

§ 15.238 Withdrawal of certificate of type approval.

- (a) A certificate of type approval may be withdrawn if the type of equipment for which it was issued proves defective in service and under usual conditions of maintenance and operation such equipment cannot be relied on to meet the conditions set forth in this part for the operation of the type of equipment involved, or if any change whatsoever is made in the construction of equipment sold under the certificate of type approval issued by the Commission, without the specific prior approval of the Commission.
- (b) The procedure for withdrawal of the certificate of type approval shall be the same as that prescribed for revocation of a radio station license pursuant to the provisions of the Communications Act of 1934, as amended.

(c) In the case of withdrawal of a certificate of type approval, the manufacturer shall make no further sale of equipment under such certificate.

(d) When a certificate of type approval has been withdrawn for unauthorized changes or for failure to comply with technical requirements, the Commission will consider that fact in determining whether the manufacturer in question is eligible to receive any new certificate of type approval.

§ 15.251 Certification of measuring device operating pursuant to § 15.214.

- (a) A device operating pursuant to § 15.214 need not be certificated by the owner or user if the device has been certificated by the manufacturer.
- (b) Where certification is based on measurement of a prototype, a sufficient number of units shall be tested to insure that all production units can be reasonably expected to comply with the applicable technical requirements.
- (c) The certificate shall be filed with the FCC, Washington, D.C. 20554.
- (d) The certificate filed by the manufacturer will be available for public inspection pursuant to the provisions of §§ 0.457 and 0.461 of this chapter.
 [34 F.R. 17172, Oct. 23, 1969]

§ 15.252 Content of certificate required by § 15.214.

(a) The manufacturer, model, and serial number(s) or other positive identification of the device that was tested.

(b) Photographs of the device.

- (c) A description of the circuitry and how the device operates.
- (d) The conditions under which the device shall be operated.
- (e) The antenna, if any, to be used with the device.
- (f) A report of measurements pursuant to § 15.253.
- (g) If filed by manufacturer, a statement certifying that production will be adequately controlled to insure that all units produced can be reasonably expected to comply with the applicable technical requirements.
- (h) If filed by manufacturer, a copy of the installation and operating instructions provided to the user.
 - (i) Date of certificate.
- (j) Signature. If filed by the manufacturer, the certificate shall be signed by a responsible official, who shall state that he is authorized to sign for the manufacturer and shall indicate his title. [34 F.R. 17172, Oct. 23, 1969]

§ 15.253 Report of measurements for a device operating pursuant to § 15.214.

The report of measurements may be prepared by any engineer skilled in making and interpreting the measurements that are required and shall contain the following information.

- (a) Identification of the device(s) tested.
- (b) List of measuring equipment used showing manufacturer, model number and date when last calibrated.
- (c) Description of measurement procedure used. If a published standard was followed, reference to the standard is sufficient, provided any departure from such standard is described in detail.
- (d) Report of the measurements obtained on the fundamental, and on harmonic and other spurious signals emitted by the device. For this measurement, the frequency spectrum shall be scanned from the lowest frequency generated by the device to the 10th harmonic of the operating frequency.
- (e) Representative calculations used to determine field strength from the actual meter reading indicating the conversion factors used and their source.

(f) The date the measurements were made

(g) The name and address of the engineer or technician who made the actual measurements, and the name and address of his employer, if any.

(h) The signature and printed name and address of the engineer responsible

for the report.

[34 F.R. 17172, Oct. 23, 1969]

- § 15.254 Identification of a device certificated under § 15.214.
- (a) Each device certificated under § 15.214 shall be identified by a label which may be part of the nameplate.
- (b) The label shall state that a certificate has been filed with the Commission attesting compliance with the applicable technical requirements.

(c) The label shall state further:

Operation of this equipment is subject to the following two conditions: 1. This equipment may not cause harmful interference. 2. This equipment must accept any interference that may be received, including interference that may cause undesired operation.

(d) The label shall be permanently attached to the device and shall be readily visible by prospective purchasers.

(e) The label may be attached only after the certificate required by § 15.214 has been filed with the Commission.

[34 F.R. 17173, Oct. 23, 1969]

PART 17—CONSTRUCTION, MARK-ING, AND LIGHTING OF ANTENNA STRUCTURES

Subpart A-General Information

Sec. 17.1 Basis and purpose.

17.2 Definitions.

17.4 Commission consideration of proposed antenna structure with respect to possible hazard to air navigation.

Subpart B—Criteria for Determining Whether Applications for Radio Towers Require Notification of Proposed Construction to Federal Aviation Agency

17.7 Antenna structures requiring notification to the FAA.

17.8 Establishment of antenna farm areas.

17.9 Designated antenna farm areas.

17.10 Antenna structures over 1,000 feet in height.

17.14 Certain antenna structures exempt from notification to the FAA.

17.17 Existing structures.

Subpart C—Specifications for Obstruction Marking and Lighting of Antenna Structures

17.21 Painting and lighting, when required.17.22 Particular specifications to be used.

Sec. 17.23 Specifications for the painting of antenna structures in accordance with § 17.21.

17.24 Specifications for the lighting of antenna structures up to and including 150 feet in height.

17.25 Specifications for the lighting of antenna structures over 150 feet up to and including 300 feet in height.

17.26 Specifications for the lighting of antenna structures over 300 feet up to and including 450 feet in height.

17.27 Specifications for the lighting of antenna structures over 450 feet up to and including 600 feet in height.

17.28 Specifications for the lighting of antenna structures over 600 feet up to and including 750 feet in height.

17.29 Specifications for the lighting of antenna structures over 750 feet up to and including 900 feet in height.

17.30 Specifications for the lighting of antenna structures over 900 feet up to and including 1,050 feet in height.

17.31 Specifications for the lighting of antenna structures over 1,050 feet up to and including 1,200 feet in height.

17.32 Specifications for the lighting of antenna structures over 1,200 feet up to and including 1,350 feet in height.

17.33 Specifications for the lighting of antenna structures over 1,350 feet and up to and including 1,500 feet in height.

17.34 Specifications for the lighting of antenna structures over 1,500 feet up to and including 1,650 feet in height above the ground.

17.35 Specifications for the lighting of antenna structures over 1,650 feet up to and including 1,800 feet in height.

17.36 Specifications for the lighting of antenna structures over 1,800 feet up to and including 1,950 feet in height.

17.37 Specifications for the lighting of antenna structures over 1,950 feet up to and including 2,100 feet in height.

17.38 Specifications for the lighting of antenna structures over 2,100 feet in height.

17.43 Painting and lighting of new and existing structures.

17.45 Temporary warning lights.

17.47 Inspection of tower lights and associated control equipment.

17.48 Notification of extinguishment or improper functioning of lights.

17.49 Recording of tower light inspections in the station record.

17.50 Cleaning and repainting.

17.51 Time when lights shall be exhibited.

17.52 Spare lamps,

17.53 Lighting equipment and paint.

17.54 Rated lamp voltage.

17.56 Maintenance of lighting equipment.

Sec.

- 17.57 Report of radio transmitting antenna construction, alteration and/or removal.
- 17.58 Facilities to be located on land under the jurisdiction of the U.S. Forest Service or the Bureau of Land Management.

AUTHORITY: The provisions of this Part 17 issued under secs. 4, 303, 48 Stat. 1066, 1082, as amended; 47 U.S.C. 154, 303. Interpret or apply secs. 301, 309, 48 Stat. 1081, 1085 as amended; 47 U.S.C. 301, 309.

Subpart A-General Information

§ 17.1 Basis and purpose.

- (a) The rules in this part are issued pursuant to the authority contained in Title III of the Communications Act of 1934, as amended, which vests authority in the Federal Communications Commission to issue licenses for radio stations when it is found that the public interest, convenience, and necessity would be served thereby, and to require the painting, and/or illumination of radio towers if and when in its judgment such towers constitute, or there is a reasonable possibility that they may constitute, a menace to air navigation.
- (b) The purpose of the rules in this part is to prescribe certain procedures and standards with respect to the Commission's consideration of proposed antenna structures which will serve as a guide to persons intending to apply for radio station licenses. The standards were developed in conjunction with the Federal Aviation Administration (FAA).

 [32 FR. 11268, Aug. 3, 1967]

§ 17.2 Definitions.

- (a) Antenna structures. The term antenna structures includes the radiating system, its supporting structures and any appurtenances mounted thereon.
- (b) An antenna farm area is defined as a geographical location, with established boundaries, designated by the Federal Communications Commission, in which antenna towers with a common impact on aviation may be grouped. [32 F.R. 8813, June 21, 1967, and 32 F.R. 11268, Aug. 3, 1967]
- § 17.4 Commission consideration of proposed antenna structure with respect to possible hazard to air navigation.
- (a) All applications are reviewed to determine whether there is a requirement that the applicant file a notice of

- proposed construction or alteration [FAA Form 7460–1] with the Federal Aviation Administration.
- (b) Whenever applications require the filing of a notice of proposed construction or alteration [FAA Form 7460-1] the applicant will be advised to do so unless the application includes an FCC Form 714 certifying that notification has been submitted to FAA or the application form itself specifically supplies all of the information which would be provided on FCC Form 714.
- (c) All applications which do not require the filing of FAA Form 7460-1 with the FAA will be deemed not to involve a hazard to air navigation and will be considered by the Commission without further reference to the FAA.
- (d) Whenever a "no hazard determination" is received from the FAA concerning any proposed antenna structure, the antenna structure is deemed not to involve a hazard to air navigation and the antenna aspect of the application for radio station authorization will be processed accordingly; provided that the FAA "no hazard determination" has not expired.
- (e) Whenever a report is received from the FAA indicating that a proposed antenna structure is a hazard, the Commission will take further appropriate action.
- (f) Applications which show on their face that the antenna structure will extend more than 20 feet above the ground or natural formation or more than 20 feet above an existing manmade structure (other than an antenna structure) shall be accompanied by FCC Form 714 indicating that notification has or has not been submitted to FAA or the application form itself shall specifically supply all of the information which would be provided on the FCC Form 714.
- (g) In addition to the other requirements of this part of the rules, each application for a radio station authorization shall include such information regarding proposed antenna construction as may be required by the FCC. Such information is to be supplied on the FCC application form specified in the rules pertaining to the radio service in which application is being made or as may otherwise be required.
- [32 F.R. 11268, Aug. 3, 1967, as amended at 34 F.R. 6481, Apr. 15, 1969]

Subpart B—Criteria for Determining Whether Applications for Radio Towers Require Notification of Proposed Construction to Federal Aviation Administration

§ 17.7 Antenna structures requiring notification to the FAA.

A notification to the Federal Aviation Administration is required, except as set forth in § 17.14, for any of the following construction or alteration:

(a) Any construction or alteration of more than 200 feet in height above

ground level at its site.

(b) Any construction or alteration of greater height than an imaginary surface extending outward and upward at

one of the following slopes:

(1) 100 to 1 for a horizontal distance of 20,000 feet from the nearest point of the nearest runway of each airport with at least one runway more than 3,200 feet in length, excluding heliports and seaplane bases without specified boundaries. if that airport is either listed in the Airport Directory of the current Airman's Information Manual or is operated by a Federal military agency.

(2) 50 to 1 for a horizontal distance of 10,000 feet from the nearest point of the nearest runway of each airport with its longest runway no more than 3,200 feet in length, excluding heliports and seaplane bases without specified boundaries, if that airport is either listed in the Airport Directory or is operated by a Federal military agency.

(3) 25 to 1 for a horizontal distance of 5,000 feet from the nearest point of the nearest landing and takeoff area of each heliport listed in the Airport Directory or operated by a Federal military agency.

(c) Any construction or alteration on an airport listed in the Airport Directory of the current Airman's Information

Manual.

(d) When requested by the FAA, any construction or alteration that would be in an instrument approach area (defined in the FAA standards governing instrument approach procedures) and available information indicates it might exceed an obstruction standard of the FAA.

Note: Consideration to aeronautical facilities not in existence at the time of the filing of the application for radio facilities will be given only when proposed airport construction or improvement plans are on file with

the Federal Aviation Administration as of the filing date of the application for such radio facilities.

[32 F.R. 11269, Aug. 3, 1967]

§ 17.8 Establishment of antenna farm areas.

(a) Each antenna farm area will be established by an appropriate rule making proceeding, which may be com-menced by the Commission on its own motion after consultation with the FAA. upon request of the FAA, or as a result of a petition filed by any interested person. After receipt of a petition from an interested person disclosing sufficient reasons to justify institution of a rule making proceeding, the Commission will request the advice of the FAA with respect to the considerations of menace to air navigation in terms of air safety which may be presented by the proposal. The written communication received from the FAA in response to the Commission's request shall be placed in the Commission's public rule making file containing the petition, and interested persons shall be allowed a period of 30 days within which to file statements with respect thereto. Such statements shall also be filed with the Administrator of the FAA with proof of such filing to be established in accordance with § 1.47 of this chapter. The Administrator of the FAA shall have a period of 15 days within which to file responses to such statements. If the Commission, upon consideration of the matters presented to it in accordance with the above procedure, is satisfied that establishment of the proposed antenna farm would constitute a menace to air navigation for reasons of air safety, rule making proceedings will not be instituted. If rule making proceedings are instituted, any person filing comments therein which concern the question of whether the proposed antenna farm will constitute a menace to air navigation shall file a copy of the comments with the Administrator of the FAA. Proof of such filing shall be established in accordance with § 1.47 of this chapter.

(b) Nothing in this subpart shall be construed to mean that only one antenna farm area will be designated for a community. The Commission will consider on a case-by-case basis whether or not more than one antenna farm area shall be designated for a particular com-

munity.

[32 F.R. 8813, June 21, 1967, as amended at 32 F.R. 13591, Sept. 28, 1967]



§ 17.9 Designated antenna farm areas.

The areas described in the following paragraphs of this section are established as antenna farm areas: [appropriate paragraphs will be added as necessary].

[32 F.R. 8813, June 21, 1967]

§ 17.10 Antenna structures over 1,000 feet in height.

Where one or more antenna farm areas have been designated for a community or communities (see § 17.9), the Commission will not accept for filing an application for a construction permit to construct a new station or to increase height or change antenna location of an existing station proposing the erection of an antenna structure over 1,000 feet above ground unless:

(a) It is proposed to locate the antenna structure in a designated antenna farm area, or

(b) It is accompanied by a statement from the Federal Aviation Administration that the proposed structure will not constitute a menace to air navigation, or

(c) It is accompanied by a request for waiver setting forth reasons sufficient, if true, to justify such a waiver.

[32 F.R. 8813, June 21, 1967]

§ 17.14 Certain antenna structures exempt from notification to the FAA.

A notification to the Federal Aviation Administration is not required for any of the following construction or alteration:

- (a) Any object that would be shielded by existing structures of a permanent and substantial character or by natural terrain or topographic features of equal or greater height, and would be located in the congested area of a city, town, or settlement where it is evident beyond all reasonable doubt that the structure so shielded will not adversely affect safety in air navigation. Applicants claiming such exemption under § 17.14(a) shall submit a statement with their application to the FCC explaining basis in detail for their finding.
- (b) Any antenna structure of 20 feet or less in height except one that would increase the height of another antenna structure.
- (c) Any electronic facility, the signal of which is used primarily for navigational guidance by aircraft, any airport visual approach or landing aid, or any airport ceiling or visibility indicator device, or other meteorological facility

or instrument, approved by the Administrator, the location and height of which would be fixed by its functional purpose. [32 F.R. 11269, Aug. 3, 1967]

§ 17.17 Existing structures.

- (a) Nothing in the criteria in this subpart concerning antenna structures or locations shall apply to those structures authorized prior to September 5, 1967.
- (b) No change in any of these criteria or relocation of airports shall at any time impose a new restriction upon any then existing or authorized antenna structure or structures.

[32 F.R. 11269, Aug. 8, 1967]

Subpart C—Specifications for Obstruction Marking and Lighting of Antenna Structures

§ 17.21 Painting and lighting, when required.

Antenna structures shall be painted and lighted when:

(a) They exceed 200 feet in height above the ground or they require special aeronautical study.

(b) The Commission may modify the above requirement for painting and/or lighting of antenna structures, when it is shown by the applicant that the absence of such marking would not impair the safety of air navigation, or that a lesser marking requirement would insure the safety thereof.

[32 F.R. 11269, Aug. 3, 1967]

§ 17.22 Particular specifications to be used.

Whenever painting and lighting are required, the Commission will assign painting and lighting specifications pursuant to the provisions of this subparts. If an antenna installation is of such a nature that its painting and lighting in accordance with these specifications are confusing, or endanger rather than assist airmen, or are otherwise inadequate, the Commission will specify the type of painting and lighting or other marking to be used in the individual situation.

[32 F.R. 11269, Aug. 3, 1967]

§ 17.23 Specifications for the painting of antenna structures in accordance with § 17.21.

Antenna structures shall be painted throughout their height with alternate bands of aviation surface orange and white, terminating with aviation surface orange bands at both top and bottom. The width of the bands shall be equal and approximately one-seventh the

height of the structure, provided however, that the bands shall not be more than 40 feet nor less than 1½ feet in width.

[28 F.R. 12529, Nov. 22, 1963]

§ 17.24 Specifications for the lighting of antenna structures up to and including 150 feet in height.

Antenna structures up to and including 150 feet in height above ground, which are required to be lighted as a result of notification to the FAA under § 17.7, shall be lighted as follows:

(a) There shall be installed at the top of the tower at least two 100-, 107-, or 116-watt lamps (#100 A21/TS, #107 A21/TS, or #116 A21/TS, respectively) enclosed in aviation red obstruction light globes. The two lights shall burn simultaneously from sunset to sunrise and shall be positioned so as to insure unobstructed visibility of at least one of the lights from aircraft at any normal angle of approach. A light sensitive control device or an astronomic dial clock and time switch may be used to control the obstruction lighting in lieu of manual control. When a light sensitive device is used, it should be adjusted so that the lights will be turned on at a north sky light intensity level of about 35-foot candles and turned off at a north sky light intensity level of about 58-foot candles.

[32 F.R. 11269, Aug. 3, 1967]

§ 17.25 Specifications for the lighting of antenna structures over 150 feet up to and including 300 feet in height.

(a) Antenna structures over 150 feet, up to and including 200 feet in height above ground, which are required to be lighted as a result of notification to the FAA under § 17.7 and antenna structures over 200 feet, up to and including 300 feet in height above ground, shall be lighted as follows:

(1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-, 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle

of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.

(2) At the approximate mid point of the overall height of the tower there shall be installed at least two 100-, 107-, or 116-watt lamps (#100 A21/TS, #107 A21/TS, or #116 A21/TS, respectively) enclosed in aviation red obstruction light globes. Each light shall be mounted so as to insure unobstructed visibility of at least one light at each level from aircraft at any normal angle of approach.

(3) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[28 F.R. 12529, Nov. 22, 1963, as amended at 32 F.R. 11269, Aug. 3, 1967]

§ 17.26 Specifications for the lighting of antenna structures over 300 feet up to and including 450 feet in height.

(a) Antenna structures over 300 feet up to and including 450 feet in height above the ground shall be lighted as follows:

(1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-. 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute, nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.

- (2) On levels at approximately two-thirds and one-third of the overall height of the tower, there shall be installed at least two 100-, 107-, or 116-watt lamps (#100 A21/TS, #197 A21/TS, or #116 A21/TS, respectively) enclosed in aviation red obstruction light globes. Each light shall be mounted so as to insure unobstructed visibility of at least one light at each level from aircraft at any normal angle of approach.
- (3) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles. [28 F.R. 12529, Nov. 22, 1963, as amended at 32 F.R. 11270, Aug. 3, 1967]

§ 17.27 Specifications for the lighting of antenna structures over 450 feet up to and including 600 feet in height.

- (a) Antenna structures over 450 feet up to and including 600 feet in height above the ground shall be lighted as follows:
- (1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-. 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.
- (2) At approximately one-half of the overall height of the tower one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any nor-

- mal angle of approach. In the event this beacon cannot be installed in a manner to insure unobstructed visibility of it from aircraft at any normal angle of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.
- (3) On levels at approximately three-fourths and one-fourth of the over-all height of the tower, at least one 100-, 107-, or 116-watt lamp (#100 A21/TS, #107 A21/TS or #116 A21/TS, respective-ly) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the tower at each level.
- (4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.
- [28 F.R. 12530, Nov. 22, 1963, as amended at 32 F.R. 11270, Aug. 3, 1967]
- § 17.28 Specifications for the lighting of antenna structures over 600 feet up to and including 750 feet in height.
- (a) Antenna structures over 600 feet up to and including 750 feet in height above the ground shall be lighted as follows:
- (1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-. 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.
- (2) At approximately two-fifths of the overall height of the tower one similar

flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event this beacon cannot be installed in a manner to insure unobstructed visibility of it from aircraft at any normal angle of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.

(3) On levels at approximately four-fifths, three-fifths and one-fifth of the over-all height of the tower, at least one 100-, 107-, or 116-watt lamp (#100 A21/TS, #107 A21/TS or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the tower at each level.

(4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[28 F.R. 12530, Nov. 22, 1963, as amended at 32 F.R. 11270, Aug. 3, 1967]

§ 17.29 Specifications for the lighting of antenna structures over 750 feet up to and including 900 feet in height.

(a) Antenna structures over 750 feet up to and including 900 feet in height above the ground shall be lighted as follows:

(1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-, 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute.

with a period of darkness equal to approximately one-half of the luminous period.

(2) On levels at approximately twothirds and one-third of the overall height of the tower one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons from aircraft at any normal angle of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.

(3) On levels at approximately five-sixths, one-half, and one-sixth of the over-all height of the tower, at least one 100-, 107-. or 116-watt lamp (#100 A21/TS, #107 A21/TS or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the tower at each level.

(4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[28 F.R. 12530, Nov. 22, 1963, as amended at 32 F.R. 11270, Aug. 3, 1967]

§ 17.30 Specifications for the lighting of antenna structures over 900 feet up to and including 1,050 feet in height.

(a) Antenna structures over 900 feet up to and including 1,050 feet in height above the ground shall be lighted as follows:

(1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-. 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.

(2) On levels at approximately foursevenths and two-sevenths of the overall height of the tower one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons from aircraft at any normal angle of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.

(3) On levels at approximately six-sevenths, five-sevenths, three-sevenths and one-seventh of the over-all height of the tower at least one 100-, 107-, or 116-watt lamp (#100 A21/TS, #107 A21/TS or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the structure.

(4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles. [28 F.R. 12530, Nov. 22, 1963, as amended at 32 F.R. 11270, Aug. 3, 1967]

§ 17.31 Specifications for the lighting of antenna structures over 1,050 feet up to and including 1,200 feet in height.

(a) Antenna structures over 1,050 feet up to and including 1,200 feet in height above the ground shall be lighted as follows:

(1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-, 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon

is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.

(2) On levels at approximately threefourths, one-half and one-fourth of the overall height of the tower one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons from aircraft at any normal angle of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.

(3) On levels at approximately seveneighths, five-eighths, three-eighths, and one-eighth of the over-all height of the tower, at least one 100-, 107-, or 116-watt lamp (#100 A21/TS, #107 A21/TS or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the structure.

(4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[28 F.R. 12531, Nov. 22, 1963, as amended at 32 F.R. 11271, Aug. 3, 1967]

§ 17.32 Specifications for the lighting of antenna structures over 1,200 feet up to and including 1,350 feet in height.

- (a) Antenna structures over 1,200 feet up to and including 1,350 feet in height above the ground shall be lighted as follows:
- (1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-, 620-,

or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately onehalf of the luminous period.

- (2) On levels at approximately twothirds, four-ninths, and two-ninths of the overall height of the tower one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons from aircraft at any normal angle of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.
- (3) On levels at approximately eightninths, seven-ninths, five-ninths, one-third and one-ninth of the over-all height of the tower, at least one 100-, 107-, or 116-watt lamp (#100 A21/TS, #107 A21/TS or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the tower at each level.
- (4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[28 F.R. 12531, Nov. 22, 1963, as amended at 32 F.R. 11271, Aug. 3, 1967]

- § 17.33 Specifications for the lighting of antenna structures over 1,350 feet and up to and including 1,500 feet in height.
- (a) Antenna structures over 1,350 feet up to and including 1,500 feet in height above the ground shall be lighted as follows:
- (1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-. 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.
- (2) On levels at approximately fourfifths, three-fifths, two-fifths, and one-fifth of the overall height of the tower one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons from aircraft at any normal angles of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.
- (3) On levels at approximately ninetenths, seven-tenths, one-half, threetenths, and one-tenth of the over-all height of the tower, at least one 100-, 107-, or 116-watt lamp (#100 A21/TS, #107 A21/TS, or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the tower at each level.

- (4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles. [28 F.R. 12531, Nov. 22, 1963, as amended at 32 F.R. 11271, Aug. 3, 1967]
- § 17.34 Specifications for the lighting of antenna structures over 1,500 feet up to and including 1,650 feet in height above the ground.
- (a) Antenna structures over 1,500 feet up to and including 1,650 feet in height above the ground shall be lighted as follows:
- (1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-, 620-, or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.
- (2) On levels at approximately eightelevenths, six-elevenths, four-elevenths, and two-elevenths of the overall height of the tower, one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from the aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons from aircraft at any normal angle of approach there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.

(3) On levels at approximately tenelevenths, nine-elevenths, sevenelevenths, five-elevenths, three-elevenths, and one-eleventh of the overall height of the tower at least one 100-, 107-, or 116-watt lamp (#100 A21/TS, #107 A21/TS, or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the structure.

(4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[32 F.R. 11271, Aug. 3, 1967]

- § 17.35 Specifications for the lighting of antenna structures over 1,650 feet up to and including 1,800 feet in height.
- (a) Antenna structures over 1,650 feet up to and including 1,800 feet in height above the ground shall be lighted as follows:
- (1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-. 620-. or 700-watt lamps (PS-40 Code Beacon type) both lamps to burn simultaneously. and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.
- (2) On levels at approximately five-sixths, two-thirds, one-half, one-third, and one-sixth of the overall height of the tower one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons

from aircraft at any normal angle of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.

(3) On levels at approximately eleventwelfths, three-fourths, seven-twelfths, five-twelfths, on e - f o u r t h, and one-twelfth of the overall height of the tower at least one 100-, 107-, or 116-watt lamp (No. 100 A21/TS, No. 107 A21/TS, or No. 116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the structure.

(4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[32 F.R. 11272, Aug. 3, 1967; 33 F.R. 7039, May 10, 1968]

§ 17.36 Specifications for the lighting of antenna structures over 1,800 feet up to and including 1,950 feet in height.

- (a) Antenna structures over 1,800 feet up to and including 1,950 feet in height above the ground shall be lighted as follows:
- (1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-, 620of 700-watt lamps (PS-40, Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.
- (2) On levels at approximately tenthirteenths, eight-thirteenths, sixthirteenths, four-thirteenths, and two-thirteenths of the overall height of the tower one similar flashing 300 m/m elec-

tric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons from aircraft at any normal angle of approach, there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.

- (3) On levels at approximately twelvethirteenths, eleven-thirteenths, ninethirteenths, seven-thirteenths, fivethirteenths, three-thirteenths, and one-thirteenth of the overall height of the tower at least one 100-, 107-, or 116-watt lamp (#100 A21/TS, #107 A21/TS, or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the structure.
- (4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[32 F.R. 11272, Aug. 3, 1967]

§ 17.37 Specifications for the lighting of antenna structures over 1,950 feet up to and including 2,100 feet in height.

- (a) Antenna structures over 1,950 feet up to and including 2,100 feet in height above the ground shall be lighted as follows:
- (1) There shall be installed at the top of the structure one 300 m/m electric code beacon equipped with two 500-, 620- or 700-watt lamps (PS-40, Code Beacon type) both lamps to burn simultaneously, and equipped with aviation red color filters. Where a rod or other construction of not more than 20 feet in height and incapable of supporting this beacon is mounted on top of the structure and it is determined that this additional construction does not permit unobstructed visibility of the code beacon from aircraft at any normal angle of approach, there shall be installed two such beacons positioned so as to insure unobstructed visibility of at least one of the beacons from aircraft at any normal angle of approach. The beacons shall be equipped with a flashing mechanism producing not more than 40 flashes per

minute nor less than 12 flashes per minute, with a period of darkness equal to approximately one-half of the luminous period.

- (2) On levels at approximately sixsevenths, five-sevenths, four-sevenths, three-sevenths, two-sevenths, and oneseventh of the overall height of the tower one similar flashing 300 m/m electric code beacon shall be installed in such position within the tower proper that the structural members will not impair the visibility of this beacon from aircraft at any normal angle of approach. In the event these beacons cannot be installed in a manner to insure unobstructed visibility of the beacons from aircraft at any normal angle of approach. there shall be installed two such beacons at each level. Each beacon shall be mounted on the outside of diagonally opposite corners or opposite sides of the tower at the prescribed height.
- (3) On levels at approximately thirteen-fourteenths, eleven-fourteenths, nine-fourteenths, one half, five-fourteenths, three-fourteenths, and one-fourteenth of the overall height of the tower at least one 100-, 107-, or 116-watt lamps (#100 A21/TS, #107 A21/TS, or #116 A21/TS, respectively) enclosed in an aviation red obstruction light globe shall be installed on each outside corner of the structure.
- (4) All lights shall burn continuously or shall be controlled by a light sensitive device adjusted so that the lights will be turned on at a north sky light intensity level of about 35 foot candles and turned off at a north sky light intensity level of about 58 foot candles.

[32 F.R. 11272, Aug. 3, 1967]

§ 17.38 Specifications for the lighting of antenna structures over 2,100 feet in height.

Antenna structures over 2,100 feet in height above the ground shall be lighted in accordance with specifications to be determined by the Commission after aeronautical study which will include lighting recommendations.

[32 F.R. 11272, Aug. 3, 1967]

§ 17.43 Painting and lighting of new and existing structures.

(a) The provisions of this part of the rules shall be effective with respect to antenna structures required to be lighted and/or painted in accordance with the terms of an authorization for a new station or a change in the height or location

of an antenna structure issued on or after September 5, 1967.

- (b) All antenna structures required to be painted and lighted in accordance with a radio station authorization valid on September 5, 1967, shall be brought into conformity with this subpart within 6 months after September 5, 1970, at any station for which the authorization is renewable on or prior to that date and within 3 months following the renewal of any authorization renewable after September 5, 1970.
- (c) Nothing in the notification criteria concerning antenna structures or locations, as set forth in Subpart B of this part, shall apply to painting and lighting those structures authorized prior to September 5, 1967, except where lighting and painting requirements are reduced, in which case the lesser requirements may apply upon approval of an application to Commission for such reduction.

 [32 F.R. 11273, Aug. 3, 1967]

§ 17.45 Temporary warning lights.

During construction of an antenna structure, for which obstruction lighting is required, at least two 100-, 107-, or 116-watt lamps (No. 100 A21/TS, No. 107 A21/TS, or No. 116 A21/TS, respectively) enclosed in aviation red obstruction light globes, shall be installed at the uppermost point of the structure. In addition, as the height of the structure exceeds each level at which permanent obstruction lights will be required, two similar lights shall be installed at each such level. These temporary warning lights shall be displayed nightly from sunset to sunrise until the permanent obstruction lights have been installed and placed in operation, and shall be positioned so as to insure unobstructed visibility of at least one of the lights at any normal angle of approach. In lieu of the above temporary warning lights, the permanent obstruction lighting fixtures may be installed and operated at each required level as each such level is exceeded in height during construction. [32 F.R. 11273, Aug. 3, 1967]

§ 17.47 Inspection of tower lights and associated control equipment.

The licensee of any radio station which has an antenna structure requiring illumination pursuant to the provisions of section 303(q) of the Communications Act of 1934, as amended, as outlined elsewhere in this part:

- (a) (1) Shall make an observation of the tower lights at least once each 24 hours either visually or by observing an automatic properly maintained indicator designed to register any failure of such lights, to insure that all such lights are functioning properly as required; or alternatively.
- (2) Shall provide and properly maintain an automatic alarm system designed to detect any failure of such lights and to provide indication of such failure to the licensee.
- (b) Shall inspect at intervals not to exceed 3 months all automatic or mechanical control devices, indicators, and alarm systems associated with the tower lighting to insure that such apparatus is functioning properly.

[32 F.R. 11273, Aug. 3, 1967]

§ 17.48 Notification of extinguishment or improper functioning of lights.

The licensee of any radio station which has an antenna structure requiring illumination pursuant to the provisions of section 303(q) of the Communications Act of 1934, as amended, as outlined elsewhere in this part:

- (a) Shall report immediately by telephone or telegraph to the nearest Flight Service Station or office of the Federal Aviation Administration any observed or otherwise known extinguishment or improper functioning of a code or rotating beacon light or top light not corrected within 30 minutes. Further notification by telephone or telegraph shall be given immediately upon resumption of the required illumination.
- (b) An extinguishment or improper functioning of a steady burning side or intermediate light or lights, shall be corrected as soon as possible, but notification to the FAA of such extinguishment or improper functioning is not required. [32 F.R. 11273, Aug. 3, 1967]

§ 17.49 Recording of tower light inspections in the station record.

The licensee of any radio station which has an antenna structure requiring illumination shall make the following entries in the station record of the inspections required by § 17.47.

(a) The time the tower lights are turned on and off each day if manually controlled.

- (b) The time the daily check of proper operation of the tower lights was made, if automatic alarm system is not provided.
- (c) In the event of any observed or otherwise known extinguishment or improper functioning of a tower light:
- (1) Nature of such extinguishment or improper functioning.
- (2) Date and time the extinguishment or improper functioning was observed, or otherwise noted.
- (3) Date, time, and nature of the adjustments, repairs, or replacements made.
- (4) Identification of Flight Service Station (Federal Aviation Administration) notified of the extinguishment or improper functioning of any code or rotating beacon light or top light not corrected within 30 minutes, and the date and time such notice was given.
- (5) Date and time notice was given to the Flight Service Station (Federal Aviation Administration) that the required illumination was resumed.
- (d) Upon completion of the periodic inspection required at least once each 3 months:
- (1) The date of the inspection and the condition of all tower lights and associated tower lighting control devices, indicators and alarm systems.
- (2) Any adjustments, replacements, or repairs made to insure compliance with the lighting requirements and the date such adjustments, replacements or repairs were made.

[32 F.R. 11273, Aug. 3, 1967]

§ 17.50 Cleaning and repainting.

All towers shall be cleaned or repainted as often as necessary to maintain good visibility.

[32 F.R. 11273, Aug. 3, 1967]

§ 17.51 Time when lights shall be exhibited.

All lighting shall be exhibited from sunset to sunrise unless otherwise specified.

[32 F.R. 11273, Aug. 3, 1967]

§ 17.52 Spare lamps.

A sufficient supply of spare lamps shall be maintained for immediate replacement purposes at all times. [32 F.R. 11273, Aug. 3, 1967]

§ 17.53 Lighting equipment and paint.

The lighting equipment, color of filters. and shade of paint referred to in the specifications are further defined in the following government and/or Army-Navy aeronautical specifications, bulletins, and drawings (lamps are referred to by standard numbers):

Outside whiteAviation surface orange		TT-P-59 1 (Color No. 12197 of Federal
- · ·		Standard 595). TT-E-489 1 (Color No. 12197 of Federal Standard 595).
Code beacon Obstruction light globe, prismatic	FAA specifications	446 (Sec. II-d-Style 4).
Obstruction light globe, prismatic	Army-Navy drawing)
Obstruction light globe, Fresnel	do	
Single multiple obstruction light fitting	do	AN-L-10A or FAA Specification L-810.
assembly.		
Obstruction light fitting assembly	do	
100-watt lamp		
107 watt lamp		
116-watt lamp		
500-watt amp		
620-watt lamp		
700-watt lamp		
•		

¹ Copies of this specification can be obtained from the Specification Activity, Room 1643, Federal Supply Service Center, General Services Administration, 7th and D Streets SW., Washington, D.C. 20407 (Outside white, 5 cents;

Center, General Services Administration, 7th and D Streets S.W., Washington, D.C. 20407 (Outside white, 5 cents; aviation surface orange paint, 5 cents, enamel, 10 cents.

² Copies of Army-Navy specifications or drawings can be obtained by contacting the Commanding General, Air Materiel Command, Wright Field, Dayton, Ohio 45433, or the Naval Air Systems Command, Navy Department, Washington, D.C. 20360. Information concerning Army-Navy specifications or drawings can also be obtained from the Federal Aviation Administration, Washington, D.C. 20553.

³ Copies of the specification can be obtained from the Federal Aviation Administration, Washington, D.C. 20553.

⁴ The 116-watt, 6,000-hour lamp and the 700-watt, 6,000-hour lamp may be used instead of the 100-watt and the 500-watt lamps whenever possible in view of the extended life, lower maintenance cost, and greater safety which they provides.

[33 F.R. 11540, Aug. 14, 1968]

§ 17.54 Rated lamp voltage.

To provide satisfactory output by obstruction lights, the rated voltage of the lamp used should, in each case, correspond to or be within 3 percent higher than the average voltage across the lamp during the normal hours of operation. [32 F.R. 11274, Aug. 3, 1967]

§ 17.56 Maintenance of lighting equipment.

Replacing or repairing of lights, automatic indicators or automatic alarm systems shall be accomplished as soon as practicable.

[32 F.R. 11274, Aug. 3, 1967]

Report of radio transmitting antenna construction, alteration and/ or removal.

Any permittee or licensee who, pursuant to any instrument of authorization from the Commission to erect or make changes affecting antenna height or location of an antenna tower for which obstruction marking is required, shall, prior to start of tower construction and upon completion of such construction or changes, fill out and file with the Director, U.S. Coast and Geodetic Survey, C & GS Form 844 (Report of Radio

Transmitting Antenna Construction. Alteration and/or Removal) in order that antenna tower information may be provided promptly for use on aeronautical charts and related publications in the interest of safety of air navigation. [32 F.R. 11274, Aug. 3, 1967]

Facilities to be located on land under the jurisdiction of the U.S. Forest Service or the Bureau of Land Management.

Any application proposing new or modified transmitting facilities to be located on land under the jurisdiction of the U.S. Forest Service or the Bureau of Land Management shall include a statement that the facilities will be so located, and the applicant shall comply with the requirements of § 1.70 of this chapter.

[32 F.R. 11274, Aug. 3, 1967]

PART 18-INDUSTRIAL, SCIENTIFIC, AND MEDICAL EQUIPMENT

Subpart A-General

Sec. 18.1 Statement of basis and purpose. 18.3 Definitions.

Full information; inspection 18.11 by Commission representatives.

Title	47	Chapte	er l
-------	----	--------	------

§ 18.1	Title 47–
Sec.	
18.13	ISM frequencies and frequency tol- erances.
18.14	Operation on microwave frequencies.
18.17	Interference from ISM equipment.
18.21	When license is required.
18.22	Showing required.
18.23	Applications for station licenses.
18.24	Full information.
18.25	License period.
18 .26	Renewal of license.
18.27	Station license, posting of.
1 8.2 8	Operator requirements.
18.29	Cessation of operation pursuant to license.
	Subpart B——[Reserved]
	Subpart C—Ultrasonic Equipment
18.71	Operation without a license.
18.72	Technical limitations.
18.73	Type approval.
18.74	Identification of type approved equipment.
18 .75	Effect of certificate of type approval.
18.76	Changes in type approved equipment.
18.77	Withdrawal of certificate of type approval.
18.78	Measurement of field strength.
18.79	Location of equipment.
18.80	Certification attesting compliance

18.77	proval.	
18.78	Measurement of field strength.	
18.79	Location of equipment.	
18.80	Certification attesting compliance with rules.	
18.81	Renewal of certification.	

after

maintenance

work. 18.83 Prototype certification permitted.

18.84 Effective date.

Certification

18.82

Subpart D-Industrial Heating Equipment

18.101 Operation without a license. 18.102 Technical limitations. 18.105 Inspection of industrial heating equipment. 18.106 Renewal of certificate. 18.107 Measurement of field strength. 18.108 Location of equipment. 18.109 Report of radiation measurements 18.111 Form of certificate. 18.112 Certification regarding operation. 18.113 Certification regarding radiation. 18.114 Prototype certification permitted. 18.115 Compliance with installation instructions. 18.116 Certificate to be filed with Commission. Copy of certificate with equipment. 18.117 18.118 Rejection of certificate. 18.119 Elimination and investigation of

harmful interference.

safety service.

18.120 Interference to a radionavigation or

Sec. 18.121 Interference to other radio services. 18.122 Report of interference investigation. 18.125 Prototype certification procedure. 18.126 Information to be filed for prototype certification.

Subpart E-Medical Diathermy Equipment

Sec. 18 141 Operation on assigned frequencies. 18.142 Operation on unassigned frequen-18.143 Measurement of field strength.

18.144 Procedure for type approval.

18.145 Effect of certificate of type approval. 18.146 Withdrawal of certificate of type approval.

Subpart F—FR Stabilized Arc Welders

18.181 Technical specifications. 18.182 Certification requirements.

18.183 Location of certificate.

Subpart G-[Reserved]

Subpart H-Miscellaneous Equipment

18.261 Miscellaneous equipment. 18.262 Existing epilation equipment.

AUTHORITY: The provisions of this Part 18 issued under secs. 4(1), 303(r), 48 Stat. 1066, 1082, as amended; 47 U.S.C. 154(i), 303(r). Interpret or apply sec. 301, 48 Stat. 1081; 47 U.S.C. 301.

Source: The provisions of this Part 18 appear at 28 F.R. 12533, Nov. 22, 1963, unless otherwise noted.

Subpart A-General

Statement of basis and purpose. § 18.1

(a) Section 301 of the Communications Act of 1934, as amended, provides for the control by the Federal Government over all the channels of interstate and foreign radio communication and further provides, in part, that no person shall use or operate apparatus for the transmission of energy, communications, or signals by radio when the effects of such operation extend beyond state lines or cause interference with the transmission or reception of energy, communications, or signals, of any interstate or foreign character by radio, except under and in accordance with the Communications Act and a license granted under the provisions of that act. The operation in the industrial, scientific and medical service of medical diathermy equipment, industrial heating equipment and miscellaneous equipment of a type which

emits radio frequency energy upon frequencies within the radio spectrum constitutes a serious source of interference to authorized radio communication services operating upon the channels of interstate and foreign communication unless precautions are taken which will prevent the creation of any substantial amount of such interference.

- (b) The following rules and regulations are designed to have a twofold effect:
- (1) They set forth the conditions under which the operation of the equipment in question is not regarded as a cause of interference to the authorized radio communication services and is therefore not required to be operated pursuant to license under the Communications Act.
- (2) They provide a procedure for the licensing of medical diathermy, industrial heating and miscellaneous equipment which in operation constitute a source of interference to authorized communication services, directly affect the control of the Federal Government over the channels of interstate and foreign radio communication, and are therefore required to be licensed.

§ 18.3 Definitions.

For purposes of the provisions of this part the following definitions in the industrial, scientific, and medical service shall be applicable:

(a) "Radio frequency energy" shall include electromagnetic energy generated at any frequency in the radio spectrum between 10 kilocycles and 30,000 megacycles.

- (b) "Medical diathermy equipment" shall include any apparatus (other than surgical diathermy apparatus designed for intermittent operation with low power) which utilizes a radio frequency oscillator or any other type of radio frequency generator and transmits radio frequency energy used for therapeutic purposes.
- (c) "Industrial heating equipment" shall include any apparatus which utilizes a radio frequency oscillator or any other type of radio frequency generator and transmits radio frequency energy

used for or in connection with industrial heating operations utilized in a manufacturing or production process.

- (d) Miscellaneous equipment shall include apparatus other than that defined in or excepted by paragraphs (b) and (c) of this section in which radio frequency energy is applied to materials to produce physical, biological, or chemical effects such as heating, ionization of gases, mechanical vibrations, hair removal and acceleration of charged particles which do not involve communications or the use of radio receiving equipment.
- (e) Ultrasonic equipment shall include any apparatus which generates radio frequency energy and utilizes that energy to excite or drive an electromechanical transducer for the production of sonic or ultrasonic mechanical energy for industrial, scientific, medical or other noncommunication purposes.
- (f) "Industrial, scientific and medical equipment" (ISM equipment). Devices which use radio waves for industrial, scientific, medical or any other purposes including the transfer of energy by radio and which are neither used nor intended to be used for radiocommunication.
- (g) "Harmful interference." Any emission, radiation or induction which endangers the functioning of a radio-navigation service or of other safety services or seriously degrades, obstructs or repeatedly interrupts a radiocommunication service operating in accordance with this chapter.
- (h) "ISM frequency." A frequency assigned by this part for the use of ISM equipment. A specified tolerance is associated with each ISM frequency. (See § 18.13.)

§ 18.11 Full information; inspection by Commission representatives.

Upon request by the Commission the owner or operator of any medical diathermy equipment, industrial heating equipment, or miscellaneous equipment shall promptly furnish the Commission with such information as may be requested concerning the operation of such equipment. The premises in which medical diathermy, industrial heating, or

miscellaneous equipment are operated, and any license or certification required hereby, shall be available for inspection by representatives of the Commission at all reasonable hours.

§ 18.13 ISM frequencies and frequency tolerances.

The following frequencies are allocated for use by ISM equipment with the tolerance limits specified:

	rrequency
ISM frequency:	tolerance
13,560 kc/s	$\pm 6.78 \mathrm{kc/s}$
27,120 kc/s	$\pm 160.0 kc/s$
40.680 kc/s	$\pm 20.0 \mathrm{kc/s}$
915 Mc/s ¹	$\pm 25.0 \mathrm{Mc/s}$
2450 Mc/s 1	$\pm 50.0 \mathrm{Mc/s}$
5800 Mc/s 1	$\pm 75.0 \mathrm{Mc/s}$
22,125 Mc/s 1	$\pm 125.0 \text{Mc/s}$

¹The use of this frequency is subject to the conditions in § 18.14.

§ 18.14 Operation on microwave frequencies.

Except for industrial heating equipment which is regulated by §§ 18.101 through 18.122, inclusive, ISM equipment may be operated on the microwave ISM frequencies (915 Mc/s, 2450 Mc/s, 5800 Mc/s and 22,125 Mc/s) subject to the following conditions:

- (a) The emission of radio frequency energy resulting from such operation shall be on the particular frequency and must not exceed tolerance limits associated with each such frequency as set forth in § 18.13.
- (b) The energy radiated and the bandwidth of emissions shall be reduced to the greatest extent practicable.
- (c) No harmful interference shall be caused to authorized communication services from spurious or harmonic radiation. In the event of such harmful interference, operation of the ISM equipment causing such harmful interference shall cease and shall not be resumed until steps necessary to eliminate such interference have been taken.

§ 18.17 Interference from ISM equipment.

(a) Subject to the exceptions in paragraphs (b) and (c) of this section and irrespective of whether the equipment otherwise complies with the rules in this part, the operator of ISM equip-

ment that causes harmful interference to any authorized radio service shall promptly take steps as may be necessary to remedy the interference.

(b) The provisions of paragraph (a) of this section shall not apply in the case of interference to an authorized radio station operating on an ISM frequency (including tolerance).

(c) The provisions of paragraph (a) of this section shall not apply in the case of interference to a receiver arising from direct intermediate frequency pick-up by the receiver of the fundamental frequency emissions of ISM equipment operating on an ISM frequency (including tolerance) and otherwise complying with the requirements of this part.

§ 18.21 When license is required.

- (a) Any medical diathermy equipment, industrial heating equipment or miscellaneous equipment which complies with the provisions of this part may be operated without a station license. A license is required for any such equipment operated otherwise.
- (b) No medical diathermy equipment, industrial heating equipment or miscellaneous equipment which does not comply with this part shall be operated except pursuant to a station license issued by the Commission authorizing such operation.
- (c) Whenever the Commission on complaint or on its own motion determines that medical diathermy equipment, industrial heating equipment or miscellaneous equipment is not in fact operating in compliance with the provisions of this part and so advises the operator of such equipment, further operation of such equipment without a station license shall be unlawful unless within 10 days of the receipt of such notice, or within such further time as the Commission may for good cause allow, the operator of such equipment shall file with the Commission a certificate of a competent engineer stating that the equipment is now capable of complying with the requirements of the rules.

§ 18.22 Showing required.

A station license for the operation of medical diathermy equipment, in-

dustrial heating equipment or miscelequipment will be granted upon proper application therefor in accordance with the provisions of this part and a showing that in the light of the following considerations the public interest, convenience, and necessity would be served by such a grant: (a) The purpose for which the equipment sought to be licensed will be used; (b) the reasons why the equipment involved may not be operated in compliance with the provisions of this part for the operation of such equipment without a license; and (c) the nature and extent of interference that may be caused to authorized communication services by the operation of such equipment.

§ 18.23 Applications for station licenses.

Each applicant for a station license authorizing the operation of medical diathermy, industrial heating equipment, or miscellaneous equipment, or requesting the modification or renewal of such a license, shall file with the Commission in Washington, D.C., three copies of each application on the appropriate form designated by the Commission and a like number of any exhibits and other papers incorporated therein and made a part thereof. Application for a license shall be made upon the appropriate form prescribed by the Commission, and separate application should be made for each unit of equipment for which a license is sought. Application for modification or renewal of a license shall also be upon appropriate form prescribed by the Commission.

§ 18.24 Full information.

Each application for a license authorizing the operation of medical diathermy, industrial heating equipment or miscellaneous equipment shall contain full and complete information concerning all matters and things required to be disclosed by the application form.

§ 18.25 License period.

Each station license authorizing the operation of medical diathermy, industrial equipment or miscellaneous equipment will expire at the hour of 3 a.m. and will be issued for a normal license period

of five years or such other period as the Commission may specify upon consideration of the facts in a particular case. Each such license shall be nontransferable.

§ 13.26 Renewal of license.

Unless otherwise directed or permitted by the Commission, applications for renewal of a station license for the operation of medical diathermy, industrial heating equipment or miscellaneous equipment shall be filed with the Commission upon prescribed forms at least 60 days prior to the expiration date of such license.

§ 18.27 Station license, posting of.

The original of each station license shall be posted in the room in which the equipment is operated. Licenses covering equipment not used in a fixed place shall be attached to the equipment itself.

§ 18.28 Operator requirements.

Equipment for which a station license is issued pursuant to the provisions of this part may be operated by persons who do not hold an operator license or permit issued by this agency.

§ 18.29 Cessation of operation pursuant to license.

If any equipment for which a license has been issued hereunder shall cease to be operated pursuant to such license, or is transferred, sold, assigned, leased, loaned, stolen, destroyed, or otherwise removed from the possession of the licensee, the licensee shall within five days of such occurrence notify the Commission thereof and, where possible, include in such notification the name and address of the recipient of such equipment.

Subpart B—[Reserved]

Subpart C—Ultrasonic Equipment

§ 18.71 Operation without a license.

Ultrasonic equipment may be operated without a license: Provided, The design and operation complies with the technical limitations for such equipment: And provided further, That the equipment has been type approved by the Commission or has been certificated pursuant to

the requirements of §§ 18.71 to 18.84 and the certificate is attached to the equipment or is prominently posted in the room in which the equipment is being operated; except that ultrasonic equipment operating on frequencies below 90 kc/s and generating less than 500 watts of radio frequency power may be operated without license, type approval or certification, if such equipment complies with all other applicable provisions of §§ 18.71 to 18.84.

§ 18.72 Technical limitations.

- (a) Ultrasonic equipment shall be designed and constructed in accordance with good engineering practice with sufficient shielding and filtering to provide adequate suppression of emissions on frequencies outside the ISM frequency bands.
- (b) Except for ultrasonic measurement equipment that operates over a continuous band of frequencies, the fundamental frequency of operation shall fall outside the frequency bands 490-510 kc/s, 2170-2194 kc/s, and 8354-8374 kc/s.
- (c) The varying conditions under which ultrasonic equipment is operated shall not result in radiation exceeding the following limits:

Frequency	Dis- tance Feet	Field µv/m
Up to and including 490 kc/s_	1,000	2400 Frequency in kc/s
Over 490 kc/s up to and including 1600 kc/s.	100	24000 Frequency in kc/s
Over 1600 kc/s exclusive of frequencies in the ISM frequency bands.	100	15.
		l

(d) The operation of ultrasonic equipment on frequencies below 490 kc/s using radio frequency power in excess of 500 watts shall be in compliance with the requirements of this section except that the maximum radiated field permitted may be increased as the square root of the ratio of the generated radio frequency power to 500 watts: Provided, That the radiated field shall in no case exceed the field permitted industrial heating equipment: And provided fur-

ther. That equipment used in predominantly residential areas shall not be permitted the increase in field with power as indicated in this paragraph.

(e) On any frequency above 490 kc/s, the radio frequency voltage appearing on each power line shall not exceed 200 microvolts. On any frequency below 490 kc/s, the radio frequency voltage appearing on each power line shall not exceed 1000 microvolts. Measurement shall be made from each power line to ground with the equipment itself both grounded and ungrounded.

Note: One method of making conducted interference measurements is described in "Military Specification for Interference Measurement" MIL-I-16910 (SHIPS) dated January 14, 1952, available from the Commanding Officer, Naval Supply Depot, Scotia, New York, 12302. Note that this procedure calls for grounding the equipment under test, whereas these rules call for measurements with the equipment both grounded and ungrounded.

§ 18.73 Type approval.

- (a) Manufacturers of ultrasonic equipment desiring to obtain type approval for their equipment may request permission to submit such equipment to the Commission for testing by following the procedure set out in Part 2 of this chapter. The request shall include a statement that at least five units of the model to be submitted are scheduled for manufacture.
- (b) To be acceptable for type approval, ultrasonic equipment must meet the following requirements:
- (1) The equipment must comply with the technical limitations for ultrasonic equipment.
- (2) The design and construction of the equipment must give reasonable assurance of compliance with the rules in this part for at least 5 years under normal operation and with average maintenance.
- (c) Additional rules relative to type approval will be found in Part 2 of this chapter.

§ 18.74 Identification of type approved equipment.

(a) Equipment for which a certificate of type approval has been issued shall be

identified by the insertion of the FCC Type Approval Number on the name-plate of the equipment.

(b) In addition to the nameplate, the manufacturer shall furnish each user of type approved equipment a certificate setting forth the conditions under which such equipment shall be operated.

§ 18.75 Effect of certificate of type approval.

A certificate of type approval issued by the Commission constitutes a recognition that, on the basis of the tests made, the equipment appears to be capable of complying with the technical limitations in the rules in this part, provided such equipment is properly installed, maintained and operated, and no change whatsoever is made in the construction of equipment sold under the certificate of type approval except on specific prior approval by the Commission to any changes made.

§ 18.76 Changes in type approved equipment.

No changes whatsoever may be made in ultrasonic equipment for which a certificate of type approval has been issued except on specific prior approval by the Commission.

§ 18.77 Withdrawal of certificate of type approval.

- (a) A certificate of type approval may be withdrawn if the type of equipment for which it was issued proves defective in service and under usual conditions of maintenance and operation such equipment cannot be relied on to meet the conditions set forth in this part for the operation of the type of equipment involved, or if any change whatsoever is made in the construction of equipment sold under the certificate of type approval issued by the Commission, without the specific prior approval of the Commission.
- (b) The procedure for withdrawal of the certificate of type approval shall be the same as that prescribed for revocation of a radio station license pursuant to the provisions of the Communications Act of 1934, as amended.
- (c) In the case of withdrawal of a certificate of type approval the manufacturer shall make no further sale of equipment under such certificate.

(d) When a certificate of type approval has been withdrawn for unauthorized changes or for failure to comply with technical requirements, the Commission will consider that fact in determining whether the maunfacturer in question is eligible to receive any new certificate of type approval.

§ 18.78 Measurement of field strength.

Measurements to determine the field strength of radio frequency energy including both fundamental and spurious (including harmonic) emissions, generated by the ultrasonic equipment shall be made in accordance with standard engineering procedures and shall include the following:

- (a) A field strength meter using loop pickup shall be used for measurements on frequencies up to and including 18 Mc/s, and such a meter with a doublet antenna shall be used for measurements on frequencies above 18 Mc/s.
- (b) The radiation shall be determined along at least 5 radials approximately 72° apart. A smooth curve shall be drawn through the measurements when plotted and the value of field strength determined from these curves.

§ 18.79 Location of equipment.

For the purpose of measurements required in order to execute a certification of compliance, the location of the ultrasonic equipment may be considered to be the actual physical location of the equipment, or, where several such units are grouped within a circle of 200 feet radius or less, the several units may at the election of the certifying engineer be considered as a single unit, located at the center of the smallest enclosing circle. If the certification includes several units treated as one equipment, the distance of 1.000 feet at which the maximum permissible radiation is determined shall be decreased by the radius of the smallest circle that encloses the several units.

§ 18.80 Certification attesting compliance with rules.

(a) A certification attesting compliance with the rules in this part may be affixed or posted for any ultrasonic equipment.

- (b) The certification shall be based on an inspection of the equipment and measurements taken at the place of use after the ultrasonic equipment has been assembled and is ready for operation: Provided however, That the certifying engineer may, in lieu of measuring the radio frequency voltage on the power lines, base the certification on specifications for the power line filter and test data regarding the radio frequency voltage on the power lines furnished by the manufacturer of the ultrasonic equipment.
- (c) The certification may be executed by any engineer skilled in making and interpreting field strength measurements. The Commission may require such engineer to present proof of his qualifications to make such measurements.

(d) The certification shall contain the following information:

(1) Type and serial number, or other positive identification of the ultrasonic equipment being certificated.

(2) Conditions under which the certificated equipment shall be operated.

- (3) Brief description of the engineering tests and a summary of the measured data upon which the certification is based.
- (4) If the radio frequency voltage on the power line is not measured, a statement that, based on an inspection of the equipment and study of such test data and specifications as may be furnished by the manufacturer, the equipment can reasonably be expected to meet the requirements for radio frequency voltage on the power lines.
- (5) A statement certifying that under the described condition of operation, the certificated equipment may reasonably be expected to meet the requirements of the rules in this part. This statement shall include the period of time over which the equipment may reasonably be expected to comply with the rules in this part.
- (6) Date the measurements were made.
 - (7) Date of certification.
 - (8) Signature of certifying engineer.
- (9) Name and address of employer of certifying engineer, if any.

§ 18.81 Renewal of certification.

The certification required by §§ 18.71 to 18.84 does not require renewal. However, when the Commission has reason to believe that operation of the equip-

ment concerned may be inconsistent with §§ 18.71 to 18.84, it may require a new certification based on a new set of measurements.

§ 18.32 Certification after maintenance work.

It shall be the responsibility of the operator of the ultrasonic equipment to have such equipment recertificated when changes have been made that might increase the radiated or conducted interference beyond the limits specified in §§ 18.71 to 18.84.

§ 18.83 Prototype certification permitted.

(a) Provision for prototype certification is made on the basis that production units can be expected to exhibit the same radiation characteristics as those of the prototype. Acceptance of prototype certification is based on representations and measurements made by the manufacturer of ultrasonic equipment.

(b) Ultrasonic equipment may be prototype certificated under the same procedure provided for industrial heating equipment in §§ 18.125 and 18.126. The technical limitations for ultrasonic equipment in § 18.72 shall apply, and the report of measurements shall include a showing of capability of compliance with the requirements of § 18.72(e).

(c) Certification of ultrasonic equipment which carries the manufacturer's prototype certification label shall be made pursuant to § 18.116 (a) and (b). [31 F.R. 7822, June 2, 1966]

§ 18.84 Effective date.

(a) All ultrasonic equipment manufactured on or after July 1, 1955 must comply with the rules in §§ 18.71 to 18.84.

(b) Ultrasonic equipment manufactured prior to July 1, 1955, may be utilized until July 1, 1965, providing it complies either with the rules in §§ 18.71 to 18.84 or with the rules for miscellaneous equipment in § 18.261. After July 1, 1965, all such equipment must comply with the rules in §§ 18.71 to 18.84.

Subpart D—Industrial Heating Equipment

§ 18.101 Operation without a license.

Industrial heating equipment may be operated without a license: *Provided*, The design and operation of the equipment complies with the technical limitations in this part for such equipment: *And provided further*, That the equip-

ment has been certificated pursuant to the requirements of this part.

§ 18.102 Technical limitations.

- (a) Industrial heating equipment shall be designed and constructed in accordance with good engineering practice with sufficient shielding and filtering to meet the requirements of this part.
- (b) Industrial heating equipment may be operated on any frequency except frequencies in the bands 490-510 kc/s, 2170-2194 kc/s, and 8354-8374 kc/s. Equipment operating on an ISM frequency may be operated with unlimited radiation on that frequency. Equipment operated on other frequencies must suppress radiation on the fundamental carrier frequency as well as other frequencies as required by this part.
- (c) Industrial heating equipment designed for operation on an ISM frequency shall be adjusted to operate as close to that ISM frequency as practicable.
- (d) Radiation of radio frequency energy from any industrial heating equipment on any frequency below 5725 Mc/s, except ISM frequencies, shall be suppressed so that the radiated field strength does not exceed 10 microvolts per meter at a distance of one mile or more from the equipment.
- (e) Radiation of radio frequency energy from any industrial heating equipment on any frequency above 5725 Mc/s, except ISM frequencies, shall be reduced to the greatest extent practicable.

Note: The Commission will establish definite radiation limits for these frequencies as soon as information regarding equipment operating on these frequencies becomes available.

(f) Filtering between the industrial heating equipment and power lines must be provided to the extent necessary to prevent the radiation of energy from power lines on frequencies other than ISM frequencies with a field strength in excess of 10 microvolts per meter at a distance of one mile or more from the industrial heating equipment and at a distance of 50 feet from the power line.

§ 18.105 Inspection of industrial heating equipment.

(a) Industrial heating equipment shall be periodically inspected in order to reaffirm the validity of the certificate required by this part.

- (b) Inspection shall be made at sufficiently frequent intervals to insure that each industrial heating equipment is installed, maintained, and operated in a manner that provides compliance with the provisions of this part.
- (c) A log shall be maintained of the inspections made. The inspector shall enter a brief note of his findings and shall date and sign each entry.
- (d) The log shall be maintained at the same location as the certificate.
- (e) The inspector shall require the equipment to be recertificated pursuant to the requirements of this part if he determines, as a result of his inspection, that such action is necessary in order to assure compliance with this part.

§ 18.106 Renewal of certificate.

- (a) The certificate required to be exhibited by this part shall be renewed:
- (1) When changes have been made that might increase the radiated interference beyond the limits specified in this part.
- (2) When the inspector has determined that such action is necessary to assure compliance with the requirements of this part.
- (3) When required by the Commission because it has reason to believe that operation of the equipment concerned may be inconsistent with the requirements of this part.
- (b) The renewal of the certificate shall be based on measurements made at the point of installation.
- (c) After April 30, 1961, the renewal certificate shall be executed on FCC Form 724.

§ 18.107 Measurement of field strength.

Measurements to determine the field strength of radio frequency energy generated by industrial heating equipment shall be made in accordance with standard engineering procedures and shall include the following:

(a) A loop antenna shall be used for measurements on frequencies below 18 Mc/s, and a doublet antenna shall be used for measurements on frequencies above 30 Mc/s. Either a loop or doublet antenna shall be used on frequencies between 18 Mc/s and 30 Mc/s. Appropriate techniques shall be resorted to for measurements in the microwave region of the spectrum.

- (b) Prior to the determination of the maximum field strength at one mile, a sufficient number of measurements shall be made in the vicinity of the industrial heating equipment to enable plotting of the polar radiation pattern and to assure the correct determination of the major lobes. Where conditions permit, these measurements shall be made at intervals of not more than 20 degrees in azimuth directions and at distances not exceeding 1,000 feet from the location of the equipment. The measurements so obtained shall be reduced to equivalent field strength at 1,000 feet.
- (c) The field strength measurements for the maximum field strength at one mile shall be made along the radial corresponding to the lobe of maximum radiation as determined from the polar radiation pattern. Sufficient measurements shall be made along radials extending through all lobes which are within 15 db of the apparent maximum lobe, as determined in paragraph (b) of this section to assure that the assumed lobe of greatest field strength is in fact the maximum lobe. If two or more lobes of radiation of approximately the same strength are present, measurements to determine field strength shall be made along the several radials for such lobes. Where possible, field strength measurements shall be made along each radial at intervals of not greater than 500 feet and an average curve drawn for measured field strength in microvolts per Where meter versus distance in feet. necessary, the average curve shall be extended to show the extrapolated field strength at one mile. In these cases where it is impractical to conduct measurements along the radial of maximum radiation a sufficient number of field strength measurements shall be made to clearly indicate the magnitude of the radiation field in the sector containing the lobe of maximum radiation.
- (d) Where there is evidence of radiation from power lines, field strength measurements shall be made at not less than three points along the power line located approximately 1 mile from the location of the industrial heating equipment causing such radiation and to include a length of power line not less than 500 feet. One point of measurement shall lie within the 1-mile distance and the others beyond. At each of these

- points at least three measurements of field strength shall be made along a line normal to the power line and out to a distance from the power line not exceeding 50 feet measured horizontally along the ground from a point directly below the outermost conductor.
- (e) The field strengths specified herein refer to the maximum field strengths, regardless of polarization, measured at a height of 12 feet above the immediate terrain or at such lower height at which the field strengths may exceed that at 12 feet. Measurements made at frequencies below 18 Mc/s may be made at any convenient height.
- (f) The spectrum shall be investigated from the lowest frequency generated in the equipment up to the tenth harmonic of the fundamental frequency or to 5725 Mc/s whichever is lower.

§ 18.108 Location of equipment.

For the purpose of measurements required in order to execute a certification of compliance, the location of the industrial heating equipment may be considered to be the actual physical location of the equipment, or, where several such units are grouped within a circle of 500 feet radius or less, the several units may. at the election of the certifying engineer. be considered as a single unit, located at the center of the smallest enclosing cir-If the certification includes several units treated as one equipment, the distance of one mile at which the maximum permissible radiation is determined shall be reduced by the radius of the smallest circle that encloses the several units.

§ 18.109 Report of radiation measurements.

The report of radiation measurements shall contain the following information:

- (a) A description of the equipment that was measured for radiation, including: manufacturer, type number, nominal operating frequency, and nominal power rating.
- (b) A listing of the measuring equipment used, including the serial numbers.
- (c) A statement of the date when the measuring equipment was last calibrated.
- (d) The date the measurements were made.
- (e) The frequency range that was investigated.
 - (f) [Reserved]

(g) If the required range of investigation includes the following frequencies, indicate the magnitude of the field measured on these frequencies or in these frequency bands:

Mc/s 74.6 to 75.4 108.0 to 118.0 121.5 156.8 243.0 828.6 to 335.4 420.0 to 460.0

- (h) A graph, taken at any convenient point, of field strength versus frequency as required by § 18.107(f). The graph shall also show the ambient noise level if the same exceeds 6 microvolts per meter. Label the graph to show where the measurements were made.
- (i) A graph of the polar radiation pattern as required by § 18.107(b). Label the graph to show the frequency that was used.
- (j) A graph of field strength versus distance along the radial of maximum radiation shown in the polar graph required by paragraph (i) of this section. Label the graph to show the frequency that was used.
- (k) A statement of the operating conditions that must be observed to ensure that radiation during routine operation does not exceed, within reasonable limits, the radiation that was measured and is reported herein.

Note: In the graphs required by paragraphs (i) and (j), use the fundamental frequency when this frequency falls outside an ISM frequency band; however, if the fundamental is an ISM frequency, use the harmonic, falling outside an ISM frequency band, which has the highest measured radiation.

§ 18.111 Form of certificate.

- (a) Certificates issued after April 30, 1961, for industrial heating equipment shall be executed on FCC Form 724 except as provided in paragraph (b) of this section.
- (b) Where the industrial heating equipment is identical to a prototype which has been tested for radiation prior to April 30, 1961, the certificate heretofore issued by the manufacturer for such equipment may be substituted for Part III of FCC Form 724 until January 1, 1970.

§ 18.112 Certification regarding operation.

The certification required in Part I of FCC Form 724 shall be executed by the owner or lessee of the equipment, in the case of a proprietorship; by one of the partners, in the case of a partnership; or by an officer or authorized employee, in the case of a corporation. If Part I is signed by an authorized employee, Part II shall be executed either by an officer or a party with respect to whom there is on file with the Commission a letter from an officer of the corporation authorizing that party to execute Part II of FCC Form 724.

[30 F.R. 14200, Nov. 11, 1965]

§ 18.113 Certification regarding radiation.

The certification required in Part III of FCC Form 724 shall be executed by an engineer skilled in making and interpreting field strength measurements. The Commission may require such engineer to furnish proof of his qualifications.

§ 18.114 Prototype certification permitted.

- (a) Provision for prototype certification is made on the basis that production units can be expected to exhibit the same radiation characteristics as those of the prototype. Acceptance of prototype certification is based on representations and field strength measurements made by the manufacturer of industrial heating equipment.
- (b) The procedure for prototype certification is set out in §§ 18.125 and 18.126.

[30 F.R. 7999, June 22, 1965]

§ 18.115 Compliance with installation instructions.

Where the certification regarding radiation (Part III of FCC Form 724) is based on measurements of a prototype, the equipment shall be installed in accordance with the instructions which the engineer certifying to Part III of FCC Form 724 has certified as being adequate to ensure reasonable expectation of compliance with the radiation limits in § 18.102.

§ 18.116 Certificate to be filed with Commission.

(a) Two copies of FCC Form 724 required by § 18.101 and § 18.111 shall be

filed. The original shall be filed with the Secretary, Federal Communications Commission, Washington, D.C., 20554. A duplicate copy shall be filed with the Engineer in Charge of the field office having cognizance over the area in which the heater is located.

- (b) For equipment which carries the manufacturer's prototype certification label, the certificate shall consist of Part I and Part II, if applicable, of FCC Form 724.
- (c) For equipment which does not carry the manufacturer's prototype certification label, the certificate shall consist of Parts I, II, if applicable, and III of FCC Form 724 plus the report of radiation measurement pursuant to § 18.109. A facsimile copy of the properly completed and signed Part III of FCC Form 724 may be submitted in lieu of the original thereof.

[28 F.R. 12533, Nov. 22, 1965, as amended at 30 F.R. 7999, June 22, 1965]

§ 18.117 Copy of certificate with equipment.

A copy of the certificate filed with the Commission pursuant to \$18.116 shall be retained by the operator and shall be attached to the equipment. Alternatively, the copy of the certificate may be placed at any location where it will be conveniently available for inspection by authorized representatives of the Commission, provided there is attached to the equipment a notice stating where the copy of the certificate is located.

§ 18.118 Rejection of certificate.

- (a) A certificate that is incomplete or otherwise does not meet the requirements of Subpart D of this part may be rejected.
- (b) The certificate shall be considered accepted unless rejected in writing within 60 days of receipt by the Commission.
- (c) In the event a certificate is rejected but with no harmful interference involved, the equipment may be operated for a period of 20 days from the date of the rejection notice pending the submission of an acceptable certificate.
- (d) If the certificate is rejected after the second submission thereof, the equipment may not be operated until a certificate has been filed with and accepted by the Commission.

[28 F.R. 12533, Nov. 22, 1965, as amended at 80 F.R. 7999, June 22, 1965]

§ 18.119 Elimination and investigation of harmful interference.

- (a) The operator of industrial heating equipment that causes harmful interference shall promptly take appropriate measures to eliminate the harmful interference.
- (b) When notified by the Commission that his installation is causing harmful interference, the operator shall arrange for an engineer skilled in interference measurements and control techniques to make an investigation to ensure that the harmful interference has been eliminated. The Commission may require the engineer making the investigation to furnish proof of his qualifications.
- (c) The results of the investigation required by paragraph (b) of this section shall be reported to the Commission's Engineer-in-Charge of the district office in accordance with the provisions of § 18.122.
- (d) If the equipment has not been certificated or if renewal of the certificate is required, a certificate covering the equipment shall be filed in accordance with the provisions of § 18.116.

§ 18.120 Interference to a radionavigation or safety service.

- (a) If the operator of industrial heating equipment is notified by the Commission that operation of such equipment is endangering the functioning of a radionavigation or a safety service, he shall immediately cease operating the equipment.
- (b) Operation may be resumed on a temporary basis, with the permission of the Engineer-in-Charge of the district office, but only for the purpose of eliminating the harmful interference, making the interference investigation, and obtaining or renewing certification if required. Requests for permission to operate on a temporary basis may be made and granted by telephone, but in that event the request and the grant shall be confirmed promptly in writing.
- (c) If the results of the interference investigation demonstrate that the harmful interference has been eliminated, and if the field work involved in certification, when required, demonstrates that the equipment meets the requirements for certification, the operator may, with the permission of the Engineer-in-Charge of the district office, resume full operation for a period

of ten days pending preparation and submission of the final interference report required by § 18.122 and the certificate required by §§ 18.101 and 18.111. Requests for permission to operate under such circumstances may be made and granted by telephone, but in that event the request and the grant shall be confirmed promptly in writing.

(d) Operation may be resumed on a regular basis after the harmful interference has been eliminated and the interference report and certificate, if required, have been filed.

§ 18.121 Interference to other radio services.

- (a) If the operator of industrial heating equipment is notified by the Commission that operation of such equipment is obstructing or repeatedly interrupting an authorized radio service other than a radionavigation or safety service, he shall take prompt measures to eliminate the harmful interference but need not cease operation unless ordered to do so by the Commission.
- (b) If the operator is ordered to cease operation, he may resume operation on a temporary basis, with the permission of the Engineer-in-Charge of the district office, but only for the purpose of eliminating the harmful interference, making the interference investigation, and obtaining or renewing certification. Requests for permission to operate on a temporary basis may be made and granted by telephone, but in that event the request and the grant shall be confirmed promptly in writing.
- (c) If the results of the interference investigation demonstrate that harmful interference has been eliminated, and if the field work involved in certification, if required, demonstrates that the equipment meets the requirements for certification, the operator may, with the permission of the Engineer-in-Charge of the district office, resume full operation for a period of ten days pending preparation and submission of the final interference report required by § 18.122 and the certificate required by §§ 18.101 and 18.111. Requests for permission to operate under such circumstances may be made and granted by telephone, but in that event the request and the grant shall be confirmed promptly in writing.
- (d) Operation may be resumed on a regular basis after the harmful interference has been eliminated and the in-

terference report and certificate, if required, have been filed.

§ 18.122 Report of interference investigation.

- (a) An interim report on the investigation and of the corrective measures that were taken shall be filed with the Engineer in Charge of the local FCC office within 30 days of notification of harmful interference. The final report shall be filed with the Engineer in Charge within 60 days of notification.
- (b) The date for filing the final report may be extended for 30 days by the Engineer in Charge when the operator has shown that he has been diligent in his efforts and that additional time is required to put into effect the corrective measures or to complete the investigation. The request for extension of time shall be accompanied by a progress report showing what has been accomplished to date.
- (c) The final report on the interference investigation shall list each radio facility which was receiving harmful interference, shall describe the measures taken to eliminate harmful interference. and shall describe the tests made to ensure that harmful interference has been eliminated, together with the test results and the date and time of each test. In the case of interference to broadcast receivers, the final report shall list the location of each receiver that was checked and the name of the receiver owner, shall describe the steps taken to eliminate the harmful interference, and shall specify the date and time each receiver was checked to ensure that harmful interference has been eliminated.

§ 18.125 Prototype certification procedure.

- (a) Manufacturers desiring to prototype certificate their equipment, must file all the information set out in § 18.126. A separate prototype certificate shall be filed for each type of equipment.
- (b) Receipt of each prototype certificate will be acknowledged subject to review at a later date.
- (c) After filing the required information, the equipment may be identified with a label carrying the following statement:

The (name of manufacturer) has prototype certificated this equipment to the FCC as complying with Part 18 of its rules.

(d) The identifying label may be part of the equipment nameplate. If separate

therefrom, the label shall be attached to the equipment in the vicinity of the nameplate.

[30 F.R. 7999, June 22, 1965]

§ 18.126 Information to be filed for prototype certification.

- (a) Original of Part III of FCC Form 724 properly completed and signed.
- (b) Report of radiation measurements pursuant to § 18.109.
- (c) Installation instructions to be furnished to purchaser of the equipment.
- (d) A statement certifying that production will be adequately controlled to insure that all units produced are capable of operating in compliance with the technical requirements of this subpart. This statement shall be signed by a responsible official authorized to sign for the manufacturer and shall show his title.

[30 F.R. 7999, June 22, 1965]

Subpart E—Medical Diathermy Equipment

§ 18.141 Operation on assigned frequencies.

A station license is not required for the operation of medical diathermy equipment on assigned frequencies provided such operation meets the following conditions:

(a) Such operation must conform to the general condition set out in the guarantee or certificate required by paragraphs (c) and (d) of this section. Operation must be confined to one or more of the frequencies:

	Frequency
ISM frequency:	
22,125 Mc/s 1	$\pm 75.0 \mathrm{Mc/s}$
22,125 Mc/s ¹ 2450 Mc/s ¹	$\pm 160.0 kc/s$
5800 Mc/s 1	$\pm 50.0 \mathrm{Mc/s}$
13,560 kc/s	$\pm 20.0 \mathrm{kc/s}$
27,120 kc/s	±25.0 Mc/s
40,680 kc/s	±6.78 kc/s
915 Mc/s 1	±125.0 Mc/s

¹The use of this frequency is subject to the conditions in § 18.14.

(b) Such operation may be without regard to the type or power of emissions being radiated. Spurious and harmonic radiations on frequencies other than those specified above shall be suppressed so that such radiations do not exceed a strength of 25 microvolts per meter at a distance of 1,000 feet or more from the medical diathermy equipment causing such radiations.

(c) With respect to equipment for which type approval has been received from the Commission in accordance with §§ 18.144 to 18.146, inclusive, there shall be affixed to each unit of equipment operated in accordance with paragraphs (a) and (b) of this section, or posted in the room in which such operation occurs, a dated certificate of a competent engineer, or a dated certificate or name plate of the manufacturer of the equipment, setting forth the F. C. C. type approval number for such equipment, the general conditions under which such equipment should be operated, and certifying that the equipment involved may reasonably be expected to meet the requirements of this section under the described conditions of operation for a period of at least three years. tification required in this section shall describe with certainty the apparatus covered thereby.

(d) The owners or operators of equipment which has not received type approval but which is manufactured for operation without a license and designed to meet the technical requirements set forth under paragraphs (a) and (b) of this section shall have posted in the room in which such equipment is operated a dated certificate of a competent engineer, or a dated certificate or name plate of the manufacturer of the equipment, setting forth the general conditions under which such equipment should be operated and certifying that the equipment involved may reasonably be expected to meet the requirements of this section for a period of at least three years under the described conditions of operation. The certification required by this section shall describe with certainty the apparatus covered thereby, and shall include a brief statement of the engineering tests upon which such certification is based and the results thereof. Field strength measurements in such tests shall be made in accordance with § 18.143.

(e) No regular renewal of certification is required for equipment covered in paragraph (c) of this section. The certification required in paragraph (d) of this section shall be renewed at intervals of three years. Notwithstanding the above provisions with respect to renewal of certification, the certification required by paragraph (c) or (d) of this section shall be renewed for particular equipment by such date as the Commission may specify if the Commission has reason to believe that the operation of

such equipment may be inconsistent with provisions of this part or the source of interference to radio communication.

§ 18.142 Operation on unassigned frequencies.

A station license is not required for the operation of medical diathermy equipment on frequencies other than those specified in §18.141(a) provided such operation is in accordance with the general conditions of operation set out in the certification required in paragraph (b) of this section, and meets the following conditions:

(a) The equipment used in such operation shall be provided with a rectified and filtered plate power supply, power line filters and shall be provided with sufficient shielding so that the emission of radio frequency energy generated by such operation, including spurious and harmonic emissions, shall not exceed a strength of fifteen microvolts per meter at a distance of 1,000 feet or more from the medical diathermy equipment on frequencies other than those specified in \$18.141(a) under any conditions of operation.

(b) There shall be affixed to each unit of equipment so operated, or posted in the room in which such operation occurs, a dated certification of a competent engineer, or a dated certificate or name plate of the manufacturer of the equipment setting forth the general conditions under which such equipment should be operated and certifying that under the described conditions of operation the requirements of this section may reasonably be expected to be met for a period of at least 3 years. certification required by this section shall describe with certainty the equipment covered thereby, and shall include a brief statement of the engineering tests upon which the certification is based and Field strength the results thereof. measurements in such tests shall be made in accordance with the provisions of § 18.143.

(c) The certification required in paragraph (b) of this section shall be renewed every 3 years: Provided, That such certification shall be renewed for particular equipment by such earlier date as the Commission may specify if the Commission has reason to believe that the operation of such equipment may be inconsistent with the provisions of this part or a source of interference to radio communication.

§ 18.143 Measurement of field strength.

Measurements to determine the field strength of radio frequency energy generated by medical diathermy equipment shall be made in accordance with standard engineering procedures and shall include the following:

(a) An approved type of field strength meter using loop pickup shall be used for measurements on frequencies below and including 18 Mc/s, and such a meter with a doublet antenna shall be used for measurements for frequencies above 18 Mc/s. Appropriate techniques shall be resorted to for measurements in the microwave region of the spectrum.

(b) The field strength at 1.000 feet from the medical diathermy equipment, or at any other point at which it becomes necessary to determine such field strength shall be determined by measurements at approximately 100foot intervals along 5 radials approximately 72° apart, provided that additional measurements shall be taken when necessary in particular cases. An average curve shall be drawn through the points obtained for each radial and then either (1) the field strength at 1,000 feet taken from the curve or (2) the curve extended to the 1,000-foot point to obtain the field strength at that point. If points of measurement along a radial are such that marked changes of field strength over short distances are noted because of standing waves, multipaths, etc., continuous measurements shall be made along any such radial at points 100 feet apart in order to obtain average values for such points.

(c) The field strength specified in this section refers to the maximum field strength regardless of polarization, measured at a height of 12 feet above the immediate terrain or at such lower height at which the field strength may exceed that at 12 feet.

(d) If due to the location of equipment in a large city, or for some other reason, measurements as outlined above are impractical because of shadows or shielding of large buildings or other objects, every effort should be made to obtain necessary measurements at clear locations such as atop adjacent buildings, etc., with the measurements corrected to the height specified in paragraph (c) of this section in accordance with best available engineering information.

§ 18.144 Procedure for type approval.

(a) Manufacturers of medical diathermy equipment designed to operate on the frequencies specified in §18.141 (a) may submit units of such equipment to this Commission for type approval upon the grant of request therefor made in writing by manufacturer to the Secretary of the Commission. Such a request will not be granted unless at least 5 units of the model to be submitted are scheduled for manufacture and the manufacturer agrees to bear all forwarding and return charges in connection with the shipment of the unit to be tested between the Federal Communications Commission, Laboratory Division, Laurel, Maryland, and the manufacturer.

(b) Any such equipment which is submitted will be tested and a certificate of type approval will be issued to the manufacturer for each type of equipment which meets the following tests:

(1) The frequency at all times during the tests below shall be maintained within 70% of the tolerance specified in § 18.141(a).

(i) From a cold start the machine will be operated continuously at full load for 6 hours, except that machines classified as portable will be subject to a 2-hour test.

(ii) From a cold start the machine will be operated at no load for 5 minutes and then the frequency deviation determined over a normal treatment cycle.

A treatment cycle will be simulated by artificial varying loads and varying settings of the resonance and other operating controls. Similar treatment cycle tests will be conducted after periods of continuous full load operations up to six hours (2 hours for portable operation) to determine the maximum deviation. The number of such tests normally will be determined by the results of test (i): Provided, however, That equipment designed to operate on the frequencies set forth in § 18.141(a) may be granted type approval regardless of frequency stability, provided such equipment meets the other requirements hereof and contains a power cut-off mechanism which is effective in rendering the machine inoperative when the deviation from the assigned frequency exceeds 70 percent of the tolerance provided for.

(2) The equipment must be designed to prevent the emission of spurious and harmonic radiations to the extent required in § 18.141(b).

(3) The electrical and mechanical components of the machine and their installation must be such as to give reasonable assurance of compliance with the requirements of permissible frequency tolerance for at least 5 years.

Note: Medical diathermy equipment operated on 915 Mc/s, 2450 Mc/s, 5800 Mc/s or 22,125 Mc/s will be eligible for type approval upon a determination by the Chief Engineer of compliance with the requirements of the Commission's public notice and order of December 26, 1946, which requirements are set forth in § 18.14.

§ 18.145 Effect of certificate of type approval.

A certificate of type approval constitutes a recognition that on the basis of the tests made the equipment appears to have the capability of functioning in accordance with the provisions of § 18.141 (a) and (b) provided such equipment is properly constructed, maintained and operated, and no change whatsoever is made in the construction of equipment sold under the Certificate of Type Approval issued by the Commission except on specific approval by the Commission to any changes made.

§ 18.146 Withdrawal of certificate of type approval.

(a) A certificate of type approval may be withdrawn if the type of equipment for which it was issued proves defective in service and under usual conditions of maintenance and operation such equipment cannot be relied on to meet the conditions set forth in this part for the operation of the type of equipment involved, or if any change whatsoever is made in the construction of equipment sold under the certificate of type approval issued by the Commission, without the specific prior approval of the Commission.

(b) The procedure for withdrawal of a certificate of type approval shall be the same as that prescribed for revocation of a radio station license pursuant to the provisions of the Communications Act of 1934, as amended.

(c) In the case of withdrawal of a certificate of type approval the manufacturer shall make no further sale of equipment under such certificate.

(d) When a certificate of type approval has been withdrawn for unauthorized changes or for failure to comply with technical requirements, the Commission will consider that fact in determining whether the manufacturer in

question is eligible to receive any new certificate of type approval.

Subpart F—RF Stabilized Arc Welders § 18.181 Technical specifications.

- (a) The requirements of this part with respect to electric arc welding devices using radio frequency energy is suspended, subject to the provisions of paragraphs (b)-(e) of this section, until action is completed in the Docket No. 11467 proceeding with respect to these devices.
- (b) In the event of interference from electric arc welding devices using radio frequency energy to any authorized radio service, steps to remedy such interference shall promptly be taken (except that, in case of interference to receivers arising from direct intermediate frequency pickup by such receivers of the fundamental frequency emissions of certified electric arc welding equipment using radio frequency energy, this provision with respect to interference shall not apply).
- (c) Equipment manufactured after September 1, 1952, shall be subject to the same technical limitations and standards as set forth for industrial heating equipment in §§ 18.101 to 18.108, inclusive, except that such equipment need not be operated within a shielder room or space but in lieu thereof shall be operated with sufficient shielding to limit the radiation to the value prescribed in § 18.102.
- (d) Radio frequency stabilized electric arc welding equipment designed for operation on ISM frequencies may be type approved and operated in accordance with the provisions of § 18.261(c).
- (e) Broad band type of emissions from arc welding equipment shall be measured by an instrument having performance characteristics similar to the "Proposed American Standards Specification for a Radio Noise Meter-0.15 to Megacycles/second" dated March 1950, published by the American Standards Association Committee on Radio Coordination C63. Quasi-Electrical peak values of field strength shall be measured and used in determining compliance with § 18.102. Instruments not having characteristics similar to the above-mentioned standards may be used provided suitable correlation factors are used to adjust the field strength readings to values which would be obtained with an instrument having the desired characteristics.

§ 18.182 Certification requirements.

- (a) The certification required by § 18.101 may be based upon field strength measurements made by the manufacturer of the equipment at locations other than the one where the equipment is in use provided such certification includes a statement by the operator of the equipment that the equipment covered thereby has been installed and is being operated in conformity to the instructions issued by the manufacturer.
- (b) The certificate required for RF stabilized arc welders shall be executed by an engineer skilled in making and interpreting field strength measurements. The Commission may require such engineer to provide proof of his qualifications.
- (c) The certificate for an RF stabilized arc welder measured at the location where it is in use shall contain the following information:
- (1) Type and serial number, or other positive identification, of the welder being certificated.
- (2) Conditions under which the welder shall be operated and maintained.
- (3) Brief description of the engineering tests and a summary of the measured data upon which the certificate is based.
- (4) Date the measurements were made.
- (5) A statement certifying that the welder does meet and may reasonably be expected to continue to meet the requirements of this part.
 - (6) Date of certification.
 - (7) Signature of certifying engineer.
- (8) Name and address of employer of certifying engineer, if any.
- (9) If the certificate is based on measurement of a prototype at some other location:
- (i) Detailed installation instructions which will insure that the welder may reasonably be expected to comply with the radiation limits in § 18.102, and
- (ii) A statement signed by the person responsible for the operation of the welder, attesting that it has been installed in accordance with the installation instructions attached to this certificate.

§ 18.183 Location of certificate.

In general the certificate shall be attached to the equipment. Alternatively the certificate may be placed at any location where it will be conveniently available for inspection by authorized representatives of the Commis-

Sec.

sion, provided there is attached to the equipment a notice stating where the certificate is located.

Subpart G—[Reserved] Subpart H-Miscellaneous Equipment

§ 18.261 Miscellaneous equipment.

(a) The operation without a license of miscellaneous equipment, as defined in \$ 18.3(d), generating radio frequency power of 500 watts or less, shall be in compliance with the provisions of this part for medical diathermy apparatus.

(b) Operation of such equipment generating radiofrequency power in excess of 500 watts shall be in compliance with the requirements for medical diathermy apparatus except that the maximum radiated field permitted shall be increased as the square root of the ratio of the generated power to 500 watts: Provided. That the radiated field shall in no case exceed the fields permitted industrial heating apparatus: And provided further, That equipment used in predominantly residential areas and operating on frequencies below 1,000 Mc/s shall not be permitted the increase in field with power as indicated in this paragraph, but shall be subject to the restrictions contained in this paragraph for diathermy equipment.

(c) Miscellaneous equipment, as defined in § 18.3(d), may be type approved under procedures similar to that for diathermy equipment with such changes in the above procedure as may be required because of the nature of the particular

equipment involved.

(d) For the purpose of field strength measurements, the location of the miscellaneous equipment may be considered to be the actual physical location of such equipment or, where several such units are grouped within a circle of 200 feet radius or less, the several units may, at the election of the certifying engineer, be considered as a single unit, the location of which will be the center of the smallest enclosing circle: Provided, however. That if the certification includes more than one unit, the distance of 1,000 feet at which the maximum permissible radiation is determined shall be decreased by an amount equivalent to the radius of the circle encompassing the several units.

(e) It shall be the responsibility of the operator to have the equipment recertifled when changes have been made that might increase the radiation beyond the specified limits.

§ 18.262 Existing epilation equipment.

The provisions of this part shall not be applicable until June 30, 1954 to epilation equipment, which uses radio frequency energy, manufactured before December 31, 1950, and shall not be applicable until December 31, 1955 for such equipment manufactured between December 31, 1950 and June 30, 1953: Provided, That the foregoing provisions of this section shall be applicable only if such steps as may be necessary are promptly taken to eliminate interference to authorized radio services resulting from the operation of equipment manufactured prior to the respective dates set forth in this section.

PART 19—EMPLOYEE RESPONSIBILI-TIES AND CONDUCT

Subpart A-General Provisions

19.735-101 Purpose. 19.735-102 Definitions. 19.735-104 Issuance, approval, and publication of Commission's regulations. 19.735-105 Interpretation and advisorv service. 19.735-106

porting conflicts of interest. 19.735-107 Disciplinary and other remedial action.

Reviewing statements and re-

Subpart B—Ethical and Other Conduct and Responsibilities of Employees

19.735-201 Specific provisions.

19.735-201a Proscribed actions.

19.735-202 Gifts. entertainment, and favors.

19.735-203 Outside employment and other activity.

19.735-204 Financial interests.

19.735-205 19.735-206 Use of Government property.

Misuse of information.

19.735-207 Indebtedness.

19.735-208 Gambling, betting, and lotteries. 19.735-209 General conduct prejudicial to the Government

19.735-210 Miscellaneous statutory provisions.

Subpart C-Ethical and Other Conduct and Responsibilities of Special Government Employees

19.735-301 Specific provisions.

19.735-302 Use of Government employment.

19.735-303 Use of inside information.

19.735-304 Coercion.

19.735-305 Gifts, entertainment. and favors.

19.735-306 Miscellaneous statutory provisions.

Subpart D—Statements of Employment and Financial Interests

	i ilialiciai illielesis
Sec.	
19.735 -4 01	Form and content of statements.
19.735-402	Specific provisions of agency
	regulations for employees.
19.735-403	Employees required to submit
	statements.
19.735-403a	Employee's complaint on filing
	requirement.
19.735-404	Commissioners not required to
	submit statements.
19.735-405	Time and place for submission
	of employees' statements.
19.735 -4 06	Supplementary statements.
19.735 -4 07	Interests of employees' relatives.
19.735-408	Information not known by em-
	ployees.
19.735 -4 09	Information prohibited.
19.735-410	Confidentiality of employees'
	statements.
19.735 -4 11	Effect of employees' statements
	on other requirements.
19.735-412	Procedure for obtaining waivers
	of the applicability of the Fed-
	eral conflicts of interest stat-
	utes.
19.735-413	Specific reporting requirements
	for special Government em-
	ployees.
	Project.

AUTHORITY: The provisions of this Part 19 issued under E.O. 11222; 3 CFR, 1964-1965 comp., CFR 735.104, unless otherwise noted.

Subpart A—General Provisions

SOURCE: The provisions of this Subpart A appear at 31 F.R. 2722, Feb. 15, 1966, unless otherwise noted.

§ 19.735-101 Purpose.

The effectiveness of the Commission in serving the public interest depends upon the extent to which the Commission holds the confidence and esteem of the Nation's citizens. To hold the public confidence, unusually high standards of honesty, integrity, impartiality, and conduct must be maintained within the Commission and all officers and employees must not only obey the literal requirements of the Federal laws and orders governing official conduct, but also show by their conduct that they support the ethical principles which underlie these laws and regulations. avoidance of misconduct and conflicts of interest on the part of Commission employees through informed judgment is indispensable to the maintenance of these standards. To again call the attention of Federal employees to the importance of maintaining these high moral and ethical standards, the President has issued Executive Order 11222 to codify, clarify, and strengthen the standards of ethical conduct and to set forth a new program assigning central responsibility to the Civil Service Commission for supervising agency action in this regard. In consequence thereof, the Commission has revised its long standing regulations promulgating standards of conduct for all Commission employees and has delegated to the Chairman responsibility for the detection and prevention of acts, short of criminal violations, which could bring discredit upon the Commission and the Federal service.

§ 19.735-102 Definitions.

- (a) "Commission" means the Federal Communications Commission.
- (b) "Employee" means an officer or employee of the Commission including the Commissioners, but does not include a special Government employee or member of the uniformed services.
- (c) "Executive order" means Executive Order 11222 of May 8, 1965.
- (d) "Person" means an individual, a corporation, a company, an association, a firm, a partnership, a society, a joint stock company, or any other organization or institution.
- (e) "Special Government employee" means a "special Government employee," as defined in section 202 of title 18 of the United States Code, that is, one appointed to serve with or without compensation, for not more than 130 days during any period of 365 days on a full-time or intermittent basis, who is employed in the Commission, but does not include a member of the uniformed services.
- (f) "Uniformed services" has the meaning given that term by section 101 (3) of title 37 of the United States Code.
- (g) "Civil Service Regulations" mean the regulations (5 CFR Part 735) on employee responsibilities and conduct issued by the Civil Service Commission on October 1, 1965, in implementation of Executive Order 11222.
- (h) "Communications Act" means the Federal Communications Act of 1934, as amended, 47 U.S.C. 151 et seq.
- § 19.735–104 Issuance, approval, and publication of Commission's regulations.
- (a) The regulations in this part have been prepared in accordance with 5 CFR Part 735 of the Civil Service Regulations to:
- (1) Implement the requirements of law, the Executive order, and Part 735 of the Civil Service Regulations; and to

- (2) Prescribe additional standards of ethical and other conduct and reporting requirements that are appropriate to the particular functions and activities of the Commission and are not inconsistent with law, the Executive order, and Part 735 of the Civil Service Regulations.
- (b) After Civil Service Commission approval, the Chairman shall:
- (1) Submit the Commission's regulations in this part to the Federal Register for publication;
- (2) Furnish each employee and special Government employee with a copy of Administrative Order No. 10, and the Commission's regulations in this part, as revised, within 90 days after approval by the Civil Service Commission;
- (3) Furnish each new employee and special Government employee with a copy of Administrative Order No. 10 and the Commission's regulations in this part, as revised, at the time of his entrance on duty;
- (4) Within 60 days after receipt of Administrative Order No. 10 and the Commission's regulations in this part, it shall be the responsibility of the Head of each Office and Bureau to secure from every person subject to his administrative supervision a statement indicating that the individual has read and is familiar with the contents of the revised order and regulations in this part, and to advise the Executive Director of the Commission that all persons are familiar with the revised order and regulations in this part. Each new employee shall execute a similar statement at the time of entrance on duty. Periodically, and at least once a year, the Executive Director shall take appropriate action to insure that the Head of each Office and Bureau shall remind employees subject to his administrative supervision of the content of Administrative Order No. 10 and the regulations in this part.
- (5) In order to assure the availability of counseling to each employee and special Government employee as provided in § 19.735–105, the General Counsels shall be the counselor for the Commission and the designee to the Civil Service Commission on matters covered by the regulations in this part.
- (6) Copies of laws, the Executive order, the Civil Service Commission Regulations, and this agency's regulations and instructions relating to ethical and other conduct shall be available in the offices of the Executive Director and the Bureau

and Office chiefs for review by employees and Special Government employees.

(c) The Commission's regulations contained in this part are effective only after approval by the Civil Service Commission and publication in the FEDERAL REGISTER.

(d) This section applies to any amendment of the regulations contained in this part.

§ 19.735–105 Interpretation and advisory service.

- (a) The General Counsel is designated as counselor for the Commission to provide guidance on matters relating to ethical conduct and to serve as the Commission's designee to the Civil Service Commission on matters covered by this part. The Office of the General Counsel is responsible for coordination of the Commission's counseling services provided under paragraph (b) of this section and for assuring that counseling and interpretation on questions of conflict of interest and other matters covered by this part are available.
- (b) The counseling services provided by the Office of the General Counsel include the giving of advice and guidance to each employee and special Government employee who seeks advice and guidance on questions of conflicts of interest and other matters covered by this part.
- (c) If an employee is in doubt about any matter covered by this part, or if he has a question as to the propriety of a past or contemplated line of conduct, he should discuss his problem with his immediate supervisor, or the Office of General Counsel.
- (d) Requests for interpretative rulings concerning the applicability of the new order and regulations in this part in implementation thereof may be submitted through the employee's supervisor to the Office of the General Counsel.
- (e) Within 90 days after approval of the Commission's regulations in this part by the Civil Service Commission, and periodically thereafter, the Commission's employees and special Government employees shall be notified of the availability of counseling services and of how and where these services are available. In the case of a new employee or special Government employee appointed after this notification, the notification shall be made at the time of his entrance on duty.

§ 19.735–106 Reviewing statements and reporting conflicts of interest.

- (a) Statements of employment and financial interests submitted under Subpart D of this part shall in the case of employees generally, be reviewed by the Executive Director, in the case of Heads of Offices and Bureaus, be reviewed by the Chairman; and in the case of an employee in the immediate office of a Commissioner, be reviewed by the individual Commissioner. Financial statements of all employees shall be filed in the office of the Executive Director.
- (b) When a statement submitted under Subpart D of this part or information from other sources indicates a conflict between the interests of an employee or special Government employee and the performance of his services for the Government, the information concerning the conflict or appearance of conflict shall be reported to the Executive Director and the employee or special Government employee concerned shall be provided an opportunity to explain the conflict or appearance of conflict.
- (c) When after explanation by the employee or special Government employee involved, the conflict or appearance of conflict is not resolved, the information concerning the conflict or appearance of conflict shall be reported to the Chairman through the counselor for appropriate administrative action.

§ 19.735-107 Disciplinary and other remedial action.

- (a) A violation of the regulations in this part by an employee or special Government employee may be cause for appropriate disciplinary action which may be in addition to any penalty prescribed by law.
- (b) In carrying out the Commission's Review and Inspection Program with respect to employee conduct, the Chairman will designate an officer or employee of the Commission who will promptly investigate all incidents or situations in which it appears that employees may have engaged in improper conduct. Such investigation will be initiated in all cases where complaints are brought to the attention of the Chairman, including: Adverse comment appearing in publications; complaints from members of Congress, private citizens, organizations, other Government employees or agencies; and formal complaints referred to the Chairman by the counselor for the Commission.

- (c) When, after consideration of the explanation of the employee or special Government employee provided by § 19.735-106(b), the Chairman decides that remedial action is required, he shall take immediate action to end the conflicts or appearance of conflicts of interest. Remedial action may include, but is not limited to:
 - (1) Changes in assigned duties;
- (2) Divestment by the employee or special Government employee of his conflicting interest;
- (3) Action under the Commission's Review and Inspection Program resulting in one of the following actions:
- (i) When investigation reveals that the charges are groundless the person designated by the Chairman to assist in administration of the program may give a letter of clearance to the employee concerned, and the case will not be recorded in his official personnel folder.
- (ii) If, after investigation, the case investigator deems the act to be merely a minor indiscretion, he may resolve the situation by discussing it with the employee. The case will not be recorded in the employee's official personnel folder.
- (iii) If the case administrator considers the problem to be of sufficient importance, he may call it to the attention of the Chairman, who in turn may notify the employee of the seriousness of his act and warn him of the consequences of a repetition. The case will not be recorded in the employee's official personnel folder, unless the employee requests it.
- (iv) The Chairman may, when in his opinion circumstances warrant, establish a special review board to investigate the facts in a case and to make a full report thereon, including recommended action.
- (v) If the Chairman decides that formal disciplinary action should be taken, he may prepare for Commission consideration a statement of facts and recommend one of the following:
- (a) Written reprimand. A formal letter containing a complete statement of the offense and official censure;
- (b) Suspension. A temporary nonpay status and suspension from duty;
- (c) Removal for cause. Separation for cause in case of a serious offense.

Only after a majority of the Commission approves formal disciplinary action will any record resulting from the adminis-

tration of this program be placed in the employee's official personnel folder.

(4) Disqualification for a particular assignment.

Remedial action, whether disciplinary or otherwise, shall be effected in accordance with any applicable laws, Executive orders, and regulations.

Subpart B—Ethical and Other Conduct and Responsibilities of Employees

§ 19.735-201 Specific provisions.

The regulations issued under this subpart contain provisions covering the standards of and governing the ethical and other conduct of FCC employees set forth in §§ 735.201a through 735.210 of the Civil Service Regulations (5 CFR 735.201a-735.210), as well as those set forth in the Executive order and the Federal Conflicts of Interest statutes and the Federal Communications Act of 1934, as amended. They are not meant to restrict unduly a Commission employee's social activities. Each employee must judge for himself whether his social activities may or may not compromise or appear to compromise his position as a public servant.

[31 F.R. 2723, Feb. 15, 1966]

§ 19.735-201a Proscribed actions.

An employee shall avoid any action, whether or not specifically prohibited by this subpart, which might result in, or create the appearance of:

(a) Using public office for private

gain;

(b) Giving preferential treatment to any person:

(c) Impeding Government efficiency or economy;

(d) Losing complete independence or impartiality;

(e) Making a Government decision outside official channels; or

(f) Affecting adversely the confidence of the public in the integrity of the Government.

[32 F.R. 13457, Sept. 26, 1967]

§ 19.735–202 Gifts, entertainment, and favors.

(a) Except as provided in paragraphs
(b) and (f) of this section, an employee shall not solicit or accept, directly or indirectly, any gift, gratuity, favor, enter-

tainment, loan, or any other thing of monetary value, from a person who:

(1) Has, or is seeking to obtain, contractual or other business or financial relations with the Commission;

(2) Conducts operations or activities that are regulated by the Commission; or

(3) Has interests that may be substantially affected by the performance or non-performance of his (the employee's) official duty: or

(4) Is in any way attempting to affect the employee's official actions at the

Commission.

(b) The prohibitions enumerated in paragraph (a) of this section do not apply in the situations enumerated below:

- (1) Where obvious family (such as those between the parents, children, or spouse of the employee and the employee) or other personal relationships make it clear that it is those relationships rather than the business of the persons concerned which are the motivating factors;
- (2) Food and refreshments of nominal value may be accepted on infrequent occasions in the ordinary course of a luncheon or dinner meeting or other meeting or on an inspection tour where an employee may properly be in attendance;

(3) Loans may be obtained from banks or other financial institutions on customary terms to finance proper and usual activities of employees, such as home mortgage loans;

(4) Unsolicited advertising or promotional material such as pens, pencils, note pads, calendars and other items of nominal intrinsic value may be accepted, as well as literature relating to the commu-

nications field.

(c) [Reserved]

(d) An employee shall not solicit a contribution from another employee for a gift to an official superior, make a donation as a gift to an official superior, or accept a gift from an employee receiving less pay than himself (5 U.S.C. 7351). However, this paragraph does not prohibit a voluntary gift of nominal value or donation in a nominal amount made on a special occasion such as marriage, illness, or retirement.

(e) An employee shall not accept a gift, present, decoration, or other thing from a foreign government unless authorized by Congress as provided by the Constitution and in Public Law 89-673,

80 Stat. 952.

(f) Neither this section nor § 19.735-203 precludes an employee from receipt of bona fide reimbursement, unless prohibited by law, for expenses of travel and such other necessary subsistence as is compatible with this part for which no Government payment or reimbursement is made. However, this paragraph does not allow an employee to be reimbursed. or payment to be made on his behalf, for excessive personal living expenses, gifts, entertainment, or other personal benefits, nor does it allow an employee to be reimbursed by a person for travel on official business under agency orders when reimbursement is proscribed by Decision B-128527 of the Comptroller General dated March 7, 1967.

[31 F.R. 2723, Feb. 15, 1966, 31 F.R. 4834, Mar. 23, 1966, as amended at 32 F.R. 13457, Sept. 26, 1967]

§ 19.735–203 Outside employment and other activity.

- (a) The Commissioners are prohibited from engaging in any other business, vocation, profession, or employment. U.S.C. 154(b)) No Commissioner or employee of the Commission is permitted to be in the employ of or hold any official relation to any person subject to any of the provisions of the Communications Act. (47 U.S.C. 154(b)) In addition, no Commissioner or employee of the Commission may engage in outside employment or other outside activity, with or without compensation, which is in conflict with or otherwise not compatible with the full and proper discharge of his duties and responsibilities as a Commission employee. Incompatible activities include but are not limited to:
- (1) Acceptance of a fee, compensation, gift, payment of expenses, or any other thing of monetary value in circumstances in which acceptance may result in, or create the appearance of a conflicts of interest situation: or
- (2) Outside employment which tends to impair his mental or physical capacity to perform his Commission duties and responsibilities in an acceptable manner; or
- (3) Outside employment or activities which reasonably might be regarded as official Commission actions, or which will bring discredit upon, or cause unfavorable and reasonable criticism of, the Commission or the Government.
- (b) No professional employee of the Commission shall engage in the private practice of his profession unless spe-

cifically authorized by the Commission. Requests for such authorizations shall, in the case of employees generally, be submitted to the Head of the Office or Bureau to which the employee is assigned; in the case of Heads of Offices and Bureaus to the Chairman; and in the case of an employee in the immediate office of a Commissioner to the Commissioner. All pertinent facts regarding the proposed employment, such as the name of the employer, the nature of the work to be performed, and the amount of time involved shall be set forth.

(c) Employees of the Commission are encouraged to engage in teaching, lecturing, and writing that is not prohibited by law. the Executive Order, the Civil Service Regulations, or this chapter. However, an employee of the Commission shall not. either with or without compensation engage in teaching, lecturing, or writing, including teaching, lecturing, or writing for the purpose of the special preparation of a person or class of persons for an examination of the Civil Service Commission or Board of Examiners for the Foreign Service. that is dependent on information obtained as a result of his Government employment, except when that information has been made available to the general public or will be made available on request, or when the Chairman gives written authorization for the use of nonpublic information on the basis that the use is in the public interest. Articles written by the staff shall not identify the author with the Commission or the Federal government unless prior approval has been obtained: In the case of employees generally, from the Executive Director upon the recommendation of the appropriate Bureau Chief; in the case of Heads of Offices and Bureaus, from the Chairman; and in the case of an employee in the immediate office of a Commissioner, from the individual Commissioner. Nor shall documents prepared in the course of official duties be used for private gain by any Commission employee. In addition, the Commissioners shall not receive compensation or anything of monetary value for any consultation, lecture, discussion, writing, or appearance the subject matter of which is devoted substantially to the responsibilities, programs, or operations of the Commission, or which draws substantially on official data or ideas which have

not become part of the body of public information. (See also 47 U.S.C. 154(b) and § 735.202(b) (2).)

- (d) [Reserved]
- (e) This section does not preclude a Commission employee from:
 - (1) [Reserved]
- (2) Participation in the activities of national or State political parties not proscribed by law.
- (3) Participation in the affairs of or acceptance of an award for a meritorious public contribution or achievement given by a charitable, religious, professional, social, fraternal, nonprofit educational and recreational, public service, or civic organization.
- [31 F.R. 2723, Feb. 15, 1966, as amended at 32 F.R. 13457, Sept. 26, 1967; 33 F.R. 11821, Aug. 21, 1968]

§ 19.735-204 Financial interests.

- (a) An employee of the Commission shall not have a direct or indirect financial interest that conflicts substantially, or appears to conflict substantially, with his Government duties and responsibilities.
- (b) An employee of the Commission shall not engage in, directly or indirectly, a financial transaction as a result of, or primarily relying on, information obtained through his Government employment.
- (c) An employee of the Commission is expected to comply with section 4(b) of the Communications Act and to support its underlying ethical principles.
- (1) Section 4(b) of the Communications Act provides in pertinent part as follows:

No member of the Commission or person in its employ shall be financially interested in the manufacture or sale of radio apparatus or of apparatus for wire or radio communication; in communication by wire or radio or in radio transmission of energy; in any company furnishing services or such apparatus to any company engaged in communication by wire or radio or to any company manufacturing or selling apparatus used for communication by wire or radio; or in any company owning stocks, bonds, or other securities of any such company; * * * nor own stocks, bonds, or other securities of any corporation subject to any of the provisions of this Act * * *.

(2) Section 4(b) has been construed in the past not to prohibit financial interests in enterprises whose activities are not subject, in any significant sense, to regulation by the Commission. However, any employee would be disqualified from

- acting in any matter involving his investments and would be required to seek a waiver under the provisions of 18 U.S.C. 208(b). (See paragraph (e)(2) of this section.)
- (d) No Commissioner shall have a pecuniary interest in any hearing or proceeding in which he participates. (47 U.S.C. 154 (j).)
- (e) An employee of the Commission is also subject to the provisions of Federal conflicts of interest statutes, which are generally applicable to Government employees, but which do not supersede the provisions of section 4(b) of the Communications Act. In summary, the main conflict of interest provisions applying to financial interests are:
- (1) An employee may not, except in the discharge of his official duties, represent anyone else before a court or Government agency in a matter in which the United States is a party or has an interest. This prohibition applies both to paid and unpaid representation of another (18 U.S.C. 203 and 205).
- (i) This prohibition does not extend to (a) representation of another person, without compensation, in a disciplinary, loyalty, or other personnel matter; or (b) the giving of testimony under oath or from making statements required to be made under penalty for perjury or contempt.
- (ii) This prohibition may be waived under the express approval of the Government official responsible for his appointment in the case of an employee who represents, with or without compensation, his own parents, spouse or child, or a person or estate he serves as a fiduciary, but only as to matters in which the employee has not participated personally and substantially as a Government employee and which are not the subject of his official responsibility.
- (2) An employee may not participate in his governmental capacity in any matter in which he, his spouse, minor child, outside business associate or person with whom he is negotiating for employment has a financial interest (18 U.S.C. 208).
- (i) This prohibition shall not apply if the employee advises the Government official responsible for his appointment of the nature of the matter, makes full disclosure of the financial interest, and receives in advance a written determination that the interest is not so substantial as to be deemed likely to affect the integrity of the employee's services.

- (ii) With respect to financial interests of the employee himself, the prohibition of this subparagraph overlaps the prohibitions of section 4(b) of the Communications Act. When the prohibitions of section 4(b) are applicable, the provisions for waiver in subdivision (i) of this subparagraph are not applicable with respect to the interest of the employee himself.
- (3) An employee may not receive any salary, or supplementation of his Government salary, from a private source as compensation for his services to the Government (18 U.S.C. 209).
- (i) This prohibition will not prevent an employee from continuing to participate in a bona fide pension plan or other employee welfare or benefit plan maintained by a former employer.
- (ii) This prohibition is not applicable to anyone serving the Government without compensation or to contributions, awards, or other expenses under the terms of the Government Employees Training Act (5 U.S.C. 2301–2319).

[31 F.R. 2724, Feb. 15, 1966]

§ 19.735–205 Use of Government property.

An employee of the Commission shall not directly or indirectly use, or allow the use of, Government property of any kind, including property leased to the Government, for other than officially approved activities. An employee has a positive duty to protect and conserve Government property, including equipment, supplies, and other property entrusted or issued to him.

[31 F.R. 2725, Feb. 15, 1966]

§ 19.735-206 Misuse of information.

Except as provided in § 19.735-203(c), or as authorized by the Commission, an employee shall not, directly or indirectly, disclose to any person outside the Commission any information, or any portion of the contents of any document, which is part of the Commission's records or which is obtained through or in connection with his Government employment, and which is not routinely available to the public and, with the same exceptions, shall not use any such documents or information except in the conduct of his official duties. Conduct intended to be prohibited by this section includes, but is not limited to, the disclosure of information about the content of or scheduling of agenda items or other staff papers to persons outside the

Commission, and disclosure of actions or decisions by the Commission prior to the public release of such information. (5 CFR Part 735) [34 F.R. 1722, Feb. 5, 1969]

§ 19.735-207 Indebtedness.

An employee shall pay each just financial obligation in a proper and timely manner, especially one imposed by law such as Federal, State, or local taxes. For the purpose of this section, a "just financial obligation" means one acknowledged by the employee or reduced to judgment by a court, and "in a proper and timely manner" means in a manner which the Commission determines does not, under the circumstances, reflect adversely on the Government as his employer. In the event of dispute between an employee and an alleged creditor, this section does not require the Commission to determine the validity or amount of the disputed debt.

[31 F.R. 2725, Feb. 15, 1966]

§ 19.735–208 Gambling, betting, and lotteries.

An employee of the Commission shall not participate, while on Government-owned or leased property or while on duty for the Commission, in any gambling activity including the operation of a gambling device, in conducting a lottery or pool, in a game for money or property, or in selling or purchasing a numbers slip or ticket. However, this section does not preclude activities:

- (a) Necessitated by an employee's law enforcement duties; or
- (b) Under section 3 of Executive Order 10927 and similar agency-approved activities.

[31 F.R. 2725, Feb. 15, 1966]

§ 19.735–209 General conduct prejudicial to the Government.

An employee of the Commission shall not engage in criminal, infamous, dishonest, immoral, or notoriously disgraceful conduct, or other conduct prejudicial to the Commission or to the Government.

[31 F.R. 2725, Feb. 15, 1966]

§ 19.735–210 Miscellaneous statutory provisions.

In addition to the statutes referred to in § 19.735–204, each employee shall acquaint himself with the following provisions that relate to his ethical and other conduct as an employee of the Commission.

- (a) House Concurrent Resolution 175, 85th Congress, 2d Session, 72 Stat. B12, the "Code of Ethics for Government Service."
- (b) Chapter 11 of title 18, United States Code, relating to bribery, graft, and conflicts of interest, as appropriate to the employees concerned.

(c) The prohibition against lobbying with appropriated funds (18 U.S.C.

1913).

- (d) The prohibitions against disloyalty and striking (5 U.S.C. 7311, 18 U.S.C. 1918).
- (e) The prohibition against the employment of a member of a Communist organization (50 U.S.C. 784).
- (f) The prohibitions against (1) the disclosure of classified information (18 U.S.C. 798, 50 U.S.C. 783); and (2) the disclosure of confidential information (18 U.S.C. 1905).
- (g) The provision relating to the habitual use of intoxicants to excess (5 U.S.C. 7352).
- (h) The prohibition against the misuse of a Government vehicle (31 U.S.C. 638a(c)).
- (i) The prohibition against the misuse of the franking privilege (18 U.S.C. 1719).
- (j) The prohibition against the use of deceit in an examination or personnel action in connection with Government employment (18 U.S.C. 1917).
- (k) The prohibition against fraud or false statements in a Government matter (18 U.S.C. 1001).
- (1) The prohibition against mutilating or destroying a public record (18 U.S.C. 2071).
- (m) The prohibition against counterfeiting and forging transportation requests (18 U.S.C. 508).
- (n) The prohibitions against (1) embezzlement of Government money or property (18 U.S.C. 641); (2) failing to account for public money (18 U.S.C. 643); and (3) embezzlement of the money or property of another person in the possession of an employee by reason of his employment (18 U.S.C. 654).
- (o) The prohibition against unauthorized use of documents relating to claims from or by the Government (18 U.S.C. 285).
- (p) The prohibitions against political activities in subchapter III of chapter 73 of title 5, United States Code 18 U.S.C. 602, 603, 607, and 608.
- (q) The prohibition against an employee acting as the agent of a foreign

principal registered under the Foreign Agents Registration Act (18 U.S.C. 219). [31 F.R. 2725, Feb. 15, 1966, as amended at 32 F.R. 13457, Sept. 26, 1967]

Subpart C—Ethical and Other Conduct and Responsibilities of Special Government Employees

SOURCE: The provisions of this Subpart C appear at 31 F.R. 2725, Feb. 15, 1966, unless otherwise noted.

§ 19.735-301 Specific provisions.

The regulations issued under this subpart contain provisions covering the standards of and governing the ethical and other conduct of special Government employees of the Commission as set forth in §§ 735.302 through 735.306 of the Civil Service Regulations (5 CFR 735.302–735.306) as well as those set forth in the Executive order and the Federal Conflicts of Interest Statutes and the Federal Communications Act of 1934, as amended.

§ 19.735–302 Use of Government employment.

A special Government employee shall not use his Government employment for a purpose that is, or gives the appearance of being, motivated by the desire for private gain for himself or another person, particularly one with whom he has family, business, or financial ties.

§ 19.735–303 Use of inside information.

- (a) A special Government employee shall not use inside information obtained as a result of his Government employment for private gain for himself or another person either by direct action on his part or by counsel, recommendation, or suggestion to another person, particularly one with whom he has family, business, or financial ties. For the purpose of this section, "inside information" means information obtained under Government authority which has not become part of the body of public information.
- (b) A special Government employee may teach, lecture, or write in a manner not inconsistent with § 19.735–203(c) in regard to employees.

§ 19.735-304 Coercion.

A special Government employee shall not use his Government employment to coerce, or give the appearance of coercing, a person to provide financial benefit to himself or another person, particularly one with whom he has family, business, or financial ties.



§ 19.735-305 Gifts, entertainment, and

(a) Except as provided in paragraph (b) of this section, a special Government employee, while so employed or in connection with his employment, shall not receive or solicit from a person having business with his agency anything of value as a gift, gratuity, loan, entertainment, or favor for himself or another person particularly one with whom he has family, business, or financial ties.

(b) The same exceptions applying to the acceptance of gifts, entertainment, and favors by Government employees which are set forth under § 19.735-202 (b) apply to special Government em-

ployees of the Commission.

§ 19.735-306 Miscellaneous statutory provisions.

Each special Government employee of the Commission shall acquaint himself with the following major statutory prohibitions relating to his ethical and other conduct as well as those set forth in § 19.735-210 in this part.

- (a) Section 4(b) of the Communications Act which provides that no employee of the Commission "shall be financially interested in the manufacture or sale of radio apparatus or of apparatus for wire or radio communication; in communication by wire or radio or in radio transmission of energy; in any company furnishing services or such apparatus to any company engaged in communication by wire or radio or to any company manufacturing or selling apparatus used for communication by wire or radio; or in any company owning stocks, bonds, or other securities of any such company; nor be in the employ of or hold any official relation to any person subject to any of the provisions of this Act, nor own stocks, bonds, or other securities of any corporation subject to any of the provisions of this Act. * * *"
- (b) Sections 203 and 205 of Title 18 of the United States Code which provide that a special Government employee
- (1) May not, except in the discharge of his official duties, represent anyone else before a court or Government agency in a matter in which the United States is a party or has an interest and in which he has at any time participated personally and substantially for the Govern-
- (2) May not, except in the discharge of his official duties, represent anyone else in a matter pending before the

agency he serves unless he has served there no more than 60 days during the past 365. He is bound by this restraint despite the fact that the matter is not one in which he has ever participated personally and substantially.

(3) These prohibitions do not extend

(i) Representation of another person without compensation in a disciplinary, loyalty, or other personnel matter: or

(ii) The giving of testimony under oath or from making statements required to be made under penalty for perjury or

contempt; or

(iii) Representation, with the approval of the Government official responsible for his appointment, of his own parents. spouse, or child, in matters in which he has not participated personally and substantially and which are not the subject

of his official responsibility.

(c) Section 208 of Title 18 of the United States Code which provides that a special Government employee may not participate in his governmental capacity in any matter in which he, his spouse, minor child, outside business associate or person with whom he is negotiating for employment has a financial interest.

(1) This prohibition shall not apply if the employee advises the Government officials responsible for his appointment of the nature of the matter, makes full disclosure of the financial interest, and receives in advance a written determination that the interest is not so substantial as to be deemed likely to affect the integrity of the employee's services.

(2) With respect to the financial interests of the employee himself, the prohibition of this section 208 overlaps the prohibitions of section 4(b) of the Communications Act. When the prohibitions of section 4(b) are applicable. the provision for waiver in subparagraph (1) of this paragraph is not applicable with respect to the interest of the employee himself.

(d) Section 207(a) of Title 18 of the United States Code which provides that a special Government employee may not. after his Government employment has ended, represent anyone other than the United States in connection with a matter in which the United States is a party or has an interest and in which he participated personally and substantially for the Government.

(e) Section 207(b) of Title 18 of the United States Code which provides that a special Government employee may not, for 1 year after his Government employment has ended, represent anyone other than the United States in connection with a matter in which the United States is a party or has an interest and which was within the boundaries of his official responsibility during the last year of his Government service. This temporary restraint of course gives way to the permanent restriction described in paragraph (d) of this section if the matter is one in which he participated personally and substantially.

Subpart D—Statements of Employment and Financial Interests

SOURCE: The provisions of this Subpart D appear at 31 F.R. 2726, Feb. 15, 1966, unless otherwise noted.

§ 19.735-401 Form and content of statements.

The statements of employment and financial interests required under this subpart for use by employees and special Government employees of the Commission are in accordance with the formats prescribed by the Civil Service Commission in the Federal Personnel Manual.

§ 19.735-402 Specific provisions of agency regulations for employees.

The regulations issued under this subpart are in accordance with the reporting requirements set forth in §§ 735.403 through 735.411 of the Civil Service Regulations (5 CFR 735.403-735.411).

§ 19.735-403 Employees required to submit statements.

Except as provided in § 19.735-404, statements of employment and financial interests shall be required from the following employees of the Commission:

- (a) Employees who are Hearing Examiners.
- (b) Employees who are Members of the Review Board.
- (c) Employees who are in grade GS-13 or above and who are Heads or Assistant Heads of Offices, Bureaus, Divisions, or Branches or comparable units.
- (d) Employees who are in grade GS-13 or above and who are in the immediate offices of the Office or Bureau Chiefs.
- (e) Employees who are in grade GS-13 or above and who are legal, engineering, or other professional assistants to the Commissioners.
- (f) Employees in GS 11 or above who are Heads or Supervisors of field offices. [82 F.R. 18458, Sept. 26, 1967]

§ 19.735—403a Employee's complaint on filing requirement.

An employee who believes that his position has been improperly included under the Commission's regulations requiring the filing of a statement of employment and financial interests is entitled to seek review of his complaint under the Commission's grievance procedure.

[82 F.R. 18458, Sept. 26, 1967]

§ 19.735-404 Commissioners not required to submit statements.

A statement of employment and financial interests is not required by this subpart from the Commissioners who are subject to separate reporting requirements under section 401 of the Executive order.

§ 19.735-405 Time and place for submission of employees' statements.

- (a) An employee required to submit a statement of employment and financial interests pursuant to § 19.735—403, shall submit that statement on the prescribed form not later than:
- (1) Ninety days after the effective date of the Commission's regulations issued under this part if employed on or before that effective date.
- (2) Thirty days after his entrance on duty, but not earlier than ninety days after the effective date, if appointed after that effective date.
- (b) An employee required to submit a statement of employment and financial interests shall submit that statement as follows:
- (1) In the case of employees generally, to the office of the Executive Director:
- (2) In the case of Heads of Offices and Bureaus, to the office of the Chairman; and
- (3) In the case of an employee in the immediate office of a Commissioner, to the individual Commissioners, respectively.
- (c) An employee required to submit a statement of employment and financial interests will be individually notified of his obligation to file.
- [31 F.R. 2726, Feb. 15, 1966, as amended at 32 F.R. 13458, Sept. 26, 1967]

§ 19.735-406 Supplementary statements.

Changes in, or additions to, the information contained in an employee's statement of employment and financial interests shall be reported in a supplementary

statement as of June 30 each year. If no changes or additions occur, a negative report is required. Notwithstanding the filing of the annual report required by this section, each employee shall at all times avoid acquiring a financial interest that could result, or take an action that would result, in a violation of the conflicts-of-interest provisions of section 208 of title 18, United States Code, or Subpart B of this part.

[32 F.R. 13458, Sept. 26, 1967]

§ 19.735—407 Interests of employees' relatives.

The interest of a spouse, minor child, or other member of an employee's immediate household is considered to be an interest of the employee. For the purpose of this section, "member of an employee's immediate household" means those blood relations who are residents of the employee's household.

§ 19.735–408 Information not known by employees.

If any information required to be included on a statement of employment and financial interests or supplementary statement, including holdings placed in trust, is not known to the employee but is known to another person, the employee shall request that other person to submit information in his behalf.

§ 19.735-409 Information prohibited.

This subpart does not require an employee to submit on a statement of employment and financial interests or supplementary statement any information relating to the employee's connection with, or interest in, a professional society or a charitable, religious, social, fraternal, recreational, public service, civic, or political organization or a similar organization not conducted as a business enterprise. For the purpose of this section, educational and other institutions doing research and development or related work involving grants of money from or contracts with the Government are deemed "business enterprises" and are required to be included in an employee's statement of employment and financial interests.

§ 19.735-410 Confidentiality of employees' statements.

Each statement of employment and financial interests, and each supplementary statement, shall be held in confidence and shall be retained in the Office of the Executive Director. Each employee

charged with reviewing a statement is responsible for maintaining the statements in confidence and shall not allow access to or allow information to be disclosed from a statement except to carry out the purpose of this part. Information from these statements shall not be disclosed except as the Civil Service Commission or the Chairman may determine for good cause shown.

[32 F.R. 13458, Sept. 26, 1967]

§ 19.735–411 Effect of employees' statements on other requirements.

The statements of employment and financial interests and supplementary statements required of employees are in addition to, and not in substitution for, or in derogation of, any similar requirement imposed by law, order, or regulation. The submission of a statement or supplementary statement by an employee does not permit him or any other person to participate in a matter in which his or the other person's participation is prohibited by law, order or regulation.

§ 19.735-412 Procedure for obtaining waivers of the applicability of the Federal conflicts of interest statutes.

(a) As indicated in § 19.735-204, an employee may not, except in the discharge of his official duties, represent anyone else before a court or Government agency in a matter in which the United States is a party or has an interest. (18 U.S.C. 203 and 205) Nor may an employee participate in his governmental capacity in any matter in which he, his spouse, minor child, outside business associate or person with whom he is negotiating for employment has a financial interest. (18 U.S.C. 208)

(b) Requests for waivers of the applicability of the above Federal conflicts of interest statutes may be submitted as follows:

(1) In the case of employees generally, to the Head of the Office or Bureau to which the employee is assigned:

(2) In the case of Heads of Offices and Bureaus, to the Chairman; and

(3) In the case of employees in the immediate offices of a Commissioner, to the Commissioner.

(c) All requests for waivers shall be in writing and in the required detail. Before any waiver is granted, the General Counsel shall be furnished a copy of all relevant documents and shall be afforded an opportunity to comment thereon.

- (d) Copies of all requests for waivers and the action taken thereon shall be maintained in the Office of the Executive Director.
- (e) An employee who deems himself disqualified from participating in any matter pending before the Commission and who does not request a waiver shall file a statement as to his disqualification and nonparticipation with the Head of the Bureau or Office to which he is assigned and with the Executive Director.
- § 19.735—413 Specific reporting requirements for special Government employees.
- (a) The reporting requirements for special Government employees are the same as those applying to Government employees which are also set forth in this subpart.
- (b) Except as provided in paragraph (c) of this section, special Government employees shall be required to submit a statement of employment and financial interests which reports:

(1) All other employment; and

- (2) The financial interests of the special Government employee which relate either directly or indirectly to the duties and responsibilities of the special Government employee.
- (c) The Chairman may waive the requirement in paragraph (b) of this section for the submission of a statement of employment and financial interests in the case of a special Government employee who is not a consultant or an expert when the Commission finds that the

duties of the position held by that special Government employee are of a nature and at such a level of responsibility that the submission of the statement by the incumbent is not necessary to protect the integrity of the Government. For the purpose of this paragraph, "consultant" and "expert" have the meanings given those terms by Chapter 304 of the Federal Personnel Manual, but do not include:

(1) A physician, dentist, or allied medical specialist whose services are procured to provide care and service to pa-

tients; or

(2) A veterinarian whose services are procured to provide care and service to

(d) A statement of employment and financial interests required to be submitted under this section shall be submitted not later than the time of employment of the special Government employee as provided in the Commission's regulations. Each special Government employee shall keep his statement current throughout his employment with the Commission by the submission of supplementary statements.

(e) As indicated in § 19.735-306, a special Government employee is subject to certain prohibitions of the Federal conflicts of interest statutes contained in sections 203, 205, and 208 of Title 18 of the United States Code. The procedures for obtaining a waiver of these statutes are the same as those applying to Government employees which are set forth in § 19.735-412.

390

FINDING AIDS

A list of current CFR volumes, a list of superseded CFR volumes, and a list of CFR titles, subtitles, chapters, subchapters and parts are included in the subject index volume to the Code of Federal Regulations which is published separately and revised annually.

Table of CFR Titles and Chapters
Alphabetical List of CFR Subtitles and Chapters
Redesignation Table
List of Sections Affected

ing the first of the second of

Superior Control of Co

Table of CFR Titles and Chapters

Title 1—General Provisions

Chap.

I Administrative Committee of the Federal Register

Appendix A—Guide to record retention requirements

Appendix B—List of acts requiring publication in the Federal Register

Appendix C-Guide to Federal Register Finding Aids

Title 2—The Congress

Chapter I-Parallel Table of Statutory Authorities and Rules

Title 3-The President

- I Proclamations
- II Executive Orders
- III Presidential Documents other than Proclamations and Executive Orders
- IV Codified Text of Selected Presidential Documents
 - V Executive Office of the President

Title 4—Accounts

- I General Accounting Office
- II Federal Claims Collection Standards (General Accounting Office—Department of Justice)

Title 5-Administrative Personnel

- I Civil Service Commission
- III Bureau of the Budget
- IV Civil Service Commission (Equal Employment Opportunity)
- V International Organizations Employees Loyalty Board
- VI Department of Defense
- VII Advisory Commission on Intergovernmental Relations
- VIII National Capital Transportation Agency
 - IX Appalachian Regional Commission
 - X National Capital Housing Authority
 - XI United States Soldiers' Home
- XII District of Columbia Redevelopment Land Agency
- XIII National Commission on Product Safety

Title 6—[Reserved]

Digitized by Google

the contract that I have to

Title 7-Agriculture

Chap.

- SUBTITLE A—Office of the Secretary of Agriculture Subtitle B—Regulations Relating to Agriculture
- I Consumer and Marketing Service (Standards, Inspections, Marketing Practices), Department of Agriculture
- II Food and Nutrition Service, Department of Agriculture
- III Agricultural Research Service, Department of Agriculture
- IV Federal Crop Insurance Corporation, Department of Agriculture
- VI Soil Conservation Service, Department of Agriculture
- VII Agricultural Stabilization and Conservation Service (Agricultural Adjustment), Department of Agriculture
- VIII Agricultural Stabilization and Conservation Service (Sugar), Department of Agriculture
 - IX Consumer and Marketing Service (Marketing Agreements and Orders; Fruits, Vegetables, Nuts), Department of Agriculture
 - X Consumer and Marketing Service (Marketing Agreements and Orders; Milk), Department of Agriculture
 - XI Consumer and Marketing Service (Marketing Agreements and Orders; Miscellaneous Commodities), Department of Agriculture
 - XII Statistical Reporting Service (Agricultural Statistics), Department of Agriculture
- XIV Commodity Credit Corporation, Department of Agriculture
 - XV Foreign Agricultural Service, Department of Agriculture
- XVI Consumer and Marketing Service (Food Stamp Program), Department of Agriculture
- XVII Rural Electrification Administration, Department of Agriculture
- XVIII Farmers Home Administration, Department of Agriculture
 - XXI International Agricultural Development Service, Department of Agriculture
- XXIV Board of Contract Appeals, Department of Agriculture

Title 8—Aliens and Nationality

- I Immigration and Naturalization Service, Department of Justice
- II Office of Alien Property, Department of Justice

Title 9—Animals and Animal Products

- I Agricultural Research Service, Department of Agriculture
- II Packers and Stockyards Administration, Department of Agriculture
- III Consumer and Marketing Service (Meat Inspection), Department of Agriculture

Title 10—Atomic Energy

I Atomic Energy Commission

Title 11-[Reserved]

Title 12—Banks and Banking

Chap.

- I Bureau of the Comptroller of the Currency, Department of the Treasury
- II Federal Reserve System
- III Federal Deposit Insurance Corporation
 - IV Export-Import Bank of the United States
 - V Federal Home Loan Bank Board
 - VI Farm Credit Administration

Title 13—Business Credit and Assistance

- I Small Business Administration
- III Economic Development Administration, Department of Commerce

Title 14—Aeronautics and Space

- I Federal Aviation Administration, Department of Transportation
- II Civil Aeronautics Board
- III National Transportation Safety Board
 - V National Aeronautics and Space Administration

Title 15—Commerce and Foreign Trade

Subtitle A—Office of the Secretary of Commerce

Subtitle B—Regulations Relating to Commerce and Foreign Trade

- I Bureau of the Census, Department of Commerce
- II National Bureau of Standards, Department of Commerce
- III Bureau of International Commerce, Department of Commerce
- IV Foreign-Trade Zones Board
- VI Business and Defense Services Administration, Department of Commerce
- VII Office of State Technical Services, Department of Commerce
- VIII Office of Business Economics, Department of Commerce
 - IX Environmental Science Services Administration, Department of Commerce
 - X Office of Foreign Direct Investments, Department of Commerce
 - XI Office of the Special Representative for Trade Negotiations

Title 16—Commercial Practices

I Federal Trade Commission

Title 17—Commodity and Securities Exchanges

- I Commodity Exchange Authority (Including Commodity Exchange Commission), Department of Agriculture
- II Securities and Exchange Commission



Title 18—Conservation of Power and Water Resources

Chap.

- I Federal Power Commission
- II Tennessee Valley Authority
- III Delaware River Basin Commission
- IV Office of Water Resources Research, Department of the Interior
- V Federal Water Pollution Control Administration, Department of the Interior
- VI Water Resources Council

Title 19—Customs Duties

- I Bureau of Customs, Department of the Treasury
- II United States Tariff Commission
- III Business and Defense Services Administration, Department of Commerce

Title 20—Employees' Benefits

- I Bureau of Employees' Compensation, Department of Labor
- II Railroad Retirement Board
- III Social Security Administration, Department of Health, Education, and Welfare
- IV Employees' Compensation Appeals Board, Department of Labor
- V Manpower Administration, Department of Labor

Title 21—Food and Drugs

- I Food and Drug Administration, Department of Health, Education, and Welfare
- II Bureau of Narcotics and Dangerous Drugs, Department of Justice

Title 22—Foreign Relations

- I Department of State
- II Agency for International Development, Department of State
- III Peace Corps
- IV International Joint Commission, United States and Canada
- V United States Information Agency
- VI United States Arms Control and Disarmament Agency

Title 23—Highways

- I Bureau of Public Roads, Department of Transportation
- II National Highway Safety Bureau, Department of Transportation

Title 24—Housing and Housing Credit

- Subtitle A—Office of the Secretary, Department of Housing and Urban Development
- Subtitle B—Regulations Relating to Housing and Housing Credit

Title 24—Housing and Housing Credit (Continued)

Chap.

- II Federal Housing Administration, Department of Housing and Urban Development
- III Housing Assistance Administration, Department of Housing and Urban Development
- IV Government National Mortgage Association, Department of Housing and Urban Development
 - V Office of Interstate Land Sales Registration, Department of Housing and Urban Development
- VI President's Committee on Equal Opportunity in Housing
- VII Federal Insurance Administration, Department of Housing and Urban Development

Title 25-Indians

- I Bureau of Indian Affairs, Department of the Interior
- II Indian Arts and Crafts Board, Department of the Interior
- III Indian Claims Commission

Title 26—Internal Revenue

- I Internal Revenue Service, Department of the Treasury
- II Tax Court of the United States

Title 27—Intoxicating Liquors

I Internal Revenue Service, Department of the Treasury

Title 28—Judicial Administration

- I Department of Justice
- II Subversive Activities Control Board
- III Federal Prison Industries, Department of Justice

Title 29-Labor

Subtitle A—Office of the Secretary of Labor

Subtitle B—Regulations Relating to Labor

- National Labor Relations Board
- III National Railroad Adjustment Board
- IV Office of Labor-Management and Welfare-Pension Reports,
 Department of Labor
 - V Wage and Hour Division, Department of Labor
- X National Mediation Board
- XII Federal Mediation and Conciliation Service
- XIII Bureau of Labor Standards, Department of Labor
- XIV Equal Employment Opportunity Commission
- XVI Automotive Agreement Adjustment Assistance Board

Title 30-Mineral Resources

4	~	_	_		_
۹	12	О	я	п	n

- I Bureau of Mines, Department of the Interior
- II Geological Survey, Department of the Interior
- IV Federal Coal Mine Safety Board of Review

Title 31—Money and Finance: Treasury

SUBTITLE A—Office of the Secretary of the Treasury

Subtitle B—Regulations Relating to Money and Finance

- I Monetary Offices, Department of the Treasury
- II Fiscal Service, Department of the Treasury
- IV Secret Service, Department of the Treasury
- V Office of Foreign Assets Control, Department of the Treasury
- VI Bureau of Engraving and Printing, Department of the Treasury

Title 32—National Defense

Subtitle A—Department of Defense

- I Office of the Secretary of Defense
- V Department of the Army
- VI Department of the Navy
- VII Department of the Air Force
 - Subtitle B—Other Regulations Relating to National Defense
- XII Defense Supply Agency
- XIII Bureau of Mines, Department of the Interior
- XIV The Renegotiation Board
- XVI Selective Service System
- XVII Office of Emergency Preparedness
- XVIII Office of Civil Defense, Office of the Secretary of the Army

Title 32A—National Defense, Appendix

- I Office of Emergency Preparedness
- VI Business and Defense Services Administration, Department of Commerce
- VII Assistant Secretary of Commerce for Domestic and International Business
- VIII Transport Mobilization Staff, Interstate Commerce Commission
 - IX Bureau of Public Roads, Department of Transportation
 - X Oil Import Administration, Department of the Interior
 - XI Oil Import Appeals Board
- XV Federal Reserve System
- XVIII National Shipping Authority, Maritime Administration, Department of Commerce
 - XIX Office of the Maritime Administrator, Department of Commerce

Title 33—Navigation and Navigable Waters

- I Coast Guard, Department of Transportation
- II Corps of Engineers, Department of the Army
- IV Saint Lawrence Seaway Development Corporation

Title 34—[Reserved]

Title 35-Panama Canal

Chap.

I Canal Zone Regulations

Title 36—Parks, Forests, and Memorials

- I National Park Service, Department of the Interior
- II Forest Service, Department of Agriculture
- III Corps of Engineers, Department of the Army
- IV American Battle Monuments Commission
 - V Smithsonian Institution

Title 37—Patents, Trade-Marks, and Copyrights

- I Patent Office, Department of Commerce
- II Copyright Office, Library of Congress

Title 38—Pensions, Bonuses, and Veterans' Relief

I Veterans Administration

Title 39—Postal Service

I Post Office Department

Title 40—[Reserved]

Title 41—Public Contracts and Property Management

Subtitle A—Federal Procurement Regulations System

- 1 Federal Procurement Regulations
- 2 Federal Aviation Administration
- 3 Department of Health, Education, and Welfare
- 4 Department of Agriculture
- 5 General Services Administration
- 5A Federal Supply Service, General Services Administration
- 5B Public Buildings Service, General Services Administration
- 5C Defense Materials Service, General Services Administration
- 5D Transportation and Communications Service, General Services
 Administration
- 6 Department of State
- 7 Agency for International Development, Department of State
- 8 Veterans Administration
- 9 Atomic Energy Commission
- 10 Department of the Treasury
- 12 Department of Transportation
- 12B Coast Guard, Department of Transportation
- 13 Department of Commerce
- 14 Department of the Interior

Title 41—Public Contracts and Property Management (Continued) Chap.

- 14H Bureau of Indian Affairs, Department of the Interior
- 18 National Aeronautics and Space Administration
- 19 United States Information Agency
- 22 Office of Economic Opportunity
- 23 U.S. Arms Control and Disarmament Agency
- 24 Department of Housing and Urban Development
- 25 National Science Foundation
- 29 Department of Labor
- 39 Post Office Department
 Subtitle B—Other Provisions Relating to Public Contracts
- 50 Public Contracts, Department of Labor
- 51 Committee on Purchases of Blind-Made Products
- 60 Office of Federal Contract Compliance, Equal Employment Opportunity, Department of Labor Subtitle C--Federal Property Management Regulations System
- 101 Federal Property Management Regulations
- 105 General Services Administration
- 109 Atomic Energy Commission
- 114 Department of the Interior
 Subtitle D—Other Provisions Relating to Property Management
 150 to End [Reserved]

Title 42—Public Health

- I Public Health Service, Department of Health, Education, and Welfare
- II Children's Bureau, Social and Rehabilitation Service, Department of Health, Education, and Welfare
- III St. Elizabeths Hospital, Department of Health, Education, and Welfare

Title 43—Public Lands: Interior

Subtitle A-Office of the Secretary of the Interior

Subtitle B-Regulations Relating to Public Lands

- I Bureau of Reclamation, Department of the Interior
- II Bureau of Land Management, Department of the Interior

Title 44—Public Property and Works

- III Department of State
- IV Business and Defense Services Administration, Department of Commerce
- V Library of Congress
- VII Department of Housing and Urban Development (Community Facilities)
- VIII Economic Development Administration, Department of Commerce



Title 45-Public Welfare

Chap.

Subtitle A—Department of Health, Education, and Welfare, General Administration

SUBTITLE B-Regulations Relating to Public Welfare

- I Office of Education, Department of Health, Education, and Welfare
- II Social and Rehabilitation Service (Assistance Programs), Department of Health, Education, and Welfare
- III Bureau of Federal Credit Unions, Social Security Administration, Department of Health, Education, and Welfare
- IV Social and Rehabilitation Service (Rehabilitation Programs), Department of Health, Education, and Welfare
- V Foreign Claims Settlement Commission of the United States
- VI National Science Foundation
- VII Commission on Civil Rights
- VIII Civil Service Commission
 - IX Administration on Aging, Social and Rehabilitation Service, Department of Health, Education, and Welfare
 - X Office of Economic Opportunity
 - XI National Foundation on the Arts and the Humanities

Title 46-Shipping

- I Coast Guard, Department of Transportation
- II Maritime Administration, Department of Commerce
- III Coast Guard (Great Lakes Pilotage), Department of Transportation
- IV Federal Maritime Commission

Title 47—Telecommunication

- I Federal Communications Commission
- II Director of Telecommunications Management

Title 48—[Reserved]

Title 49—Transportation

Subtitle A—Office of the Secretary of Transportation

Subtitle B—Other Regulations Relating to Transportation

- I Hazardous Materials Regulations Board, Department of Transportation
- II Federal Railroad Administration, Department of Transportation
- III Federal Highway Administration, Department of Transportation
- X Interstate Commerce Commission

Title 50-Wildlife and Fisheries

- I Bureau of Sport Fisheries and Wildlife, Fish and Wildlife Service, Department of the Interior
- II Bureau of Commercial Fisheries, Fish and Wildlife Service, Department of the Interior
- III International Regulatory Agencies (Fishing and Whaling)
- IV Fish and Wildlife Service, Department of the Interior

Alphabetical List of CFR Subtitles and Chapters

Title and Subtitle or

	Chapter Reference
Administrative Committee of the Federal Register	1, I
Advisory Commission on Intergovernmental Relations	5, VII
Agency for International Development	22, II
Federal Procurement Regulations System	41, 7
Aging, Administration on	45, IX
Agricultural Research Service	7, III; 9, I
Agricultural Stabilization and Conservation Service	7, VII, VIII
	7, VII, VIII
Agriculture Department	7 III.O I
Agricultural Research Service	7, III; 9, I
Agricultural Stabilization and Conservation Service	7, VIÍ, VIII 7, XIV
Commodity Credit Corporation	
Commodity Exchange Authority	17, I
Commodity Exchange Commission	17, I
Consumer and Marketing Service	7, I, IX, X, XI, XVI; 9,
C 1 D 1 C	III
Contract Appeals, Board of	7, XXIV
Farmers Home Administration	7, XVIII
Federal Crop Insurance Corporation	7, IV
Federal Procurement Regulations System	41, 4
Food and Nutrition Service	7, <u>ÍI</u>
Foreign Agricultural Service	7, XV
Forest Service	36 , I I
International Agricultural Development Service	7, XXI
Packers and Stockyards Administration	9, II
Rural Electrification Administration	7, XVII
Secretary of Agriculture, Office of	7, Subtitle A
Soil Conservation Service	7, VI
Statistical Reporting Service	7, Subtitle A 7, VI 7, XII
Air Force Department	32, VII
Alien Property, Office of	8, II
American Battle Monuments Commission	36, IV
Appalachian Regional Commission	5, IX_
Arms Control and Disarmament Agency, U.S.	22, VI
Federal Procurement Regulations System	41, 14H, 23
Army Department	32, V 32, XVIII
Civil Defense, Office of	32, XVIII
Engineers, Corps of	33, II; 36, III
Arts and the Humanities, National Foundation on the	45, XÏ
Atomic Energy Commission	10, I
Federal Procurement Regulations System	41, 9
Federal Property Management Regulations System	41, 109
Automotive Agreement Adjustment Assistance Board	29, XVI
Blind-Made Products, Committee on Purchases of	41, 51
Budget, Bureau of	5, III
Business and Defense Services Administration	15, VI; 19, III; 32A, VI;
	44, IV
Business Economics, Office of	15, VIII
Canal Zone Regulations	35, I
Census Bureau	15, Î
Children's Bureau, Social and Rehabilitation Service	42, II
Civil Aeronautics Board	14, II
Civil Defense Office, Office of the Secretary of the Army	32, XVIII
Civil Rights Commission	
Civil Service Commission	45, VII 5, I, IV; 45, VIII
International Organizations Employees Loyalty Board	5, V
Don't Don't	~, ·

	Title and Subtitle or Chapter Reference
Coast Guard Federal Procurement Regulations System	33, I; 46, I, III 41, 12B
Commerce Department Assistant Secretary of Commerce for Domestic and Inte	•
national Business Business and Defense Services Administration	32A, VII
· · · · · · · · ·	15, VI; 19, III; 32A, VI; 44, IV
Business Economics, Office of Census Bureau	15, VIII 15, I
Economic Development Administration Environmental Science Services Administration	13, III ; 44, VIII 15, IX
Federal Procurement Regulations System	41, 13
Foreign Direct Investments, Office of Government Inventions Jurisdiction	15, X 37, III
International Commerce, Bureau of Maritime Administration	15, III 32A, XVIII, XIX; 46, II
National Burcau of Standards National Shipping Authority	15, ÍI 32A, XVIII
Patent Office	37, I, III
Secretary of Commerce, Office of Special Representative for Trade Negotiations, Office of	15, Subtitle A of 15, XI
State Technical Services, Office of Commercial Fisheries, Bureau of	15, VII 50, II
Committee on Purchases of Blind-Made Products Commodity Credit Corporation	41, 51 7, XIV
Commodity Exchange Authority	17, I
Commodity Exchange Commission Community Facilities	17, I 44, VII
Comptroller of the Currency, Bureau of Consumer and Marketing Service	12, I 7, <u>I,</u> IX, X, XI, XVI; 9 ,
Contract Appeals, Board of	III 7, XXIV
Copyright Office, Library of Congress Customs Bureau	37, II 19, I
Defense Department	5, VI; 32, Subtitle A, XII
Air Force Department	32, VII 32, V, XVIII; 33, II; 36,
Army Department	III
Navy Department Secretary of Defense, Office of	32, VI 32, I
Defense Materials Service Defense Supply Agency	41, 5C 32, XII
Delaware River Basin Commission District of Columbia Redevelopment Land Agency	18, III 5, XII
Economic Development Administration	13, III; 44, VIII
Economic Opportunity, Office of Education, Office of	41, 22; 45, X 45, I
Emergency Preparedness, Office of	32, XVII; 32A, I
Employees' Compensation, Burcau of Employees' Compensation Appeals Board	20, I 20, IV
Employees Loyalty Board, International Organizations Engineers, Corps of	5, V 33, II : 36, III
Engraving and Printing, Bureau of Environmental Science Services Administration	31, VI 15, IX
Equal Employment Opportunity Equal Employment Opportunity Commission	5, IV; 41, 60 29, XIV
Executive Office of the President	3, V
Export-Import Bank of the United States Farm Credit Administration	12, IV
Farmers Home Administration	12, VI 7, XVIII
Federal Aviation Administration Federal Procurement Regulations System	14, I 41, 2
Federal Claims Collection Standards Federal Coal Mine Safety Board of Review	4, II 30, IV

	Title and Subtitle or Chapter Reference
Federal Communications Commission Federal Contract Compliance, Office of Federal Credit Unions, Bureau of	47, I 41, 60 45, III
Federal Crop Insurance Corporation Federal Deposit Insurance Corporation	7, IV 12, III
Federal Highway Administration Federal Home Loan Bank Board Federal Housing Administration	49, III 12, V 24, II
Federal Insurance Administration Federal Maritime Commission Federal Mediation and Conciliation Service	24, VII 46, IV 29, XII
Federal Power Commission Federal Prison Industries	18, I 28, III
Federal Procurement Regulations Federal Procurement Regulations System Federal Property Management Regulations	41, I 41, Subtitle A 41, 101
Federal Property Management Regulations System Federal Railroad Administration Federal Register, Administrative Committee of	41, Subtitle C 49, II 1, I
Federal Reserve System Federal Supply Service Federal Trade Commission	12, II; 32A, XV 41, 5A 16, I
Federal Water Pollution Control Administration Fiscal Service, Treasury Department Fish and Wildlife Service	18, V 31, II 50, I, II, IV
Commercial Fisheries, Bureau of Sport Fisheries and Wildlife, Bureau of	50, II 50, I
Food and Drug Administration Food and Nutrition Service Foreign Agriculture Service	21, I 7, II 7, XV
Foreign Assets Control, Office of Foreign Claims Settlement Commission of United States Foreign Direct Investments, Office of	31, V 45, V 15, X
Foreign-Trade Zones Board Forest Service General Accounting Office	15, IV 36, II 4, I, II
General Services Administration Federal Procurement Regulations System Federal Property Management Regulations System	41, 1, 5, 5A, 5B, 5C, 5D 41, 101, 105
Geological Survey Government National Mortgage Association Great Lakes Pilotage	30, II 24, IV 46, III
Hazardous Materials Regulations Board Health, Education, and Welfare, Department of Children's Bureau, Social and Rehabilitation Service	49, I 45, Subtitle A 42, II
Education, Office of Federal Credit Unions, Bureau of Federal Procurement Regulations System	45, I 45, III 41, 3
Food and Drug Administration Public Health Service Rehabilitation Service	21, I 42, I 42, II; 45, II, IV, IX
St. Elizabeths Hospital Social Security Administration Highway Administration, Federal	42, III 20, III 49, III
Highway Safety Bureau, National Housing, President's Committee on Equal Opportunity Housing and Urban Development Department	23, II 24, VI
Community Facilities Federal Housing Administration Federal Insurance Administration	44, VII 24, II 24, VII
Federal Procurement Regulations System Government National Mortgage Association	41, 24 24, IV
Housing Assistance Administration Interstate Land Sales Registration, Office of Secretary, Office of	24, III 24, V 24, Subtitle A

Title and Subtitle or Chapter Reference

	Chapter Rete
Immigration and Naturalization Service	8, I
	OF T. 41 14TT
Indian Affairs, Bureau of	25, I; 41, 14H
Indian Arts and Crafts Board	25, II
Indian Claims Commission	25 III
Intergovernmental Relations, Advisory Commission on	25, III 5, VII
	J, V 11
Interior Department	,
Federal Procurement Regulations System	41, 14, 14H, 114
Fish and Wildlife Service	50, I, II, IV
	30, 1, 11, 1 v
Geological Survey	30, II
Indian Affairs, Bureau of	25, I
Indian Arts and Crafts Board	25, II
	40 TT
Land Management Bureau	43, II
Mines, Bureau of	30, I; 32, XIII
National Park Service	36, I
Oil Import Administration	
	32A, X
Reclamation, Bureau of	43, I
Secretary of the Interior, Office of	43, Subtitle A
Water Pollution Control Administration, Federal	18, V
Water Resources Research, Office of	18, IV
Internal Revenue Service	26, I ; 27, I
International Agricultural Development Service	7, XXI
International Commerce, Bureau of	15, III
	00 II. 41 7
International Development, Agency for	22, II; 41, 7
International Joint Commission, United States and Canada	22, IV
International Organizations Employees Loyalty Board	5 V
	5, V 50, III
International Regulatory Agencies (Fishing and Whaling)	30, 111
Interstate Commerce Commission	49, X
Transport Mobilization Staff	32A, VIII
Interstate Land Sales Registration, Office of	24, Ú
incontact name successful and one of	~ -, ·
Justice Department	28, I
Alien Property, Office of	8, II
Federal Claims Collection Standards	4, II
	T, 11
Federal Prison Industries	28, III
Immigration and Naturalization Service	8, I
Narcotics and Dangerous Drugs, Bureau of	21, II
	,
Labor Department	
Employees' Compensation, Bureau of	20, I
Employees' Compensation Appeals Board	20, IV
	41 60
Federal Contract Compliance, Office of	41, 60
Federal Procurement Regulations System	41, 29, 50 29, IV
Labor-Management and Welfare-Pension Reports, Office of	29. IV
	29, XIII
Labor Standards, Bureau of	00 V
Manpower Administration	20, V
Public Contracts	41, 50
Secretary of Labor, Office of	29, Subtitle A
Wage and Hour Division	29, V
	20, 137
Labor-Management and Welfare-Pension Reports, Office of	29, IV
Labor Standards, Bureau of	29, XIII
Land Management, Bureau of	43, II
Library of Congress	44, V
Copyright Office	37, II
Manpower Administration	20, V
	46, II
Maritime Administration	
Maritime Administrator, Office of	32A, XIX
National Shipping Authority	32 A, XVIII
Mines, Bureau of	30, Í; 32, XIII
Monetary Offices, Treasury Department	31 , I
Narcotics and Dangerous Drugs, Bureau of	21, II
	14, 37
National Aeronautics and Space Administration	14, V
Federal Procurement Regulations System	41, 18
National Bureau of Standards	15, II
National Capital Housing Authority	5. X
	รั ข้าบ
National Capital Transportation Agency	15, II 5, X 5, VIII 5, XIII
National Commission on Product Safety	5, XIII

Title and Subtitle or Chapter Reference

	Chapter Reference
National Foundation on the Arts and the Humanities	45, XI
National Highway Safety Bureau	23, II
National Labor Relations Board	29, I
National Mediation Board	29, X
National Park Service	36, I
National Railroad Adjustment Board	29, III
National Science Foundation	45, VI 41, 25
Federal Procurement Regulations System	41, 25
National Shipping Authority	32A, XVIII
National Transportation Safety Board	14, ÍII
Navy Department	32, VI
Oil Import Administration	32A, X
Oil Import Administration	
Oil Import Appeals Board	32 A , XI
Packers and Stockyards Administration	9, II
Panama Canal, Canal Zone Regulations	35, I
Patent Office	37, Ī
Peace Corps	22, III
Post Office Department	39, I
Federal Procurement Regulations System	41, 39
Presidential documents	3, I, II, III, IV , V
Executive orders	3, II, IV
Proclamations	3, I, IV
President's Committee on Equal Opportunity in Housing	24, VI
Procurement Regulations System, Federal	41, Subtitle A
Property Management Regulations System, Federal	41, Subtitle C
Public Buildings Service	41,5B
Public Contracts, Department of Labor	41, 50
Public Health Service	42, I
Public Roads, Bureau of	23, I; 32A, IX
Dellard Administration Endonal	
Railroad Administration, Federal	49, II
Railroad Retirement Board	20, II
Reclamation, Bureau of	43, I
Renegotiation Board	32, XIV
Rural Electrification Administration	7, XVII
St. Elizabeths Hospital	42, III
Saint Lawrence Seaway Development Corporation	33, IV
Secret Service, Treasury Department	31, IV
Securities and Exchange Commission	17, II
	17, 11 20 VVI
Selective Service System	32, XVI
Small Business Administration	13, I
Smithsonian Institution	36, V
Social Security Administration	20, III
Federal Credit Unions, Bureau of	45, III
Soil Conservation Service	7, VI
Sport Fisheries and Wildlife, Bureau of	50, I
State Department	22, I; 44, III
Agency for International Development	22, II; 41, 7
Federal Procurement Regulations System	41, 6
State Technical Services, Office of	15, VII
Statistical Reporting Service, Agriculture Department	7, XII
Subversive Activities Control Board	28, II
Tariff Commission, United States	19, II
Tax Court of the United States	26, II
Telecommunications Management, Director of	47, II
Tennessee Valley Authority	18, II
Trade Negotiations, Office of Special Representative	15, XI
Transport Mobilization Staff, Interstate Commerce Commission	32 A, VIII
Transportation, Department of	00 T 44 405 40 T
Coast Guard	33, I; 41, 12B; 46, I, III
Federal Aviation Administration	14, I; 41, 2
Federal Highway Administration	49, III
Federal Procurement Regulations System	41, 2, 12, 12 B
Federal Railroad Administration	49, II

Title and Subtitle or Chapter Reference

Transportation, Department of (Continued)	Chapter Ref
Hazardous Materials Regulations Board	49, I
National Highway Safety Bureau	23, II
Public Roads, Bureau of	23, I; 32A, IX
Secretary, Office of	49, Subtitle A
Transportation and Communications Service	41, 5D
Treasury Department	•
Comptroller of the Currency, Bureau of	12, I
Customs Bureau	19, I
Engraving and Printing, Bureau of	31, VI
Federal Procurement Regulations System	41, 10
Fiscal Service	31, II
Foreign Assets Control, Office of	31, V
Internal Revenue Service	26, I; 27, I
Monetary Offices	31 , I
Secret Service	31, IV
Secretary of the Treasury, Office of	31, Subtitle A
United States Arms Control and Disarmament Agency	22, VI; 41, 23
United States Information Agency	22 , V
Federal Procurement Regulations System	41, 19
United States Soldiers' Home	5, XI
United States Tariff Commission	19, II
Veterans Administration	38, I
Federal Procurement Regulations System	41, 8
Wage and Hour Division	29, V
Water Pollution Control Administration, Federal	18, V
Water Resources Council	18, VI
Water Resources Research, Office of	18, IV
Welfare-Pension Reports, Office of Labor-Management and	
Wildlife, Bureau of Sport Fisheries and	50, I
•	•

Redesignation Table

Note: For convenience, a redesignation table follows, covering all redesignations in Subchapters B, C and D, as well as in Subchapter A, of this title (28 F.R. 12386, Nov. 22, 1963).

	PART 0	Old New	Old New
No change in	n number of part. Fol-	0.271 0.311	0.402 0.411
	were redesignated.	0.272 0.313 0.281 0.314	0.404 0.415
TOWING SECTIONS	were redesignated.		0.405 0.413
Old New	Old New	0.282 0.317	0.406
0.1 0.1	0.112 0.42	0.284 0.316	(a-f) 0.417
0.2 (a-b) 0.2	0.113 0.43	0.285 0.315	(g) 0.419
(c) 0.181	0.114 0. 44	and 0.385	(h) 0.421
0.3 0.3	0.115 0.45	0.286 0.386	(1) 0.425
0.5 0.5	0.116 0.46	Preceding	(j) 0.423
0.10 0.71	0.117 0.47	0.291 0.333	(k)0.417(b) (2)
0.11 0.73	0.120 0.31	0.291 0.331	(1) 0.417(c)
0.12 0.72	0.121 0.32	0.292 0.332	0.407 0.427
0.13 0.74	0.122 0.33	0.293 0.337	0.408 0.407
0.14 0.75	0.123 0.34	0.322 0.251	0.409 0.441
0.15 0.76	0.124 0.35	0.323 0.257	0.410 0.449
0.16 0.77	0.125 0.36	0.331 0.241	0.411 0.451
0.17 0.78	0.126 0.37	0.332 0.243	0.412 0.453
0.18 0.79	0.131 0.51	0.333 0.247	0.413
0.19 0.80	0.132 0.52	0.341 0.261	(a-b) 0.443
0.20 0.81	0.183 0.53	0.351 0.371(a)	(c) 0.445
0.21 0.91	0.134 0.54	0.361 0.231	0.414 0.447
0.22 0.92	0.135 0.55	0.401	0.415 0.405
0.23 0.93	0.136 0.56	(a-c) 0.401	0.416 0.429
0.24 0.94	0.141 0.171	(d-e) 0.403	0.417 0.431
0.25 0.95	0.151 0.61		
0.26 0.96	0.161 0.181	New Old	New Old
0.27 0.97	0.162 0.184	0.1 0.1	0.55 0.135
0.28 0.98	0.201 0.201	0.2 0.2 (a-b)	0.5 6 0.136
0.29 0.99	0.202 0.203	0.3 0.3	0.61 0.151
0.41 0.111	0.206 0.861	0.5 0.5	0.71 0.10
0.42 0.113	0.207 0.365	0.11 0.91	0.72 0.12
0.43 0.112	0.211 0.211	0.12 0.92	0.73 0.11
0.44 0.113	0.212 0.212(a)	0.13 New	0.74 0.13
0.45 0.114	0.213 0.212(b)	0.14 0.93	0.75 0.14
0.46 0.115	0.214 0.214	0.15 0.94	0.76 0.15
0.47 0.116	0.215 0.215	0.16 0.95	0.77 0.16
0.49 0.121	0.216 0.186	0.17 0.96	0.78 0.17
0.61 0.131	0.217 0.383	0.18 0.98	0.79 0.18
0.62 0.133	0.218 0.218	0.19 New	0.80 0.19
0.63 0.132	0.219 0.381	(See 0.97)	0.81 0.20
0.64 0.133	0.224 0.351	0.20 New	0.91 0.21
0.65 0.134	0.225 0.357	0.31 0.120	0.92 0.22
0.66 0.135 0.67 0.136	0.231 0.341 0.232 0.347	0. 32 0. 121 0. 33 0.1 22	0.93 0.23 0.94 0.24
0.68 0.137	0.241 0.281	0.34 0.123	0.95 0.25
0.69 0.138	0.242 0.287	0.35 0.124	0.96 0.26
0.70 0.139	0.251 0.291	0.36 0.125	0.97 0.27
0.81 0.151	0.252 0.293	0.37 0.126	0.98 0.28
0.82 0.152	0.253 0.294	0.41 0.111	0.99 0.29
0.91 0.11	0.254 0.295	0.42 0.112	0.111 0.41
0.92 0.12	0.255 0.296	0.43 0.113	0.112 0.43
0.93 0.14	0.256 0.297	0.44 0.114	0.113 0.42
0.94 0.15	0.257 0.302	0.45 0.115	and 0.44
0.95 0.16	0.258 0.303	0.46 0.116	0.114 0.45
0.96 0.17	0.259 0.304	0.47 0.117	0.115 0.46
0.97 0.19	0.260 0.305	0.51 0.131	0.116 0.47
and 0.183	0.261 0.307	0.52 0.132	0.121 0.4 9
0.98 0.18	0.262 0.292	0.53 0.133	0.131 0.61
0.111 0.41	0.263 0.298	0.54 0.134	0.132 0. 63

v 014	Non Old	Old Non	Old Non
New Old 0.183 0.62	New Old 0.307 0.261	Old New section section	Old New section
and 0.64	0.311 0.271	1.50 1.42	1.146 1.258
0.134 0.65	0.313 0.272	1.51 1.48	1.147 1.260
0.135 0.66	0.314 0.281	1.52 1.49	1.148 1.261
0.136 0.67	0.815 New	1.53 1.50 1.54 1.51	1.149 1.263 1.150 1.264
0.137 0.68 0.138 0.69	(See 0.285) 0.316 0.284	1.55 1.52	1.151(a, e-f) 1.267
0.139 0.70	0.317 0.282	1.56 1.47	(b-d) _ 1.274
0.151 0.81	0.331 0.291	1.57 1.5	1.152(a) 1.271
0.152 0.82	0.332 0.292	1.63 1.87	(b) 1.273
0.161 New	0.333 Text	1.64 1.110 1.65 1.62	1.1531.276 1.1541.277
(See 0.206) 0.171 0.141	preceding 0.291	1.66 1.68	1.155 1.279
0.181 0.2(c)	0.837 0.298	1.67 1.61	1.156 1.204
and 0.161	0.341 0.231	1.68 1.502	1.157 1.282
0.182 0.91(d)	0.347 0.232	1.70 1.83	1.158 1.203
0.183 0.97	0.351 0.224 0.357 0.225	1.71 1.84 1.72 1.85	1.171 1.351 1.172 1.352
0.184 0.162 0.186 0.216	0.361 0.206	1.73 1.77(g)	1.173 1.353
0.201 0.201	0.365 0.207	1.74 1.77(d)	1.174 1.354
0.203 0.202	0.371(a) _ 0.351	1.75 1.77(f)	1.175 1.355
0.211 0.211	(b) _ New	1.76 1.89	1.176 1.356
0.212 0.212	0.381 0.219	1.77 1.91 1.78 1.92	1.177 1.357
and 0.218 0.214 0.214	0.383 0.217 0.385 0.285	1.79 1.6	1.178 1.358 1.179 1.359
0.215 0.215	0.386 0.286	1.80 1.80	1.180 1.360
0.218	0.3 87 New	1.81 1.101	1.181 1.361
(a-b) 0.218	0.401 0.401 (a-c)	1.82 1.102	1.192 1.111
(c) New	0.403 0.401 (d-e)	1.83 1.104	1.193 1.120
0.231 0.361 0.241 0.331	0.405 0.415 0.407 0.408	1.84 1.106 1.85 1.115	1.202 1.401 1.203 1.403
0.243 0.332	0.411 0.402	1.86 1.117	1.204 1.405
0.247 0.333	0.413 0.405	1.87 1.113	1.205 1.407
0.251 0.322	0.415 0.404	1.101 1.201	1.2111.412
0.257 0.823	0.417 0.406 (a-f),	1.102 1.202	1.212 1.413
0.261 0.341 0.281 0.241	(k-l) 0.419 0.406(g)	1.103 1.219(a-b) 1.104 1.223	1.213 1.415 1.214 1.417
0.287 0.242	0.421 0.406(h)	1.105 1.225	1.215 1.419
0.291 0.251	0.423 0.406(j)	1.106 1.227	1.216 1.421
0.292 0.262	0.425 0.406(1)	1.111 1.251	1.217 1.423
0.293 0.252	0.427 0.407	1.112 1.253	1.218 1.425
0.294 0.253 0.295 0.254	0.429 0.416 0.431 0.417	1.113 1.46, 1.205 1.121 1.311	1.219 1.427 1.300 1.501
0.296 0.255	0.441 0.409	1.122 1.312	1.301 1.511
0.297 0.256	0.443 0.413 (a-b)	1.128 1.313	1.302 1.512
0.298 0.263	0.445 0.413(c)	1.124 1.314	1.303 1.513
0.302 0.257	0.447 0.414	1.125 1.315	1.304 1.514
0.303 0.258 0.304 0.259	0.449 0.410 0.451 0.411	1.126 1.316 1.127 1.317	1.305 1.516 1.306 1.564
0.805 0.260	0.458 0.412	1.128 1.318	1.307 1.566
-	4	1.129 1.319	1.308 1.518
PA	RT 1	1.131 1.331	1.309 1.519
No change in nun	aber of part. Follow-	1.1 32 1.333- 1.335	1.310 1.520 1.311 1.522
ing sections were re	edesignated.	1.132 (note) 1.333	1.312 1.568
Old New	Old New	and 1.338	1.313 1.599
section section	section section	1.1331.336(b)	1.314 1.598
1.10 1.1	1.23 1.23	and 1.339	1.315 1.601
1.111.41	1.24 1.24 1.25 1.25	1.134(a) 1.336 (b) 1.336(b)	1.316 1.525 1.321 1.531
1.12 1.44 1.13 1.45	1.26 1.26	and 1.337	1.322 1.533
1.14 1.8	1.41 1.291	1.135 1.340	1.323 1.534
1.15 1.3	1.42 1.292	1.140(a, c-e) 1.221	1.324 1.548
1.16 1.108	1.43 1.294	(b) 1.254	1.325 1.536
1.17 1.2	1.44 1.296	1.141 1.229 1.142 1.255	1.326 1.537 1.327 1.538
1.18 1.4 1.19 1.43	1.45 1.297 1.46 1.298	1.143 1.241	1.828 1.539
1.21 1.21	1.47 1.301	and 1.245	1.329 1.540
1.221.22	1.48 1.303	1.144 1.243	1.330 1.541

Old	New	Old eastion	New	New section	Old	New	Old
section 1.3 3 1	<i>section</i> 1 542	section 1.462	section	1.1	section	section	section
1.332		1.463		1.2		1.251	1.143(b) 1.111
1.333	1.544	1.468		1.3		1.253	
1.334	1.545	1.469	1.783	1.4			_ 1.140(b)
1.335		1.471		1.5		1.255	
1.336		1.472		1.6		1.258	
1.337	1.549	1.473 1.474		1.8		1.260	
1.341 1.342		1.475		1.21		1.261	
1.343		1.476		1.23		1.264	
1.351	1.569	1.477		1.24			.151(a, e-f
1.352	1.570	1.478		1.25			1.152(a)
1.353	1.561	1.479		1.26			1.152(b)
1.354		1.480 1.483		1.41 1.42		1.274	
1.355		1.484	1 802	1.43		1.276	(b-d) 1 153
1.357		1.485	1.803	1.44		1.277	
1.358		1.486		1.45		1.279	
1.359		1.487		1.46		1.282	
1.360		1.490		1.47		1.291	
1.361(a)_		1.491		1.48		1.292	
1.362(a)_	1.591	1.492 1.493		1.49 1.50		1.294 1.296	
1.302(a) _ (h_h	1.594	1.500		1.51		1.297	
(1-1)	1.592	1.501		1.52		1.298	
1.363	1.605	1.502	1.912	1.61		1.301	
1.364	1.603	1.503		1.62		1.308	
1.365		1.504	1.914	1.68		1.811	
1.370		1.505		1.77(a-c)		1.812	
1.401	1 703	1.506 1.507			1.74 New	1.313 1.814	
1.411		1.521		(f)		1.315	
1.412		1.522		(g)		1.316	
1.416	1.716	1.523			New	1.317	
1.417	1.717	1.524		1.80		1.318	
1.418	1.718	1.525 1.526		1.81 1.83		1.319	
1.421 1.422	1 722	1.527		1.84		1.831 1.833	
1.423		1.528		1.85		1.334	
1.424	1.724	1.529		1.87		1.335	
1.425	1.725	1.530		1.89		1.336	
1.426	1.726	1.531		1.91			1.134(b)
1.427 1.428		1.532 1.533		1.92 1.101		1.338	
1.429		1.534		1.102		1.339	; 1.134(b)
1.430		1.541	1.951	1.104		1.340	
1.431		1.542	1.952	1.106		1.351	
1.432		1.543		1.108		1.352	
1.433		1.544		1.110		1.353	
1.434		1.545 1.546		1.111		1.354 1.355	
1.440		1.547		1.115		1.356	
1.441		1.548		1.117		1.357	
1.442	1.743	1.549		1.120		1.358	
1.443		1.550		1.201		1.359	
1.444		1.551		1.202		1.360	
1.445 1.446		1.561 1.5 8 1		1.203 1.204		1.361 1.401	
1.447		1.600		1.205		1.403	
1.448	1.749	1.601	1.1103	1.207		1.405	
1.449	1.761	1.602		1.221	1.103;	1.407	1.205
1.450		1.620			.140(a, o-e)	1.411	
1.451		1.621 1.622		1.223			e 1.211
1.452		1.623		1.225		1.412	
1.454		1.624		1.22 7 1.22 9		1.413 1.415	
1.455	1.767	App		1.241		1.417	
1.461	1.771		-	1.243		1.419	

411

				<u>-</u>			
New section	Old section	New section	Old section	New section	Old	New	Old
1.421		1.716		1.923	section _ 1.523	section 1.958	section 1.545
1.423	_ 1.217	1.717	_ 1.417	1.924		1.959	
1.425		1.718		1.925		1.961	
1.427 1.501		1.721 1.722		1.926 1.927		1.9 62 1.9 71 _	
1.502		1.723		1.928		1.973	
1.511		1.724		1.929		1.981	
1.512 1.513		1.725 1.726		1.930 1.931		1.991	
1.514		1.727		1.932		1.1101	
1.516	_ 1.305	1.728	_ 1.428	1.933	_ 1.533	1.1105	1.602
1.518		1.729		1.984		1.1111	
1.519 1.520		1.730 1.731		1.951 1.952		1.1115	
1.522	_ 1.311	1.732	_ 1.432	1.953	_ 1.546	1.1117	1.623
1.525		1.733		1.955		1.1119	
1.531		1.734 1.735		1.956	_ 1.507	App	Арр.
1.534	_ 1.323	1.741			PA	RT 2	
1.536	_ 1.325	1.742		No cha	inge in nu	mber of p	art. Fol-
1.537 1.538	1.320	1.743 1.744		lowing se	ections wer	e redesign	ated.
1.539	_ 1.328	1.745		Old	New	Old	New
1.540		1.746		2.501			2.584(b)
1.541		1. 747 1. 7 48		2.510		(b)_	2.567(a),
1.543		1.749		2.511 2.512			2.584(a). 2.593(a)
1.544		1.761	_ 1.449	2.520			2.567(b)
1.545 1.546		1.762 1.763		2.521			2.584(c)
1.547		1.764		2.522 2.523			2.584(d) 2.567(c),
1.548	_ 1.324	1.765	_ 1.453	2.524		\-/-	2.584(e).
1.549		1.766		2.525			2.593(b)
1.561 1.562		1.767 1.771	_ 1.455 _ 1.461	2.530	_ 2.591	(g)_	2.567(d), 2.584(f)
1.564	_ 1.306	1.772	_ 1.462			2.541	
1.566		1.773				2.542	
1.568		1.781 1.783				2.543	2.555
1.570	_ 1.352	1.785	_ 1.471				
1.571		1.786		New	Old	New	Old
1.572 1.573		1.787 1.788		2.551		2.575	
1.574	_ 1.357	1.789		2.553 2.555		2.577 2.579	
1.578	_ 1.358	1.790		2.557		2.582	
1.580 1.587		1.791 1.792		2.561			2.540(b)
1.591		1.793		2.563 2.565		(b)_ (c)_	
4 500	(b-d)	1.794		2.567(a)		(d)_ _(d)_	
1.592	_ 1.361 (1-j)	1.801		(b)	(c)	(e)_	(f)
1.593	_ 1.362(a)	1.803		(c)		(f) 2.591	(g)
1.594		1.804	_ 1.486	(d) 2.571			2.540(b)
1.597	(b-h) 1.965	1.805 1.811		2.573		(b)_	
1.598		1 812			D _A	RT 3	
1.599		1 813	_ 1. 492				
1.601		1.814				signated as	
1.605		1.911		No otner	change in	section nu	mpers.
1.611	1.341	1 912	_ 1.502		Pa	RT 4	
1.615		1.913 1.914		Part 4	was rede	signated a	s Part 74.
1.621		1.914		No other	change in	section nu	mbers.
1.701	_ 1. 4 01	1.916	_ 1.551		Pa	RT 5	
1.703		1.918		No ch		umber of	part. No
1.711		1.921 1. 922			ange in in a section ni		Par. 140
							

Par	16	Part 9 Part 87	Part 9 Part 87
Part 6 was redesi	gnated as Part 23.	9.911 87.511 9.912 87.513	9.1208 87.165 9.1204 87.167
No other change in	section numbers.	9.913 87 .515	9.1205 87.169
Par	r 7	9.914 87.517 9.1001 87.251	9.1301 87.441 9.1302 87.443
Part 7 was redesi	gnated as Part 81.	9.1002 87.253	9.1401 87.141
No other change in s	ection numbers.	9.1003 87.255	9.1402 87.143
Par	r 8	9.1004 87.257 9.1005 87.259	9.1403 87.145 9.1404 87.147
Part 8 was redesi	gnated as Part 83.	9.1101 87.321	9.1405 87.149
No other change in	n section numbers,	9.1102 87.323 9.1103 87.325	9.1406 87.151 9.1407 87.153
Par	т 9	9.1104 87.327	9.1501 87.271
Part 9 was redesi	gnated as Part 87.	9.1105 87.329	9.1502 87.273
The following secti	ons were redesig-	9.1201 87.161 9.1202 87.163	9.1503 87.275 9.1504 87.277
nated:		Par	r 10
Part 9 Part 87 9.1 87.1	Part 9 Part 87 9.313 87.185		ignated as Part 89.
9.2 87.3	9.314 87.187	The following sect	
9.3 87.5 9.4 8 7 .51	9.315 87.183(d) 9.321 87.195	nated:	•
9.5 87.53	9.331 87.201	Old	New
9.101 87.21	9.411 87.401	Subpart A—Gen. Info.	
9.102 87.23 9.103 87.25	9.412 87.403 9.413 87.405	10.2	
9.104 87.27	9.414 87.407	10.3	
9.105 87.29 9.106 87.31	9.415 87.409 9.416 87.411	10.4	
9.107 87.125	9.431 87.291	10.6	89.11
9.108 87.41	9.432 87.293 9.433 87.295	10.7	
9.109 87.39 9.110 87.43	9.434 87.297	10.9	
9.111 87.45	9.435 87.299	Subpart B-Appls.,	"Subpart B" deleted
9.112 87.47 9.113 87.49	9.436 87.301 9.437 87.303	Authorizations and Notifications.	
9.117 87.37	9.438 87.305	10.51	
9.118 87.95 9.119 87.97	9.439 87.307 9.440 87.309	10.52	
9.120 87.127	9.441 87.451	10.54	89.57
9.121 87.129 9.141 87.91	9.442 87.453 9.443 87.455	10.55	
9.142 87.93	9.446 87.461	10.58	
9.151 87.99	9.447 87.463	10.60	
9.152 87.10 9.153 87.103	9.451 87.431 9.452 87.433	10.61	
9.154 87.105	9.453 87.435	10.63	89.71
9.155 87.107 9.156 87.109	9.454 87.437 9.511 87.501	10.64	
9.178 87.61	9.512 87.503	10.66	89.77
9.179 87.63 9.180 87.65	9.513 87.505 9.611 87.33 1	10.68	
9.181 87.67	9.612 87.333	10.70	
9.182 87.69 9.183 87.71	9.613 87.335 9.614 87.337	Subpart C—Tech. Standards.	"Subpart C" deleted
9.184 87.73	9.711 87.341	10.101	89.101
9.185 87. 75 9. 186 87. 111	9.712 87.343 9.713 87.345	10.102(a) (b-c)	
9.187 87.77	9.714 87.347	(d)	
9.188 87.79	9.715 87.349	10.103	
9.189 87.113 9.1 9 0 87.81	9 7 16 87.351 . 9.717 87.353	10.104 (b) (3-4, 6)	
9 .191 8 7 .115	9.718 87.355	(b) (5)	(b) (3)
9.192 87.117 9.193 87.123	9.719 87.357 9.811 87.235	10.105 (d-g)	
9.194 87.119	9.812 87.237	10.106	89.111
9.1 9 5 87.121 9.311 87.181	9.813 87.239 9.814 87.241	10.107	
9.812 87.188	9.815 87.243	10.109	

Old	New	Old	New
10.110		10.454	89.505
10.111	. 8 9 .1 21	10.455	89.513
Subpart D—Opr.	"Subpart D" deleted	10.456	89.511
Requirements.		10.457	
10.151	89.151	10.458	
10.152		10.459	
10.153		10.460	
10.154			
		10.461	
10.155		10.462	89.525
10.156		Subpart K—State	Subpart R—State
10.157	89.167	Subpart K—State Guard Radio Service. 10.501	Guard Radio
10.158	8 9.16 9	Service.	Service
10.159	. 8 9.17 1	10.501	8 9.551
10.160		10.502	89.553
10.161		10.503	
10.162		10.504	
10.163		10.505	
		Gubnort T Tool	George W. W. Const.
10.164		Subpart L—Local	Suppart E-Local
10.165	89.161	Govt. Radio Service.	Govt. Radio
Subpart E—Dev'l.	Subpart C—Dev'l.		Service
Subpart E—Dev'l. Opr.	Opr.	10.551	89.251
10.201	89.201	10.552	89.253
10.202	89.203	10.553	89.255
10.203		10.554	
10.204		10.555	
10.205		(g) (5)	
10.206		(g) (6)	(a) (5)
		(8) (7)	(g) (b)
10.207		(g) (7)	(g)(b)
10.208	89.215	Part	11
Subpart F—Police	Subpart G—Police		
Radio Service.	Radio Service	Part 11 was redesi	ignated as Part 91
Subpart F—Police Radio Service. 10.251	89 .301	No other change in s	section numbers
10.252	89.303	210 Outice Change in S	section numbers.
10.253		Part	12
10.254			
10.255		Part 12 was redest	gnated as Part 97.
(n) (19-2x)	Deleted	The following secti	ons were redesig-
(h) (19-23)	Deleted		ons were redesig-
(h) (19-28)	Deleted Subpart J—Fire	nated:	ons were redesig-
Subpart G—Fire Radio Service.	Subpart J—Fire Radio Service .		ons were redesig-
Subpart G—Fire Radio Service. 10.301	Subpart J—Fire Radio Service. 89.351	nated:	ons were redesig-
Subpart G—Fire Radio Service. 10.301 10.302	Subpart J—Fire Radio Service. 89.351 89.353	nated: Old New 12.097.1	ons were redesig- Old New 12.4697.25
Subpart G—Fire Radio Service. 10.301	Subpart J—Fire Radio Service. 89.351 89.353 89.355	nated: Old New 12.097.1 12.197.3(a)	Old New 12.4697.25 12.4797.29(d)
Subpart G—Fire Radio Service. 10.301 10.302	Subpart J—Fire Radio Service. 89.351 89.353 89.355	nated: Old New 12.097.1 12.197.3(a) 12.2(b)	Old New 12.4697.25 12.4797.29(d) 12.4897.31
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359	nated: Old New 12.097.1 12.197.3(a) 12.2(b) 12.3(c)	Old New 12.4697.25 12.4797.29(d) 12.4897.31
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359	nated: Old New 12.097.1 12.197.3(a) 12.2(b) 12.3(c) 12.4(d)	Old New 12.46
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359	nated: Old New 12.097.1 12.197.3(a) 12.2(b) 12.3(c) 12.4(d) 12.5(e)	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.31 12.50 97.39 12.60 97.45
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29(d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359	nated: Old New 12.0	Old New 12.46
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29(d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.352	Subpart J—Fire Radio Service. 89.351 89.353 89.357 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453	nated: Old New 12.0	Old New 12.46
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37 12.62 97.41 12.64 97.43
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.354 10.354	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29(d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37 12.62 97.39 12.63 97.43 12.65 97.43 12.65 97.43
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459	nated: Old New 12.0	Old New 12.46
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22)	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29(d) 12.48 97.31 12.49 97.39 12.50 97.29(c) 12.60 97.45 12.61 97.37 12.62 97.41 12.64 97.43 12.65 97.81 12.65 97.81 12.65 97.81
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22)	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted	nated: Old New 12.0	ONS were redesig- Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37 12.62 97.39 12.64 97.43 12.65 97.43 12.65 97.43 12.66 97.43
Subpart G—Fire Radio Service. 10.301 10.302 10.308 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22)	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29(d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37 12.62 97.39 12.63 97.41 12.64 97.43 12.65 97.43 12.65 97.43 12.65 97.43 12.65 97.43 12.65 97.43 12.65 97.43
Subpart G—Fire Radio Service. 10.301 10.302 10.308 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22)	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted	nated: Old New 12.0	Old New 12.46
Subpart G—Fire Radio Service. 10.301 10.302 10.308 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22)	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.45 12.62 97.41 12.64 97.43 12.65 97.49 12.66 97.47 12.68 97.47 12.68 97.47 12.69 97.49 12.70 97.49 12.70 97.49
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Laintenance Radio Serv. 10.401	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37 12.62 97.45 12.65 97.43 12.65 97.43 12.65 97.43 12.65 97.87 12.68 97.81 12.68 97.85 12.69 97.85 12.69 97.85 12.81 97.87
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Maintenance Radio Serv. 10.401 10.402	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart I.—Highway Maintenance Radio Serv. 89.401 89.403	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29(d) 12.48 97.31 12.49 97.33 12.50 97.37 12.62 97.37 12.62 97.39 12.63 97.41 12.64 97.43 12.65 97.59(b) 12.66 97.81 12.67 97.85 12.69 97.139 12.70 97.49 12.81 97.53
Subpart G—Fire Radio Service. 10.301	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.403 89.405	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.39 12.50 97.29 (c) 12.60 97.45 12.61 97.37 12.62 97.41 12.64 97.43 12.65 97.81 12.66 97.81 12.67 97.47 12.68 97.85 12.69 97.49 12.81 97.51 12.82 97.53 12.85 97.55
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Maintenance Radio Serv. 10.401 10.402 10.403	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.457 B9.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29(d) 12.48 97.31 12.49 97.33 12.50 97.37 12.62 97.37 12.62 97.39 12.63 97.41 12.64 97.43 12.65 97.59(b) 12.66 97.81 12.67 97.85 12.69 97.139 12.70 97.49 12.81 97.53
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.352 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Maintenance Radio Serv. 10.401 10.402 10.403	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.45 12.62 97.41 12.64 97.43 12.65 97.59 (b) 12.66 97.47 12.68 97.87 12.69 97.47 12.68 97.89 12.70 97.49 12.70 97.47 12.82 97.87 12.82 97.87 12.85 97.51 12.82 97.87 12.86 97.55 12.86 97.55 12.86 97.55 12.86 97.55 12.86 97.55
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Maintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15)	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409 Deleted	nated: Old New 12.0	ONS were redesig- Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37 12.62 97.45 12.65 97.43 12.65 97.43 12.65 97.43 12.65 97.85 12.69 97.139 12.68 97.85 12.69 97.85 12.81 97.85 12.82 97.87 12.85 97.87 12.85 97.87 12.85 97.55 12.90 97.95 12.90 97.97
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Maintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15)	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409 Deleted	nated: Old New 12.0	ONS were redesig- Old New 12.46 97.25 12.47 97.29(d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37 12.62 97.39 12.63 97.41 12.64 97.43 12.65 97.59(b) 12.66 97.81 12.67 97.85 12.69 97.139 12.70 97.49 12.81 97.51 12.82 97.87 12.85 97.53 12.86 97.55 12.90 97.97 12.93 97.99
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Jaintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15) (f) (16)	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409 Deleted (f) (10)	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.39 12.50 97.45 12.61 97.45 12.62 97.41 12.64 97.43 12.65 97.59 (b) 12.66 97.81 12.67 97.47 12.68 97.81 12.70 97.49 12.81 97.51 12.82 97.87 12.82 97.87 12.85 97.55 12.90 97.95 12.91 97.95 12.91 97.99 12.93 97.99 12.94 97.101
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Laintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15) (f) (16) Subpart J—Special	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.457 89.457 89.459 Deleted Subpart I.—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409 Deleted (f) (10) Subpart P—Special	nated: Old New 12.0	ONS were redesig- Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.45 12.62 97.45 12.62 97.41 12.64 97.43 12.65 97.59 (b) 12.66 97.81 12.67 97.47 12.68 97.81 12.69 97.139 12.70 97.47 12.82 97.87 12.82 97.87 12.83 97.51 12.81 97.51 12.82 97.55 12.90 97.95 12.91 97.95 12.91 97.97 12.93 97.99 12.94 97.101 12.101 97.89
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.352 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Maintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15) (f) (16] Subpart J—Special Emer. Radio Serv.	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409 Deleted (f) (10) Subpart P—Special Emer. Radio Serv.	nated: Old New 12.0	ONS were redesig- Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.33 12.50 97.45 12.61 97.37 12.62 97.39 12.65 97.43 12.65 97.43 12.65 97.85 12.66 97.81 12.68 97.85 12.69 97.139 12.70 97.49 12.81 97.51 12.82 97.87 12.85 97.53 12.86 97.55 12.90 97.97 12.93 97.97 12.93 97.99 12.94 97.101 12.101 97.89 12.102 97.111
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Maintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15) (f) (16) Subpart J—Special Emer. Radio Serv. 10.450	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.407 89.409 Deleted (f) (10) Subpart P—Special Emer. Radio Serv. 89.501	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.39 12.50 97.29 (c) 12.60 97.45 12.61 97.37 12.62 97.41 12.64 97.43 12.65 97.41 12.66 97.81 12.67 97.47 12.68 97.85 12.69 97.139 12.70 97.49 12.81 97.51 12.82 97.53 12.86 97.55 12.90 97.95 12.91 97.97 12.93 97.99 12.94 97.101 12.101 97.89 12.101 97.89 12.101 97.89
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart H—Highway Maintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15) (f) (16) Subpart J—Special Emer. Radio Serv. 10.450 10.450 10.450	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.457 89.457 89.459 Deleted Subpart I.—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409 Deleted (f) (10) Subpart P—Special Emer. Radio Serv. 89.501 89.503	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.39 12.50 97.29 (c) 12.60 97.45 12.61 97.37 12.62 97.41 12.64 97.43 12.65 97.59 (b) 12.66 97.47 12.68 97.81 12.69 97.139 12.81 97.51 12.82 97.89 12.85 97.55 12.90 97.55 12.90 97.55 12.90 97.95 12.91 97.97 12.93 97.99 12.94 97.101 12.101 97.89 12.102 97.113 12.104 97.115
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart I—Highway Laintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15) (f) (16) Subpart J—Special Emer. Radio Serv. 10.450 10.450 10.450 10.451 10.452	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409 Deleted (f) (10) Subpart P—Special Emer. Radio Serv. 89.501 89.503 89.509	nated: Old New 12.0	ONS were redesig- Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.35 12.50 97.45 12.61 97.37 12.62 97.45 12.65 97.41 12.64 97.43 12.65 97.41 12.66 97.45 12.66 97.81 12.67 97.85 12.69 97.139 12.70 97.85 12.81 97.51 12.82 97.85 12.82 97.85 12.83 97.51 12.84 97.51 12.85 97.53 12.86 97.55 12.90 97.95 12.91 97.97 12.93 97.95 12.91 97.97 12.93 97.99 12.101 97.89 12.102 97.111 12.103 97.135 97.93
Subpart G—Fire Radio Service. 10.301 10.302 10.303 10.304 10.305 Subpart H—Forestry- Conservation Radio Serv. 10.351 10.352 10.353 10.354 10.355 (e) (16-22) Subpart H—Highway Maintenance Radio Serv. 10.401 10.402 10.403 10.404 10.405 (f) (10-15) (f) (16) Subpart J—Special Emer. Radio Serv. 10.450 10.450 10.450	Subpart J—Fire Radio Service. 89.351 89.353 89.355 89.357 89.357 89.359 Subpart N—Forest- ry-Conservation Radio Serv. 89.451 89.453 89.455 89.457 89.459 Deleted Subpart L—Highway Maintenance Radio Serv. 89.401 89.403 89.405 89.407 89.409 Deleted (f) (10) Subpart P—Special Emer. Radio Serv. 89.501 89.503 89.509	nated: Old New 12.0	Old New 12.46 97.25 12.47 97.29 (d) 12.48 97.31 12.49 97.39 12.50 97.29 (c) 12.60 97.45 12.61 97.37 12.62 97.41 12.64 97.43 12.65 97.59 (b) 12.66 97.47 12.68 97.81 12.69 97.139 12.81 97.51 12.82 97.89 12.85 97.55 12.90 97.55 12.90 97.55 12.90 97.95 12.91 97.97 12.93 97.99 12.94 97.101 12.101 97.89 12.102 97.113 12.104 97.115

Old	New	Old New
12.106	97.91	12.214 97.173
	97.69	12.215 97.175
	97.61	12.221 97.177
	97.63	12.222 97.179
	97.65	12.223 97.181
12.131	97.67	12.224 97.183
	97.71	12.225 97.185
	97.73	12.226 97.187
12.134	97.93	12.227 97.189
12.135	97.75	12.228 97.191
	97.103	12.2 31 97.193
12.137	97.105	12.2 32 97.195
	97.77	12.233 97.197
	97.131	12.234 97.199
	97.133	12.235 97.201
	97.135	12.241 97.203
	97.137	12.2 42 9 7.205
	97.107	12.243 97.207
	97.119	12.244 97.209
	97.121	12.245 97.211
	97.123	12.246 97.213
	97.125	12.251 97.215
	97.127	12.252 97.217
	97 .1 2 9	12.253 97.219
	97.161	12.254 97.221
	97.163	12.255 97.223
	97.165	12.256 97.225
	97.167	12.257 97.227
	97.169	Appen- Appen-
12.213	97.171	dices 1-4 dices 1-4

PART 13

No change in number of part. No change in section numbers.

PART 14

Part 14 was redesignated as Part 85. No other change in section numbers.

PART 15

No change in number of part. Section 15.166 was deleted and the following sections were redesignated.

Subpart C-Radio Receivers

Old	New	Old	New
15.61		15.67	15.81
15.62		15.68	15.72
15.63		15.69	15.82
15.64		15.70	15.65
15.65		15.71	15.66
15.66			

Subpart E—Low Power Comm. Devices

Old	New	Old	New
15.201		15.210	15.222
15.202		15.211	15.229
15.203		15.212	15.212
15.204		15.213	15.213
15.205		15.214	15.235
15.206		15.215	15.236
15.207		15.216	15.237
15.208		15.217	15.238
15.209			

Subpart C-Radio Receivers New Old New Old 15.61_____ 15.61 15.71_____ 15.66

15.63_____ 15.62 15.72 _____ 15.68 15.65_____ 15.70 15.75_____ 15.63 15.66_____ 15.71 15.81_____ 15.67 15.69_____ 15.64 15.82_____ 15.69 15.70_____ 15.65

Subpart E-Low Power Comm. Devices

New	Old	New	Old
15.201	15.201	15.222	15.210
15.202	15.202	15.227	15.208
15.203	15.203	15.228	15,209
15.204	15.204	15.229	
15.205	15.205	15.235	
15.211	15.206	15.236	
15.212	15.212	15.237	
15.213	15.213	15.238	
15.221	15.207		

PART 16

Part 16 was redesignated as Part 93. No other change in section numbers.

PART 17

No change in number of part. No change in section numbers.

PART 18

No change in number of part. Following sections were redesignated.

-		
Old New	Old	New
18.1 18 .1	18.44	_ 18.24
18.2 18.3	18.45	18.25
18.3 18.21(a)	18.46	18.26
18.4 18.11	18.47	18.27
18.5(a) 18.181	18.48	
(b) 18.182(a)		
(c) 18.182(b)	18.51	
(d) 18.182(c)	18.70	
(e) 18.183	18.71	
18.6 18.13	18.72	
18.7 18.14	18.73	
18.8 18.17	18.74	
18.11 18.141	18.75	
18.12 18.142	18.76	
18.13 18.143	18.77	
18.14 18.144	18.78	
18.15 18.145	18.80	
18.16 18.146	18.81	
18.31 18.261	18.82	
18.41(a) 18.21(b)	18.84	
(b) (c)	18.101-	
18.42 18.22	18.122	18.101-
18. 4 3 18. 23		18.122
N 014		

New		New	Old
18.1	18.1	18.22	18.42
18.3	18.2	18.23	18.43
18.11	18.4	18.24	18.44
18.13	18.6	18.25	18.45
18.14	18.7	18.26	
18.17	18.8	18.27	
18.21(a)_	18.3	18.28	
(b)_	18.41(a)	18.29	
(c)_	(b)	18.71	

New	Old	New	Old
18.72	. 18.71	18.141	18.11
18.73	. 18.72	18.142	18.12
18.74	. 18.73	18.143	18.13
18.75	18.74	18.144	18.14
18.76	18.75	18.145	18.15
18.77	18.76	18.146	18.16
18.78	18.77	18.181	18.5(a)
18.79	18.78	18.182(a)	18.5(b)
18.80	18.80	(b)	(c)
18.81	18.81	(c)	(d)
18.82	18.82	18.183	18.5(e)
18.84	18.84	18.261	18.31
18.101-		18.262	18.51
18.122	18.101-		
	18.122		

PART 19

Part 19 was redesignated as Part 95. The following sections were redesignated:

Old	New	Old	New
19.1		19.41	95.53
19.2	9 5.3	19.42	95.55
19.3	95.5	19.43	95.57
19.4	95.7	19.44	95.59
19.11	95.11	19.45	95.61
19.12	95.13	19.51	95.63
19.13	95.15	19.52	95.65
19.14	95.17	19.53	95.67
19.15	95.19	19.54	95.69
19.16	95.25	19.61	95.81
19.17	95.27	19.62	95.87
19.18	95.21	19.63	95.91
19.19	95.23	19.64	95.93
19.21	95.29	19.71	95.95
19.22	95.31	19.72	95.101
19.23	95.33	19.73	95.103
19.24	95.35	19.74	95.105
19.25	95.37	19.81	95.111
19.31	95.41	19.82	95.113
19.32	95.43	19.83	95.115
19.33	95.45	19.91	95.117
19.34		19.92	95.119
19.35		19.93	95.121
19.36			

PART 20

Part 20 was redesignated as Part 99. The following sections were redesignated:

Old	New	Old	New	
20.1	9 9.1	20.15	 9 9.15	
20.2	99.3(b)	20.21	99.17	
20.3	(a)	20.22	99.19	
	and (c)	20.23	99.21	
20.4	(d)	20.24	99.23	
20.5	(e)	20.25	99.25	
20.6	(f)	20.26	9 9.27	
20.7	(g)	20.29	99.29	
20.8	(h)	20.30	99.31	
20.9	(i)	20.31	99.33	
20.11	99.7	20.32	99.35	
20.12	99.9	20.33	99.37	
20.13	9 9.11	20.34	9 9.39	
20.14	99.13	20.35	99.41	

PART 21

No change in number of part. No change in section numbers.

PART 23

Part 23 was redesignated from Part 6. No other change in section numbers.

No change in part or section numbers in the following:

PART	25	PART	51
PART	31	PART	52
PART	33	PART	61
PART	34	PART	62
PART	35	PART	63
PART	41	PART	64
PART	42	PART	66
PART	43		

PART 73

Part 73 was redesignated from Part 3. No other change in section numbers.

PART 74

Part 74 was redesignated from Part 4. No other change in section numbers.

PART 81

Part 81 was redesignated from Part 7. No other change in section numbers.

PART 83

Part 83 was redesignated from Part 3. No other change in section numbers.

PART 85

Part 85 was redesignated from Part 14. No other change in section numbers.

PART 87

Part 87 was redesignated from Part 9. The following sections were redesignated:

D 05

D ---- 0

Part 87	Part 9	Part 87	Part 9
87.1	9.1	87.61	9.178
87.3	9.2	87.63	9.179
87.5	9.3	87.65	9.180
87.21	9.101	87.67	9.181
87.23	 9 .102	87.6 9	9.182
87.25	9.103	87.71	9.183
87.27	9.104	87.73	9.184
87.29	9.105	87.75	9.185
87.31	9.106	87.77	9.187
87.37	9.117	87.79	9.188
87.39	9.109	87.81	9.190
87.41	9.108	87.91	9.141
87.43	9.110	87.93	9.142
87.45	9.111	87.95	9.118
87.47	9.112	87.97	9.119
87.49	9.113	87.99	9.151
87.51	9.4	87.101	9.152
87.53	9.5	87.103	9.153

D ---- 07

Part 87 Part 9	Part 87 Par		OIA
87.105 9.154	87.295 9.43	3 Appls., Authorizations	Subpart B—same
87.107 9.155	87.297 9.43	4 and Notifications.	title
87.109 9.156	87.2999.48	89.51	10.51
87.111 9.186	87.301 9.43		
87.1139.189	87.303 9.43		
87.115 9.191	87.305 9.48		
87.1179.192	87.307 9.43		
87.119 9.194	87.309 9.44		
87.121 9.195	87.321 9 .11		
87.123 9.193	87.323 9.11		
87.125 9.107	87.325 9 .11		10.61
87.127 9.120	87.327 9.11	.04 89.69	10.62
87.129 9.121	87.329 9.11	.05 89.71	10.63
87.141 9.1401	87.331 9.61	.1 89.73	10.64
87.143 9.1402	87.333 9.61		
87.145 9.1403	87.335 9.61		
87.147 9.1404	87.337 9.61		
87.149 9.1405	87.8419.71		
87.151 9.1406	87.343 9.71	.2 89.83	
87.153 9.1407	87.345 9.71		Subpart C—same
87.161 9.1201	87.347 9.71	.4	title
87.163 9.1202	87.349 9.71		
87.165 9.1203	87.351 9.71	.6 89.163(a)	10.102(a)
87.167 9.1204	87.353 9.71	.7 (b)	(d)
87.169 9.1205	87.355 9.71		
87.181 9.311	87.357 9.71		10.104
87.183 9.312	87.401 9.41		
and 9.315	87.4039.41		
87.185 9.313	87.405 9.41		
	87.407 9.41	4 00.119	10.100
87.187 9.314			
87.195 9.321	87.4099.41		
87.201 9.331	87.4119.41		
87.211 9.611	87.431 9.45		
and 9.711	87.4339.45		
87.235 9.811	87.435 9.45	3 Operating 4 Requirements	Subpart D—same
87.237 9.812	87.437 9.45	4 Requirements	title
87.239 9.813	87.441 9.18		
87.241 9.814	87.443 9.13	02 89.153	10.152
87.243 9.815	87.451 9.44		
87.251 9.1001	87.453 9.44		
87.253 9.1002	87.455 9.44		
87.255 9.1003	87.4619.44		
87.257 9.1004	87.463 9.44	7 89.163	10.100
	87.5019.51		10.150
87.259 9.1005			10.156
87.271 9.1501	87.503 9.51		10.157
87.273 9.1502	87.505 9.51		10.158
87.275 9.1503	87.511 9.91		
87.277 9.1 504	87.5139.91		
87.291 9.431	87.515 9.91		
87.293 9.432	87.517 9.91	4 89.177	10.162
_		89 179	10.163
Part	89	Subpart C-Dev'l.	Subpart E-Dev'l.
Dowt 90 was rada	impoted from De	Subpart C—Dev1. rt Opr. 89.201	Opr.
Part 89 was rede	· · · · · · · · · · · · · · · · · · ·	89.201	10 201
10. The following	sections were r	e- 89.203	10.202
designated:		89.205	
Man	Old	89.207	
New			
Subpart A—Gen. Info.		89.209	
89.1		89.211	10.206
89.3	10.2	89.213	10.207
89.5	10.3	89.215	10.208
89.7		Subpart E-Local	Subpart I-I-Local
89.9		Govt. Radio Serv.	
			Govt. Radio Serv.
89.11	10.0	89.251	TO:00T
90 19		00.050	40.550
89.13	10.7	89.253	
89.15 89.17	10.7 10.8	89.253 89.255 89.257	10.553

New Old	Par	т 95
89.259 10.555 (g) (5) (g) (6)	Part 95 was rede	esignated from Part
(g) (5) (g) (6) (g) (6) (g) (7)		sections were re-
	designated:	300000125 #1020 10-
Subpart G—Police Subpart F—Police Radio Service Radio Service	New Old	New Old
89.801 10.251	95.119.1	New Old 95.5319.41
89.303 10.252	95.319.2	95.5519.42
89.305 10.253 89.307 10.254	95.5 19.3	95.5719.43
89.309 10.255	95.7 19.4	95.5919.44
Subpart J—Fire Subpart G—Fire	95.11 19.11	95.61 19.45
Subpart J—Fire Subpart G—Fire Radio Serv. Radio Serv.	95.1819.12	95.6319.51
89.351 10.301	95.15 19.13 95.17 19.14	95.65 19.52 95.67 19.53
89.353 10.302	95.1919.15	95.69 19.54
89.355 10.303 89.357 10.304	95.21 19.18	95.81 19.61
89.359 10.305	95.23 19.19	95.87 19.62
Subpart L-Highway Subpart I-Highway	95.25 19.16	95.91 19.63
Maintenance Maintenance	95.27 19.17 95.29 19.21	95.9319.64
89.359 10.305 Subpart I.—Highway Maintenance Radio Serv. 10.305 Subpart I.—Highway Maintenance Radio Serv.	95.3119.22	95.95 19.71 95.101 19.72
89.401 10.401	95.3319.23	95.10319.73
89.403 10.402 89.4 05 10.403	95.35 19.24	95.10519.74
89.407 10.404	95.37 19.25	95.111 19.81
89.409 10.405	95.41 19.31	95.113 19.82
(f) (10) (f) (18)	95.4319.32	95.115 19.83
Subpart N—Forestry- Conservation Radio Serv. Subpart H—Forest- ry-Conservation Radio Serv.	95.45 19.33 95.47 19.34	95.117 19.91 95.119 19.92
Conservation ry-Conservation	95.4919.85	95.121 19.93
Radio Serv. Radio Serv.	95.5119.36	00.121 10.50
89.451 10.351 89.453 10.352		
89.455 10.353	Par	т 97
89.457 10.354	Part 97 was rede	signated from Part
		STRUCK HOME LET!
89.459 10.355	12. The following	sections were re-
89.459 10.355 Subpart P—Spec. Subpart J—Spec.	12. The following	sections were re-
89.459	12. The following designated:	
89.501 10.450	12. The following designated: New Old	New Old
89.459	12. The following designated:	New Old 97.39 12.62 97.41 12.63
89.501	12. The following designated: New Old 97.112.0 97.3(a) 12.1 (b) 12.2	New Old 97.39 12.62 97.41 12.63 97.43 12.64
89.503	12. The following designated: New Old 97.1	New Old 97.39 12.62 97.41 12.63 97.43 12.64 97.45 12.60
89.501	12. The following designated: New Old 97.1	New Old 97.39 12.62 97.41 12.63 97.43 12.64 97.45 12.60 97.47 12.67
89.501	12. The following designated: New Old 97.1	New Old 97.39 12.62 97.41 12.63 97.45 12.64 97.45 12.60 97.47 12.67 97.49 12.70
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501 10.450 89.503 10.451 89.505 10.454 89.507 10.453 89.509 10.452 89.511 10.456 89.513 10.455 89.516 10.457 89.517 10.457 89.519 10.459 89.521 10.460	12. The following designated: New Old 97.1	New Old 97.39 12.62 97.41 12.63 97.45 12.64 97.45 12.60 97.47 12.67 97.49 12.70 97.51 12.81 97.53 12.85 97.55 12.86 97.57 12.86
89.501 10.450 89.503 10.451 89.505 10.454 89.507 10.453 89.509 10.452 89.511 10.456 89.513 10.455 89.517 10.457 89.519 10.459 89.521 10.460 89.523 10.461	12. The following designated: New Old 97.1	New Old 97.39
89.501 10.450 89.503 10.451 89.505 10.454 89.507 10.453 89.509 10.452 89.511 10.456 89.515 10.455 89.516 10.457 89.519 10.459 89.521 10.460 89.525 10.461 89.525 10.462	12. The following designated: New Old 97.1	New Old 97.39
89.501 10.450 89.503 10.451 89.505 10.454 89.507 10.453 89.509 10.452 89.511 10.456 89.515 10.455 89.516 10.457 89.519 10.459 89.521 10.460 89.525 10.461 89.525 10.462	12. The following designated: New Old 97.1	New Old 97.39
89.501 10.450 89.503 10.451 89.505 10.454 89.507 10.453 89.509 10.452 89.511 10.456 89.513 10.455 89.517 10.457 89.519 10.459 89.521 10.460 89.523 10.461	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39 12.62 97.41 12.63 97.45 12.60 97.47 12.67 97.49 12.70 97.51 12.81 97.53 12.85 97.55 12.86 97.59(a) and (c) 12.29 (b) 12.65 97.61 12.11 97.63 12.113 97.65 12.114 97.67 12.131 97.69 12.107 97.71 12.132 97.73 12.133 97.75 12.135 97.77 12.151 97.79 12.28
89.501	12. The following designated: New Old 97.1	New Old 97.39 12.62 97.41 12.63 97.41 12.64 97.45 12.60 97.47 12.67 97.53 12.81 97.53 12.85 97.55 12.86 97.59(a) and (c) 12.29 (b) 12.65 97.61 12.11 97.63 12.113 97.65 12.114 97.67 12.131 97.69 12.107 97.71 12.132 97.73 12.133 97.75 12.135 97.77 12.151 97.79 12.185 97.71 12.185 97.77 12.151 97.79 12.185 97.79 12.28 97.81 12.66
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39
89.501	12. The following designated: New Old 97.1	New Old 97.39

New	Old	New	Old	New	Old	New	014
97.93	12.104	97.141		97.211		97.221	
	(second	97.143		97.213		97.223	
	sent.)	97.161	12.200	97.215		97.225	
97.95		97.163		97.217		97.227	
97.97	12.91	97.165		97.219			
97.99	12.93	97.167	12.211				
97.101	12.94	97.169	12.212		PART	99	
97.103		97.171			- na-		
97.105	12.137	97.173	12.214	Part 99	was redes	ignated fro	m Part
97.107	12.156	97.175	12.215		following		
97.111	12.102	97.177	12.221		_	SECTIOITS M	CIG 16-
97.113	12.103	97.179	12.222	designated	1:		
97.115	12.104	97.181	12.223	New	Old	New	Old
	(first	97.183	12.224		20.1	99.15	
	sent.)	97.185	12.225	99.3(a)		99.17	20.21
97.117	. 12.105	97.187	12.226	and (c)_	20.3	99.19	20. 22
97.119	. 12.157	97.189	12.227		20.2	99.21	20.23
97.121	12.158	97.191	12.228		20.4	99.23	20.24
97.123	. 12.159	97.193	12.231		20.5	99.25	20.25
97.125	. 12.160	97.195	12.232		20.6	99.27	20.26
97.127	. 12.161	97.197	12.233	(g)	20.7	99.29	20.29
97.129	. 12.162	97.199	12.234		20.8	99.31	20.30
97.131	12.152	97.201	12.235		20.9	99.33	20.31
97.133	. 12.153	97.203	12.241	99.7		99.35	
97.135	. 12.154	97.205	12.242	99.9	20.12	99.37	20.88
97.137	12.155	97.207	12.243	99.11	20.13	99.39	20.84
97.139	12.69	97.209	12.244	99.13	20.14	99.41	

List of Sections Affected

All changes in this volume of the Code of Federal Regulations which were made by documents published in the FEDERAL REGISTER since January 1, 1964, are enumerated in the following list. Entries indicate the nature of the changes effected. Page numbers refer to Federal Register pages. The user should consult the entries for chapters and parts as well as sections for revisions.

For the period before January 1, 1964, see "List of Sections Affected, 1949-

1963," which is published in a separate volume.

1964	47 CFR—Continued 29 F.R. Page
47 CFR 29 F.I	^{16.} 0.202
4/ CFK Pag 0.11	Revised 4913
(e) added 320	2 0.304
(d) revised1466	
0.94	0.311
(c) and (d) revised 1126	8 (a) (4) revised 12773
0.95	Heading and (b) revised 15923
Revised 956	5 0.313
0.96	(a) revised 15923
Revised 956	5 0.315
0.111 Danies d	Revised 14665
Revised 1466 (g) added 1575	
(g) added 1575 0.113	0.331
Revised 1575	
0.121	(b) (13) revised 13815
(c) and (d) amended; (e)	(a) revised 14405
added 817	
0.181	(b) (17) added 18373
Revised 1466	3 4 0.333
0.182	Revised 14406
Revised 1466	⁶⁴ 0.341
0.183	Revised 6443
Revised 1575	0.351
Deleted 1466	5 Revised 6443
0.184	0.361
Added 1466	(b) revised 6443
0.185	0.365
Added 1466	55 Revised 6443
0.186	0.385
(b) (8) revised 1592	Revised 14665
0.231	0.386
Revised 1466	Revised 15923
0.241 (c) revised 1557	0.387
0.243	(b) revised 15923
(c) added 1058	35 0.401
0.281	(a) revised12370
(d) (14) added14	· · · · ·
(d) (15) added 34	
(x) added 1116	Revised 12371

47	CFR-	—Continued	29 F.R. Page	47 CF	RContinued	29 F.R. Page
0.44		sed	12371	1.550 Adde	d	15576
(Ì) revis	sed; (c) added	15760	1.564	evised	
0.44 H		and (a) revised	_ 10397	1.568		
		endedended		(c) 1 1.569	evised	6444
1) ame	snucu	_ 13203	(b) (1	.) and (c)(1) revis	ed 11360
P		text of appendix designas Appendix A, Appendix A		1.571 Amer	nded	9499
	dix B	added	_ 4801	1.594	dded	
		F heading revised x B amended		1.729	iadea	0444
A	nnual i	report form L amended		(a) 1 1.742	revised	8219
1.10 A			_ 14406	Revis	ed	12371
1.12				1.744 (b) a	nd (c) revised	6444
1.21			_ 14400	1.745		
		heading revised	_ 12774	1.748	ed	6444
1.27 A			_ 12774	Intro 1.790	ductory text of (b)	revised_ 6445
1.65			15510	Revis	ed	
1.84	:			Revis 1.901—	ed 1.991	14667
R 1.92			_ 15760	Subp	art F heading revis	sed 93 87
((e) revi	lsed	_ 6443	1.912 (e) 1	evised	12371
1.20 D			_ 8219	1.918	revised	6445
1.20	4			1.922		
1.20			_ 8219	Amei	nded nded	9387 13816
		lsed	_ 6443	1.926	revised	
1.20 A	.dded _		_ 8219	(c) 1	revised	
1.21	.1			1.955 (b) s	and (c) revised	2700
1.22			_ 8219	1.962		
		(d) revised			6) revised -1.1119	7822
1.29	1	, and (d) revised		Effec	tive date postponed	l 265
		ised		2	tive date	
1.29	2				nded	3703
1.29			_ 6444	2.106 Ame	nded	3703
R	evised		_ 6444	Ame	nded	3705
1.29 R	•		_ 6444		nded Isions waived in pa	
1.29	8			Ame	nded	11086
1.30		(b) revised	_ 6444		nded nded	
		ised	_ 6444	Ame	nded	
1.51 R	_		_ 12371	2.571 (b)	revised	10585
1.54	-	eted	10056	2.591		
•	u, aelf	71,041 I	しみぶつり	(91.)	CHVISEO	LUNKS

47	CFR	—Continued	29 F.R. Page	47 CFR—Continued 30 FJ	
2.60			•	0.318	ь~
		1		Added 883	37
		1		0.331	
		1	19090	(b) (2) deleted 867	78
2.60	-	d (b) amonded	E100	0.332	^^
		d (b) amended nended, (b) and (c) i		(i) added 270 Introductory text of (b) revised;	JG
``	vised		11456	(b) (5) deleted 67	70
(8	i) and	d (c) revised	19090	(j) and (k) added 1212	
5.55				0.371	
Ir		uctory text of (a) a		(a) revised 322	23
		revised	9388	0.417	
13.1		•	4554	(a) (5) amended 454	46
13.1		i	15761	Introductory text of (a) re-	
		ve date postponed	265	vised; (b) revised; (c), (d),	
		ve date		and (e) added 86'	
13.1		· · · · · · · · · · · · · · · · · · ·	2011	(a) (5) amended 1300	60
	-	ve date postponed	265	0.445 (c) amended 46	19
		ve date		(c) amended 46 (c) amended 77	
13.7				(b) revised 883	
R	evised	1	7463	(c) amended 149	
\mathbf{R}	evised	d	15761	1	
15.6				Policy statement 74:	19
		ended; (c) added	2558	Subpart H added 92'	71
15.7				Relaxation of reporting rules 988	
		revised	2558	Forms amended 1019	97
17.4		هـ د		Forms amended 121	25
(1	() ad	ded	13816	Appendix B amended 1480	บช
		10/5		1.80 (i) revised 970	67
		1965		1.106	٠.
47	CFR		30 F.R. Page	Amended 44	79
0.15	,			(a) revised 88	37
R	evise	d	6250	1.111	
0.16	;			Revised 88	38
R	evise	d	6250	1.525	
0.12	1			(b) (2) revised 67	80
		lded	8837	1.526	
0.21				Added 45	
-		lded	8678	(a) revised130 (a) revised135	41
0.21		d	0014		.Z.I
0.23		u	9314	1.569 Introductory text, (a)(1) and	
		ided	14100	(b) (1) revised	27
0.25		ided	14100	1.580	•••
	-	vised	8678	(f) (10) added45	47
0.28		1204	00.0	(c) and introductory text of (d)	_
		revised	8678	revised67	81
		ided		Introductory text of (f) revised_ 130	
		ided		1.594	
0.30		· - · ·		(d) (5) added 45	47
		eleted	8678		
		lded		revised67	81
0.30				1.615	
(g) ac	lded	12125	(f) added45	47
0.31	11			1.911	
(a) (13	added	7521	(e) added27	06

47 CFR—Continued	30 F.R. Page	47 CFR—Continued	30 F.R. Page
1.922		2.106—Continued	0-
Amended	2446	Corrected 14013,	14437
Amended		Revised	15701
Amended	6779	2.601	
Amended		Revised	7102
1.926		2.603	
(b) (9) revised	2446	(a) and (b) revised; (c) de-	
(b) (5), (6), (7) and (8) de-		leted	7102
leted		13.5	
1.927		(a) revised	3597
(a) revised	6779	13.14	000.
1.928		Deleted	14808
(b) revised	6779	13.15	1 1000
1.929	0110	Deleted	14909
Revised	6779	15.63	14000
1.962	0.15	(c) revised	6250
(a) (8) deleted	4479	15.75	0250
	4419		570
1.973 (d) revised	2002	(b) (1), (2), and (3) revised	5 78
	3223	15.211	
1.1103		Introductory text of (a) re-	
(d) revised	14806	vised; (a) (5) and (6) added;	
1.1104		introductory text of (b) re-	
Added	14806	vised	9316
1.1105		Effective date of (a) (6) stayed_	11354
Revised	14806	Effective date of (a)(6) further	
1.1111		stayed	15150
Revised	14806	18.112	
1.1113	11000	Revised	14200
Revised	14007	18.114	
1.1115	14807	Revised	7999
	14005	18.116	
(a) revised; (b) (9) added	14807	(b) and (c) revised	7999
1.1117		18.118	1000
Revised	14808	(a) revised	7999
1.1201—1.1251			1999
Subpart H added	9271	18.125	=000
2.1		Added	7999
Amended	7155	18.126	
Amended	8679	Added	7999
Amended	8839		
Amended	13370	1966	
2.100			1 F.R.
Revised	7156	47 CFR	Page
2.102		0.5	•
(a), (b) (4), (5), and (6) re-		(m) added	16714
vised		0.85	
2.104	1100	Center heading and section	
Revised	7156	added	16714
2.105	1130	0.91	10111
	F150	Revised	16315
(h) (1) revised	7156	0.92—0.99	10010
2.106		Deleted	16915
Amended			10919
Provisions waived in part		0.92	10015
Amended		Added	16312
4480, 6220, 6388, 7105, 7106	i, 71 56,	0.93	
8222, 8679.		Added	16315
Corrected 9883	, 11354	0.131	
Amended 12125	, 12294	Revised	6832
Corrected	12294	0.132	
Amended 13370	, 13706	Revised	6832

4/ CFR-Continued	31 F.R. Pa ge	47 CFR—Continued	31 F.R. Page
0.133—0.139 Deleted	6832	1.785 (a) revised	747
0.211		1.912	
(d) revised	11756	(d) revised	6113
0.243		1.918	
(d) added		(b) revised	. 14394
(e) added	4456	1.922	
0.251		Amended; incorrectly desig-	
(c) added		nated as 1.992	6113
(d) added(a) deleted		1.924	
(b) revised		(b) (2) (iv) deleted; (b) (2) (vi)	
0.281	10111	revised	. 2600
(aa) added	3074	1.926 (b) (2) added; (b) (9) revised	6113
(bb) added	4456	1.951	0113
(d)(16) added	4457	Revised	6832
(d)(17) added	6627	1.991	. 0032
(d)(8) revised	13474	Revised	2550
(cc) added	16567	1.1115	. 2000
0.289		(a) revised	6114
Center heading and section		2	. Ulla
added	16715	Subpart H added	3400
0.291	0074	2.1	0400
(i) added		Amended	14205
(j) added 0.303	4400	2.106	. 14000
(c) revised	12474	Amended6416-6417 (3 docs.)	7227
0.311	19414		. 16316
(a) (14) added	3074	2.601	, 10010
(a) (15) added		Revised	292
0.331		Revised	
(b) (18) added	10125	2.603	3000
(b) (4) revised	13474	(a) revised	292
(b) (16) revised	16316	Revised	
0.332		2.701	
(1) added		Subpart H added	3400
(m) added	4457	13	0.00
0.371		Appendix added	15
(a) (3) added	3074	13.2	
0.441 Revised	0110	(a) (1) (iii) added	12480
0.445	6112	(a) (1) (iii) revised	14591
(c) amended	10500	13.3	
1	10001	Revised	12480
Forms amended	746	13.4	
Forms amended		(a) revised; (d) added	12480
1.106	13230	13.5	
(k)(3) revised	2074	(d) added	12481
1.522	3014	13.11	
(a) revised	14204	(a) revised	. 15
1.526	エエリシエ	13.22	
(d) revised	14000	(d) through (g) redesignated as	}
1.550	1 4000	(e) through (h), respectively;	
(c) (1) revised	9798	new (d) added	
1.744	0170	13.25	
(a) revised	14394	Revised	12481
1.773	OOZ	13.61	
Revised	6868	(c) added	12481

		—Continued	31 F.R. Page	4/ Crk-Confinued	32 F.R. Page
	ntrod	uctory text revised	. 12481	0.211 (e) deleted	20860
15.				0.261	
			. 3400	(b) amended	20860
15.		added	7469	0.281	10262
		revised		(dd) through (jj) added (kk) added	
	211	Teviscu	. 13351	(ll) added	
		ve date of (a)(6) further	•	0.289	10010
•		ed		(c) added	3831
15.	220			(c) (10) added	6283
1	Added		3400	(c) (11) added	7835
18.				(b) revised	20860
1	Added		7822	0.291	
19				(k) added	5273
			_ 2721	0.311	
	735–20			(a) (9) revised	20860
		corrected	₋ 4834	0.314	00000
	735-40		11000	(e) revised	20800
,	c) re	vised	_ 11393	0.315 (b) revised	20260
				0.331	20000
		1 967		(b) (8) revised; (b) (19) added_	3831
47	CFF	•	82 F.R.	0.385	0001
_ `	CFI	•	Page	(a) revised	20860
0	Zuhna	rt C heading revised	10570	0.401—0.493	
0.1	_	to C Heading Teviseu	_ 10970	Subpart C heading revised	10570
		d	10580	0.401	
0.1		w	_ 10008	(b) revised	12795
		ignated from 0.2(a); re	_	(a) revised	
		d		0.403—0.431	
0.2		W	- 10008	Revoked	10570
		designated as 0.1 and re	_	0.403	
		d: (b) redesignated as 0.		Added	10570
		revised	_	0.405	
		d		Added	10570
0.3		w	_ 12100	0.406	
		d	10560	Added	10570
0.4		W	_ 10000	(b) (7) corrected	12180
		ignated from 0.2(b); re	-	0.411—0.420	
		d		Center heading and sections	40004
0.5		~ ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	_ 10000	added	10971
		d	10589	0.421—0.423 Center heading and sections	
ľ	(a) (12	and (b) (6) amended	13125		10579
0.6		, (3) (4)	_ 10120	added 0.422	10012
		r heading and section re		Amended	20260
		d		0.423	20000
0.9				(c) (15) and (16) corrected	19190
	Revise	ed	_ 20859	Introductory text of (a)	12100
	14			amended	13125
- • •	Amen	ded	_ 20859	0.431-0.434	-0480
	21			Center heading and sections	
		nended; (b), (c), (d), an	đ	added	10573
	(e)	revised	20859	0.441	
0.1	84			Redesignated as 0.481; amend-	
	Headi	ng revised: (c) amended	_ 20860	ed	10578

47 CFR—Continued	32 F.R. Page	47 CFR—Continued	32 F.R Page
0.441—0.467	_	2.602	•
Center heading and sections	.	Revised	10181
added	10573	2.603	
0.443—0.453	44	(a) and (b) revised; (d) added_	273
Redesignated as 0.483—0.493	10578	Revised5.63	10181
0.443 (c) revised	0000	Revised	
Amended		13.5	1129
0.455	10120	(c) (2) and (3) redesignated as	
(g) amended	13125	(c) (3) and (4) respectively;	
0.457	10110	(c) (2) added	9232
(g) revised	11213	13.11	
Headings of (a), (b), (c), (d),		(b)(1) revised	2627
(e), and (f) corrected	12180	15.63	
0.4710.475		(c) amended	6933
Center heading and sections		(c) revised	10854
added	10578	17.1	
0.481		Revised17.2	11268
Redesignated from 0.441;		(o) added	
amended 0.483—0.493	109.19	(a) deleted; (b) redesignated as	8813
Redesignated from 0.443—		(a) and revised; (c) through	
0.453		(n) deleted; (o) redesignated	
1.61	Page	as (b)	11260
(g) added	_	17.3	
Amended		Deleted	11268
1.70		17.4	
Added	2890	Revised	11268
1.202		17.7	
Revised	20861	Added	11269
1.412		17.8	
(a)(1) amended	20861	Added	8813
1.526		(a) revised	13591
(a) (1), (2), and (3) amended	20861	17.9	
1.545		Added	8813
Revised	2771	17.10	
1.570 (b)(3) amended	00001	Added	8813
1.572	20861	17.11 Deleted	
(a) (1) revised	11014	Deleted	11268
1.573	11214		
Amended	20061	Deleted	11268
1.912	40001	Deleted	
(e) revised	20261	17.14	11268
1.1209	20001	Revised	44555
(g) added	4358	17.15	11269
1.1227	1000	Deleted	11000
(b) revised	4358	17.16	11269
2.106	1000	= 1100	11269
Amended	924.	17.17	11209
2772, 3832, 4069, 6933,		Revised	11960
10300, 11991, 12916.	,	17.21	11409
2.561		Revised	11960
(b) revised	12795	17.22	11209
2.601		Revised	11269
Revised		17.24	
Revised		Revised	11269

47 CFR—Continued	32 F.R. Page	47 CFR—Continued	33 F.R. Page
17.25 (a) (1) and (2) revised	_ 11269	0.204 Added	8227
17.26 (a) (1) and (2) revised	11270	0.214 (b) and paragraph designation of (a) deleted	
17.27 (a) (1) and (2) revised 17.28	11270	0.215 (b) and paragraph designation	
(a) (1) and (2) revised	11270	of (a) deleted	
(a) (1) and (2) revised	11270	Revised 0.257	
(a) (1) and (2) revised		Deleted 0.261	
(a) (1) and (2) revised		(c) deleted 0.281	
(a) (1) and (2) revised		(ll) revised 0.287	
(a) (1) and (2) revised 17.34—17.45 Deleted		Revised 0.289 (b) revised	
17.34—17.58 Added		0.291 (a) revised	
19.735–201a Added		0.292 (b) revised	
19.735-202 Introductory text of (a) re-		Revised	
vised; (c) redesignated as 19.735-201a and revised; (d)	5	(a) and (b) revised; (g) added_0.298	15291
and (e) revised; (f) redesignated from 19.735-203(e)(1)		Heading revised; (c) added 0.305	
and revised 19.735–203		Deleted 0.307	
Heading revised; (e) (1) redesignated as 19.735–202(f) and	l	Revised	8228
revised 19.735-210		(a) (12) revised 0.313 Deleted	3637 8228
(d), (g), (h), (j), and (p) re- vised; (q) added	. 13 4 57	0.331 (b)(10) revised	
19.735–403 Revised	. 13458	(b) (20) added	10849
19.735–403a Added	. 13458	(n) added 0.337	5302
19.735-405 (c) revised	13458	Revised 0.357	8228
19.735–406 Revised	13458	Deleted 0.361	8228
19.735-410 Revised	13458	(h) deleted 0.371	8228
1968		(a)(1) revised; (a)(4) added Revised	7153 8228
47 CFR	33 F.R. Page	(e) added 0.381	19181
0.5 (b) (5) revised	•	Revised 0.383	8228
0.51 (c) revised		(d) deleted 0.453	8228
(a) revised 0.53		(a) (1) revised 0.461	5302
Revised	5302	(e) added	463

47 CFR—Continued	33 F.R. Page	47 CFR—Continued	33 F.R. Page
1.4 (g) revised	10571	1.1117 (a) amended	. 91:
1.14		2.1	
Added	. 11821	Amended	10856
(d) revised	7153	2.106 Amended	2041
1.221	4.00	3119, 3184, 4259, 6295, 9660,	10856
Center heading revised	. 463	12674, 14298	,
(b) (3) revised	9658	2.561 (b) revised	14165
Effective date of revision of		2.601	1410
(b) (3) stayed	15655	Revised	412
Added	463	Revised	10090
1.248	. 100	2.603 (a) and (b) revised	410
Redesignated from 1.251; (c) (2)		(a) and (b) revised	10090
revised; (c)(6)(ii) deleted 1.249	463	13	
Redesignated from 1.255(d);		Appendix amended	914
heading added	463	13.3	
1.250		Revised	913
Added	463	(c) revised	698
Center heading revised	463	13.5	
Redesignated as 1.248; (c) (2)		(a) revised	698
revised; (c)(6)(ii) deleted	463	13.8 Revised	
1.255		17.35	914
(d) redesignated as 1.249; head- ing added	463	(a)(3) corrected	7039
1.311—1.319	100	17.53	
Recodified as 1.311—1.325 and		Corrected	11540
revised; center heading re-		19.735-203 (c) revised; (d) deleted	11001
vised 1.311—1.325	403	(c) levised, (d) deleted	11071
Recodified from 1.311—1.319		1969	
and revised; center heading		47 CFR	34 F.R.
revised	463		Page
1.315 Revised	10571	0.12 Revised	14220
1.316	10011	0.13—0.20	14990
Revised	10571	Deleted	14330
1.323	10550	0.35	
Revised	10572	Revised	14374
(e) and (f) added	466	(a) amended	5656
1.362		(a) corrected	6480
Added	466	(a) amended	14690
1.502 (c) deleted	7491	(a) amended	18123
1.571	1491	(a) (3) added	14375
Amended	10344	0.281	
1.613		(r) revised	6524
(b)(3) revised	9083	(kk) revised 0.289	11144
1.615 (d) revised	9084	(c) (12) added	15416
1.922		0.403	
Amended	10345	Revised	9282
1.1115 (a) amended	4720	0.455 (b) (3) added	5040

4/ CFK—Confinued	34 F.R. Page	47 CFR—Confinued	4 F.R. Page
0.485		2.102	
(c) amended	752	(b) (4) through (7) redesig-	
(c) amended	18459	nated as (5) through (8),	
1.61		respectively; new (b) (4)	
(c), (d), and (e) revised	6481	added1	2138
1.227		2.106	
(b) (1) revised; (b) (5) added	7966	Amended	1542,
1.516		5379, 5724, 13542, 1	
(e) added	7966	Revised1	
1 571		Amended 2	20199
(c) amended	7966	2.301, 2.302	
Amended	14376	Subpart D recodified as 2.301—	
1.580		2.303; revised	5104
Heading revised; (c) amended;		2.301—2.303	
introductory texts of (d) and		Subpart D recodified from 2.301,	
(f) revised; (f) (9) revoked;		2.302; revised	5104
(i) revised; (m) added	7966	2.601	
1.591		Revised	556
(b) amended	7967	Revised	11302
(b) amended	14469	2.603	
1.597		(a) and (b) revised	556
(e) and (f) added	5103	(a) and (b) revised	11302
1.613		5.52	
(a) amended	5949	(b) and (c) revised	3802
1.785		5.55	
(a) (6) added	12137	(a) revised; (b) deleted	3802
1.911		5.56	
(a) and (b) revised	19419	(b) (12) revised	3802
1.922		5.161	
Amended		Revised	3802
Amended	19419	5.163	
1.923		(c) revised	3802
Revised	9283	15.63	
1.926		(c) revised	7497
(b) (1) revised; (b) (4) added	18802	15.201	
1.953		(d) added	17172
(b) (3) deleted	19419	15.214	
1.1111		Added	17172
(a) amended	14375	15.251	
1.1115		Added	17172
(a) revised	9283	15.252	
1.1203		Added	17179
(a) (5) revised	8242	15.253	11112
1.1207	0010	Added	17179
(a) revised	8243	15.254	11112
2		Added	17179
Subpart D recodified and re-			11117
vised	5104	17.4	6401
2.1	-0-0		6481
Amended		19.735–206	1700
Amended	5656	Revised	1722